

articulate and clear in ^{Speech)} The Phonology of Mongolian

joints. [Lati

Jan-Olof Svantesson, Anna Tsendina, Anastasia Karlsson, and Vivan Franzén

THE PHONOLOGY OF THE WORLD'S LANGUAGES

The Phonology of Mongolian

THE PHONOLOGY OF THE WORLD'S LANGUAGES

General editor: Jacques Durand

Each volume in this series offers an extensive treatment of the phonology of one language within a modern theoretical perspective, and provides comprehensive references to recent and more classical studies of the language.

Published in the series: The Phonology of Dutch Geert Booij The Phonology of Standard Chinese San Duanmu The Phonology of English

Michael Hammond

The Phonology of Norwegian Gjert Kristoffersen

The Phonology of Portuguese Maria Helena Mateus and Ernesto d'Andrade

The Phonology and Morphology of Kimatuumbi David Odden

> The Lexical Phonology of Slovak Jerzy Rubach

The Phonology of Hungarian Péter Siptár and Miklós Törkenczy

The Phonology of Mongolian Jan-Olof Svantesson, Anna Tsendina, Anastasia Mukhanova Karlsson, and Vivan Franzén

> The Phonology of Armenian Bert Vaux

The Phonology and Morphology of Arabic Janet C. E. Watson

> The Phonology of German Richard Wiese

In preparation:

The Phonology of Danish
Hans BasbøllThe Phonology of Spanish
Iggy RocaThe Phonology of Tamil
Prathima ChristdasThe Phonology of
CatalanThe Phonology of Polish
Edmund GussmanMax Wheeler

THE PHONOLOGY OF MONGOLIAN

Jan-Olof Svantesson, Anna Tsendina, Anastasia Mukhanova Karlsson, and Vivan Franzén



OXFORD UNIVERSITY PRESS

Great Clarendon Street, Oxford 0x2 6DP

Oxford University Press is a department of the University of Oxford. It furthers the University's objective of excellence in research, scholarship, and education by publishing worldwide in

Oxford New York

Auckland Cape Town Dar es Salaam Hong Kong Karachi Kuala Lumpur Madrid Melbourne Mexico City Nairobi New Delhi Shanghai Taipei Toronto

With offices in

Argentina Austria Brazil Chile Czech Republic France Greece Guatemala Hungary Italy Japan South Korea Poland Portugal Singapore Switzerland Thailand Turkey Ukraine Vietnam

> Published in the United States by Oxford University Press Inc., New York © I.-O. Svantesson 2005

The moral rights of the author have been asserted Database right Oxford University Press (maker)

First published 2005

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted, in any form or by any means, without the prior permission in writing of Oxford University Press, or as expressly permitted by law, or under terms agreed with the appropriate reprographics rights organizations. Enquiries concerning reproduction outside the scope of the above should be sent to the Rights Department, Oxford University Press, at the address above

You must not circulate this book in any other binding or cover and you must impose the same condition on any acquirer

A catalogue record for this title is available from the British Library

Library of Congress Cataloging in Publication Data (Data available) ISBN 0199260176

1 3 5 7 9 10 8 6 4 2

Typeset by Peter Kahrel Ltd., Lancaster Printed and bound in Great Britain on acid-free paper by Biddles Ltd., www.biddles.co.uk

CONTENTS

Ack	knowl	edgements	ix
Lis	t of Ta	ubles	х
Lis	t of Fi	igures	xi
Abi	brevia	ations and symbols	xii
Tra	nscri	ption	XV
Int	roduc	tion	xvi
1.	vov	WELS	1
	1.1	Acoustic properties of monophthongs	1
		1.1.1 The phonetic basis for vowel harmony	7
		1.1.2 The vowel /i/	8
	1.2	Diphthongs	9
	1.3	Palatalized vowels	10
2.	CO	12	
	2.1	Stops and affricates	12
		2.1.1 Stops in other Mongolian dialects	17
		2.1.2 Buriad and Kalmuck stops	17
	2.2	Fricatives	18
	2.3	Nasals	18
	2.4	Liquids	19
	2.5	Glides	20
	2.6	Palatalized consonants	20
3.	PH	ONEMES	22
	3.1	Vowel phonemes	22
	5.1	3.1.1 Non-initial vowels	23
	3.2	Consonant phonemes	25
		3.2.1 Palatalized consonants	28
		3.2.2 Velar and uvular consonants	28
		3.2.3 Labials	29
	3.3	Loan-word phonology	30
4.	WR	ITING SYSTEMS	34
	4.1	Cyrillic Mongolian	34
		Cyrillic Buriad and Kalmuck	40
	4.3	Modern Written Mongolian	40

5.	PHO	ONOLOGICAL PROCESSES	43				
	5.1	Phonological representation	43				
	5.2	Vowel harmony	46				
		5.2.1 Transparent i	48				
		5.2.2 Opaque vowels	50				
		5.2.3 The domain of vowel harmony	52				
		5.2.4 Vowel harmony as feature spreading	53				
		5.2.5 Velar ~ uvular alternation and epenthetic consonants	55				
		5.2.6 Cyclic vowel harmony	56				
	5.3	Velar nasal assimilation	57				
	5.4	Reduplication	58				
		5.4.1 Adjective reduplication	58				
		5.4.2 Noun reduplication	59				
6.	SYI	LLABIFICATION AND EPENTHESIS	62				
	6.1	Syllable structure	62				
		6.1.1 Syllable types	63				
		6.1.2 Codas, sonority, and epenthesis	65				
	6.2		68				
	6.3	3 Schwa ~ zero alternation					
	6.4	Cyclic syllabification	73				
		6.4.1 The verb suffix $-(\partial)x$	75				
	6.5	The phonemic status of palatal glides	76				
	6.6	Word structure	78				
	6.7	Final consonant combinations	79				
7.	PRO	DSODY	85				
	7.1	Focal accent	85				
	7.2	Final prominence tone	90				
	7.3	Boundary signalling	91				
	7.4	Questions	93				
		Word stress	94				
8.	OL	D MONGOLIAN	98				
		Uigur Mongolian	99				
	8.2	Sino-Mongolian	102				
	0.2	8.2.1 Sino-Mongolian consonants	102				
		8.2.2 Sino-Mongolian vowels	106				
	8.3	Arabic Mongolian	107				
	8.4	'Phags-pa Mongolian	108				
	8.5	Old Mongolian vowels	111				
		8.5.1 Primary long vowels	113				

vi

		CONTENTS	vii
	8.6	Vowels in non-initial syllables and vowel harmony	113
		8.6.1 Rounding assimilation	114
		8.6.2 Alternation of velar and uvular consonants	115
		8.6.3 Neutral $*i$	116
		8.6.4 Proto-Mongolic * <i>u</i> ?	117
	8.7	Old Mongolian consonants	118
		8.7.1 Stops and affricates	119
		8.7.2 The fricative h	121
		8.7.3 Sonorants and $*s$	124
		8.7.4 Coda consonants	124
	8.8	Morphophonological processes	125
	8.9	Old Mongolian vocabulary	126
9.	THE	MONGOLIC LANGUAGES	140
	9.1	Mongolian	141
	9.2	Buriad	144
	9.3	Kamnigan	146
	9.4	Oirad	147
	9.5	Dagur	149
	9.6	Shira Yugur	150
	9.7	Monguor	151
	9.8	Santa	152
	9.9	Bonan	153
	9.10	Kangjia	153
	9.11	Moghol	154
		Other related languages	154
	9.13	Comparative vocabulary	155
10.		ELOPMENT OF THE MODERN MONGOLIC	
	LAN	IGUAGES	178
	10.1	The Mongolic vowel shifts	178
	10.2	Vowel splits and mergers	182
	10.3	Long vowels	183
	10.4	*i-diphthongs	185
	10.5	Vowel deletion and reduction	185
		10.5.1 Reduction of short non-initial vowels	186
		10.5.2 Deletion of initial vowels	188
	10.6	Non-initial vowels and vowel harmony	188
		10.6.1 Vowel harmony shifts	189
		10.6.2 Vowel harmony in Old Mongolian and Halh	190
	10.7	Vowel assimilation	194
		10.7.1 Progressive rounding assimilation and	
		the development of rounding harmony	194

CONTENTS

	10.7.2	Regressive rounding assimilation	194
	10.7.3	Regressive * <i>i</i> -assimilation	195
	10.7.4	Breaking of word-initial *i	196
	10.7.5	Breaking of * <i>u</i> in Dagur	197
10.8	Onset c	onsonants	197
	10.8.1	Stops	198
	10.8.2	Affricates	200
	10.8.3	Fricatives	202
	10.8.4	Sonorants	203
10.9	Coda co	onsonants	203
	10.9.1	Unstable *n	205
10.10	Deaspir	ration and related processes	205
10.11	Palatali	zation	208
	10.11.1	Consonant palatalization	208
	10.11.2	Vowel palatalization	210
	10.11.3	Vowel palatalization and vowel harmony	212
10.12	Syllable	e and word structure	214
10.13	Conclu	sion	216
Appendices			218
References			230
Index			299
Index of Old	d Mongo	olian words	311

ACKNOWLEDGEMENTS

We thank all those who helped us in various ways, especially the Mongolian speakers Basbajar, Davaadorž, Hürèlbaatar, Sarantujaa, Sarancacral, and several others, who willingly let us do rather lengthy and sometimes boring tape recordings of their language. Several teachers and students at the Mongolian State University in Ulaanbaatar helped us with arranging recordings and in other ways, including L. Lhagva, B. Sarantujaa, Ju. Mönh-Amgalan, François Jacquesson, and Judith Bertalan. We were helped with the acquisition of literature and visits to different libraries by Tamara Esenova and Petr Bitkeev (Elista), Veronica Veit (Bonn), Juha Janhunen (Helsinki), Saitô Yoshio and Sakaedani Haruko (Tokyo), Kuribayashi Hitoshi (Sendai), Coyijongjab, Kögjiltü, and Köke (Kökeqota), Secencogtu (Peking), Tat'jana Skrynnikova (Ulan-Ude), and several others. Special thanks to James Bosson who supported this project from its start and read parts of the manuscript at various stages, and to Saitô Yoshio who read the entire manuscript and gave many valuable comments, as did the series editor Jacques Durand and one anonymous reviewer.

Part of the project was financed by the Swedish Research Council for the Humanities and Social Sciences (HSFR) and by the Swedish Institute. Travel grants to Mongolia and other parts of the world were given also by Vetenskapssocieteten i Lund, Kungl. Humanistiska Vetenskapssamfundet i Lund, the Elisabeth Rausing Memorial Foundation, and the Ido Foundation.

LIST OF TABLES

1.1	Formant frequencies for monophthongs	5
3.1	Examples of consonants in different positions	26
3.2	Examples of Russian loan-words in Mongolian	31
4.1	Cyrillic alphabets for Mongolic languages	35
4.2	The Modern Written Mongolian alphabet	41
5.1	Examples of vowel harmony in suffixes	47
5.2	Examples of vowel harmony in uninflected words	49
5.3	Examples of noun reduplication	60
6.1	Syllable types	64
6.2	Two-consonant codas	66
8.1	Sino-Mongolian transcriptions	103
8.2	Examples of initial and non-initial vowels	112
8.3	Examples of Old Mongolian onset consonants	119
9.1	Transcription changes	156

LIST OF FIGURES

0	Research on Mongolian phonology	xviii
1.1	Vowel duration	3
1.2	Duration of long and short initial vowels	4
1.3	Formant frequencies for short initial vowels	4
1.4	Formant frequencies for full non-initial vowels	6
1.5	Formant frequencies for reduced non-initial vowels	7
1.6	Non-initial /i/	8
1.7	Diphthongs	10
1.8	Long palatalized vowels vs. <i>i</i> -diphthongs	11
1.9	[Vi] diphthongs vs. [Vj] combinations	11
2.1	Timing of dental stops	14
2.2	Aspirated [t ^h] and unaspirated [t] in word-initial position	15
2.3	Preaspirated [ht] and unaspirated [t] in word-final position	16
2.4	Preaspirated [ht] and unaspirated [t] in word-medial position	16
2.5	Preaspirated [ht] and unaspirated [t] after a sonorant	16
2.6	Comparison of Buriad, Kalmuck, and Mongolian aspiration	18
2.7	Aspiration of the fricative /s/	19
2.8	The nasals /n/ vs. /ŋ/	19
2.9	The laterals /ʒ/ vs. /ɬ/	20
2.10	Plain /Է/ vs. pharyngealized /Է ^j /	21
2.11	Consonant palatalization after <i>i</i> -diphthongs	21
7.1	Contrasting focus in a short sentence	86
7.2	Realization of the focal accent	87
7.3	Contrasting focus in a long sentence	88
7.4	Contrasting focus marked lexically and by intonation	89
7.5	The final prominence tone L	- 90
7.6	The phrase-initial boundary tone [LH	91
7.7	The phrase-final boundary tone LH]	92
7.8	Intonation in a question and a declarative	94
7.9	Yes-no questions	95
7.10	Final boundary tone in a question	95
10.1	The Mongolic vowel shift illustrated synchronically	179
10.2	Mongolic vowel shifts	181
10.3	Vowel palatalization	212
10.4	Number of syllables and speech sounds per word in Old Mongolian	
	and Halh	215

ABBREVIATIONS AND SYMBOLS

Languages and dialects

- AM Arabic Mongolian
- Baa Baarin Mongolian
- Bon Bonan
- Bur Buriad
- Ch Chinese
- Cha Chahar Mongolian
- Dag Dagur
- Hlh Halh Mongolian
- KJ Kangjia
- Klm Kalmuck
- Kmn Kamnigan
- Mgr Monguor
- Mo Mongolian
- Mog Moghol
- Oir Oirad
- OM Old Mongolian
- Ph 'Phags-pa Mongolian
- Ru Russian
- San Santa
- ShY Shira Yugur
- SM Sino-Mongolian
- Tib Tibetan
- UM Uigur Mongolian
- WM (Modern) Written Mongolian
- YCh Yuan dynasty Chinese

Grammatical forms (Halh and/or Old Mongolian forms of the suffixes are given)

ablative (-*Es*: *- $A\check{c}^{h}A$) ABL accusative (-(i)g; *-i)ACC agent nominalizer $(-\check{c}^h)$ AGNR adjectivizer (various suffixes) AR causative (-Uk; *-kA)CAUS collective $(-c^h GE, -U\xi)$ COLL comitative $(-t^h Ei)$ COM conditional (-pk; *-pAsU)COND cooperative $(-kc^h)$ COOP

CV	converb $(-\eta; *-n)$
DAT	dative $(-t; *-A)$
DER	derivational suffix
DPST	direct past $(-\xi E)$
FUTP	future participle $(-(\partial)x; *-k^{h}U, *-k^{h}Un, *-k^{h}Ui)$
GEN	genitive $(-i\eta; *-Un, *-U)$
IMP	imperative $(-g; *-t^h Uk^h Ai)$
INST	instrumental (- <i>Er</i> ; *- <i>ijAr</i> ; *- <i>pAr</i>)
IPFG	imperfect gerund $(-\check{c}; *-\check{c}U)$
NPST	nonpast (– <i>n</i>)
NR	nominalizer (various suffixes)
PFG	perfect gerund (– <i>Et</i>)
PL	plural $(-\check{c}^{h}Ut, -t; *-Ut, *-s, *-t)$
PST	past $(-w; *-pA(i))$
PSTP	past participle (- <i>sŋ</i> ; *- <i>ksAn</i>)
Q	question (wee, pee, $-(j)U$)
RFL	reflexive $(-E; *-ijAn, *-pAn)$
TERM	terminal $(-t^h \mathcal{B})$
VR	verbalizer (various suffixes)

Other abbreviations

А	* <i>a</i> /* <i>e</i> (10.6.2)
[ATR]	[advanced tongue root] (5.1)
С	consonant (5.1)
Е	elalolə (5.2)
[F]	[pharyngeal] (5.1)
G	g/G (5.2.6)
Н	high tone (Chapter 7)
L	low tone (Chapter 7)
0	*o/*ø (10.6.2)
[O]	[open] (5.1)
[P]	[palatal] (5.1)
[R]	[round] (5.1)
[RTR]	[retracted tongue root]
U	u/v or *u/*y (5.2; 10.6.2)
V	vowel (5.1)
[V]	[velar] (5.1)

Symbols

- . syllable boundary
- graphic boundary in Written Mongolian (4.3; 8.1)
- morpheme boundary
- morpheme and graphic boundary (8.1)

ABBREVIATIONS AND SYMBOLS

- = boundary between phonological words in compounds
- [initial boundary tone (7.3)
-] final boundary tone (7.3)
- /.../ phonological representation
- [...] phonetic form
- <...> transliteration
- /n unstable *n* (10.9.1)

TRANSCRIPTION

Halh (Khalkha) Mongolian forms are usually given in a quasi-phonemic transcription where epenthetic vowels and vowel harmony alternants are shown, for example, $m_{2yG2}\xi^{h}vt$ монголчууд 'Mongols'. The Cyrillic Mongolian spelling is also shown. Phonological representations are given within slashes, /mɔŋgb– č^hUt/, and when phonetic details are discussed, a narrower phonetic transcription in square brackets, like [mɔ̃Ngöłčot], may be given.

IPA symbols are used with the exception that affricates are written with single symbols: $c, \check{c}, \check{j} = [ts, tJ, d_3]$; and further, $\check{s}, \check{z} = [J, 3]$. Although the letters [a] and [a] denote different sounds in the IPA alphabet, this vowel quality difference is not used in any Mongolic language, and we use *a* as the italic form of the letter a, with no sound difference intended.

Words from other Mongolic dialects and languages are taken from published sources. For ease of comparison the phonetic symbols have been normalized to fit in with our transcription of Halh (see Chapter 9 for details). Words from the different written sources for Old Mongolian are given in a strict transliteration of the respective writing system (see Chapter 8).

In the bibliography, and in proper names mentioned in the text, conventional transcriptions are used. The Cyrillic alphabets for Russian, Mongolian, Kalmuck, and Buriad are transliterated as in Table 4.1. The obsolete Russian letters b and i are transliterated \check{e} and i. The transcription systems used for Chinese, Japanese, and Korean are *pinyin*, Hepburn, and McCune-Reischauer. For Written Mongolian, the Mostaert transcription is used with some modifications (see Table 4.2, p. 41).

For writing Mongolian, Chinese, Japanese, Korean, and Russian personal names with the Latin alphabet, we have used the transcriptions for these languages used elsewhere in the bibliography, although these are not always the forms preferred by the authors. We apologize for any inconvenience caused by this. We use the order: family name–personal name, for all Chinese, Japanese, and Korean names even in running text.

INTRODUCTION*

In this book the phonology of Mongolian is presented from two points of view, synchronic and diachronic. In Chapters 1–7, basic instrumental phonetic data on Standard Mongolian, the Halh (Khalkha) dialect as spoken in Ulaanbaatar, the capital of the Republic of Mongolia, is presented, and the phonology is analysed. The first two chapters give acoustic data on vowels and consonants, respectively. In Chapter 3, a phoneme analysis is given, and in Chapter 4, the two writing systems used for Mongolian are presented and related to the phoneme analysis. In Chapter 5, (morpho)phonological processes are presented, in particular vowel harmony, which is unusually regular in Mongolian. In Chapter 6, syllable structure and syllabification is treated together with epenthesis of schwa vowels, which is shown to depend on syllabification. Finally, intonation and stress is treated in Chapter 7, again based on acoustic measurements. In this part of the book, the term Mongolian normally refers to Ulaanbaatar Halh Mongolian.

Halh is a dialect of the Mongolian language, which forms the Mongolic language group (or language family) together with ten rather closely related languages. From Chapters 8–10, we give an overview of the Mongolic languages and their historical development. In Chapter 8, we reconstruct Old Mongolian, the immediate ancestor of the languages of the oldest Mongolian written sources. In Chapter 9, we give short sketches of the eleven modern Mongolic languages, and in the final chapter we describe the phonological processes that shaped the modern languages from Old Mongolian. In this way we try to give a unified description of the main phonological features of the modern languages, using a historical point of view, without going into the details of each language. We believe that this approach, which is the traditional one in Mongolian studies, adds insights, and gives a background, to the strictly synchronic analysis in previous chapters. Many synchronic phonological phenomena in Halh Mongolian, for example, pharyngeal vowel harmony and its relation to the distribution of palatalized consonants, can be understood better if the historical background is kept in mind. Furthermore, the Mongolic languages are close enough to make a simultaneous diachronic description of them meaningful. We focus on the phonological processes that shaped the languages from their common ancestor rather than on separate developments in

^{*} Several dictionaries were useful for finding examples, in particular the Mongolian–English dictionary by Bawden (1997), the two Mongolian–Chinese dictionaries *Monggol Kitad toli* (1999) and *Xin Meng–Han cidian* (1999), and the Academic Mongolian–Russian dictionary (Bajarsajhan et al. 2001– 2). For some purposes, the reverse alphabetic dictionary by Vietze and Zenker (1976) was very valuable. The official orthographic and morphological dictionary by Damdinsürèn and Osor (1983) was used as a norm for the Cyrillic spelling. All examples taken from dictionaries were checked by Tsendina, and in doubtful cases other native speakers of Ulaanbaatar Halh were consulted as well.

the individual languages. These processes, described in Chapter 10, taken together with the description of Old Mongolian in Chapter 8 give a rather detailed picture of the phonologies of the different modern Mongolic languages, with the detailed synchronic description of Hallh in the early part of the book as a kind of standard of comparison.

The Mongolic language group is usually regarded as one of the three branches of the Altaic language family. The others are the Turkic and Tungusic languages. Some scholars are convinced that Korean, Japanese, and even Ainu also belong to the Altaic family, while others are equally convinced that Altaic is not a valid genetic grouping, so that Mongolic, Tungusic, and Turkic are separate language families, whose structural and lexical similarities are due to areal influence. Authors who are in favour of the Altaic family include most Mongolists, John Ramstedt (1957) and Nicholas Poppe (1965) among others. The existence of Altaic as a valid genetic grouping is refuted by authors such as Gerard Clauson (1956) and Gerhard Doerfer (1966). There have also been some attempts to connect Mongolian with the Dravidian languages (see e.g. Vacek (1996)). However, the possible relations of Mongolic to other language groups will not be treated here.

The first scientific study of the phonology of a Mongolic language was the description of Halh by the Finland-Swedish scholar, John Ramstedt, published in 1902. In accordance with the then current linguistic practice, he analysed Halh phonology from an historical point of view, comparing it with Classical Written Mongolian. His work became highly influential, especially in Russia, and was the model for the phonological works of Boris Vladimircov, Nikolaj Poppe, and several others.

The first instrumental phonetic investigation of a Mongolic language was made in 1915-16 by the Helsinki phonetician, Jean Poirot, who measured Kalmuck segment durations. The results were not published until 1935, in Ramstedt's Kalmuck dictionary (pp. xxi-xxvi). The Japanese Obata Jûichi and Teshima Takehiko (1934; 1935) measured formant frequencies for Mongolian vowels and the duration of some consonants. More substantial investigations were not made until the 1950s in China and the Soviet Union. Ignatij Buraev and Valentin Zolhoev in Buriatia, Dordži Pavlov and Petr Bitkeev in Kalmuckia, as well as Sürèngijn Möömöö in Mongolia made pioneering phonetic work on their respective languages. Younger researchers, such as Tamara Esenova in Kalmuckia, have continued, but unfortunately the economic situation in Russia and Mongolia has now almost put an end to instrumental phonetic work in these countries. The most dynamic and interesting work on Mongolian phonetics is no doubt being done by Mongol scholars in China. Instrumental phonetic investigations had already been made in the 1950s by Cenggeltei, Coyijongjab, and others, and they are now continued by scholars like Köke and Bayarmendü. Recently, natural language processing of Mongolic languages has been undertaken by the researchers Köke and Dawa, both from China, and by Ljubov' Radnaeva from Buriatia.

An impressive number of descriptions of dialects and of lesser-known Mongolic languages have been made in the countries where they are spoken. In addition to

INTRODUCTION

those already mentioned, the Kalmuck scholar Buljaš Todaeva, Valentin Rassadin in Buriatia, and Žavzangijn Coloo in Mongolia made important contributions to this field. Again, this work is very strong in the Mongolic-speaking parts of China, where detailed descriptions of all indigenous Mongolic languages have been published by Böke, Engkebatu, Chen Naixiong, and several others. Phonological analyses of Halh Mongolian were made by Šadavyn Luvsanvandan in Mongolia.

Comparatively little phonetic and dialectological work has been done by researchers from outside the Mongolic-speaking countries, where the tradition has been more historical and philological. The dominating scholar has undoubtedly been Nikolaj Nikolaevič (Nicholas) Poppe, who was first working in the Soviet Union but emigrated to the United States in 1949. He published extensively in virtually all fields of Mongolian studies, in particular historical linguistics. Other scholars with a broad and varied production are the Japanese Nomura Masayoshi and Hattori Shirô. The Japanese tradition is continued by mongolists such as Jôo Hakutarô, Kuribayashi Hitoshi, and Saitô Yoshio.

Mongolic historical phonology in the tradition of Ramstedt, Vladimircov, and Poppe was revived recently by the increasing amount of data becoming available from the lesser-known Mongolic languages. The most important work is done in China and Mongolia by scholars as Jagunasutu, Kögjiltü, and Tömörtogoo.

As this sketchy and subjective survey shows, the bulk of the research on Mongolic phonology is being done by scholars in the Mongolic-speaking countries, in spite of difficulties created by economic factors and changing political situations. Unfortunately, their research is often published in journals and books which, in addition to the language barriers, are difficult to access in libraries outside these countries. Although it has not been possible to describe all results and discussions in this literature, we feel that it deserves to be better known among Western



FIGURE 0. Research on Mongolian phonology

scholars than is now the case, and we have tried to cite the relevant publications in the proper places.

Although there are several specialized bibliographies on Mongolian studies, the most important bibliographical tool for Mongolian linguistics is the annual *Linguistic bibliography* (*Bibliographie linguistique*) (1939–) which indexes most publications about Mongolian linguistics even from China and Mongolia. Recent issues can be accessed on the Internet, at http://www.kb.nl/kb/blonline/.

The state of Mongolian studies (or more precisely, Mongolian phonology) in different parts of the world is illustrated in Figure 0, where the number of publications appearing in the bibliography of this book (except works which do not deal specifically with Mongolian) are shown broken down by country of origin (of the main author) and year of publication. Several interesting observations can be made from this graph, in particular the decline of Russian (and Mongolian) science in the 1990s, and the emergence of China as the most important country for Mongolian studies in the 1980s and 1990s. This page intentionally left blank

1

VOWELS

In this chapter we present briefly the results of our phonetic analyses of the Halh Mongolian vowels. The consonants are treated in the following chapter, and a phoneme analysis is given in Chapter 3.

The acoustic phonetic investigation of Halh in Chapters 1 and 2 is based mainly on recordings made by one of the authors (JOS) in Ulaanbaatar in May 1990, using a portable cassette recorder of relatively high quality (Sony WM D6C). The main recordings were made by three male speakers of Halh, Basbajar (BB), Davaadorž (DD), and Hürèlbaatar (HB). They were born, and had grown up, in Ulaanbaatar, and were living there at the time of the recording. Their ages were twentyone, twenty-six, and thirty-six years, respectively. A list of words and sentences illustrating various phonetic phenomena was recorded. Some of the words were read in isolation, and some in a focused position in a carrier sentence. Each word was read three to five times by each speaker. The recordings were analysed in the *ESPS/Waves+* environment on Sun workstations at the Department of Linguistics, Lund University, or using the *Praat* speech analysis program written by Paul Boersma and David Weenink (University of Amsterdam). Some of the figures were made with the *WaveSurfer* analysis program written by Kåre Sjölander and Jonas Beskow (Royal Institute of Technology (KTH), Stockholm).

1.1 ACOUSTIC PROPERTIES OF MONOPHTHONGS¹

There are seven long vowels [i1, e1, a1, u1, 01, 01, 01, 01] and six short vowels [i, a, u, 0, 0, 0] in initial syllables (see below for the vowel transcribed [i]). In non-initial syllables, there are full vowels [i, i, e, a, u, 0, 0, 0] as well as phonetically reduced vowels which we will write as [ĭ, ĕ, ă, ŭ, ŏ, ŏ, 5] in this chapter. In addition to the monophthongs, there are also diphthongs, to be treated in section 1.2.

A list of disyllabic words (1) illustrating each combination of monophthongic vowels in the first and second syllable was recorded by the three Ulaanbaatar speakers. The material does not contain palatalized or alveopalatal consonants which may affect the vowel quality. The words were read in the sentence frame [pi: __gisəŋ] $\delta \mu$ ___ rэсэ μ 'I said __'. Two recordings of each word from each speaker were analysed. The duration and the first three vowel formants were measured from spectrograms and waveform displays, using the *ESPS/Waves*+ environment.

¹ For sources and notes on Acoustic properties of monophthongs, see App. B on p. 218.

```
VOWELS
```

(1)	Material for	• the acoustic	investigation	of monophthongs
----	---	--------------	----------------	---------------	-----------------

,		or the acou	suc investigation	п өј төпөрт	nongs	
	[ši��ig]	шилийг	'glass-ACC'	[piːrig]	бийрийг	'brush–ACC'
	[šilzer]	шилээр	'glass–inst'	[pi:rer]	бийрээр	'brush–inst'
	[pilzpur]	билбүүр	'hypocrite'	[ti:kuk]	дийлүүл	'to win–CAUS'
	[čʰi�ʒgĭr]	чилгэр	'sturdy'	[tiː��ǐ�ʰt]	дийлэлт	'victory'
	[ti͡big]	дэлийг	'mane-ACC'	[teːkig]	дээлийг	'robe–acc'
	[titer]	дэлээр	'mane–INST'	[teː]	дээлээр	'robe–INST'
	[titgur]	дэлгүүр	'shop'	[te:gur]	дээгүүр	'over'
	[tiʒgĭr]	дэлгэр	'wide'	[ceitspĕr]	зээлбэр	'loan-word'
	[tabig]	далыг	'seventy– ACC'	[pa:big]	баалыг	'lap-dog-ACC'
	[talzar]	далаар	'seventy– INST'	[paːɟar]	баалаар	ʻlap-dog– inst'
	[tabob]	далуул	'seventy– COLL'	[carrotz]	зааруул	'to dislocate'
	[cară]z]	зарла	'to announce'	[saːră]ʒ]	саарал	'grey'
	[suţtig]	сүлдийг	'emblem–	[su:ţig]	сүүлийг	'tail-ACC'
	00		ACC'	00		
	[sugter]	сүлдээр	'emblem– INST'	[suzger]	сүүлээр	'tail–inst'
	[pulʒur]	бүлүүр	'paddle'	[tu:gur]	дүүгүүр	'sling'
	[purtŭ]	бүрдэл	'range'	[tu:ţgŭr]	дүүлгэр	'frisky'
	[subig]	сулыг	'loose-ACC'	[currig]	зуурыг	'moment-
	-			_	_	ACC'
	[subar]	сулаар	'loose-inst'	[cuirar]	зуураар	'moment– INST'
	[curguz]	зургуул	'six-coll'	[curub]	зууруул	'ink-slab'
	[curŭ�aht]	зуралт	'pencil lead'	[tuː�əŭb]	дуулал	'singing'
	[tokig]	дөлийг	'flame-ACC'	[po:rig]	бөөрийг	'kidney-ACC'
	[torwutz]	дөрвүүл	'four-COLL'	[coːɟruɟ]	зөөлрүүл	'to soften'
	[tobor]	дөлөөр	'flame-INST'	[po:ror]	бөөрөөр	'kidney–INST'
	[soţŏr]	сөлөр	'cross-eyed'	[coː�brŏ�bʰt]	зөөлрөлт	'softening'
	[tɔɟɪg]	долыг	'experience- ACC'	[bɔːßı8]	боолыг	'slave–ACC'
	[t၁ԷսԷ]	долуул	'seven-coll'	[tə:gur]	доогуур	'below'
	[təkər]	долоор	'experience– INST'	[pɔːkər]	боолоор	'slave-inst'
	[pə�ər]	болор	'crystal'	[pɔːtŏ৳]	боодол	'bundle'

The mean durations for each vowel are shown in (2). The results are summarized in Figure 1.1, and they are illustrated for speaker DD in Figure 1.2.

These measurements show that the duration of a short vowel in the initial syllable is almost exactly half (on average 48 per cent) of the duration of a long vowel. The duration of a full non-initial vowel is, on average, 61 per cent of the duration

(2) Mean duration of monophthongs (ms)n = number of measurements on which each mean value is based

Long	Long initial vowels $(n = 8)$								
	[iɪ]	[er]	[ar]	[uː]	[ʊɪ]	[01]	[ɔː]		
BB	143	159	162	134	155	170	153		
DD	122	121	142	113	127	130	139		
HB	107	122	143	110	129	145	137		
Shor	t initic	ıl vow	els (n =	= 8)					
	[i]	[<u>i</u>]	[a]	[u]	[ʊ]	[o]	[၁]		
BB	73	84	89	74	69	74	85		
DD	50	63	82	55	59	60	78		
HB	55	52	78	46	56	50	57		
Non-	initial	vowe	ls						
	[i]	[1]	[e]	[a]	[u]	[ʊ]	[0]	[၁]	reduced
п	16	12	12	8	16	12	4	4	28
BB	92	79	104	101	109	103	90	79	53
DD	62	61	86	90	82	90	100	75	46
HB	64	56	88	86	93	75	66	83	40

of a long initial vowel, and reduced non-initial vowels have a duration which is 34 per cent of the duration of a long initial vowel. Although full vowels in non-initial syllables developed historically from long vowels (10.5), their duration in modern Halh is closer to the duration of a short vowel in the initial syllable than to the duration of a long vowel. This supports the analysis of full non-initial vowels as short in section 3.1.1. As seen in (2) and in Figure 1.2, the usual intrinsic duration relation obtains, so that open vowels have longer duration than closed vowels.

The first three formant frequencies of each vowel were measured at a point where the influence of the surrounding consonants was judged to be minimal, usually



FIGURE 1.1 Mean vowel duration (ms) for the three speakers. L1 = long initial vowel, S1 = short initial vowel, F2 = full non-initial vowel, R2 = reduced non-initial vowel.

VOWELS



FIGURE 1.2 Duration of long and short initial vowels (Speaker DD). The boxes contain 50% of the values, with the point inside showing the median. The upper and lower marks show the maxima and minima.

close to the middle of the vowel. The formant tracking facility of *Waves* was used. The results are shown in Table 1.1.

The qualities of corresponding long and short initial vowels are approximately the same, but the short ones are usually slightly centralized, as illustrated in Figure 1.3. The only major exception is [o], whose long and short versions have rather different qualities, although they function phonologically and historically as a long ~ short pair, and are written with the same Cyrillic letter ($\Theta \Theta$ and Θ). In a narrow phonetic transcription they could be written as [ot] and [Θ].



FIGURE 1.3 Mean formant frequencies (F1 and F2) for short initial vowels compared to long initial vowels (circled). Speaker DD.

IADI	с I.I	ronnan	nequencio	.5 101 1101	ophthongs	(incans of	<i>n</i> measure	ments)	
(a)]	Long	initial vov	vels $(n = 8)$)					
() -	0	[ir]	[er]	[aː]	[uː]	[ʊː]	[oː]	[31]	
BB	F1	291	368	631	323	395	360	544	
	F2	2,050	1,948	1,305	832	833	1,102	981	
	F3	2,852	2,670	2,443	2,502	2,635	2383	2,224	
DD	F1	349	385	806	357	433	395	604	
	F2	2,222	2,030	1,355	968	948	1,015	969	
	F3	2,907	2,869	2,707	2,488	2,712	2,518	2,412	
HB	F1	291	362	689	314	459	379	534	
	F2	2,238	2,184	1,448	937	928	1,006	1,064	
	F3	2,959	2,857	2,521	2,487	2,467	2,597	2,550	
(b)	Short		wels $(n = 8)$						
(-) .		[i]	[<u>i</u>]	[a]	[u]	[ʊ]	[o]	[၁]	
BB	F1	341	336	586	340	415	359	517	
	F2	1,929	1,970	1,290	1,062	974	1,367	1,004	
	F3	2,727	2,660	2,428	2,519	2,678	2,363	2,134	
DD	F1	363	358	792	370	489	366	565	
	F2	1,919	2,047	1,410	1,032	956	1,350	1,032	
	F3	2,844	2,862	2,596	2,578	2,712	2,407	2,424	
HB	F1	348	340	688	333	444	343	540	
	F2	2,061	2,161	1,509	1,195	1,073	1,603	1,095	
	F3	2,848	2,867	2,667	2,620	2,525	2,543	2,569	
(c) I		on-initial		<i>,</i>	,	,	<i>,</i>		
(0) 1		[i]	[1]	[e]	[a]	[u]	[ʊ]	[o]	[၁]
	n	16	12	12	8	16	12	4	4
BB	F1	328	379	371	586	337	429	384	542
22	F2	2,006	1,801	1,850	1,339	909	878	1,224	1,045
	F3	2,678	2,668	2,619	2,477	2,422	2,336	2,348	2,301
DD	F1	348	384	420	651	372	473	406	554
	F2	2,146	2,027	1,901	1,516	1,039	957	1,402	1,080
	F3	2,750	2,759	2,712	2,605	2,519	2,679	2,501	2,621
HB	F1	299	403	401	584	323	451	359	548
	F2	2,197	1,968	2,056	1,460	1,041	948	1,438	1,284
	F3	2,812	2,610	2,645	2,528	2,503	2,578	2,580	2,563
(d)		,	nitial vowe		_,	_,5 0 5	_,	_,	_,= ==
<i>(u)</i> .	licuu	i]	[ĕ]	[ă]	[ŭ]	[ŭ]	[ŏ]	[ŏ]	
	n	6	2	4	4	4	4	4	
BB	F1	356	376	538	353	402	351	485	
DD	F2	1,708	1,493	1,413	1,667	1,051	1,451	1,131	
	F3	2,409	2,430	2,397	2,434	2,632	2,453	2,797	
DD	F1	390	2,430 397	637	364	472	380	510	
	F2	1,932	1,703	1,547	1,732	1,088	1,455	1,171	
	F3	2,701	2,692	2,708	2,627	2,793	2,541	2,611	
HB	F1	355	318	2,700 529	334	493	427	503	
	F2	2,012	1,674	1,386	1,733	1,149	1,490	1,204	
	F3	2,012	2,428	2,508	2,516	2,453	2,559	2,477	
	15	2,330	2,720	2,500	2,510	2,-133	4,000	2,417	

TABLE 1.1 Formant frequencies for monophthongs (means of *n* measurements)

VOWELS

Auditory analysis as well as the intuition of native speakers suggests that the short vowels written with the Cyrillic letters n < i > and <math>3 < i > (usually the reflexes of Old Mongolian *i and *e) have merged to a vowel with the quality [i] in Ulaanbaatar Halh. These vowels are distinguished in Table 1.1b and Figure 1.3 as [i] and [i]. The formants of these vowels are close to each other. The simultaneous equality of F1 and F2 was tested with Mahalanobis' D2-test (see e.g. Rao 1965: 480), with the result that the difference was not significant for speakers BB and HB (F(2, 13) = 1.34 and 2.70; p > 0.05). The difference was significant (F = 13.60, p < 0.001) for Speaker DD, however, but the reflex of *e has higher F2 and thus a more [i]-like quality than the reflex of *i for this speaker. Our conclusion is that these vowels have merged to [i] in Ulaanbaatar speech; according to Möömöö and Mönh-Amgalan (1984: 82), this merger took place in Central Halh in general.

The qualities of non-initial full vowels are closer to the short initial vowels than to the long ones (see Table 1.1c and Figures 1.3 and 1.4). The quality of [o] is $[\Theta]$ -like in this case as well. The two non-initial vowels [i] and [I] are phonetically different but do not contrast, since the quality depends on the vowel of the preceding syllable (see 1.1.2).

Reduced vowels occur only in non-initial syllables, and their quality depends on the preceding full vowel. In this chapter we will write the reduced vowels with the same symbol as the preceding full vowel with a breve added, e.g. [xorŏb] xypaπ 'meeting'. As will be shown in Chapter 6, the reduced vowels are epenthetic (nonphonemic), and in our usual transcription we write them with the schwa symbol ϑ (so that the word 'meeting' is written xor ϑ). Although there is some individual variation, a reduced vowel is normally a centralized variant of the vowel of the preceding syllable (see Table 1.1d and Figure 1.5). The only major exception is [ŭ], which is more or less identical to [ě]. This is consistent with the Cyrillic Mongo-



FIGURE 1.4 Mean formant frequencies (F1 and F2) for full non-initial vowels compared to long initial vowels (circled). Speaker DD.



FIGURE 1.5 Mean formant frequencies (F1 and F2) for reduced non-initial vowels compared to long initial vowels (circled). Speaker DD.

lian script, which writes $9 < \delta >$ in both cases (and also after [i], but [ĭ] usually differs more from [ĕ] than [ŭ] does). On the other hand, the Cyrillic script implies that [ŏ] and [ă] are identical, both being written a <a>, but this is not confirmed by the acoustic data, which show that these two reduced vowels are quite different, being centralized forms of [υ] and [a], respectively. If the consonant immediately preceding the reduced vowel is an alveopalatal sibilant or a palatalized consonant, the reduced vowel is [i]-like (see 3.1.1).

1.1.1 The phonetic basis for vowel harmony²

Mongolian vowel harmony, whose phonological aspects are treated in section 5.2, divides the vowels into two classes [u, e, o] and [υ , a, υ]; the remaining vowel [i] is neutral. A non-compound word can contain vowels only from one of these classes. This causes the vowel pairs [u ~ υ], [e ~ a], and [o ~ υ] to alternate in suffixes. The phonetic characterization of the two vowel harmony classes is a bit controversial. Mongolian traditional grammar codified in works as *Jirüken-ü tolta-yin tayilburi* 'Commentary to *Auricle of the heart*' written by Danzandagba some time between 1727 and 1735, and *Kelen-ü cimeg* 'Jewel of language' written in 1828 by Agvangdandar (cf. Baldanžapov 1962*a*, *b*), divides the vowels into the harmony classes *em-e* 'female' and *er-e* 'male', and the neutral vowel *i* is called *ersü* 'hermaphrodite'. In *Jirüken-ü tolta-yin tayilburi*, the terms *köndüi* (*köndei*) 'hollow', *cingg-a* 'strong, tense', and *sagarmag* 'neutral' are given as well.

The acoustic data (Table 1.1) show that the second member of each pair $[u \sim v]$, $[o \sim 5]$, and $[e \sim a]$ has consistently higher F1 and normally lower F2 than the first

² For sources and notes on the phonetic basis for vowel harmony, see App. B on p. 219.

VOWELS

member. The same relation holds for $[x \sim a]$, which is found in most non-Hall Mongolian dialects corresponding to Halh $[e \sim a]$; see Svantesson (1985). This is similar to the relation within the vowel harmony pairs in several West African languages, such as Akan, where X-ray investigations have shown that the articulatory cause of this acoustic effect is that the first member of a pair is pronounced with a wider pharynx than the second member (Lindau 1975; 1979). There is direct confirmation that this is the case in Mongolian in the X-ray pictures published by Cenggeltei and Sinedke (1959), which show that [u] and [o] have a wider pharynx cavity than [u] and [ɔ], but are otherwise rather similar. Buraev (1959) (e.g. Figure 86, p. 110) and Buraev, Bažeeva, and Pavlova (1975) published X-ray pictures showing the same fact for Buriad [u] and [v]. Möömöö (1977) reports (pp. 56-7) that his unpublished X-ray pictures show that [u] and [o] (his \ddot{u} and \ddot{z}) have a wider pharynx cavity than [v] and [o] (his *u* and *o*). He further says that the tongue blade position is approximately the same for [u] and [v], and for [o] and [o], and that [v]and [5] are characterized by having a greater tension in the tongue muscles than [u] and [o].

Following Svantesson (1985), we will assume a feature [pharyngeal] that denotes activity in the hyoglossi muscles which pull the tongue root backwards, possibly combined with activity in the pharyngeal constrictor muscles. The vowels [v, v, a] have this feature, while [u, v, e] are non-pharyngeal (see further Chapter 5). The feature [pharyngeal] is more or less equivalent to [RTR] 'retracted tongue root'; the opposite feature [ATR] 'advanced tongue root' has also been used for describing vowel harmony. Lindau (1979) used the feature [expanded] (approximately the same as [ATR]) for describing vowel harmony in West African languages.

1.1.2 The vowel /i/

The vowel phoneme /i/ is neutral to vowel harmony, in the sense that it can occur in non-initial syllables of words whose first syllable has a non-pharyngeal vowel



FIGURE 1.6 Stylized formant tracings of the words [pa:bar] баалаар 'lap-dog-INST', [pa:big] баалыг 'lap-dog-ACC', and [a:bⁱg] аалийг 'manner-ACC' based on three recordings of isolated words by Speaker BB. F1 and F2 were measured at the beginning, middle, and end of the vowels. The shaded area represents the intervocalic consonant.

[e, u, o] as well as in words with a pharyngeal vowel [a, υ , υ] in its initial syllable. The pronunciation is different in these two cases, however, and in detailed phonetic transcriptions we write [i] in non-pharyngeal words and [I] in pharyngeal words. These two vowels are distinguished in Table 1.1c and Figure 1.4; the Cyrillic script writes them as $\mu\mu$ and ω . The vowel [I] has higher F1 and lower F2 than [i], the acoustic relation between them being the same as between the alternating vowel harmony pairs, so that [I] is the pharyngealized version of [i]. Since they occur in disjoint environments, we regard them as allophones of one phoneme *ii*. This variation takes place only in non-initial syllables, since *ii* does not occur in the initial syllables of pharyngeal words (5.2.1); see also Bayancogtu (1981*b*).

Palatalized and plain (non-palatalized) consonants contrast (see 3.2.1), and the vowel phoneme /i/ occurs after both plain and palatalized consonants, for example after plain [b] in [pa:b] баалыг and after palatalized [b] in [a:b] aaлийг (see Figure 1.6). As indicated in the phonetic transcription, the quality of the vowel is different depending on the type of the preceding consonant: it is [1] after plain and [i] after palatalized consonants. Although there is some coarticulation effect at the end of the preceding consonant, /i/ does not palatalize it. This can be seen by comparing F1 and F2 at the end of the first vowel [a:] in the word [pa:b] баалыг (Figure 1.6). The formants do not differ significantly in these words; see also Svantesson (1991a).

1.2 DIPHTHONGS³

There are four diphthongs which will be written [ai, oi, oi, ui] here. They are illustrated in Figure 1.7, which shows the path of these diphthongs in the acoustic F1– F2 space for Speaker DD, based on recordings of the material in (3). The paths show F1 and F2 at the beginning and end of the diphthong and at three intermediate equidistant points. As this figure illustrates, the beginning and end of the paths are more or less central in relation to the target vowels. In the literature (e.g. Ramstedt 1902), Mongolian diphthongs are often written as ending in [e]. This is phonetically more correct than the notation we use, which reflects their historical origin and possibly the targets of the diphthongs. In fast speech, the paths of the diphthongs are even shorter.

(3) Material for the acoustic investigation of diphthongs

	habitual	perfect gerund
'to fight'	[tai�z ^j təg] дайлдаг	[taiţ ^j at] дайлаад
'to fly up'	[sɔilʒ ^j təg] сойлдог	[sɔilʒ ^j ɔt] сойлоод
'to put one's hand in'	[suilz ^j təg] суйлдаг	[suilz ^j at] суйлаад
'to betroth'	[suilttəg] сүйлдэг	[suiget] сүйлээд

³ Luvsanvandan (1956; 1980*c*); Radnaeva (1988); Coyijongjab (1989*a*).



FIGURE 1.7 Diphthongs. Speaker DD. Each point represents the mean of four measurements. Long initial vowels are shown for comparison (circled).

1.3 PALATALIZED VOWELS

Consonant palatalization is contrastive in Mongolian, but the palatalized consonant phonemes are found only in words with pharyngeal vowels [a, σ , σ]. When these vowels (long or short) are followed by a palatalized consonant, their pronunciation is changed so that the final part of the vowel becomes more [i]-like (cf. the word [atg¹ig] in Figure 1.6). These palatalized variants are denoted [ä, $\ddot{\sigma}$, $\ddot{\sigma}$] in this section. Since they occur only before palatalized consonants, they must be regarded as allophones of the corresponding plain vowels. The non-pharyngeal vowels [e, u, o] never occur before palatalized consonants.

Kuribayashi (1985*a*) claims that the long palatalized vowels [ä:, ö:, ö:] and the *i*-diphthongs [ai, oi, oi] have merged in Halh. In order to investigate this, we recorded the words in (4) read in isolation by the three Ulaanbaatar speakers. The frequencies of F1 and F2 were measured at the beginning and end of the vowel, and at three intermediate equidistant points.

(4)	Long	palatali	zed vowels	s and <i>i</i> -c	liphth	ongs
	[äːţʲ]	ааль	'manner'	[ailʒ ^j]	айл	'encampment'

[ɔ̈ːɬʲ]	ооль	'adze'	[ɔir ^j]	ойр	'near'
[öːţʲ]	ууль	'owl'	[vilʒ ^j]	уйл	'to cry'

The results are reported in full in Svantesson (1991a), and are illustrated here in Figure 1.8. As this figure shows, the palatalized vowels are phonetically diphthongs, which have similar paths in the F1–F2 plane as the corresponding *i*-diphthongs, starting at a point in the neighbourhood of the corresponding non-umlauted vowel and ending in the $[e \sim i]$ area. The path of the palatalized vowel is often longer, ending closer to [i] than the diphthong. There is also a difference in the timing structure of the path. Although the diphthongs and long palatalized vowels are acoustical-



FIGURE 1.8 Long palatalized vowels vs. *i*-diphthongs. Speaker HB, based on five readings of isolated words.

ly similar, they are kept distinct in Ulaanbaatar Halh, and the difference is usually quite salient perceptually. Other dialects of Mongolian, such as Chahar and Baarin, have merged these vowels, however, and this may have taken place in some Halh areas as well.

Another interesting problem is the contrast between *i*-diphthongs and combinations of a vowel and the consonant [j], as in [pairar] байраар 'building–INST' vs. [pajrar] баяраар 'happiness–INST'; see section 6.5 for a phonological treatment of this problem and for more examples. Both are phonetically realized as diphthongs, beginning with a vowel and ending in a palatal glide, which is not fricative-like. In this case as well, the difference between the two kinds of diphthongs depends mainly on the timing of the spectral changes from the initial to the final ([i]-like) part of the vowel. Figure 1.9 illustrates the contrasts for speaker DD; the two other speakers show a similar pattern. The combination [Vj] is more diphthongic, and its final part is a fairly constant [i] vowel, while [Vi] has smaller formant movements.



FIGURE 1.9 Stylized formant tracings of [Vi] diphthongs and [Vj] combinations based on two recordings of words said in isolation by Speaker DD. F1 and F2 were measured at the beginning and end, and at three intermediate equidistant points. The illustrated words are [pairar] байраар 'building–INST' vs. [pajrar] баяраар 'happiness–INST' (left) and [uiţcčx] үйлээх 'to whirl–FUTP' vs. [ujţcčx] үелээх 'to fluctuate–FUTP' (right).

CONSONANTS¹

In this chapter we present briefly the results of our phonetic analyses of the Halh consonants, based on recordings of the same three Ulaanbaatar speakers as in Chapter 1. Some additional recordings were used for the illustrations.

2.1 STOPS AND AFFRICATES²

There are four places of articulation for stops in Mongolian: labial, dental, velar, and uvular. In addition, there are palatalized labial, dental, and velar stops, which contrast with the plain (non-palatalized) ones. There are also dental and alveopalatal affricates.

The velar, palatalized velar, and uvular stops, which have only one manner of articulation, are often voiced, unlike the other stops. Furthermore they function phonologically as voiced consonants (see 6.1.2). We will write them as [g, g^{j} , g]. They may be fricativized, in particular the uvular stop which has the variant pronunciation [μ].

For the other places of articulation, there are two contrasting manners. There is no consensus in the literature about their articulatory phonetic correlates. In his pioneering 1902 description of Halh Mongolian phonetics, John Ramstedt described the two series of stops and affricates as strong ~ weak (*fortis ~ lenis* or *stark ~ schwach*). According to Ramstedt, the strong series is voiceless, has post-aspiration in word-initial position, and has both pre- and postaspiration (*gehauchter eingang/ausgang*) in word-medial position. The weak series is usually voiceless, according to Ramstedt; other investigators describe them as 'devoiced'. It is obvious both from listening and from our instrumental investigation that the 'weak' Halh stops are ordinary voiceless unaspirated stops, similar to those found in Russian or French, and similar to the prototypical sounds denoted by the IPA symbols for voiceless stops.

The phonetic nature of the 'strong' stops varies with their place in a word. Our acoustic data show that they are voiceless postaspirated in word-initial position and voiceless preaspirated in all other environments. We will use the term aspirated stop. Unlike Ramstedt we find postaspiration only in word-initial position. In

¹ Coarticulation is treated briefly by Köke (1999a).

² For sources and notes on stops and affricates, see App. C, p. 220.

his analysis, strong consonants never occur word-finally; our word-final aspirated stops and affricates are written with a following reduced vowel by him.

In previous phonemic transcriptions, the two series are almost universally rendered with symbols for voiceless and voiced consonants, and some investigators describe them as phonetically voiceless ~ voiced; see the survey of the literature in the Appendix on p. 220. The Cyrillic Mongolian script treats them in this way, using letters which denote voiceless and voiced consonants in Russian for the two series. Since our investigation shows that aspiration is the distinctive property, we will write the Halh stops and affricates as voiceless aspirated ([ph], [th], etc.) and voiceless unaspirated ([p], [t], etc.). Although the aspiration sign is written after the consonant, it is intended as a symbol for (pre- or post-) aspiration in general. The system of stops and affricates is then as shown in (1). We write the affricates as [c, č] (IPA [ts, tʃ]). See section 3.2 for a phoneme analysis.

labial		dental		alveo-	velar		
plain	palat.	plain	palat.	palatal	plain	palat.	uvular
ph	p ^{jh}	t ^h	t ^{jh}				
p	p ^j	t	t ^j				
					g	gj	G
		ch		č ^h			
		с		č			

(1) Mongolian stops and affricates

We investigated the acoustic properties of Mongolian stops and affricates using recordings of the three Ulaanbaatar speakers BB, DD, and HB. A full report is given in Karlsson and Svantesson (2002), and the main results are outlined here. A list of word pairs where unaspirated and aspirated dental stops contrast minimally in different positions was recorded (see (2)). Each word was read three times by each of the three speakers in the carrier sentence [pir ____ gisəŋ] 'I said ____'. The recordings were analysed in *Praat*.

(2) Material for the acoustic investigation of dental stops

[tʰaɬʒa]	талаа	'steppe-rfl'	[taba]	далаа	'shoulder-blade-RFL'
[a ^h ta]	атаа	'camel gelding-RFL'	[ata]	адаа	'demon-RFL'
[a ^h t]	ат	'camel gelding'	[at]	ад	'demon'

The duration of the occlusion phase, of postaspiration, and of preaspiration was measured from wave-form plots and spectrograms. The results of the measurements are shown in (3), and they are illustrated in Figure 2.1, which shows the timing of the dental stops t^h and t for speaker DD.

The duration of the occlusion phase was measured for initial, medial, and final stops. We found no significant difference in the duration of the occlusion phase for aspirated and unaspirated consonants.



FIGURE 2.1 Timing of dental stops. Speaker DD.

Postaspiration was measured by the voice onset time (VOT), including the short explosion phase. It could be measured in initial and medial position in our material, but not word-finally. The main reason for this is that it was often impossible to find the boundary between a final stop and the initial stop (in some cases fricativized) of the word [gisəŋ] in the carrier sentence. Audibly there is no postaspiration word-finally, however, and additional observations made on words said in isolation by other speakers showed no word-final postaspiration. In word-initial position, VOT is very short for unaspirated consonants, and it is always greater for the aspirated ones. The difference is large and shows a clear statistical significance. In medial position, the difference in VOT is small and not significant. Our conclusion is that postaspiration is consistent and salient in initial, but not in medial or final position.

The final part of the vowel preceding an aspirated stop is usually pronounced with clearly audible aspiration noise and is at least partially devoiced. The duration of preaspiration was measured from the beginning of the devoicing of the vowel to the beginning of the closure phase of the consonant. Preaspiration is most salient when the preceding vowel is in the same word, that is, when the consonant is in medial or final position, but occurs also in the vowel [ir] of the first word [pir] 'I' of the carrier sentence when it precedes an aspirated consonant. For example, the sentence frame with the word [t^hata] 'steppe-RFL' is pronounced [pir.^ht^hatagisõ] 'I said steppe'. The fact that aspirated stops have preaspiration in all positions shows that this is a consistent phonetic correlate differentiating aspirated and unaspirated stops, although it cannot be realized in utterance-initial position. Preaspiration does not occur with unaspirated stops.

			preaspiration	occlusion	postaspiration
initial:	BB	t ^h	18	64	57
		t	_	75	22
	DD	t ^h	11	99	58
		t	_	98	11
	HB	t ^h	10	68	40
		t		65	23
medial:	BB	t ^h	39	117	20
		t		134	16
	DD	t ^h	41	110	15
		t		129	13
	HB	t ^h	39	123	19
		t		117	14
final:	BB	t ^h	30	123	
		t		130	
	DD	t ^h	26	100	
		t		104	
	HB	t ^h	15	98	
		t	—	107	

(3) Duration of dental stops (ms) (mean of three readings)

The contrast between aspirated and unaspirated stops in different positions is illustrated in Figures 2.2–2.4 with wave-form plots and spectrograms of words read in isolation by a female Ulaanbaatar speaker, Ènhmaa (EM), age forty-five. She was recorded in Moscow in 2001, using a DAT cassette recorder. The figures were made with *WaveSurfer*. When an aspirated stop is preceded by a sonorant, aspiration is realized as devoicing of the sonorant, as illustrated for [b] in Figure 2.5 (cf. Galsan 1969). This effect has been observed across word boundaries as well, devoicing a word-final sonorant before a word beginning with an aspirated consonant.

In conclusion, our data may be interpreted as showing that preaspiration is the main distinctive property, which is realized as such in those positions where this is



FIGURE 2.2 Wave-form and spectrogram for $[t^hab]$ тал 'steppe' (left) and [tab] дал 'shoulder-blade' (right). Speaker EM. Aspiration is seen as high frequency noise in the spectrograms. There is a clear difference in VOT between $[t^h]$ and [t].


FIGURE 2.3 Wave-form and spectrogram for $[a^ht]$ at 'camel gelding' (left) and [at] ad 'demon' (right). Speaker EM. Preaspiration can be seen as high-frequency noise in the second half of the vowel preceding the aspirated stop, and also as blurring of the vowel formants. The releases (explosion phases) are similar for the two stops, and there is no contrastive postaspiration.



FIGURE 2.4 Wave-form and spectrogram for [a^hta] ataa 'camel gelding-RFL' (left) and [ata] adaa 'demon-RFL' (right). Speaker EM. Here, as well, the releases are similar for the two stops, and there is no contrastive postaspiration.



FIGURE 2.5 Wave-form and spectrogram for /agt^h/ [a4t] алт 'gold' (left) and /agt/ [agt] алд 'fathom' (right). Speaker EM. The lateral fricative /g/ is rather voiceless for some speakers, but at least its first part is usually voiced. Before an aspirated consonant it becomes completely voiceless and has more high-frequency noise.

possible. Word-initially, or at least utterance-initially, preaspiration is phonetically impossible so it has to be realized as postaspiration instead.

Preaspiration is not very common among the world's languages. Medial and final preaspiration and initial postaspiration seem to be the normal pattern in languages with preaspiration, such as Scottish Gaelic and Icelandic. These languages also have devoicing of sonorants preceding preaspirated stops. See Helgason (2002) for a general survey of preaspiration.

2.1.1 Stops in other Mongolian dialects³

Köke and Coyijongjab (1999: 121) made instrumental phonetic investigations of the Chahar dialect of Mongolian. They analyse the stops as (post)aspirated vs. unaspirated voiceless, and do not report preaspiration. They give data on VOT and the duration of the occlusion phase, presumably for words said in isolation, showing that the occlusion phase is much longer for aspirated stops compared to unaspirated stops (in word-medial position). Their VOT values are similar to ours in word-initial position, but not word-medially. Their Figure 52 (p. 223) shows rather heavy postaspiration on word-medial [t^h], and no visible preaspiration. They do not show figures of word-final aspirated stops, but our informal listening to Mongolian dialects spoken in China (including Chahar) suggests that there is preaspiration in this position. A preliminary conclusion is that Chahar and other Mongolian dialects spoken in China have preaspiration only word-finally, and postaspiration in other positions. This needs to be investigated further, however.

2.1.2 Buriad and Kalmuck stops⁴

As is the case for Mongolian proper, there is no consensus in the literature about the analysis of the stops in the closely-related languages Buriad and Oirad (including Kalmuck). Instrumental phonetic studies of Buriad stops were made by Bajčura (1978) and Soktoeva (1980*a*; 1981*a*, *b*; 1983; 1984; 1988; 1989*a*). Bajčura (1978: 2) says that Buriad strong consonants are almost twice as long as the corresponding weak consonant, citing Bajčura (1961: 144), where he says, however, that strong and weak consonants have almost the same duration. Soktoeva (1981*a*) says that the difference between the Buriad stop series depends on the presence or absence of voicing (*v učastii ili v otsutstvii golosa*), as in Russian, but in Soktoeva (1983; 1988) maintains that relatively long duration is the most important phonetic correlate of strong Buriad stops, although this is often accompanied by a voiceless (*gluhoj*) vs. voiced (*zvonkij*) difference which is, however, phonologically irrelevant. Several earlier researchers, including Poppe (1930*b*) and Ramstedt (1957: 42), analyse Buriad and Kalmuck weak stops as being voiced, unlike those in Mongolian proper.

This agrees with our informal observations. Figure 2.6 shows the word for 'hero' (whose Halh form is $[pa:^{h}t\check{a}r]$ 6aarap) said in isolation by three male speakers, of (Hori) Buriad, (Dörbed) Kalmuck, and Halh, respectively. Buriad initial [b] is clearly voiced, and the Kalmuck initial consonant shows voicing as well. In contrast, Halh has an initial voiceless [p]. The word-medial stop is also different. In Buriad it is a clearly postaspirated [t^h] and in Halh a preaspirated [^ht]. Kalmuck also has postaspiration, although shorter than in Buriad. Another difference between

³ For sources and notes on stops in Mongolian dialects, see App. C, 2.1.1, p. 221.

⁴ For sources and notes on Buriad and Kalmuck stops, see App. C, 2.1.2, p. 221.



FIGURE 2.6 Wave-form and spectrogram for Buriad [ba:thăr] (top left), Kalmuck [ba:thr] (top right), and Mongolian [pa:htăr] (bottom).

the languages is that the second vowel, which is a reduced $[\check{a}]$ in Buriad and Halh, is even more reduced in Kalmuck, which has a final syllabic [r]. All these observations are preliminary, however, and need to be confirmed by more research.

2.2 FRICATIVES⁵

There are five fricatives, $[x, \chi, x^j, s, \xi]$. Velar [x] occurs in words with non-pharyngeal vowels [e, o, u] and uvular $[\chi]$ in words with pharyngeal vowels $[a, \sigma, \sigma]$ (cf. 3.2.2). The palatalized fricative $[x^j]$ probably has a purely palatal pronunciation.

The sibilant fricatives $[s, \check{s}]$ are postaspirated in word-initial and medial position, see Figure 2.7. In some cases there is devoicing of a preceding vowel, but this is less consistent. In connection with this, it can be noted that Old Mongolian **s* groups with the aspirated stops in the historical phonological process of deaspiration (see 10.10).

2.3 NASALS⁶

There are six nasals $[m, m^j, n, n^j, \eta, N]$. Velar $[\eta]$ occurs only in words with non-pharyngeal vowels and uvular [N] only in pharyngeal words. Unlike the other nasals, these two are found only in syllable codas, and in connected speech they are normally realized only as nasalization of the preceding vowel, or of its final part if it is long (see Fig. 2.8).

⁵ Köke (1998*a*). ⁶ Bulucilagu (1985*c*); Soktoeva (1985).



FIGURE 2.7 Wave-form and spectrogram for the words /saw/ [s^haw] cab 'vessel' (left) and /asax/ [as^ha χ] acaax 'to light–FUTP' (right) said in isolation by Speaker DD. Postaspiration of [s] is seen as noise with lower frequency than the sibilant noise. The initial *a* in *asax* is completely devoiced.



FIGURE 2.8 The nasals /n/ vs. /ŋ/. The words /xonx/ $[\chi \text{on}\chi]$ xoHox 'to stay overnight–FUTP' (left) and /xonx/ $[\chi \tilde{3}\chi]$ xoHx 'bell' (right) said in isolation by Speaker HB. In the first word, the vowel [5] is not very nasalized, and the nasal is consonantal, but in the second word there is a nasalized vowel and no nasal consonant.

2.4 LIQUIDS⁷

The liquids in Mongolian are the lateral fricative [b], the vibrant [r], and the corresponding palatalized consonants $[b^j, r^j]$. The presence of the lateral fricative [b] while a plain lateral [1] is absent is a typologically unusual feature. This conspicuous pronunciation of the lateral is seldom mentioned in the literature; exceptions are Ramstedt (1902: 27) and Saitô (1986: 116). There is a tendency to devoice [b], speaker BB does this consistently, while the other speakers DD and HB have at least partially voiced [b], except before aspirated stops.

There is also a voiceless lateral [4] found in word-initial position in a few Tibetan loan-words. It differs from [β] not only by being voiceless but also by having

⁷ Soktoeva (1985; 1986*c*); Guw-a (1995); Köke (1996).



FIGURE 2.9 Wave-form and spectrogram for [bam] лам 'lama' (left) and /lagw/ [laвw] лхагва 'Wednesday' (right). Speaker DD. The first word [pi:] of the carrier sentence is seen in the figures.

higher intensity and more high-frequency noise (see Fig. 2.9), and it is differentiated from [b] even by those speakers who devoice [b]. [4] often consists of two phases; the second phase may be described as strong postaspiration.

The /r/ phoneme is usually pronounced as a vibrant [r] or a flap [r] by Speakers DD and HB, but as a fricative [I] (often devoiced word-finally) by Speaker BB.

2.5 GLIDES

The glide [w] (and palatalized [w^j]), which has developed historically from the Old Mongolian stop *p when it occurred after a vowel, functions phonologically as a consonant. Phonetically, [w] is less rounded than e.g. English [w] and is sometimes described as a bilabial fricative [β]; it is, however, a glide (approximant) rather than a fricative.

A consonant that can probably be described as a labialized uvular glide (written $[B^w]$ by us) is found in a few Tibetan loan-words, including $[4aB^w]$ JXaFBa 'Wednesday' (cf. Fig. 2.9). We regard it as the phonetic realization of the consonant cluster /Gw/.

2.6 PALATALIZED CONSONANTS

As mentioned above, palatalized consonants $[p^{jh}, t^{jh}, p^{j}, t^{j}, g^{j}, x^{j}, m^{j}, n^{j}, b^{j}, r^{j}, w^{j}]$ contrast with the corresponding plain consonants, but only in words with pharyngeal vowels. Palatalized consonants coarticulate with surrounding vowels, decreasing F1 and increasing F2 in adjacent parts of the vowels (cf. 1.3). This is illustrated with stylized formant tracings in Figure 2.10.

Palatalized and plain consonants do not contrast after *i*-diphthongs, and our acoustic investigation shows that consonants in this position are phonetically palatalized. Figure 2.11 shows stylized formant tracings of plain [b] and palatalized



FIGURE 2.10 Stylized formant tracings (F1 and F2) for the words [pa:bar] баалаар 'lapdog-INST' and [a:b^jar] аалнар 'manner-INST', based on three readings of isolated words by Speaker BB. F1 and F2 were measured at the beginning, middle, and end of the vowels. The shaded area represents the intervocalic consonant.

[b^{j}] after [a:] and of non-contrastive [$b^{(j)}$] after [ai]. The vowel after the consonant is /i/. It is clear that the consonant is palatalized after [ai], and that the following /i/ is pronounced [i] rather than [I] in this case (cf. 1.1.2). This seems to be the only case of progressive palatalization in Halh. Since palatalization is not contrastive in this environment, and since we have not investigated how far the palatalizing influence of the diphthong reaches, we will normally write plain consonants after *i*-diphthongs, except in detailed phonetic transcriptions; we thus write the word exemplified in Figure 2.11 as *aifsig* айлыг although the pronunciation is [aik^jig] (and not [aibig], as the Cyrillic spelling suggests).



FIGURE 2.11 Stylized formant tracings (F1 and F2) for the words [pat;hg] баалыг 'lap-dog-ACC', [a;hjig] аалийг 'manner-ACC', and [a;hjig] айлыг 'family-ACC' based on three readings of isolated words by Speaker BB. F1 and F2 were measured at the beginning, middle, and end of the vowels. The shaded area represents the intervocalic consonant.

3

PHONEMES

In this chapter we analyse the Mongolian segmental phonemes, with the phonetic investigations in the previous chapters as the basis. Our analysis differs in many respects from previous analyses by Western authors, such as Ramstedt (1902), Poppe (1951*a*; 1970), or Street (1963), but is more similar to the analyses of some Russian, Chinese, and Japanese authors, in particular Saitô (1985; 1986).

3.1 VOWEL PHONEMES¹

There are seven basic vowels in Mongolian:

(1) i u v e o a o

There is contrasting vowel length, but long vowels occur only in word-initial syllables. The short vowel o is realized as $[\Theta]$, and the short vowel e occurs only in non-initial syllables (see 1.1). Four of the vowels form diphthongs with i. In initial syllables, there are thus three kinds of contrasting vowels, short, long, and diphthongs:

(2) Vowel phonemes in initial syllables

short	long	diphthongs
i u	ii uu	ui
υ	υυ	υi
0	ee oo	
aэ	aa oo	ai oi

Words illustrating these contrasts are:

(3) Vowel contrasts

ily ил 'clear'	tiilд дийл 'to win'	
	ееђ ээл 'blessing'	
alz ал 'to kill'	аадс аалз 'spider'	ailt айл 'encampment'

¹ Rygaloff (1973); Möömöö (1975); Bitkeev (1982; 1985*b*); Luvsanvandan (1982*a*); Norjin (1991; 1992*a*); Kökebars (1999). For the diphthong *va*, see Vladimircov (1929: 299).

uђ үл 'not'	սսե հծ	'cloud'	uib үйл 'deed'
υђ ул 'sole'	оођ уул	'mountain'	иід уйл 'to cry'
ођ өл 'grey'	оођ өөл	'to trim'	
эђол 'to find'	pook goou	л 'slave'	эіђойл 'to understand'

In addition to these vowels, there are a few marginal diphthongs and triphthongs. The diphthong υa and the triphthong υai occur in a few words, mostly Chinese loans. Except for the word š υaic $\Box yai3$ 'brush' (Ch shuāizi 刷子) and perhaps a few others, they follow the consonants x or g. Some examples are $\upsilon acanc$ $\Box yai3$ 'restaurant' (Ch guǎnzi 館子), x υar xyap 'flower' (Ch huār 花兒), x υais xyaĭc 'acacia' (Ch huáishù 槐樹), and υai $\Box yaŭ$ 'Mr', which goes back to Classical Mongolian abugai. A possible interpretation with labialized consonants x^w , σ^w , was proposed by Saitô (1986) (see also 5.4.2). The diphthong a υ has been found only in the word a υaga ayraa 'mighty'.

3.1.1 Non-initial vowels²

Non-initial syllables can have two kinds of vowels, full and reduced. These have usually developed from long and short vowels at an earlier stage in the historical development of the language (see 10.3, 10.5). We showed in section 1.1 that the quality of a reduced vowel is predictable, being a centralized version of the vowel in the preceding syllable, except that a reduced vowel following [u] is a centralized version of [e]. As is the case for vowel harmony, this rule ignores intervening *i*. For instance, *mači* [mači] мажий 'to die' and $2\xi^{j}i$ [$2\xi^{j}i$] олий 'to squint' have the habitual forms [mačităg] мажийдаг and [$2\xi^{j}it$] олий does not be used the reduced vowel depends on the vowel preceding [i].

The quality of a reduced vowel depends on the preceding consonant as well: it is [i]-like if the preceding consonant is an alveopalatal ($\check{s}, \check{c}^h, \check{c}$) or a palatalized consonant ($p^{jh}, t^{jh}, p^j, t^j, g^j, x^j, m^j, n^j, \xi^j, r^j, w^j$). Examples are the reduced vowels in [ačib] ажил 'work' and [at^jib] адил 'like'.

We will show in Chapter 6 that not only the quality, but also the places, where reduced vowels occur are predictable. Thus they can be inserted (epenthesized) by a rule, and need not be present in phonological representations. In our usual notation of Halh words, we will write out the epenthetic vowels as schwas (∂), however, so that we write e.g. *xamor* for [xamăr] xamap 'nose', *pɔ\bar* for [pɔ\barbar] болор 'crystal', *ačə\barbar* for [ači\barbar] ажил 'work', and *at^ja\barbar* for [at^j\barbar] адил 'like'; the phonological representations of these words are /xamr/, /pɔ\bar/, /ač\barbar/, /at^j\barbar.

Under our analysis, where the reduced vowels are epenthetic, the full vowels are the only remaining phonemic vowels in non-initial syllables. Since there is no

² Vladimircov (1929: 311ff.); Bosson and Unensečen (1962); Luvsanvandan (1964*a*; 1969); Tömörcèrèn (1968); Sun (1981); Rialland and Djamouri (1984); Mikami (1985; 1986); Kuribayashi (1988; 1992); Kögjiltü (1991–2); Köke and Coyijongjab (1996); Köke and Sodobagatur (1996); Svantesson and Franzén (1996); Sodobagatur (1999). *Kalmuck*: Street (1962); Pavlov (1982); Pavlov and Esenova (1986).

PHONEMES

length contrast, we analyse the full non-initial vowels as having short (unmarked) quantity, although they have developed historically from long vowels. This analysis is supported by the fact that the durations of full non-initial vowels are intermediate between long and short vowels in initial syllables, but closer to the short vowels (see 1.1). The phonemic vowels in non-initial syllables can be either monophthongs or diphthongs (4). The contrasting possibilities of non-initial vowels are reduced further by vowel harmony (see 5.2).

As seen in section 1.1.2, the phoneme /i/ has two allophones, [i] and [I], in noninitial syllables. The former occurs in words with non-pharyngeal vowels, and also after palatalized and alveopalatal consonants in pharyngeal words, the latter elsewhere in pharyngeal words. Initial /i/ is found only in non-pharyngeal words, so it is always pronounced [i].

(4)	Phonemic vowels in non-initial syllables								
	monophthongs	diphthongs							
	i u	ui							
	υ	vi							
	e o								
	aэ	ai oi							

In the literature, there are at least four other ways of treating non-initial vowels:

- (a) Full vowels are regarded as allophones of long vowels, and reduced vowels are regarded as allophones of short vowels. This analysis lies behind the Cyrillic Mongolian script, and is usually assumed in traditional descriptions, for instance Poppe (1951a).
- (b) As (a), except that reduced vowels are regarded as allophones of a separate schwa vowel phoneme /ə/. The reduced vowel [ĭ], however, is treated as an allophone of /i/ (Street 1963; Poppe 1970).
- (c) Stuart and Haltod (1957) adopt the analysis that reduced vowels are epenthetic and absent from phonological representations, except that [ĭ] is an allophone of /i/. Full vowels in non-initial syllables are analysed as short, except that [i] is regarded as long /ii/. This is necessary because they do not recognize palatalized consonants as phonemes. Kakudô (1974) has a similar analysis.
- (d) Rialland and Djamouri (1984) adopt the same analysis as Stuart and Haltod, except that they regard all full non-initial vowels as long. They regard [ĭ] after alveopalatal sibilants as non-phonemic, but it is not clear how they would treat [ĭ] after palatalized consonants.

These different ways of analysing non-initial vowels are illustrated in (5), showing words in Cyrillic Mongolian, in phonetic transcription, in the phonemic analysis adopted here, and in the four other analyses (in our interpretation, and using our segment notation).

7

.

c

7

(-)

D . CC

(5)	(5) Different analyses of non-initial vowels							
			our	(a)	(b)	(c)	(d)	
			analysis					
саарм	ıаг [sa <mark>ırm</mark> ăg]	'neutral'	/saarmg/	/saarmag/	/saarməg/	/saarmg/	/saarmg/	
xoëp	[xɔjŏr]	'two'	/xɔjr/	/xəjər/	/xɔjər/	/xəjr/	/xɔjr/	
санаа	[sana]	'thought'	/sana/	/sanaa/	/sanaa/	/sana/	/sanaa/	
ажил	[ačĭlʒ]	'work'	/ačŀʒ/	/ačiţ/	/ačiţ/	/ačiţ/	/ačţz/	
адил	[at ^j ĭ]ʒ]	'like'	/at ^j t/	/atilʒ/	/atilʒ/	/atiŀz/	?	
мини	й [mini]	'my'	/mini/	/minii/	/minii/	/minii/	/minii/	

3.2 CONSONANT PHONEMES³

The total inventory of contrasting consonants is given in (6). Phonemes shown in brackets are marginal, occurring only in loan-words (see 3.3) and some onomatopoeics.

(6) Mongolian consonant phonemes

	labial	palatalized labial	dental	alveopalatal	palatal	velar	uvular
voiceless aspirated stops	(ph)	(p ^{jh})	t ^h	t ^{jh}			
voiceless unaspirated stops	p	p^j	t	ť			
voiced stops					gj	g	G
voiceless aspirated affricates			c^h	čh			
voiceless unaspirated affricates			с	č			
voiceless fricatives			S	š	xj	Х	
nasals	m	mj	n	n ^j		ŋ	
voiced lateral fricatives			ß	$\mathfrak{z}^{\mathfrak{j}}$			
voiceless lateral fricative			(4)				
rhotics			r	r ^j			
glides	W	wj			j		

As shown in Chapter 2, the aspirated consonants are postaspirated in word-initial position, and preaspirated in word-medial and word-final position. The distribution of consonants differs considerably between word-initial and non-initial (medial or final) positions. Due to the suffixing nature of the language, the contrasting possibilities are usually the same in medial and final position. When a suffix is

³ Möömöö (1976; 1977); Luvsanvandan (1982*b*); Saitô (1986; 1996*b*); Sanžaa (1988; 1990). *Final n*, *y*: Lutajirgal (1990); Pèljei (1992).

PHONEMES

added, a final consonant becomes medial and is usually not changed (e.g. $x > t^h x$ or 'town', ablative $x > t^h > s$ xorooc, comitative $x > t^h t^h > i$ xorroŭ). The only exception is y, which changes to n when a suffix beginning with a vowel or a dental consonant is added (see 5.3).

Vowel harmony affects the consonant system as well. As seen in section 1.1.1, the vowels can be divided into two classes which cannot co-occur in the same word. The classes are based on the feature [pharyngeal], so that a, z, v are pharyngeal vowels and e, o, u are non-pharyngeal. The non-pharyngeal vowel *i* is neutral to vowel harmony (see 5.2 for details). The full set of consonants is found only in words with pharyngeal vowels. The main reason for this is that the contrast between palatalized and plain consonants is upheld only in pharyngeal words, so that the consonant phonemes p^{jh} , t^{jh} , p^j , t^j , g^j , x^j , m^j , n^j , ξ^j , r^j , w^j do not occur in non-pharyngeal words (cf. Hattori 1983c; Sanžaa 1987a).

The velar nasal η never occurs in syllable initial position, and the consonants ξ , w, and r (and their palatalized counterparts) do not occur in word-initial position in native Mongolian words. A number of loan-words with initial ξ and w have been incorporated into the language, however (see 3.3). Words illustrating consonant contrasts in different environments are given in Table 3.1.

(a) II	nitial	and fina	l cons	onants	in phary	ngeal w	ords					
Word-initial				Word-	final							
(p ^h) p	(p ^{jh}) p ^j) t ^h t c ^h c	(t ^j) č ^h č	gj	G	р	pj	t ^h [^h t t c ^h [^h c	t ^j	g ^j	g	G
m	m ^j	s n Է (4)	š n ^j (Է ^j)	xj	x [χ]	m	m ^j	s n Է	š n ^j K ^j	x ^j		х [χ] ŋ [n]
(w)				j		w	w ^j	r	r ^j	j		
p ^h atz p ^{jh} atz t ^h atz		пал пял тал	'pla	ash!' te' ppe'		at ^h pət ^{jh}	ат бо [,]		camel geldi volume'	ng'		
pal z p ^j alz talz t ^j aŋ g ^j alza	L	бал бял дал дян гялаа		mear' enty'		xolgipj xolgipj at cat ^j ag ^j ag	xo.	лбо't лби'с (с ць'1 , 'х	to join' off' demon' nutmeg' wormwood tight'	,		
Gałz		гал	'fire	,		aG	Аг		Buriad area	a)		

TABLE 3.1 Examples of consonants in different positions

c ^h am	цам	'mask dance'	ach	ац	'fork'
č ^h at	чад	'to be able'	ač ^h	ач	'grandson'
cam	зам	'road'	ac	аз	'good luck'
čam	жам	'law'	ač	аж	'to notice'
sab	сал	'raft'	as	ac	'to climb'
šalz	шал	'floor'	xaš	хаш	'jade'
x ^j al z pər	хялбар	'easy'	ax ^j	ахь	'to advance'
xab	хал	'to change'	ax	ax	'elder brother'
mab	мал	'cattle'	am	ам	'mouth'
m ^j a lz a	мялаа	'to anoint'	am ^j	амь	'life'
nab	нал	'to lean'	an	ана	'to follow'
n ^j alz	нял	'to smear'	an ^j	ань	'to close'
			aŋ	ан	'beast'
Ъam	лам	'lama'	ab	ал	'to kill'
Է ^j aŋxʊa	лянхуа	'lotus'	ађ ^ј	аль	'which'
łam	лхам	'Buddhist deity'			
			ar	ap	'back'
			xar ^j	харь	'strange'
waar	ваар	'tile'	aw	ав	'to take'
			xaw ^j	хавь	'near'
jab	ЛК	'punishment'	aj	ая	'melody'

(b) Initial and final consonants in non-pharyngeal words

Word-	initial		Wo	rd-fi	inal		
(p ^h)	t ^h			t ^h [^h t]		
р	t c ^h	$\overset{g}{\check{c}^h}$	р	t c ^h [č ^h [^h č]	g
	c	č š x		c		č š	
m	s n	Š X	m	s n		5	x
111	日 日 (4)		ш	ß			ŋ
(w)		j	w	r			j
p ^h uu	пүү	<i>'pud</i> (weight unit)'					
t ^h uu	түү	'to gather'	ith	3	ЭT	'to	spike'
puu	бүү	'don't'	šilz	рг	пилб	э'sh	in'
tuu	дүү	'younger brother'	it	I	ад	'vi	gour'
guu	гүү	'mare'	ig	I	łΓ	'sp	indle'
c ^h uu	цүү	'ring'	ich	3	эц	'to	become weary'
č ^h ig	ЧИГ	'direction'	ič ^h	ł	44	'rig	ght'
cuu	зүү	'needle'	ic	3	93	ʻill	omen'
čurč	жүрж	'orange'	ič	I	4Ж	'to	tal'
suu	сүү	'milk'	is	I	4C	'so	ot'
šuu	шүү	'saltpetre'	iš	I	иш	'ste	em'

Word-initial		Word-final			
xuu	хүү	'boy'	ix	их	'big'
muu	мүү	' <i>mu</i> (area unit)'	im	ИМ	'earmark'
nuu	нүү	'to move'	in	энэ	'this'
			iŋ	ин	'millstone'
Էսս	лүү	ʻjar'	iţ	ил	'clear'
łump	лхүмбэ	'grandiose'			
-		-	ir	ир	'blade'
wiic	ВИЗ	ʻvisa'	iw	ив	'completely'
juu	юу	'what?'	ij	эе	'harmony'

TABLE 3.1 (cont.)

3.2.1 Palatalized consonants⁴

The palatalized consonants are historically conditioned by a following *i, which disappeared (see 10.11.1). One example is am^j амь 'life' from Old Mongolian *amin (contrasting with am ам 'mouth', OM *aman). Palatalized and plain consonants do not contrast in words with non-pharyngeal vowels, and as seen in section 2.6 they do not contrast after the diphthongs ai, oi, vi.

Palatalized consonants are fairly rare in word-initial position, where they have developed as a result of assimilation of the vowel *i* to a following vowel, for example Old Mongolian $k^{h}ilpar > x^{j}a \not zpar xялбар$ 'easy' (this process is traditionally called 'breaking'). Palatalized t^{jh} , r^{j} , and w^{j} do not occur initially at all, and k^{j} and t^{j} are found only in very few loan-words, including $j^{j}a\eta x va \pi \pi \pi x$ 'lotus' (from Ch $liánhu\bar{a}$ $\bar{a}\bar{c}$) and $t^{j}a\eta \pi \pi$ 'inn' (Ch dian \bar{E}). The palatalized consonants p^{j} , g^{j} , x^{j} , m^{j} , n^{j} are more common in initial position, but occur almost exclusively before *a* or *aa*; see examples in Table 3.1. Palatalized x^{j} is found before other vowels in a few words, however, such as $x^{j}vor x\mu yp$ 'pennant' and $x^{j}vormag x\mu ypmar$ 'gravel'.

Palatalized t^{j} and t^{jh} are not frequent in any position. The historical explanation for this is the non-occurrence of the combinations *ti and *thi in Old Mongolian; they had probably developed to $*\check{c}i$ and $*\check{c}hi$ already in Pre-Mongolic.

3.2.2 Velar and uvular consonants⁵

Velar and uvular consonants alternate with the vowel harmony class of the word, so that velars occur only in words with non-pharyngeal vowels (*i*, *e*, *u*, *o*), and uvulars only in words with pharyngeal vowels (*a*, *v*, *o*). Thus, the phonemes /ŋ, x/ are realized as velars [ŋ, x] in non-pharyngeal words, and as uvulars [N, χ] in pharyngeal words (7a). Even when a suffix beginning with *i* is added to a stem ending in [χ],

⁴ For sources and notes on palatalized consonants, see App. D, 3.2.1, p. 222.

⁵ For sources and notes on velar and uvular consonants, see App. D, 3.2.2, p. 222.

the uvular pronunciation is retained (7b), although only palatalized or alveopalatal consonants occur before i in uninflected pharyngeal words (see 5.2.1). The vowel i is written ω in this case.

The velar g is an exception. In word-initial position it behaves like the other velars and occurs only in non-pharyngeal words, alternating with σ in pharyngeal words (7c), but velar g may also occur in morpheme final position in pharyngeal words, contrasting with uvular σ (7d). This contrast is upheld when suffixes are added (7e).

(7)	Velar ~	uvular	alternation
$\langle I \rangle$	VCIUI -	uvuun	ancimation

(a)	c ^h ooŋ	$[c^{h}orn \sim c^{h}\tilde{o}r]$ $[\Im n \sim \tilde{\Im}]$	цөөн	'few'	velar uvular
	ວກ		он	'year'	
	xib	[xib]	хэл	'tongue'	velar
	xar	[xar]	хар	'black'	uvular
	sux	[sux]	сүх	'axe'	velar
	ax	[aχ]	ax	'elder brother'	uvular
(b)	sux-ig	[suxig]	сүхийг	'axe-ACC'	velar
	ax–ig	[aɣıg]	ахыг	'elder brother-ACC'	uvular
(c)	gir	[gir]	гэр	'house'	velar
	Gar	[Gar]	гар	'hand'	uvular
(d)	pag	[pag]	баг	'team'	velar
	pag	[pag]	бага	'small'	uvular
(e)	pag–as	[pagas]	багаас	'team-RFL'	velar
		[pagas]	багаас	'small-rfl'	uvular
	pag-as	[paGas]	0ai aac	SIIIaII-KFL	avatar
	pag–as pag–ig	[pagig]	багийг	'team–ACC'	velar

3.2.3 Labials

The contrast between *p* and *w* is marginal. The historical reason for this is that Old Mongolian **p* was 'weakened' to *w* in most non-initial positions (see 10.8.1), and is preserved only word-initially and after *m*, *b*, or *w* (e.g. in *bimp* лимбэ 'flute', *abp* алба 'service', *awpəb* авбал 'to take–COND'). There was no *w* in Old Mongolian, but it was introduced in initial position in loan-words, so that a new contrast was created, as in *paar* баар 'to fail' vs. *waar* ваар 'tile' (from Ch *wăr* 瓦兒). In final position, contrasts such as /*abp*/*abp* алба 'service' vs. /*ab–w*/*abaw* алав 'to kill–PST' occur. These words do not contrast minimally on the surface, because of the epenthetic vowel, but the phonological representations have minimally contrasting segmental contents.

The palatalized consonants p^j and w^j have a distribution similar to that of p and w: p^j occurs only initially or after m^j or ξ^j , and w^j has not been found initially. We have not found any minimal contrasts between these two consonants, but they might be regarded as separate phonemes in analogy with p and w.

PHONEMES

The consonants p^h and p^{jh} occur in some native onomatopoeic and expressive words, e.g. $p^{h}at$ пад 'thud!', $p^{jh}as$ пяс 'bang!', and also in loans, for example, $p^{h}anc$ панз 'pedlar' (Ch fànzi 販子), $p^{h}uus$ пүүс 'company' (Ch pùzi 舖子), $p^{jh}anc$ пянз 'gramophone record' (Ch piànzi 片子), $p^{jh}ats$ пял 'plate' (Manchu fila) (see also 3.3).

3.3 LOAN-WORD PHONOLOGY⁶

Numerous loan-words have entered Mongolian, originally from Chinese and Tibetan, and later from Russian and English. It is difficult to decide exactly which borrowed sounds have become regular Mongolian phonemes, since an individual's pronunciation of loan-words depends on his knowledge of the donor language. Thus, those who know Russian well may pronounce [k] and [f] when they occur in loans, but it is common to substitute [x] and [p^h] for them.

The phoneme t occurs only in a few loans from Tibetan, the only common word being tagw [tagw] $\pi xar Ba$ 'Wednesday' (Tib tegers' lhag-pa); see section 2.5 for the final consonants of this word. As mentioned in section 3.1 above, the diphthong va(and the triphthong vai) appears mainly in Chinese loans.

There are no indigenous words which begin with *r*, but Cyrillic Mongolian writes initial p < r > in some Russian and Tibetan loans, e.g. радио <radio > 'radio' (Ru радио *rádio*) and the name Ринчен <Rinčen > (Tib ^{\$4}(35); *rin-chen*). Such words are usually pronounced with an added initial vowel (*aračaw*; *irančhay*) (as discussed in this section). In more established loans, the Mongolized form is standard, as in *aršay* аршаан 'holy water' (Sanskrit *rasiyana*) and *irtən* эрдэнэ 'jewel' (Sanskrit *ratna*).

Similarly, word-initial ξ and w do not occur in indigenous words, but they have been introduced with loans, e.g. ξ ат лам 'lama' (Tib 회학 bla-ma), ξ imp лимбэ 'flute' (Tib 회학 gling-bu), ξ а ξ лал 'Muslim' (Tib 회학 kla-klo); waar ваар 'tile' (Ch wǎr 瓦원); wiic виз 'visa' (Russian viza).

Especially in Soviet times, many Russian loans entered colloquial Mongolian. Their pronunciation varies depending on the speaker's education and knowledge of Russian, and also depending on when, and in which way, the word was borrowed. Some examples are given in Table 3.2, and the ways in which such words were adapted to Mongolian phonology is discussed in this section. In Cyrillic Mongolian, these words are usually spelled as in Russian, except that final unstressed vowels are deleted. If Russian loan-words were written phonemically in Mongolian, the spellings in the far right column in Table 3.2 would be used, but this is normally not done.

In these words, non-Mongolian segments are usually changed to similar Mongolian sounds, so that Russian f becomes p^h (e.g. Ru fantázija > Mo p^hant^hac ;

⁶ Vladimircov (1929: 242ff.); Luvsandèndèv (1958); Sanžeev (1959: 20); Bajarsürèn (1978); Darbeeva (1988; 1996: 130 ff.); Guo (1988); Bayarmendü (1989). *Buriad:* Bertagaev (1947); I. D. Buhaeva (1980).

fártuček > $p^{h}aarč^{h}\partial k$; *figúra* > $p^{h}igur$; *váfli* > $paap^{jh}\partial \xi^{j}$). Russian k, which does not occur in Mongolian, is sometimes retained (brážka > parašok; $knópka > konop^{h}$; poká > paka; $protokól > p^{h}ort^{h}ko\xi$; $r^{j}úmka > urumk$), but may also be changed to x (kássa > xaas; $kinó > x^{j}anv$; $kontóra > xont^{h}or$) or g/g ($kolbasá > ga\xi owsa$; $kostjúm > gošč^{h}om$; $plítka > p^{h}i\xiit^{h}og$). Although palatalized dental stops do occur in Mongolian, they are not frequent, and Russian palatalized d^{j} and t^{j} are usually rendered as č and č^h (dežúrnyj > čičur; $kostjúm > gošč^{h}om$; teátr > contraction of the stop of the stop

Russian				Mongolian	
avtomát	[efte'mat]	автомат	'slot-machine'	awt ^h mat ^h	*автмаат
brážka	['braʃkɐ]	бражка	'mash'	parašək	*бараашик
dežúrnyj	[d ^j ī'ʒurnij]	дежурный	'person on duty'	čičur	жижүүр
dráma	['drame]	драма	'drama'	taram	*дараам
fantázija	[fenˈtazije]	фантазия	'fantasy'	p ^h ant ^h ac	*пантааз
fártuček	['fartutʃɪk]	фартучек	'apron'	p ^h aarč ^h ək	*паарчик
figúra	[f ^j 1'gure]	фигура	'figure'	p ^h igur	*пигүүр
kássa	[ˈkasɐ]	касса	'cash-desk'	xaas	*xaac
kinó	[k ^j ɪˈno]	кино	'cinema'	x ^j anu	*хянуу
knópka	[ˈknopkɐ]	кнопка	'button'	konop ^h	*конооп
kolbasá	[kəlbɐˈsa]	колбаса	'sausage'	Galzəwsa	*галавсаа
kontóra	[ken'tore]	контора	'office'	xənt ^h ər	*хонтоор
kostjúm	[kes ^{j'} t ^j um]	костюм	'dress'	G⊃šč ^h ⊃m	*гошчоом
laboránt	[ləbeˈrant]	лаборант	'laboratory assistant'	よawrant ^h	*лавраант
limonád	[1 ^j 1mɐˈnat]	лимонад	'lemonade'	よ ^j amnat	*лямнаад
minúta	[m ^j 1'nute]	минута	'minute'	m ^j anʊt ^h	*мянуут
molokó	[məlɐˈko]	молоко	'milk'	məţzkə	*молкоо
pívo	['p ^j ive]	пиво	'beer'	p ^{jh} aaw	*пиав
plítka	['pl ^j itke]	плитка	'stove'	p ^h i�;it ^h əg	*пилийтэг
poká	[peˈka]	пока	'bye!'	paka	*бакаа
protokól	[prətɐˈkol]	протокол	'protocol'	p ^h ort ^h kob	*порткоол
rádio	[ˈrad ʲɪɐ]	радио	'radio'	aračəw	*араажив
rezína	[r ^j ı'zina]	резина	'rubber'	irceŋ	*эрзээн
rjúmka	['r ^j umke]	рюмка	'wine glass'	urumk	*үрүүмк
signál	[s ^j 1g'nal]	сигнал	'signal'	č ^h agna k	*чагнаал
spírt	['sp ^j irt]	спирт	'alcohol'	isp ^h irt ^h	*испийрт
škáf	[ˈʃkaf]	шкаф	'cupboard'	iškaw	*ишкаав
šljápa	[ˈʃlʲapɐ]	шляпа	'hat'	šilz ^j ap ^h	*шилиап
teátr	[t ^j 1'atr]	театр	'theatre'	č ^h aat ^h ər	*чаатар
tráktor	[ˈtraktər]	трактор	'tractor'	t ^h araxt ^h ər	*тараахтар
váfli	['vafl ^j 1]	вафли	'waffles'	paap ^{jh} əԷ ^j	*баапиль
vagón	[ve'gon]	вагон	'coach'	рэдэŋ	*богоон
vazelín	[vəz ^j ı'l ^j in]	вазелин	'Vaseline'	pack ^j iŋ	*базлийн
víza	['vjize]	виза	ʻvisa'	wiic	*вийз

TABLE 3.2 Examples of Russian loan-words in Mongolian

 $\check{c}^{h}aat^{h}\partial r$); this change is analogous to the historical change of Pre-Mongolic **t*, **t*^h to * \check{c} , * \check{c}^{h} before **i* (see 8.7.1).

The quality and place of vowels in Russian loan-words are partially decided by the Mongolian restrictions on word and syllable structure (Chapter 6). There is no length contrast in the Russian vowel system, but stressed vowels have relatively long duration. The stressed vowel in the Russian word becomes a long vowel if it occurs in the initial syllable of the Mongolian word (*fártuček* > $p^{h}aar\check{c}^{h}\partial k$; kássa > xaas; váfli > paap^{jh} ∂k^{j} ; víza > wiic), and a phonemic (short) vowel if it is noninitial ($dežúrnvi > \check{c}i\check{c}ur$; fantázija > p^hant^hac ; kinó > x^janv ; kolbasá > gak avsa; limonád > k^{i} amnat; protokól > p^{h} ort^hkok; signál > \check{c}^{h} agnak). If a Russian word begins in a way that is impossible in Mongolian, a vowel is inserted to satisfy the word structure requirements. When a word begins with r, a vowel is added before it, usually identical to the stressed vowel (rádio > aračow; $rezína > irce\eta$; rjúmka> urumk). Initial consonant clusters are broken up by an inserted vowel, usually identical to the stressed vowel (brážka > parašok; dráma > taram; knópka > $k > n > p^h$; $plitka > p^h i \xi i t^h > g$; $protok ol > p^h > r t^h k > \xi$; $traktor > t^h a raxt^h > r$); if the cluster begins with s or \check{s} , the vowel i may be inserted before the cluster (*spirt* > $isp^{h}irt^{h}$; škáf > iškaw; but $šljápa > šik^{j}ap^{h}$).

Unstressed non-initial vowels are treated according to the Mongolian rules, that is, they are deleted and then built up by an epenthesis rule (6.2). This means that inadmissible consonant clusters are broken up by epenthetic schwas ($te\acute{a}tr > \check{c}^haat^h \sigma r$), and final unstressed vowels are deleted ($dr\acute{a}ma > taram; figúra > p^higur; kontóra > xont^h \sigma r; pívo > p^{jh}aaw; víza > wiic$). If this results in an inadmissible final consonant cluster in Mongolian, a schwa is inserted ($br\acute{a}\check{z}ka > paraš\sigma k; plítka > p^hi\check{g}it^h \sigma g; váfli > paap^{jh}\sigma \xi^j$). Word-medial unstressed vowels are deleted unless they are required by the Mongolian epenthesis rule ($avtom\acute{a}t > awt^hmat^h; labor\acute{a}nt > kawrant^h; limon\acute{a}d > k^jamnat; molokó > mokko; vazelín > pack^{ji}\eta$). The same rule may also require a change of the place of a vowel ($kolbas\acute{a} > gak \sigma ws; protokól > p^hort^hko\xi$).

The historical 'breaking' process, by which initial Old Mongolian *i was assimilated to the following vowel (usually *a), and the consonant preceding it was palatalized (10.11.1), is also visible in Russian loans, at least as pronounced by people with little knowledge of Russian ($kin\delta > x^{j}an\upsilon$; $limonád > \xi^{j}amnat$; $minúta > m^{j}an\upsilon t^{h}$; $pivo > p^{jh}aaw$; $signál > c^{h}agna\xi$). Some loans from other languages have gone through 'breaking' as well, for example, $p^{jh}axt^{h}r\upsilon\xi *\Pi x \tau pyy\pi$ 'gramophone' (from the trademark *Victrola*).

The Russian stressed vowel is dominant in the loan-words into Mongolian, in the sense that it tends to decide the vowel harmony class, even when it is not found in the initial syllable. As seen above, a copy of a stressed vowel which becomes non-initial in Mongolian is often inserted as the initial vowel of the word ($rádio > aračaw; rjúmka > urumk; knópka > konop^h$), and the Russian stressed vowel usually decides the quality of initial unstressed *a* or *o*, which are pronounced in the same way in Russian (kolbasá > gabawsa; vagón > pogoŋ). In these ways, and also by eliminating initial *i* by 'breaking', the words become regular in relation to

vowel harmony (see 5.2). There is some vacillation, however, and the same Russian word may be treated in different ways by different speakers. For example, the Russian word pivo 'beer' may be pronounced as p^{hiiw} or as $p^{jh}aaw$ in Mongolian, words which belong to different vowel harmony classes, as shown, for example, by the instrumental suffix: p^{hiiwer} but $p^{jh}aawar$. Similarly, Russian kinó 'cinema' may be rendered as $x^{j}anv$ (instrumental $x^{j}anvgar$) or kino (instrumental kinogor).

WRITING SYSTEMS

Two different writing systems are presently used for writing Mongolian, the Cyrillic alphabet in the Republic of Mongolia and the Mongolian alphabet in South Mongolia in China. The Mongolian alphabet is the modern form of the Uigur Mongolian alphabet used since the thirteenth century for writing Mongolian (see 8.1).

The Cyrillic alphabet was introduced in Mongolia by a government decision of 25 March 1941. It gradually replaced the old Mongolian script and was made the only official writing system in 1946. At the end of the 1980s, the choice of writing system was widely debated in Mongolia, and in 1991 there was a decision made by Parliament that the old Mongolian script should replace Cyrillic in 1994 for all official purposes. As preparation for this, the old script was taught as the main way of writing Mongolian in the primary schools in 1991–4. This met resistance from the general public, however, and the script reform was suddenly abolished by a decision made by Parliament in July 1994. Cyrillic is now the only widely-used script in Mongolia, although the old script is still officially used beside it (Grive-let 2001). A short-lived attempt to introduce the Latin alphabet in Mongolia was made in 1930–2 (Poppe 1932*b*), and there has also been some recent discussion about this.

It was decided in 1955 that the Cyrillic script should be used by the Mongols in China as well, but this was never implemented, although some publications were printed in Cyrillic Mongolian. The use of Cyrillic Mongolian in China was abolished in 1958 (Svantesson 1991*b*).

4.1 CYRILLIC MONGOLIAN

The Cyrillic Mongolian alphabet is given in Table 4.1, together with the Cyrillic alphabets for the two Mongolic languages spoken in the Russian Federation, Buriad and Kalmuck. The conventional transliteration of Cyrillic Mongolian is based on the transliteration of Russian and is phonetically somewhat misleading, since the Mongolian pronunciation of several Cyrillic letters differs from their Russian pronunciation. For practical reasons it is used here for names, and in the bibliography. We follow the ISO transliteration of Cyrillic letters (according to ISO 9: 1995(E)) except in a few cases where it diverges considerably from the usage in the Mongolist literature. We have adopted the convenient ISO transliteration \hat{s} for III instead of the more common digraph $\check{s}\check{c}$.

			Phonemic trans	scription	
Letter		Transliteration	Mongolian	Buriad	Kalmuck
Aa	Aa	а	a, ə, Ø	a, ə	а
əə	дә	ä (ISO ấ)			ε
Бб	Бб	b	р	b	b
Вв	Be	v	w	*	W
Гг	Гг	g	G, g	g	g
$\mathbf{h}\mathbf{h}$	hh	h	_	ĥ	G
Дд	$\mathcal{I}\partial$	d	t	d	d
Ee	Ee	e	je, ji, jo, jə, j	je, jə/ ^j e	je/e
Ëë	Ëë	ë	jə, jə, j	jə, jə/ ^j ə	
Жж	Жж	ž	č	ž	*
Жҗ	Җҗ	Ž			j
Зз	33	Z	с	Z	Z
Ии	Ии	i	i, ə/ ^j ə, ^j i, ^j	^j i, ^j ə	i
Йй	Йй	j	i	i	j
Кк	Κκ	k	*	*	\mathbf{k}^{h}
Лл	Лл	1	b	1	1
Мм	Мм	m	m	m	m
Нн	Ηн	n	n, ŋ	n	n
Hң	$H_{\mathcal{H}}$	ņ	_		ŋ
Oo	Oo	0	 თ, <i>ə</i> , Ø	o, ə	0
Өө	$\Theta \theta$	ö (ISO ô)	0, ə , Ø	0	ø
Пп	Пп	р	p ^h	p^{h}	$\tilde{\mathbf{p}}^{\mathrm{h}}$
Рр	Pp	r	r	r	r
Ce	\overline{Cc}	8	S	S	S
Τт	Tm	t	t ^h	t ^h	t ^h
Уу	Vy	u	υ	υ	u
Yy	Yy	ü (ISO ù)	u	u	у
$\Phi \phi$	$\Phi \phi$	f	*	*	*
Xx	Xx	h	х	х	Х
Цц	Цų	с	c^{h}	*	c^{h}
$\mathbf{q}_{\mathbf{q}}$	q_{y}	č	č ^h	*	č ^h
Шш	Шш	š	š	š	š
Щщ	Щщ	ŝ	*	*	*
Ъъ	Ъъ	"	Ø	*	*
Ыы	Ыы	у	i	i	i
Ьь	Ьь	,	j	j	j
Ээ	Ээ	è	i, e, ə, Ø	e, ə	e
Юю	Юю	ju (ISO û)	jo, ju	ju, jʊ/ ^j u, ^j ʊ	ju
Яя	Яя	ja (ISO â)	ja, jə, j/ ^j a	ja, jə/ ^j a	ja

TABLE 4.1 Cyrillic alphabets for Mongolic languages

Note: Letters marked with the symbol — do not belong to the indicated alphabet. The asterisk symbol, *, denotes letters which occur only in loan-words. Transcriptions given after a slash (/) occur after consonants: those given before the slash occur in other positions.

In spite of frequent statements to the contrary (see e.g. Rinčen 1958–9: 130 (fn. 2)), the Cyrillic Mongolian script is well-suited to writing modern Halh Mongolian. With few exceptions it is phonemic in the sense that the spelling of a word and its phonemic form are equivalent, so that each one can be derived from the other. But there is no simple one-to-one correspondence between the letters and the phonemes. In this section, the main rules which connect the writing system to the phonemes are given.

In Cyrillic Mongolian, vowels are analysed according to model (a) in section 3.1.1, so that full (phonemic) and reduced (epenthetic) non-initial vowels are written in the same way as long and short initial vowels, respectively. Single vowel letters denote short vowels in initial syllables (1a). The letters $\pi < i >$ and 3 < i > denote the same sound *i* in initial syllables; see section 1.1. In non-initial syllables, single vowel letters denote epenthetic vowels (1b), and in word-final position they are mute (1c).

(1)	(a)	ард	<ard></ard>	art	'people'
		дов	<dov></dov>	təw	'mound'
		тул	<tul></tul>	t ^h ʊԷ	'because'
		бид	<bid></bid>	pit	'we'
		элс	<èls>	ilzs	'sand'
		хөл	<höl></höl>	хођ	'foot'
		хүн	<hün></hün>	xuŋ	'person'
	(b)	хурал	<hural></hural>	xurəb	'meeting'
		Монгол	<mongol></mongol>	məŋgəb	'Mongol'
		ажил	<ažil></ažil>	ačəb	'work'
		хүрэн	<hürèn></hürèn>	xurəŋ	'brown'
		зөвлөл	<zövlöl></zövlöl>	cowbəb	'advice'
	(c)	хана	<hana></hana>	xan	'wall'
		чоно	<čono>	čhən	'wolf'
		анги	<angi></angi>	aŋg ^j	'class'
		шөнө	<šönö>	šon	'night'
		үнэ	<ünè>	un	'price'

Mute final vowel letters serve to distinguish *n* from *y* and *G* from *g* (see (7) and (8)), and final $\mathbf{u} <\mathbf{i} >$ shows palatalization of the preceding consonant. In some words the mute letters have no function, and seem to be written just for etymological reasons, as in MOHFO <mongo mong 'silver'. Normally, the single letters $\mathbf{y} < \mathbf{u} >$ and $\mathbf{y} < \mathbf{u} >$ do not occur in non-initial syllables. Exceptions are some suffixes such as –дугаар/дүгээр <dugaar/dügèèr> 'ordinal number'.

Double vowel letters denote long vowels in initial syllables (2a), and phonemic (short) vowels in non-initial syllables (2b).

(2)	(a)	аалз	<aalz></aalz>	aabc	'spider'
		хоол	<hool></hool>	хээђ	'food'
		муур	<muur></muur>	moor	'cat'

	ээж	<èèž>	eeč	'mother'
	өөр	<öör>	oor	'self'
	үүл	<üül>	սսէյ	'cloud'
(b)	санаа	<sanaa></sanaa>	sana	'thought'
	долоо	<doloo></doloo>	tobo	'seven'
	ааруул	<aaruul></aaruul>	aarub	'dried curds'
	хүрээ	<hürèè></hürèè>	xure	'camp'
	хөдөө	<hödöö></hödöö>	xoto	'countryside'
	илүү	<ilüü></ilüü>	ilzu	'more'

The combinations ай, ой, уй, үй <aj, oj, uj, üj > denote *i*-diphthongs (3a). The combination эй <èj > writes non-initial *e* when it alternates with diphthongs in vowel harmony, for example, in the comitative suffix $-t^hai/t^hoi/t^he$ (3b). It is also used for writing non-initial *e* when the preceding syllable has *o*, as in хөөндэй <hööndèj > *xoonte* 'thrush', and also in some other words, such as хүүхэлдэй <hüühèldèj > *xuuxəbte* 'doll'.

The combination $\mathbf{n}\mathbf{\check{n}} < ij >$ denotes long ii in initial syllables and phonemic i in non-initial syllables (3c). It is used in non-pharyngeal words, and in pharyngeal words after palatalized and alveopalatal consonants.

(3)	(a)	ай	<aj></aj>	ai	'to fear'
<u> </u>		ой	<oj></oj>	oi	'forest'
		хуй	<huj></huj>	xui	'sheath'
		үйл	<üjl>	uib	'action'
	(b)	гэртэй	<gèrtèj></gèrtèj>	girthe	'house-com'
		бөөртэй	<böörtèj></böörtèj>	poorthe	'kidney-сом'
	(c)	хийд	<hijd></hijd>	xiit	'lamasery'
		морийг	<morijg></morijg>	mər ^j ig	'horse-ACC'
		зөгий	<zögij></zögij>	cogi	'bee'

The letter $\omega < y >$ writes a non-palatalizing non-initial *i*, which occurs only in inflectional suffixes, for example, optic $\langle oryg \rangle > \sigma rig$ 'place-ACC'.

The 'soft' vowel letters e, ë, io, $\pi < e$, ë, ju, ja> have several different functions. Word-initially they denote the combination of *j* and a following vowel (4a). The letters e <e> and io <ju> are ambiguous, denoting *ji* ~ *jo* and *ju* ~ *jv*, respectively. When a soft vowel is written after another vowel inside a word, it denotes *j* plus an epenthetic vowel (4b), and in word-final position the soft vowel letters (chosen so that they formally follow vowel harmony) denote *j* (4c).

jos 'nine'
jir 'ninety
jos 'rule'
> julzxgər 'fat'
jum 'thing'
jaw 'to go'

(b)	үер	<üer>	ujər	'flood'
	xoëp	<hoër></hoër>	xəjər	'two'
	аяга	<ajaga></ajaga>	ajəg	'cup'
(c)	бие	<bie></bie>	pij	'body'
	гоё	<goë></goë>	GƏj	'elegant'
	уя	<uja></uja>	υj	'to tie'

The combination of a soft and a hard vowel letter in word-initial position denotes the combination of *j* and a long vowel (5a); in non-initial position it denotes *j* plus a phonemic vowel (5b). If a word stem ends in e, ë, $\pi < e$, ë, ja> and a suffix beginning with *u* or υ is added, the stem-final letter is retained and not changed to io < ju> as might be expected (5c). In this way, the graphic identity of the stem is preserved.

(5)	(a) (b) (c)	еэвэн ёотон юүдэн яам эеэр өеө оёос оюутан туяа биеүд хоёул	<eèvèn> <ëoton> <juüdèn> <jaam> <èeèr> <öeö> <oëos> <ojuutan> <tujaa> <bieüd> <hoëul></hoëul></bieüd></tujaa></ojuutan></oëos></jaam></juüdèn></eèvèn>	jeewəŋ jəət ^h əŋ juutəŋ jaam ijer ojo əjəs əjət ^h əŋ t ^h oja pijut xəjət	<pre>'mooncake' 'lump sugar' 'hood' 'ministry' 'harmony-INST' 'tilted' 'seam' 'student' 'ray' 'bodies' (δμε <bie> pij 'body') 'both' (xoëp <hoër> xɔjər 'two') 'to cauca to lacua' (nag choic) arei 'to</hoër></bie></pre>
		хаяул	<hajaul></hajaul>	xaju k	'to cause to leave' (xaя <haja> xaj 'to leave')</haja>

There is no special letter for the phoneme *j*. It is written with a soft vowel letter when it is word-initial or preceded by a vowel (see (4) and (5) above). When *j* is preceded by a consonant, it is written with the 'soft sign' b <'> and a soft vowel letter: abbaac < av'jaas > awjas 'talent'. Sometimes the 'hard sign' b <''> is used instead: abbaac < av''jaas > .

As seen in section 3.2.1, consonant palatalization is phonemic in pharyngeal words. As in Russian, there are no separate letters for the palatalized consonants, and palatalization is indicated in several different ways. In word-final position or when a consonant follows, palatalization is denoted by the 'soft sign' $\mathbf{b} < ' >$ (6a). If there is a final cluster of two palatalized consonants, a final $\mathbf{u} < \mathbf{i} > \mathbf{i}$ swritten (6b). When an initial palatalized consonant is followed by the short vowel *a*, the soft vowel letter $\mathbf{a} < \mathbf{j} \mathbf{a} > \mathbf{i}$ swritten (6c); other short vowels do not occur in this context. The combinations $\mathbf{u}a$, $\mathbf{u}o$, $\mathbf{u}y < \mathbf{i}a$, $\mathbf{i}o$, $\mathbf{i}u >$ denote palatalization plus a long vowel in initial syllables, where they occur only after $x < \mathbf{h} > \mathbf{in}$ a few words (6d). In non-initial syllables they are more common, denoting palatalization of the preceding consonant plus a phonemic vowel (6e).

The *i*-coloured epenthetic vowel which occurs after palatalized consonants and alveopalatals is written u < i > (6f).

(6)	(a)	говь	<gov'></gov'>	GJWj	'semi-desert'
		арьс	<ar's></ar's>	ar ^j s	'skin'
	(b)	салхи	<salhi></salhi>	salz ^j x ^j	'wind'
	(c)	хяр	<hjar></hjar>	x ^j ar	'crest'
	(d)	хиам	<hiam></hiam>	x ^j aam	'sausage'
		хиоцгон	<hiocgon></hiocgon>	x ^j əəc ^h Gəŋ	'hermaphrodite'
		хиур	<hiur></hiur>	x ^j uur	'pennant'
	(e)	авиа	<avia></avia>	aw ^j a	'sound'
		мориор	<morior></morior>	mər ^j ər	'horse-inst'
		хариул	<hariul></hariul>	xar ^j ub	'to answer'
	(f)	адил	<adil></adil>	at ^j əb	'like'
		ажил	<ažil></ažil>	ačəlz	'work'

The letter r < g > writes two phonemes, g and G. In non-pharyngeal words, only g occurs (7a). In pharyngeal words, r denotes G, except in morpheme final position, where it denotes g. Morpheme final G is written with a mute final vowel (7b).

When a suffix beginning with a vowel other than i or with t^h is added to a pharyngeal stem ending in g, it is impossible to distinguish the g from a g in the writing system (7c).

The combination Gi (occurring only when a suffix beginning with *i* is added to a stem ending in *G*) is written $\Gamma \mathfrak{u} < gy >$, and gi and $g^{j}i$ are both written $\Gamma \mathfrak{u} \neq gij > (7d)$.

зөгий <zögij> cogi 'bee'</zögij>	
үг <üg> ug 'word'	
(b) гар <gar> Gar 'hand'</gar>	
зургаа <zurgaa> curga 'six'</zurgaa>	
баг <bag> pag 'team'</bag>	
бага <baga> pag 'small'</baga>	
(c) багаас <bagaas> pagas 'team–ABL' (< баг <bag> pag)</bag></bagaas>	
багаас <bagaas> pagas 'small-ABL' (<бага <baga> pag)</baga></bagaas>	
зургаа <zurgaa> curga 'picture-RFL' (< зураг <zurag> cura</zurag></zurgaa>	<i>g</i>)
зургаа <zurgaa> curga 'six'</zurgaa>	
TAFT <tagt> t^hagt^h 'cap-AR' (< TAF <tag> t^hag)</tag></tagt>	
тагт <tagt> t^hagt^h 'balcony'</tagt>	
(d) багыг <bagyg> pagig 'small-ACC' (< бага <baga> pag)</baga></bagyg>	
багийг <bagijg> pagig 'team–асс' (< баг <bag> pag)</bag></bagijg>	
агийг <agijg> ag^jig 'wormwood-ACC' (< агь <ag'> ag^j)</ag'></agijg>	

When the letter H < n > is followed by a vowel letter (which is mute in word-final position) or by a dental consonant, it denotes *n* (8a); otherwise it denotes *ŋ* (8b).

(8)	(a)	нар	<nar></nar>	nar	'sun'
		үнээ	<ünèè>	une	'cow'
		шөнө	<šönö>	šon	'night'
		хүнс	<hüns></hüns>	xuns	'food'

(b)	хаан	<haan></haan>	xaaŋ	'Khan'
	хүн	<hün></hün>	xuŋ	'person'
	хонх	<honh></honh>	хэŋх	'bell'

4.2 CYRILLIC BURIAD AND KALMUCK

The Cyrillic alphabet is also used for the two Mongolic languages, Buriad and Kalmuck, spoken in the Russian Federation. These languages are described briefly in sections 9.2 and 9.4.

The Buriad and Kalmuck pronunciations of the letters sometimes differ from the Mongolian one, and there are some letters in the Buriad and Kalmuck alphabets which are not found in Mongolian (see Table 4.1). The main difference between the three Cyrillic scripts is, however, the way in which non-initial vowels are treated. We analyse the reduced vowels in Mongolian, Buriad, and Kalmuck as being non-phonemic (except that Buriad word-final schwas are phonemic (see 9.2)), and consequently there is no long ~ short quantity distinction in non-initial syllables. Cyrillic Mongolian and Buriad write full non-initial vowels with double letters, and write reduced vowels with single letters, while Kalmuck writes full vowels with single letters, and does not write out reduced vowels. Thus our analysis agrees with the spelling for Kalmuck, but not for Mongolian and Buriad. These differences are illustrated by the following words:

		pronun-	represen-	orthog-	transliter-
		ciation	tation	raphy	ation
'red'	Mongolian:	[ʊɟ͡ɟaŋ]	/uɟaŋ/	улаан	<ulaan></ulaan>
	Buriad:	[ʊlaŋ]	/ʊlan/	улаан	<ulaan></ulaan>
	Kalmuck:	[ulan]	/ulan/	улан	<ulan></ulan>
'hero'	Mongolian:	[paːʰtăr]	/paat ^h r/	баатар	<baatar></baatar>
	Buriad:	[baːtʰăr]	/baat ^h r/	баатар	<baatar></baatar>
	Kalmuck:	[baːtʰr̪]	/baat ^h r/	баатр	<baatr></baatr>
	-	Buriad: Kalmuck: 'hero' Mongolian: Buriad:	<i>ciation</i> 'red' Mongolian: [ʊkɟŋ] Buriad: [ʊlɨŋ] Kalmuck: [ulan] 'hero' Mongolian: [paːʰtǎr] Buriad: [baːtʰǎr]	ciationtation'red'Mongolian:[υξaŋ]/υξaŋ/Buriad:[υlaŋ]/υlan/Kalmuck:[ulan]/ulan/'hero'Mongolian:[pa: ^h tăr]/paat ^h r/Buriad:[ba:thăr]/baat ^h r/	ciationtationraphy'red'Mongolian:[υξаŋ]/υξаŋ/улаанBuriad:[υlaŋ]/υlan/улаанKalmuck:[ulan]/ulan/улан'hero'Mongolian:[pa: ^h tăr]/paat ^h r/баатарBuriad:[ba:thăr]/baat ^h r/баатар

A Kalmuck orthography which writes schwa vowels was officially adopted in 1999, and codified in the dictionary by Omakaeva (2000). It was not widely accepted, however, and was abolished in 2001.

4.3 MODERN WRITTEN MONGOLIAN

The old Mongolian alphabet, which was devised as an adaptation of the Uigur alphabet in the thirteenth century, is still used by the Mongols in China. The more or less standardized modern form of the Mongolian script will be referred to as Modern Written Mongolian.

Compared to the original Uigur Mongolian alphabet, the forms of the letters have been changed somewhat, and some sounds which were originally written

(9)

with the same letter are now differentiated by diacritic dots, but otherwise the spelling has been changed only marginally. The modern form of the Mongolian alphabet is shown in Table 4.2. As in other alphabets of Semitic origin, the letters have different forms depending on their place in the word: initial, medial, or final. The old Uigur form of the Mongolian script is described in section 8.1.

The standard transcription of Written Mongolian is that of Mostaert (1941–4). We will use this with some modifications: we substitute c, j for Mostaert's \check{c} , \check{j} , which is possible because these letters without a diacritic are not used otherwise in the transcription. We also write g for Mostaert's two letters g and γ , which represent the two Mongolian letters Ω and Ω / π . They are in complementary distribution, so that γ occurs in words with back vowels (a, o, u), except before i and except in the combination $\Re g$; g occurs in all other positions. We use this transcription

		C 1		
initial	medial	final	transcription	Halh equivalents
Ŧ	٦	ب ل	а	a, ə, Ø
4	4	ب ل	e	i, e, ə, Ø
オ	1	2	i (y)	i, ə, Ø
र र वे ष्र	d	Ø	u , o	ບ, ວ, ຈ, Ø
র	P R	ወ	ü, ö	u, o, ə, Ø
	•• • •••	٠.	n	n, ŋ
ወ	ወ	Ð	b	p, w, Ø
ர	மு		р	p^{h}
\$	1		q	Х
\$ \$:4 4	1	g	G, g, Ø
う	2	1	k, g	x, g, Ø
≁ 1	41	や	m	m
44	1 F	য	1	₿
¥	+	ŧ	s	s, š
∤ :	* :		š	š
÷	ৰ ব	ৰ্	t, d	t ^h , t
પે	ч		с	c^h,\check{c}^h
1	ч		j	c, č
ч	Ч		у	j
	ที	ウ	r	r

TABLE 4.2 The Modern Written Mongolian alphabet

Note: When <i> occurs between two vowels, it is transliterated $<y>: \neg r < d < sayin>, <math>\partial (\nabla \neg < bey-e>$. There are some extra letters used in transliterations of foreign words. They include $\neg w$ or ∂ , ∂f , ∂k , $\neg ts$, $\neg z$, $\neg y$ ($\neg y$) h, $\neg z$. The letter $\neg s$ is sometimes written word-initially in foreign words, and $\neg s$ is written medially; in these cases we transliterate them as d and t: $\neg r \rightarrow r' < \neg r \rightarrow t'$ mainly for names and in the bibliography. When Old Mongolian words are discussed, they are given in (reconstructed) phonemic form; see Chapter 8.

The Mostaert transcription is not a strict transliteration, but a kind of reconstruction of the pronunciation at the time when the script was created. It makes several sound distinctions which are not indicated in the Written Mongolian script itself, based on the supposed original pronunciation, also reflected in Halh and other modern languages. As Table 4.2 shows, the vowels transcribed <u> and <o> are written with the same Mongolian letter, as are <u> and <o>. Similarly, the consonants <k> and <g> are not distinguished in writing, nor are <t> and <d>. In syllable coda position (i.e. before a consonant or word-finally), the transcription never writes <q, k, t>.

The vowel harmony alternants $\langle u/o \rangle \sim \langle \ddot{u}/\ddot{o} \rangle$ are distinguished by the script only in the first syllable, and $\langle a \rangle \sim \langle e \rangle$ only in absolute word-initial position.

After the letters \forall , \forall , \forall , \neg , C < q/g, s, m, l, r>, word-final <a/e> is usually written with the letter γ separated from the rest of the word. This is indicated with a hyphen: $\Im \neg C \gamma < qar-a> xar xap$ 'black'. The form γ also occurs without any space after $\cap < k/g>$ and $\Theta < b>$: $(\Theta \land \cap \gamma) < k\ddot{o}ke> xox x\Thetax$ 'blue'; $\Theta \land \cap \gamma < b\ddot{o}ke> pox \delta\Thetax$ 'wrestler'. As these examples also show, the vowel $\forall < u/o/\ddot{u}/\ddot{o}>$ forms the ligature $(\Theta \text{ with } \cap < k,g> \text{ and } \Theta < b>$. For more details on Written Mongolian, see Poppe (1954*a*).

Because the transcription of Written Mongolian involves some interpretation, different dictionaries and other sources differ somewhat in the transcription. This is mainly a matter of using different symbols, but one point where there is a real difference of interpretation is the choice of o/\ddot{o} or u/\ddot{u} in non-initial syllables. Mostaert and most other sources write o/\ddot{o} when these vowels are supposed to have been present in Old Mongolian, as in the word <Monggol> (or <Mong γ ol>) 'Mongol'. We follow the Mongolian–Chinese dictionary *Monggol Kitad toli* (1999). This dictionary usually writes o (or \ddot{o}) rather than u (or \ddot{u}) in non-initial syllables of a root when the word-initial syllable has o or \ddot{o} , even in some cases where the etymological vowel is u/\ddot{u} . For example, it writes *toso* 'fat', although the Old Mongolian form has *u: $*t^{h}osu/n$.

The correspondence between Written Mongolian and modern Halh pronunciation is complicated and is best understood in relation to the historical development of Halh as outlined in Chapter 10. The most common correspondences between the Written Mongolian transcription and the Halh sounds are shown in Table 4.2. 5

PHONOLOGICAL PROCESSES

The most important phonological process in Mongolian is vowel harmony, which is described in detail in section 5.2. The domain of vowel harmony is the (non-compound) word, and vowel harmony functions within roots as well as between roots and suffixes. Thus it belongs both to the basic phonology and to morphophonology. In addition to vowel harmony, some minor morphophonological processes are described in this chapter, but the other major morphophonological process, schwa ~ zero alternation, is treated in the next chapter, since it is closely connected with syllabification.

5.1 PHONOLOGICAL REPRESENTATION

One of the most important processes in Mongolic historical phonology is the shift from palatal vowel harmony in Old Mongolian to pharyngeal harmony in modern Mongolian. Using a standard vowel feature system, where the first process is described as spreading of the feature [back], and the second one as the spreading of [ATR], this historical relationship cannot be described without ad hoc rules connecting these two features, which have no natural relation to each other. Following Svantesson (1985), we have chosen to use the place features [pharyngeal], [palatal], and [velar], accepting the ideas about vowel articulation and vowel features developed by Sidney Wood (1975; 1979), based on cineradiographic data from a large number of languages. These features provide a natural articulatory explanation for the Mongolian vowel harmony shift (10.6.1). Similar vowel feature systems are used in several recent phonological theories (see e.g. van der Hulst 1987).

According to Wood, there are four places of vowel articulation (places of narrowest constriction): (1) the hard palate, (2) the soft palate, (3) the uvula and upper pharynx, and (4) the (lower) pharynx. These four vowel articulations are connected with activity in the following muscles:

 (1) Vowel articulation and muscle activity place of articulation muscles involved palatal (hard palate) genioglossi velar (soft palate) genioglossi, styloglossi uvular (uvula/upper pharynx) styloglossi pharyngeal (lower pharynx) hyoglossi and pharyngeal constrictors Following Wood we will use the three place features [palatal], [velar], and [pharyngeal] ([P], [V], [F]) for the three articulatory gestures which involve the genioglossi, styloglossi, and hyoglossi muscles, respectively. The pharyngeal constrictor muscles may be involved in pharyngeal articulation, which can be combined with the three other places of articulation, so that seven place feature combinations are possible. In addition, the features [open] and [round] ([O], [R]) are relevant for vowels. The feature [open] indicates a relatively low degree of constriction. The feature specifications for those vowels which are relevant for the Mongolic languages are given in (2). We write the symbol *a* only for non-phonemic schwa vowels; phonemic vowels with a schwa-like quality are written as *y*.

(2) Features and vowels

		_	R	0	OR
Р	palatal	i	у	e	ø
PF	pharyngealized palatal	Ι	Y	ε	œ
\mathbf{PV}	velar	u	u		
PVF	pharyngealized velar		υ		
V	uvular			Y	0
VF	pharyngealized uvular				э
F	pharyngeal			a	

We will regard all features as privative, that is as being either present or absent, rather than taking values + or -. This is natural at least for the three place features and the feature [round] since they denote the presence or absence of activity in certain muscles. Privative features (unary elements or particles) are assumed in several phonological theories, such as particle phonology (Schane 1984) and dependency phonology (J. Anderson and Ewen 1987).

The full feature specifications of the seven Halh vowels are given in (3). The features [palatal] and [velar] do not have any distinctive function in Halh, and we will normally use the underspecified feature specifications without them.

(3) Halh vowel features

	fully specified	underspecified
i	[P]	[]
u	[PVR]	[R]
υ	[PVFR]	[FR]
e	[PO]	[O]
a	[FO]	[FO]
0	[VOR]	[OR]
э	[VFOR]	[FOR]

The underspecified representations have [pharyngeal] as the only place feature, and in order to obtain the correct phonetic realizations of the vowels, the other place features must be filled in by phonetic realization rules (redundancy rules): (4) *Phonetic realization rules*

(a)	[]	>	[P]	i
	[R]	>	[PVR]	u
	[O]	>	[PO]	e
	[OR]	>	[VOR]	0
(b)	[FR]	>	[PVFR]	υ
	[FOR]	>	[VFOR]	Э

The four vowels in (4a) can be regarded as unmarked: u is the unmarked rounded vowel, e the unmarked open vowel, and o the unmarked open rounded vowel. The vowel i, which is invisible for vowel harmony (see 5.2.1), can be regarded as completely unmarked in Halh. Together with a, which can be regarded as the unmarked pharyngeal vowel, these four vowels make up the most common vowel system in the world's languages, {i, e, a, o, u}, and the rules in (4a), which create this subsystem in Halh can be regarded as universal default rules in the sense of Archangeli (1988: 35ff.). The remaining two vowels v and z are obtained by adding the feature [pharyngeal] to u and o, both in the full and the underspecified representations.

Another motivation for using the feature system in (2) is that it applies to both vowels and consonants and thus is suitable for explaining phonological processes, such as velar ~ uvular alternation in vowel harmony, which involve these two classes of sounds. For the consonants, the additional place features [labial] and [apical] are assumed:

(5)	Place features for consonants	
	labials (p^h, p, m, w)	[labial]
	palatalized labials (p ^{jh} , p ^j , m ^j , w ^j)	[labial, palatal]
	dentals (t^h , t , c^h , c , s , n , ξ , r)	[apical]
	alveopalatals $(t^{jh}, t^j, \check{c}^h, \check{c}, \check{s}, n^j, \check{\xi}^j, r^j)$	[apical, palatal]
	palatals (g^j, x^j, j)	[palatal]
	velars (g, x, η)	[velar]
	uvulars (G)	[velar, pharyngeal]

We will assume a representation where the root nodes of the segments are attached to C and V units of a timing ('skeleton') tier (6) (see e.g. Clements and Keyser 1983). In particular, long vowels and diphthongs are represented as in (6b, c).

(6)	(a) <i>t^haţx</i> талх 'bread'	(b) <i>nuur</i> нүүр 'face'	(c) <i>sims</i> оймс 'sock'
	CVCC	CVVC	VVCC
		$ \vee $	
	t ^h ађх	n u r	o i m s

5.2 VOWEL HARMONY¹

Mongolian has vowel harmony which restricts the co-occurrence of vowels within a non-compound word. It affects roots as well as derivational and inflectional suffixes, and only some recent unassimilated loan-words are excepted from vowel harmony. The main type of vowel harmony is based on the feature [pharyngeal] (cf. 5.1 above and the discussion in 1.1.1). There is also rounding harmony which is limited to open vowels.

Pharyngeal harmony divides the basic vowels into two harmony classes, nonpharyngeal and pharyngeal. If the underspecified feature specifications given in (3) above are used, the only difference between the vowel classes is the presence or absence of the feature [pharyngeal]:

(7)Vowel classes non-pharyngeal vowels pharyngeal vowels [] i [R] [FR] u σ [O] [FO] e а [FOR] 0 [OR] Э

The main rule is that vowels with different values for the feature [pharyngeal] cannot co-occur in the same word. The only exception is the vowel *i*, which has no pharyngeal counterpart. In non-initial position (i.e. when it is not the first vowel of the word) it can co-occur with any initial vowel, and following vowels harmonize with the vowel which comes before the *i*. The vowel *i* is not completely neutral, however, since it functions as a non-pharyngeal vowel when it appears in the initial syllable, forcing the vowels following it to be non-pharyngeal.

Rounding harmony is restricted to open vowels (e, o, a, o). The open rounded vowels o and o can occur in a non-initial syllable only if they are preceded by the same vowel o or o (except that i can intervene, as in pharyngeal harmony). There is an asymmetry between these two vowels, however. The vowel o cannot be followed by a, but o can be followed by e. Rounding harmony is triggered only by open vowels, and an open vowel that follows a non-open rounded vowel (u, v) must be unrounded (e or a). This gives the following possibilities for vowel sequencing in Mongolian:

(8) Vowel sequencing

Vowel in preceding syllable	Pos.	sible vowels in foll	lowing syllable
i, ii, u, uu, ui, e, ee	i	u uie	
0, 00	i	u uieo	
υ, υυ, υi, a, aa, ai	i		υ υi a ai
ວ, ວວ, ວi	i		υ υi ο οi

ı

¹ For sources and notes on vowel harmony, see App. E, p. 222.

The effects of vowel harmony are perhaps best seen in suffixes, whose vowels alternate depending on the vowels of the stem. There are five different vowel alternation patterns in suffixes, which can be represented as the phoneme /i/, the archiphonemes /E/ and /U/, and the diphthongs /Ei/, /Ui/. They are realized as shown in (9). Examples of these patterns are given in Table 5.1.

TABLE 5.1 Examples of vowel harmony in suffixes

The following suffixes are used as examples: $/-\natural E/$ 'direct past', $/-U\natural$ / 'causative', $/-xUic^{h/}$ 'potential', /-ig/ 'accusative', $/-t^{h}Ei/$ 'comitative'.

	DPST	CAUS	POTENTIAL		ACC	СОМ
'eat'	it-ke	it—ulʒ	it-xuic ^h	ʻglass'	šilz–ig	šiķ-t ^h e
	идлээ	идүүл	идхүйц	0	шилийг	шилтэй
'decide'	šiit–ţe	šiit–u lz	šiit–xuic ^h	'brush'	piir–ig	piir-t ^h e
	шийдлээ	шийдүүл	шийдхүйц		бийрийг	бийртэй
'see'	uc-lze	uc-utz	uc–xuic ^h	'axe'	sux-ig	sux-t ^h e
	үзлээ	үзүүл	үзхүйц		сухийг	сүхтэй
ʻjump'	tuuk–ke	tuuk–uk	tuub-xuich	'tail'	suu¦z–ig	suutz-the
	дүүллээ	дүүлүүл	дүүлхүйц		сүүлийг	сүүлтэй
'be stunned'	tuir–]z e	tuir–ułz	tuir–xuic ^h	'type'	cuitz–ig	cuitz-the
	дүйрлээ	дүйрүүл	дүйрхүйц		зүйлийг	зүйлтэй
'decorate'	xeeb-ze	xeelz–ulz	xeel3-xuich	'gown'	teetz-ig	teetz-t ^h e
	хээллээ	хээлүүл	хээлхүйц		дээлийг	дээлтэй
'give'	og– z o	og–u k	og–xuic ^h	'foot'	xob–ig	xotz–t ^h e
	өглөө	өгүүл	өгхүйц		хөлийг	хөлтэй
'decrease'	c ^h oor–ţo	c ^h oor–utz	choor-xuich	'kidney'	poor–ig	poor-t ^h e
	цөөрлөө	цөөрүүл	цөөрхүйц		бөөрийг	бөөртэй
'pleat'	xun ^j –]3a	xսn ^j –սԷ	xun ^j -xuic ^h	'arrow'	sum–ig	sum–t ^h ai
	хуньлаа	хуниул	хуньхуйц		сумыг	сумтай
'evaporate'	uurš–lza	uurš–u k	uurš–xuic ^h	'cat'	muur–ig	mʊʊr–tʰai
	ууршлаа	ууршуул	ууршхуйц		муурыг	мууртай
'cry'	uitz–tza	ப்த–பத	uitz–xuich	'pole'	t ^h ʊiऺʒ–ig	t ^h uitz–t ^h ai
	уйллаа	уйлуул	уйлхуйц		туйлыг	туйлтай
ʻgo'	jaw–ŀza	jaw–otz	jaw–xuic ^h	'hand'	Gar−ig	Gar−t ^h ai
	явлаа	явуул	явхуйц		гарыг	гартай
'be delayed'	saat ^h –¦za	saat ^h –vb	saath-xuich	'spider'	aalzc–ig	aabc–t ^h ai
	саатлаа	саатуул	саатхуйц		аалзыг	аалзтай
ʻbrag'	sairx– z a	sairx–ʊ॑॑₽	sairx–xuic ^h	'family'	ailʒ−ig	aib–t ^h ai
	сайрхлаа	сайрхуул	сайрххуйц		айлыг	айлтай
'enter'	ər–ţə	or–u₿	ər–xuic ^h	'place'	or–ig	ər–t ^h əi
	орлоо	оруул	орхуйц		орыг	ортой
'be pierced'	choor-bo	choor–utz	choor-xuich	'cake'	pɔɔw–ig	pəəw–t ^h əi
	цоорлоо	цооруул	цоорхуйц		боовыг	боовтой
'dart out'	c ^h əik–kə	choilz–ulz	choilz-xuich	'sock'	⊃ims–ig	⊃ims–t ^h ⊃i
	цойллоо	цойлуул	цойлхуйц		оймсыг	оймстой

Vowel alternation	patte	erns	in suj	fixe	5
Stem vowel	Sų	ffix v	owel		
	i	Е	Ei	U	Ui
i, ii, u, uu, ui, ee	i	e	e	u	ui
0, 00	i	0	e	u	ui
υ, υυ, υi, a, aa, ai	i	a	ai	υ	υi
ə, əə, əi	i	э	əi	υ	υi

The contrast between the five non-initial vowels is illustrated in (10).

(10)	/xar_i/	xari	харъя	'to look–optative'
	/xarE/	xara	xapaa	'sight'
	/xarEi/	xarai	харай	'to jump'
	/xarU/	xaru	харуу	'grit'
	/xarUi/	xarvi	харуй	'twilight'

The same restrictions that hold for suffix vowels apply to non-initial vowels in stems. Examples of possible vowel combinations in uninflected words are given in Table 5.2.

The vowel harmony examples show that a long vowel and an *i*-diphthong combine with the same vowels in the next syllable as the corresponding short vowel does. We assume that the vowels of non-initial syllables of uninflected words are represented in the same way as suffix vowels. For instance, the words *sana* canaa 'thought', *xoto* $x \theta d \theta \theta$ 'countryside', *šire* IIII $\theta \theta$ 'table', $\tilde{c}^h \upsilon \beta \upsilon \eta \eta \eta \psi$ 'stone', and *xantgai* xahdraŭ 'elk' are represented as /sanE/, /xotE/, /širE/, /č^h \varthold U/, /xantgEi/.

After the vowels *i*, *u*, *e*, the archiphoneme /E/ and the combination /Ei/ are realized in the same way, as *e*. We will write /Ei/ when *e* alternates with other realizations of /Ei/ (i.e. *ai*, *ɔi*) in suffixes, for example, in comitative /–t^bEi/. After the rounded open vowel *o*, the suffix vowel /Ei/ is realized as unrounded *e* (e.g. /poor–t^hEi/ *poort^he* бөөртэй 'kidney–COM'). The vowel *e* can also occur in stem words after *o*. We assume that *e* is the realization of /Ei/ in this case as well, so that e.g. *xoonte* xөөндэй 'thrush' is represented as /xoontEi/.

5.2.1 Transparent i

Non-initial *i* is transparent in the sense that it is completely ignored by vowel harmony. It is not affected by vowel harmony, nor does it interfere with vowel harmony. This is illustrated by comparing the realizations of reflexive -E in (11a), where they follow a suffix containing *i*, with those in (11b) where they follow the stem directly. The intervening *i* makes no difference for the realization of the final suffix; in colloquial Ulaanbaatar Halh, there is some variation in the realization of *o* in this position (see 5.2.2).

5.2 VOWEL HARMONY

pit ^h gi	битгий	'don't!'	siimxi	сиймхий	'gap'
tilgur	дэлгүүр	'shop'	šiiču	шийжүү	'gruel' (Ch)
gisgui šire	гэсгүй ширээ	'proctor' (Tib) 'table'	č ^h iič ^h e	чийчээ	'motorcar' (Ch)
tulzi	дүлий	'deaf'	t ^h uuxi	түүхий	'raw'
t ^h uru	түрүү	'ear (of corn)'	t ^h uušu	түүшүү	'back of jaws'
ugui	үгүй	'no'			
une	үнээ	'cow'	šuuge	шүүгээ	'cupboard'
guimxi	гүймхий	'frisky'	peeţsi	бээлий	'glove'
guitzgu	гүйлгүү	'rapid'	ceet ^h u	зээтүү	'mattock'
tuixuich	дүйхүйц	'equivalent' (Ch)			
tuiweŋ	дүйвээн	'noise'			
cogi	зөгий	'bee'	xoomi	хөөмий	ʻpharynx'
očun	өжүүн	'stubborn'	xooru	хөөрүү	'conceited'
xoto	хөдөө	'countryside'	xooc ^h o	хөөцөө	'heat'
xoxte	хөхдэй	'mythological bird'	xoonte	хөөндэй	'thrush'
mur ^j i	мурий	'be bent'	thuučhi	туучий	'lamenter'
č ^h υჭυ	чулуу	'stone'	υυčυ	уужуу	'wide'
tugui	дугуй	'circle'	ບບາ່ຽວກູດບ່	уурлангуй	'angry'
curga	зургаа	'six'	puux ^j a	буухиа	'courier'
c ^h urxai	цурхай	'pike'	t ^h uul g ai	туулай	'hare'
uiwsx ^j i	уйвсхий	'be unsteady'			
cuigulz	зуйгуул	'wild'			
xuixui	хуйхуй	'Chinese Muslim' (C	Ch)		
xuiţa	хуйлаа	'roll'			
uitzai	уйлхай	'weepy'			
nar ^j iŋ	нарийн	'thin'	jaačg ^j i	яажгий	'rickety' (Ch)
atu	адуу	'horse'	aarub	ааруул	'dried curds'
wantoi	вандуй	'pea' (Ch)			
jama	ямаа	'goat'	čaača	жаажаа	'auntie' (Ch)
xantgai	хандгай	'elk'	paawgai	баавгай	'bear'
maičg ^j i	майжгий	'down at heel'	məin ^j i	мойний	'to get knotty'
xairu	хайруу	'hard'	choiltu	цойлдуу	'isolated'
aisui	айсуй	'coming'			
paicha	байцаа	'cabbage' (Ch)	poito	бойдоо	'clumsy'
paimgai	баймгай	'full-sized'			
эђ ^j i	олий	'to squint'			
xočułz	хожуул	'stump'	čວວt ^h ບ	жоотуу	'pick'
sənsxui	сонсохуй	'hearing'	təəckəŋgui	дооглонгуй	'mocking'
tobo	долоо	'seven'	təəčə	доожоо	'appearance'
nəxəi	нохой	'dog'	хээдэі	хоолой	'throat'

TABLE J.2 Examples of vower narmony in unminected words	TABLE 5.2	Examples of	vowel	harmony in	uninflected words
---	-----------	-------------	-------	------------	-------------------

Note: Whenever possible, monomorphemic words are used as examples. No examples were found for some of the theoretically possible combinations, and some could be exemplified only with Chinese or Tibetan loan-words.

(11)	(a) ACC-RFL $/-ig-E/$			(b)	rfl /-E/		
		piir–ig–e	бийрийгээ		piir–e	бийрээ	'brush'
		suutz–ig–e	сүүлийгээ		suutz–e	сүүлээ	'tail'
		teetz–ig–e	дээлийгээ		teeţ-e	дээлээ	'gown'
		poor-ig-o	бөөрийгөө		poor-o	бөөрөө	'kidney'
		muur–ig–a	муурыгаа		muur-a	муураа	'cat'
		chaas-ig-a	цаасыгаа		c ^h aas–a	цаасаа	'paper'
		xəəb–ig–ə	хоолыгоо		xəəb-ə	хоолоо	'food'

The fact that *i*-diphthongs function in the same way as the corresponding basic vowel in vowel harmony, can be regarded as a special case of the fact that any non-initial i is ignored by vowel harmony, since *i*-diphthongs are represented as sequences of two vowels (see (6) above).

There are rather few monomorphemic pharyngeal words with *i* as the vowel of the second syllable. Many of these are 'expressive' verbs, intransitive verbs which describe how something looks or sounds (12). It is possible that *i* should be regarded as a suffix in these words, although with a very vague meaning. There are also a few other words of this type, such as $nar^{j}i\eta$ нарийн 'thin'. These words take suffix vowels whose vowel harmony class depends on the first vowel, and the *i* is ignored. This is exemplified with the direct past in (12).

(12)	verb		direct past		
	t ^h ax ^j i	тахий	t ^h ax ^j i–5a	тахийлаа	'to be bent'
	jaŋx ^j i	янхий	jaŋx ^j i–Էa	янхийлаа	'to be bony'
	guwč ^h i	гувчий	συwč ^h i− β a	гувчийлаа	'to be hollow'
	∋β ^j i	олий	∋β ^j i–β∋	олийлоо	'to squint'

The *i* in these words (written $\mu\mu$ in Cyrillic Mongolian) is always preceded by a palatalized or alveopalatal consonant (i.e. one with the feature [palatal], see (5) above). This appears to be the only case where an *i* palatalizes a preceding consonant in modern Halh Mongolian, all other applications of such rules being historical, since the relevant environments have been eliminated by other phonological rules (see 10.5 and 10.7). Non-palatalizing *i* (μ in Cyrillic Mongolian) occurs only in inflectional suffixes (accusative -ig, genitive $-i\eta$, -ni, and the optative (see 6.5)). This seems to be the only phonological property that makes a difference between inflectional and derivational suffixes in Halh (see 8.8 for a similar case in Old Mongolian).

5.2.2 Opaque vowels

The non-open vowel /U/(u or v) is opaque in the sense that it blocks rounding harmony, and following open vowels must be unrounded. For instance, when the direct past $-\xi E$ is added after the causative $-U\xi$, E is realized as an unrounded vowel (*e* or *a*), irrespective of whether the stem has rounded vowels or not (13).

Examples where the direct past is added directly to the stem are given for comparison.

(13) CAUS_DPST /_[]b_-bE/

3)	CAUS-DPST /	U��–��E/	dpst /E/				
	it–utz–tze	идүүллээ	it-be	идлээ	'to eat'		
	uc–utz–tze	үзүүллээ	uc-ze	үзлээ	'to see'		
	xeetz-utz-tze	хээлүүллээ	хееђ-ђе	хээллээ	'to decorate'		
	og–utz–tze	өгүүллээ	og-zo	өглөө	'to give'		
	xun ^j –ub–ba	хуниуллаа	xun ^j –ţa	хуньлаа	'to pleat'		
	jaw–u¦z–ka	явууллаа	jaw-15a	явлаа	'to go'		
	or−uӄ−ӄa	орууллаа	ər-zə	орлоо	'to enter'		

When Ei is realized as e, it blocks rounding harmony, as seen in (14), where comitative $-t^{h}Ei$ is combined with reflexive -E (an epenthetic consonant g/g is inserted if the stem ends in a vowel (see 5.2.5)). A crucial example is *poor*- $t^{h}e$ -ge бөөртэйгээ 'kidney-сом-RFL', where the realization of the last vowel is unrounded e, as in the comitative suffix, and not rounded o, as in the root. The spelling norm, as represented by the orthographic dictionary by Damdinsürèn and Osor (1983), prescribes forms with a rounded open vowel in these words (бөөртэйгөө poortego, etc.), but such forms are not common in colloquial Ulaanbaatar speech. Similarly, words like xoonte хөөндэй 'thrush', with the phonological representation /xoontEi/, usually take the unrounded forms of open suffix vowels, for example, reflexive /xoontEi_E/ xoontege хөөндэйгээ.

(14)	сом–rfl /–t ^h Ei	—Е/	rfl /E/		
	piir-t ^h e-ge	бийртэйгээ	piir–e	бийрээ	'brush'
	suutz-t ^h e-ge	сүүлтэйгээ	suutz–e	сүүлээ	'tail'
	teetz-the-ge	дээлтэйгээ	tee _b –e	дээлээ	'gown'
	poor-t ^h e-ge	бөөртэйгээ	poor–o	бөөрөө	'kidney'
	muur–t ^h ai–ga	мууртайгаа	muur–a	муураа	'cat'
	c ^h aas–t ^h ai–Ga	цаастайгаа	chaas-a	цаасаа	'paper'
	xəə�–t ^h əi–Gə	хоолтойгоо	xəək-ə	хоолоо	'food'

In the spoken language of Ulaanbaatar, i may block rounding harmony. This is more common for o than for o. There is some variation, and the pronunciation seems to depend on the distance between the initial rounded vowel and the harmonizing one. In words like /jornxikgčh-E/ 'president-RFL', the most common pronunciation is *joranxikagč^he* and not *joranxikagč^ho*, although the normative spelling is ерөнхийлөгчөө. In forms like /poor-ig-E/ 'kidney-ACC-RFL', spelled бөөрийгөө, there is variation between *poorige* and *poorigo*. Some words, like /oor-ig-E/ өөрийгөө 'self-ACC-RFL', are usually pronounced with o (oorigo). This needs to be investigated more, and for the time being we will use the normative forms with rounded vowels in our analysis. Since there is no contrast between rounded and unrounded vowels in this environment, some variation is perhaps to be expected.
5.2.3 The domain of vowel harmony²

The phonological domain of vowel harmony is the non-compound word, and the constituents of compounds are treated as separate words from the point of view of vowel harmony. Thus, suffix vowels always harmonize with the last element of a compound, for example, with *cui* 3YH in $aw^{j}a=cui$ aBHa3YH 'phonetics' ($aw^{j}a$ 'sound', *cui* '-ology'), and with $m \sigma r^{j}$ морь in $xii=m\sigma r^{j}$ хийморь 'good luck' (xii 'air', $m\sigma r^{j}$ 'horse'). For example, the reflexives of these two words are $aw^{j}a=cui-ge$ aBHa3YH' σ and $xii=m\sigma r^{j}-\sigma$ хийморио, respectively.

In a noun phrase, case and reflexive suffixes are invariably added to the last element of the phrase, and there is no agreement within the noun phrase. The suffix follows harmonizing features of the word it is attached to, irrespective of its syntactic role in the noun phrase:

(15)	eeč–es mother–ABL	ээжээс	'from mother'
	aaw–as father–ABL	ааваас	'from father'
	eeč aaw–as mother father–ABL	ээж ааваас	'from mother and father'
	aaw eeč–es father mother–ABL	аав ээжээс	'from father and mother'
	eeč aaw xɔjr–ɔs mother father two–ABL	ээж аав хоёроос	'from mother and father'
	aaw pur-es father each-ABL	аав бүрээс	'from each father'

The negation -gui and the indirect past tense suffix $-\check{c}e/\check{c}^{h}e$ have no vowel alternation (16), and thus do not conform to the vowel harmony rules (the alternation between \check{c} and \check{c}^{h} in the indirect past is partly phonologically, partly lexically determined). It can be noted that unlike other verb suffixes, these two were written as separate words in Classical Mongolian, $\ddot{u}gei$ and $juqui/j\ddot{u}k\ddot{u}i$.

(16)	5) verb stem		negated impe	negated imperfect participle		indirect past		
	it	ид	it-e-gui	идээгүй	it–če	иджээ	'to eat'	
	uc	үз	uc-e-gui	үзээгүй	uc–če	үзжээ	'to see'	
	xeeţ	хээл	xeetz-e-gui	хээлээгүй	xeeţ–če	хээлжээ	'to decorate'	
	og	өг	og–o–gui	өгөөгүй	og–č ^h e	өгчээ	'to give'	
	xun ^j	хунь	xon ^j –a–gui	хуниагүй	xun ^j –če	хуньжээ	'to pleat'	
	jaw	яв	jaw–a–gui	яваагүй	jaw–če	явжээ	'to go'	
	or	op	or–o–gui	ороогүй	or−če	оржээ	'to enter'	

The indirect past cannot be followed by other suffixes. When another suffix is added after -gui, it takes non-pharyngeal vowels even if the stem has pharyngeal

vowels, for example, $a\check{c}\partial\xi$ -gui- \check{c}^hut [work-NEGATION-PL] ажилгүйчүүд 'the unemployed (PL)'; $t^hagnai-\check{s}-\upsilon\xi-t\partial g$ -gui-ger-e [palate-VR-CAUS-HABITUAL-NEGATION-INST-RFL] тагнайшуулдаггүйгээрээ 'since [it] is usually not palatalized' (from Luvsanvandan 1975c: 12).

A verb-forming suffix -xi, identical with the verb xii хий 'to do', is found in a number of verbs, exemplified in (17). These verbs take suffixes whose vowels harmonize with the *i*, unlike the verbs in (12).

(17)	verb		direct past		
	amsxi	амсхий	amsxi–ţe	амсхийлээ	'to take a rest'
	Gətxi	годхий	G⊃txi–₿e	годхийлээ	'to pop out'
	jarxi	ярхий	jarxi–ţe	ярхийлээ	'to bang'
	togxi	дугхий	tugxi-ke	дугхийлээ	'to take a nap'

If the domain of vowel harmony is taken to be the non-compound word, the suffixes -xi, $-\check{c}e$ and -gui function as compound-forming elements (=xii, $=\check{c}ee$, =gui) rather than derivational or inflexional suffixes. This is partly supported by the prosodic behaviour of the negation =gui, which, unlike suffixes, can take the rising tone signalling focal accent (see 7.3).

5.2.4 Vowel harmony as feature spreading

The standard analysis of vowel harmony is autosegmental spreading of the relevant feature over a domain (see e.g. van der Hulst and van de Weijer (1995)). In Mongolian, the domain is the non-compound word, and the spreading features are [pharyngeal] and [round]. Since Mongolian is exclusively suffixing, and a root determines the vowel harmony class of its suffixes, the direction of spreading is from left to right.

We assume that the initial vowel of a word is specified as in (7) above. The vowels that occur in the phonemic representation of non-initial syllables, /i, U, E/, have the feature specifications [] (zero), [round], and [open], respectively. For the two archiphonemes, these are the features which are common to the vowels in the respective alternation class {u, v} and {e, a, o, s}. The vowel /i/ is not changed by vowel harmony, /U/ gets its value for the feature [pharyngeal] from vowel harmony, and /E/ gets the values for both [pharyngeal] and [round]:

	realization in				
vowel and feature	non-pharyngeal words		pharyngeal words		
specification	unrounded	rounded	unrounded	rounded	
E [O]	[O] e	[OR] o	[OF] a	[ORF] o	
U [R]	[R]] u	[RF	7] ບ	
i []	[] i				

ı.

Using this analysis, pharyngeal harmony can be regarded as spreading of the feature [pharyngeal] from the first vowel throughout the word. If the initial vowel does not have this feature, no spreading takes place. The set of target segments to which pharyngeal vowel harmony spreads consists of all vowels except *i*, which, as seen in section 5.2.1, is transparent and unaffected by vowel harmony. According to the analysis of van der Hulst and van de Weijer (1995: 508), the target segments are those segments which have a V-place node. The absence of place features in the representation of *i* indicates that it does not have any place node, so that vowel harmony cannot spread to it (or spread from initial *i*).

Rounding harmony can similarly be analysed as spreading of the feature [round], but the target segments are only the open vowels, and the non-open rounded vowel U blocks rounding harmony. One possible explanation for this is that the feature [round] spreads from an initial open rounded vowel, and the spreading is stopped by a vowel which is inherently associated to the feature [round] due to the general constraint in autosegmental phonology that association lines cannot cross (19a). It is, however, difficult to explain why the feature [round] does not spread from an initial non-open rounded vowel (u or v) except by an ad hoc rule (see (19b)).

(19) (a) thosob-ub-be төсөөлүүллээ 'to imagine-CAUS-DPST'



(b) ис-де үзлээ 'to see-dpst'



If one of the features [pharyngeal] or [round] is spread in a word, it is always present in the initial vowel. This suggests that these features are associated to the initial vowel in the phonological representation of root morphemes as in (20a), and not associated only with the morpheme as in (20b). The representation in (a), but not the one in (b), excludes pharyngeal words with initial *i*. See section 10.6.2 for a comparison with Old Mongolian vowel harmony, where *i can occur initially in both front- and back-vocalic words.



5.2.5 Velar ~ uvular alternation and epenthetic consonants³

As mentioned earlier, velars and uvulars alternate, so that the uvulars $[N, \chi, G]$ occur only in pharyngeal words, and the velars $[\eta, x, g]$ only in non-pharyngeal words. The only exception is *g*, which may occur in morpheme-final position in pharyngeal words and may contrast with *G* (see 3.2.2).

When a suffix beginning with a vowel is added to a stem ending in a vowel, an epenthetic consonant is added to avoid hiatus. This consonant is g in non-pharyngeal and G in pharyngeal words. When it occurs before i, however, it is g in pharyngeal words as well, even though the combination Gi is possible, as seen in section 3.2.2. Examples are given in (21a), with examples of consonant-final stems (21b) for comparison.

(21)	(a)	/xuu/	xuu	хүү	'boy'
		/xuu–Er/	xuuger	хүүгээр	'boy-inst'
		/xuu—iŋ/	xuugiŋ	хүүгийн	'boy-gen'
		/sana/	sana	санаа	'thought'
		/sana–Er/	sanagar	санаагаар	'thought-INST'
		/sana—iŋ/	sanagiŋ	санаагийн	'thought-GEN'
	(b)	/ar/	ar	ap	'back'
		/ar_Er/	arar	apaap	'back-INST'
		/ar—iŋ/	ariŋ	арын	'back-GEN'

In the place feature analysis given in (5), velars have the specification [velar] and uvulars [velar, pharyngeal]. The feature [velar] is redundant for the Halh vowels, and it is possible to specify the consonants as well without referring to it, so that velars are unmarked consonants without any place feature, and uvulars are specified as [pharyngeal]. If consonant epenthesis consists of the addition of a consonant without place features, the velar appears under this analysis, and it becomes uvular by the spreading of [pharyngeal] in pharyngeal words (22).

(22) (a) /unE-E/ une-ge үнээгээ 'cow-RFL'

Vn V–V	\rightarrow	VnVCV
[R][O][O]		[R][O][O]

³ Kotwicz (1936); Ramstedt (1957: 170ff.); Rialland and Djamouri (1984).

(b) /sanE-E/ sana-Ga санаагаа 'thought-RFL'



5.2.6 Cyclic vowel harmony

In section 5.2.4 we gave the vowel harmony rules as if the harmonizing features spread to the target segments of a word in one application of autosegmental spreading after all suffixes have been added, so that vowel harmony is postcyclic or even postlexical in the terminology of lexical phonology (see e.g. Cole 1995).

There is, however, evidence that Mongolian vowel harmony is cyclic, so that vowel harmony spreading must take place after each addition of a suffix. The reason for this is the segmental rule that converts *ei* and *oi* to *e*, but preserves *ai* and *oi*; or in an alternative formulation, realizes /Ei/ as *e/e/ai/oi* depending on vowel harmony; see (9) above.

First, rounding harmony does not spread through /Ei/ when it is realized as e (23a), but it spreads through its realization σi (23b):

(23)	(a)	/mor_t ^h Ei_E/	morthege	мөртэйгээ	'way-com-rfl'
		/xoontEi_E/	xoontege	хөөндэйгээ	'thrush-RFL'
	(b)	/ər-thEi-E/	ərthəigə	ортойгоо	'place-com-rfl'
		/nəxEi–E/	nəxəigə	нохойгоо	'dog-rfl'

If rounding harmony were to apply after all morphological operations, the result would be incorrect for words like those in (23a), but correct for those in (23b); G denotes the alternation $g \sim G$ in suffixes.

(a)	representation	mor-t ^h Ei-E
	consonant epenthesis	mor-thEi-GE
	spreading of [R]	mor-thoi-Go
	segmental rules	*mor-the-go
(b)	representation	or−t ^h Ei−E
	consonant epenthesis	or−t ^h Ei−GE
	spreading of [FR]	∋r−t ^h ∋i−G∋
		spreading of [R]segmental rules(b) representationconsonant epenthesis

Instead, cyclic application of rounding harmony is necessary in order to explain why /Ei/ blocks rounding harmony when it is realized as e (25).

(25)	stem		mor
	Cycle 1:	COM	mor-t ^h Ei
		spreading of [R]	mor-t ^h oi
		segmental rule	mor-the

Cycle 2:	RFL	morthe-E
	consonant epenthesis	morthe-GE
	no spreading	mort ^h e-ge

The realization of /Ei/ as *e* is dependent on the fact that [pharyngeal] has not spread to /Ei/, since the pharyngeal realizations of /Ei/ are *ai* and *si*. Pharyngeal harmony spreading must thus precede the /Ei/ realization rule, and this proves that not only rounding harmony, but also pharyngeal harmony, is cyclic.

Vowel epenthesis (Chapter 6) is independent of vowel harmony, and schwa vowels in non-initial syllables are transparent to vowel harmony (as was pointed out by Saitô 1984):

(26) /aw-sŋ-t-E/	[awsănta]	авсандаа	'to take-pstp-dat-rfl'
/aw–č ^h x–ţE/	[awč ^h ĭx��a]	авчихлаа	'to take-intensive-dpst'
/sur_sŋ_t_E/	[sʊrsʊ̆nta]		'to study-PSTP-DAT-RFL'
/sur_č ^h x_BE/	[surč ^h ĭx�a]	сурчихлаа	'to study-intensive-dpst'

As seen in sections 1.1 and 3.1.1, the quality of epenthetic vowels is determined by a rule which depends on the preceding vowel (or consonant, if it is palatal). It will be shown in Chapter 6 that the rule governing the insertion of epenthetic vowels is cyclic, like vowel harmony. The fact that the epenthetic vowels are transparent to vowel harmony can be explained by ordering the epenthesis rule after the vowel harmony rule. The rule which determines the quality of epenthetic vowels can be regarded as postcyclic.

5.3 VELAR NASAL ASSIMILATION

When a suffix beginning with a dental consonant (e.g. comitative $-t^hEi$, dative $-t/t^h$, plural -s, or the adjectivizing suffix $-t^h$) is added to a word ending in the velar nasal η , this consonant is assimilated to the dental and becomes n:

(27)				comitative		dative	
	xaaŋ	хаан	'Khan'	xaant ^h ai	хаантай	xaant	хаанд
	xuŋ	хүн	'person'	xunt ^h e	хүнтэй	xunt	хүнд
	xuuxəŋ	хүүхэн	'girl'	xuuxənt ^h e	хүүхэнтэй	xuuxənt	хүүхэнд

When a suffix beginning with a vowel is added to a word ending in η , it also changes to n (28). This is a consequence of the fact that the velar nasal never occurs in syllable onsets.

(28)	xaaŋ	хаан	'Khan'	instrumental:	xaanar	хаанаар
	xuŋ	хүн	'person'		xuner	хүнээр
	xuuxəŋ	хүүхэн	ʻgirl'		xuuxner	хүүхнээр

This also changes the pronunciation of the vowel: it is heavily nasalized before η , which is often realized only by this nasalization; but *n* remains consonantal and the preceding vowel is only slightly nasalized (see 2.3).

5.4 REDUPLICATION

There are two reduplication processes in Mongolian which interact with the phonology, one which applies to adjectives and one to nouns. They are described here because of their interest as providing arguments for the phonological analysis. This section is based on Svantesson (1997); the same kind of analysis has been made by Mikami (1983), Kubo (1997), and Song Chae-mok (1999*b*).

5.4.1 Adjective reduplication

This kind of reduplication is used for intensifying some adjectives. In Halh it is applied only to some twenty or thirty adjectives. Examples are given in (29).

(29)	Exc	amples of a	djective redup			
		word			reduplicated fo	rm
	(a)	xar	xap	'black'	xaw xar	хав хар
		XOX	хөх	'blue'	XOW XOX	хөв хөх
		nəgəŋ	ногоон	'green'	now nogoŋ	нов ногоон
	(b)	utzan	улаан	'red'	uw ukan	ув улаан
		iţxəŋ	ИЛХЭН	'clear'	iw ilʒxəŋ	ИВ ИЛХЭН
	(c)	poorəŋxi	бөөрөнхий	'round'	pow poorəŋxi	бөв бөөрөнхий
		xuit ^h əŋ	хүйтэн	'cold'	xuw xuit ^h əŋ	хүв хүйтэн

The most common case is that a reduplicative prefix is formed by adding the coda w to the initial CV combination (29a). If the base begins with a vowel, only this vowel and w is prefixed (29b).

This can be analysed in the framework of non-concatenative morphology (see e.g. Marantz 1982) by assuming that a CVC prefix is added, where the last C is prespecified as *w*. The segments of the base are copied and associated to the unassociated segment slots of the prefix. Unassociated segments are deleted by 'stray erasure':



These derivations are straightforward assuming that the association goes from left to right and is 'segment-driven', so that it starts with the leftmost segment finding a suitable slot, and then continues with the next segment, until the slots (or the segments) are exhausted. Left-to-right, segment-driven reduplication is regarded as the default case by Marantz (1982: 447). Nevertheless, this type of reduplication has something to tell about the representation of long vowels and diphthongs in Mongolian. The examples in (29c) show that vowel length does not transfer to the reduplicative prefix, and that only the first element of a diphthong is transferred. Thus, long vowels and diphthongs are not treated as units, but as segment sequences, so reduplication supports the representation of long vowels and diphthongs assumed in (6). The derivations are shown in (31):

In Eastern Mongolian dialects, including Horchin (9.1), the *i*-diphthongs have developed to monophthongs. In Horchin, the word 'cold' is [xyyt^hən], which is reduplicated as [xyp xyyt^hən] (Coyijongjab p.c. June 1998). This indicates that the vowel corresponding to Halh *ui* must be represented as *yy* even at the phonological level in this dialect. According to Mönggöngerel (1998: 85), adjective reduplication in the Naiman dialect of Mongolian transfers vowel length: *xiit^hən* 'cold' reduplicates as *xiip xiit^hən*.

5.4.2 Noun reduplication

This kind of reduplication, which is phonologically more interesting, can be applied to any noun (including proper names and nominal forms of verbs). It has a meaning like 'X and such things', 'X and people like him/her' and is slightly pejorative or at least shows an indifferent or disrespectful attitude. Examples are given in Table 5.3. Similar reduplication processes are common in other Central and West Asiatic languages.

Some lexicalized reduplications are used for certain words, but noun reduplication is usually formed regularly by reduplicating the entire word, changing the initial consonant to *m* and attaching it to the base as a suffix (Table 5.3a). If there is no initial consonant, the *m* is just added (b), and if the initial consonant is *m*, it is changed to *c* (c). The affricates c^h , \dot{c}^h , *c*, \dot{c} and the voiceless lateral i are reduplicated as if they were single consonants (d), indicating that this analysis, which was assumed in Chapter 3, is the correct one.

Reduplication is also relevant for the problem how to analyse the initial combinations [xua] ~ [xwa] and [gua] ~ [gwa] which occur in a number of Chinese loans (see 3.1 above). If initial clusters (xw-, Gw-) are ruled out, there are two possibilities, either to assume labialized consonants x^w , G^w , or to assume a diphthong va, which does not occur otherwise in the language. In the reduplication, only x and G are changed (Table 5.3e), indicating that the diphthong analysis should be chosen, as was done in section 3.1. It can be noted that the combination of initial m and the diphthong va does not occur elsewhere in the language.

Interesting facts are provided by the interaction of noun reduplication and consonant palatalization. Since palatalized consonants are analysed as unitary segments, one might expect that they are changed to *m* in reduplication. This is, however, not the case: the initial consonant of the reduplicative prefix is palatalized m^{j} , not plain m (Table 5.3f). This raises the question of whether palatalized consonants should be represented as sequences of plain consonants and i, so that for instance the word 'baby' should be represented as $nia\xi x$ rather than $n^{j}a\xi x$. A representation with i is undesirable, since it is necessary to recognize the existence of palatalized consonant phonemes (3.2.1), and furthermore, this representation would introduce several *i*-initial diphthongs not found elsewhere in the language. Reduplication itself provides another argument, since words beginning with palatalized m^{j} reduplicate with the initial consonant c not followed by any i element (Table 5.3g). There is no palatalized counterpart to the dental affricate c, so a reduplication pattern like $*m^{j}a\eta c c^{j}a\eta g$ is impossible, and this blocks palatalization from being transferred to the reduplicative suffix. On the other hand, if the diphthong analysis were adopted, nothing except an ad hoc rule would prevent non-existing reduplication patterns such as $*mia\eta g cia\eta g$.

	word			reduplicated form	
(a)	t ^h a৳x	талх	'bread'	t ^h ałyx małyx	талх малх
	teeb	дээл	'gown'	teetz meetz	дээл мээл
	Gəiməŋ	гоймон	'noodles'	gəiməŋ məiməŋ	гоймон моймон
	nut	нүд	'eye'	nut mut	нүд мүд
(b)	ar	ap	'back'	ar mar	ар мар
	ontəg	өндөг	'egg'	ontəg montəg	өндөг мөндөг
(c)	mab	мал	'cattle'	malz calz	мал зал
	miţxi	мэлхий	'frog'	mitzi citzi	мэлхий зэлхий
(d)	chai	цай	'tea'	c ^h ai mai	цай май
	č ^h ʊԷʊ	чулуу	'stone'	č ^h ugu mugu	чулуу мулуу
	cam	зам	'road'	cam mam	зам мам
	čiţ	жил	'year'	čig mig	жил мил
	łagw	лхагва	'Wednesday'	łagw magw	лхагва магва
(e)	guane	гуанз	'restaurant'	guanc muanc	гуанз муанз
	xuar	хуар	'flower'	xuar muar	хуар муар
(f)	p ^j as z əg	бяслаг	'cheese'	p¹as॑॑zəg m¹as॑zəg	бяслаг мяслаг
	n ^j al j x	нялх	'baby'	n ^j al z x m ^j alzx	нялх мялх
	x ^j aam	хиам	'sausage'	x ^j aam m ^j aam	хиам миам
(g)	m ^j agmər	мягмар	'Tuesday'	m ^j agmər cagmər	мягмар загмар
	m ^j aŋG	мянга	'thousand'	m ^j ang cang	мянга занга
(h)	juŋgər	юнгар	'mustard'	jungər mungər	юнгар мунгар
	jor	ëp	'omen'	jər mər	ёрмор
	jir	ep	'ninety'	jir mir	ер мэр
	jeewəŋ	еэвэн	'mooncake'	jeewəŋ meewəŋ	еэвэн мээвэн
	juutəŋ	юүдэн	'hood'	juutəŋ muutəŋ	юүдэн мүүдэн
	jos	ec	'nine'	jos mos	ес мөс
(i)	jas	яс	'bone'	jas m ^j as	яс мяс
	, ja ¦ 3a	ялаа	ʻfly'	jaga m ^j aga	ялаа мялаа

TABLE 5.3 Examples of noun reduplication

A final problem is the reduplication of words with initial *j*. The expected pattern would be that *m* is substituted for *j* in the reduplicative suffix. This is in fact the case (Table 5.3h), except when the vowel following *j* is *a*. In that case, palatalized m^j occurs (i). This can be partially explained by the distribution of palatalized consonants, which do not occur in words with non-pharyngeal vowels (*i*, *e*, *u*, *o*). Thus a reduplication pattern like **jos* $m^j os$ is phonotactically impossible, but this does not rule out patterns like **jor* $m^j or$ with a pharyngeal vowel. Although this case is difficult to explain, it can be noted that word-initial palatalized consonants occur almost exclusively before *a*. In those cases where the historical development can be expected to result in a palatalized consonant before υ or υ , a plain consonant occurs instead, for instance in $n\upsilon\upsilon$ Hyy 'to hide' (from Old Mongolian **nihu*) (see 10.11.1).

Mongolian reduplication also raises the question of which units are involved in reduplication processes. Adjective reduplication is a textbook example of a process that can be analysed with a CV template, but noun reduplication requires other units. On the one hand, the entire word is reduplicated, but on the other, the process needs access to the internal structure of palatalized consonants in order to produce the reduplication patterns for palatalized consonants. This suggests that it is necessary to represent palatalization on a separate tier in the phonology.

6

SYLLABIFICATION AND EPENTHESIS

This chapter deals with the formation of surface syllables, which is closely connected with epenthesis. Syllables are surface phenomena, and an epenthetic vowel is a syllable nucleus although it does not occur in phonological representations. Thus, in this chapter the term 'vowel' (the symbol V) refers both to phonemic and epenthetic (non-phonemic) vowels. A dot (.) denotes syllable boundary. The chapter is based mainly on Svantesson (1988*a*; 1994; 1995*a*).

The description of vowel epenthesis and syllabification is based on careful lexical pronunciation of words said in isolation. When a word occurs in a sentence, or when the speech tempo is faster, epenthetic and sometimes even phonemic vowels may disappear depending on the speech rhythm, and there is some individual variation between speakers as well. This remains to be investigated, however. Occasional examples of this variation can be seen in the speech material analysed in Chapter 7 and elsewhere.

6.1 SYLLABLE STRUCTURE¹

Mongolian surface syllables have the structure (C)V(V)(C)(C)(C). The syllable will be regarded as consisting of an onset (the initial consonant) and a rhyme, which in its turn consists of a vowel nucleus and a coda (the final consonants). The onset and coda may be absent. The marginal triphthong *vai* is disregarded in this chapter.

(1) Mongolian surface syllable structure



¹ Sühbaatar (1973); Bajčura (1974); Jôo (1976*a*); Shimizu (1980); Dobo (1982); Ebleltü (1992); Svantesson (1994); Xiao (1995); Kökebars (1996); Köke (1998*b*); Bernhardt (1999); Song Chae-mok (1999*a*).

We will assume that words are divided into syllables in such a way that a consonant which is followed by a vowel is an onset, and does not belong to a coda, for example, *sa.na* caHaa 'thought', *xaf.xa* xaJxaa 'shield–RFL' (not **san.a*, **xafx.a*). This kind of syllable division is regarded as uncontroversial in the phonological (but not in the phonetic) literature (see e.g. Clements and Keyser 1983: 37). For Mongolian, a rule that divides syllables in this way is given explicitly by Todaeva (1951: 39–40), and can also be inferred from examples given in works by other native Mongolian scholars, such as Sanžeev (1959: 18) and Cenggeltei (1979*a*: 149). This syllable division is also supported by the fact that the only consonant strings that can occur between vowels in surface forms are those that consist of a possible word-final surface consonant cluster plus one consonant which may occur word-initially.

Due to the agglutinative nature of the language, a word-final coda cluster can usually be made word-medial by adding a suitable syllable-forming suffix, for example, comitative case $-t^{h}Ei$ (i.e. $-t^{h}ai/t^{h}oi/t^{h}e$) for nouns or adjectives, or indirect past tense $-\xi E (-\xi a/\xi o/\xi e/\xi o)$ for verbs. For instance, the word-final coda ξst^{h} in $i\xi st^{h}$ элст 'sandy' becomes word-internal after the addition of the comitative suffix: $i\xi st^{h}.t^{h}e$ элсттэй. For this reason, codas are normally exemplified in wordfinal position this chapter.

6.1.1 Syllable types

Table 6.1 shows those syllable types that can occur in different positions in a word, differentiated according to the number of consonants and vowels, and also according to the type of the vowel (short, long, *i*-diphthong, or epenthetic vowel). As seen in the table, any vowel type can combine with any permissible number of following tautosyllabic consonants (zero, one, two, or three).

The number of possible syllable types is largest in word-initial position, where long and short vowels contrast, and where syllables may or may not have an onset. Epenthetic vowels do not occur in initial syllables. A monosyllabic word can consist of any type of syllable that can be an initial syllable of a polysyllabic word, except that words of the type (C)V (where V is a short vowel) do not occur (see also 6.6).

Non-initial syllables always have an onset, and under the analysis adopted here, their nucleus may be a (short) phonemic vowel, a diphthong, or an epenthetic schwa vowel. The only difference between word-internal and word-final syllables is that an epenthetic vowel cannot occur in absolute word-final position.

Word-initial syllables may lack an onset, but onset-less syllables are not allowed inside words. If a suffix beginning with a vowel is added to a word ending in a vowel, an epenthetic consonant, G or g depending on the vowel harmony class, is inserted between them to avoid hiatus (see 5.2.5).

Any consonant may be a surface syllable onset, except η (and the marginal phoneme w^{j}). If η becomes an onset as the result of a morphological operation, it is changed to *n* (see 5.3).

TABLE 6.1	Syllable types
	- Jerre - Jree

(a)	Monosyllab	ic words		
(/	CVV	guu	гүү	'mare'
	VV	33	00	'powder'
	CVi	xui	хүй	'group'
	Vi	ai	ай	'category'
	CVC	t ^h ət ^{jh}	тоть	'parrot'
	VC	aj	ая	'melody'
	CVVC	suulz	сүүл	'tail'
	VVC	aaw	аав	'father'
	CViC	cuitz	зүйл	'type'
	ViC	ailz	айл	'family'
	CVCC	taws	давс	'salt'
	VCC	art	ард	'people'
	CVVCC	nuurs	нүүрс	'coal'
	VVCC	aabc	аалз	'spider'
	CViCC	mailzs	майлс	'cypress'
	ViCC	oims	оймс	'sock'
	CVCCC	tawst ^h	давст	'salty' ('salt-AR')
	VCCC	ilzst ^h	элст	'sandy' ('sand-AR')
	CVVCCC	nuursč ^h	нүүрсч	'coalman' ('coal-AGNR')
	VVCCC	no exampl		probably accidental gap
	CViCCC	mai lz st ^h	майлст	'cypress-AR'
	ViCCC	$\mathfrak{i}mst^h$	оймст	'sock-AR'
(h)	Word-initial	,		
(0)	CV	ja.ţa	ялаа	ʻfly'
	V	a.tu	адуу	'horse'
	ĊVV	t ^h uu. k ai	цдуу туулай	
	VV	aa.rug	ааруул	
	CVi	pai.šəŋ	байши	
	Vi	ai.məg	аймаг	'district'
	CVC	xon.ti	хөндий	'hollow'
	VC	ulz.gər	үлгэр	'story'
	CVVC	paaw.gai	баавгаі	5
	VVC	uur.xai	уурхай	
	CViC	nailz.cur	найлзу	
	ViC	ailz.t ^h ai	айлтай	'family–сом'
	CVCC	xant.gai	хандгаі	
	VCC	ars.Էəŋ	арслан	'lion'
	CVVCC	gaans.t ^h ai	гаанста	ай 'ріре–сом'
	VVCC	aabc.thai	аалзтаі	
	CViCC	mailzs.thai	майлст	-
	ViCC	əims.t ^h əi	ойметс	ой 'sock-com'
	CVCCC	tawst ^h .t ^h ai	давстта	ай 'salty–coм'
	VCCC	ilzst ^h .t ^h e	элсттэй	а́ 'sandy–сом'
	CVVCCC	nuursč ^h .t ^h	е нүүрсч	тэй 'coalman–сом'
	VVCCC	uurst ^h .tza	уурстла	
				-

	CViCCC	nəirst ^h .59		ірстлос)	'to sleep-TERM-RFL' ($n > tr = t^h z = 3$)
	ViCCC	aimst ^h .lza	аим	истлаа		'to be scared-TERM-RFL' ($aims-t^h \not z - a$)
(c)	Word-inter	mal				
	CV	ja.Էa.t ^h ai	:	ялаатаі	ă	'fly-com'
	CVi	t ^h ʊʊ.났ai.t ^h ai	,	туулай	гай	'hare-com'
	Cə	ar.Gə.Ha	ä	аргалаа	a	'dried dung–RFL'
	CVC	aa.rʊ৳.t ^h ai		ааруулт		'dried curds-сом'
	CViC	xa.rais.nas		харайс		'to jump–PSTP–ABL' (<i>xarai–sŋ–as</i>)
	СәС	sar.məg.č ^h əŋ		сармаг		'monkey'
	CVCC	sa.narx.səŋ		санаар		'to intend-PSTP' (sanarx-sŋ)
	CViCC	puc.gairx.səi		бузгайј		'to be haughty–PSTP' (<i>puccairx–sŋ</i>)
	CəCC	sur.wəង្វč.ង្គទន្		сурвал:		'journalist'
	CVCCC	sa.narxt ^h .片a		санаар		'to intend-TERM-RFL'
	CViCCC	puc.gairxth.g		бузгайр		'to be haughty-TERM-RFL'
	CəCCC	nʊ.Էʲəmstʰ.tʰ	ai 1	нулимс	ттай	'tearful—coм'
(d)	Word-final					
	CV	ja. z a	ялаа	ı	ʻfly'	
	CVi	t ^h ʊʊ.났ai	туул	ай	'hare'	
	CVC	šag.šur	шагі	шуур	'biscuit	-stick'
	CViC	xa.raix	xapa	айх	'to jum	p-futp'
	СәС	uऺz.gər	үлгэ	p	'story'	
	CVCC	sa.narx	сана	apx	'to inter	
	CViCC	puc.gairx	бузг	айрх	'to be h	aughty'
	CəCC	gu.təmč	гуда	ιмж	'street'	
	CVCCC	mo.gorst ^h		эөрст		'(gristle-AR)
	CViCCC	no examples	found	d – prob		
	CəCCC	nu.�� ^j əmst ^h	нули	имст	'tearful	'(tear-AR)

Any single consonant except p and p^{j} (and the marginal phonemes p^{h} , p^{jh} , t) can be a coda. The historical explanation for the exceptions is that Old Mongolian *pdeveloped to w (or w^{j} in palatalizing environments) when preceded by a vowel, for example, $*t^{h}apun > t^{h}aw$ TaB 'five'; $*t^{h}apin > t^{h}aw^{j}$ TaBb 'fifty' (see 10.8.1).

6.1.2 Codas, sonority, and epenthesis²

Syllabification is closely related to vowel epenthesis in Mongolian, and we will show in this chapter that phonetically reduced non-initial vowels are inserted as part of the syllabification process, in order to create well-formed surface syllables. Thus they are epenthetic vowels which are absent from phonological representations. When the phonological representation of a word ends in two consonants, there are two possibilities for the surface form: it may have a two-consonant coda,

² Jalcib (1962); Sanžeev (1967); Kakudô (1974); Dobo (1982); Svantesson (1988*a*; 1994); Charette (1991–2); Denwood (1997*a*; 1998).

e.g. /art/ art apg 'people', or if the two consonants cannot form a coda, an epenthetic vowel is inserted between them, e.g. /paat^hr/ $paat^{h} \sigma r$ баатар 'hero'. Whether or not a vowel is epenthesized depends only on the underlying consonant cluster, and there is no contrast between forms with and without epenthetic vowels. The only

	t ^h	t ^{jh}	čh	ch	р	pj	t	t ^j	č	c	š	s	x	xj	G	g	gj	m	m ^j	n	n ^j	ŋ	ß	βj	r	\mathbf{r}^{j}	w	wj	j
t ^h	_	0	-	-	0	0	-	0	-		_	-		0	_	_	0	-	0	-	0	-	_	0	-	0	_	0	0
t ^{jh}	_				0	0	_									_							_		_				0
\check{c}^{h}	_	0	_	_	0	0	_	0	_			_		0		_	0	-	0	_	0	_	_	0	_	0	_	0	0
$\mathbf{c}^{\mathbf{h}}$	_	0			0	0	_	0	_		_	_		0		_	0		0	_	0	_	_	0	_	0	_	0	0
t	-	0	-	Ι	0	0	-	0	-		-	-		0	-	_	0	-	0	_	0	Ι	-	0	-	0	-	0	0
tj	-				0	0						-				_							_		_				0
č	-	0		_	0	0	_	0	_			_		0	-	_	0	-	0	_	0	-	-	0	_	0	-	0	0
c	_	0		_	0	0	_	0	_		_			0	_	_	0	-	0	_	0	_	_	0	_	0	-	0	0
š	+			I	0	0	_		_			_			_	_	-	-		_		Ι	_		_		-		0
s	+	0	+		0	0	_	0	_		-			0	-	_	0	-	0	_	0	_	_	0	_	0	-	0	0
х	+	0	+	_	0	0	_	0	_		_	_		0		_	0	-	0	_	0	_	_	0	_	0	-	0	0
$\mathbf{x}^{\mathbf{j}}$	+			-	0	0			_		-	_				_		-		_		_	_		_		-		0
G	+	0	-		0	0	_	0	_			_		0		-	0	-	0	_	0	-	-	_	_	0	_	0	0
g	+	0	+	+	0	0	+	0	+	+	+	+		0		_	0	-	0	_	0	_	_	0	_	0	-	0	0
g^j					0	0	+		+			+				_		-		_		_	_				-		0
m	+	0	+	+	+	0	+	0	+			+	+	0	_	_	0		0	_	0	_	_	0	_	0	_	0	0
$\mathbf{m}^{\mathbf{j}}$	+		+				+		+			+		+		_				_		_	_		_		_		0
n	+	0	+	+	0	0	+	0	+	+		+		0	-	_	0	-	0	_	0	_	_	0	_	0	-	0	0
\mathbf{n}^{j}	+		+	+	0	0	+	+	+			+		+		_				_		_	_		_		-		0
ŋ	0	0	0	0	0	0	0	0	0	0	+	0	+	0	+	+	+	0	0	0	0	0	0	0	0	0	0	0	0
ß	+	0	+	+	+	0	+	0	+	+	+	+	+	0	-	-	0	-	0	_	0	-	-	0	_	0	_	0	0
$\boldsymbol{\beta}^j$	+		+	+		+	+		+			+		+		_		-		_		_	_		_		-		0
r	+	0	+	+	0	0	+	0	+	+	+	+	+	0	-	_	0	-	0	_	0	_	_	0		0	-	0	0
$\mathbf{r}^{\mathbf{j}}$	+		+	+	0	0	+		+	+	+	+		+		_		-		_		_	_				_	_	0
w	+	0	+	+	0	0	+	0	+		+	+	+	0	_	_	0		0	_	0	_	_	0	_	0	_	0	0
$\mathbf{w}^{\mathbf{j}}$	+		+	+	0	0	+		+			+		+		-				_		-	-		_		-		0
j	+	0	+		0	0	+	0	+	+		+	+	0	-	-	0		0	-	0	-	_	0	-	0	-	0	0

TABLE 6.2 Two-consonant codas

The table shows which combinations can (+) and which cannot (-) form codas. Combinations marked 'o' do not occur in phonological representations (see (4) below), and empty positions denote combinations which have not been attested and are believed to be accidental gaps. The area below and to the left of the thick line contains those coda clusters which are predicted by the coda constraint (2).

exceptions are verbs with the future participle suffix -x, which have an epenthetic vowel even if the stem-final consonant and x may form a coda (see 6.4.1).

Table 6.2 shows which two-consonant combinations can form a coda and which cannot. All combinations of two final underlying consonants which we have encountered are exemplified in section 6.7.

Whether or not a consonant combination can form a syllable coda in Mongolian is essentially determined by a simple form of the well-known 'sonority law' saying that sonority is maximal at the syllable nucleus, and decreases towards the edges (an early formulation is Whitney 1866; see Clements 1990 for a recent survey and analysis). Combined with the assumption that voiced consonants are more sonorous than voiceless consonants, this accounts for most two-consonant codas in Mongolian, making the combination of a voiced+voiceless consonant a possible coda and causing other combinations to trigger vowel epenthesis (2).

(2) Coda constraint

A string of two consonants is a possible coda if, and only if, it has decreasing sonority, that is, if it consists of a voiced consonant followed by a voiceless consonant.

The following exceptions obtain:

- (a) The clusters ηg , ηg^j , ηg are allowed as codas.
- (b) Clusters consisting of a fricative and an aspirated stop or affricate (except c^h) form codas. The codas št^h, st^h, sč^h, xt^h, xč^h, and x^jt^h of this type have been attested.
- (c) The voiced stop G does not form a coda with voiceless obstruents, except t^h .

In the clusters in (a), the place of articulation of the velar nasal is assimilated to the following obstruent, so that the actual pronunciation is $[\eta g, \eta^j g^j, NG]$. These clusters can be regarded as forming partial geminates that cannot be split by epenthesis (Schein and Steriade 1986: 720), which may explain why they exist. Another possible explanation is the fact that the velar nasal never occurs as a syllable onset, which prevents it from being followed by an epenthetic vowel.

There are also some three-consonant codas, exemplified in (3) (see also Table 6.1). They consist of a voiced consonant followed by one of the coda-forming fricative–stop clusters (2b), and all encountered three-consonant combinations of this type form codas, except σst^h and σsc^h which trigger epenthesis (e.g. /cags– c^h /[ca.gəs c^h] загасч 'fisherman'), obviously a consequence of the fact (2c) that σs is an impossible coda (/cags/ [ca.gəs] загас 'fish'). It can be noted that all attested words with three-consonant codas are formed with suffixes.

(3) Three-consonant codas

mst ^h	/čims–t ^h /	čimst ^h	жимст	'having fruit' ('fruit–AR')
n ^j st ^h	/ən ^j s–t ^h /	∋n ^j st ^h	оньст	'lock–ar'
Ե st ^h	/iʒs–tʰ/	ilʒst ^h	элст	'sandy' ('sand-AR')
Է ^j st ^h	/xalz ^j s—t ^h /	xalz ^j st ^h	хальст	'peel-ar'

wst ^h	/taws-th/	tawst ^h	давст	'salty' ('salt–AR')
mxt ^h	/nomx-t ^h /	nomxth	HOMXT	'to become tame' ('tame–vr')
ŋxt ^h	/c ^h aŋx-t ^h /	chaŋxth	цанхт	'to become sick' ('disease–vr')
₿ ^j xt ^h	/naţ ^j x_t ^h /	nalz ^j xt ^h	нальхт	'to get mouldy' ('mould-vr')
rxt ^h	/irx-t ^h /	irxt ^h	эрхт	'competent' ('rights-AR')
gsč ^h	/cigs_č ^h /	cigsč ^h	зэгсч	'warbler' (cf. cigs зэгс 'reed')
rsč ^h	/nuurs–čʰ/	nuursč ^h	нүүрсч	'coalman' ('coal–AGNR')
mxč ^h	/xamx–č ^h /	xamxč ^h	хамхч	'to smash up' ('to pieces-vr')
rxč ^h	/šarx–č ^h /	šarxč ^h	шархч	'coroner' ('wound–AGNR')
wxč ^h	/xəwx–č ^h /	xəwxč ^h	ховхч	'to break apart' ('apart–vr')

A consonant string may satisfy the coda constraint without actually occurring as a coda, but in that case the string does not occur in phonological representations, either because of a segmental rule or because it is an accidental gap. The following restrictions hold in phonological representations for sequences of two word-final consonants:

- (4) *Restrictions on final two-consonant combinations in phonological representations*
 - (a) If the second consonant is palatalized $(t^{jh}, t^j, g^j, x^j, m^j, n^j, \xi^j, r^j, w^j)$, the first one must be one of these consonants as well (or, in a few words, \check{s} or G).
 - (b) The velar nasal /ŋ/ occurs only before /š, x, g, g^{j} , g/.
 - (c) The glide /j/ does not occur as the second consonant (see 6.5 for a possible exception).
 - (d) The labials /p, p^{j}/do not occur as the first consonant.
 - (e) The labials /p, p^j/ occur as the second consonant only after /m, β/ and the corresponding palatalized consonants.

6.2 SYLLABIFICATION OF MORPHOLOGICALLY SIMPLE WORDS³

In this section we will show that syllabification and epenthesis in morphologically simple words can be described by using the general principles of maximality and directionality (see e.g. Itô 1989). The only language-specific facts needed are the exact formulation of the coda constraint and the direction of syllabification.

When the phonological representation of a word ends in more consonants than can be accommodated in a syllable coda, one or more vowels are epenthesized. An epenthetic vowel is never inserted at the end of a word, so if there are two final consonants, the epenthetic vowel (if necessary) is always inserted between them (see examples in 6.7). If a word ends in three consonants, a schwa (if necessary) is

³ For notes on syllabification of morphologically simple words see App. F, 6.2, p. 222.

inserted before the two last consonants if they can form a coda (5a), and before the last consonant otherwise (5b).

(5)	(a)	/gʊtmč/	gu.təmč	гудамж	'street'
		/gurnc/	GU.rənc	гуранз	'emery'
		/šitms/	ši.təms	шидэмс	'fillet'
		/ic ^h st ^h /	i.c ^h əst ^h	эцэст	'as a result of'
	(b)	/t ^h arwg/	t ^h ar.wəg	тарвага	'marmot'
		/gurw��/	gur.wəb	гүрвэл	'lizard'
		/xitmŋ/	xit.məŋ	хэдмэн	'pear'
		/č ^h ot ^h gr/	č ^h ot ^h .gər	чөтгөр	'devil'
		/pompg/	pom.pəg	бөмбөг	'ball'
		/xowchs/	xʊw.cʰəs	хувцас	'clothes'

Epenthetic vowels are thus inserted as far to the left as possible, both in two- and three-consonant final clusters. For example, /gotmč/ 'street' is syllabified as $g_{U.tamč}$ rygamæ, not * $g_{U.tamč}$ (which does consist of two well-formed syllables). This can be described by assuming that syllabification proceeds from right to left, creates the largest possible syllable constituents (rhymes and onsets), and inserts epenthetic vowels only when necessary to build well-formed rhymes. Words consisting of one morpheme are thus syllabified by a right-to-left maximal procedure:

- (6) Monomorphemic syllabification
 - (i) The phonological representation of the word is scanned from right to left and a maximal coda (possibly empty) is found.
 - (ii) The coda is combined with the preceding vowel to make a rhyme. If the segment preceding the coda is a consonant, a schwa vowel is epenthesized as the nucleus of the rhyme.
 - (iii) The preceding consonant becomes an onset, and the syllable is complete.
 - (iv) If there are segments left, the procedure is repeated.

For an example of how monomorphemic syllabification works, consider the word /jort^hnc^h/ ертөнц 'world' (7a). The input string is scanned from right to left. The one-consonant string c^h is a possible coda, but it is not maximal, since the string nc^h also is a possible coda. Since t^hnc^h is not a possible coda, nc^h is maximal and becomes a coda. The preceding segment t^h is a consonant, so an epenthetic vowel must be inserted as a nucleus, t^h becomes an onset, and the final syllable $t^h \partial nc^h$ is completed. In the next step, or becomes a rhyme and j an onset. It can be noted that the direction is critical; maximal left-to-right syllabification of this word would result in the syllabification *jort^h. $n\partial c^h$ (7b), which is incorrect, although it does consist of two well-formed syllables.



This procedure describes syllabification correctly in morphologically simple words, not only those with final clusters of two or three consonants, but also those with longer consonant clusters or with more than one phonological vowel, as exemplified in (8).

(8)	/pɔ肽čmr/	pɔᡰʒč.mər	болжмор	'lark'
	/surwţč/	sur.wə��č	сурвалж	'root'
	/sar ^j msg/	sa.r ^j əm.səg	саримсаг	'garlic'
	/sarmgč ^h ŋ/	sar.məg.č ^h əŋ	сармагчин	'monkey'
	/xawt ^h Gţč/	xawt ^h .Gəbč	хавтгалж	'snipe (bird)'
	/xuux�ate/	xuu.xəħ.te	хүүхэлдэй	'doll'
	/gurggtai/	GUT.Gəb.tai	гургалдай	'nightingale'
	/galywsa/	Galzəwsa	(галавсаа)	'sausage' (Ru kolbasá
	-	-		(see 3.3))

Right-to-left maximal syllabification assigns each segment in the phonological representation to either a rhyme or an onset, and there are no extrasyllabic segments. This syllabification procedure cannot produce rhymes consisting only of an epenthetic vowel, and it also guarantees that all syllables, except possibly the word-initial one, have an onset. This is consistent with the fact that any word-final surface consonant cluster can occur as the coda of a non-final syllable. In this respect, Mongolian is different from many other languages, such as Swedish, where some word-final consonant clusters are not allowed as codas in word-internal syllables.

There are some exceptions to the monomorphemic syllabification rule, which are at least partially phonologically conditioned. One exception is that words ending in a three-consonant cluster consisting of η followed by a velar stop (g, g^j, G) and a voiceless consonant are always syllabified by inserting the schwa after the

70

velar stop: /t^hiŋgs/ t^hiŋ.gəs тэнгэс 'sea' (not *t^hi.nəgs); /č^hiŋgs/ č^hiŋ.gəs Чингэс 'Chinggis' (not * $\check{c}^{h}i.n$ əgs). This is obviously related to the existence of codas of the form $\eta g, \eta g^{i}, \eta g$ (see (2a)).

Another class of exceptions involves words whose phonological representations end in preceded by two voiced consonants. In most such words, the epenthetic vowel is irregularly inserted immediately before s (9a). Some words follow the monomorphemic syllabification rule and insert the schwa before the sonorant–scluster (b), and some vary more or less freely (c).

(9)	(a)	/t ^h arws/	t ^h ar.wəs	тарвас	'watermelon'
		/alzms/	ałz.məs	алмас	'diamond'
		/xar ^j g ^j s/	xar ^j .g ^j əs	харгис	'vicious'
		/ar ^j w ^j s/	ar ^j .w ^j əs	арвис	'erudition'
	(b)	/alz ^j rs/	a.ᡰʒ ^j ərs	алирс	'lingonberry'
	(c)	/sar ^j ms/	sar ^j .məs ~ sa.r ^j əms	саримс	'garlic'
		/nʊऺ <code>kjms/</code>	nuk ^j .məs ~ nu.k ^j əms	нулимс	'tear'

6.3 SCHWA ~ ZERO ALTERNATION

In the preceding section it was shown that schwa epenthesis in monomorphemic words is governed by the syllabification rule (6). In derived and inflected words there are frequent alternations between schwas and zero, also governed by syllabification rules, although there is feedback from the alternation rules to the syllabification rules leading to cyclic syllabification (6.4).

The schwa ~ zero alternations which take place in suffixation can be illustrated by the nine words in (10), built up by successively adding suffixes to the root /uik/ 'action'. Because of the suffixing and agglutinative nature of the language, similar examples are not uncommon.

(10) Schwa ~ zero alternation

•				
(a) Root	/uiţ/	uitz	үйл	'action'
(b) vr	/–čʰţ/	uiłz.č ^h əłz	үйлчил	'to serve'
(c) CAUS	/–Uţ/	uiķč ^h .ķuķ	үйлчлүүл	'to cause to serve'
(d) agent	/–gč ^h /	ui��č ^h .��u.��əgč ^h	үйлчлүүлэгч	'customer'
(e) pl	/—t/	ui��č ^h .��u.��əg.č ^h ət	'customers'	
		үйлчлүүлэгчид		
(f) gen	/—iŋ/	uibč ^h .bu.bəgč ^h .tit) 'of the custo	omers'
		үйлчлүүлэгчдий	Η	
(g) nr	/_x/	uikč ^h .ku.kogč ^h .tit	jx 'customers'	belongings'
		үйлчлүүлэгчдий	HX	
(h) dat	/—t/	uibčh.bu.bəgčh.ti).xət	
		үйлчлүүлэгчдий	нхэд	
(i) rfl	/–E/	uibčh.bu.bəgčh.ti)x.te	
		үйлчлүүлэгчдий	нхдээ	

A schwa vowel is inserted in (10b, d, e, h), and one is deleted in (c, f, i). Inspection of the syllabification of the nine words in (10) shows that they are all syllabified by rule (6), although they are not monomorphemic (except the root). The zero \sim schwa alternations can thus be explained as automatic consequences of the syllabification rule. Some more examples will be given to illustrate this.

The simplest case is that no epenthesis or schwa deletion takes place. The suffix can be incorporated into the final syllable of the stem (11a), or it can form its own syllable (b). If the suffix begins with a vowel, the final consonant of the stem becomes an onset (c). Some suffixes begin with two consonants, the first of which can be incorporated into the final syllable of the stem, the rest of the suffix forming a syllable (d). A schwa can be inserted inside a suffix (e) or between the stem and a suffix (f), but not into the stem, as will be shown in section 6.4 below. Finally, deletion of a schwa vowel in the stem-final syllable takes place if the stem ends in a schwa plus a single consonant, and the suffix begins with a vowel (g). If the suffix can form a coda, the schwa is moved from the stem to the place between the stem and the suffix (h).

(11) Schwa deletion and insertion

(a)	ar – /t/	\rightarrow	art	ард	'back-dat'
	nee – /gt/	\rightarrow	neegt	нээгд	'to open-passive'
(b)	co.wəţ – /ţE/	\rightarrow	co.wəţ.ţo	зөвлөлөө	'to advise-DPST'
	mər ^j – /t ^h Ei/	\rightarrow	mər ^j .t ^h əi	морьтой	'horse-com'
(c)	małz – /E/	\rightarrow	ma.ţa	малаа	'cattle-RFL'
	gu.təmč – /Er/	\rightarrow	gu.təm.čar	гудамжаар	'street-INST'
(d)	pai – /cʰgE/	\rightarrow	paic ^h .Ga	байцгаа	'to be-COLL'
	a.sv – /ŋgUtʰ/	\rightarrow	a.suŋ.gut ^h	асуунгуут	'to ask-
					BENEFACTIVE'
(e)	$\Im r - /\check{c}^h x /$	\rightarrow	ər.č ^h əx	орчих	'to enter-
					INTENSIVE'
	čhi.xər – /ţg/	\rightarrow	č ^h i.xər.�əg	чихэрлэг	'sugary'
					('sugar–ar')
(f)	$x \Im t^h - t/t$	\rightarrow	xə.t ^h ət	хотод	'town-dat'
	mart ^h – /gt/	\rightarrow	mar.t ^h əgt	мартагд	'to forget-PASSIVE'
(g)	xa.mər – /E/	\rightarrow	xam.ra	хамраа	'nose-rfl'
	xa.rəw – /Uţ/	\rightarrow	xar.wob	харвуул	'to shoot–CAUS'
(h)	xa.rəw – /n/	\rightarrow	xar.wən	харвана	'to shoot–npst
	xa.rəw – /gčʰ/	\rightarrow	xar.wəgč ^h	харвагч	'to shoot–AGENT'

There are instances of schwas which never alternate with zero, including those found in non-final syllables (e.g. in *xuu.xəţ.te* хүүхэлдэй 'doll') or in final syllables with a complex coda (e.g. *gu.təmč* гудамж 'street'). Rhymes consisting only of a schwa occur only in non-final syllables (see Table 6.1), and thus never alternate with zero.

6.4 CYCLIC SYLLABIFICATION⁴

It was shown in section 6.2 that morphologically simple words are syllabified and provided with epenthetic vowels by a rule that is maximal and right-to-left directional. As seen in section 6.3, this rule often applies to derived or inflected words as well, but as was pointed out by Damdinsüren (1946: 20; 1958: 27), Coloo (1976: 129; 1987: 29), Saitô (1986), and others, there are some Halh words which differ only by the presence or absence of a reduced vowel, for example, xakəx халах 'to change' vs. xakx xanx 'shield; Halh'. According to our observations, there are also words which differ only in the place of reduced vowels, for example, cowkako зөвлөлөө 'advice-RFL' vs. cowakko зөвлөлөө 'to advise-DPST' (according to the standard spelling rules these two non-homophonous words are spelled in the same way). An analysis which treats reduced vowels as non-phonemic and derivable by purely phonological rules is thus impossible in Standard Halh Mongolian. If, however, the morphology is taken into account, the phonetically reduced vowels can be analysed as being absent from phonological representations; see Svantesson (1988a; 1994; 1995a), a similar analysis was proposed by Saitô (1984). This can be illustrated with the following pairs of words with identical segments, or segments belonging to the same sonority classes, but with different morphological structure and different syllabification. The first word in each pair is syllabified by rule (6), but not the second one.

(12)	(a)	/cow�_�E/	co.wəჭ.ჭo	зөвлөлөө	'advised' ('advise-DPST')
		/cow�_�_E/	cow.ţə.ţo	зөвлөлөө	'his advice'
					('advise-NR-RFL')
	(b)	/aţt ^h _t_mţ/	a� .t ^h ət.mə�	алтадмал	'gilded' ('gold–vr–ar')
		/art–č ^h ǥ–ӄ/	art.č ^h ə.ԷəԷ	ардчилал	'democratization'
					('people-vr-nr')

To see why the words in these pairs are syllabilied in different ways, we show each step in the derivations in (13).

(13)	(a)	Root	/cowţ/	co.wə৳	зөвлө	'to advise'
		DIRECT PAST	/–ţE/	co.wəţ.ţo	зөвлөлөө	'advised'
	(b)	Root	/cowţ/	co.wəb	зөвлө	'to advise'
		NOMINALIZER	/_৳/	cow.५३५	зөвлөл	'advice'
		REFLEXIVE	/–E/	cow.ţə.ţo	зөвлөлөө	'his advice'
	(c)	Root	/aŀzth/	alzth	алт	'gold'
		VERBALIZER	/—t/	alz.t ^h ət	алтад	'to gild'
		ADJECTIVIZER	/_mţ/	alz.t ^h ət.məlz	алтадмал	'gilded'
	(d)	Root	/art/	art	ард	'people'
		VERBALIZER	/–čʰţ/	art.č ^h ə�z	ардчил	'democratize'
		NOMINALIZER	/_ b /	art.č ^h ə.৳ə৳	ардчилал	'democratiza-
						tion'

⁴ For notes on cyclic syllabification, see App. F, 6.5, p. 223.

Inspection of these and other similar examples shows that maximal right-to-left syllabification (6) is not applied in those cases when it would require insertion of a new epenthetic vowel into the already syllabified part of the word, as would be the case if this rule were applied in the final step in (13b, d): $cow.\xi_{\partial\xi}-/E/ \rightarrow *co.w_{\partial\xi}$. *ko*; $art.\check{c}^{h}\partial\xi-/B/ \rightarrow *ar.t\partial\check{c}^{h}.\xi_{\partial\xi}$. The actual syllabifications $cow.\xi_{\partial}.\xi_{O}$ and $art.\check{c}^{h}\partial$. *bo* and *art.* $\check{c}^{h}\partial\xi$ avoid this.

Thus, the general syllabification rule for morphologically complex words can be expressed in terms of a constraint on resyllabification after a suffix has been added. A suffixed word is syllabified as if it were monomorphemic, unless this forces a new epenthetic vowel to be inserted in the (already syllabified) stem to which the suffix is added:

(14) Resyllabification constraint On each morphological cycle, an epenthetic vowel cannot be inserted into the already syllabified part of a word.

Thus syllabification is cyclic in relation to the morphology in the sense that after a suffix has been added, the word is resyllabified according to the monomorphemic syllabification rule (6), unless the resyllabification constraint is violated. In that case, the stem is left unchanged, and the epenthetic vowel (if necessary) is added after it. This results in a syllable whose rhyme consists only of an epenthetic vowel.

Some other words where this rule applies are shown in (15).

(15) Examples of cyclic syllabification $ar.Goldshift - /ig/ \rightarrow ar.Goldsig apranear 'dried dung-ACC'$ (not *a.rog.fig) $t^{h}og.rog - /or/ \rightarrow t^{h}og.roldsig or the perform of the circulate' ('ring-INST')$ $(not *t^{h}o.gor.gor)$ $t^{h}og.roldsig - /b/ \rightarrow t^{h}og.roldsig or the perform of the circulate' ('ring-VR')$ $(not *t^{h}o.gor.golds)$

The resyllabification constraint ensures that a schwa vowel in a non-final syllable cannot be deleted, since that would necessitate the insertion of a schwa in the stem. When a suffix is added to a word which has a C_{∂} syllable, cyclic syllabification thus preserves this epenthetic vowel: $art.\check{c}^{h}\partial.\xi_{\partial}\xi_{-}/t/ \rightarrow art.\check{c}^{h}\partial.\xi_{\partial}\xi_{f}$ 'democratization-DAT' $art.\check{c}^{h}\partial.\xi_{\partial}\xi_{-}/as/ \rightarrow art.\check{c}^{h}\partial.\xi_{\partial}\xi_{f}$ as 'democratization-ABL'; monomorphemic syllabification would give $*ar.t_{\partial}\check{c}^{h}.\xi_{\partial}\xi_{f}$ in the first word, but the correct result in the second one.

Cyclic syllabification can also explain the syllabification of words with the plural suffix -t, which is one of several plural suffixes, chosen partly lexically and partly phonologically. Examples are $mir.g
at mapping 'sages', and <math>t^h u \ mathbf{s}.m at rymmode to -left syllabification would lead to <math>mi.r \ mathbf{s}.m at r \ mathbf{s}.m at r \ mathbf{s}.m at mathbf{s}.m at mathbf{s}.m at mathbf{s}.m \ mathbf{s}.$

epenthetic vowel in the stem (e.g. $mir.g \partial \eta \rightarrow *mi.r \partial gt$), and thus blocks maximal right-to-left syllabification.

It can be noted that derivational and inflectional suffixes are treated in the same way by the syllabification rule (and, as far as we know, by other phonological rules). Thus there seems to be no need to divide the phonology and morphology into strata depending on the type of suffix. On each cycle, the syllabification rule does not need access to the input string's morphological structure, only to its syllabification.

6.4.1 The verb suffix $-(\mathfrak{g})\mathbf{x}$

As mentioned above, there are contrasts of the type exemplified in (16). They involve the future participle suffix [-(a)x] (also known as 'irrealis' or 'infinitive'), realized as -x after vowels and as -ax after consonants. The existence of such contrasts was noted by the main designer of the Cyrillic Mongolian script, Cèndijn Damdinsürèn, and seems to be one important reason why he chose to write epenthetic vowels in Cyrillic Mongolian rather than follow the example of the Cyrillic Kalmuck script, where epenthetic vowels are not written (Damdinsürèn 1946: 20; 1958: 27; cf. also Coloo 1987: 29).

(16)	(a)	xa.ţəx	халах	'to change-FUTP'
		i.rəx	ирэх	'to come-furp'
		sa.wəx	савах	'to beat-FUTP'
	(b)	xaţx	халх	'shield'
		irx	эрх	'power'
		sawx	савх	'chopsticks'

These future participle forms violate the epenthesis rules, unlike other verb forms with suffixes consisting of a single obstruent, such as the imperfect gerund $-\check{c}$. This suffix is regular, so that the verbs in (16a) have the imperfect gerund forms $xa\xi\check{c}$ халж, $ir\check{c}$ нрж, $saw\check{c}$ савж, without schwas. The irregular $-\partial x$ forms are not phonologically conditioned, since the phoneme x always functions regularly as a voiceless obstruent in morphologically simple words like those given in (16b) above. Furthermore, there are other suffixes with the phonological form -x, with nominalizing or verbalizing function, and they behave regularly, for example, in /aaw-iŋ-x/ aawiŋx aabiHx 'what belongs to father' ('father-GEN-NR'), /teer-x/ teerx дээрх 'that above' ('above-NR'), and /paat^hr-x/ paat^h ∂rx баатарх 'to act as a hero' ('hero-VR').

The x of the future participle suffix thus requires a vowel before it, or, equivalently, it requires to be the whole syllable coda (and not just part of the coda). The phonological representation of this suffix is problematic. One possibility would be to represent it as $/-\infty x$, although schwas do not occur otherwise in phonological representations. It would then be necessary to introduce an exception to consonant epenthesis (see 5.2.5), however, so that forms like the future participle of *nee* H₉₉ 'to open' do not surface as **neegox* instead of *neex* H_{99x}. Another problem with this analysis is the fact that the future participle suffix behaves regularly, like other instances of x, on following morphological cycles, where its epenthetic vowel usually is deleted if this is required by the syllabification rules. For example, the ablatives of the words $xa\xi ax$ xanax 'to change–FUTP' and $xa\xi x$ xanx 'shield' are normally pronounced in the same way, as $xa\xi xas$, although they are written differently (xanaxaac, xanxaac). Like other schwas, the one in this suffix may disappear depending on the speech style (see Fig. 2.8 for an example).

It might be remarked that there is nothing unusual about this suffix except its aberrant phonology. It is very common and completely productive in the contemporary language, it existed already in the oldest known stages of Mongolian (Weiers 1969: 160 ff.), and its present form is the regular reflex of its earliest form, Old Mongolian $*-k^h U$ (*[q^hu ~ k^hy]).

6.5 THE PHONEMIC STATUS OF PALATAL GLIDES

It is well known that glides and the corresponding vowels are in complementary distribution in many languages. This fact is implicit in the orthography of, for instance, Latin, and was pointed out explicitly at least as early as by Whitney (1862: 310). Expressed in modern language, his proposal is to use the same phonological representation for the glide and the corresponding vowel, and this is often assumed by modern phonologists as well (see e.g. the discussion in Goldsmith 1990: 151ff.). In this section, the representation of Mongolian [i] and [j] is discussed in relation to syllabification, following Svantesson (1996) (see also Saitô 1985).

The assumption that [i] and [j] are allophones of one phoneme in Mongolian is supported by the behaviour of the optative suffix, which is -i with consonant-final stems (17a) and -j with vowel-final stems (b); the Cyrillic Mongolian spelling of this suffix is a bit idiosyncratic.

(17)	(a)	xar	xap	'to look'	optative:	xar—i	харъя
		it	ид	'to eat'		it—i	идье
	(b)	UU	уу	'to drink'		υυ–j	ууя
		xowa	хуваа	'to divide'		xuwa-j	хуваая
		nee	нээ	'to open'		nee-j	нээе

This might suggest that the optative suffix is represented as i, which is realized as [j] when it is adjacent to a vowel, and that [j], which is always adjacent to a vowel, might in general be represented as an allophone of i. This solution is, however, problematic for two reasons. The first reason is that a vowel suffix after a vowel would normally trigger epenthesis of G/g (5.2.5), and the second one is the existence of surface contrasts between *i*-diphthongs and the combinations of a vowel and [j]:

(18)	aj	ая	'melody'	ai	ай	'category'
	эj	oë	'to sew'	əi	ой	'forest'
	υj	уя	'to tie'	υi	уй	'sorrow'
	uj	үe	'joint'	ui	үй	'to brew'
	pij	бие	'body'	xii	хий	'air'

The *Vi* diphthongs and the *Vj* combinations are phonetically realized as diphthongs, which differ from each other in their paths in the F1–F2 plane (1.3); the combination *ij* is phonetically identical to the long vowel *ii* in normal speech, however, so that *pij* and *xii* are pronounced [pi:], [xi:]. The historical development of the $Vi \sim Vj$ contrast is due to deletion of word-final short vowels (cf. 10.5.1):

(19)	Old Mongolian	Halh		
	*huja	υj	уя	'to tie'
	*yje	uj	үe	'joint'
	*hoi	oi	ой	'forest'
	*t ^h aulai	t ^h uu b ai	туулай	'hare'

The two glides j and i show different phonological behaviour before vowel-initial suffixes, such as instrumental -Er:

(20)	(a)			'melody' 'elegant'	instrumental:	aj–ar Goj–or	аяар гоёор
		uj pij	*	ʻjoint' 'body'		uj–er pij–er	үеэр биеэр
	(b)	oi xui	ой	'category' 'forest' 'group' 'air'		ai–gar oi–gor xui–ger xii–ger	хүйгээр

These examples show that *i* triggers consonant epenthesis (5.2.5) and thus functions as a vowel, while *j* behaves as a consonant. As comparison with (17) shows, this is confirmed by the treatment of the optative suffix -i/j (21). The optative forms in (21a) are disyllabic and those in (21b) monosyllabic.

(21)	(a)	xaj	хая	optative: 'to leave'	xaj—i	хаяя
		эj	oë	'to sew'	∋j−i	oëë
		υj	уя	'to tie'	υj—i	уяя
	(b)	xai	хай	'to seek'	xai—j	хайя
		əi	ой	'to bounce'	∋i−j	ойё
		GUİ	гуй	'to ask for'	G∪i−j	гуйя

The difference between i and j is also seen in vowel epenthesis. Both can co-occur with a following voiceless consonant within the same syllable (22a), but only i can combine with a following voiced consonant in the same syllable (b) or with a voiced–voiceless cluster (c). In these cases, j requires an epenthetic vowel, making

the word disyllabic at least in careful lexical pronunciation. Thus, i functions as a part of the syllable nucleus, while j functions as a consonant.

(22)	(a)	/uis/ /ujs/	uis ujs	үйс үес	'birchbark' 'time'
	(b)	/pair/ /pajr/	pair pajər	байр баяр	'place' 'happiness'
	(c)	/xai�cʰ/ /xaj–��cʰ/	хаіђс ^h xajəђс ^h	хайлц хаялц	'fusibility' 'to leave–coop'

In conclusion, Mongolian [j] and [i] contrast on the surface, and behave phonologically in different ways both as regards consonant and vowel epenthesis. Obviously, they must be differentiated in some way in the phonological representation. We will assume that [i] is specified as a vowel and [j] as a consonant.

Returning to the optative, which alternates between [-j] and [-i], one solution is to assume the phonological representation /-j/. As seen in (4c), the consonant /j/does not occur together with a preceding consonant at the end of a word in phonological representation, that is, words never end in /Cj/; equivalently, no surface syllable has the rhyme [əj]. If the optative is represented as /-j/, adding it after a consonant would result in a syllable of this type, and assuming that the rhyme ∂j is realized phonetically as [i] will give the correct phonetic output. For example, the word /xar-j/ 'look-OPTATIVE' gets the form $xar\partial j$ which is realized phonetically as [xari] (cf. (17)). This would also explain why the consonant preceding [i] is not palatalized in these words.

6.6 WORD STRUCTURE⁵

There are few restrictions on how syllables can be put together to form words in Mongolian, so the word structure can be inferred from the syllable structure and from vowel harmony, which regulates the distribution of vowels within words. The only basic restriction is that a word cannot consist of only a short vowel or a consonant and a short vowel, so the minimal word structure is (C)VV (as in $\upsilon \upsilon$ yy 'to drink'; $t^{h} \sigma \tau$ too 'number'; σi of 'forest'; $x \upsilon i$ xyň 'sheath') or (C)VC (ar ap 'back'; $x \sigma f$ x $\sigma \pi$ 'foot'). A few monosyllabic function words are written with an orthographic short vowel but are nevertheless pronounced with a long vowel, at least in citation form (cf. Hattori 1951: 85; Tömörcèrèn 1968; Sanžeev 1978: 88). These include the pronouns *pii* $\delta \pi$ 'I', $\tilde{c}^{h}ii$ Ψ 'you', $t^{h}aa$ τ a 'you (honorific)', the conjunction *paa* δ a 'and', and the question marker *bee/wee* $\delta \sigma/B \sigma$. This restriction might be expressed by saying that a word must consist of two moras, assuming that a syllable with a short vowel consists of one mora and that an additional vowel, or a consonant in the syllable coda, contributes one mora. We will not develop this

⁵ Zolhoev (1972); Dyrheeva and Buhaeva (1978); Buraev (1985*b*); O. D. Buhaeva (1989; 1991*a*, *b*); Dyrheeva (1991).

further here; the mora concept is used in Chapter 7 on prosody, but a slightly different definition of the mora is employed there. It can be argued that at least some of these words have a short vowel which turns up in inflected words, for example, *pit* бид 'we', *čhini* чиний 'your', *thant* танд 'to you', so that the long vowel in the uninflected word is due to a phonetic realization rule.

Root words do have a simpler structure than derived or inflected words, but this remains to be investigated, like many other problems in Mongolian phonotactics.

6.7 FINAL CONSONANT COMBINATIONS

As mentioned in section 6.1.2, the following tables exemplify all attested combinations of two final underlying consonants.

at ^h ət ^h ət ^h əč ^h	атат оточ	'having a camel' 'doctor'	pət ^{jh} ət ^h	ботит	'having volumes'
put ^h əc ^h x ^j at ^h ət xat ^h əč t ^h at ^h əš	бүтэц Хятад хатаж таташ	'structure' 'China' 'to dry–IPFG' 'mincemeat'	pət ^{jh} ət	ботид	'volume-dat'
ut ^h əs ut ^h əg nut ^h əg t ^h ut ^h əm tət ^h ən put ^h əŋ	утас утга нутаг тутам дотно бутан	'thread' 'meaning' 'homeland' 'each' 'inside' 'jar'	xat ^{jh} əg	хатиг	'furuncle'
pot əŋ Gut ^h ə找 paat ^h ər xat ^h əw	бутан гутал баатар хатав	'boots' 'hero' 'to dry–PST'	pət ^{ih} ək xat ^{ih} ər	ботил хатир	'volume–vr' 'trot'
xuč ^h ət ^h pač ^h əč ^h ač ^h əc ^h	хүчит бачич ачиц	'powerful' 'swindler' 'once loaded'	ac ^h ət ^h	ацат	'forked'
cəč ^h ət əč ^h əč	зочид очиж	'guests' 'to go to–IPFG'	guic ^h ət t ^h əəc ^h əč puc ^h əš	гүйцэд тооцож буцаш	'complete' 'to reckon–IPFG' 'roundworm'
xuuč ^h əs pič ^h əg pač ^h əm	хуучис бичиг бачим	'old things' 'writing' 'urgent'	ic ^h əs c ^h ac ^h əg	эцэс цацаг	'end' 'tassel'
ວč ^h ən cວč ^h əŋ ບບč ^h əໄຽ ບč ^h ər ວč ^h əw	очно зочин уучил учир очив	 'to go to-NPST' 'guest' 'to forgive' 'reason' 'to go to-PST' 	t ^h ວວc ^h ən t ^h ວວc ^h əŋ puc ^h əຽ xac ^h ər t ^h ວວc ^h əw	тооцно тооцон буцал хацар тооцов	'to reckon–NPST' 'to reckon–CV' 'to boil' 'cheek' 'to reckon–PST'

čatəth	жадат	'spear-AR'	pət ^j ət ^h	бодит	'real'
čatəčh	жадач	'spearman'	poror	0 0 C	
mitəch	мэдэц	'knowledge'			
ətət	одод	'stars'			
utəč	удаж	'to stay-IPFG'			
utəš	удаж үдэш	'evening'			
Gatəs	гадас	'stake'	pət ^j əs	бодис	'matter'
satəg	садга	'effigy'	por os	оодне	matter
atəg	адаг	'end'	c ^h at ^j əg	цадиг	'biography'
utəm	удам	'origin'	e at og	циди	ologiaphy
Gatən	гадна	'outside'			
putaŋ	будан	'mist'			
xutəb	худал	'lie'	at ^j əlz	адил	'like'
otər	нудал өдөр	'day'	Gat ^j ər	гадир	'quince'
sitəw	сэдэв	'theme'	Gat a	гадир	quinee
5118 W	СЭДЭВ	theme			
cheečəth	цээжит	'chest-AR'	acət ^h	азат	'lucky'
sačəc ^h	сажиц	'amplitude'	ucot	usur	lucity
xəčət	хожид	'to be late'	icət	эзэд	'owners'
ačəč	ажиж	'to notice-IPFG'	ucəč	үзэж	'to see-IPFG'
	••••		ucəš–gui	үзэшгүй	'unsightly'
teečəs	дээжис	'choice food'	8	100	J
рэčәд	божго	'child's penis'	рэсәд	бозго	'small feather'
pučəg	бүжиг	'dance'	ucəg	үзэг	'pen'
uučəm	уужим	'wide'	ucəm	үзэм	'raisin'
ačən	ажна	'to notice-NPST'	ucən	үзнэ	'to see-NPST'
tuučəŋ	дүүжин	'swing'	šaacəŋ	шаазан	'porcelain'
ačəţ	ажил	'work'	ucəţ	үзэл	'view'
šičər	шижир	'pure'	Gacər	газар	'ground'
ačəw	ажив	'to notice-PST'	ucəw	үзэв	'to see–pst'
				1	
aašt ^h	аашт	'character-AR'	ust ^h	уст	'to disappear'
			usč ^h	усч	'waterman'
t ^h ušəc ^h	түшиц	'base'			
ašət	ашид	'always'	usət	үсэд	'extremely'
itešəč	идээш-	'to infuse-IPFG'	xusəč	хүсэж	'to want-IPFG'
	иж		t ^h isəš–gui	тэсэшгүй	'unbearable'
išəs	ишис	'stems'	U	1	
pəšəg	бошго	'official'	pəsəg	босго	'threshold'
pošəg	бошиг	'decree'	usəg	үсэг	'script'
uušəg ^j	уушги	'lung'	U	•	I
čišəm	жишим	'comparable'	xusəm	хусам	'scale'
itešən		'to infuse–NPST'	xusən	хүснэ	'to want–NPST'
	,,			1	

paišəŋ	байшин	'building'	xəəsəŋ	хоосон	'empty'
aašəlz	аашил	'to behave'	tบรอไร	дусал	'drop'
pišər	бишир	'to worship'	asər	acap	'very'
itešəw	-	'to infuse-PST'	t ^h osəw	төсөв	'budget'
					U
t ^h uuxt ^h	түүхт	'historical'	pəx ^j t ^h	бохьт	'to become tarry'
t ^h uuxč ^h	түүхч	'historian'	-		•
xoxəch	хөхөц	'indigo'	ax ^j əc ^h	ахиц	'progress'
xuuxət	хүүхэд	'child'			1 0
uxəč	үхэж	'to die-IPFG'	sax ^j əč	сахиж	'to guard–IPFG'
Gaixəš	гайхаш	'surprise'	ax ^j əš–gui	ахишгүй	'irreproducible'
nəxəs	нохос	'dogs'	cəx ^j əs	зохис	'propriety'
uxəg	үхэг	'chest'	sax ^j əg	сахиг	'to guard–IMP'
šaxəm	шахам	'nearly'	t ^h ax ^j əm	тахим	'back of the knee'
uxən	ухна	'billy-goat'	sax ^j ən	сахина	'to guard-NPST'
saixəŋ	сайхан	'beautiful'	əx ^j əŋ	охин	'girl'
saxəb	сахал	'beard'	sax ^j əţ	сахил	'vow'
uxər	үхэр	'ox'	t ^h ax ^j ər	тахир	'curved'
uxəw	үхэв	'to die–pst'	sax ^j əw	сахив	'to guard–pst'
unon	THOP		Suron	cuille	to guild 151
t ^h agt ^h	тагт	'balcony'			
pəgəč ^h	богоч	'front leg tie'			
pagət	багад	'small-DAT'			
pagəč	багаж	'tool'			
cagəs	загас	'fish'			
sagəg	сагаг	'buckwheat'			
šugəm	шугам	'line'			
pagən	багана	'pillar'			
tagəŋ	даган	'to follow–cv'			
t ^h ugəlz	тугал	'calf'			
paigəlz ^j	байгаль	'nature'			
agər	агар	'cypress'			
łagw	лхагва	'Wednesday' (pr	onounced [4	aw ^{w],} see 2.4	5)
10000	Jinai Da	Wednesday (pr	onounced [n	ID], 500 2)
thagth	тагт	'capped'			
igč ^h	ЭГЧ	'elder sister'			
pogch	богц	'saddle bag'			
poge	богд	'holy'	pog ^j t	богьд	'experienced-DAT'
vgč	угж	'feeding-bottle'	uug ^j č	уугьж	'to smoke-IPFG'
t ^h agc	угж тагз	'hovel'	5050	<i>y</i> y x b <i>n</i>	to onlone in to
pagš	багш	'teacher'			
sags	сагс	'basket'	cag ^j s	загьс	'let foal suckle
Sugo	car c	Jusitet	cue o	Surbe	other foal's
					mother'
					mound

SYLLABIFICATION AND EPENTHESIS

ogəg igəm	ӨГӨГ ЭГЭМ	'to give–імр' 'collar-bone'	c ^h əg ^j əg aag ^j əm	цогиг аагим	'to canter–імр' 'burning hot'
ogən	өгнө	'to give–NPST'	pəg ^j ən	богино	'short'
ogəŋ	өгөн	'to give–cv'	ບບg ^j ອກ	уугин	'to smoke–cv'
šugəb	шүгэл	'whistle'	pag ^j ə৳	багил	'grey-black'
šugər	Шүгэр	(a name)			, <u> </u>
ogəw	өгөв	'to give–psr'	uug ^j əw	уугив	'to smoke–pst'
xamt ^h	хамт	'together'	am ^j t ^h	амьт	'living'
imč ^h	ЭМЧ	'doctor'	am ^j č ^h	амьч	'egoist'
c ^h amc ^h	цамц	'shirt'			C
ķimp	лимбэ	'flute'			
x ^j amt	хямд	'cheap'	am ^j t	амьд	'alive'
simč	сэмж	'omentum'	xum ^j č	хумьж	'to gather-IPFG'
oims	оймс	'sock'	am ^j s	амьс	'to breathe'
č ^h imx	чимх	'pinch'			
		-	t ^h am ^j x ^j	тамхи	'tobacco'
t ^h aməg	тамга	'seal'			
aiməg	аймаг	'district'	xum ^j əg	хумиг	'to gather-IMP'
omən	өмнө	'southern'	xum ^j ən	хумина	'to gather–npst'
Gəiməŋ	гоймон	'noodles'	xum ^j əŋ	хумин	'to gather-cv'
taaməb	даамал	'controller'	t ^h əm ^j əb	томил	'to appoint'
xamər	хамар	'nose'	t ^h am ^j ər	тамир	'strength'
niməw	нэмэв	'to add-PST'	xum ^j əw	хумив	'to gather-pst'
ហ n t ^h	унт	'to sleep'	ən ^j th	оньт	'notched'
anč ^h	анч	'hunter'	sən ^j č ^h	соньч	'curious'
Ganch	ганц	'single'	xan ^j c ^h	ханьц	'to associate'
tont	дунд	'middle'	gən ^j t	гоньд	'caraway'
			pan ^j t ^j	банди	'novice'
tanč	танаж	'to trim–IPFG'	t ^h an ^j č	таньж	'to recognize-IPFG'
			man ^j č	Манж	'Manchu'
panc	банз	'board'			
tans	данс	'account'	ən ^j s	оньс	'lock'
			t ^h an ^j x ^j	танхи	'spoiled'
UNƏG	унага	'foal'			
xənəg	хоног	'day and night'	t ^h an ^j əg	таниг	'to recognize-IMP'
sanəm–	санам-	'unimaginable'			
gui	гүй				
unən	унана	'to fall–npst'	t ^h an ^j ən	танина	'to recognize– NPST'
unəŋ	үнэн	'truth'	sən ^j əŋ	сонин	'news'
ənəb	онол	'theory'	t ^h an ^j əb	танил	'acquaintance'
unər	үнэр	'smell'	an ^j ər	анир	'noise'

82

6.7 FINAL CONSONANT COMBINATIONS

unəw	унав	'to fall-PST'	t ^h an ^j əw	танив	'to recognize-PST'
បŋš	унш	'to read'			
с ^h эŋx	цонх	'window'			
m ^j aŋg	мянга	'thousand'			
moŋg	мөнгө	'silver'			
aŋg ^j	анги	'class'			
alzth	алт	'gold'	calz ^j t ^h	зальт	'cunning'
ilʒč ^h	элч	'messenger'	xuulz ^j č ^h	хуульч	'lawyer'
suutsch	суулц	'to attend'	xəß ^j c ^h	хольц	'mixture'
šilzp	шилбэ	ʻshin'			
			xəbibi	холби	'off'
poţst	болд	'steel'	xʊʊԷ ^j t	хуульд	'law-dat'
eeţč	ээлж	'turn'	рэђ ^ј č	больж	'to stop–IPFG'
aabc	аалз	'spider'			
t ^h uţzš	түлш	'fuel'			
ußs	улс	'state'	xalz ^j s	хальс	'peel'
t ^h ał3x	талх	'bread'			
			salz ^j x ^j	салхи	'wind'
xaabəq	хаалга	'door'			
atzəg	алаг	'motley'	šalz ^j əg	шалиг	'obscene'
vţəm	улам	'more'	alz ⁱ əm	алим	'apple'
okən	олно	'to find-NPST'	pɔk ^j ən	болино	'to stop-NPST'
ວ່ຽອກ	олон	'many'	c ^h alʒ ^j əŋ	цалин	'salary'
tuukak	дуулал	'song'	t ^h ək ^j ək	толил	'to reflect'
pəţər	болор	'crystal'	tək ^j ər	долир	'cross-eyed'
Galzəw	галав	'era'	pɔ͡k _j əw	болив	'to stop–pst'
urth	урт	'long'	mər ^j t ^h	морьт	'mounted'
girč ^h	гэрч	'witness'	mər ^j č ^h	морьч	'horseman'
aarch	аарц	'curds'	sər ^j c ^h	сорьц	'sample'
art	ард	'people'	ur ^j t	урьд	'before'
čurč	жүрж	'orange'	ur ^j č	урьж	'to summon–IPFG'
Garc	гарз	'loss'	Gar ^j C	гарьз	'latrine'
xarš	харш	'palace'	ur ^j š	урьш	'to get warmer'
tars	дарс	'wine'	sər ^j s	сорьс	'onion'
curx	зүрх	'heart'			
			ar ^j x ^j	архи	'liquor'
tarəg	дарга	'chief'			
curəg	зураг	'picture'	cər ^j əg	зориг	'courage'
turəm	дүрэм	'rule'	xur ^j əm	хурим	'wedding'
tərən	дорно	'east'	jar ^j ən	ярина	'speak–NPST'
xurəŋ	хүрэн	'brown'	xar ^j əŋ	харин	'but'

83

84

SYLLABIFICATION AND EPENTHESIS

arə l z arəw	арал арав	ʻisland' ʻten'	GUI ^j əH jar ^j əw pər ^j əw ^j	гурил ярив борви	ʻflour' 'to speak–PST' 'skin bag'
šuwt ^h t ^h owč ^h c ^h uwc ^h suwt guwč	шувт товч цувц сувд гувж	'through' 'button' 'path' 'pearl' 'stick at dog's neck'	xaw ^j t ^h xaw ^j č ^h t ^h aw ^j c ^h gow ^j t caw ^j č	Хавьт Хавьч Тавьц Говьд Завьж	'to be near' 'to chatter' 'stand' 'semi-desert-DAT' 'corner of the mouth'
nowš taws sawx	новш давс савх	ʻgarbage' ʻsalt' ʻchopsticks'	xaw ^j s	хавьс	ʻribs'
č ^h awəg	чавга	ʻplum'	cəw ^j x ^j	зовхи	'eyelid'
suwəg awən	суваг авна	'canal' 'to take–NPST' 'soap'	t ^h aw ^j əg t ^h aw ^j ən xuw ^j əŋ	тавиг тавина хурин	'offering' 'to put–NPST' 'bucket'
sawəŋ owəԷ owər	саван өвөл өвөр	'winter' 'breast'	xuw ^j əb aw ^j ər	хувин хувил авир	'to change' 'character'
awəw	авав	'to take-pst'	t ^h aw ^j əw	тавив	'to put-pst'
sajt ^h Gəjč ^h	саят гоёч	'millionaire' 'partiality'			
Gojt	гоёд	'elegant–DAT'			
ojč	оёж	'to sew-IPFG'			
ajc	аяз	'melody'			
ujs	yec	'time'			
sajx	саях	'recently'			
ajəg	аяга	'cup'			
xujəg	хуяг	'armour'			
ojən	оёно	'to sew-npst'			
nəjəŋ	ноён	'lord'			
səjəb	соёл	'culture'			
xəjər	xoëp	'two'			
υjəw	уяв	'to tie–PST'			

PROSODY¹

In this chapter we give a phonological analysis of intonation in Mongolian, and also discuss the problem of word stress. As in other languages, intonation has two major functions: the prominence-lending function, which highlights an important constituent (focus) of the utterance, and the boundary-signalling function, which divides the utterance into smaller groups (prosodic phrases). The tonal gestures associated with the two functions are focal accent and boundary tone, respectively.

We analyse the functionally relevant gestures within autosegmental theory (see e.g. Ladd 1996). The main acoustic correlate of intonation is fundamental frequency, F0. The tonal gestures are abstracted to two distinctive events, Low (L) and High (H), and the tonal courses are regarded as linear sequences of them. The intonational inventory of a language consists of prominence-lending tonal gestures and boundary tones.

The data for this chapter are recordings of the three male Ulaanbaatar speakers BB, DD, and HB (see Chapter 1) and of two female Ulaanbaatar speakers, Sarantujaa (ST) and Sarancacral (SC). For the analysis of question intonation, we used recordings of seven Ulaanbaatar speakers, four men and three women. Acoustic analyses and figures showing wave-forms and intonation contours were made in *Praat*.

In this chapter we give a preliminary report of the main facts; a comprehensive analysis will be given in a forthcoming work by one of the authors (AMK).

7.1 FOCAL ACCENT

We distinguish two pragmatic types of focus, broad and narrow. A constituent can get narrow focus either by the pragmatic context, or by a lexical focus marker. Broad focus occurs in neutral utterances without narrow focus.

Focal accent was investigated using the material shown in (1) illustrating focusing of different constituents in a short declarative sentence. This material was read by the three male speakers BB, DD, and HB as the answer to different questions (also read by the speakers). When a specific constituent is asked for with a questionword, we regard that constituent as having narrow focus. Thus, there is narrow focus on the first word *aaw* 'father' in (1a), and on the second word *jama* 'goat' in (1b).

¹ For additional literature on prosody and intonation, see App. G, 7, p.223.

PROSODY

(1) Material for investigating focal accent in declaratives

(a) xin jama xar ^j ukon pee	aaw jama xar ^j ubən
who goat herd-NPST Q	father goat herd-NPST
Хэн ямаа хариулна бэ?	Аав ямаа хариулна.
'Who herds goats?'	'Father herds goats.'
(b) aaw juu xar ^j ukən pee	aaw jama xar ^j ulzən
father what herd–NPST Q	father goat herd–NPST
Аав юу хариулна бэ?	Аав ямаа хариулна.
'What does father herd?'	'Father herds goats.'

The intonation contours of the declarative sentences in (1) are shown in Figure 7.1. When the focused word is *aaw*, there is an F0 rise on the long vowel in this word, and when the focused word is *jama* with a short initial vowel, the first vowel has low, and the second one high F0. In both cases, this can be analysed as a LH (low–high) sequence, and in both cases the peak is the highest one in the sentence.

The synchronization of the focal LH tone with the segmental level is best described with a mora analysis. Assuming that Mongolian short and epenthetic vowels are counted as one mora, and that long vowels are counted as two moras,



FIGURE 7.1 The utterance *aaw jama xar^jvţən* 'Father herds goats' with focus on *aaw* (top) and on *jama* (bottom). Speaker BB.

the high tone is placed on the second mora of the focused word, and the low tone on its first mora. This pattern is seen in the recordings of all speakers, the only variation being that a long initial vowel in a focused word may have an overall high tone instead of a rise. We regard this as a variant phonetic realization of the narrow focus LH tone, a realization that seems to be more common for male speakers than for women. An example is given in Figure 7.2 (top), which shows the focused word *paabara* with a high tone extending over the entire long vowel in the initial syllable. Figure 7.2 (bottom) with the word *pabpabctagar* illustrates the fact that there is a high tone on the second syllable of a focused word with short initial vowel, even when the second vowel is an epenthetic schwa. These words were said in focused position (before the final verb) in the carrier sentence *pii* _____ *gisaŋ* 'I said ____'.

The peak of the rising tone that signals focal accent tends to be the highest one in the utterance, although this is not always the case. Following syllables can get higher frequencies, and it is the step from lower to higher F0 values, synchronized with the two first moras, which is relevant for signalling focal accent.



FIGURE 7.2 The focused words *paaţara* баалаараа 'lap-dog-INST-RFL' (top) and *paţpaţct(a)gar* балбалздагаар 'to shiver-навітиаL-INST' (bottom) said in a carrier sentence. Speaker DD.
PROSODY

The excursion of the focal rise on a long vowel has approximately the same size as the difference in F0 between the second and the first syllable in words where the first vowel is short. This supports the analysis that the tonal gestures are identical in these two cases, and is a further argument for describing the focal gesture as a LH synchronized with the two first moras of a focused word.

When focus is on the last word of an utterance, usually a verb since Mongolian is a verb-final language, it is not marked with the rising focal accent tone. The last word can be focused by other phonetic means, however, usually by lengthening of the final vowel or sonorant (cf. Poppe 1951*a*: 14).

The sentence in (2), similar to that in (1) but having longer constituents, shows a more complicated picture since the tonal course reflects grouping of words into prosodic phrases in addition to signalling focus. This utterance can be seen as consisting of three prosodic phrases: *manai aaw* 'our father', $2\beta \partial \eta jama$ 'many goats', and $vv\beta ar xar^jv\beta \partial n$ 'herds in the mountains'. In this case, the prosodic and syntactic groupings correspond to each other. The focused prosodic phrase has a LH tone with the highest F0 value in the utterance, signalling focal accent (see Figure 7.3). Within each prosodic phrase, the second mora of the first word usually has the highest F0 peak, even if it does not have pragmatic prominence (as *manai* 'our' in the prosodic phrase *manai aaw* 'our father'). This suggests that the F0 rise has another function in addition to focal accenting, namely signalling



FIGURE 7.3 The sentence manai aaw ɔʰəŋ jama uuʰər xarⁱuʰən 'Our father herds many goats in the mountains' with focus on manai aaw 'our father' (top), and on ɔʰəŋ jama 'many goats' (bottom). Speaker BB.

of a new prosodic phrase, as investigated in more detail in section 7.3.

 (2) manai aaw эдэŋ jama uutar xarⁱutan our father many goat mountain-INST herd-NPST Манай аав олон ямаа уулаар хариулна.
 'Our father herds many goats in the mountains.'

Narrow focus can also be signalled lexically by enclitic focus markers, such as $\check{c}^h \,\mathfrak{q}$ and $\check{b} \,\mathfrak{n}$. This is exemplified in (3).

(3)	(a)	inč ^h	nom saiŋ	(b)	in	nəmč ^h	saiŋ
		this-FOCUS	s book good		this	s book-Fe	ocus good
		Энэ ч ном	сайн.		Эн	э ном ч с	сайн.
		'This book	is good.'		'Th	is book is	s good.'

In an utterances with a lexical focus marker, the focused word has an F0 rise as well. This is illustrated in Figure 7.4, where the sentences in (3) are contrasted. In both cases, focus is signalled lexically by the marker \check{c}^h and prosodically by a tonal rise (focal LH). In the second sentence, there are two rising gestures, an initial boundary tone (which we denote with [LH; see 7.3) on *in* 'this', and a focal rise LH on *nom* 'book'.



FIGURE 7.4 The utterances $in\check{c}^h n m sain$ 'This book is good' (top) and $in n m\check{c}^h sain$ 'This book is good'. Speaker ST.

7.2 FINAL PROMINENCE TONE

A Mongolian utterance often has a tonal downstep to a low pitch which spreads over the rest of the utterance. This occurs in neutral utterances without narrow focus as well as in utterances with narrow focus marked by a focal rise LH. An example is given in Figure 7.5, which shows the intonation contour of the utterance in (4).

(4) t^hir oor xuuxənt^he t^han^jə¿c^hsənar parəxgui, t^huunt^he he other woman-сом get.to.know-PSTP-INST not.only, she-сом suusəŋ marry-PSTP
 Тэр өөр хүүхэнтэй танилцсанаар барахгүй, түүнтэй суусан. 'Not only did he get to know another woman, he even married her.'

The second part of this utterance, $t^{h}uunt^{h}e subset `[he]$ married her', is pronounced with low F0 (below 200 Hz for this female speaker). This part of the utterance is semantically and pragmatically important, so the low tone cannot be regarded as signalling deaccentuation. We analyse it as a prominence signalling gesture, a low tone (L) that spreads over the final part of an utterance, often a long constituent which is semantically important.

This low prominence tone is very common. The step from the previous part of the utterance can be as much as 200 Hz for female speakers, which is perceptually very salient.

We believe that the existence of this prominence tone is related to the information structure and syntactic structure of Mongolian. Being a strictly verb-final language with SOV (subject-object-verb) word order, Mongolian tends to place the rheme at the end of an utterance. The final verb is pragmatically important, and the position before the final verb is the syntactic focus position, as in Turkish and Hungarian, so that a constituent can be focused by moving it to this position. We hypothesize that the low tone is a default device for marking the rheme of a sen-



FIGURE 7.5 The utterance t^{hir} oor xuuxant^he $t^{h}an^{j}a^{k}_{5}c^{h}s(a)$ nar paraxgui, $t^{h}uunt^{h}e$ suusan 'Not only did he get to know another woman, he even married her'. Speaker ST.

tence, used when other types of final focus marking are not employed. The fact that the final verb cannot get the focal rise LH supports this interpretation.

7.3 BOUNDARY SIGNALLING

Long utterances are often divided into two or more prosodic phrases, entities with prosodically defined boundaries. This was investigated using material recorded with the two female Ulaanbaatar speakers ST and SC. Most utterances have an F0 rise (analysed as a low-high sequence) on the initial word, but occasionally the peak of the rise occurs on the following word, if the first word is monosyllabic (e.g. a personal pronoun like *pii* 'I'). The rising gesture is also found non-initially in longer utterances and is often placed on semantically unimportant words. One example is the sentence in (5), whose intonation contour is shown in Figure 7.6.

(5) pii in t^hakar экэŋ хэлэд хоос^hәktsəŋ рэкәwč^h, t^hitnər is I this matter–INST many day pursue–PSTP although, they not cowšorəw agree–PST
Би энэ талаар олон хоног хөөцөлдсөн боловч, тэд нар эс зөвшөөрөв. 'Although I worked on this matter diligently for many days, they did not agree.'

Three tonal rises can be observed. Two of them are placed on function words, the pronouns pii 'I' and $t^{h}itn \partial r$ 'they'. We analyse these gestures as non-prominence related phrase-initial boundary tones represented as [LH. There is a third rising tone between these two, placed on the semantically important word $\partial \xi \partial \eta$ 'many'. This rise signals a focal accent (LH). It is followed by the low prominence tone L.

The F0 rise thus has two different functions, signalling a prosodic boundary, [LH, or signalling narrow focus, LH. This is illustrated clearly in Figure 7.4, where narrow focus is marked explicitly with the marker \check{c}^h . In the sentence *in* \check{c}^h *nom sain 'This* book is good' with focus on *in* 'this', there is only one rising tone, signalling



FIGURE 7.6 The sentence pii in $t^ha\xi ar \, z\xi \partial \eta \, x z n \partial g \, x z o c^h \partial \xi t s \partial \eta \, p z \xi \partial w c^h$, $t^h itn ar is \, cow s or \partial w$ 'Although I worked on this matter diligently for many days, they did not agree'. Speaker ST.

both focus and boundary, but in the sentence in $n \supset m\check{c}^h$ sain 'This book is good', with focus on $n \supset m$ 'book', these functions are separated, the rise on in signalling boundary, and the rise on $n \supset m$ signalling focus.

In subordinate clauses, boundary signalling may occur phrase-finally as well, placed on postposed conjunctions. We represent this as an optional phrase-final boundary tone LH]. This is illustrated in Figure 7.7 which shows the intonation contour of the sentence in (6).

 (6) piket^h toosson oč^hras, pii kino ucsongui ticket finish-pstp because, I film see-pstp-negation Билет дууссан учраас, би кино үзсэнгүй.
 'Since the tickets were sold out, I did not see the film.'

This utterance is divided into two prosodic phrases: $pi\xi et^{h}t uussan u\xi^{h}ras$ 'since the tickets were sold out' and piikina uusangui 'I did not see the film'. The first phrase is signalled by an initial boundary tone [LH on $pi\xi et^{h}$, which also has a focal function here. At the end of the phrase there is a final boundary tone LH] on $v\xi^{h}ras$. The next phrase is signalled by an initial [LH boundary tone on pii. The expected final low prominence tone is replaced by a focal rise in this sentence, realized on the negation -gui, which optionally can take the focal rise, unlike sentence-final verbs. Like the fact that this suffix does not follow vowel harmony (5.2.3), this is an indication that the negation is not a suffix but a compound-forming element.

Relatively small pitch rises sometimes occur on non-initial and non-focused constituents of a prosodic phrase within longer utterances. An example is the sentence manai aaw $2\beta a\eta$ jama $20\beta ar xar^j Q\beta an$ 'Our father herds many goats in the mountains' (2). The focal accents LH and boundary tones [LH are realized on the initial words manai and $2\beta a\eta$ of the prosodic phrases manai aaw 'our father' and $2\beta a\eta$ jama 'many goats' (Figure 7.3), but the non-initial words aaw 'father' and jama 'goat' have small rising F0 gestures. Like the focal accent and boundary tone, they are aligned with the two first moras, and may have the function of signalling word boundaries.



FIGURE 7.7 The sentence $pi\xi et^h$ to uss of $u\xi^h ras$, pii kino ucsongui 'Since the tickets were sold out, I did not see the film'. Speaker ST.

In conclusion, the division of an utterance into prosodic phrases is normally signalled by an initial boundary tone [LH. This gesture is perceptually salient, having an excursion up to 150 Hz for female speakers, giving a very characteristic melody to Mongolian declaratives. Utterance-internal prosodic phrases can be signalled by an optional final rise LH] as well.

7.4 QUESTIONS²

Mongolian has strong formal markers for questions. Yes–no questions have an obligatory sentence-final question marker -(j)U which is cliticized to the final word of the sentence (for an example, see (7a)). It is written as a separate word yy/ $ioy/\gamma\gamma/io\gamma$ in Cyrillic Mongolian; the vowel follows vowel harmony, and *j* is added if the preceding word ends in a vowel. Question-word questions have a sentence-final marker *pee/wee* 63/B3 in addition to the question-word, but it can be omitted in colloquial speech. Question-words are not moved to the sentence-initial position (see the examples in (1) above).

The focused words in yes–no questions and question-word questions are signalled by the same tonal gesture, so we will treat them together. For analysing question intonation we used recordings of seven Ulaanbaatar speakers, four men and three women. The focused word is signalled by a focal rising tone (LH), and a question-word is always focused. The phonetic realization of the focal rise in questions is a bit different from its realization in declaratives. If a word has a short vowel in the initial syllable, and contains a non-initial diphthong, or ends in a vowel, the rise is not synchronized with the two first moras, but is realized on the diphthong or final vowel. In other cases, the focal rise is realized as in declaratives. This is illustrated in Figure 7.8, where the question $t^{hir} torčin pagšo$? 'Is he Dorj's teacher?' is compared with the declarative sentence $t^{hir} torčin tiwt^{h}or$ 'This is Dorj's notebook' (7). In both cases, the focused word torčin 'Dorj's' is signalled by a tonal rise synchronized with the two moras in the word.

(7)	(a)	t ^h ir tərčiŋ	pagšu	(b)	in	tərčiŋ	tiwt ^h ər
		he Dorj-GEN	teacher-Q		this	Dorj-gen	v notebook
		Тэр Доржий	ін багш уу?		Эн	э Доржий	ін дэвтэр.
		'Is he Dorj's	teacher?'		'Th	is is Dorj'	s notebook.'

The question marker -(j)U is cliticized to the final word of the yes-no question, and can occupy the second mora position in the resulting cliticized word. For example, the word *ucəg* yəər 'pen' becomes *ucgu* yəər yy (formed by the syllabification rule (6) in 6.2) when the question marker has been attached to it, with the marker forming the second mora. In consequence with this, the high tone of the focal LH gesture is aligned with the question marker in the sentence $t^{hir} ucgu$? 'Is that a pen?'

² Bjuraeva and Pavlova (1976); Mohosoeva (1978; 1979; 1981; 1983; 1984); Esenova and Mandžieva (1986); Žargalov (1986; 1987*a*, *b*; 1988); Karlsson (2003*a*).



FIGURE 7.8 The question $t^{hir} torčin pagšv$? 'Is he Dorj's teacher?' (top) and the declarative *in torčin tiwt^har* 'It is Dorj's notebook' (bottom). Speaker E.

(Figure 7.9). In the question *in šireju*? 'Is this a table?', the second mora coincides with the second vowel in the word *šire* 'table', and consequently the high tone is aligned with that vowel.

The final marker in both types of questions can get a tonal rise, analysed as a final boundary tone LH]. This final rise is used systematically by two of our seven speakers (see Fig. 7.10 for an example).

Another feature which is characteristic for questions (at least for those with up to five words), is that they contain no pauses. In declaratives, pauses are relatively frequent, especially between the theme and the rheme (often the subject and the predicate phrase), and in complex sentences between the clauses.

7.5 WORD STRESS³

In languages with contrastive stress, word stress is usually signalled by a combination of phonetic correlates, including fundamental frequency (F0), duration, and vowel quality. Intensity may be involved as well.

³ For sources and notes on word stress, see App. G, 7.5, p. 223.



FIGURE 7.9 The yes-no questions t^{hir} ucgu? Təp үзэг үү? 'Is that a pen?' (top) and in *šireju*? Энэ ширээ юү? 'Is this a table?' (bottom). Speaker ST.



FIGURE 7.10 The question *in juni nom pee*? Энэ юуны ном бэ? [this what book q] 'What kind of book is this?' Speaker ST.

PROSODY

We have shown earlier in this chapter that fundamental frequency is used for signalling prominence and boundaries on the prosodic phrase level, but it has little, if any, relevance on the word level.

Vowel duration and vowel quality were both investigated in section 1.1. Their main function is paradigmatic, distinguishing long and short vowels in initial syllables and distinguishing phonemic and epenthetic vowels in non-initial syllables. None of these phonetic properties gives prominence to any particular syllable position in a word, since long vowels occur only in the initial syllable, and the qualities of initial and non-initial short vowels are approximately the same, as is their duration, although non-initial short vowels are slightly longer than initial short vowels. Thus both vowel quality and duration depends on the phonological status of the vowel and not on its position in relation to a stressed syllable, as is the case in languages, such as Russian or English, which have reduction of unstressed vowels.

The initial vowel is phonologically strong due to its decisive role in vowel harmony. Because of vowel harmony, and because long and short vowels contrast only in initial syllables, the number of contrasting vowels is very much larger in initial than in non-initial syllables. This may be interpreted as giving phonological prominence to the initial syllable, making it a plausible candidate for carrying stress. This is not supported by the data on F0, duration, and vowel quality, however. In our spoken material, there are several cases of almost complete reduction of word-initial vowels, which is not compatible with the view that this vowel is stressed.

Our conclusion is that word stress is not phonologically relevant in Mongolian. The different phonetic properties which signal a stressed syllable in other languages, especially in languages with contrastive stress, have different functions in Mongolian and do not converge to form a unified phenomenon of word stress in this language.

One indication that this view is the correct one is that there are widely differing opinions on the place and nature of word stress in Mongolian which have been expressed in the literature. The only consensus is that stress is not contrastive, so that no two words can be distinguished only by the place of stress. On the other hand, it is usually taken for granted that there is stress of some kind, but different authors disagree about the place of the stressed syllable in a word. In our opinion this is a spurious question, but we will nevertheless review the literature dealing with it briefly. There are at least six different opinions about the place of Mongolian stress in the literature:

(1) Stress falls on the first syllable. This is the opinion of almost all native Mongolian scholars who wrote on this subject, and it is shared by many other Mongolists who describe Mongolian proper or Buriad.⁴ According to Cenggeltei

⁴ Schmidt (1831: 14); Ramstedt (1902: 50); Rudnev (1905; 1911; 1913–14: XXXI); Kljukin (1926); Vladimircov (1929: 64, 97); Poppe (1931; 1936: 8; 1937: 18; 1938: 62); Grønbech (1940); Todaeva (1951: 40); Sanžeev (1959: 20); Cenggeltei (1959; 1979*a*: 98); Luvsanvandan (1966*b*; 1980*b*); Coloo (1973*b*; 1976: 153); Möömöö (1974); Qaserdeni et al. (1996: 122).

- (1959), Baarin Mongolian has stress on the first syllable, realized as low tone. Several of these authors, including Ramstedt (1902), Vladimircov (1929), Poppe (1931: 9), and Coloo (1973b), say that there is a secondary stress (often described as musical or tonal) on the last syllable. Zolhoev (1962) says that Buriad word stress falls on the first syllable, and phrase stress on the last syllable of the first word of a phrase. Bitkeev (1980) says that the first syllable has relatively long duration and the last syllable has high pitch in Kalmuck.
- (2) If the first vowel is long (or a diphthong) or if it is the only phonemic vowel, it carries the stress. If the first vowel is short and there are other phonemic vowels in the word, the first of these carries the stress. Under the traditional vowel analysis (a) in section 3.1.1 (our phonemic non-initial vowels are long, epenthetic vowels are phonemic and short), this can be expressed as: stress falls on the first syllable which contains a long vowel or diphthong; if there are no long vowels, stress falls on the first syllable.⁵
- (3) The same as (2) with the modification that stress falls on the penultimate phonemic vowel if there are two or more phonemic vowels in non-initial syllables (Bosson 1964: 21; Poppe 1970; Walker 1995; 1996). For example, opinion (3) places stress on the penultimate syllable in words like [šagšur'ara] шагшуураараа 'biscuit-stick-INST-RFL', while opinion (2) puts the stress on the antepenultimate: [šag'šurara].
- (4) Stress falls on the final syllable. A. A. Bobrovnikov (1849: 33) says that stress, realized as a pitch rise, always falls on the last syllable in Mongolian. Orlov (1878) makes the same assertion about Buriad. Pozdneev (1879: 179) and Cybikov (1915: 7) say that most words are stressed on the final syllable. According to Yi (1964; 1991), Mongolian loans in sixteenth-century Korean always have a high tone mark on the final syllable.
- (5) Stress falls on the final syllable, but if the word contains a long vowel or diphthong, it takes the stress. This view is held by A. I. Bobrovnikov (1835) and Kowalewski (1835).
- (6) Stress falls on the initial syllable if its vowel is long; and if it is short, stress falls on the second syllable (Köke 2003*a*, *b*).

The reason for this situation is apparently that different criteria and different phonetic correlates have been used for defining stress. Several investigators have come to the conclusion that stress is difficult to define or that there are several kinds of stress: Cybikov (1908: 5); Zolhoev (1961; 1970); Gerasimovič (1970; 1975); Bitkeev (1980); Svantesson (1990; 1992). In our opinion, word stress is not a relevant concept in Mongolian phonology, and any definition of it must be arbitrary, depending on the individual investigator's choice of criteria.

⁵ This is the opinion of Sturt (1941: 4), Aberle and Austin (1951), Hattori (1951), Poppe (1951*a*: 13; 1960*a*: 143), Stuart and Haltod (1957: 68), Street (1963: 62), Beffa and Hamayon (1975), etc. This view has been adopted in several theory-oriented works using Mongolian data, e.g. Hyman (1977: 52); Odden (1979: 158); Prince (1983: 75ff.); Hayes (1985: 63f.); Hammond (1986: 196ff.); Halle and Vergnaud (1987: 71f.); Goldsmith (1990: 187f.); Kenstowicz (1994: 582, 607); Halle and Idsardi (1995: 413f.); Ahn (1999: 21f.).

OLD MONGOLIAN

Four different scripts, Uigur, Chinese, Arabic, and 'Phags-pa, were used for writing Mongolian in the thirteenth to the fifteenth centuries. We use the term Old Mongolian for the immediate ancestor language that can be reconstructed from these four kinds of sources. The language of the Uigur script documents differs somewhat from the language of the other three sources, which is known as Middle Mongolian (Vladimircov 1929: 47; Poppe 1955: 15). The languages of the Chinese (Sino-Mongolian) and 'Phags-pa Mongolian sources are more or less identical, while Arabic Mongolian (sometimes called Western Middle Mongolian) has some divergent features that remind us of modern Oirad. In addition to these four main sources for Old Mongolian, some Mongolian words can be found in Armenian, Georgian, Korean, and other texts, but they are of less importance for Mongolic historical phonology. There is a large secondary literature about the oldest Mongolian texts, most of which have been published in a form that makes them available for non-specialists in Mongolian palaeography. We have benefited greatly from these publications; references are given in the relevant places.

Uigur Mongolian preserves Old Mongolian medial *h, which is not shown in the three other scripts. On the other hand, they preserve initial *h, not written in Uigur Mongolian. Thus, both Uigur Mongolian and Middle Mongolian have lost some archaic features and have retained others. For this reason, and also because the oldest texts in Uigur Mongolian and Middle Mongolian are almost contemporaneous, it is not possible to assume a linear development from Uigur Mongolian via Middle Mongolian to the modern languages (Doerfer 1964*a*: 37; Weiers 1970*b*; de Rachewiltz 1999). The sources probably write different dialects, which have developed in different ways from the ancestor language (cf. Poppe 1960*a*: 2f.). We will avoid the rather misleading term Middle Mongolian.

Reconstructions of a Mongolian ancestor language are given explicitly or implicitly by Poppe (1955; 1960*a*; 1965; 1976), Ramstedt (1957), Clauson (1962), Tömörtogoo (1992), Darbeeva (1996), and others. These reconstructions are mainly based on the traditional transcription of Written Mongolian, which partly relies on the modern Mongolian pronunciation (see 4.3). In addition, initial **h* (or an equivalent consonant) is reconstructed in the appropriate words, using the 'Middle Mongolian' sources. In our opinion, the bias towards Uigur Mongolian and modern Halh makes these reconstructions somewhat inappropriate as the point of departure for describing the development of the modern languages. Since a reconstruction of Proto-Mongolic, using all the modern and ancient sources, is still premature, we have chosen to work with the four kinds of thirteenth to fifteenth century Mongolian texts, and reconstruct Old Mongolian only from them. The resulting reconstruction is, however, not very different from the others, and many of the differences that do occur are due to different analyses rather than to the choice of sources.

Although Old Mongolian as defined here is not identical to Proto-Mongolic, it is very close to it, and Old Mongolian is taken as the point of departure when we describe the development to the modern languages in Chapter 10. Old Mongolian has the advantage of being clearly defined in a way that is independent of the modern languages. Furthermore, it reflects the language of Chinggis Khan's time, or somewhat earlier; it seems that the language became unified at this time, when the Mongolian state and the Mongolian written language were created (Janhunen 1998). On the other hand, Proto-Mongolic may have had properties which are not reflected in Old Mongolian, but are reflected in some modern Mongolic languages. Such properties seem to be few and difficult to reconstruct. The existence of long vowels may be an example (see 8.5.1).

Words from the sources for Old Mongolian are given in a strict transliteration of the graphemes, using different letters only when the writing system uses different graphic symbols. For example, unlike the traditional transcription of Written Mongolian, we always transliterate the Uigur Mongolian letter $\gamma_{\overline{v}}$ as *u*, although it did denote two different Old Mongolian vowel phonemes, **u* and **o*. When talking about the Uigur Mongolian vowel system, for example, we mean the system implied by the vowel letters (graphemes), not the actual vowel phoneme system used by the speakers who wrote their language with the Uigur Mongolian script. Arabic Mongolian and 'Phags-pa Mongolian are transliterated by the same principles, whereas the non-alphabetic character of the Chinese script requires another solution for Sino-Mongolian (see 8.2).

The Old Mongolian sources tell us nothing about prosodic features such as stress or intonation, moreover, and the information on this is inadequate for most modern languages, so we will not speculate about this subject.

All reconstructed Old Mongolian forms used in this book are given in section 8.9 together with their sources.

8.1 UIGUR MONGOLIAN

The Uigur Mongolian alphabet is an adaptation of the script used by the Turkic Uigurs in Central Asia from the ninth to the sixteenth centuries. The Uigurs had borrowed it from the Sogdians, an Iranian people, who had in their turn taken it from Syrian. According to Chapter 124 of the *History of the Yuan dynasty (Yuan shi)*, compiled in 1370–1 by Song Lian and others, the Uigur Mongolian script originated when Tatatong'a, an Uigur official who knew the written Uigur language well, was captured by the Mongols from the Naiman Khan in 1204 and was ordered to teach the princes how to use the Uigur alphabet for writing the Mongolian language (*Yuan shi*, vol. 10, pp. 3048–9 in the 1976 Beijing ed.). The Uigur script was

originally written from right to left in rows, like other scripts of Semitic origin, but probably under Chinese influence it was rotated 90° counter-clockwise, resulting in columns following each other from left to right. This way of writing was inherited by the Mongols.

The term Uigur Mongolian (UM) is used here for the oldest texts written in the Uigur Mongolian script during approximately the thirteenth to the fifteenth centuries, corresponding to the first stage of Pre-classical Mongolian in Weiers (1969). A criterion which may be used for identifying texts as belonging to this stage of the language is the use of the letter $\langle q \rangle \Im/\pi$ before $\langle i \rangle \nabla$ in back-vocalic words (Poppe 1975*a*; Higuchi 1999: 67ff.). Later texts will be referred to as (Classical or Modern) Written Mongolian (cf. 4.3).

The earliest extant inscription, known as *Chinggis Khan's stone*, is probably from 1227. The corpus of Uigur Mongolian texts is not very large. We have used mainly the collection by Dobo (1983*b*) in which the texts are reproduced in facsimile, copied and transcribed in Modern Written Mongolian, and provided with comments.

The Uigur Mongolian alphabet is shown in (1). Comparison with Table 4.2 shows that, except for the forms of the letters, the Modern Written Mongolian alphabet differs from Uigur Mongolian only by the introduction of letters with diacritic dots $(\gamma / \gamma < n > \text{ and } \gamma / \gamma < g >)$ and a few modified letters $(\neg < c > \text{ and } \gamma < y >)$ in order to differentiate previously homographic sounds.

Uigur	Modern	Uigur	Modern
1 + 5	i · · a	200	e e e t
. '	r e	्रा	ব d
1 1 3	オーンi	17 14 4	२४ २४ २४]
4 4 40	ব ব ত u	nno	ના ના નર M
ş ş	द्रं प्र у	ų i	ччČ
1	→ • ↓ 1	4	₁ j
+ + 1	\$ 1 1 9	5 5 j	っつう k
ممع	ჟ ჟ ტ ^ხ	* 5	ォゥ [『]
* * 4	≯ ₹ ₹ [§]	4 4	a a W
۰.	- Z		

(1) The Uigur Mongolian alphabet

Note: The table shows the Uigur forms and the corresponding Modern Written Mongolian forms of the letters at the beginning of a word, in the middle, and at the end.

Uigur Mongolian forms will be given here in strict transliteration, as shown in (1), and not in the traditional transcription (see 4.3). We use the IPA transcription y for the front high rounded vowel, usually written \ddot{u} in the Mongolist literature.

Because of the similarities of several letters, there are frequent ambiguities in Uigur Mongolian, and even letter-for-letter transliteration involves a certain amount of interpretation. For just one example, the graphic word γ and γ can, in theory, be transliterated in at least five ways: $\langle aqa \rangle$ ($\Im \pi \pi \checkmark$), $\langle anan \rangle$ ($\Im \pi \tau \checkmark$), $\langle naqa \rangle$ ($\Im \tau \pi \checkmark$), $\langle nanan \rangle$ ($\Im \tau \tau \tau \checkmark$), or $\langle enana \rangle$ ($\Upsilon \tau \tau \tau \checkmark$). Although the transliteration used here involves interpretation, it is reversible in the sense that the original spelling can be retrieved from it.

In most Uigur Mongolian texts, no diacritic dot is written on the letter $\langle n \rangle$ (γ / τ) to distinguish it from the vowel $\langle a/e \rangle$. When the 'dotted n' (γ / τ) appears, we transliterate it as $\langle n \rangle$. The letter $\hat{\gamma}$ (π) with two diacritic dots is used in a few Uigur Mongolian texts; we transliterate it as $\langle q \rangle$. In Modern Written Mongolian, the two dots indicate that the stop is unaspirated, but they seem to have no systematic meaning in the Uigur Mongolian texts. In word-final position, the sound s is often written as l, transliterated $\langle z \rangle$. A few Uigur Mongolian texts have the letter $\frac{1}{2} \langle \bar{x} \rangle$. The Uigur Mongolian grapheme system is shown in (2).

(2)i t/d č k b V u a e a slz m n 1 r w i

As in Modern Written Mongolian, Uigur Mongolian has five vowel letters. The script distinguishes between *e* and *a* only in word-initial position, and we will use <e> only for transliterating the word-initial letter \uparrow , contrasting with $\neg <a>$. The medial/final vowel letter \dagger / \checkmark is always transliterated as <a>. The vowel symbol $\sigma_{\nabla} < y>$ occurs only in initial syllables, so only the three vowels <i, a, u> occur in the transliteration of non-initial syllables, although we reconstruct non-initial **e*, **y* as well for Old Mongolian. For example, the Uigur Mongolian words written $\neg Oh \sigma /$ 'grass', $O \dashv '$ tongue', and $\neg \sigma_{\nabla} \dashv \sigma / '$ 'ice' are transliterated <ebasun>, <kala>, <mylsun> (the traditional transcriptions of these words are *ebesün*, *kele*, *mölsün*, and we reconstruct them as **epesyn*, **k*^h*ele*, **mølsyn* in Old Mongolian). When $\nabla / O <i>$ occurs between two vowels, we interpret it as a glide $<j>: \neg \sigma_{\nabla} / <sajin> 'good', <math>\partial O > d$ >baj-a> 'body'.

Only one series of stops/affricates is written in Uigur Mongolian. The letters φ ($\sigma\tau$, σ') and ς usually occur in disjoint environments in Modern Written Mongolian, so that ς is written medially before a vowel, and φ in all other positions. In the Uigur Mongolian texts, there are a few exceptions to this, and we transliterate φ as <t> and ς as <d>, although this probably does not correspond to a real sound contrast.

In some words, final *a* is written separately from the rest of the word (cf. 4.3). In the transcription this is shown with a hyphen $\langle -\rangle : \Im \mathcal{R} \supset \langle qar-a \rangle$ 'black'. If they are relevant, morpheme boundaries are shown with a dash $\langle -\rangle : \neg \Box \Im \nabla \langle qar-a \rangle$ 'black'. If they are relevant, morpheme boundaries are shown with a dash $\langle -\rangle : \neg \Box \nabla \nabla \langle qar-a \rangle$ 'black'. If they are relevant, morpheme boundaries are shown with a dash $\langle -\rangle : \neg \Box \nabla \nabla \langle qar-a \rangle$ 'black'. If they are relevant, morpheme boundaries are shown with a dash $\langle -\rangle : \neg \Box \nabla \nabla \langle qar-a \rangle$ 'to go-FUTP'. Uigur Mongolian does not indicate morpheme boundaries within words, except for case and reflexive suffixes which are written separately from the stem. We use a long dash $\langle -\rangle >$ to denote the combined graphic and morpheme boundary: $\Box \nabla \nabla \nabla \nabla \langle qar-a \rangle = 0$ (kyl-mijan) 'foot-RFL'.

OLD MONGOLIAN

8.2 SINO-MONGOLIAN¹

The oldest literary text in Mongolian is the *Secret history of the Mongols*. This is preserved only as a text where the Mongolian language is written with Chinese characters, but most scholars assume that the original was written in the Uigur script. According to de Rachewiltz (1965), the main part of the text (sections 1 to 268), dealing with Chinggis Khan's life and known as *Cinggis qagan-u ijagur* 'The origin of Chinggis Khan', was written in the Uigur script in 1228, and the remaining fourteen sections of the book were written after 1258. The transcription with Chinese characters was probably made in the 1390s (de Rachewiltz 1965: 204).

The Chinese–Mongolian glossary *Hua–Yi yiyu* 'Chinese–Foreign glossary' from 1389 is another source for Mongolian words transcribed with Chinese characters, basically in the same way as in the *Secret history*. There are also some other Sino-Mongolian glossaries of a similar type.

Words from the *Secret history* are cited from the edition by Kuribayashi and Coyijongjab (2001), which contains a reproduction of the standard *Si bu congkan* Chinese text, a transliterated text, and a complete word index. Forms from *Hua–Yi yiyu* are cited from Kuribayashi (2003).

Examples are given here in Chinese characters and in a transcription based on the Yuan dynasty northern Chinese pronunciation reconstructed as Early Mandarin by Pulleyblank (1984; 1991). This reconstruction is based on the *Zhongyuan yin-yun* 'Pronunciation of the Central Plain' written about 1324 by Zhou Deqing.

All Chinese characters occurring in examples cited in this book are listed in Table 8.1 together with their Yuan Chinese and Sino-Mongolian readings. Because each Chinese character represents a monosyllabic morpheme, only those syllables which occur in Chinese can be written with Chinese characters. As shown below, a number of devices are used in Sino-Mongolian to represent non-Chinese sounds.

8.2.1 Sino-Mongolian consonants

The system of onset consonants that can be inferred from the Sino-Mongolian transcriptions is shown in (3).

```
ťh
                čh
                      kh
(3)
                č
                      k
      p
           t
                š
                     х
           s
                           χ
      m
           n
           1
           r
                i
```

¹ For notes on Sino-Mongolian, see App. H, 8.2, p. 223.

TABLE 8.1 Sino-Mongolian transcriptions

	0	0.000	0.7	0.7		011/
	a	am	an	aŋ	aj	aw
—	阿っ	唵 am	安 an	昂 aŋ	埃 aj	奥 aw
р	巴把八 pa		班 pan	邦 paŋ	伯 paj	鴇保 pow
m	麻馬 ma		蠻 man	莽忙 maŋ	埋 maj	卯 maw
t	苔ta		丹 tan	當 taŋ	歹taj	擣 taw
t ^h	塔榻 t ^h a	探 t ^h am	壇 t ^h an	堂 t ^h aŋ	臺 t ^h aj	討 t ^h aw
S	撒 sa	毵三 sam	散 san	桑 saŋ	賽 saj	掃 saw
n	納 na	南 nam	難 nan		乃 naj	
1	刺 1a	藍 lam	闌 lan	郎 1aŋ	來 laj	老 law
č	札 tşa	站 tşam	盞 tşan	掌 tşaŋ	澤 tşaj	
čh	察 tşʰa	攙 tş ^h am	潺 tş ^h an	昌 tş ^h aŋ		抄 tş ^h aw
š	沙 şa				篩 şaj	
k	歌 kə	敢 kam	\pm kan		該 kaj	
$\mathbf{k}^{\mathbf{h}}$	可 k ^h o	坎 k ^h am	刊 kʰan	康 k ^h aŋ		
х	合哈 xo	含 xam	罕 xan	杭 xaŋ	孩 xaj	好 xaw
j	牙迓ja	黯 jam	顏 jan	陽 jaŋ		
w	瓦 wa	-	-			
	e	em	en	eŋ	ej	ew
_	額 jaj	奄jɛm	延jɛn			
р	別pjɛ		邊pjɛn			
m	篾mjɛ		綿 mjɛn			
t	, 迭咥 tjɛ	點 tjɛm	顛 tjɛn	登 təŋ	德 təj	
t ^h	帖 t ^h jɛ́	忝 t ^h jɛm	⊞ t ^h jɛn	騰 t ^h əŋ	5	挑 t ^h jɛw
s	薛 sjɛ	5	免 sjen	僧 səŋ		嘯 sjew
	5		5	想 sjaŋ		5
n	捏 觛 njɛ	粘 njɛm	年 njɛn	能 nəŋ		耨 nəw
1	列lje	廉 ljɛm	連 ljen	良 ljaŋ		
č	折 tse	詹 tsem	氈 tsɛn			勺tşew
	者 tşia	,				
čh	扯徹 tshe		禪 tṣʰɛn			超 tş ^h ɛw
-			淺 ts ^h ian			
k	格 kjaj	兼 kjɛm	堅 kjen			鉤 kəw
k ^h	客 k ^h jaj	njem	速 k ^h jɛn		克 kʰəj	
x	岛 xje		Jen Jen		2 J	
11	赫 xjaj					
j	也耶je					
J	L-III JC					

1.101						
	0		on	oŋ	oj	ow
_	斡 wɔ		完 on	汪 waŋ		
р	孛 pɔ					
m	抹秣沫 mo					
t	朵多 to		端 ton			
t ^h	脫 t ^h o		團 t ^h on			
s	莎 sɔ					
n	那 no					
1	羅 lo		欒 lən			
č				莊 tşwaŋ		卓 tşwaw
š					率 şwaj	搠 şwaw
k	戈kwo		官 kən			
k			關 kwan			
k^h	闊科 k ^h wo		寬 k ^h ən	匡 k ^h waŋ		
х	豁 xwɔ		桓 xən	晃 xwaŋ	槐 xwaj	
j	約jo					
	u		un	սդ	uj	
-	屼兀 u 五世士		溫穩 un 本	翁 uŋ	為委 uj	
pu	不步 ト pu ナ		奔 pun		北 puj	
mu	木mu		門 mun	蒙 muŋ 茎	梅 muj	
tu	都覩 tu 香土團 db		敦 tun 古 th	董 tuŋ 絞 than	對 tuj	
t ^h u	秃土圖 t ^h u 油		屯 t ^h un 瑶	統 t ^h uŋ	迷:	
S	速 su 訲		孫 sun	曲	遂 suj	
n	訥 nu 鱼 1		嫩 nun 論 lun	農 nuŋ 筎 1	電 1:	
1	魯 lu		im lun	籠 luŋ	雷 luj	
č č ^h				中 tşuŋ 厺 təhun	垂 tahu:	
č. š				充 tş ^h uŋ	垂 tşʰuj 水 şuj	
	古 ku		昆裩 kun	公 kuŋ	示 suj 癸 kuj	
k k ^h	口 Ku 窟 k ^h u		地址 Kull 坤 k ^h un	五 kuŋ 孔 k ^h uŋ	夹 kuj 恢 k ^h uj	
	雇 κ-u 忽呼 xu		坪 K-un 渾 xun	孔 K-uŋ 洪 xuŋ	灰 xuj	
х	恋叶 Xu		伴 Xull	iff Xulj	次 xuj	
	i	im	in	iŋ		iw
-	亦ji	音jim	因 jin	媵 jiŋ		拍 jiw
р	必 pi		賓 pin			
m	米mi		民敏 min			
t	的 ti		丁 tiŋ			
s	昔 si					
n	你泥 ni		紉您 nin			紐 niw
1	里驪黎 li	林 lim	鄰驎 lin	靈 liŋ		留 liw
č	直 tşi 只 tşr		真 tşin	征 tşiŋ		州 tşiw
čh	赤 tṣʰi	沉 tş ^h im	陳臣 tş ^h in	成 tş ^h iŋ		抽 tsʰiw
š	石si失sr		申 şin	升 şiŋ		守 şiw
k	吉 ki		斤 kin	京 kiŋ		
k ^h	乞 k ^h i	欽 k ^h im	勤 k ^h in	輕 k ^h iŋ		
х	希喜 xi		欣 xin	興 xiŋ		

TABLE 8.1 (cont.)

104

	ø	øn	у	yn	уŋ
_	月 ц ε		余 y	云 云 yn	
s	雪 sye	旋 syen		循 syn	松 sjuŋ
1	劣lye	,	騄 ly	侖 lyn	
č	拙 tşwɛ		主 tşy	諄 tşyn	
č ^h	啜 tşʰwɛ	川 tşʰwɛn	出 tşʰy		
š			暑 şy		
kh	缺 k ^h yɛ		曲 k ^h y		
х			許 xy		

TABLE 8.1 (cont.)

According to Pulleyblank, Northern Chinese of the Yuan period had lost the voiced stops and affricates which occurred in Middle Chinese, and like Modern Standard Chinese it distinguished only two series, voiceless unaspirated (p, t, k, t_{S}) and voiceless aspirated $(p^{h}, t^{h}, k^{h}, t_{S}^{h})$ (cf. also Hattori 1946: 79–113; 1973 and Finch 1986).

Mongolian syllable onsets contain some consonants that do not occur in Chinese. They are denoted by the addition of small superscript characters. The Chinese character $\pm zh\bar{o}ng$ 'middle' is added to characters with initial *x*, probably to indicate uvular pronunciation χ , e.g. $\pm \chi u$ (from $\Re xu$).

Pulleyblank's reconstruction of Yuan Chinese does distinguish between l and r, but characters with initial r are not used in Sino-Mongolian. Presumably the Yuan Chinese r, which developed from an earlier *n, was similar to its Modern Standard Chinese reflex, a retroflex fricative [z], and was felt to be different from Mongolian vibrant [r]. Instead, [r] was indicated with the character \pm *shé* 'tongue' superscribed to a syllable beginning with [1], for example $\pm ron$ (from # lon).

The consonants given in (4) occur in syllable codas.

(4)	р	th		$\mathbf{k}^{\mathbf{h}}$
		s		х
	m	n		ŋ
		1		
		r		
	W		j	

In Yuan Chinese, only the nasals *m*, *n*, *y*, and the glides *w*, *j* could form codas. The other Sino-Mongolian coda consonants are indicated by the addition of an extra character whose onset denotes the syllable coda. The codas *p*, *t^h*, *k^h*, *x*, *l* are written with subscript characters (卜 YCh *pu*, 惕 *t^{hi}*, 克 *k^həj*, 黑 *xəj*, 勒 *ləj*) and *s* with a full-size 思 *sz*. The coda *r* is written as 兒 *rr* or ^西兒. The coda *m* is sometimes indicated with a Chinese character with the coda *m*, but sometimes by the addition of the subscript character $\pm mu$. The coda *m* was probably disappearing (merging with *n*) in the form of Chinese which is the base for the Sino-Mongolian transcriptions. The

Hua–Yi yiyu sometimes uses a superscript character $\top d\bar{n}g$ to indicate final *l*, as in $\top \mathbb{A}$ SM *ul* (from \mathbb{A} YCh *un*).

8.2.2 Sino-Mongolian vowels

The relation between the Yuan Chinese and Sino-Mongolian vowels is less straightforward than for the consonants. The discrepancy between the Mongolian and Chinese vowel systems is rather large, and we have followed the tradition of transcribing Sino-Mongolian by converting the Chinese vowels to the seven Old Mongolian vowels *i*, *y*, *e*, ϕ , *u*, *o*, *a* (5).

(5)	5) Sino-Mongolian vowels									
	vowe	l	exa	mple		vowel		example		
	YCh	SM		YCh	SM	YCh	SM		YCh	SM
	a	a	荅	ta	ta	ja	e	良	ljaŋ	leŋ
	э	a	哈	хэ	xa	ia	e	者	tşia	če
		0	朵	tə	to	э	e	克	k ^h əj	k ^h ej
	w٥	0	豁	xwɔ	хо	i	i	你	ni	ni
	wa	0	槐	xwaj	xoj	ji	i	亦	ji	i
	ųε	ø	雪	sye	sø	ŗ	i	只	tşŗ	či
	wε	ø	啜	tşʰwε	č ^h ø	u	u	忽	xu	xu
	3	e	扯	tş ^h e	čhe	у	у	許	ху	xy
	je	e	別	pje	pe	ju	у	松	sjuŋ	syŋ
	jaj	e	額	jaj	e					

Combinations of a glide+vowel in Yuan Chinese are interpreted as single Sino-Mongolian vowels, but combinations vowel+glide are interpreted as Sino-Mongolian vowel+glide. For example, YCh $w\varepsilon$, combining the features round (from w), front, and open is interpreted as SM ϕ , but YCh εw is interpreted as SM ew (cf. Kuz/menkov 1987).

The YCh vowel \circ contrasts with *a* after most onsets, and contrasts with $w\circ$ (but not with *a*) after velar onsets and the zero onset. In the first case, YCh \circ is interpreted as SM *o* and YCh *a* as SM *a*; in the second case, YCh \circ is interpreted as SM *a*, and YCh $w\circ$ as SM *o* (see (5) and Table 8.1 for examples).

The choice of Sino-Mongolian y, ϕ vs. u, o is dictated more by Chinese phonotactics than by the actual Mongolian vowel. The front rounded vowels are less common than the back ones in Chinese, so the Old Mongolian vowels *y and * ϕ are often represented by Sino-Mongolian u and o; conversely, Old Mongolian *u and *o are represented by Sino-Mongolian y and ϕ in a few cases.

Undoubtedly it is a bit circular to use a transcription which treats the vowels in this way as a basis for reconstructing Old Mongolian, but all cited Sino-Mongolian forms are given with their Chinese characters in the Old Mongolian vocabulary in section 8.9, so that the reconstructed Yuan Chinese reading can be retrieved from Table 8.1.

Some researchers maintain that a vowel length distinction is indicated in some words in the *Secret history*. For example, the characters \Re and \mathscr{B} have the same segmental pronunciation t_2 in Pulleyblank's Yuan Chinese reconstruction, but \Re has rising tone and \mathscr{B} upper level tone, which might implicate a difference in duration, as in Modern Standard Chinese. According to Murayama (1970), the character \Re is normally used in those words which have a long vowel reflex in some modern languages, possibly inherited from Proto-Mongolic, and the character \mathscr{B} is seldom used in such words. For example, 'four' is written $\Re R \ beta$ *torpen* in the *Secret history*. Our Old Mongolian reconstruction is $t \ oppen$, but the long vowel in Monguor *teeren* indicates that Proto-Mongolic might have had a long vowel (see 8.5.1, and also Manlažav 1969; Kuz'menkov 1989; 1993*a*). The development of such words in modern languages is not uniform, however, and it is difficult to judge if, and to what extent, the *Secret history* indicates vowel length.

8.3 ARABIC MONGOLIAN²

There are several texts from the fourteenth and fifteenth centuries where Mongolian words are written with the Arabic alphabet. The most important one is a fourteenth century manuscript of the dictionary *Muqaddimat al-adab* 'Preamble to literature' by Abū'l-Qāsim Maḥmūd ibn-'Umar al-Zamaxšarī with Mongolian glosses written with the Arabic alphabet. Poppe (1938–9) has published the Mongolian and Turkish words in this manuscript and provided them with Russian translations. Other Arabic Mongolian texts are the Mongolian glossary in *Hilyat al-insān wa ḥalbat al-lisān* 'The ornament of man and the arena of language' by Jamāl al-Dīn ibn-Muhannā (beginning of fourteenth century; see Poppe 1938–9: 43–51), and the anonymous *Kitāb majmū ' tarjumān Turkī wa 'Ajamī wa Mughulī wa Fārsī* 'Collection for the Turkish–Persian–Mongolian–Persian interpreter' (the 'Leiden manuscript') of 1343 (Poppe 1927–8). We transliterate the Arabic alphabet as shown in (6).

(6) Arabic letters used in the Mongolian transcriptions

џ b	s س	۲ k	٥h
t ت	š ش	1 ل	
ΞĬ	ċ к	m م	(ڊ ص)
d د	p ق	n ن	(باط)
r ر			
∣ a	ĩ 'a	٥a	
u (w) و	u' او		
i(j) ى	i' ای		

Arabic has only three vowel qualities, [a], [i], [u]. In the Mongolian transcriptions of *Muqaddimat al-adab*, the vowels are usually written with letters indicating Arabic long vowels, i, o, j. We transliterate them simply as <a, i, u>, since they

² For notes on Arabic Mongolian, see App. H, 8.3, p. 223.

Short Arabic vowels indicated with sub- or superscript letters are written in a few cases in *Muqaddimat al-adab*, and more often in the Leiden manuscript. They are transliterated <ā, ĭ, ŭ>, for example, کیلَن *kilăn* 'tongue'. If no vowel is written in the Arabic script, no vowel is shown in the transliteration: خربان *wrban* 'three' (OM **kurpan*). When three vowel letters follow each other, the middle one is interpreted as a glide *w* or *j*: جاوون *jawun* 'hundred', *suppa* 'to tie'. Similarly, initial *s* followed by a vowel is transcribed *j*: *jabu* 'to go'.

Thus the Arabic Mongolian grapheme system is:



As seen in (7), voiced and voiceless stops are not differentiated, except d and t (corresponding to Old Mongolian *t and $*t^h$). In the uvular region, there are two consonants, the voiceless stop q and the voiced fricative \varkappa . They correspond to Old Mongolian uvular stops, aspirated $*[q^h]$ and unaspirated *[q], which in our reconstruction are regarded as allophones of the phonemes $*/k^h$ and */k in back-vocalic words. This is the only source for Old Mongolian which differentiates $*[q^h]$ and *[q].

The 'emphatic' (pharyngealized) consonant s is used for OM *s in a few words with back vowels, for example, أوصون 'usun 'water', where this sound apparently was felt to be closer to the Mongolian pronunciation than plain [s]. This suggests that vowel pharyngealization (10.1) had started already in the language of the Arabic Mongolian texts. Similarly, emphatic $t \perp i$ is used in a few words with back vowels, e.g. taqia 'hen' (also written with plain t: taqia).

8.4 'PHAGS-PA MONGOLIAN

Chinggis's grandson Qubilai, who founded the Mongolian–Chinese Yuan dynasty, introduced a new official script in 1269, to be used in all parts of his empire for Mongolian, Tibetan, and Chinese. This script, which was based on the Tibetan alphabet, was designed by the 'Phags-pa Lama, and is known as the 'Phags-pa script or the square script (*dörvölžin bičig*). It was used as an official script until the fall of the Yuan dynasty in 1368. Only about sixty 'Phags-pa texts are preserved, most of them imperial edicts or other official documents. The most convenient edition is Poppe (1957), which is the English version of an earlier work in Russian (Poppe 1941*b*). It contains high-quality photographs of the texts, transcriptions, English translations and comments, and a glossary, as well as a general introduction to the 'Phags-pa script. There are several later collections which contain more texts, however (see 8.9 for the sources we have used).

The 'Phags-pa script is written in columns which follow each other from left to right. The alphabet is given in (8).

Fnags-pa ieners usea jor mongonan									
Consonants Vowels									
뀌	k	ц	d	ш	j	^ ₀			
പ	k ^h	ଷ	n	т	r	^{ତା} u			
리	g	ව	р	ല	1	a i			
2	ŋ	ව	b	ടി	š	ч ₁			
Э	č	ୟ	m	κ١	s	∽ e			
田	čh	因	w	ন্দ	h	도 ø			
Ę	j	Э	Z	UN	Ø	তি y			
Б	t	Б	,	р	q				
Ħ	t ^h								

(8) 'Phags-pa letters used for Mongolian

The grapheme system shown in (9) can be inferred.



The 'Phags-pa script distinguishes three series of stops and affricates: voiced, voiceless unaspirated, and voiceless aspirated. In the Mongolian transcriptions, the voiced (b, d, g, j) and voiceless aspirated (t^h, k^h, \check{c}^h) series are normally used, corresponding to Old Mongolian voiceless unaspirated $(*p, *t, *k, *\check{c})$ and aspirated $(*t^h, *k^h, *\check{c}^h)$ consonants. There is only one letter in the uvular region, probably representing the voiceless stop q. Voiceless unaspirated k is written in a few words, for example, $\exists \nabla \exists \forall j k e$ 'big', but the consonant letters p, t, \check{c} are not used in indigenous Mongolian words. The letter z is found only in the word $\exists \exists zara$ 'moon', where it probably represents [s].

There are seven vowel letters. The letters for ϕ and y are compounds, *eo* and *eu*, respectively. Jagunasutu (1987c) regards the vowel letters as given in (8) as

the basic ones. In word-initial position, the letters o, u, i, r have a bar above them (K $\bowtie \Box \Box$), and initial ϕ and y are preceded by the zero initial letter Ξ , having the forms $\Xi \Box \prec$ and $\Xi \Box \Box$. When there is a following consonant in the same syllable, the form \checkmark of o is normally (but not always) used. Following Ligeti (1961), we transcribe it as \hat{o} (and the corresponding ϕ form $\Box \leftarrow (e\hat{o})$ as $\hat{\phi}$). There is some graphic variation, for example, the letter ∇r is sometimes written T.

The letter τ is usually transliterated $\langle \dot{e} \rangle$. Poppe (1957: 25) says that this vowel is more closed than $U \langle e \rangle$. Kögjiltü (1983; 1987*c*) and Jagunasutu (1988) make the more precise assumption that it was pronounced [1], and we transcribe it with this character. It stands in almost complementary distribution with *e*, usually occurring at the beginning of words, in contact with palatal and alveopalatal consonants, or as the second element of a diphthong. Hattori (1984; 1989) argues that $\tau \langle i \rangle$ and $U \langle e \rangle$ represent the same sound.

'Phags-pa Mongolian does not usually write the letter $I\overline{g} < y >$ in non-initial syllables, but we assume that $\overline{g} < u >$ denotes the sound [y] in non-initial position in front-vocalic words (cf. 8.6). We still transcribe it as < u >, however. A number of 'Phags-pa Mongolian words are written with < u > in some texts and with < y > in others. For example, 'to reach' can be written $P\overline{g}H k^h ur$ or $P\overline{I}\overline{g}H k^h yr$. Such words usually contain velar consonants, and for this reason we assume that the vowel was pronounced y (see 8.6.2), but we write it as u or y in the transliterations, depending on the original spelling.

As in the Tibetan alphabet, there is no letter for the vowel *a*, but an onset consonant without a vowel letter is interpreted as being followed by this vowel. There is usually a small space between the syllables. This is essential for deciding whether or not an *a* should be read. For example, IL FIG IS is different from IL FIGI. These words can be transliterated letter for letter as $< t^{h}$.bu.n> and $< t^{h}$.bun>, where < .> denotes the space between syllables. In the first word, $t^{h}abuna$ 'five-DAT', the space before *n* shows that it is the onset of a syllable *na*, and the absence of a space in the the second word, $t^{h}abun$ 'five', shows that *n* is a syllable coda.

We transliterate the letter \mathbb{P} as <' >. It may have been a glottal stop, or just a hiatus marker; Poppe (1957) and Jagunasutu (1989), both of whom transliterate it as <.>, regard it as a kind of zero initial. It is written in word-initial position in some words, and also between two vowels inside words, e.g. \mathbb{E} $\mathbb{P}_{\mathbb{F}} \mathbb{E} < \emptyset$.'u.l> a'ula 'mountain'. As this example also shows, the syllable a is written with the zero initial \mathbb{E} .

The *i*-diphthongs are written with $\langle ji \rangle$ in back-vocalic and with $\langle i \rangle$ in front-vocalic words, for example, $\square \exists \exists \square \exists \land nji.m.n \rangle$ 'eight- $\square \exists \top$; $\square \land \exists \land \exists \land d \land nji.m.n \rangle$

qôji > 'snake'; $\square (X < k^h e_I > 'wind')$. Since they are written together with the preceding vowel and consonant as one unit without syllable space, and with the combination form of o, we assume that they belong to the same syllable as these. For this reason we transcribe these words as *naimana*, $moq\hat{o}i$, $k^h e_I$. In some words there is a clear syllable boundary space, as in \mathbb{Z} $\mathbb{Z} \otimes \mathbb{Z} \otimes \mathbb{Z} \otimes \mathbb{Z}$, in which case we write *ji* in the transcription (*sajin*), although it can be doubted if this really denotes a different sound.

8.5 OLD MONGOLIAN VOWELS³

The reconstruction of the Old Mongolian vowel system is fairly uncontroversial, except for two points discussed below: the possible existence of a back high unrounded vowel, and the occurrence of 'primary' long vowels.

The sources indicate a seven-vowel system in word-initial syllables (10). Noninitial vowels are discussed in section 8.6.

(10)	Old Mongolian vowels in the initial syllable						
	i	у	1	u			
	e	ø		0			
			a				

These vowels are reconstructed from the correspondences shown in (11). Examples are given in Table 8.2.

(11)	Old N	Mong	golian ⁻	vowel	s in the	initi	al sylle	able
	OM	*a	*0	*u	*e	*ø	*y	*i
	UM	a	u	u	a, e	у	у	i
	SM	a	0	u	e	0	u	i
	AM	a	u	u	i	u	u	i
	Ph	a	0	u	е, 1	ø	у	i

Only the 'Phags-pa script distinguishes all vowels. It even writes an eighth vowel, *t*, which we regard as an allophone of *e* (or of *i* in diphthongs). Uigur Mongolian does not differentiate **u* and **o* (or **y* and * ϕ). Sino-Mongolian distinguishes **u*/*y* from **o*/ ϕ , but not, except in a few cases, **u* from **y* or **o* from * ϕ . Since Arabic has only three vowels (*i*, *u*, *a*), *u* must represent all four rounded vowels in Old Mongolian, and Arabic *i* corresponds to Old Mongolian **i* or **e* in the initial syllable.

All vowels except *i itself form diphthongs with a following *i. These diphthongs are represented in the sources as Vi, Vij, Vj, or Vi (Vladimircov 1929: 255, 266ff.; Poppe 1955: 76ff.; Jirannige 1985). There is also a diphthong *au. Diphthongs occur both in initial and in non-initial syllables. Examples are given in (12). Uigur Mongolian uji and yi are graphically identical, vww (or www), so the difference made between them in (12) is artificial.

³ For notes on Old Mongolian vowels, see App. H, 8.5, p. 224.

	ОМ	UM	SM	AM	Ph	
*a—a	*alt ^h an	aldan	alt ^h an	altan	alt ^h an	ʻgold'
*a—u	*japu	jabu	japu	jabu	jabu	ʻto go'
*a—i	*amin	amin	amin	amin	amin	ʻlife'
*u–a	*tumta	tumda	tunta	dundạ	dumda	'middle'
*u–u	*usun	usun	usun	'ușun	usun	'water'
*u–i	*urita	urida	urita	'urida	urida	'before'
*o–a	*k ^h ojar	qujar	χojar	qujar	qojar	'two'
*o–u	*t ^h osun	tusun	t ^h osun	tusun	t ^h osun	'fat'
*o–i	*čok ^h i	juqi	čewk ^h i	j̃uqi	joqi	'to fit'
*e–e	*nere	nar-a	nere	nirạ	nere	'name'
*e–y	*ehyten	ekudan	euten	'iudan	1'uden	'door'
*e–i	*setk ^h il	sadkil	set ^h k ^h il	sitkil	sedk ^h il	'mind'
*y–e	*yke	yka	uke	'ukạ	'yge	'word'
*y–y	*yk ^h y	yku	uk ^h u	'uku	yk ^h u	'to die'
*y–i	*pyrin	byrin	puri	buri	byrin	'each'
*ø–e	*tørpen	tyrban	torpen	durban	dørbe	'four'
*ø–y	*mølsyn	mylsun	molsun	mulsun		'ice'
*ø–i	*øk ^h in	ykin	ok ^h in	'ukin		'girl'
*i-a *i-u *i-o *i-e *i-y *i-j *i-i	*k ^h ičahar *nihu *siŋk ^h or *irken *nityn *hiryher *pič ^h ik *č ^h itk ^h ør	qičaqar niqu sinkqur irkan nidun irukar bičik	k ^h ičaar niu šiŋχor irken nitun xiruer pič ^h ik ^h č ^h it ^h k ^h or	qij'ar niu šŭqŭr 'irkan nidun hirar bijʻik čtkr	k ^h ijaar šiŋqôr 'irgen hiru'er bič ^h ig	'border' 'to hide' 'falcon' 'people' 'eye' 'blessing 'letter' 'devil'

 TABLE 8.2 Examples of initial and non-initial vowels

(12) Examples of Old Mongolian diphthongs

OM	UM	SM	AM	Ph	
*naiman	naiman	najman	naiman	naiman	'eight'
*sain	sajin	sain	sain	sajin	'good'
*kak ^h ai	qaqai	χαχαј	вaqai	qaqai	ʻpig'
*oira	ujir-a	oira	'uira		'near'
*k ^h oina	qujina	χoina	quinạ	qojina	'back'
*uila	ujila	ujila	'uila		'to cry'
*k ^h ei	kai	k ^h ej	kĭji	khei	'wind'
*nek ^h ei		nek ^h ej	nikai		'sheepskin'
*k ^h øisyn	kyisun	k ^h oisun	kuisun		'navel'
*k ^h øit ^h en	kyidan	k ^h oit ^h en	kuitan		'cold'
*yile	yila	ujile	'uilạ	yıle	'deed'
*t ^h aulai	taulai	t ^h awlaj	taulai	t ^h awlai	'hare'
*k ^h auli	qauli	χauli		qa'uli	'law'
	*naiman *sain *kak ^h ai *oira *k ^h oina *uila *k ^h ei *nek ^h ei *k ^h øisyn *k ^h øisyn *k ^h øit ^h en *yile *t ^h aulai	$\begin{array}{llllllllllllllllllllllllllllllllllll$	$\begin{array}{llllllllllllllllllllllllllllllllllll$	$\begin{array}{llllllllllllllllllllllllllllllllllll$	$\begin{array}{llllllllllllllllllllllllllllllllllll$

8.5.1 Primary long vowels⁴

None of the sources for Old Mongolian shows vowel length, except perhaps Sino-Mongolian; the 'Phags-pa long vowels are secondary, corresponding to two vowels in Uigur Mongolian and Sino-Mongolian. Consequently we do not reconstruct vowel length for Old Mongolian. There are some indications, however, that Proto-Mongolic might have had phonemic vowel length, or at least traces of it. Dagur, Shira Yugur, Monguor, and Bonan have long vowels in some words (indicated with italics in (13)), where Old Mongolian and the other Mongolic languages (here exemplified with Halh) all have short vowels.

(13)	OM	Dagur	Shira Yugur	Monguor	Bonan	Halh	
'five'	*t ^h apun	t ^h aawə	t ^h aawyn	t ^h aawun	t ^h awoŋ	t ^h aw	тав
'tree'	*motun	məət	muutyn	mooti	mot ^h oŋ	mət	мод
'bride'	*peri	pyr ^j	peery	peeri	weru	pir	бэр
'fat'	*thosun	thos	t ^h uusyn	t ^h oosi	t ^h osoŋ	thos	тос
'to guard'	*sak ^h i	sak ^j	saaky	ski	saaqa	sax ^j	сахь
'horse'	*morin	mər ^j	məəry	mori	moru	mər ^j	морь
'four'	*tørpen	turp ^w	tørwen	teeren	teraŋ	torəw	дөрөв
'fine'	*narin	nar ^j ən	naryn	narin	naaraŋ	nar ^j iŋ	нарийн
'year'	*hon	хээп	hən	xon	hoŋ	oŋ	он

The existence of such correspondences has been explained by assuming that Proto-Mongolic had 'primary' long vowels in these words, although the evidence for this is rather scanty, and the developments in the different modern languages are not uniform. In our opinion, the existence of vowel length in Proto-Mongolic can be doubted.

The theory that long vowels existed in Proto-Mongolic has been used for explaining why the intervocalic consonant, here reconstructed as *h, disappeared in all old and modern languages except Uigur Mongolian (see 8.7.2 and 10.3).

8.6 VOWELS IN NON-INITIAL SYLLABLES AND VOWEL HARMONY⁵

It is generally assumed that Old Mongolian (and Proto-Mongolic) had palatal (back ~ front) vowel harmony, and we will also make this assumption. There is, however, only incomplete support for this in the sources. None of the sources distinguishes non-initial back and front rounded vowels, and only Sino-Mongolian and 'Phagspa Mongolian distinguish non-initial a and e. Keeping strictly to the sources, only the vowels in (14) can be reconstructed in non-initial syllables.

⁴ For notes and sources on primary long vowels, see App. H, 8.5.1, p. 224.

⁵ For notes and sources, see App. H, 8.6, p. 224.

(14)Old Mongolian vowels in non-initial syllables

OM	*a	*e	*u/y	*0/ø	*i
UM	a	а	u	u	i
SM	а	e	u	0	i
AM	a, a	a, ạ	u	u	i
Ph	a	e	u	0	i

Non-initial a and e follow vowel harmony so that a is found only when the initial vowel is back (or *i), and *e only when the initial vowel is front. Although palatal vowel harmony thus is visible in the sources only for $*a \sim *e$, it is unlikely that it applied only to one of the three vowel pairs which contrasted in the back ~ front dimension in initial syllables. For this reason we assume that not only $a \sim e$, but also $*u \sim *y$ (and, in a few words, $*o \sim *\phi$), must have alternated in non-initial syllables. Thus we reconstruct *y and $*\phi$ rather than *u, *o in non-initial syllables, when the initial syllable has a front vowel.

Under these assumptions, only the combinations of initial and non-initial vowels shown in (15) occur. Examples are given in Table 8.2.

(15)	<i>Vowel sequencing in Old Mongolian</i> See section 8.6.1 for the sequences $o-o$ and $\phi-\phi$.									
	initial vowel	non	-initia	l vowel						
		*a	*u	*0	*e	*у	*ø	*i		
	*a	+	+					+		
	*u	+	+					+		
	*0	+	+	(+)				+		
	*e				+	+		+		
	*у				+	+		+		
	*ø				+	+	(+)	+		
	*i	+	+	+	+	+	+	+		

Palatal harmony applies to suffixes as well as to stems. This can be exemplified with the dative suffix -a/e in Sino-Mongolian (yurpan-a 'three-DAT' vs. irken-e 'people-DAT') and 'Phags-pa Mongolian (thabun-a 'five-DAT' vs. dorben-e 'four-DAT'). Old Mongolian and Halh vowel harmony are compared in section 10.6.2.

8.6.1 Rounding assimilation

The open rounded vowels o and ϕ are rare in non-initial syllables in the sources. They occur in a few words where the first vowel is i, and also in some words where the initial syllable has the same vowel ($o \text{ or } \phi$) as the second one. In the first case we reconstruct non-initial *o or *ø for Old Mongolian, but not in the second case. The reason for this is that when the vowel of the first syllable is reconstructed as *o or * ϕ , and the following vowel is open, there is often disagreement between the

sources whether the second vowel is rounded or not, as illustrated in (16). A general tendency is that Arabic Mongolian shows a rounded second vowel less often than the other sources. This may be a dialect feature connecting Arabic Mongolian with modern Oirad.

(1	6)	Rounding	assimilation	in the O	ld Me	ongolian sources
----	----	----------	--------------	----------	-------	------------------

OM	UM	SM	AM	Ph	
*k ^h ojar	qujar	χojar	qujar	qojar	'two'
*ora	ura	oro	'ura	oro	'to enter'
*nok ^h ai	nuqai	noxaj	nuqai	noqôi	'dog'
*moŋkal	munkqul	maŋxol	mnʁul	môŋqôl	'Mongol'
*tørpen	tyrban	torpen	durban	dôrbe	'four'
*nøk ^h er	nykur	nok ^h ar	nukr	nøk ^h ôr	'friend'

Progressive rounding assimilation of the second vowel (known as 'labial attraction') is thus a process that started already in Old Mongolian. Since it is difficult to sort out which words had rounded non-initial open vowels at the Old Mongolian stage, we will assume, and build into the reconstruction, that there was no rounding assimilation in Old Mongolian, so that the open rounded vowels *o and $*\phi$ occurred in non-initial syllables only if the vowel of the initial syllable was *i (cf. Vladimircov 1929: 315f.; Poppe 1951*b*; de Rachewiltz 1999).

Even if there is some rounding assimilation in the Old Mongolian sources, it does not affect suffixes. This can be exemplified with the roots **pos* 'to rise' and * $k^h \phi l$ 'foot' which take unrounded open suffix vowels in all sources (e.g. UM *bus-qa* [CAUS], AM *bus-ba* [PST], Ph *bôs-qa* [CAUS]; UM *kyl—ijan* [RFL], SM $k^h ol-ijan$ [RFL], AM *kul-an* [RFL]). Thus Old Mongolian did not have rounding harmony, unlike modern Mongolian proper, but like most other modern Mongolic languages.

8.6.2 Alternation of velar and uvular consonants

As mentioned above, the velar stops $*[k^h]$ and *[k] occur only in front-vocalic words, and the uvular stops $*[q^h]$ and *[q] only in back-vocalic words (which may contain the front vowel *i). Suffixes with velar or uvular stops alternate so that the velar variant occurs in front-vocalic and the uvular one in back-vocalic words. For example, the future participle suffix alternates between $*[-k^hy]$ in front-vocalic and $*[-q^hu]$ in back-vocalic words (17). The vowel alternation is not visible, but all sources show the alternation between velar and uvular consonants.

(17) Velar ~ uvular alternation in vowel harmony

OM	UM	SM	AM	Ph	
*[mete-k ^h y]					
*[japu–q ^h u]	jabu–qu	japu–χu	jabu–qu	jabu–qu	'to go'

8.6.3 Neutral *i

The phonetically front vowel *i, which has no back counterpart, can combine with any vowel in any position in the same word; see (15) and examples in Table 8.2. Old Mongolian *i is thus transparent in the sense that it is ignored by vowel harmony. An intervening *i is followed by vowels which harmonize with the vowel preceding the *i. This can be seen within root words as $*t^hak^hija$ 'hen' vs. $*e\xi^hike$ 'father' or *urita 'before' vs. *ynijen 'cow'. The Sino-Mongolian forms t^hak^hija , $e\xi^hike$, urita, unien distinguish a and e explicitly. This can also be exemplified with the $a \sim e$ alternation of suffixes added to words with i in non-initial position, seen explicitly in Sino-Mongolian forms as $morin-a\xi^ha$ 'horse-ABL' vs. ξ^herik ijer 'soldier-INST', and 'Phags-pa Mongolian morin-ijan 'horse-RFL' vs. byrin-e'each-DAT'.

Initial **i* can be followed by either back or front vowels. Again, this is visible in the sources only for Sino-Mongolian and 'Phags-pa Mongolian $*a \sim *e$. Inflected forms of such words show that the first vowel which is not an **i* determines the vowel harmony class of the word. Examples with the dative suffix *-e/a are Sino-Mongolian *irken-e* 'people-DAT' vs. $k^{h}ilpar-a$ 'easy-DAT' and 'Phags-pa Mongolian '*irgen-e* 'people-DAT' vs. *jirqo'an-a* 'six-DAT'.

When the first vowel of a word is *i* and the following one is *u* or o in the sources, it cannot be decided directly if the word is back- or front-vocalic. If the word contains a velar or uvular consonant, this shows the vowel harmony class. For example, since all sources show an uvular consonant in the word 'falcon' (UM *sinkqur*, SM *šiŋχor*, AM *šŭqŭr*, Ph *šiŋqôr*), it must be back-vocalic (OM **siŋk^hor*), but 'devil' (SM *č^hit^hk^hor*, AM *čtkr*), with velars, must be front-vocalic (OM **č^hitk^hor*).

If a word of this type does not contain a velar or uvular, its vowel harmony class can be determined if a suffix contains e/a or a velar/uvular. For example, the Sino-Mongolian forms *nitu*-*pen* 'eye-RFL' and $\check{c}^{h}isu$ -*pan* 'blood-RFL' show that 'eye' is front-vocalic (**nity*/*n*) and 'blood' is back-vocalic (* $\check{c}^{h}isu/n$).

Words which contain no other vowels than *i can be either front or back. There are rather few examples in the sources of such words with suffixes that reveal the vowel harmony class, but most of these words take front-vocalic/velar suffixes, for example, $*k^{h}i$ 'to do' (UM ki-kun [FUTP], SM $k^{h}i$ - $k^{h}un$ [FUTP], AM ki-ku [FUTP], Ph $k^{h}i$ - $k^{h}un$ [FUTP]) and $*pič^{h}i$ 'to write' (UM biči-kui [FUTP], AM bji-ksan [PSTP], Ph $bič^{h}i$ -bei [PST]). Exceptions are the words *nis 'to fly' and *nisi 'to beat', which take suffixes with an uvular consonant or the back vowel a (AM nis-qa-ba [CAUS-PST]; SM niši- $t^{h}u\chi aj$ [IMP], SM niši-xta- χun [PASSIVE–FUTP]). There is conflicting evidence for the word $*\check{c}^{h}ik^{h}in$ 'ear', which takes front-vowel suffixes in Sino-Mongolian ($\check{c}^{h}ik^{h}i$ -pen [RFL]), while its Arabic Mongolian form jiqin contains the uvular consonant q. Unlike the case in Old Mongolian, Halh words with i as the only vowel always take non-pharyngeal suffixes, e.g. $pi\check{c}^{h}-\xi e$ бичлээ 'to write–DPST'; $\check{c}^{h}ix$ -e чихээ 'ear–RFL'.

8.6.4 Proto-Mongolic *u?⁶

None of the sources indicates a back high unrounded vowel *u, and we do not reconstruct it for Old Mongolian. This vowel, often written \ddot{i} , is reconstructed in back-vocalic words by most Mongolists, however.

One reason for reconstructing this vowel is the asymmetry of the vowel system, and particularly of vowel harmony, where *i was completely neutral, occurring in both back-vocalic and front-vocalic words. If *i were originally *u in back-vocalic words, the Mongolic vowel system would be the same as the Turkic one, and vowel harmony would be more regular.

Another argument that has been given for the existence of Proto-Mongolic **uu* is the fact that the combination reconstructed here as */k^hi/ was written in two different ways in Uigur Mongolian, as $\Im_{\mathbf{T}}(\mathbf{m}_{\mathbf{T}}) < qi >$ in back-vocalic, and as $\Im_{\mathbf{T}} < ki >$ in front-vocalic words. Our interpretation is that the letters \Im (\mathbf{m}) and \Im represent different allophones, uvular *[q^h] and velar *[k^h], of a phoneme */k^h/. The uvular variant occurs in words with back vowels and the velar one in words with front vowels. The vowel **i* is neutral and can occur in both types of words. It can thus be preceded by either *[q^h] or *[k^h], depending on whether the other vowels of the word are back or front. Since all sources denote the **i* vowel in the same way in all types of words, we will assume that there was only one **i* vowel, which was truly neutral. Many researchers have, however, assumed that there was a vowel difference as well, so that Uigur Mongolian $\Im_{\mathbf{T}}(\mathbf{m}_{\mathbf{T}})$ and $\Im_{\mathbf{T}}$ were pronounced as [q^hul] and [k^hi], respectively.

In later stages of Written Mongolian, $\Im_{\nabla}(\mathbf{m}_{\nabla}) < qi > is replaced by <math>\Im_{\nabla} < ki > in$ back-vocalic words, and this is one of the main differences between Uigur Mongolian and Classical (or Modern) Written Mongolian. Our interpretation of this is that the uvular allophone $[q^h]$ was replaced by velar $[k^h]$ before *i* in back-vocalic words (but was retained in other positions).

The combination reconstructed as */ki/ was written in the same way as */k^hi/ in Uigur Mongolian, and the same reasoning can be applied, although there are rather few examples in the Old Mongolian material. In addition to Uigur Mongolian, Arabic Mongolian is fairly consistent in the use of uvulars before *i* in backvocalic words, while 'Phags-pa Mongolian vacillates. Sino-Mongolian writes only velars before *i* (18).

(18) Examples of */khi/ and */ki/

	OM	UM	SM	AM	Ph	
*[k ^h i]	*k ^h i	ki	k ^h i	ki	k ^h i	'to do'
	*setk ^h il	sadkil	set ^h k ^h il	sitkil	sedk ^h il	'mind'
*[ki]	*č ^h iŋkis	činkkiz	č ^h iŋkis		jiŋgis	'Chinggis'

⁶ For notes and sources, see App. H, 8.6.4, p. 224.

(18) (cont.)

	OM	UM	SM	AM	Ph	
*[q ^h i] *[qi]	*k ^h ičahar *čok ^h i *aŋkita *čalki	1 1	čewkhi	juqi	joqi	'border' 'to fit' 'separate' 'to swallow'

8.7 OLD MONGOLIAN CONSONANTS

Old Mongolian had the maximal syllable structure CVC. The consonants which occur as syllable onsets are shown in (19). Coda consonants are treated in section 8.7.4.

(19)	Old Mongolian onset consonants					
	voiceless aspirated stops/affricate voiceless unaspirated stops/affricate	D	t ^h t	č ^h č	k ^h k	
	voiceless fricatives	Р	s	C	к	h
	nasals	m	n			
	lateral		1			
	vibrant		r			
	glide			j		

The correspondences on which this reconstruction is based are given in (20), and examples are given in Table 8.3. The velar stops /k/ and $/k^h/$ have uvular allophones *[q] and *[q^h] in back-vocalic words; they are shown separately since they are treated separately in all sources. Similarly, initial and medial **h* are shown separately.

(20) Correspondences for Old Mongolian onset consonants The reflex š of *s occurs before *i.

OM	*t ^h	*č ^h	*[k ^h]	*[q ^h]	*p	*t	*č	*[k]	*[q]
UM	t, d	č	k	q	b	t, d	j, č	k	q
SM	t ^h	čh	$\mathbf{k}^{\mathbf{h}}$	χ	р	t	č	k	χ
AM	t	j	k	q	b	d	j	k	R
Ph	t ^h	čh	k ^h	q	b	d	Ĭ	g	q
OM	*s	*h-	*-h-	*m	*n	*1	*r	*j	
UM		~	-						
UNI	S	Ø	k, q	m	n	1	r	j	
SM	s s/š	Ø x	k, q Ø	m m	n n	1 1	r r	j j	
						_		j j j	

118

119

	OM	UM	SM	AM	Ph	
*t ^h	*t ^h aulai	taulai	t ^h awlaj	taulai	t ^h awlai	'hare'
	*alt ^h an	aldan	alt ^h an	altan	alt ^h an	'gold'
*č ^h	*č ^h erik	čarik	č ^h erik ^h	jirik	č ^h erig	'soldier'
	*napč ^h in	nabčin	napč ^h in	nabjin	nabč ^h in	'leaf'
$k^{h} [k^{h}]$	*k ^h ei	kai	k ^h ej	kĭji	k ^h eı	'wind'
	*nøk ^h er	nykur	nok ^h ar	nukr	nøk ^h ôr	'friend'
$k^{h}[q^{h}]$	*k ^h ura	qur-a	χura	qura	qura	'rain'
-	*kak ^h ai	qaqai	χαχαj	ваqаі	qaqai	ʻpig'
*p	*pič ^h i	biči	pič ^h i	biji	bič ^h i	'to write'
•	*japu	jabu	japu	jabu	jabu	'to go'
*t	*tørpen	tyrban	torpen	durban	dôrbe	'four'
	*ertem	erdam	ertem	'irdm	ırdem	'learning'
*č	*čok ^h i	juqi	čewk ^h i	juqi	joqi	'to fit'
	*k ^h ičahar	qičaqar	k ^h ičaar	qijar	k ^h ijaar	'border'
*k [k]	*ker	kar	ker	kir	ger	'house'
	*yke	yka	uke	'uka	yge	'word'
*k [q]	*kačar	qačar	χačar	ваjar	qajar	'ground'
- 1-	*čirkohan	jirquqan	čiryoan	jirsuan	jirqo'an	'six'
*s [s]	*sur	sur	sur	sur	sur	'to learn'
	*t ^h osun	tusun	t ^h osun	tusun	t ^h osun	'fat'
*s [š]	*sini	sini	šini	šini	šine	'new'
	*masi	masi	maši		maši	'very'
*h	*hon	un	xon	hun	hôn	'year'
	*tehere	takar-a	teere	dirạ	deere	'above'
	*ahula	aqula	aula	'ula	a'ula	'mountain'
*m	*morin	murin	morin	murin	morin	'horse'
	*amin	amin	amin	amin	amin	'life'
*n	*nere	nar-a	nere	nira	nere	'name'
	*k ^h oina	qujina	χoina	quina	qojina	'back'
*]	*t ^h aulai	taulai	t ^h awlaj	taulai	t ^h awlai	'hare'
-	*yile	yila	ujile	'uila	vile	'deed'
*r	*morin	murin	morin	murin	morin	'horse'
-	*nere	nar-a	nere	nira	nere	'name'
*j	*japu	jabu	japu	jabu	jabu	'to go'
J	*k ^h ojar	qujar	χojar	qujar	qojar	'two'
	ռ Ծյա	գսյա	Volu	գսյա	ԳԾյա	

TABLE 8.3 Examples of Old Mongolian onset consonants

8.7.1 Stops and affricates⁷

We reconstruct two series of Old Mongolian stops and affricates: voiceless aspirated and voiceless unaspirated. There are five places of articulation, labial (*p), dental $(*t^h, *t)$, alveopalatal $(*\check{c}^h, *\check{c})$, velar $(*k^h, *k)$, and uvular $(*q^h, *q)$. The

 $^7\,$ For notes and on stops and affricates, see App. H, 8.7.1, p. 224.

OLD MONGOLIAN

uvulars are allophones of the velars, occurring in back-vocalic words. In most earlier reconstructions, the two series of stops and affricates are written with symbols for voiceless and voiced consonants, corresponding to our voiceless aspirated and voiceless unaspirated, respectively. While some authors say that the Old Mongolian consonant series were phonetically voiceless and voiced, others use terms as fortis ~ lenis or strong ~ weak.⁸ Our reconstruction corresponds to that given by Poppe (1955) as shown in (21). Poppe's *g/y corresponds both to our *k and to our medial *h (see 8.7.2 below).

(21) Comparison of reconstructions

Our	Poppe	Our	Poppe
		*p	b
*t ^h	t	*t	d
*č ^h	č	*č	ž
*k ^h [k ^h]	k	*k [k]	g
*k ^h [q ^h]	q	*k [q]	Ŷ

The two series are distinguished clearly in Sino-Mongolian (aspirated ~ unaspirated) and 'Phags-pa (aspirated ~ voiced), except for the uvular stops. Arabic Mongolian differentiates the two series only for dental and uvular stops (voiceless ~ voiced), and Uigur Mongolian writes the two series in the same way except for the alveopalatal affricate when it is word-initial ($\check{c} \sim j$).

The reason for assuming that the Old Mongolian series were aspirated ~ unaspirated rather than voiceless ~ voiced is that aspiration is involved in Sino-Mongolian and 'Phags-pa Mongolian, and also that most modern languages have an aspirated ~ unaspirated contrast where voicing is only marginally involved. Another argument for this analysis is that the first series has a more limited distribution, not occurring in syllable codas (see 8.7.4), suggesting that it is more marked than the second one (cf. Janhunen 1990a: 38). It is highly unlikely that the voiceless member of a voiceless ~ voiced pair is the more marked one, but likely that the aspirated member of an aspirated ~ unaspirated pair is, and for this reason the voiceless ~ voiced interpretation is less probable. As will be seen in section 10.10, some historical developments suggest that the Old Mongolian aspirated stops were preaspirated, perhaps with the same distribution as in modern Halh, that is, postaspirated word-initially and preaspirated in other positions (see Svantesson 2003*a*).

All sources distinguish velar and uvular stops. They alternate with vowel harmony, so that the velars $*[k^h]$, *[k] occur only in words with front vowels, and the uvulars $*[q^h]$, *[q] only in words with back vowels (8.6.2). Since velars and uvulars thus stand in complementary distribution, we will regard each velar ~ uvular pair as the allophones of one phoneme, which we write with the symbol for the velar: $*/k^h/$, */k/ (cf. Poppe 1960*a*: 10; 1976; Hattori 1972; Buraev 1987*a*: 30). As seen in section 8.6.4, both velars and uvulars combine with a following *i, so there is a

120

potential contrast between velars and uvulars in this position. As far as we know, no minimal pairs exist, so this potential contrast is at most marginal.

The dental stops $*t^h$ and *t do not occur before *i in Old Mongolian. They had probably become affricates, $*c^h$ and *c, in this position already in Pre-Mongolic.

8.7.2 The fricative *h⁹

We reconstruct a fricative *h both word-initially and medially between vowels. Initial *h corresponds to zero in Uigur Mongolian, x in Sino-Mongolian, and h in 'Phags-pa and Arabic Mongolian, while medial *h corresponds to k or q (depending on vowel harmony) in Uigur Mongolian, and to zero in Sino-Mongolian, Arabic Mongolian, and 'Phags-pa Mongolian; see (20). Although initial and medial *hhave somewhat different reflexes, there are good reasons for analysing them as variants of the same phoneme, as pointed out by Janhunen (1999). Both are represented as zero in some of the Old Mongolian sources, and in the other sources they are represented as velar, uvular, or laryngeal consonants. Although the exact phonetic nature of this consonant cannot be determined, we write it as *h, since it is a weak consonant which disappeared in most modern languages; Janhunen writes it as *x.

Since this reconstruction is a controversial issue in Mongolic historical phonology, we will motivate it in some detail. The root of the problem is the fact that the sounds represented by the Uigur Mongolian letters $\pi < q >$ and Q < k > in intervocalic position may develop in three different ways in the modern languages, for example Halh. This is illustrated in (22). The words in (a) have the reflex x in Halh, those in (b) have g/g, and those in (c) have a zero reflex. The reflexes in the other Old Mongolian sources are also shown in (22).

(22) Intervocalic uvulars/velars in Uigur Mongolian

In addition to transliterated Uigur Mongolian, Classical Written Mongolian (WM) forms are given. The letter $\pi < q >$ occurs in back-vocalic words and $\Diamond < k >$ in front-vocalic. The forms with the letter π , traditionally transcribed y, occur in Classical Mongolian but usually not in Uigur Mongolian.

	UM	WM	Halh		SM	AM	Ph	
(a)	nuqai yku–	ᡝᠦᡣ᠇ᡳ ᠥᢙ᠊	nəxəi ux	нохой үх	noxaj uk ^h u	-	noqôi yk ^h u	'dog' 'to die'
(b)	muqai yka	ᠮᠦᡣ᠇ᡞ ᠥᠺᠡ	məgəi ug	могой үг		тиваі 'uka	moqôi 'yge	
(c)	aqula takar-a	ᡣᡍᠥᠯ᠊ᡟ ᡐ᠇ᠺᠺ᠋᠊᠋᠋᠋	uuk teer	уул дээр	aula teere	'ula dirạ	a'ula deere	'mountain' 'above'

⁹ See the survey of the literature in App. H, 8.7.2, p. 225.

As shown in (20), we reconstruct $*k^h$ (*[q^h, k^h]) for the correspondence in (22a), *k (*[q, k]) in (b), and *h in (c), that is, three different consonant phonemes. This provides a straightforward explanation for the different behaviours of the intervocalic uvulars/velars in words such as 'snake' and 'mountain'. In our reconstruction these words are *mokai (> Halh mɔgɔi) and *ahula (> Halh votb), and the different consonants motivate their different developments.

Many earlier reconstructions, including Poppe (1955), follow Classical Written Mongolian, where the traditional transcription is $\frac{*q}{k}$ in case (a), and $\frac{*v}{g}$ in both (b) and (c) (cf. (21)). This reconstruction is problematic since it does not differentiate cases (b) and (c). For example, the words 'mountain' and 'snake' are *ayula and *moyai in this reconstruction, and to explain the different developments it is necessarv to specify environments where intervocalic y/g are lost or are retained. Ramstedt (1902: 22) says that this may have to do with different accentuation; Ramstedt (1912; 1957: 88) and Vladimircov (1929: 216) make the more precise assumption that these consonants are lost when they preceded a vowel with secondary stress, and are retained in other positions. Both Ramstedt and Vladimircov assume that the consonants were different as well, a fricative in the first case and a stop in the second. According to this assumption, words like *ayula had a secondary stress on the second vowel, but *moyai did not. This assumption is a bit circular, however, since there is no other evidence for secondary stress than the different developments of the consonants. Several researchers have connected this assumption with the hypothesis that Proto-Mongolic had primary long vowels (8.5.1), and proposed that the conditioning factor was not stress but phonological length of the following vowel, so that these intervocalic stops were lost in 'weak position', that is, before a primary long vowel, but were retained in 'strong position', that is, before a short vowel (Hattori 1959; Poppe 1959; 1960a: 40f.; 1961; 1962; 1967; Ligeti 1964) (see also Murayama 1970). Poppe (1960a) relates the two hypotheses by saying that the primary long vowels developed from vowels which carried a tone accent (musikalischer Ton).

Although there is little direct evidence in the Mongolic data for the hypothesis that long vowels caused the loss of a preceding uvular/velar, it gets some support from comparison with Junggar Tuvan (Kök Mungchaq), a Turkic language spoken in Xinjiang (Coyijongjab 1982b; 1985; Möngkebuyan 2000a). According to Coyijongjab, this language has preserved archaic phonological features in numerous loans from Mongolic. Some words which correspond to Old Mongolian words with a *VhV group have the group VhVV in Junggar Tuvan, for example baruhuuun 'right', dʒaluhuuu 'young', and xaħaalga 'gate' (OM *parahun, *čalahu, *kʰahalka). Rather few words with this type of correspondence have been found, however; furthermore, other researchers analyse the Junggar Tuvan data differently: Mawkanuli (1999) gives the cited words as baruuun, žalunu, xalga.

As mentioned above, Ramstedt and Vladimircov reconstructed different Old Mongolian consonants for the lost and retained Classical Mongolian uvulars/ velars, but regarded them as allophones conditioned by the accentuation or length of the following vowel. Poppe (1960*a*: 41, 46) says that Proto-Mongolic had fricatives $*_{y}$ and $*_{w}$ in 'weak position', that is, before a long vowel, and stops in 'strong

position', before a short vowel. Hattori (1939) reconstructs a fricative $*\nu$ for our *h, and stops *G/g for our *k; Street (1957b) makes a similar proposal.

In our opinion, the only possibility for Old Mongolian as we define it, is to regard the disappearing and retained 'weak' uvulars/velars as reflexes of different consonants since they have different reflexes in the Old Mongolian sources, except Uigur Mongolian. The alternative with the same consonant but different conditioning environments must be rejected since there is no independent internal evidence for the existence of long (or stressed) vowels in the required positions.

One problem with reconstructing the disappearing uvular/velar as a separate Old Mongolian consonant is its limited distribution, appearing only in intervocalic position. Identifying it with initial *h eliminates this problem at least partially, since *h then occurs in any syllable onset position, except immediately after another consonant.

There are some direct indications that the consonant we reconstruct as Old Mongolian intervocalic *h is the reflex of an earlier (Pre- or Proto-Mongolic) uvular/velar or a labial, or perhaps of both. One such indication is a number of words which have reflexes of *h in some modern languages but reflexes of *k in others. These include OM *suhu 'armpit', with reflexes of *h in Chahar (svv), Baarin (svv), and Kalmuck (sy), but reflexes of *k in Halh (sva cyra) and Buriad (hvga); similarly, OM * $k^{h}i\check{c}ahar$ 'border' has reflexes of *h in Buriad ($x^{j}izar$) and Kalmuck ($k^{h}izer$), but the Halh form ($x^{j}acgar x\pi\pi$ ap) has the reflex of *k. Another indication of this relation is that intervocalic *h appears as g in some Tungusic loanwords (Poppe 1972a; Doerfer 1985b).

Some words with OM *h show reflexes of *p as well, for example, OM *tehel 'gown', which is dewl in Kalmuck (<*p), degəl in Buriad (<*k), and teeţ дээл in Halh (<*h). This word is written with Θ in Classical Mongolian but with \Diamond <k> in Uigur Mongolian. The word 'self' is written with in Uigur Mongolian (ybar), but has zero reflexes in Sino-Mongolian (oer), Arabic Mongolian ('ur), and 'Phags-pa Mongolian (ϕ 'er), so that Uigur Mongolian suggests the Old Mongolian form * ϕ per, and the other sources suggest * ϕ her. Facts like these, and also comparison with languages outside Mongolic, have led to the assumption that at least some occurrences of intervocalic *h have developed from an earlier unaspirated *p (usually written *b), or that one dialect of Old Mongolian retained *p while it developed to *k (*y/g in Poppe's notation) in others (Poppe 1960a: 41ff.; Thomsen 1987; Ozawa 1997; Janhunen 1999). Bulag (1983; 1985; 1993a; 1996) reconstructs Proto-Mongolic (or Pre-Mongolic) *p for our intervocalic *h, and *q/k for our *k.

Initial *h disappeared in Mongolian, Buriad, Kamnigan, Oirad, and Moghol, but is retained as h, x or \check{s} in the other Mongolic languages (see 10.8.3). Ramstedt (1916–20; 1957: 39ff.) and Pelliot (1925; 1944) proposed the hypothesis that this consonant was * p^h (usually written *p) in Proto-Altaic, and then developed to zero in Modern Mongolian via intermediate stages with *f or *h. This is known as Ramstedt–Pelliot's law. Some Mongolists reconstruct * p^h for Proto-Mongolic.¹⁰ This hypothesis is based on comparison with other Altaic languages, and on evidence
from Mongolic loans in Tungusic and Turkic languages, but has little support from the Old Mongolian sources or the modern Mongolic languages. In our opinion, $*p^h$ cannot be reconstructed at the Old Mongolian (or even Proto-Mongolic) stage.

In conclusion, reconstructing the disappearing uvular/velar stop in Uigur Mongolian as a separate Old Mongolian phoneme is the only reasonable solution, given the Old Mongolian sources. Furthermore, identifying this consonant with initial h solves a number of problems in Mongolic historical phonology.

8.7.3 Sonorants and *s

The Old Mongolian sonorants *m, *n, *l, *r, *j agree between all sources, and have not changed much even in the modern languages. As is still basically the case in modern Mongolian, the liquids *l and *r cannot occur word-initially (Vladimircov 1929: 366; Poppe 1955: 155, 160; Darbeeva 1996: 88ff.).

8.7.4 Coda consonants

The coda position is more restricted than the onset position, the main difference being that the affricates and the aspirated stops do not occur as codas (A. A. Bobrovnikov 1849: 12ff.; Vladimircov 1929: 408; Poppe 1960*a*: 40; Cydenžapov 1973). The velar nasal * η is the only consonant that does not occur in syllable onsets but only in codas. The coda consonants are given in (23).

(23) Old Mongolian coda consonants

voiceless stops	р	t	k
voiceless fricative		S	
nasals	m	n	ŋ
lateral		1	
vibrant		r	

The correspondences on which this reconstruction is based are given in (24), and examples in (25).

(24)	Old Mongolian	1 coda consonants	ſ
------	---------------	-------------------	---

OM	*p	*t	*[k]	*[q]	*s	*m	*n	*ŋ	*1	*r
UM	b	t	k	q	s	m	n	nk	1	r
SM	р	t ^h	$\mathbf{k}^{\mathbf{h}}$	Х	s	m	n	ŋ	1	r
AM	b	d, t	k	q	s	m	n	n(q)	1	r
Ph	b	d	g	q	S	m	n	ŋ	1	r

	OM	UM	SM	AM	Ph	
*p	*ap	ab	ap	ab	ab	'to take'
	*napč ^h in	nabčin	napč ^h in	nabjin	nabč ^h in	'leaf'
*t	*morit	murit	molit ^h	murid	morid	'horses'
	*setk ^h il	sadkil	set ^h k ^h il	sitkil	sedk ^h il	'mind'
*k [k]	*č ^h erik	čarik	č ^h erik ^h	jirik	č ^h erig	'soldier'
	*pič ^h ik	bičik	pič ^h ik ^h	bijik	bič ^h ig	'letter'
*k [q]	*č ^h ak	čaq	čhax	jaq	č ^h aq	'time'
	*pulak	bulaq	pulaχ-	bulaq	bulaq	'spring'
*s	*urus	urus	urus	urus	urus	'to flow'
	*pos	bus	pos	bus	bôs	'to rise'
*m	*ertem	erdam	ertem	'irdm	ırdem	'learning'
	*k ^h amt ^h u	qamtu	χamt ^h u	qamtu	qamt ^h u	'together'
*n	*hon	un	xon	hun	hôn	'year'
	*yntyr	yndur	untur	'undur	yndur	'high'
*ŋ	*tyhyreŋ	tykurank	tuureŋ	duran		'full'
	*miŋk ^h an	minkqan	minxan	mnqan	miŋqan	'thousand'
	*øŋke	ynkka	oŋke	'unkạ	<i>ô</i> ŋge	'colour'
*1	*moŋkal	munkqul	maŋχol	mnsul	môŋqôl	'Mongol'
	*alt ^h an	aldan	alt ^h an	altan	alt ^h an	ʻgold'
*r	*k ^h ojar	qujar	χojar	qujar	qojar	'two'
	*urt ^h u	urtu	urt ^h u	'urtu	urt ^h u	'long'

(25) Examples of coda consonants

There were no complex codas in Old Mongolian, except in some loans such as **pars* 'tiger' (from Persian).

The nasals *n and $*\eta$ cannot precede non-homorganic stops within a word (cf. Ramstedt 1913), but *m can precede dentals (as in *tumta 'middle') or velars (*nimken 'thin'). In word-final position, all three nasals occur.

Many Old Mongolian nouns and numerals ended in 'unstable' *n, which appeared in some forms but not in others (see 10.9.1).

8.8 MORPHOPHONOLOGICAL PROCESSES

Except for vowel harmony, there are rather few morphophonological processes in Old Mongolian. There are usually no changes in the stems when suffixes are added. One exception is that word-final k becomes weakened to h before the plural suffix -Ut (-ut/yt) (but cf. Tömörtogoo 1990). Examples are $\check{c}herik$ 'soldier', plural $\check{c}herih-yt$ and $\check{a}imak$ 'district', PL $\check{a}imah-ut$. Weakening does not take place before case suffixes, however. For example, the genitives of these words are $\check{c}herik-yn$ and $\check{a}imak-un$. This suggests that plural and case suffixes have different status in the morphology. It could be mentioned that the plural suffix is usually written joint to the stem in Uigur Mongolian, but case suffixes are written separately.

Another indication that case suffixes are clitic-like and not fully integrated in the words is the fact that unlike stem-internal *s before *i, stem-final *s is not pronounced *[š] when a suffix beginning with *i is added. Examples are OM *ulus-i 'state-ACC' (SM *ulusi*, AM '*ulusi*), and *yile-s-i 'deed-PL-ACC' (SM *ujilesi*, Ph *yılesi*).

It is more common that suffixes vary with the stems. In particular, many suffixes have different forms depending on whether the stem ends in a vowel or a consonant. These variations are rather complicated and differ between the sources. For a detailed account of this, we refer to Weiers (1969), and for Classical and Modern Written Mongolian to Poppe (1954a).

8.9 OLD MONGOLIAN VOCABULARY

All Old Mongolian words used as examples are given in this vocabulary together with the sources on which the reconstructions are based.

The stems of words are given (for verbs, the stem is identical to the imperative). If the stem does not form a graphic unit in a source, an inflected form is given. For each word, normally only one form from one source of each kind is given. The variation which sometimes occurs is usually irrelevant for our purposes and is not accounted for (see especially Weiers 1969 for this). Unstable *n (10.9.1) is shown as /n.

The different Old Mongolian sources are discussed in section 8.1–4. The abbreviations used in the vocabulary are given below. When known, the year of the source is given.

Uigur Mongolian

organ mongo		
Texts from Do	bo (1983b)):
D1	1227	Chinggis Khan's stone
D2	1240	Edict of Ögödei
D3	1246	Edict of Güyüg
D4	1257	Stone inscription to Möngke Khan
D5	1267/89	Abaga's letter
D6	1289	Argun's letter to Philippe le Bel
D7	1290	Argun's letter to Pope Nicholas IV
D8	1302	Qasan's letter to Pope Bonifacius VIII
D9	1305	Öljeitü's letter to Philippe le Bel
D10	1307	Translation of the Book of filial piety (Xiàojīng 孝經)
D11	1312	Commentary to Bodhicaryāvatāra
D12	1335	Inscription in memory of Zhāng Yìngruì 張應瑞
D13	1338	Inscription in memory of Jigüntei
D14	1340	Inscription of Arug, Prince of Yunnan
D15	1346	Xīngyuángé 興元閣 inscription
D17	1362	Inscription in memory of prince Hindu

D19	14C	Fragments of the Alexander romance
D20	1453	Edict of 1453
D21	1580	Altan Khan's letter
D22	15C	Ming dynasty edicts
Turfan texts (Cèrènsodno	m and Taube 1993):
T24		Bhadracaryā-praṇidhāna-rāja
T25	early 14C	Mañjuśrīnāmasaṃgīti
T29, 30	early 14C	<i>Mahākāli</i> hymn
T34		Poem
Т36		Text on the consequences of sin
Т50, 51, 61	13/14C	Calendar fragments
T68	1369	Tax exemption document
Т70		Tax document
Т80, 82		Text fragments
Sino-Mongoli	an	
Ν	1228	Secret history of the Mongols (Kuribayashi and
		Coyijongjab 2001)
HY	1389	Hua–Yi yiyu (Kuribayashi 2003)
Arabic Mongo	olian	
М	14 C	Muqaddimat al-adab (Poppe 1938–9).
IM	14C	ibn-Muhannā (Poppe 1938–9)
L	1343	Leiden Manuscript (Poppe 1927–8)
'Phags-pa Mo		1 1 11
	-	(P) or Žančiv (2002) (Z):
P1 (Z1)	1276	Edict of Mangala
P2 (Z17)	1314	Edict of Buyantu Khan
P4 (Z23)	1321	Edict of Empress Dowager Dagi
P7 (Z40)		Minusinsk páizi
P11 (Z51(2))		Subhāșitaratnanidhi fragment
P12 (Z38)	1345	Jūyōngguān 居庸關 inscription (east)
P13 (Z39)	1345	Jūyongguan inscription (west)
Z2	1277/89	Edict of Qubilai Khan
Z4	1277/89	Edict of Qubilai Khan
Z5	1280/92	Edict of Qubilai Khan
Z8	1298	Edict of Öljeitü Khan
Z10	1303	Edict of Prince Seüse
Z11	1305	Edict of Prince Qaisan
Z12	1306	Edict of Öljeitü Khan
Z14	1313	Edict of Buyantu Khan
Z15	1314	Edict of Buyantu Khan
Z18	1314	Edict of Buyantu Khan
Z21	1318	Edict of Buyantu Khan

128		OLD MONGOLIAN
Z22	1320	Edict of Empress Dowager Dagi
Z25	1323	Edict of Gegen Khan
Z26	1324	Edict of Yisüntemür
Z28	1328	Edict of Yisüntemür
Z29	1335	Edict of Togontemür
Z32	1343	Edict of Togontemür
Z34	1351	Edict of Togontemür
Z36	1368	Edict of Togontemür
Z51		Subhāșitaratnanidhi fragments
Z53		Printed text fragment
Z59		Name inscription

**ač^hi(h)an* 'burden'. SM 阿赤阿納 *ač^hian-a* [DAT] N274/12:27; AM (جالن ajaan M94b.

*ahula 'mountain'. UM 'mot aqula D15:25; SM 阿屼刺 aula N118/3:30; AM りょ 'ula M372a; Ph 5 ピッ マ a'ula P2:20.

- **aimak* 'district'. UM 'ᠠᡳᠮ᠇᠇f *aimaq* T70r15; SM 阿亦馬_黑 *aimax* N156/5:25, 阿亦馬 ^中渾 *aimay*-un[GEN] N156/5:25; Ph B37 尼 *aimaq* P4:4.
- * ak^ha 'elder brother'. UM 'mm' aqa D9:27; SM 阿中合 $a\chi a$ N18/1:11; AM ^[a] aqa M303b; Ph 5 E aqa Z10:11.
- *alak 'motley'. UM 'ᠬᠯᠠᡍᠴᠢᡟ alaġ-čin [DER] D12:50; SM 阿刺黑 alax N100/2:44; AM ᡟ ala M97a.
- **ali* 'which'. UM ᠬᠯᡗ *ali* D6:24; SM 阿里 *ali* N205/8:37; AM 北*ali* M98b; Ph ర 已 *ali* Z5:27.
- *alima 'fruit'. SM 阿里麻 alima HY1:4a; AM ばん alma IM432a; Ph 5 23 8 alima P12:7.
- *alt^ha/n 'gold'. UM ܐܕܕܕܐ aldan D17:4; SM 阿_勒壇 alt^han N117/3:27; AM النتان altan M385a; Ph 岛回 መalt^han P12:5.
- *ama/n 'mouth'. UM مترامان aman D17:27; SM 阿蠻 aman N87/2:25; AM المان aman M99b.
- **amiln* 'life'. UM ᠠᠮᡳ/ *amin* D12:21; SM 阿民 *amin* N55/1:35; AM امين amin M100b; Ph ട താത *amin* P13:12.
- *anykita 'separate'. UM ᡝᠠᠭᢉ᠇᠇ᠶᡨᡟ ankqida D10:27a5; SM 昂吉荅 anykita N92/2:33; AM انگدا ankda L1258; Ph ടവ வருide P2:20.
- **ap* 'to take'. UM شوبه *ab-ču* [IPFG] D12:21; SM 阿_ト *ap* N13/1:8; AM بوبه *ab-uba* [PST] M94a; Ph 5⊓ *ab* P2:21.
- *arihun'clean'. UM ܐܕܝܕܝܐ ariqun D13:21; SM 阿里溫 aliun HY1:28a; AM اريون ariun M105b.
- *arka 'method'. UM ٦ إ arq-a D11:162b8; SM 阿^舌兒^中合 arχa N208/8:45; AM ر غه arʁa M105a.
- *čahu 'to bite'. UM *faqu-qsan* [PSTP] T29:6; SM 札兀 čau N140/4:27; AM خاو *jaqu-qsan* [PSTP] T29:6; SM 札兀 čau N140/4:27; AM

- *čahu/n 'hundred'. UM לעסוד jaqun D11:156b11; SM 札溫 čaun N185/6:52; AM جاوون jawun M203b; Ph Ш ലன ja'un Z36:42.
- *čalahu 'young'. UM ᠵᠯ᠇ᠬᠣ jalaqu D13:20; SM 札刺為 čalauj N156/5:25; AM بلو jlu M200b.
- *čalki 'to swallow'. SM 札動吉 čalki N78/2:11; AM جالغيبة jalui-ba [PST] M200b.
- *čiha 'to point'. UM
- *čikasu/n[°]fish'. UM (jiqasun D22:35b2; SM 只^中合速 čixasu N75/2:7; AM زنينيون jirasun M205b.
- *čil 'year'. UM ܐ jil D2:3: SM 昗_{ij} čil N13/1:8; AM ↔ jil L1274; Ph שזם jil P2:24.
- *čirkoha/n 'six'. UM (jirquqan D6:33; SM 只^舌兒^中豁安 čirxoan N133/4:13; AM جير غوان *jirguqan* M207a; Ph யउम ലく 전의 *jirqo 'an* Z36:17, யउम 巴く 전 Ø *jirqo 'an* a [DAT] P1:21.
- *čiryke/n 'heart'. UM (\mathfrak{f}^{π} D17:18; SM \mathfrak{R}^{π} Bh čiruke N125/3:48; AM جوروكان jirukan M211b. [SM indicates *k, but most modern languages indicate $*k^{h}$]
- *čita 'spear'. UM , jida–z [PL] D19:13a10; SM 只荅 čita N72/2:3; AM 兰沃 jida M205a.
- *čok^hi 'to fit'. UM (توابير juqi–qu [FUTP] D6:23; SM 勺乞 čewk^hi N155/5:22; AM خو قيبه juqi–bą [Pst] M208a; Ph ሠ≺ లర joqi P13:10.
- *čø(h)elen 'soft'. SM 勺斡連 čewolen N189/7:10; AM جوولان juulan M209a.
- *č[ϕ/o]p 'right'. UM (\overline{vo}) jub D7:15; SM $\neg_{arsigma}$ čewp N20/1:12, 拙別察 č $\phi p e\check{c}^{h}a$ [ABL] N254/11:21; AM $\downarrow \downarrow \check{j}\check{u}b$ IM439b. [UM indicates *o, and the e in the SM ablative form indicates * ϕ . The modern languages indicate * ϕ .]
- *č^ha(h)alsu/n 'paper'. SM 察阿_勒孫 č^haalsun N203/8:31; AM جالسون jalsun M131b.
- *č^hak 'time'. UM (خطق čaq D11:157b14; SM 察_黑 č^hax N31/1:19; AM جاق *jaq* M130b; Ph ዙ 묘 고 č^haq—un [GEN] P13:9.
- *č^hakahan 'white'. UM خمت čaäqaqan D11:167b7; SM 察^中合安 č^haxaan N203/8:31; AM جاغان *jasan* M130a; Ph ዙ 민민장 č^haqaan Z15:15.
- *č^hasu/n 'snow'. SM 察孫 č^hasun N31/1:19; AM جاسون jasun M131a.
- *č^herik 'soldier'. UM *carik* D17:20; SM 扯^舌里_克 č^herik^h N100/2:44, 扯^舌里昆 č^herik–un [GEN] N194/7:29, 扯^舌里吉耶^舌兒 č^herik–ijer [INST] N170/6:5; AM エスシン jirik M132a, ヴィンシン jirik–un [GEN] M132a; Ph 旧 H3后 č^herig P2:4, 旧 H3 后 ある č^herig–un [GEN] Z28:2.
- *č^herihyt 'soldiers'. UM ファᡵᡪᠿ v čarikut D6:9; SM 拙^舌里兀傷 č^heriut^h N101/2:46; Ph ℍՄ H3 凹จ มจิช č^heri'ud–un [GEN] P2:4.
- *č^hik^hi/n 'ear'. UM *六 čikin* D15:30; SM 赤勤 *č^hik^hin* N138/4:24, 赤乞邊 *č^hik^hi–* pen [RFL] N254/11:25; AM (جيقين *jiqin* M134b.
- *č^hila(h)u/n 'stone'. SM 赤老溫 č^hilawun N72/2:3; Ph 昭 己 巴 の č^hila'un P13:5.

*č^hina 'to boil'. SM 赤納 č^hina N19/1:11; AM + jina M133b.

- *č^hin[o/a] 'wolf'. UM パップ čina T82v5; SM 赤那 č^hino N78/2:11; AM + jina M134a. [The modern languages indicate *o]
- *č(^h)iŋkis 'Chinggis'. UM ママベマイ činkkiz D1:1; SM 成吉思 č^hiŋkis N1/1:1; Ph யてい Fot jiŋgis P2:8. [UM and SM indicate *č^h, as does Halh; Ph indicates *č]
- *č^hisu/n 'blood'. UM パポロリャイ čisu-dan [DER] T30:12; SM 赤孫 č^hisun N145/4:38; 赤 速班 č^hisu-pan [RFL] N254/11:25; AM *jisun* M135a.
- *čhitkhør 'devil'. SM 赤惕科兒 čhithkhor HY1:16a; AM جنكر čtkr IM435a.
- *eč^hike 'father'. UM (محترك ačika D7:8; SM 額赤格 eč^hike N73/2:4; AM (أيجيكه ijika M151b.
- *ehyte/n 'door'. UM ܐܝܝܕܝܪ ekudan D22:6a5; SM 額兀顛 euten N208/8:43; AM ايودان iudan M167b; Ph T كالله تا i'uden Z53:5.
- *emehel 'saddle'. UM ܐܪ̈́ܐ emakal D21:12; SM 額篾額_勒 emeel N80/2:15; AM ايمل iml M139b.
- *emys 'to wear'. UM ᡝ**ᠳᠣᠲᠣ** emus-basu [COND] D10:36a1; SM 額木思 emus N106/3:7; AM آموسبه 'amus-ba [PST] M233b.
- *ene 'this'N62/1:43. UM ܐܝܝܝ ana D2:1; SM 額捏 ene N62/1:43; AM ابنه 'ina M155b; Ph Г נו Ine Z5:29.
- *epesy/n 'grass'. UM ᡝᢒᠰᠣᠠ ebasun D13:4; SM 額別孫 epesun N24/1:15; AM زايداسون ibasun M150b.
- *ertem 'learning'. UM 'توجيع' erdam D13:5; SM 額^舌兒點 ertem N189/7:10; AM اير دم irdm M161a; Ph КН الله rrdem P13:10.
- **ert^he* 'early'. UM 'ايرته erda D10:9b1; SM 額兒帖 ert^he N64/1:44; AM 'ايرته' irtạ M164b.
- *harpa/n 'ten'. UM ᠠᢏᠪᠡ arban D5:14; SM 哈兒班 xarpan N53/1:33; AM هربان hrban M181b; Ph い のは harban Z12:13.
- *hič^he 'to be ashamed'. SM 希扯 xič^he N244/10:31; AM هيجيبه hiji-ba [PST] M184b; Ph Fo hič^hu Z51(3)a:10.
- **hiruhar* 'bottom'. UM ᡝᡳᠷᠳ᠇ᠷ *iruqar* D19:9b1; SM 喜^舌魯阿^舌刺 *xiruar–a* [DAT] N199/8:10; AM ஆரும் *hiruar* M185b.
- *hiryher 'blessing'. UM ᡝᡳᠣᠺᡘ irukar D14:14; SM 希^舌魯額^舌兒 xiruer N201/8:20; AM ふんし hirar M185b; Ph ぽ ዘ岡 凹叶 hiru'er P2:13.
- *hoi 'forest'. UM ᡝ ui D17:50; SM 槐 xoj N12/1:8.
- *hon 'year'. UM ܐ un D12:15; SM 桓 xon N26/1:16; AM هون hun M185b; Ph S
حة hôn Z34:28.
- *hu(h)ut^ha 'bag'. SM 呼呼塔 xuxut^ha HY1:10b; AM أوته 'uta M372b.
- *huja 'to tie'. UM ᡝᠣᠢ᠇ᠵᠣ uja-ču [IPFG] D19:9a7; SM 忽牙 xuja N195/7:33; AM 純泉 huja M188a.
- *hulahan 'red'. UM ܐܝܝܝܝܝܝ ulaqan D20:20; SM 忽刺安 xulaan N81/2:17; AM هولان hulan M187a.
- *hyk^her 'ox'. UM ᡝᠢᠭᠷ ykar D6:31; SM 忽客兒 xuk^her N100/2:44; AM しく 'ukr M377b; Ph SUD EUH hyk^her Z2:19.
- *hyneken 'fox'. SM 忽捏堅 xuneken N247/11:1; AM هوناكان hunakan M191b.
- *hynesy/n 'ash'. SM 忽捏速 xunesu N87/2:25; AM هوناسون hunasun M191b.

- **ilk*(*^h)a* 'to separate'. UM ¹ いい (1956) U19:13a6; SM 亦^前 合 *il* χa N139/4:25. [The modern languages indicate **k*]
- **ima*(*h*)*a*/*n* 'goat'. SM 亦馬阿_期 *imaa*-*t*^{*h*} [PL] N151/5:11; AM '*ima*'an M153b. **ine*(*h*)*e* 'to laugh'. SM 亦捏額 *inee* N189/7:9; AM '*ina* M196b.
- *ire 'to come'. UM ᡝᡪᡵᠠᢉᠲᠠᠠᠡ ira-ksan [PSTP] D10:14a4; SM 亦^舌列 ire N1/1:1; AM [jul 'ira M197a.
- *irken 'people'. UM テーマ irkan D1:2; SM 亦兒堅 irken N5/1:3, 亦兒格捏 irken-e [DAT] N53/1:32; AM (火ン) 'irkan M197b; Ph 凹引 回 li 'irgen-e [DAT] Z18:8.
- *ite 'to eat'. UM ᡝᢏᡨᢒᠷᠣ ida-basu [COND] D10:36a3; SM 亦咥 ite N26/1:16; AM أبدا 'ida M194a; Ph IS NIN ZUIS ide-gsen [PSTP] Z51(4)b:7.
- **jasuln* 'bone'. UM جمالي *jasun* D17:31; SM 牙孫 *jasun* N175/6:19; AM ياسون *jasun* M388a.
- **jek^he* 'big'. UM 示介 *jaka* D2:2; SM 也容 *jek^he* N46/1:27; AM يك *jika* M388a; Ph उर 時じ *jike* P2:2.
- *josu/n 'rule'. UM 、 jusu D9:34; SM 約孫 josun N9/1:6; Ph 3 (25 josu P2:22.
- *kakč^ha 'single'. UM (مجل qaqča D7:18; SM $cap{a} = \frac{1}{2} \chi ax c^h a$ N4/1:3; AM المحافظ raqja M175a; Ph Le H $qaq c^h a$ P12:3.
- *kak^hai 'pig'. UM كَتَسَبَّرُ qaqai D15:5; SM ^中合^中孩 χαχαj N166/5:40; AM خافای каqai M175a; Ph E ലാദ qaqai Z25:36.
- *kal 'fire'. UM ٢ إمار عن 8 أي الله الماري الماري الماري الماري الماري الماري الماري الماري الماري *kal 'fire'. UM ٢ من عن 101 الماري (fire'. UM ٢ ماري الماري - *kar 'hand'. UM デー qar D19:13a11; SM 中合兒 χar N59/1:40; AM シ ビar M175b.
- *kasi(h)un 'bitter'. SM ^中合失溫 χašiun N77/2:9; AM غاشون *ʁašun* M178b.
- *kem 'fault'. UM 🕅 kam D10:7a4; Ph אום פא מעצ און gem−ijen [RFL] Z51(4)b:6.
- *kerel 'light'. UM (マーマーマー karal D11:157b9; SM 格^舌連_勒 kerel N98/2:42; Ph 向 HU 己の *gerel—un* [GEN] P13:8.
- *kørehesy/n 'beast'. UM (حواكوتر kyruka-d [PL] D19:8b12; SM 戈^舌劣額孫 korøesun N26/1:16; AM حواسون kurasun M172a; Ph 配く 肥く 肥く 空 gørø'e Z26:33.
- *kurpa/n 'three'. UM うっちつ qurban D1:4; SM ^中忽兒班 xurpan N4/1:3, ^中忽^舌兒 巴納 xurpan–a [DAT] N251/11:12; AM シビア Brban M179b; Ph L TH R qurban P2:20.
- *k^ha(h)a 'to shut'. SM ^中合阿 χaa N245/10:34; AM ⁱ qa M297b.
- * $k^hahalka$ 'gate'. UM $\int qaqalq-a D15:9$; SM $\stackrel{\circ}{=} \ominus m_{ij} \stackrel{\circ}{=} \partial qal\chi a N247/11:2;$ AM كَلُنَّا $q\ddot{a}l\ddot{u}$ IM444b.
- *k^haič^hi 'scissors'. UM ܐܝܝܝܐ qajiči D22:35b2; SM 孩亦赤 xajič^hi HY1:11a; AM قايجى qaiji M346a.
- **k^hakač^ha* 'to leave'. UM (متربيبيا qaäača–n [cv] D11:156a13; SM ⁺合⁺合察 χαχαč^ha N177/6:22; AM قاغاجا qaʁaja M286a.

- *k^halahun 'warm'. UM ᠭ᠇ᠯ᠇ᠬᡨ qalaqun D12:16; SM ^中合刺溫 χalaun N201/8:17; AM בולפפנ glawun M288b; Ph L 已 凹砺 qala'un P2:19.
- *k^hamt^hu 'together'. UM பிர்ச்சி qamtu D9:35; SM ^中含禿 χ amt^hu N64/1:44; AM قامتو qamtu M290b; Ph 已因 如何 qamt^hu Z18:35.
- *k^hara [•]black'. UM ᡣ gar-a D17:14; SM ^中合^舌刺 χara N21/1:13; AM [•] 資 M292a.
- *k^hari 'to return'. UM كَتَرَيْتَ (qari-qsan [PSTP] D10:34b3; SM 中合^舌里 χari N94/2:37; AM قاريبه qari-ba [PST] M280b.
- *k^hauč^hin 'old'. UM (مورية' qaučin T70:10; SM ^中合兀陳 χauč^hin N78/2:12; AM أو جين qujin M313b.
- *k(^h)auli 'law'. UM 「テレーマ qauli D21:9; SM ^中合兀里 χauli N229/9:46; Ph 巴 巴豆 尼亞 qa'uli P2:20. [The modern languages indicate *k^h]
- *k^hečije 'when'. UM (حجر) kačij-a D10:35a5; SM 客只額 k^hečie N100/2:45; AM لا كرجبيل kijija M213b.
- *k^hei 'wind'. UM 🛇 kai D11:167a6; SM 克 k^hej N31/1:19; AM كبى kĭji N213b; Ph மா k^her P13:12.
- *k^hele/n 'tongue'. UM (₩ kala D11:165a11; SM 客連 k^helen N104/3:2; AM كيكَن kilăn N214b.
- *k^hen 'who'. UM (∀ kan D17:19; SM 虔 k^hen N68/1:48; AM كين kin M216a; Ph الكات k^hen Z36:42.
- **k^hetyn* 'how many'. UM (محيدون *kadun* D4:1; SM 客敦 *k^hetun* N204/8:33; AM كيدون *kidun* M203b.
- *k^hi 'to do'. UM (、んん ki-kun [FUTP] D14:7; SM 乞坤 k^hi-k^hun [FUTP] N152/5:13; AM シシシ ki-ku [FUTP] M219a; Ph 凹る 凹の k^hi-k^hun [FUTP] Z18:31.
- *k^hičahar 'border'. UM 「テナナ・ティート qičaqar D17:31; SM 乞札阿^舌兒 k^hičaar N257/11:40; AM 道道ar M135a; Ph 凹 凹 H k^hijaar-a [DAT] P13:4.
- *k^hič^hi(h)e 'to strive'. UM (マママイ kičika−n [cv] D10:7b5; SM 乞赤額 k^hič^hie N218/9:21; AM シュー kija M219a.
- *k^hilpar 'easy'. UM 介、行介 qilbar D11:156b12; SM 勤_勤巴兒 k^hilpar N35/1:21, 乞_勤 八^舌剌 k^hilpar—a [DAT] N22/1:14; AM ジリロ qilbar M298a.
- *khimusun 'nail'. SM 乞木速 khimusu N199/8:7; AM قبموسون qimusun M298b.
- *khithat 'Chinese'. UM 介、ティレイ qidat D22:3a2; SM 乞塔傷 khithath N132/4:10.
- *k^hohasun 'empty'. UM Srowow ququsun D11:158a14; SM ^中豁斡孫 χoosun N105/3:6; AM فوسون qusun M306b.
- *k^hoina 'back'. UM (بنجي qujina D6:23; SM ^中豁亦納 χoina N11/1:7; AM قوينه quina M299b; Ph L< 33 ঢ qojina P12:1.
- *k^hojar 'two'. UM كتوبيار qujar D9:34; SM ^中豁牙兒 χojar N3/1:2; AM قوبيار qujar M303b; Ph ല< Зн qojar Z12:11.
- *k^hola 'far'. UM أوك qula D9:9; SM ^中豁羅 χolo N227/9:42; AM أوك qula M300a.
- *k^honi/n 'sheep'. UM (محبور) quni-t [PL] D13:3; SM 中部初 χonin N19/1:11; AM قونين qunin M302a; Ph L < வன வா Z32:25.
- *k^høisy/n 'navel'. UM (كويسون kyisun T30:7; SM 阔亦孫 k^hoisun HY1:24a; AM كويسون kuisun M226a.
- *k^høit^hen 'cold'. UM (حويتان kyidan T80:3; SM 阔亦田 k^hoit^hen N207/8:40; AM لمحويتان kuitan N226a.

- * $k^h \phi k^h e$ 'blue'. UM ($free ho k^h c$ N203/8:31; AM كوكه kuka M220b.
- *k^høl 'foot'. UM (み kyl—ijan [RFL] D10:37a6; SM 阔_物 k^hol N203/8:29, 阔里顔 k^hol-ijan [RFL] N173/6:13; AM シレン kul M221b, シレン kul-an [RFL] M222.
- **k^hølesy/n* 'sweat'. SM 阔列孫 *k^holesun* N254/11:25; AM كولاسونى *kulasun-i* [ACC] M333b.
- *k^høŋken 'light'. UM (كريم kynkka–lan [DER] D10:14a7; SM 匡格捏 k^hoŋken–e [DAT] HY2:16a; AM كونكان kunkan M223a.
- *k^huč^ha 'to bark'. SM ^中忽察 χuč^ha N189/7:10; AM 達 quja M307a.
- *k^h[u/o]i 'sheath'. AM قوى qui M309a. [The AM form with uvular q shows that the vowel may be *u or *o. Reflexes in modern languages (e.g. Halh xvi хуй) indicate *u.]
- *k^hulakai 'thief'. UM أَسَمَّتَ qulqai D7:26; SM ^中忽剌^中孩 χulaχaj N131/4:8; AM قولاغاي qulazai M309a; Ph 모호 교 모ਤ장 qulaqai Z5:35.
- *k^hupi 'destiny'. UM (محمر) qubi T24:2; SM ^中忽必 χupi N23/1:15; AM قوبى qubi M306b.
- *k^hura 'rain'. UM ך ק*ur-a* D11:158a1; SM ^中忽^舌刺 χura N108/3:12; AM قوره qura M101b; Ph Lo H qura P13:12.
- *k^huru(h)u/n 'finger'. SM^中忽^舌魯兀 χuruu N178/6:43; AM قورون qurun M312b.
- *k^hynty 'heavy'. UM (ひっつつ kyntu D10:23b4; SM 坤都 k^huntu N146/4:47; AM كند و kndu M227b.
- * $k(^h)yr$ 'to reach'. UM (天子子 kyr-basu [COND] D3:5; SM 古兒 kur N32/1:20; AM $j \neq kur$ M229a; Ph 回可H k^hyr P13:4, 四H k^hur Z10:7. [SM indicates *k. Ph indicates * k^h , as do all modern languages. There are a few words which begin with the character 古 ku in Sino-Mongolian but with k^h in 'Phags-pa, indicating Old Mongolian *k and * k^h , respectively; the other sources do not differentiate these consonants. Hattori (1976) says that although there are Chinese characters which are pronounced k^hu (e.g. 窟 'cave'), they are avoided because they have unsuitable meanings, and are replaced by $\exists ku$ (meaning 'ancient'), which thus denotes either ku or k^hu .]
- *k(^h)yreken 'son-in-law'. UM (کوراکان kyrakan D12:19; AM کوراکان kurakan M229a. [The modern languages indicate *k^h]
- *maŋlai 'forehead'. UM أَسَمَّتَمَ manklai D12:9; SM 莽來 maŋlaj N4/1:3; AM منقلاى mnglai M222a.
- *masi 'very'. UM **行於** masi D10:10b7; SM 馬石 maši N69/2:1; Ph B GG maši P13:2.
- *mete 'to know'. UM ディティ (@ mada-ku [FUTP] D11:161a7; SM 篾迭坤 mete-k^hun [FUTP] N143/4:35; AM ميداكو mida-ku [FUTP] M235a; Ph الما الله mida-k^hun [FUTP] Z21:14.
- *mik^ha/n 'meat'. UM (miqa-bar [INST] D10:23a3; SM 米^中罕 mixan N232/10:4; AM ميقان miqan M236b.
- *miŋk^ha/n 'thousand'. UM أَنَّ minkqan D13:30; SM ightarrow minkqan N208/8:43;AM منقان mnqan M236b; Ph א שוס שוס minqan P12:4. [The AM form shows that OM has $*k^h$, but the modern languages indicate *k]

- *mokai 'snake'. UM أسموغاى muqai D9:41; SM 抹^中孩 moxaj N102/2:48; AM موغاى musai M238a; Ph 장 모 ح موغاى more i Z11:33.
- *moŋkal 'Mongol'. UM أَسَمَ munkqul D1:2; SM 忙^中豁_动 maŋχol N108/3:12; AM منغول mnʁul M238a; Ph ষدا لا د moŋqôl Z36:8.
- *mori/n 'horse'. UM 「マテマ' murin D13:32; SM 秣^舌驎 morin N55/1:36, 秣^舌驪納察 morin—ač^ha [ABL] N265/12:1; AM مورين murin M238b; Ph B< Hō morin Z12:13, B< Hō loā morin—ijan [RFL] Z51(3)b:3.
- *morit 'horses'. UM Yourit D13:3; SM 秣曬 molit^h N205/8:35; AM بوريد murid M238b; Ph 谒 H J I morid Z22:18.
- *mølsy/n 'ice'. UM أَسَابَتْهُمَ mylsun D10:6a2; SM 沫癫孫 molsun N238/10:12; AM مولسون mulsun M239a.
- **mør* 'path'. UM ゲット *myr* D6:20; SM 抹兒 *mor* N90/2:28; AM シッ *mur* M239b; Ph 尼U+H *mør* Z4:18, 尼U+H 3UH *mør*-*ijer*[INST] Z18:17.
- *nahat 'to play'. UM ۲۰۰۲ naqad–un [cv] D11:167a3; SM 納阿敦 naat–un [cv] HY1:17b; AM أَسَتَتَبَهُ nat–ba [PST] M132a.
- *napč^hi/n 'leaf'. UM ܐᢏ nabčin D15:34; SM 納卜陳 napč^hin HY1:4a; AM نابجين nabjin M244a; Ph 因同 时因 nabč^hin P13:11.
- *narin 'fine'. UM '۲, warin D20:14; SM 納^苦鄰 narin N243/10:25; AM نارين narin M246a.
- *nasu/n 'age'. ٦-++ nasu D10:11a4; SM 納孫 nasun N66/1:46; AM ناسون nasun M246b; Ph ت محمق nasun P13:12.
- *nehe 'to open'. UM アテベj naka D21:40; SM 捏額 nee N246/10:43; AM 辿 ni-ba [PST] M248b.
- **nek^he* 'to weave'. UM ᠠᠠᠭᠯ *naka-mal* [DER] D22:37b5; SM 捏客 *nek^he* N90/2:28; AM ビンド *nika-ba* [PST] M248a.
- *nekhei 'sheepskin'. SM 遛克 nekhej N112/3:20; AM نبكاى nikai M248a.
- *nere 'name'. UM ᡝ᠇ᠷ ᠭ nar-a D10:2b3; SM 捏^舌列 nere N7/1:4; AM ᡤ nira M248a; Ph ਯ ll ll nere P13:12.
- *nič^hykyn 'naked'. SM 你出裙 nič^hykun N86/2:24; AM نبجوكون nijukun M221b.
- *nihu 'to hide'. UM 'To niqu-ča [DER] T25:18; SM 你兀 niu N86/2:24; AM نيو niu M257b.
- *nilk^ha 'baby'. UM ך nilq-a D10:18a3; SM 你動^中合 nilχa N195/7:39; AM نيلقه nilqa M253b.
- *nimken 'thin'. UM ᡝᠢᠯ nimkan D10:6a1; SM 您堅 ninken HY1:27a; AM نيمكان nimkan M257a.

- **nisi* 'to beat'. SM 你失秃^中孩 *niši-t^hu*χ*aj* [IMP] N199/8:9, 你失_黑荅^中渾 *niši-xta-*χ*un* [PASSIVE-FUTP] N227/9:43.
- *nity/n 'eye'. UM **موجع nidun** D15:8; SM 你敦 nitun N62/1:43, 你覩邊 nitu-pen [RFL] N254/11:25; AM نينون nidun M249b. [The SM reflexive shows that the word is front-vocalic]
- *noir 'sleep'. UM ᡝᠣᠢᠷ nujir T34r14; SM 那亦^舌兒 noir N205/8:37; AM ᡤ يويير nujir M382b.
- *nokahan 'green'. UM سوسوس nuquqan D20:13; SM 那^e豁安 noxoan N170/6:2; AM نو غان nuwan M259a.
- *nok^hai 'dog'. UM ܐܝܝܝܟܪ nuqai D15:10; SM 那^中孩 noχaj N78/2:11; AM نوقاى nuqai M259a; Ph 장 E ሩ Зउ noqôi Z51(3)a:10.
- *nøk^her 'friend'. UM ひてのた nykur D14:18; SM 那可兒 nok^har N13/1:8; AM シシ nukr M261a; Ph 図イ 凹イH nøk^hør Z51(3)b:2.
- **oira* 'near'. UM ບາກ j *ujir-a* D9:9; SM 斡亦^舌刺 *oira* N31/1:19; AM しょう '*uira* M263a.
- **ol* 'to find'. UM いってうけの *ul-basu* [COND] D10:20a5; SM 斡_勤 *ol* N114/3:24; AM *'ul-ba* [PST] M265b; Ph K 回日 ズリ *ol-uqsad* [PSTP] Z34:29.
- *olan 'many'. UM シーチャー ulan D7:27; SM 斡樂 olon N56/1:37; AM しんしょ) 'ulan M264a; Ph K 回 olan Z5:29.
- *ora 'to enter'. UM ^{wwwww} ura-qsan [PSTP] D7:31; SM 斡^舌羅 oro N5/1:3; AM أور ⁺ 'ura M269a; Ph K H< oro Z34:30.
- *ø[h/p]er 'self'. UM いのかり ybar D7:2; SM 斡額兒 oer N86/2:24; AM しい W279b; Ph ちいく ビリト ø'er Z18:32. [UM indicates *p and the other sources *h; the modern languages, e.g. Halh oor oop, indicate *h]
- *øk 'to give'. UM كرمي المرابي (IPFG] D6:28; SM في ok^h N13/1:8; AM او کبه 'uk-ba [PST] M274a; Ph قادة ĝg Z2:14.

*øk^hin 'girl'. UM マママ ykin D12:19; SM 斡勤 ok^hin N6/1:4; AM وكين 'ukin M275a. *ønteke/n 'egg'. SM 完迭堅 onteken N276/12:31.

- *pahu 'to go down'. UM 句 baqu-ču D19:10a12; SM 保兀 pawu N63/1:43; AM 少 bau M114a; Ph 同 凹 5 ba'u P1:17.
- *pajan 'rich'. UM front bajan D10:5a6; SM 伯顏 pajjan N3/1:2; AM بابان bajan M114a.
- **pari* 'to hold'. UM كَمَرْبَعَرْ bari–lan [Der] D12:16; SM 把^音里 pari N19/1:12; AM بارى bari M112a; Ph ק اج bari P1:16.
- *pars 'tiger'. UM 分★ bars D6:14; SM 巴兒思 pars N78/2:12; AM بارس bars IM402b; Ph ╗HZ bars P2:24.
- **pasa* 'also'. UM **5 //** *basa* D6:24; SM 巴撒 *pasa* N19/1:12; AM نسه *bsa* M113a; Ph □ Z *basa* P2:19.
- *pat^hu 'firm'. UM 句子 badu-da [DER] D19:9a7; SM 巴禿 pat^hu N256/11:36; AM batu M113b.

- *peje 'body'. UM f う j baj-a D12:15; SM 別耶 peje N33/1:20; AM 4... bia M198a; Ph I J beje P12:1.
- *peri 'daughter-in-law, bride'. UM あって bari T51:10; SM 別^舌里 peri N272/12:24; AM シーシー biri M118a.
- **pič^hi* 'to write'. UM ᠪᡪᠵᠵᠢᠿᠶ *biči–kui* [FUTP] D11:159b2; SM 必赤 *pič^hi* N203/8:31; AM (PST] M118b; 同て 昭て *bič^hi* P2:26.
- **pič^hik* 'letter'. UM () bičik D2:3; SM 必赤_克 *pič^hik^h* N203/8:31; AM () bičik D1:8b; Ph 印 田石 田石 白花^hig P1:20.
- *pisilak 'cheese'. SM 必石刺黑 pišilax HY1:13a; AM بيشلاقين bišlaq-in [GEN] M119b.
- *pos 'to rise'. UM のhmv (bus-qui [FUTP] D10:9b1, のhmrz (bus-qa-ču [CAUS-IPFG] D10:37b2; SM 李思 pos N98/2:42; AM チェッシュ bus-ba [PST] M122a; Ph R モ bôs Z36:24, Rモマ E 凹す bôs-qa-ju [CAUS-IPFG] P13:4.
- * $p\phi(h)ere$ 'kidney'. SM 孛額^音列 poere N100/2:44; AM 近 bŭra L1266.
- *pø(h)esy/n 'louse'. SM 李額孫 poesun N201/8:18; AM بواسون buasun M123a.
- **pulak* 'spring'. UM 허ㅠ *bulaq* D17:50; SM 不刺^中合 *pula*_又-a [DAT] N128/4:2; AM ショ*bulaq* M289a; Ph 雨 已 *bulaq*-a [DAT] Z8:28.
- **pyse* 'belt'. UM Gオッ *byz-a* D13:25; SM 不薛 *puse* N117/3:27; AM بوسه busa M128b.
- *sa(h)a 'to milk'. SM 撒阿 saa N152/5:17; AM w sa M319b.
- *sahu 'to sit'. UM 'mmunter' saqu-qsan [PSTP] D9:13; SM 撒兀 sau N24/1:15; AM sau M319a; Ph マ 四 sa'u Z5:19.
- *sahuri/n 'base'. UM /mmuss / saqurin D11:160a3; SM 撒兀音鄰 saurin N64/1:44.
- *sain 'good'. UM ساین sajin D7:8; SM 撒因 sain N6/1:4; AM ساین sain M315a; Ph ت تع sajin Z51(4)a:1.
- *sak^hal 'beard'. UM سقال saqal T70:5; SM 撒^中合 المعنية sayal N203/8:28; AM سقال sqal M318a.
- **sak^hi* 'to guard'. UM אייייי' *saqi–n* [cv] D10:4b7; SM 撒乞 *sak^hi* N79/2:14; AM مناقيبه saqi–ba [PsT] M318a; Ph マロ saqi Z10:4.
- *sam 'comb'. SM 毿 sam HY1:10b; AM مَسْم sm M317b.
- *sara 'moon'. UM ᠠ᠇ᡵᠠ ' sara D5:14; SM 撒^舌刺 sara N21/1:13; AM سارا sara M318b; Ph m H zara P2:24.
- *se(h)yl 'tail'. SM 薛溫_勤 seul N24/1:15, 薛兀列徹 seul-eč^he [ABL] N77/2:10; AM بى وك suul M330b.
- **setk^hil* 'mind'. UM 'ܐܕܪܐ *sadkil* D9:22; SM 薛_惕乞_勒 *set^hk^hil* N125/3:48; AM سينكيل *sitkil* M321b; Ph Zோ மா عود *sedk^hil* Z36:21.
- *si(h)e 'to piss'. SM 失額 šie N194/7:28; AM شيكو ši-ku [FUTP] M331b.
- *sini 'new'. UM أشينى šini D22:7:5; SM 失你 šini N265/12:1; AM شينى šini M334b; Ph ნ ق كان šine Z5:43.
- **siŋk^hor* 'falcon'. UM ᠠᡳᡊᡣᠣᠷ *sinkqur* D22:16a4; SM 升^中豁兒 *šiŋχor* N63/1:43; AM 流道 *šŭqŭr* L76; Ph เวิลา ヒ (H *šiŋqôr* Z59.
- *sipahu/n 'bird'. UM 小分 sibaqun D17:29; SM 失鴇溫 šipawun N111/3:18; AM śipawun M332b; Ph 厉 同 巴爾 šiba'un Z51(1)a:5.

- *sipar 'mud'. SM 失巴兒 šipar N102/2:48; AM شيبار šibar M187a.
- *sira 'yellow'. UM '*** sira T29:5; SM 失^舌刺 šira N21/1:13; AM سيره šira M335b.
- *siri(h)e 'table'. SM 石^舌里額 širie HY1:11a; AM شير šira M166a.
- *sity/n 'tooth'. SM 失都 šitu N82/2:19; AM شيدون šidun M332b.
- *su(h)u 'armpit'. SM 速兀 suu N214/9:13; AM صو su M328b.
- *sumu/n 'arrow'. UM hove sumun D11:162b3; SM 速門 sumun N147/4:49; AM weaci
- *sur 'to learn'. UM かってのかの sur-basu [COND] D13:17; SM 速^舌兒 sur N149/5:4; AM wet M328a; Ph ZoH sur P11:6.
- *sykhe 'axe'. SM 速客 sukhe N214/9:14; AM سوكا suka M189b.
- *syme 'temple'. UM ح الم sym-a D15:4; SM 速篾思 sume-s [PL] N267/12:9; AM سومه suma M329a; Ph الح والع والح syme P12:6.
- *ta(h)ari 'sore'. SM 荅阿^舌里 taari N24/1:15; AM பி. பி. வால் M139b.
- *taru 'to press'. UM ママママ taru-n [cv] D12:16; SM 荅音魯 taru N246/10:43; AM aru M339b; Ph II Ho daru Z36:36.
- *tehel 'gown'. UM ᡨイオ takal D11:160a9; SM 迭額動 teel N33/1:20; AM ふんし M141b.
- *tehere 'above'. UM ᡨᠭ (ג לא *takar-a* D9:30; SM 迭額^舌列 teere N1/1:1; AM بير dirạ M142a; Ph யய deere P12:3.
- *tehesyn 'rope'. UM ٩-٢٩٠٠ takasun D19:9a4; SM 迭額孫 teesun HY1:10a; AM ديسون disun M126a.
- *tehy 'younger brother'. UM 研合 taku D9:27; SM 迭兀 teu N18/1:11; AM シ du M256a; Ph 坦 回 de'u Z10:11.
- *tok^halaŋ 'lame'. SM 朵合郎 toxalaŋ HY1:24b; AM دوقولان duqulan M143a. [AM indicates *k^h. Most modern languages indicate *k.]
- *tolaha/n 'seven'. UM المتحلق tuluqan D8:12; SM 朵羅安 toloan N45/1:27; AM دولان dulan M142b; Ph لا حرك dolo'an Z14:28.
- *tot^hara 'inside'. UM و *tudur-a* D13:5; SM 朵脫^舌剌 tot^hora N6/1:4; AM دوتار dutar M143b: Ph 以 瓜 (H dot^hora Z36:39.
- *tøč^hi/n 'forty'. UM موجين tyčin D9:25; SM 朵臣 toč^hin N177/6:23; AM دوجين dujin M143b.
- *tørpe/n 'four'. UM موريان tyrban D8:13; SM 朵兒邊 torpen N11/1:7; AM دوريان durban M143b; Ph ሠ ط forbe P13:7, ሠ ط ال الم dôrben–e [DAT] Z29:30.
- *tumta 'middle'. UM 아っつう tumda-tu [DER] D11:159b9; SM 敦荅 tunta N4/1:3; AM دونده dunda M144b; Ph 川网 川 dumda Z5:42.
- *tyhyreŋ 'full'. UM ᡐᠣᡵᠿᡵ᠇ᡪᠵ tykurank D10:5a1; SM 都兀^舌良 tuureŋ N145/4:39; AM دوران duran M148b; Ph யி மிர மிர மர் dy'ur-P12:7.
- *t^hak^hi 'to offer'. UM ᠳᠬᠠᠭᠬᠯ᠋ᠭ taqi-qui [FUTP] D10:17a2; SM 塔乞 t^hak^hi N216/9:20; AM 山道山 aqi-ba [PST] M124b.
- *t^hak^hija 'hen'. UM എന്ന taäij-a T68:18; SM 塔乞牙 t^hak^hija N141/4:30; AM تاقيه taqia M148b رابيت taqia M148b طاقيه taqia M341a; Ph 岡 ଓ З t^hak^hija P4:18.

*thani 'to recognize'. SM 塔泥 thani N32/1:20; AM نانيه tani-ba [PST] M340b.

- *t^hapi/n 'fifty'. UM ᠳ tabin D19:7b11; SM 塔賓 t^hapin N136/4:18; AM تابين tabin M338a.
- *t^hapu/n 'five'. UM ᡐ**᠇ᠪᠠ /** tabun D1:4; SM 塔奔 t^hapun N19/1:11; AM ビルロン tabun M338a; Ph II わのて t^habun Z5:42, II わの t^habun-a [DAT] Z2:19.
- *t^hat^ha 'to pull'. UM ᠳᠠᡨᡣᠣᠯᠣᠯ tada-qul-un [CAUS-CV] D13:32; SM 塔塔 t^hat^ha N76/2:7; AM 近tata M342a; Ph 即即t^hat^ha P2:21.
- *t^haulai 'hare'. UM जन्ननि taulai D5:13; SM 討來 t^hawlaj N272/12:20; AM تاولاى taulai M343a; Ph 때 වਤਰ t^hawlai Z34:32.
- *t^hemehe/n 'camel'. UM ٩-٢٠٠٠ tamakan D22:37b2; SM 帖篾延 t^hemeen N244/10:28; AM تيمان *timan* M345b.
- *t^hemyr 'iron'. UM ᠳᠮᠦᠷ tamur D9:23; SM 帖木^舌兒 t^hemur N199/8:10; AM نيمور timur M346a.
- **t^hoha/n* 'number'. UM 'tuqan D13:24; SM 脫安 *t^hoan* N23/1:15; AM توننو *tun–du* [DAT] M352a; Ph 岡< 巴≺ゐ *t^ho* 'on P12:3.
- * $t^{h}oka(h)an$ 'kettle'. SM 脱^中豁阿泥 $t^{h}o\chi oan-i$ [ACC] N277/12:35; AM 近 tu Ban M109a.
- *t^hosu/n 'fat'. UM ماماه *tusun* D11:167a8; SM 脫孫 t^hosun N254/11:24; AM توسون tusun M351b; Ph 때 حتاة t^hosun Z11:20.
- **uhu* 'to drink'. UM ᠠᠳᠳᠪᠰᠣ *uqu–basu* [COND] D19:11b9; SM 兀兀 *uu* N179/6:34; AM 步 ' '*u* M371b.
- *uila 'to cry'. UM ᡝᠣᡳᡳᠯᡣ᠇ᠤᠢ ujila-qui [FUTP] D10:35b4; SM 委亦刺 ujila N56/1:37; AM اويلا uila M362a.
- *uk(^h)ija 'to wash'. UM いっついつうか uqija-basu [COND] T50r3; SM 兀乞牙 uk^hija N189/7:12; AM しょう 'usa M362a. [SM indicates *k^h, and AM *k. The modern languages indicate *k.]
- *ulus 'state'. UM いってい いってい いってい いっしょう (ACC] N73/2:4; AM にしょう (ACC] N73/2:4; AM しんしょう (ACC] M204a; Ph ゆ 己のマ ulus P12:1.
- *umart^ha 'to forget'. UM ひっていっている umarda-ču [IPFG] D7:25; SM 兀馬^舌兒塔 umart^ha N152/5:15; AM ベビビ mrta M234a. [The modern languages lack reflexes of initial *u]
- *unt^ha 'to sleep'. UM 'unda-qui [FUTP] D10:9b1; SM 穩榻 unt^ha HY1:17b; AM (回) 'unta M365a.
- *unu 'to ride'. UM 'vorono unu-qu [FUTP] D13:32; SM 兀訥 unu N89/2:43; AM jei unu M365a.
- **uri* 'to invite'. UM ᡝᠣᡵᡪᠬᠤᠢ *uri-qui* [FUTP] D11:158b12; SM 兀^舌里 *uri* N203/8:32; AM اوريده '*uri-ba* [PST] M366a.
- *urita 'before'. UM 'すいい 'urida D8:2; SM 兀^舌里苔 urita N246/10:44; AM / しょう 'urida M366a; Ph ゆ H び U urida P12:1.
- **urt^hu* 'long'. UM ᡝᠣᡵᠣᠠᠣ *urtu* D13:35; SM 兀兒圖 *urt^hu* N106/3:8; AM ᡣ '*urtu* M366b; Ph ЮН 颐 *urt^hu* Z36:24.

- *usu/n 'water'. UM 'totor' usun D13:4; SM 兀孫 usun N81/2:17; AM 'eau 'usun M368b; Ph lo 又可 usun P2:18.
- *yče 'to see'. UM ᠠᢏᠵᠠᡇ yča-ku [FUTP] D11:163b5; SM 兀者 uče N5/1:3; AM しょう 'uja-ku [FUTP] M218b; Ph 国 即 即 yje Z36:40.
- *yčyhyr 'point'. UM ¹ עלשטר (DAT] včukur D17:46; SM 兀主兀^舌列 učyur-e [DAT] N166/5:41; AM (中央ログロン (ローク) (マチャック) (1) (ローク) (ロ
- *yile 'deed'. UM 'マママオ' yila D8:9; SM 委亦列 ujile N197/7:49, 委亦列昔 ujile-s-i [PL-ACC] N279/12:49; AM 'リンゴーロン 'uila M375b; Ph ちょうて きょ yile P13:13, ちょうて きょ マッ yile-s-i [PL-ACC] P13:4.
- *yje 'joint'. UM سر) yj-a D11:166b2; SM 兀也 uje N18/1:11; AM اوبا 'uja M382b.
- *yke 'word'. UM かっ yka D2:1; SM 兀格 uke N16/1:9; AM しょ 'uka M373a; Ph 凹の い 'yge Z5:30.
- *yk^hy 'to die'. UM ᠠᡡᢙᠭᠯ᠇᠇ᠳ yku-ksat [PSTP] D14:5; SM 兀窟 uk^hu N24/1:15; AM أوكو 'uku M378b; Ph قات ত א^hu P7:b2.
- *ynije/n 'cow'. SM 兀你延 unien N121/3:38; AM اونايان 'unajan M381b.
- *yntyr 'high'. UM יעדדים 'ndur D15:3; SM 溫都兒 untur N64/1:45; AM 'undur M276b; Ph รบดช มดฟ yndur P12:3.
- *ytyr 'day'. UM ບບບບ ydur D10:18a5; SM 兀都兒 utur N5/1:3; AM اودور 'udur M273a; Ph 505 いのり ydyr P13:8.

9

THE MONGOLIC LANGUAGES¹

In addition to the three fairly well-known languages, Mongolian, Buriad, and Oirad, the Mongolic language group contains the 'peripheral' languages Kamnigan, Dagur, Shira Yugur, Monguor, Santa, Bonan, Kangjia, and Moghol. Except for Moghol, which was spoken in Afghanistan, but is now almost extinct, the peripheral languages are spoken in northern China. Mongolian, Oirad, and Buriad are more or less mutually comprehensible. Speakers of these languages have used Classical Written Mongolian, and have had contact with each other throughout history. The peripheral languages differ more.

Data from languages other than Halh Mongolian are taken from the published literature.² We have modified the transcriptions somewhat to agree with our transcription of Halh Mongolian (see 9.13 for details). For the sake of consistency, only one source has been used for each language. For the peripheral languages in China, we have used the series of descriptions written by scholars from Inner Mongolia University, which are the most comprehensive ones available.

The number of first language speakers for the different languages and dialects given below should be regarded as rough estimates. They are based on censuses and on estimates found in the literature. Mongolian and Chinese censuses record only membership of ethnic groups and not the languages actually spoken. In the Soviet censuses, people were asked what they regard as their native language, but it seems that many who actually speak mainly (or even only) Russian still regard the language of their ethnic group (e.g. Kalmuck or Buriad) as their native language. In China, many members of ethnic Mongolic groups speak only Chinese. This is not seen in the census data, and the estimates of the number of speakers of different languages within an ethnic group are based mainly on *Language atlas of China* (Wurm and Li 1987) and on Ouyang and Zhou (1994). The spread of normative forms of Mongolian (Halh in Mongolia and Chahar in South Mongolia) tends to level out the dialect differences, but exactly how far this process has gone is difficult to ascertain.

We are concerned only with the internal development of inherited Mongolic words in the different languages, and we will not deal with phonological developments due to language contacts, which are quite prominent in the languages spoken in the Gansu–Qinghai area.

¹ For additional sources, see App. I, p. 225.

² The sources are given in the sections on the respective languages.



MAP 1. Map of the modern Mongolic languages

9.1 MONGOLIAN³

The Mongolian language is spoken by about 2.5 million people in the Republic of Mongolia, and by perhaps 2.7 million in South Mongolia (Öbör Monggol) in China (officially known in Chinese as *Neimenggu zizhiqu*, Inner Mongolia autonomous region). We will avoid the Sinocentric terms 'Outer Mongolia' and 'Inner Mongolia' for these areas. See Chapter 4 for the writing systems in use for Mongolian.

The dialect differences are rather small, and there is no generally accepted dialect division. The dialects are often referred to by names which primarily denote ethnic or political, rather than linguistic, units. To add to the confusion, dialects of Buriad and Oirad are officially regarded as dialects of Mongolian proper both in Mongolia and China.

Except for Üjemchin, all forms of Mongolian proper spoken in Mongolia can be regarded as Halh (Khalkha) dialects, characterized by depalatalization of alveopalatal affricates, changing Old Mongolian $*\check{c}^h$ and $*\check{c}$ to c^h and c, respectively, except when they occurred before *i. The Üjemchin dialect, which is spoken in South Mongolia and by recent immigrants from South Mongolia to Mongolia, does not have depalatalization. It is regarded as belonging to the Chahar dialect group by Chinese scholars. There is no agreement among Mongolian dialectologists about the division of Halh into subdialects, but a division into Central, Eastern, and Western Halh is often made.

The Mongolian dialects of South Mongolia are divided into six groups by Cenggeltei (1979*b*) and most other Mongolian dialectologists in China. One of these groups is Alshaa, which we classify as an Oirad dialect. The five others are Horchin, Harchin, Baarin, Chahar, and Ordos.

The Tsongool and Sartuul dialects, whose speakers moved from Mongolia into the South Selenge area in Buriatia in the seventeenth century (Clark 1979; C. Budaev 1992: 22), are usually regarded as southern dialects of Buriad. They have some Buriad morphological features, such as verb agreement, but do not show the characteristic phonological developments of Buriad. At least from a phonological point of view, they must be regarded as dialects of Mongolian proper (cf. Poppe 1955: 23; Buraev 1965; 1987*a*: 20; 1996: 13; C. Budaev 1977*b*; Janhunen 1992: 142). The Darhad dialect in north-western Mongolia, spoken by mongolized Oirads, must also be regarded as a dialect of Mongolian although it retains some Oirad features (Sanžeev 1931; 1953: 64f.).

In order to reach a dialect division which is independent of ethnic divisions and state boundaries, we will base it only on phonological criteria. Many different criteria have been used by dialectologists, and we will choose three innovations which seem to be basic (although such a choice must be somewhat subjective): depalatalization of affricates, deaspiration, and loss of palatalized consonants.

Affricate depalatalization made alveopalatal affricates dental before vowels other than *i. For example, Old Mongolian $*\check{c}alahu$ 'young', which is $\check{c}alv$ in most Mongolian dialects, became $ca\xi v$ залуу in Halh. Depalatalization is typical for the Halh dialects, and occurs also in Dariganga and in the Shiliin Gol, Ulaanchab, and Sönid dialects, which are classified within Chahar by Chinese Mongolists. We regard them as subdialects of Halh.

Deaspiration makes an initial aspirated consonant unaspirated if another aspirated consonant or a voiceless fricative follows in the same stem. For example, Old Mongolian t^{h} becomes t^{h} in dialects without deaspiration (e.g. Standard Halh Toc), and tos in deaspirating dialects. This innovation has spread in a central area, which includes Halh dialects in the south-west, in the east, and in the Gobi (Kara 1962; 1963; Coloo 1967c; 1970), as well as Chahar, Ordos, Naiman, and some other dialects of South Mongolia. The residual area where no deaspiration took place consists of central Halh, including the basis for the standard language, Darhad, and Hotgoid (Möömöö and Mönh-Amgalan 1984) as well as the Eastern dialects (including Harchin, Horchin, Baarin), and probably the Urad dialect. Since different sources give conflicting information on deaspiration in the Halh areas, the boundary shown on the language map is tentative.

Palatalized consonants (which developed before *i) were lost in the Eastern dialects (Harchin, Horchin, Baarin, Jaruud), and in Naiman; as a compensation, the palatalization (umlauting) of the vowel preceding the palatalized consonant became phonemic. For example, Old Mongolian *morin 'horse' is mor^{j}

in Standard Halh (морь), but became mær in these dialects.

Using these phonological criteria, the various dialects of Mongolian proper might tentatively be classified into six groups (1). The Urad dialect, which is not extensively documented, seems to have escaped all three innovations.

(1) Criteria for Mongolian dialect division

	affricate depalatalization	deaspiration	loss of palatalized consonants
Northern Halh	+	_	-
Southern Halh	+	+	-
Southern Mongolian	-	+	_
Naiman	-	+	+
Eastern Mongolian	-	_	+
Urad	-	-	_

The distribution and approximate number of speakers of these dialects are given in (2).

- (2) Dialects of the Mongolian language
 - Northern Halh (1.5 mill.): Central Halh, northern part of Western Halh, Darhad (20,000), Hotgoid, South Selenge 'Buriad' (Tsongool–Sartuul, 20,000).
 - Southern Halh (1.1 mill.): Southern part of Western Halh, Gobi Halh, Eastern Halh, Southern Halh, Dariganga (40,000), Shiliin Gol (50,000), Ulaanchab (30,000), Sönid (40,000).

Southern Mongolian (350,000): Chahar (100,000), Üjemchin (75,000), Heshigten (20,000), Ordos (130,000).

Naiman (100,000)

Eastern Mongolian (2.1 mill.): Baarin (110,000), Arhorchin (80,000), Ih Minggan, Ongniud (50,000), Harchin (350,000), Tümd, Hüree (80,000), Monggoljin, Horchin (1.1 mill.), Gorlos (35,000), Heilongjiang Dörbed (35,000), Jalaid (140,000), Jaruud (90,000). *Urad* (40,000)

The Chahar dialect as spoken in Shuluun Höh ('Plain Blue') banner was designated as the standard Mongolian language of China in 1980 (Türgen 1985). What is presented as the standard language of South Mongolia in dictionaries such as *Monggol Kitad toli* (1999) seems to be a compromise between different dialects, however.

We will use examples from three Mongolian dialects: Standard Halh (described in Chapters 1–7), Shuluun Höh Chahar (Dobo 1983*a*), and the Baarin dialect of Baarin Right banner (Sun et al. 1990). These represent Northern Halh, Southern Mongolian, and Eastern Mongolian, and also exemplify each of the three phonological innovations.

Shuluun Höh Chahar has the phonemes shown in (3) in native words (Dobo 1983a).

(3)	Ch	ahar	· pho	nemes							
	i	у		u			-	t ^{jh}			
	Ι	\mathbf{Y}		U	р	pj	t	t ^j			
	e	ø	Y	0			s		š	xj	х
	3	æ	a	э	m	mj	n	n ^j		ŋj	ŋ
							1	þ			
							r	r ^j			
									j		

The phonemes of the Baarin dialect are shown in (4) (cf. Sun et al. 1990).

(4)	Ba	Baarin phonemes										
	i	У	-	u		t ^h	čh					
		Y		υ	р	t	č	k				
			Y	0		S	š	Х				
	8	æ	a	э	m	n		ŋ				
						1						
						r						
							j					

All dialects of the Mongolian language have a phonemic vowel length contrast which, according to our analysis, is found only in the initial syllable. All dialects also have vowel harmony of the same type as in Halh.

9.2 BURIAD⁴

Buriad is spoken by some 260,000 persons in Russia, mainly in the Buriad Republic close to Lake Baikal and in the Aga area further to the east, by 60,000 in northern Mongolia, and by 100,000 (including 70,000 Bargu) in the Hölönbuir league in the north-eastern part of South Mongolia (China).

Characteristic phonological features of Buriad are the weakening of *s to h and fricativization of affricates. Another feature differentiating Buriad from Halh is the retention of word-final short vowels, for example, uula 'mountain' vs. Halh uub yy π (Old Mongolian *ahula).

The so-called Southern Buriad dialect (South Selenge Buriad or Tsongool–Sartuul) has not gone through any of these sound changes and is, as mentioned earlier, classified as a dialect of Mongolian proper by us. Kamnigan is often regarded as a Buriad dialect, but following Janhunen (1992) we regard it as an independent language.

There is no complete agreement about the dialect division of Buriad, but five main dialects may be recognized (C. Budaev 1978; Buraev 1987*a*: 17 ff.; 1996):

⁴ For notes and sources, see App. I, 9.2, p. 227.

144

- (5) Buriad dialects
 - (a) Eastern Buriad (Hori, Aga, Muhar-Sheber, Tugnui, Hilok, Buriad of Mongolia)
 - (b) Northwestern Buriad (Ehired, Bulagad, Kabansk, North Selenge, Ivalga, Bargazhan, Kachug, Boohon, Ol'hoon, Osa, Baigal–Hudari)
 - (c) Southwestern Buriad (Alair, Tünhen, Aha, Ungi, Zakamna)
 - (d) Nizhneudinsk
 - (e) Bargu

The Northwestern (Ehired–Bulagad) and Southwestern (Alair–Tünhen or Sajan) dialect groups are often regarded as the two branches of a Western Buriad dialect group. Western Buriad differs from the other dialects by contrasting short u and o (from Old Mongolian *y and * ϕ) in initial syllables. In the other dialects they have merged to u. This development seems to be spreading to some Western dialects (Kachug, Bargazhan, Ivalga), apparently due to the influence of the standard language based on Hori.

The Northwestern dialects differ from Southwestern by having the palatal glide *j* rather than the voiced alveopalatal fricative \check{z} (IPA [3]) as the reflex of Old Mongolian \check{z} before \check{z} (e.g. Ehired (NW) *jel* 'year' vs. Tünhen (SW) and Hori (E) \check{z} *el*; OM \check{z} *il*).

The speakers of the Ivalga and North Selenge dialects spoken to the south-west of Ulan-Ude originally came from an area to the west of Lake Baikal. According to Cyrenov (1996), the Ivalga dialect is closely related to Ehired.

Nizhneudinsk, the westernmost Buriad dialect, is conservative in the sense that it has retained k^h unchanged as a stop (with velar and uvular allophones depending on the vowel harmony class of the word). The easternmost dialect, Bargu, spoken in Hölönbuir league in China, consists of two variants, Old and New Bargu. Old Bargu has merged s and k^h to x. For example, Old Mongolian sara 'moon' and k^hara 'black' both became xar in Old Bargu (Standard Hori Buriad harə, xarə).

The Eastern Buriads traditionally used Classical Written Mongolian as their written language. A special Buriad variant of the Mongolian alphabet was devised in 1905 by Agvan Doržiev, who took the Oirad Clear Script as a model. The Cyrillic alphabet was used for writing Buriad from about 1840 in Western Buriatia (Montgomery 1994: 141), but after 1917 the Mongolian alphabet was introduced there as well. It was officially replaced with a Latin-based script in 1931. At first, Halh Mongolian was used as the dialect basis for this script, but this was almost immediately changed to the South Selenge (Tsongool–Sartuul) dialect, which is phonologically a dialect of Mongolian proper rather than Buriad. The dialect basis was changed to Hori in 1936, and it was decided in 1938 that the Cyrillic alphabet should replace the Latin one. Printing in the old Mongolian alphabet ceased in 1937, and printing in the Cyrillic alphabet started in 1939–40 (Šagdarov 1969). The Buriad Cyrillic alphabet is described briefly in section 4.2. In China and Mongolia, Buriad is regarded as a dialect of Mongolian, and the speakers use the respective forms of the Mongolian written language. Our Buriad data is taken from Čeremisov's (1951) dictionary of the standard language, based on Hori. The phonemes shown in (6) occur in indigenous words (cf. Buraev 1959; 1987*a*; Buraev, Bažeeva, and Pavlova 1975; Soktoeva 1988; 1989*b*).

(6)	Buriad	phonemes
(0)	Duriau	phonemes

i		u				t ^{jh}			
	ə	υ	b	bj	d	dj	g	gj	
e		0			s	š	Х	х ^j	h
	a	Э			Z	ž			
			m	m ^j	n	n ^j			
					1	ļj			
					r	r ^j			
						j			

Short *e* is pronounced [ul] and long *ee* is [e:]. According to Zolhoev (1972; 1973*a*), orthographic single vowels in non-initial syllables are phonetically reduced and epenthetic in the sense that both their quality and their place in a word can be predicted by rules, as is the case in Mongolian proper. We will write them with the schwa symbol. Unlike Mongolian schwas, Buriad schwas can appear word-finally, where they contrast with zero, as in *garə* rapa 'to go out' vs. *gar* rap 'hand' (both are *gar* rap in Halh), and they must be regarded as phonemes in this position. We adopt Zolhoev's solution which implies that vowel length is contrastive only in the initial syllable, as in our analysis of Mongolian. Thus we analyse orthographic single vowels in non-initial syllables as *ə* (e.g. алтан *alt^hən* 'gold'; Монгол *məngəl* 'Mongol'; морин *mər'ən* 'horse') and orthographic double vowels in non-initial syllables as (short) phonemic vowels (e.g. эмээл *emel* 'saddle'; улаан *vlan* 'red'; Буряад *bur'ad* 'Buriad').

We write the Buriad unaspirated stops as voiced, according to our informal observations (see 2.1.2), although more data is needed to clarify this point. The nasals [n] and [ŋ] stand in complementary distribution, [ŋ] occurring word-finally and before velar consonants, and [n] in all other environments (Zolhoev 1963*c*; Buraev 1987*a*: 46, 63). We regard them as allophones of /n/. Like Mongolian, Buriad has both pharyngeal harmony and rounding harmony.

9.3 KAMNIGAN⁵

Kamnigan is spoken by about 2,000 persons. Most of them live in China, by the river Mergel, north of Hailar. They only came there after the Russian revolution in 1917, and some speakers may still remain in Russia. There may also be a few Kamnigan speakers in north-eastern Mongolia, by the Onon river. The Kamnigan are bilingual in Evenki and their Mongolic language. The material used here

⁵ Kőhalmi (1959); Mišig (1961); Damdinov (1962; 1968; 1975; 1977; 1988); Rinčen (1969); Doerfer (1985*a*).

is from the Urul'ga dialect of Manchurian Kamnigan Mongol (Janhunen 1990*a*; 1992; 2003*c*). This dialect has certain features in common with Buriad (the developments $*\phi > u$, *s > h in syllable onset position, and *s > t in coda position). These features are not found in the Man'kovo dialect (Janhunen 1992), which is more similar to Halh, at least superficially. In the Urul'ga dialect, the reflexes of the Old Mongolian onset *s are [š] before *i* and [h] in other positions (disregarding some complications due to recent loan-words), and in the Man'kovo dialect they are [š] and [s]. We regard [š] and [h] as allophones of /h/ in the Urul'ga dialect. The phonemes shown in (7) occur in the Urul'ga dialect.

(7) Kamnigan phonemes

ťh kh i u t k υ p c^{h} e Э с a h m n ŋ 1 r i

Unlike the case of Buriad and Mongolian, the reflexes of Old Mongolian *u and *a (and *y and *e) contrast in non-initial syllables in Kamnigan. For example, the Old Mongolian words $*alt^han$ 'gold' and *nasun 'age' became alt^ha , nahv in Kamnigan, but alt^han , nahan in Buriad. Thus the reflexes of Old Mongolian short non-initial vowels must be analysed as phonemic vowels in Kamnigan, and consequently vowel length is phonemic in both initial and non-initial syllables. Kamnigan has pharyngeal vowel harmony as well as rounding harmony.

9.4 $OIRAD^6$

The Oirad language has about 200,000 speakers in western Mongolia, 210,000 in China, and 140,000 in the Kalmuck Republic north-west of the Caspian Sea in Russia. The Kalmucks are descendants of Oirad groups who settled in Russia during the 1630s. Their language is basically the same as that spoken by those Oirads who remained in the east or returned there during the eighteenth century, but because it has its own written standard and has been influenced by Russian, it is regarded as an independent language by some researchers. We regard Kalmuck as an Oirad dialect, however. The entire Kalmuck population was deported and spread over large areas of Siberia and Kazakstan in 1943. When they were allowed to return in 1957, Russian was made the only language of instruction in Kalmuck schools, but since the beginning of the 1990s teaching in Kalmuck has been intro-

⁶ For notes, see App. I, 9.4, p. 227.

duced (Pjurbeev 2001*a*). Like most other Oirads, the Kalmucks are Buddhists. A group of at least 2,500 Muslim (Sart or Ööld) Kalmucks live in Kyrgyzstan, near Karakol (previously Prževal'sk) to the south of Ysyk Köl.

Oirad is characterized by its conservative vowel system, which is essentially the same as in Old Mongolian. Another salient phonological feature is the strong reduction of non-initial short vowels. As illustrated in Figure 2.6, syllabic sono-rants occur even in careful lexical pronunciation where Halh has a schwa vowel, as in Kalmuck [batthr] vs. Halh [pathtar] 'hero'.

The dialect differences seem to be small. The main dialects in Mongolia are Dörbed (70,000), Bayad (45,000), Zahchin (30,000), Urianhai (30,000), Torguud (15,000), and Ööld (15,000). In China, the Torguud dialect (130,000) is spoken in Xinjiang, and the Hoshuud dialect (40,000; also known as the Kök Nuur or Qinghai dialect, or as Deed Mongol 'Upper Mongol') in Qinghai.

The Alshaa (Alshaa–Ejnee) dialect (40,000) spoken in Alshaa league in the western part of South Mongolia has some Mongolian and some Oirad features. Cenggeltei (1957–8) classified it together with Hoshuud as forming a dialect intermediate between Xinjiang Torguud Oirad and Mongolian proper. In Cenggeltei (1979*b*), he reclassified it as belonging to the 'Inner Mongolian dialect', that is, Mongolian proper, while Hoshuud was classified as Oirad. Since they have typical Oirad vowel systems we regard both Alshaa and Hoshuud as Oirad dialects.

The Kalmucks are divided into two main dialects: Dörbed, spoken in the western part of Kalmuckia, and Torguud, spoken in the eastern part. The Buzaawa dialect spoken by the Don Kalmucks is close to Dörbed. There is also a small Hoshuud group, whose dialect is similar to Torguud. One major difference between Dörbed and Torguud is the opening of the rounded non-open vowels (*u > o and $*y > \phi$) in certain contexts in Dörbed (see 10.2; see also Kičikov 1967*a*).

The language of the Dörbed of Heilongjiang province in China, as described by Bao Xianglin (1985) and Zhang and Shaobu (1990), has apparently been Mongolized, having a Mongolian rather than Oirad vowel system. Some Oirad dialects in Mongolia are also changing under the influence of Mongolian proper. According to Sanžeev (1953: 62), Bayad, Hotgoid, and Zahchin are intermediate between Oirad and Halh (cf. also Vladimircov 1923). This is confirmed for Bayad by Öljeyibürin (1992) whose description shows it to be similar to Northern Halh. Zahchin seems to have undergone a vowel shift of the Mongolian type, judging from X-ray pictures given by Coloo (1965).

In China and Mongolia, Oirad is officially regarded as a dialect of Mongolian, and the Oirads are expected to use the Mongolian written language of the respective country. A special Oirad alphabet, the 'Clear script' (*todo bicig*), was designed by Zaya Pandita in 1648 and is still used by some of the Oirads in China, although the official policy since 1982 is that the Oirads should abolish the Clear script and use Mongolian as their written language (Svantesson 1991b). The Clear script is based on the Mongolian alphabet but is more phonematic and closer to the spoken language.

In Russia, the Cyrillic alphabet has been used for printing Kalmuck since 1919, and it was made the only official alphabet in 1925. It was replaced with the Latin

alphabet in 1930, but Cyrillic was reintroduced in 1938. The written language is based mainly on the Torguud dialect, but incorporates some Dörbed features.

Oirad is represented here by Kalmuck, with words taken from Muniev's (1977) dictionary. Cyrillic Kalmuck is described briefly in section 4.2. The Cyrillic script does not write schwas, and thus treats them as epenthetic vowels, but Pavlov (1983: 60) seems to regard them as phonemes. There is some variation in the use of reduced vowels (see Bitkeev 1976), and even the two authoritative dictionaries by Muniev (1977) and Omakaeva (2000) often disagree about the place where reduced vowels are pronounced (cf. also Street 1962 and Bitkeev 1964). We follow the Cyrillic script and analyse Kalmuck reduced vowels as epenthetic, absent from the phonological representation. This implies that vowel length is distinctive only in initial syllables. The Kalmuck phonemes are shown in (8) (cf. Pavlov 1963*b*; 1968; 1983; and Bitkeev 1965).

(8) Kalmuck phonemes

i e ε	y ø	u o a	p ^h b	t ^h d c ^h	(t ^{jh} (d ^j) č ^h		
				S	ľ š	X	
				Z			G
			m	n	nj	ŋ	
				1	þ		
				r			
			W		j		

We write the Kalmuck unaspirated stops as voiced, according to our informal observations (2.1.2), although more data are needed to clarify this point.

Oirad has palatal (back ~ front) vowel harmony, but no rounding harmony. The Alshaa and Hoshuud dialects have rounding harmony, however. For instance, the word 'beast', Old Mongolian * $k \phi rehesyn$, is $g \phi r \varepsilon sn$ in Kalmuck and Xinjiang Oirad, but $g \phi r \phi s$ in Alshaa and Hoshuud (Sun et al. 1990; Gereltü 1992).

9.5 DAGUR⁷

Dagur is spoken in China, mainly in the Morin Dawaa area in the eastern part of South Mongolia, and in the Nonni River region to the north-east of Qiqihar in Heilongjiang province. There are also some speakers in the Hailar area (South Mongolia) and the Ili area in Xinjiang. Traditionally, the Dagur are shamanists or Lamaists.

The four areas correspond to four main dialects (Todaeva 1986): the Buthaa dialect (45,000) spoken in Morin Dawaa, the Qiqihar dialect (25,000), the Hailar dialect (5,000), and the Xinjiang dialect (4,000). The dialect differences are relatively

⁷ For sources on Dagur, see App. I, 9.5, p. 228.

large. The Hailar dialect has lost Old Mongolian initial h which is preserved in Buthaa.

During the nineteenth century, some authors used the Manchu alphabet for writing Dagur (Engkebatu 1994*b*; 1996). The Cyrillic alphabet was introduced in 1957, but it was abolished the same year (Kara 1995). A writing system using the Latin alphabet was devised in 1981. It is based on the rather conservative Buthaa dialect. The Dagur normally use Chinese or Mongolian as their written language, however.

The Dagur data used here is taken from Engkebatu (1984; 1988), who describes the Buthaa dialect. The phonemes shown in (9) occur in indigenous words. A feature which is unique for a Mongolic language is the existence of labialized consonant phonemes.

(9) Buthaa Dagur phonemes

i		u			t ^h	t ^{wh}	t ^{jh}	čh	kh	k ^{wh}	k ^{jh}
	Y		р	$\mathbf{p}^{\mathbf{j}}$	t	t ^w	t ^j	č	k	$\mathbf{k}^{\mathbf{w}}$	k ^j
ε	a	Э			s	$\mathbf{S}^{\mathbf{W}}$		š	х	$\mathbf{X}^{\mathbf{W}}$	х ^j
			m	mj	n	\mathbf{n}^{w}	n ^j				
					1	l^w	þ				
					r		r ^j				
			W					j			

Engkebatu writes only three short vowels, ∂ , *i*, *u*, in non-initial syllables. Since Dagur has strong vowel reduction similar to that in Mongolian proper, we interpret these vowels as schwas and write them as ∂ , assuming that their quality is coloured by the preceding consonant. This analysis is possible because there are palatalized and labialized consonant phonemes in Dagur. We assume that non-initial short [i] is a schwa preceded by a palatalized or alveopalatal consonant, or *j*, and that noninitial short [u] is a schwa preceded by a labialized consonant or *w*. For example, we interpret the sequences written by Engkebatu as *di* and *du* as $t^{j}\partial$ and t^w_∂ , respectively. We regard these schwas as non-phonemic, as in Mongolian, and thus analyse Engkebatu's long non-initial vowels as short, so that there is a vowel quantity contrast only in the initial syllable. Since we reserve the symbol ∂ for schwas, we write other occurrences of ∂ in Engkebatu's material as *x*. Dagur has pharyngeal vowel harmony, with the vowels divided into three classes, pharyngeal (*a*, ∂ , ε), non-pharyngeal (*y*), and neutral (*i*, *u*). There is also a restricted form of rounding harmony, so that ∂ is followed by ∂ (and not by *a*) in suffixes.

9.6 SHIRA YUGUR⁸

Shira Yugur (East Yugur; Yellow Uigur) is spoken by some 4,000 people in two small areas on the border between Gansu and Qinghai provinces. The speakers

⁸ Kotwicz (1939); Todaeva (1966; 1975; 1997*f*); Kim Pang-han (1969); Jagunasutu (1981*b*; 1987*a*); Bulucilagu (1982; 1985*b*; 1986; 1989; 1991; 1992); Satô (1987); Erdenicugla (1988); Möngkebuyan (1989); Chuluu (1994*a*); Nugteren (2003).

are Lamaists. The language is not written, and the speakers use Chinese as their written language.

The sources used are Bulucilagu and Jalsan (1992) and Bulucilagu (1985*a*). They describe the Kangle dialect, but the differences between this and the other dialect, Qinglong, are reported to be small. The phonemes shown in (10) occur in indigenous words.

(10) Shira Yugur phonemes

i	у		u	$\mathbf{p}^{\mathbf{h}}$	th	čh	$\mathbf{k}^{\mathbf{h}}$	
			υ	р	t	č	k	
e	ø	Y	0		S	š	х	h
		a	э	m	n		ŋ	
					1			
					r			
				w		j		

There is contrastive vowel length. Vowel harmony is fairly complicated. The vowels can be divided into three classes: pharyngeal (v, a, z), non-pharyngeal (u, e, ϕ, o) , and neutral (i, y, y). There is rounding harmony as well. According to Bulucilagu and Jalsan (1992), there are numerous exceptions, especially to rounding harmony.

9.7 MONGUOR⁹

Monguor (White Mongol; Tu) is spoken mainly in the easternmost part of Qinghai province, with the largest concentration in the counties Huzhu (60,000 speakers) and Minhe (30,000). The dialect differences are quite large, and it has been suggested that Huzhu Monguor (Mongghul) and Minhe Monguor (Mangghuer) are separate languages (Slater 1998; 2003a, b). As far as the phonology of indigenous words is concerned, the differences are not greater than between the dialects of Dagur or Bonan, and for this reason we have preferred to treat the Monguor variants as dialects of one language, and have retained the traditional name form Monguor.

In 1979, a writing system based on the Latin alphabet was devised for the Huzhu dialect. It has been used to some extent in primary education since 1981 (Kakudô 1990c; Jagunasutu 1992).

The sources used for Monguor are Cenggeltei (1991) and Qasbagatur (1986), who describe the Huzhu dialect. The phonemes shown in (11) occur in indigenous words.

⁹ Mostaert and de Smedt (1929–31); Mostaert (1931); Nomura (1959*a*); Róna-Tas (1960*b*; 1962; 1966); Jagunasutu (1964; 1981*a*; 1987*b*); Kim Pang-han (1969); Todaeva (1973; 1997*e*); Doerfer (1974); Saitô (1983; 1993*a*); Mudrak (1986); Xi Yuanlin (1986); Kakudô (1987*b*; 1988; 1990*a*, *b*; 1994; 1997); Cenggeltei (1988; 1997); Kuz'menkov (1988*b*); Li Keyu (1988); Chuluu (1994*d*); Böke (1998; 2001); Li Meiling (2001); Georg (2003*b*). *Minhe dialect*: Jagunasutu and Li (1982); Slater

(11)Huzhu Monguor phonemes p^h th čh kh i u č t k e 0 p a a с š s х m n n 1 r w i

Monguor has lost vowel harmony. There is contrastive vowel length in Huzhu, but it has been lost in Minhe: compare Minhe *ima* 'goat', $t^{h}osi$ 'fat', *xuta* 'bag' with Huzhu *imaa*, $t^{h}oosi$, *xuuta*. Another dialect difference is the loss of some syllable final consonants in Minhe: ter 'gown', qar 'fire', saŋ 'comb' vs. Huzhu teel, qal, sam.

9.8 SANTA¹⁰

Santa (Dongxiang) is spoken by some 270,000 persons in an area to the south-west of Lanzhou in Gansu province. The speakers are Muslims. A Santa script based on the Latin alphabet was developed around 1980 (Schwarz 1982), but has not been used very much. The school education is in Chinese.

The data are taken from Böke (1983*a*; 1986). He describes the Suonanba dialect (120,000 speakers). The other two main dialects are Wangjiaji (110,000) and Sijiaji (40,000); the dialect differences are reported to be small. The phonemes shown in (12) occur in indigenous words. There is no contrastive vowel length, and only remnants of vowel harmony.

(12) Santa phonemes

i	u	u	$\mathbf{p}^{\mathbf{h}}$	t ^h	čh	$\mathbf{k}^{\mathbf{h}}$	$\mathbf{q}^{\mathbf{h}}$
e		0	р	t	č	k	q
	a			S	š	х	
			m	n		ŋ	
				1			
				r			
			W		j		

(1998; 2003a, b).

¹⁰ Todaeva (1959; 1960*b*; 1961; 1997*c*); Nasunbayar (1961; 1981; 1987*a*); Liu (1965; 1981); Bulag (1981; 1988; 1997); Böke (1982; 1983*b*; 1985; 1998; 2001); Kakudô (1982*b*; 1995); Nadamid (1984);

152

9.10 KANGJIA

9.9 BONAN¹¹

Bonan is spoken by 4,000 persons in the Tongren area in the easternmost part of Qinghai province, and by 5,000 in Dahejia township and other parts of Jishishan autonomous county close to Linxia in Gansu province. The dialect differences are reported to be rather large. The speakers of Tongren Bonan are Lamaists who belong to the Tu (Monguor) ethnic group. Their school education is given in Tibetan. The speakers of Jishishan Bonan are Muslims and belong to the Bonan ethnic group. Their school education is given in Chinese. Bonan has no written language.

Our source is Chen (1986*a*; 1987), who describes the Nianduhu variant of the Tongren dialect. The phonemes shown in (13) occur in indigenous words. There is contrastive vowel length in the Tongren dialect, but it has been lost in Jishishan (Chen 1989–90). There is no vowel harmony.

(13) *Bonan phonemes*

i	u	u	$\mathbf{p}^{\mathbf{h}}$	t ^h	čh	$\mathbf{k}^{\mathbf{h}}$		
e		0	р		č		q	
	a			S	š	х		h
			m	n		ŋ		
				1				
				r				
			W		j			

9.10 KANGJIA¹²

The Kangjia language is spoken by about 300 persons in the Kangyang commune in Jainca county in the easternmost part of Qinghai province. This language was first investigated by Wu Chengyi in 1988. It is usually regarded as being close to Bonan, but there are some phonological similarities to Shira Yugur as well. We use the description by Secencogtu (1999). The phonemes shown in (14) occur in indigenous words.

(14) Kangjia phonemes

	00							
i		ш	u	$\mathbf{p}^{\mathbf{h}}$	t ^h	čh	kh	
			υ	p	t	č	k	
e		¥	0		s	š	х	h
		a	э	m	n		ŋ	
					1			
					r			
				W		j		

Ma and Liu (1986); Ibrahim (1988); Chuluu (1994b); Field (1997); Stephen Kim (2003).

¹¹ Todaeva (1963; 1964; 1979; 1997*a*); Böke and Chen (1981); Böke and Liu (1982); C. Li (1986); Nasunbayar (1987*b*); Chen (1989–90; 1994*b*); Chuluu (1994*e*); Böke (1998; 2001); Kögjiltü (2003).

There is no contrasting vowel length in Kangjia. There is a limited form of vowel harmony, where the vowels are divided into three classes, pharyngeal (a, 2, v), non-pharyngeal (e, o, u), and neutral (i, u, v).

9.11 MOGHOL¹³

Moghol is almost extinct, but may still be spoken in a few villages in Herat province in Afghanistan by descendants of troops from the time of the great Mongol conquests. Moghol data are taken from Weiers (1972). The phonemes shown in (15) occur in indigenous words.

(15)	Mogh	ol phonemes

i	u	р	t	č	k	q
e	0				g	G
	а		S	š	Х	
		m	n		ŋ	
			1			
			r			
				j		

According to Ramstedt (1906) and Weiers (1972), the two series of stops/affricates are voiceless ~ voiced in Moghol.

Earlier researchers have noted long vowels in Moghol (Ramstedt 1906; Ligeti 1955; 1964). According to Weiers (1970*a*), these long vowels are not contrastive, and are partly due to stress and partly due to the earlier researchers' expectations of finding long vowels. Consequently, Weiers does not write long vowels, and we follow him. Moghol does not have vowel harmony. Moghol has been written sporadically with the Arabic alphabet; see Weiers (1992–3).

The Hazara of Iran and Afghanistan regard themselves as descendants of Mongol troups. Their Iranian language is usually regarded as a dialect of Dari, but it contains some Mongolic words (Efimov 1965).

9.12 OTHER RELATED LANGUAGES

There are a few languages which have some relationship to the Mongolic languages. They will be mentioned briefly here, but material from them is either irrelevant or not sufficiently known to be very useful in Mongolic historical phonology.

Kitan, the language of the Liao dynasty of north-eastern China (916–1125), is believed to have been Mongolic or related to Mongolic. The Dagur are sometimes regarded as descendants of the Kitan. Although much progress has been made in

¹² Böke (1995; 2001); Secencogtu (2002).

¹³ Pritsak (1964); Böke (1987; 1996; 1997); Kuz'menkov (1997); Weiers (2003).

deciphering the two different scripts used for Kitan (see e.g. Janhunen 2003*e*), the available material is too limited to be very useful in Mongolic historical phonology.

The so-called Benren ('original people') in Shidian county in Yunnan province in south-western China speak a language of the U language group, which belongs to the Angkuic sub-branch of the Palaungic branch of Austroasiatic (Svantesson 1988b). Shidian U has been investigated by Chen (1994*a*; 1995*a*) and by Yan and Zhou (1995: 178ff.). Chen found that part of the vocabulary is similar to Mongolic, in particular Dagur. Furthermore, some fifteenth century inscriptions containing Kitan characters have been found in the area. Chen has drawn the conclusion that the Benren are Kitan descendants. However, the linguistic evidence for this is not entirely convincing, and the question needs further study.

There is a group of about 5,000 people in Tonghai county in Yunnan who regard themselves as ethnic Mongols, but who have assimilated linguistically to their neighbours and now speak the Khatso language, which belongs to the Loloish group of the Tibeto-Burman languages (He 1989; 1998). Another group of about 11,000 ethnic Mongols live in Yanyuan county in Sichuan. They speak the Tibeto-Burman language Naxi (Ouyang and Zhou 1994).

The Wutun language is spoken in Tongren county in Qinghai province by about 2,000 persons who are officially classified within the Tu (Monguor) ethnic group, although they regard themselves as Tibetans. Apparently, Wutun is a Chinese dialect which acquired inflectional morphology due to extensive contact with Tibetan and Mongolic languages. The speakers of Wutun are bilingual in their language and Tibetan. (See Chen 1986*b* and C. Li 1986.)

The Tangwang language is spoken by about 20,000 persons in some ten villages, including Tangjiacun and Wangjiacun, in the north-eastern part of the Dongxiang autonomous county in Gansu province (Ibrahim 1985). The speakers are officially classified as Hui (Chinese Muslims) or Dongxiang (Santa). According to Ibrahim, most of the vocabulary is Chinese, but like Wutun, Tangwang has inflectional morphology, in this case derived from Santa.

9.13 COMPARATIVE VOCABULARY

This section contains a comparative vocabulary of the Mongolic languages with data taken from the sources referred to above. The vocabulary provides examples for the historical development outlined in Chapter 10. We have modified the transcriptions of the sources somewhat so that they agree with our transcription of Mongolian, and we have also tried to eliminate over-differentiation which sometimes occurs in the sources, by changing symbols for allophones to those for the phoneme. All changes we made of the transcriptions in the original sources are shown in Table 9.1.

Except for Halh, Buriad, Kalmuck, and Moghol, the data come from Chinese publications, which use IPA symbols with some modifications. The two series of

original	changed to
ă	ShY v.
æ	Mgr a.
e	Dag ε.
ə	Cha, Baa, Dag, ShY, KJ <i>v</i> ; San <i>e</i> ; Bon <i>w</i> .
е	Mgr <i>i</i> .
3	Mgr e.
¥	Mgr <i>a</i> .
0	Baa, Kmn, Dag <i>ɔ</i> .
θ	Baa, Bon, KJ <i>o</i> .
0	Cha v.
ວ	Mgr o.
ŏ	ShY y.
u	Baa, Kmn, ShY, KJ <i>v</i> .
ŭ	ShY v.
υ	Mgr o.
u	Baa, ShY, KJ <i>u</i> .
v	Kmn <i>u</i> .
y b	Mgr u. Cha Baa Kmp Dag ShY Mar Sap Bap KLp
β	Cha, Baa, Kmn, Dag, ShY, Mgr, San, Bon, KJ <i>p</i> . ShY <i>w</i> .
р С	$\operatorname{Kmn} c^h$.
ç	Bon \check{s} ; ShY hj .
ç	Mgr š; San s.
¢ d	Cha, Baa, Kmn, Dag, ShY, Mgr, San, Bon, KJ <i>t</i> .
dz	Mgr c.
dz	Mgr, Bon \check{c} ; San t .
dz.	San č.
dz	Cha, Baa, Dag, ShY, KJ č.
f	Bon, KJ h ; Mgr, San x .
g	Cha, Baa, Kmn, Dag, ShY, Mgr, San, Bon, KJ k.
ġ	Mog <i>G</i> .
gq	$\operatorname{ShY} k^h$.
G	Mgr, San, Bon q ; ShY, KJ k .
x	Dag, ShY, KJ k.
h	San x.
ħ	Bon h.
j	Kmn c.
k	Kmn, Dag, ShY, Mgr, San, Bon, KJ k ^h .
1	ShY hl.
ņ	ShY hn.
ng	Kmn ŋ.
ŋ	Dag n.
р	ShY, Mgr, San, Bon, KJ p^h .
q	ShY, KJ k^h ; San q^h .
R	ShY, KJ k; San q.

TABLE 9.1	Transcription changes	
	1 0	

s	Kmn h.
ş	Mgr, San, Bon š.
ſ	Cha, Baa, Dag, ShY, KJ š.
t	Cha, Baa, Kmn, Dag, ShY, Mgr, San, Bon, KJ t ^h .
tç	Mgr, Bon \check{c}^h ; San t^h .
tş	San \check{c}^h .
t∫	Cha, Baa, Dag, ShY, KJ č ^h .
υ	Mgr, Bon w.
v	KJ w.
х	Kmn h.
χ	Bon, KJ x.
у	Kmn <i>j</i> .

stops are indicated by the symbols for voiceless and voiced stops, while the texts of the publications describe them as voiceless aspirated and voiceless unaspirated, respectively. When we use these sources, we write the IPA symbols for aspirated and unaspirated voiceless stops instead.

Monguor *c*, *g* and Bonan *ç*, *g* seem to be in complementary distribution, at least in indigenous words. We write them as *š*. We analyse Santa *tc*, *dz*, *c* as allophones of *l*t^h, t, s/ before *l*i/. The Shira Yugur and Kangjia uvulars written *q* and *g* occur only in words with pharyngeal vowels and are analysed as allophones of *l*k^h/ and *l*k/, respectively, by us. We regard the fricatives *x* and *y* as allophones of *l*k/ between voiced sounds in these languages. Similarly, Santa *x* is an allophone of *l*q/, and Dagur *y* is an intervocalic allophone of *l*k/. Monguor *f*, *x*, Santa *f*, *h*, *x*, Bonan *f*, *h*, and Kangjia *f*, *h* do not seem to contrast in indigenous words, *f* occurring only before *u* or other rounded vowels; we regard them as allophones of *l*x/ (Monguor, Santa) or *l*h/ (Bonan, Kangjia). Dagur *y* is an allophone of *l*n/ before velars.

Short vowels in non-initial syllables in the sources for Chahar, Baarin, Buriad, and Dagur are interpreted as non-phonemic schwas, written ∂ , and what the sources write as long vowels in non-initial syllables are written here as short vowels. We use the symbol ∂ exclusively for schwa vowels. Shira Yugur [ă, ŭ, ŏ] are regarded as allophones of /s/. Monguor [æ] and [s] are allophones of /a/, [3] is an allophone of /e/, [9] an allophone of /i/, [υ] and [υ] are allophones of /o/, and [y] is an allophone of /u/. Chahar nasalized vowels are analysed as combinations of vowels and y by us.

In some cases a source gives different forms for the same word, for instance Monguor *ite* \sim *te* 'to eat', with or without an initial vowel. We regard the form with a vowel as the main form and the other one as a reduced form.

OM	'burden' *ač ^h ihan	'mountain' *ahula	'eld. brother' *ak ^h a	'motley' *alak	'which' *ali	'fruit' *alima
Hlh	ač ^h a	იიჭ	ax	alzəg	alz ^j	alʒ ^j əm
	ачаа	уул	ax	алаг	аль	алим
Cha	ač ^h a	បបl	ax	alək	εlj	εl ^j əm
Baa	εč ^h a	បបl	ax	alək	εl	
Bur	ašan	ບບໄຈ	axə	aləg	al ^j ən	al ^j mə
Kmn		oola	ak ^h a			
Klm		uul	ax	alg	al ^j	al ^j mn
Dag	at ^h e	aul	ak	alar		al ^j əm
ShY	hčan	uula	aka	alak	aalv	alma
Mgr	ščaa	ula	aqa	alaq	ali	alima
San	ač ^h aŋ	ula	aqa		ali	alima
Bon	ač ^h aŋ	uula	aqa		anu	alma
KJ	ačho	ula	aka	alvk	ani	alima
Mog		aulo				

	'gold'	'mouth'	'life'	'to take'	'clean'	'method'
OM	*alt ^h an	*aman	*amin	*ap	*arihun	*arka
Hlh	alzth	am	am ^j	aw	ar ^j uŋ	arəg
	алт	ам	амь	ав	ариун	арга
Cha	alt ^h	am	εm ^j	ap	eruŋ	arək
Baa	alt ^h	am	εm	ap	arun	arək
Bur	alt ^h ən	amən	am ^j ən	abə	ar ^j un	argə
Kmn	alt ^h a			ap		
Klm	alt ^h n	amn	εmn	aw	eryn	arg
Dag	alt ^h	am	am ^j	au	arun	arək
ShY	alt ^h an	aman	amyn	ap	aroon	arka
Mgr	xaltan	ama	amin	awu	arin	
San	ant ^h aŋ	amaŋ	amiŋ	aki	aruŋ	
Bon	alt ^h aŋ	amaŋ		ap	aron	
KJ	antho	amo	amin	api	arun	
Mog	altan	aman		ob	orun	

'to bite' *čahu	'hundred *čahun	' 'young' *čalahu	'to swallov *čalki	w' 'to u *čaru	1	
cuu	cuu	cabu	cakjgj	car	caa	
зуу	зуу	залуу	залги	зар	заа	
čυυ	čuu	čalu	čɛŀk	čar	čaa	
čυυ		čalu	čɛlk	čar	čaa	
Ζυυ	zuun	zalu	zalg ^j ə	zara) zaa	
coo	c55					
zu	zun	zalu	zal ⁱ g	zar	za	
čau	čau	čalo	čɛlk ^j	čar	čaa	
čυυ	čuun	čaluu	čalk ^h v	čar	čaa	
čau	čoŋ	čaliu	č ^h alke	čari	čaa	
čau		čalau	čaŋq ^h ei	čaru	ı čala	
ču	nčoŋ		čalqa	čar	čaa	
čiu	čun	čalu	čalke	čari	čia	
jau		jalau		jo ru		
'fish'	'year'	'six'	'heart'	'spear'	'to fit'	
*čikasun	*čil	*čirkohan	*čiryken	*čita	*čok ^h i	
cagəs	čilz	curga	curx	čat	cəx ^j	
загас	жил	зургаа	зүрх	жад	зохь	
čakəs	čil	čurka	čurx	čıt		
čaks	čil		čurx	čɛt	čœx	
zagəhən	žel	zorgan	zurxən	žadə	zəx ^j ə	
cakahu	cil	corkaa		cita		
zagsn	jil	zurgan	zyrk ^h n	jid	zok ^h	
čaus	čil	čirko	čurk ^w		čuk ^j	
čakasvn	čyl	čirkuun	čyrken	čuta	čoky	
čiqasi	čil	čirqoon	čirke	čitaa	č ^h uku	
čaqasuŋ	čeŋ	čiqoŋ	čuqe	čita		
Xalaacon	čilm	čiraon	čirke		čoqui	
carqasoij		enquij				
čikasun	čile jil	čirkuŋ	čirke	čita	čuku	
	*čahu cuu syy čuu čuu coo zu čau čuu čau ču čau ču čiu jau 'fish' *čikasun cagəs sarac čakəs čakəs čakəs čakəs čaks zagəhən cakahu zagsn čaus čaus čaus čaus čaus čausuj	*čahun соо соо зуу зуу čоо čоо čоо čоо čоо соо čоо čоо čоо čоо čau čoo šarac škuл čakəs čil čakaso čil čaus čil čaus čil čaus čil čaus čil čau čol<	*čahu*čahun*čalahu CUU CUU $calyU$ $3yy$ $3yy$ $3aлyy$ $čUU$ $čUU$ $čalU$ $čUU$ $ZUUN$ $zalU$ ZUU $ZUUN$ $zalU$ COO COO $CalU$ ZUU $ZUUN$ $zalU$ COO COO $CalU$ COO COO $CalU$ COO COO $CalOO$ COO COO $CalOO$ COU $COON$ $CalOO$ CaU $COON$ $CalOO$ CaU $COON$ $CalOO$ CaU $COON$ $CalOO$ CaU $COON$ $CalOO$ Jau $COON$ $CalOO$ Jau $TOON$ $TOON$ $CaGOS$ $CilS$ $CUTGA$ $3araC$ $XUIN$ $3ypraa$ $CakS$ Cil $CUTGA$ $SaraC$ $XIIN$ $3ypraa$ $CakAhU$ Cil $CUTGA$ $ZagOhON$ Zel $ZUTGAN$ $CakAhU$ Cil $Curgan$ $CakAhU$ Cil $Curgan$ $CakAS$ Cil $CirkOON$ $CalS$ Cil $CirkOUN$	*čahu*čalahu*čalki $c u u$ $c u v$ $c a k u$ $c a k j g j$ $3 y y$ $3 y y$ $3 a \pi y$ $3 a \pi \pi$ $c u u$ $c u v$ $c a k u$ $c a l v$ $c a l x j g j$ $c u u$ $c u u$ $c u v$ $c a l u$ $c e l k$ $c u u$ $z u u$ $z a l u$ $z a l g j a$ $c u 0$ $z u u$ $z a l u$ $z a l g j a$ $c u 0$ $z u u$ $z a l u$ $z a l g j a$ $c u 0$ $z u u$ $z a l u$ $z a l j g$ $c a u$ $c a u$ $c a l u$ $c a l k h y$ $c a u$ $c u 0$ $c a l u$ $c a l k h y$ $c a u$ $c u 0$ $c a l u$ $c a l q a$ $c a u$ $c u 0$ $c a l u$ $c a l q a$ $c a u$ $c u 0$ $c a l u$ $c a l q a$ $c a u$ $c u 0$ $c a l u$ $c a l q a$ $c a u$ $c u n c o \eta$ $c a l u$ $c a l q a$ $c a u$ $c u n c o \eta$ $c a l u$ $c a l q a$ $d u$ $d $	*čahu*čahu*čalahu*čalki*čar CUU CUU $Calzu$ $calz^j g^j$ Car $3yy$ $3yy$ $3a_{JTY}$ $3a_{JTT}$ $3ap$ $\dot{C}UU$ $\dot{C}UU$ \dot{CalU} $\dot{Cel^jk}$ \dot{Car} $\dot{C}UU$ ZUU \ddot{CalU} $\ddot{Cel^jk}$ \ddot{Car} $\ddot{C}UU$ $ZUUn$ $Zalu$ $Zalg^jg$ $Zara$ $\dot{C}UU$ $ZUUn$ $Zalu$ $Zalg^jg$ $Zara$ $\dot{C}OU$ $ZUUn$ $Zalu$ $Zalg^jg$ $Zara$ $\dot{C}OU$ $\ddot{C}OU$ \ddot{CalU} $\ddot{Cel^jk}$ \ddot{Car} $\ddot{C}uu$ \ddot{Cun} \ddot{CalUU} $\ddot{Carlakh^y}$ \ddot{Car} \ddot{Cuu} \ddot{Con} \ddot{CalUU} $\ddot{Cardh^hei}$ \ddot{Caru} \ddot{Cuu} \ddot{Con} \ddot{CalU} $\ddot{Cardh^hei}$ \ddot{Caru} \ddot{Cuu} \ddot{Con} \ddot{CalU} $\ddot{Cardh^hei}$ \ddot{Caru} \ddot{Cuu} \ddot{Con} \ddot{CalU} \ddot{Caru} \ddot{Caru} $\dot{fish'}$ 'year''six''heart''spear''fish''year''six''heart''spear''fish''year''six''heart''stear'fish''year''six''heart''spear''fish''year''six''heart''spear'<	
OM	'soft' *čøhelen	ʻright' *čøp	'paper' *č ^h ahalsun	'time' *č ^h ak	'white' *č ^h akahan	'snow' *č ^h asun
---	--	--	---	---	---	--
Hlh	cookəŋ	cow	c ^h aas	chag	c ^h agaŋ	chas
	зөөлөн	зөв	цаас	цаг	цагаан	цас
Cha	čooləŋ		č ^h aas	č ^h ak	č ^h akaŋ	čas
Baa	čoolən	čop	č ^h aas	č ^h ak	č ^h akan	č ^h as
Bur	zoolən	zub	saarhən	sag	sagan	sahən
Kmn			c ^h aarhu	c ^h ak	c ^h akaan	c ^h ahu
Klm	jøølən	zøw	c ^h aasn	chag	c ^h agan	chasn
Dag	čvulvn	čuu	č ^h aas	č ^h ɛk ^j	č ^h ikan	č ^h as
ShY	čylen	čøp	č ^h aasv	č ^h ak	č ^h vkaan	časvn
Mgr	čoolon	čup	č ^h aalci	č ^h aq	č ^h iqaan	č ^h asi
San	čolien	čo		čha	č ^h iqaŋ	čaŋsuŋ
Bon	čulaŋ				č ^h ixaŋ	časoŋ
KJ	čilo			č ^h uk ^h	čhixo	č ^h asun
Mog					čagon	čosun
	'flower'	'soldier	' 'ear'	'stone'	'to boil'	'wolf'
OM	'flower' *č ^h eč ^h ek	'soldier *č ^h erik	'' 'ear' *č ^h ik ^h in	'stone' *č ^h ilahu		'wolf' *č ^h ino
OM Hlh						
Hlh	*č ^h eč ^h ek c ^h ic ^h əg цэцэг	*č ^h erik c ^h irəg цэрэг	*č ^h ik ^h in č ^h ix чих	*č ^h ilahu č ^h uђu чулуу	ın *č ^h ina	[*] č ^h ino č ^h эn чоно
	*č ^h eč ^h ek c ^h ic ^h əg	*č ^h erik c ^h irəg	*č ^h ik ^h in č ^h ix чих čix	*č ^h ilahu č ^h u뮟u чулуу č ^h ulu	n <u>*č^hina</u> č ^h an	*č ^h ino č ^h ən
Hlh Cha Baa	*č ^h eč ^h ek c ^h ic ^h əg цэцэг	*č ^h erik c ^h irəg цэрэг	*č ^h ik ^h in č ^h ix чих	*č ^h ilahu č ^h uђu чулуу	in [*] č ^h ina č ^h an чана	[*] č ^h ino č ^h эn чоно
Hlh Cha	*č ^h eč ^h ek c ^h ic ^h əg цэцэг čič ^h ək č ^h ič ^h ək sesəg	*č ^h erik c ^h irəg цэрэг č ^h irək	*č ^h ik ^h in č ^h ix чих čix č ^h ix šexən	*č ^h ilahu č ^h u뮟u чулуу č ^h ulu	in [*] č ^h ina č ^h an чана č ^h an	*č ^h ino č ^h ɔn чоно č ^h ɔn č ^h ɔn šɔnə
Hlh Cha Baa Bur Kmn	*č ^h eč ^h ek c ^h ic ^h əg цэцэг čič ^h ək č ^h ič ^h ək sesəg c ^h ec ^h ek	*č ^h erik c ^h irəg цэрэг č ^h irək č ^h irək serəg	*č ^h ik ^h in č ^h ix чих čix č ^h ix šexən c ^h ik ^h i	*č ^h ilahu č ^h ułzu чулуу č ^h ulu č ^h ulu šulun c ^h iloo	in *č ^h ina č ^h an чана č ^h an č ^h an šanə	*č ^h ino č ^h эn č ^h эn č ^h эn šənə c ^h эnэ
Hlh Cha Baa Bur Kmn Klm	*č ^h eč ^h ek c ^h ic ^h əg цэцэг čič ^h ək č ^h ič ^h ək sesəg	*č ^h erik c ^h irəg цэрэг č ^h irək č ^h irək	*č ^h ik ^h in č ^h ix чих čix č ^h ix šexən c ^h ik ^h i č ^h ik ^h n	*č ^h ilahu č ^h olgo чулуу č ^h olo č ^h olo šolon c ^h iloo č ^h olun	<u>in</u> *č ^h ina č ^h an чана č ^h an č ^h an	*č ^h ino č ^h ɔn чоно č ^h ɔn č ^h ɔn šɔnə
Hlh Cha Baa Bur Kmn	*č ^h eč ^h ek c ^h ic ^h əg цэцэг čič ^h ək č ^h ič ^h ək sesəg c ^h ec ^h ek	*č ^h erik c ^h irəg цэрэг č ^h irək č ^h irək serəg c ^h erg	*č ^h ik ^h in č ^h ix чих čix č ^h ix šeхәn c ^h ik ^h i č ^h ik ^h n č ^h ik ^{jh}	*č ^h ilahu č ^h uţzu чулуу č ^h ulu č ^h ulu šulun c ^h iləə č ^h olun č ^h olun	in *č ^h ina č ^h an чана č ^h an šanə č ^h an šanə	*č ^h ino č ^h on č ^h on č ^h on c ^h ono č ^h ono
Hlh Cha Baa Bur Kmn Klm Dag ShY	*č ^h eč ^h ek c ^h ic ^h əg цэцэг čič ^h ək č ^h ič ^h ək sesəg c ^h ec ^h ek	*č ^h erik c ^h irəg uəpər č ^h irək č ^h irək serəg c ^h erg č ^h eryk	*č ^h ik ^h in č ^h ix чих čix č ^h ix šexən c ^h ik ^h i č ^h ik ⁱⁿ č ^h ik ⁱⁿ	*č ^h ilahu č ^h olgo чулуу č ^h olo č ^h olo šolon c ^h iloo č ^h olun	in *č ^h ina č ^h an чана č ^h an č ^h an šanə č ^h an šanə č ^h an	*č ^h ino č ^h эn č ^h эn č ^h эn šənə c ^h эnэ
Hlh Cha Baa Bur Kmn Klm Dag	*č ^h eč ^h ek c ^h ic ^h əg цэцэг čič ^h ək č ^h ič ^h ək sesəg c ^h ec ^h ek c ^h ec ^h g	*č ^h erik c ^h irəg цэрэг č ^h irək č ^h irək serəg c ^h erg	*č ^h ik ^h in č ^h ix чих čix č ^h ix šexən c ^h ik ^h i č ^h ik ^{jh} č ^h ik ^{jh} č ^h k ^h vn č ^h iki	*č ^h ilahu č ^h uţzu чулуу č ^h ulu č ^h ulu šulun c ^h iləə č ^h olun č ^h olun	in *č ^h ina č ^h an vaHa č ^h an č ^h an šanə č ^h an šanə č ^h na č ^h na č ^h na	*č ^h ino č ^h on č ^h on č ^h on c ^h ono č ^h ono
Hlh Cha Baa Bur Kmn Klm Dag ShY	*č ^h eč ^h ek c ^h ic ^h əg цэцэг čič ^h ək č ^h ič ^h ək sesəg c ^h ec ^h ek	*č ^h erik c ^h irəg uəpər č ^h irək č ^h irək serəg c ^h erg č ^h eryk č ^h iraq	*č ^h ik ^h in č ^h ix uux čix č ^h ix šexən c ^h ik ^h i č ^h ik ^h n č ^h ik ^{jh} č ^h k ^h vn č ^h iki č ^h iqeŋ	*č ^h ilahu č ^h uţzu чулуу č ^h ulu č ^h ulu šulun c ^h iləə č ^h olun č ^h olun	in *č ^h ina č ^h an vaHa č ^h an šanə č ^h an šanə č ^h na č ^h na č ^h na č ^h inaa č ^h ina	*č ^h ino č ^h on č ^h on šonə c ^h ono č ^h on
Hlh Cha Baa Bur Kmn Klm Dag ShY Mgr	*č ^h eč ^h ek c ^h ic ^h əg цэцэг čič ^h ək č ^h ič ^h ək sesəg c ^h ec ^h ek c ^h ec ^h g	*č ^h erik c ^h irəg uəpər č ^h irək č ^h irək serəg c ^h erg č ^h eryk	$\begin{tabular}{lllllllllllllllllllllllllllllllllll$	*č ^h ilahu č ^h utzu yynyyy č ^h olu č ^h olu šolon c ^h iloo č ^h olun č ^h olo č ^h olu	n *č ^h ina č ^h an vaHa č ^h an šanə č ^h an šanə č ^h na č ^h ina č ^h ina č ^h ina	*č ^h ino č ^h on č ^h on č ^h on c ^h ono č ^h ono
Hlh Cha Baa Bur Kmn Klm Dag ShY Mgr San	*č ^h eč ^h ek c ^h ic ^h əg цэцэг čič ^h ək č ^h ič ^h ək sesəg c ^h ec ^h ek c ^h ec ^h g	*č ^h erik c ^h irəg uəpər č ^h irək č ^h irək serəg c ^h erg č ^h eryk č ^h iraq	*č ^h ik ^h in č ^h ix uux čix č ^h ix šexən c ^h ik ^h i č ^h ik ^h n č ^h ik ^{jh} č ^h k ^h vn č ^h iki č ^h iqeŋ	*č ^h ilahu č ^h uţzu чулуу č ^h ulu č ^h ulu šulun c ^h iləə č ^h olun č ^h olun	in *č ^h ina č ^h an vaHa č ^h an šanə č ^h an šanə č ^h na č ^h na č ^h na č ^h inaa č ^h ina	*č ^h ino č ^h on č ^h on šonə c ^h ono č ^h on
Hlh Cha Baa Bur Kmn Klm Dag ShY Mgr San Bon	*č ^h eč ^h ek c ^h ic ^h əg uəuər čič ^h ək č ^h ič ^h ək sesəg c ^h ec ^h ek c ^h ec ^h g	*č ^h erik c ^h irəg uəpər č ^h irək č ^h irək serəg c ^h erg č ^h eryk č ^h iraq č ^h eruk	$\begin{tabular}{lllllllllllllllllllllllllllllllllll$	*č ^h ilahu č ^h utzu yynyyy č ^h olu č ^h olu šolon c ^h iloo č ^h olun č ^h olo č ^h olu	n *č ^h ina č ^h an vaHa č ^h an šanə č ^h an šanə č ^h na č ^h ina č ^h ina č ^h ina	*č ^h ino č ^h on č ^h on šonə c ^h ono č ^h on

OM	'blood' *č ^h isun	'devil' *č ^h itk ^h ør	'father' *eč ^h ike	'door' *ehyten	'saddle' *emehel	'to wear' *emys
Hlh	c ^h us	č ^h ot ^h gər	ichəg	uut	imeţ	oms
	цус	чөтгөр	эцэг	үүд	эмээл	өмс
Cha	čus	-		uut	ymyl	oms
Baa	č ^h us	č ^h yt ^h xər		uut	ymyl	oms
Bur	šuhən	šudxər	esəgə	uudən	emel	umdə
Kmn		chitkhur	U U		emeel	umut
Klm	c ^h usn	č ^h øt ^h k ^h r	echkh	yydn	emɛl	øms
Dag	čhos		vč ^h ək	vut	ymyl	vms
ShY	č ^h ysvn		čhke	yten	emel	mys
Mgr	č ^h isi			ute	imel	mosi
San	č ^h usuŋ			uitien		misu
Bon	čhison			ntaŋ		muiš
KJ	č ^h isun			itə	mila	mysu
Mog	čisu			oidan	jamal	umus
	'this' '	orass' 'l	eamino'	'early'	'te n '	'ashamed'

	'this'	'grass'	'learning'	'early'	'ten'	'ashamed'
OM	*ene	*epesyn	*ertem	*ert ^h e	*harpan	*hič ^h e
Hlh	in	ows	irtəm	irt ^h	arəw	ič ^h
	энэ	өвс	эрдэм	эрт	арав	ИЧ
Cha	vп	ops	vrtəm	vrth	arəp	ič ^h
Baa	vп	ops	vrtəm	vrt ^h	arəp	ič ^h
Bur	enə	ubhən	erdəm	erthə	arbən	ešə
Kmn	ene			erthe	arpa	
Klm	en	øwsn	erdm	erth	arwn	ič ^h
Dag	vnэ	vus	vrtəm	vrt	xarpə	xič ^h
ShY	ene	wesyn	vrtem	hvrt ^h e	harwan	hč ^h e
Mgr	ne	wesi	rtem	šte	xaran	ščee
San	ene	osuŋ		ethie	xaroŋ	šiče
Bon	enu	epsoŋ		erthe	harwaŋ	šče
KJ	env	weisun		ethe	haro	šči
Mog	ena	ebasun				

OM	'bottom' *hiruhar	'blessing' *hiryher	'forest' *hoi	'year' *hon	'bag' *huhut ^h a	'to tie' *huja
Hlh	jərəh	jorotz	oi	ວŋ	uut ^h	υj
	ёроол	ерөөл	ой	он	уут	уя
Cha		jorol	œœ	эŋ	uuth	υj
Baa	jərəl		œœ	on	uut ^h	υj
Bur		jurol	oi	on	บบt ^h ə	υjə
Kmn			oi	on		
Klm	joral	jørel	ø	on	uut ^h	uj
Dag		xiryl		xəən		xuja
ShY	hroor	hørøør		hən	uut ^h a	hjaa
Mgr			xii	xon	xuuta	xujaa
San	siro			xoŋ	xuta	xeija
Bon			hii	hoŋ	hta	U
KJ	hulur		he	hən	p ^h uta	huja
Mog					•	5

	'red'	'ox'	'fox'	'ash'	'to separate'	'goat'
OM	*hulahan	*hyk ^h er	*hyneken	*hynesyn	*ilka	*imahan
Hlh	uban	uxər	unəg	uns	jatzəg	jama
	улаан	үхэр	үнэг	үнс	ялга	ямаа
Cha	ulaŋ	uxər	unək	uns	ılk	jama
Baa	ulan	uxər	unək	uns	jalk	jama
Bur	ulan	uxər	unəgən	unəhən	ilgə	jaman
Kmn	ulaan	ukher	uneke			imaa
Klm	ulan	yk ^h r	yngn	ymsn	jilg	jaman
Dag	xulan	xuk ^{wh} ər	xun ^w ək ^w	xuns	jalək	ima
ShY	hlaan	hkor	høneken	henesyn	alka	maan
Mgr	xulaan	xukor	xunike	xuneesi	laqaa	imaa
San	xulaŋ	xukie	xunieqaŋ	xuniesuŋ	iŋqa	imaŋ
Bon	hulaŋ	okhor	_	homsu		jimaŋ
KJ	hulo		huniko			imo
Mog	ulam	ukar		unasun		

	'to laugh	' 'to come	' 'to eat'	' 'to go'	'bone'	'big'
OM	*inehe	*ire	*ite	*japu	*jasun	*jek ^h e
Hlh	ine	ir	it	jaw	jas	ix
	инээ	ир	ид	яв	яс	ИХ
Cha	yn ^j y	ir	it	jap	jas	ix
Baa	inv	ir	it	jap	jas	jix
Bur	en ^j e	jerə	ed ^j ə	jabə	jahən	jexə
Kmn		ire	ite	japu		jek ^h e
Klm	inε	ir	it	jow	jasn	ik ^h
Dag	xinyt	ir	it	jau	jas	xik
ShY	hnii	ere	ete	jawy	jasvn	šik ^h e
Mgr	šinee	ire	ite	jau	jasi	ške
San	sinie	ire	itie	jawu	jasuŋ	xukie
Bon	šine	er	nte	jawu	jasoŋ	ško
KJ	šine	re	ite	jυ	jasun	
Mog	ina	ira	ida	jobu	josu-	
	'rule'	'ground'	'single'	ʻpig'	'fire'	'hand'
OM	*josun	*kačar	*kakč ^h a	*kak ^h ai	*kal	*kar
Hlh	jəs	Gacər	Ganch	Gaxai	Galz	Gar
	ëc	газар	ганц	гахай	гал	гар
Cha	jəs	kačər	kanch	kaxe	kal	kar
Baa	jəs	kačər	kanc ^h	kaxa	kal	kar
Bur	jəhən	gazər	gansa	gaxai	gal	gar
Kmn	-	kacar	-	kak ^h ai	kal	kar
Klm	josn	Gazr	Ganch	Gaxa	Gal	Gar
Dag	jəs	kačər	kanč ^h	kakə	kal ^j	kar ^j
ShY	jəsə	kačar	k ^h akč ^h a	kak ^h ai	kal	kar
Mgr		qačar		xqai	qal	qar
San		qača		q ^h uqei	q ^h aŋ	q ^h a
Bon		qačar		qaquui	xal	xar
KJ		kačar		kykai	xar	xar
Mag		anton			ao1	a0#

GOl

Gar

Mog

Gajar

THE MONGOLIC LANGUAGES

OM	'bitter' *kasihun	'fault' *kem	'house' *ker	ʻlight' *kerel	'beast' *kørehesyn
Hlh	Gašuŋ	gim	gir	girəz	goros
	гашуун	гэм	гэр	гэрэл	гөрөөс
Cha	kašuŋ	kym	kvr	kvrəl	koros
Baa	kašun	kym	kyr	kvrl	koros
Bur	gašun	gem	ger	gerəl	gurohən
Kmn			ker		kureehu
Klm	Gašun	gem	ger	gerl	gøresn
Dag	kasun	kym	kyr ^j		kurvs
ShY	kašuun		ker	kerel	kureesyn
Mgr	xašin		ker	kireel	
San	q ^h wšuŋ	kien	kie	kieran	
Bon	qašoŋ	kem	ker		
KJ	kaši-	kan	ker		
Mog	qošun		ger		

	'three'	'to shut'	'door'	'scissors'	'to leave'	'warm'
OM	*kurpan	*k ^h aha	*k ^h ahalka	*k ^h aič ^h i	*k ^h akač ^h a	*k ^h alahun
Hlh	GU r əw	xaa	xaabəg	xaič ^h	xagəch	xalzuŋ
	гурав	xaa	хаалга	хайч	хагац	халуун
Cha	korəp	xaa		xeeč ^h		xaluŋ
Baa	korəp	xaa	xaalək	xeeč ^h		xalon
Bur	gurbən	xaa	xaalgə	xaišə	xagasa	xalun
Kmn	korpa					k ^h aləən
Klm	Gurwn	xa	xaalG	xeeč ^h	xagch	xalun
Dag	k ^w arpə	xaa	xaalək	k ^h aič ^h		xalun
ShY	korwan	xaa	xaalka	xaič ^h v	xakač ^h a	xaluun
Mgr	qoraan	xaa	xaalqa	xaič ^h i		xaloŋ
San	quraŋ	q ^h a		q ^h aič ^h i	qaqač ^h a	q ^h aluŋ
Bon	quraŋ	xaa		xiič ^h i		xoloŋ
KJ	kuro	xa		k ^h aič ^h i		xulə
Mog	qurban					qalon

OM	'together' *k ^h amt ^h u	'black' *k ^h ara	'to return' *k ^h ari	ʻold' *k ^h auč ^h in	ʻlaw' *k ^h auli	'when' *k ^h ečije
Hlh	xamt ^h	xar	xar ^j	xບບč ^h əŋ	xυυβ ^j	xice
	хамт	xap	харь	хуучин	хууль	хэзээ
Cha	xamt ^h	xar	xer ^j	xuuč ^h əŋ		хүčү
Baa	xamt ^h	xar	xer	xuučhən	xyyl	хүčү
Bur	xamt ^h ə	xarə	xar ^j ə	xuušən	xuuliə	xeze
Kmn	k ^h amt ^h u	k ^h ara				k ^h ecee
Klm	xamt ^h	xar	xer	xuuč ^h n	xuul ^j	k ^h eze
Dag		xar	xar ^j	k ^h auč ^h in	k ^h əəl ^j	хуčу
ShY	xamty	xara	xarv	xuuč ^h yn		k ^h ečee
Mgr	xamti	xara	xari	xaučin		khičee
San	xant ^h u	q ^h ara	q ^h ari	q ^h uaič ^h eŋ		kieče
Bon	hamtu	xara	haa	xiič ^h aŋ		k ^h uč ^h i
KJ		xara	xari	xuaič ^h o		kyče
Mog		qaro	qari-			keja
	'wind' 'tongu	e' 'who	' 'how many	' 'border'	'to stri	ve'

OM	*wind *k ^h ei	*tongue' *k ^h elen	*who' *k ^h en	*how many *k ^h etyn	*border *k ^h ičahar	*to strive *k ^h ič ^h ihe
Hlh	xii	xib	xiŋ	xit	x ^j acGar	xič ^h e
	хий	хэл	хэн	хэд	хязгаар	хичээ
Cha	xii	xyl	ххŋ	xyt		kič ^h v
Baa	xii	xyl	xyn	xyt	xɛčkar	xxč ^h y
Bur	x ^j ii	xelən	xen	xedən	x ^j izar	xeše
Kmn		k ^h ele	k ^h en			
Klm	khi	k ^h eln	k ^h en	k ^h edy	k ^h izɛr	
Dag	xvin	xyl ^j	xyn	xyt	k ^h ičar	k ^h ič ^h v
ShY	k ^h ii	k ^h elen	k ^h en	k ^h eten	kvčaar	
Mgr	k ^h ii	k ^h ile	k ^h en	k ^h iti	kičaar	
San	k ^h ai	k ^h ielien	k ^h ien	kietuŋ	quuča	
Bon	k ^h ii	k ^h elaŋ	k ^h aŋ	k ^h utoŋ		
KJ	k ^h e	k ^h iliɔ	k ^h ə	kuto		
Mog	kei	kelan	ken	kedu		

хотs хумс хотs хотэs х ^ј отhэn k ^h imoho хитsn	x ^j at ^h ət Хятад kıt ^h at k ^h it ^h at k ^h it ^h d	xəəsəŋ xoocoh xəəsəŋ xəəsən xəəhən xoosn	xəin xoйнo xœœn xœœn xəinə xøøn	xojər xoëp xojər xojər xojər k ^h oir xojr
xums xəməs x ^j umhən k ^h imuhu xumsn	kıt ^h at k ^h it ^h at	xəəsən xəəsən xəəhən	xœœn xœœn xəinə	xəjər xəjər xəjər k ^h əir
xəməs x ^j umhən k ^h imuhu · xumsn	k ^h it ^h at	xəəsən xəəhən	xœœn xoinə	xəjər xəjər k ^h əir
x ^j umhən k ^h imuhu xumsn		xəəhən	xəinə	xəjər k ^h əir
k ^h imuhu xumsn				k ^h əir
xumsn		xoosn	xøøn	
	k ^h it ^h d	xoosn	xøøn	voir
1. h. vh				лојг
k ^h imč ^h		xəəs ^w ən	x ^w ainə	xəir
xymysyn	k ^h ʊt ^h at	XUUSYN	xəinə	kuur
- č ^h imsi	č ^h itar	xoosin	xoino	qoor
quimusun	q ^h utei	q ^h usuŋ	q ^h uina	qua
1 0	hti	1 5	xiina	quar
kemesun	xt ^h e		kuaina	kuar
		qosun	qauna	qijar
	qomsoŋ	qomson hti	qomson hti kemesun xt ^h e	qomson hti xiina kemesun xt ^h e kuaina

OM	*k ^h ola	*sneep *k ^h onin	*k ^h øisyn	*k ^h øit ^h en	*k ^h øk ^h e	*k ^h øl
Hlh	xəţ	xən ^j	xuis	xuit ^h əŋ	хох	хођ
	хол	хонь	хүйс	хүйтэн	хөх	хөл
Cha	xəl	xœn ^j	xiis	xiit ^h əŋ	kox	xol
Baa	xəl	xœn	xyys	xyyt ^h n	XOX	xol
Bur	xələ	xən ^j ən	xuihən	xuit ^h ən	xuxə	xul
Kmn		k ^h əni	k ^h uihu		k ^h uk ^h u	k ^h ul
Klm	xol	xøn	k ^h iisn	k ^h iit ^h n	k ^h øk ^h	k ^h øl
Dag	xəl	xən ^j	k ^h uis	k ^h uit ^{wh} ən	k ^h uk ^{wh}	k ^h ul ^j
ShY	xələ	xoony	k ^h yysyn	k ^h yt ^h en	hk ^h ø	k ^h øl
Mgr	xolo	xoni	k ^h uici	k ^h uiten	k ^h uko	k ^h ol
San	qolo	qoni	k ^h uaisuŋ	k ^h uit ^h ien	k ^h ukie	k ^h uan
Bon	xolo	qoni	k ^h iisoŋ	k ^h it ^h aŋ	k ^h uko	k ^h ol
KJ	kolo	xoni	-	k ^h uit ^h o	k ^h uk ^h u	k ^h or
Mog	qolo	qonin			koka	kol

OM	'sweat' *k ^h ølesyn	ʻlight' *k ^h øŋken	'to bark' *k ^h uč ^h a	'sheath' *k ^h ui	'thief' *k ^h ulakai	'destiny' *k ^h upi
Hlh	xolzs	xoŋgəŋ	xuch	xui	xulzgai	xuw ^j
	хөлс	хөнгөн	хуц	хуй	хулгай	хувь
Cha	xols	xoŋkəŋ	-	XYI	xulke	хүр ^j
Baa	xols	xoŋŋən	xuč ^h	XYY	xulke	xœp
Bur	xulhən	xungən	xusə	xui	xulgai	xub ^j ə
Kmn		e			C	
Klm	k ^h ølsn		xuc ^h	ху	xulxa	xyw
Dag	xuls	xunkyn	khočh	•	x ^w alək	xəp ^j
ShŸ	k ^h ulesvn	k ^h øŋken	k ^h uč ^h a	xui	xulykai-	xuw
Mgr	k ^h onorsi	k ^h onkon	xoča	xoi	xolqai	
San	k ^h oliesuŋ	koŋkien	q ^h uča		quqi	
Bon	k ^h olsoŋ	k ^h oŋk ^h aŋ	hča		qolqui	
KJ	k ^h ulesu	kuŋkə	k ^h uč ^h a		kulkai	
Mog		·			qolagei	
	'finger'	'heavy'	'to reach'	'son-in-la	aw' 'foreh	ead' 'very'
OM	*k ^h uruhun	*k ^h ynty	*k ^h yr	*k ^h yreke		2
Hlh	xuru	xunt	xur	xurgəŋ	magn	ai maš
	хуруу	хүнд	хүр	хүргэн	магн	
Cha	xuru	xunt	xur	xurkəŋ	makn	ε
Baa	xuru	xunt	xur	xurkən	meŋn	
Bur	xurgən	xundə	xurə	xur ^j gən	magn	
Kmn	k ^h uruu		k ^h ur	k ^h uriken		
Klm	xurgn	k ^h ynd	k ^h yr	k ^h yrgn	maŋn	a maš
Dag	xərə	xunt	k ^h ur	xurk ^w ən	mank	
ShY	xuruun	k ^h uŋt ^h y	k ^h ur	k ^h urken	maŋli	
Mgr	xori	k ^h untun	k ^h uri	k ^h urkeer	-	
San	quru	kuntu	k ^h uru	q ^h uqoŋ	maŋle	
Bon	quru	k ^h unt ^h u	k ^h ur	k ^h urqaŋ		
KJ	koro	kuntu	k ^h ur	k ^h urkə		
Mog	qurun	kundu	kur			
	7					

THE MONGOLIC LANGUAGES

OM	'to know' *mete	'meat' *mik ^h an	ʻthous *miŋk		nake' nokai	'horse *mori	
Hlh	mit	max	m ^j aŋo		1363i	mər ^j	mər ^j t
11111	инт мэд	мах	мянга		югой	морь	
Cha	myt	max	mmk		nokæ	mœr ^j	-
Baa	myt	max	menn		ioka	mær	mar or
Bur	medə	m ^j axən	m ^j ang		nogoi	mor ^j a	ən mər ^j əd
Kmn	mete	mik ^h a	miŋk		10501	mori	in mor ou
Klm	med	maxn	miŋg		10Ga	mørn	mørd
Dag	myt	m ^j ak	m ^j anl		106u 10k ^w	mər ^j	ing)i a
ShY	mete	mak ^h an	mvŋk		iokoi	moor	x
Mgr	mute	maxa	meŋx		noqoi	mori	
San	metie	miqa			noqei	mori	
Bon	metu	maqa			noqui	moru	I
KJ	mete	maka			nukuai	məri	
Mog	meda	mjeqan			10GOi	morin	n
	'tree'	'ice'	'path'	'to play			'leaf'
OM	*motun	*mølsyn	*mør	*nahat	*nai	man	*napč ^h in
Hlh	mət	mos	mor	naat	nai	m	nawč ^h
	мод	мөс	мөр	наад	най	ім	навч
Cha	mət	mos	mor	naat	nee	m	nepč ^h
Baa	mət	mos	mor	naat	nee	m	nepč ^h
Bur	mədən	mul ^j hən	mur	naadə	nai	mən	nabšə
Kmn	mətu	mulihu		naat	nai	ma	napc ^h i
Klm	modn	møsn	mør	naad	nee	mn	
Dag	moot	mvis		naat	nai	m	larč ^h
ShY	muutyn	møsyn	mør	naat		man	lapč ^h vk
Mgr	mooti	molsi	moor	naati		man	lapči
San	mut ^h uŋ	mansuŋ	mo	natu		maŋ	lač ^h eŋ
Bon	mothon	melsoŋ	mor	naat	nin	-	lapč ^h aŋ
KJ	murt ^h un	morsun	mor	nate	nei	mə	laščo
Mog	modu		mur	notu			

OM	'fine' *narin	'age' *nasun	'to open' *nehe	'to weave' *nek ^h e	'sheepskin' *nek ^h ei	'name' *nere
Hlh	nar ^j iŋ	nas	nee	nix	nixi	nir
	нарийн	нас	нээ	нэх	нэхий	нэр
Cha	nerŋ	nas	nyy	nyx		nyr
Baa	nerin	nas	nyy	nyx		nvr
Bur	nar ^j ən	nahən	nee	nexə	nex ^j ə	nerə
Kmn	narin	nahu	nee			nere
Klm	nern	nasn	ne	nek ^h	nek ^h ɛ	nern
Dag	nar ^j ən	nas	nyy			nyr
ShY	naryn	nasyn	nii	nekhe	nyk ^h yy	nere
Mgr	narin	nase	nee	nik ^h i	nik ^h ii	nire
San	naruŋ	nasuŋ	nie	nokhie	nekhi	
Bon	naaraŋ	nasu	nee			neru
KJ	naro	nason	ne		nekhe	nere
Mog	norin			neka		nera

OM	ʻnaked' *nič ^h ykyn	'to hide' *nihu	'baby' *nilk ^h a	'thin' *nimken	'spine' *niruhun	'to fly' *nis
Hlh	nuc ^h gəŋ	nuu	n ^j al z x	nimgəŋ	nuru	nis
	нүцгэн	нуу	НЯЛХ	нимгэн	нуруу	нис
Cha	nuč ^h kən	ทบบ	nılx	niŋkəŋ	nuru	nis
Baa	nuč ^h kən	ทบบ			nuru	nis
Bur	n ^j usegen	n ^j ʊʊ	n ^j ilxə	n ^j imgən	n ^j urgən	n ^j iidə
Kmn		niv				neit
Klm	nyc ^h k ^h n	nu	nilx	nimgn	nurgn	nis
Dag		noo	n ^j alk ^h	ninkyn	nirə	
ShY		ทบบ		neŋkwen	nuruun	honys
Mgr		niu		nenken	nuri	nesi
San	muč ^h uquŋ	niu		niŋk ^h ien	nuruŋ	misu
Bon	nišqaŋ	nuu		niŋqaŋ		muš
KJ				niŋkə	nuru	mysu
Mog		niku		-		

OM	'eye' *nityn	'sleep' *noir	ʻgreen' *nokahan	'dog' *nok ^h ai	'friend' *nøk ^h er	'near' *oira
Hlh	nut	nəir	nəgəŋ	nəxəi	noxər	əir
	нүд	нойр	ногоон	нохой	нөхөр	ойр
Cha	nut	nœœr	nəkəŋ	nəxœ	noxər	œœr
Baa	nut	næær	nəkən	nəxə	noxər	œœr
Bur	n ^j udən	nəir	nəgən	nəxəi	nuxər	əirə
Kmn	nitu			nəkhəi	nuk ^h ur	oiro
Klm	nydn	nør	nogan	noxa	nøk ^h r	øør
Dag	nit	nəir	nuwa	nəkw	nuk ^w ər	wair
ShY	nutun	nuur	nəkəən	nəkhəi	nøk ^h ør	oiro
Mgr	nuti	noor	nuqoon	noxoi	nokhor	
San	nutuŋ	no	noqoŋ	noqei	nok ^h ie	uira
Bon	notoŋ	nor	noquŋ	noqui	nokhor	
KJ	nutu	nur	nokon	nokoai	nok ^h u	
Mog	nudun	noir		noqai		oira

	'to find'	'many'	'to enter'	'to give'	'girl'	'egg'
OM	*ol	*olan	*ora	*øk	*øk ^h in	*ønteken
Hlh	ગંદ	აჭაŋ	or	og	əx ^j əŋ	ontəg
	ол	олон	op	θΓ	охин	өндөг
Cha	əl	ələŋ	or	ok		ontək
Baa	əl	ələn	or	ok		ontək
Bur	ələ	ələn	ərə	ugə	ux ^j ən	undəgən
Kmn	ગ	ələn	oro	uk		untuku
Klm	ol	oln	or	øg	ok ^h n	øndgn
Dag	əl	walən	war	uk ^{wh}	ujən	vnt ^w ək ^w
ShY	ગગ	ələn	oro	ok	hk ^h on	
Mgr	uli	ulon	uro	uqo	ščun	ntike
San	olu	oloŋ	oro	oki	othin	enteqi
Bon	ol	oloŋ	or	OX	ok ^h uŋ	emtuke
KJ		บไบ	υ r υ	uk		nteke
Mog	ol		oru	og	ukin	ondagan

OM	'colour' *øŋke	'to go down' *pahu	ʻrich' *pajan	'to hold' *pari	'tiger' *pars	'also' *pasa
Hlh	oŋg	ρυυ	pajəŋ	par ^j	par	pas
	өнгө	буу	баян	барь	бар	бас
Cha	oŋk	քսս	pajəŋ	per ^j	par	pas
Baa	oŋ	քսս	pajən	per	par	pas
Bur	ungə	bʊʊ	bajən	bar ^j ə	bar	bahə
Kmn	uŋke	рээ				
Klm	øŋg	bu	bajn	ber	bars	bas
Dag	unk ^w	рээ	pajən	par ^j		pas
ShY	øŋkø	ρυυ	pyjan	par	parvs	p ^h ysa
Mgr	ŋko	pau	pajaan	pari	pas	
San	oŋkie	pau	pajaŋ	pari	pasu	phese
Bon		puu	pajaŋ	war	pasu	sa
KJ		pυ	pajo	pari	pasu	
Mog		-		bari	-	bos

	'firm'	'body'	'bride'	'to write'	'letter'	'to rise'
OM	*pat ^h u	*peje	*peri	*pič ^h i	*pič ^h ik	*pos
Hlh	pat ^h	pij	pir	pič ^h	pič ^h əg	pos
	бат	бие	бэр	бич	бичиг	бос
Cha		pvj	pvr	pič ^h	pič ^h ək	pos
Baa	pat ^h	pvj	pyr	pič ^h	pič ^h ək	pos
Bur	bathə	bejə	ber ^j ə	bešə	bešəg	bədə
Kmn			peri	pic ^h i	pic ^h ik	pət
Klm	bath	bij	ber	bič ^h	bič ^h g	bos
Dag		руј	pyr ^j			pos
ShY	path	pij	peerv	p ^h yč ^h y	p ^h vč ^h vk	p ^h əs
Mgr	p ^h ati	pee	peeri		p ^h učik	posi
San	p ^h utu	peje	pieri	p ^h iči		posu
Bon	pat ^h u		weru	p ^h uč ^h i	p ^h uč ^h i	oš
KJ	-	pe	pere	p ^h yč ^h i	p ^h vč ^h iu	posu
Mog		-	beiri	biči	-	bos

OM	'kidney' *pøhere	'louse' *pøhesyr	ʻspring *pulak	,	'to milk' *saha	'to sit' *sahu
Hlh	poor	poos	ხიჭან	; pus	saa	SUU
	бөөр	бөөс	булаг	бүс	caa	суу
Cha	poor	poos	pulək		saa	SUU
Baa	poor	poos	pələk		saa	SUU
Bur	boorə	boohən	buləg	behə	haa	hưư
Kmn	peere		pulak	pehe		həə
Klm	bøør	bøøsn	bulg	bys	sa	su
Dag		puus	pular	pys	saa	sau
ShY	p ^h yyre	pvisvn	pulak	p ^h vsvi	saa	รบบ
Mgr	pooro	poosi	pulaq	p ^h usee	saa	sau
San	poro	posuŋ	pula	p ^h isie	sa	sau
Bon	pooru	poosoŋ	pulaq	se	saa	suu
KJ	pore	posun		p ^h use	sa	SU
Mog		busu			so	sou
OM	'base' *sahurin	ʻgood' *sain	'beard' *sak ^h al	'to protect' *sak ^h i	'comb' *sam	'moon' *sara
Hlh	suur ^j	saiŋ	saxəlz	sax ^j	sam	sar
	суурь	сайн	сахал	сахь	сам	cap
Cha	syir ^j	seeŋ	saxəl		sam	sar
Baa	SYYT	seen	saxəl	SEX	sam	sar
Bur	hươr ^j ə	hain	haxəl	hax ^j ə	ham	harə
Kmn		hain	hak ^h al			hara
Klm	syyr	sen	saxl	sɛkʰ	sam	sar
Dag		sain	sakəl	sak ^j		sar
ShY		sein	sakal	saaky	sam	sara
Mgr		sain	sqal	ski	sam	sara
San		sei-	saqaŋ	saqi	saŋ	sara
Bon		saŋ	saxal	saaqa	sam	sara
KJ		seini		saxi	san	sara
Mog		soin	sagal			

OM	'tail'	'mind'	'to piss'	'new'	'falcon'	'bird'
OM	*sehyl	*setk ^h il	*sihe	*sini	*siŋk ^h or	*sipahun
Hlh	suutz	sit ^h gəb	šee	šin	šəŋxər	šuwu
	сүүл	сэтгэл	шээ	шинэ	шонхор	шувуу
Cha	suul		šyy	šin		šupυ
Baa	suul	svt ^h xəl	šyy	šyn		šupu
Bur	huul	sed ^j xəl	šee	šenə	šənxər	šubun
Kmn	heel			hine		hipoo
Klm	syl	sedk ^h l	še	šin	šoŋxr	šowun
Dag	svul ^j		SYY	šink ^h ən		šəkə
ShY	syyl	set ^h kvl	šii	šyny		šuun
Mgr	suul	skil	šee	šini		šau
San	sien		še	šini		
Bon	sel	sukt ^h e-	še	šinu		
KJ	sar		še	šini		
Mog			sei			

OM	'mud' *sipar	'yellow' *sira	'table' *sirihe	'tooth' *sityn	'arrow' *sumun	'to learn' *sur
Hlh	šawər	šar	šire	šut	sum	sur
	шавар	шар	ширээ	шүд	сум	cyp
Cha	šapər	šar	širv	šut	som	sur
Baa	šapər	šar	širy	šyt	səm	sur
Bur	šabər	šarə	šere	šudən	həmən	hʊrə
Kmn	hipar	hira	hiree	hitu		hor
Klm	šawr	šar	šire	šydn	sumn	sur
Dag	šaur	šar	širv	šit	som	sor
ShY	šwar	šra	šere	štyn	symyn	sur
Mgr	šawar	šira	širee	šti	sumu	suri
San	šuwa	šira	šire	šituŋ	sumu	suru
Bon	šiwar	šira		štoŋ	smo	sur
KJ	šiwar	šira	šire	šituŋ	sumu	sur
Mog		širo		sudun		

OM	'axe' *syk ^h e	'temple' *syme	'sore' *tahari	'sound' *tahun	'to press' *taru	'gown' *tehel
Hlh	sux	sum	taar ^j	tบบ	tar	teelz
	сүх	сүм	даарь	дуу	дар	дээл
Cha	sux	sum	teer	too	tar	tyyl
Baa	SOX	som	teer	too	tar	
Bur	huxə	humə	daar ^j ə	duun	darə	degəl
Kmn	huk ^h e			təə	taru	tekel
Klm	syk ^h	sym	deer	dun	dar	dewl
Dag	sukw	sum	taar	tau	tar	tvvl ^j
ShY	sukhe		taary	toon	tary	tiil
Mgr	suko	smeen		tau	taari	teel
San	sukie			toŋ	taru	tien
Bon	ške			toŋ	tar	
KJ	suku			tơŋ	tari	
Mog				daun	dar	

OM	'above' *tehere	'rope' *tehesyn	'brother' *tehy	ʻlame' *tok ^h alaŋ	'seven' *tolahan	ʻinside' *tot ^h ara
Hlh	teer	tees	tuu	təckəŋ	tobo	tət ^h ər
	дээр	дээс	дүү	доголон	долоо	дотор
Cha	tyyr	tyys	tuu	təkləŋ	tələ	tət ^h ər
Baa	tyyr	tyys	tuu	təkləŋ	tələ	tət ^h ər
Bur	deerə	deehən	duu	dəxələn	dələn	dət ^h ər
Kmn	teere		tee		tələə	
Klm	deer	deesn	dy	doglŋ	dolan	dot ^h r
Dag	tyyr	tyys	tvu	tək ^w l ^w ən	tələ	t ^w at ^{wh} ər
ShY	tiire	tiisvn	tyy	təkələŋ	tələən	ht ^h ər
Mgr	tire	teesi	tiu	toqloŋ	toloon	t ^h utor
San	tiere	tiesuŋ	tiau	toqoloŋ	toloŋ	sutoro
Bon	tiiru		tu	toqlaŋ	toloŋ	
KJ	tere	teisun	tewu	toklo-	tələ	
Mog	dera	deisun				

OM	'forty' *tøč ^h in	'four' *tørpen	'middle' *tumta	ʻfull' *tyhyreŋ	'to offer' *t ^h ak ^h i	'hen' *t ^h ak ^h ija
Hlh	točh	torəw	tont	tuurəŋ	t ^h ax ^j	t ^h ax ^j a
	дөч	дөрөв	дунд	дүүрэн	тахь	тахиа
Cha	toč ^h	torəp	tont	tuurən		tex ^j a
Baa	točh	torəp	tont	tuurəŋ	t ^h EX	t ^h exa
Bur	dušən	durbən	dundə	duurən	t ^h ax ^j ə	t ^h ax ^j a
Kmn	tuchi	turpe	tomta			t ^h ak ^h ie
Klm	døč ^h n	dørwn	dund	dyyrŋ	t ^h ɛk ^h	t ^h ak ^h a
Dag	tuč ^h	turp ^w	t ^w ant	tuur-	t ^h ak ^{jh}	tyki
ShY	tøč ^h vn	tørwen	tonta	tyreŋ	tak ^h v	tak ^h a
Mgr	t ^h ičin	teeren	tunta	tiuri-		t ^h iqau
San		tieroŋ	tunta	turan		t ^h uqa
Bon		teraŋ		tur-		t ^h uxa
KJ		tero	tonta	tur-		t ^h axa
Mog		durbon		dur-		

	'to recognize'	'fifty'	'five'	'to pull'	'hare'	'camel'
OM	*t ^h ani	*t ^h apin	*t ^h apun	*t ^h at ^h a	*t ^h aulai	*t ^h emehen
Hlh	t ^h an ^j	t ^h aw ^j	t ^h aw	thath	t ^h uutzai	t ^h ime
	тань	тавь	тав	тат	туулай	тэмээ
Cha	t ^h ɛn ^j	t ^h ɛp ^j	t ^h ap	tat ^h	t ^h υʊlε	t ^h ymy
Baa	t ^h ɛn	t ^h εp	t ^h ap	t ^h at ^h	t ^h ʊʊlɛ	t ^h omo
Bur	t ^h an ^j ə	t ^h ab ^j ən	t ^h abən	t ^h at ^h ə	t ^h uulai	t ^h emen
Kmn		t ^h api	t ^h apu		t ^h əəlai	t ^h emee
Klm	t ^h an ^j	t ^h ewn	thawn	t ^h at ^h	t ^h uula	t ^h emen
Dag	t ^h an ^j	t ^h ap ^j	t ^h aawə	t ^h at ^h	t ^h aul ^j	t ^h vmv
ShY	t ^h any	thawyn	t ^h aawyn	ht ^h a	t ^h uulii	t ^h emeen
Mgr	t ^h ani	t ^h ajin	t ^h aawun	t ^h ita	t ^h oolii	t ^h imeen
San	t ^h ani	-	t ^h awuŋ	sta	t ^h aulei	
Bon	t ^h ani		thawoŋ	šta	t ^h oli	
KJ	t ^h ani		thawon	stha	t ^h ʊli	
Mog	toni		taban	tata		temon

OM	ʻiron' *t ^h emyr	'number' *t ^h ohan	'kettle' *t ^h okahan	'fat' *t ^h osun	'to drink' *uhu	'to cry' *uila
Hlh	t ^h omər	thoo	thogo	thos	υυ	vilz
	төмөр	тоо	тогоо	тос	уу	уйл
Cha	t ^h omər	thoo	t ^h əkə	təs	υυ	чıl
Baa	thomər	thoo		thos	υυ	yyl
Bur	t ^h umər	thoon	t ^h əgən	t ^h əhən	υυ	vilə
Kmn	t ^h umur	thoo		thoho	UU	
Klm	t ^h ømr	t ^h o		t ^h osn	u	uul ^j
Dag		thoo	t ^h uwa	thos	33	wail
ShY	t ^h emər	t ^h បប n	t ^h əkəən	t ^h uusyn	UU	yyla
Mgr	t ^h imur	t ^h oo	t ^h uqoo	t ^h oosi	uu	ulaa
San	t ^h iemu	t ^h eu-	t ^h uqoŋ	t ^h osuŋ	u	uila
Bon	t ^h emur	t ^h oo-	t ^h uxoŋ	t ^h osoŋ	uu	laa
KJ	čhimo	t ^h ʊ-	t ^h បxបŋ	t ^h usun		ila
Mog				tosun		uila
OM	'to wash' *ukija	'to forget' *umart ^h a	' 'to sleep' *unt ^h a	'to ride' *unu	'to summo *uri	on' 'long' *urt ^h u
Hlh	UGA	marth	unth	υn	ur ^j	urth
	угаа	март	унт	уна	урь	урт
Cha	uka	mart ^h	ប n t ^h	υn		urth
Baa	uka	mart ^h	ប n t ^h	υn	Yľ	urth
Bur	uga	marthə	បnt ^h ə	υnə	Ur ^j ə	បt ^h ə
Kmn	ukaa	mart ^h a	untha			
Klm	uga	mart ^h	unt ^h	un	yr	ut ^h
Dag	waa	mart ^h	want ^h	∋n ^w ə	ərj	ərth
ShY	ukwaa	mart ^h a	nta	hony		hʊrtʰʊ
Mgr	ŋqoa	muštaa	nt ^h aa	xuni	uri	štur
San	waqa	mat ^h a		unu	uru	xutu
Bon	waqa	martha	t ^h aa	honu	ur	štur
KJ	ukua	mart ^h a		une	uri	štu
Mog	uga	morta		uno		urtu

	'water'	'to see'	'point'	'deed'	'joint'	'word'
OM	*usun	*yče	*yčyhyr	*yile	*yje	*yke
Hlh	US	uc	ucur	uitz	uj	ug
	yc	үз	үзүүр	үйл	үe	үг
Cha	US	uč	učur	yil	uj	uk
Baa	US	uč	učur	yyl	uj	uk
Bur	uhən	uzə	uzur	uilə	ujə	ugə
Kmn	uhu	uce				
Klm	usn	yz	yzyr	yyl	уј	yg
Dag	ЭS	uč	xučur	wyil	wyj	
ShY	k ^h usun	eče	čyyr	ule	jwee	uke
Mgr	scu	uča	učuur	ule	uje	uko
San	usu	uče	uču	uilie	uje	
Bon	su	uči	učir	wilu		
KJ	su	uče	učir	ile		
Mog	usun	uja				

OM	'to die' *yk ^h y	'cow' *ynijen	'high' *yntyr	'day' *ytyr
Hlh	ux	une	ontər	otər
	үx	үнээ	өндөр	өдөр
Cha	ux	un ^j y	ontər	otər
Baa	ux	une	ontər	otər
Bur	uxə	un ^j en	undər	udər
Kmn		unie	untur	utur
Klm	ykh	una	øndr	ødr
Dag	ukw	un ^j e	xunt ^w ər	ut ^w ər
ShY	hk ^h u	niin	ontor	otor
Mgr	xuku	unee	untur	utur
San	xuku		untu	utu
Bon	hku	unaŋ	untur	utur
KJ	ku	unio	untyr	utvr
Mog	uku	uinan		udur

10

DEVELOPMENT OF THE MODERN MONGOLIC LANGUAGES¹

This chapter describes the development from Old Mongolian to the Modern Mongolic languages, in particular Halh Mongolian. The standard works by Ramstedt (1902; 1957), Vladimircov (1929), Sanžeev (1953), Poppe (1955; 1960*a*), Yu Shichang (1983), Tömörtogoo (1992), and Darbeeva (1996) are relevant for the whole chapter, and we will not refer to them specifically in each section. More specific literature is referred to in the relevant places. In order to avoid excessive repetition, examples which illustrate the normal developments in each language are given by referring to the comparative vocabulary in section 9.13.

10.1 THE MONGOLIC VOWEL SHIFTS²

The Old Mongolian seven-vowel system has changed considerably in all modern Mongolic languages except Oirad. Three phonological processes, velarization, pharyngealization, and palatalization, account for most changes. Vowel palatalization (umlauting) is conditioned by a following *i*. In some languages, for example, Halh Mongolian, it has only had an effect on the phonetic surface, but in Eastern Mongolian and in Oirad it has restructured the vowel system. Vowel palatalization is treated in section 10.11.2.

In contrast to palatalization, velarization and pharyngealization are not conditioned by the phonological environment, and have no obvious internal motivation. These processes, which have affected all modern Mongolic languages except Oirad, are treated in this section, as the two parts of the Mongolic vowel shift. In order to describe the vowel shifts we will use the features [palatal], [velar], [pharyngeal], [open], and [round] ([P], [V], [F], [O], [R]) introduced in section 5.1.

Velarization is the process of adding the feature [velar]. This converts front vowels to back vowels:

(1) y [PR] \rightarrow u [PVR] e [PO] \rightarrow y [VO] ϕ [POR] \rightarrow o [VOR]

¹ For notes and sources, see App. J, 10, p. 228.

² For notes and sources, see App. J, 10.1, p. 228.

There is a general rule which excludes the combination *[PVO], replacing it with [VO], so if the feature [velar] is added to an open vowel with the feature [palatal], this feature is deleted, as in the two last cases in (1).

Pharyngealization adds the feature [pharyngeal]:

(2) i [P] \rightarrow I [PF] u [PVR] \rightarrow υ [PVFR] o [VOR] \rightarrow 2 [VFOR]

These processes are illustrated in Figure 10.1, which gives a synchronic picture of the diachronic change from Old to modern Mongolian. This figure (from Svantesson 1995*b*) is an F1–F2 diagram showing simultaneously the long vowels of Dörbed Kalmuck and Shiliin Gol Mongolian. Vowels which have been formed by palatalization are not shown. The Kalmuck vowels are encircled, and arrows point towards the etymologically corresponding Mongolian vowels. The Kalmuck vowels are unchanged compared to Old Mongolian, and the Shiliin Gol Mongolian vowels have gone through pharyngealization (dotted arrows) and velarization (solid arrows). The vowel **i* split into two phonemes, *i* and *i*, in Shiliin Gol (see 10.2). The only unaffected vowel is *a*. It can be seen from Figure 10.1 that velarization changes the formants in the direction of the vowel [o], with the feature specification [VOR], having [velar] as its only place feature. Pharyngealization has its locus in the area of [a] with [pharyngeal] as its place feature. The main acoustic effects of velarization and pharyngealization are to decrease F2 and to increase F1, respectively.

The different Mongolic languages have gone through velarization and pharyngealization to different extents, and exactly which vowels are involved varies from



FIGURE 10.1 The Mongolic vowel shift illustrated synchronically

language to language. The basic vowel correspondences in initial syllables are summarized in (3), which shows the reflexes of short initial vowels in environments where other vowel-changing processes, such as palatalization and *i-assimilation have not taken place.

(3) Main reflexes of short vowels in initial syllables

The Dagur reflex wa of *u is due to 'breaking' (see 10.7.5).

OM	*a	*0	*u	*e	*ø	*у	*i
Hlh	a	э	υ	i	0	u	i
Cha	a	э	υ	Y	0	u	i
Baa	a	э	υ	Y	0	u	i
Bur	a	э	υ	e	u	u	e
Kmn	a	э	υ	e	u	u	i
Klm	a	0	u	e	ø	у	i
Dag	a	э	o, wa	¥	u	u	i
ShY	a	э	υ	e	ø	u	Y
Mgr	a	0	u , o	i, e	0, u	u	i
San	a	0	u	ie, e	0	u	i
Bon	a	0	u	e	0	u	i, w
KJ	a	υ, ο	υ	e	0, u	u	i
Mog	a, o	0	u	e	0	u	i

Examples (9.13): */a/: *alt^han, *ap, *č^hasun, *kar, *k^hara, *t^hapun; */o/: *hon, *nok^hai, *ol, *ora, *pos, *t^hosun; */u/: *hulahan, *kurpan, *k^huruhun, *sur, *ukija, *usun; */e/: *ene, *ker, *k^helen, *mete, *nere, *peri; */ø/: *k^høk^he, *k^høl, *k^hølesyn, *mør, *nøk^her, *øk; */y/: *k^hynty, *k^hyr, *k^hyreken, *syk^he, *yče, *yk^hy; */i/: *čil, *č^hik^hin, *nis, *pic^hi(k), *sini.

With respect to the vowel shift, the modern Mongolic languages can be divided roughly into four types: the Oirad, Monguor, Mongolian, and Dagur types. We illustrate them with stylized vowel diagrams in Figure 10.2.

The Old Mongolian vowel system is retained in Oirad (including Kalmuck), with the vowel ε added by palatalization.

In Monguor, Santa, Bonan, and Moghol, *y and $*\phi$ were velarized and merged with *u and *o, leading to a five-vowel system; see Figure 10.2a.

The vowels were changed by pharyngealization and velarization in Mongolian, Buriad, Kamnigan, Shira Yugur, and Kangjia. The back vowels *u and *o were pharyngealized to v and z in all these languages. Velarization of the three front vowels *e, $*\phi$, *y took place in most non-Halh Mongolian dialects including Baarin and Chahar, and also in some Halh dialects in the south, such as Shiliin Gol. In Standard Halh, Kamnigan, and Kangjia, only $*\phi$ and *y were velarized, and in Shira Yugur only *y. These changes did not cause mergers, so the contrasts of the original seven-vowel system are retained, but the vowel qualities have changed, leading to vowel systems such as that in Halh, illustrated in Figure 10.2b. (a) Monguor type



(b) Mongolian type





(2) Velarization $i \quad y \longrightarrow u$ $a \quad a$







(c) Dagur type



FIGURE 10.2 Mongolic vowel shifts

Dagur has apparently gone through the same developments as the languages of the Mongolian type, followed by a polarization of the rounded vowels so that the features [open] and [pharyngeal] either occur together or not at all for these vowels. This caused v [FR] to merge with z [FOR] and o [OR] to merge with u [R]. As in the languages of the Monguor type, this led to a five-vowel system (see Fig. 10.2c), but the Dagur vowels u and z have different origins than Monguor u and o. The vowel ε was added by other processes.

10.2 VOWEL SPLITS AND MERGERS³

In Chahar and some other dialects of South Mongolia (including Shiliin Gol, Harchin, and Urad), the initial vowel *i has split into two phonemes, depending on the vowel harmony class of the other vowels of the word. In words which contain back non-initial vowels in Old Mongolian, initial *i either became *i* or was assimilated to the following vowel. In words with Old Mongolian front vowels, *i is retained, if it is not assimilated, as is often the case when the following vowel is *y (>*u*). When final vowels disappeared, the allophonic variation between [i] and [I] became contrastive, and the Old Mongolian phoneme */i/ split into two, /*i*/ and /t/:

(4)	Old Mongolian	*čita 'spear'	*ite 'to eat'
	pharyngealization	čīta	
	vowel deletion (7)	čıt	it
	Chahar	čıt	it

Other examples are OM k^{hilpar} 'easy' and $mink^{han}$ 'thousand', which are $x_{ilp\sigma r}$ and mink in Chahar. Chahar *i* corresponds to ε in Baarin, where the exemplified words are $\check{c}\varepsilon t$, $x\varepsilon lp\sigma r$, $m\varepsilon nn$. In Halh, *i-assimilation is obligatory in original back-vocalic words (10.7.3), and the exemplified words are $\check{c}at$ жад, $x^{j}agp\sigma r$ хялбар, $m^{j}ang$ мянга.

Initial short *i and *e have merged to i in Halh, at least in the variant spoken in Ulaanbaatar. According to Möömöö and Mönh-Amgalan (1984: 82), this merger took place in Central Halh in general (cf. also Vladimircov 1929: 148ff.; Luvsanbaldan 1982). These vowels have been kept distinct in most variants of Southern and Eastern Mongolian, with *e velarized to v. Short *i and *e usually merged to e (pronounced [ui]) in Buriad, and to i in Monguor. Examples are OM $*k^helen$ 'tongue' and *ite 'to eat', which became $xit_{\mathcal{J}}$ xэл, it µµ in Halh, xelan, $ed^{i}a$ in Buriad, k^hile , ite in Monguor, and xvl, it in Chahar and Baarin. At least partial merger of *i and *e took place in Urad (Batusayiqan 1985; Möngkebuyan 1992a; 1997) and Ordos (Serengnorbu 1996; Möngkebuyan 1990).

³ For notes and sources, see App. J, 10.2, p. 229.

In Buriad (except the western dialects) and in (Urul'ga) Kamnigan, short initial *y and * ϕ merged to u. The same development took place in Dagur, in the context of other vowel changes. In Mongolian proper, these vowels are kept apart, as u and o. Examples are OM * ϕ k 'to give' and * hyk^{her} 'ox', which became uge, ux σ r in Buriad, uk, uk^{her} in Kamnigan, uk^{wh}, xuk^{wh} σ r in Dagur, but og $\Theta\Gamma$, ux σ r yx σ p in Halh (see also 9.2).

In Eastern Mongolian dialects, especially Baarin and Harchin, as well as in Dörbed Kalmuck, *u and *y regularly became open before labial consonants, merging with the reflexes of *o and *ø, respectively. Examples are Baarin and Harchin som 'arrow', som 'temple', xæp 'destiny' vs. Halh som cyM, sum cYM, xow^j xyBb (OM *sumun, *syme, *k^hupi). In Dörbed, these words are somn, søm, xøw, whereas the standard Kalmuck forms (based on the Torguud dialect) are sumn, sym, xyw. This development took place after labial consonants as well, but less regularly. An example is Baarin and Harchin polok, Halh pu5og булаг 'spring', OM *pulak.

A few words with Old Mongolian *y, including *yntyr 'high' and *ytyr 'day', have reflexes of * ϕ rather than *y in many modern languages, including Mongolian (Halh *ontər* өндөр, *otər* өдөр) and Oirad (Kalmuck ϕndr , ϕdr) (see also Vladimircov 1929: 155).

10.3 LONG VOWELS⁴

All modern Mongolic languages except Moghol, Santa, Kangjia, the Minhe dialect of Monguor, and the Jishishan dialect of Bonan have contrastive vowel length. Its absence in these languages is due to a late process which made all vowels short. In Mongolian, Buriad, Oirad, Dagur, and Bonan there is a quantity contrast only in the initial syllable.

Although some languages have 'primary' long vowels, possibly inherited from Proto-Mongolic (8.5.1), the long vowels of most modern Mongolic languages are 'secondary' in the sense that they have developed from Old Mongolian *VhV or *ijV combinations.

The development of these groups in the modern languages is shown in (5). Because there are rather few examples for some of the individual *VhV groups, it is not always possible to trace their development accurately, especially in the lesser known languages.

When the two vowels surrounding *h were identical, the *VhV group usually developed to a long vowel with the same quality as the corresponding short vowel. The groups *oha and $*\phi he$ developed to long monophthongs corresponding to short *o and $*\phi$. Groups of the form *ihV and *ijV developed in the same way as the *VhV groups with identical vowels, except that the preceding consonant was palatalized in some languages (see 10.11.1).

⁴ For notes and sources, see App. J, 10.3, p. 229.

(5) Old Mongolian *VhV and *ijV groups

Reflexes in initial/non-initial syllables are separated with a slash. Shortening of etymologically long vowels in non-initial syllables (10.5.1) is not indicated. For reflexes given in parentheses, we have only one example.

OM	*aha *iha/ija	*ehe *ihe/ije	*oha	*øhe	*uhu *ihu	*yhy	*ahu *au	*ehy	*uha	*yhe
Hlh	aa	ee	33	00	υυ	uu	υυ	uu	(၁)	(0)
Cha	aa	ŶŶ	33	00	υυ	uu	υυ	uu		(0)
Baa	aa	ŶŶ	33	00	υυ	uu	υυ	uu	(၁)	
Bur	aa	ee	33	00	υυ	uu	υυ	uu		(00)
Kmn	aa	ee	(00)	(ee)	υυ		33	ee		
Klm	aa	ee/ɛ	00	øø	uu	уу	uu	уу	(a)	(٤)
Dag	aa	ΥY	33		33	uu	au/ɔ	хu		(y)
ShY	aa	ii, e(e)	υυ	y(y)	υυ	y(y)	υυ	y(y)	(vv)	(øø)
Mgr	aa	ee	00	00	uu/i	u(u)	au,u	u(u)		
San	a	ie, e		0	u	u	au,u		(0)	
Bon	aa	ee, ii		00	uu/o		uu			
KJ	a, (3)	e	υ	0	υ		υ,ο		(U)	
Mog	o, a	ei/a	(0)	(u)	u	(u)	au	(oi)		

Examples (9.13): */aha/: *k^haha, *nahat, *saha, *č^hakahan, *hulahan, *imahan; */iha/: *čiha, *ač^hihan; */ija/: *t^hak^hija, *ukija; */ehe/: *nehe, *tehere, *tehesyn, *emehel, *inehe, *t^hemehen; */ihe/: *sihe, *k^hič^hihe, *sirihe; */ije/: *k^hečije, *ynijen; */oha/: *k^hohasun, *t^hohan, *čirkohan; */øhe/: *čøhelen, *pøhere, *pøhesyn; */uhu/: *huhut^ha, *uhu, *k^huruhun, *niruhun; */ihu/: *nihu, *arihun, *kasihun; */yhy/: *tyhyreŋ, *yčyhyr; */ahu/: *ahula, *čahu, *sahu, *čalahu, *č^hilahun, *k^halahun; */au/: *k^hauč^hin, *t^haulai; */ehy/: *ehyten, *sehyl, *tehy; */uha/: *hiruhar; */yhe/: *hiryher.

In a few cases, the *VhV groups developed vowel qualities different from the corresponding short vowels. The mergers of *i and *e to i in Halh and Monguor, and of *y and $*\phi$ to u in Buriad, concern only short vowels. Thus *ehe became ee in Halh and Monguor, and $*\phi he$ became oo in Buriad. In Shira Yugur, *oha and *yhy developed to vv and yy, while the short vowels *o and *y became o and u.

The groups **ahu* and **ehy* are often retained as diphthongs in Dagur, Monguor, Santa, and Moghol, while they became monophthongs, usually merging with the combinations **uhu* and **yhy*, in the other languages. Thus, **ahu* became vv in Mongolian and Buriad and *uu* in Oirad. It developed in its own way in Kamnigan, becoming vv, while **uhu* became vv. The diphthong **au* developed in the same way as **ahu*. The rather rare combinations **uha* and **yhe* have developed in ways similar to **oha* and **øhe* (Poppe 1950; Thomsen 1958; Kuribayashi 1984).

10.4 **i*-DIPHTHONGS⁵

The Old Mongolian **i*-diphthongs are preserved as diphthongs in most languages (6). In the Eastern Mongolian dialects, in some Southern Mongolian dialects, including Shuluun Höh Chahar, and in Oirad, the diphthongs were monophthongized, and some monophthongization took place in other languages as well. The Old Mongolian back vowels **ai* and **oi* became front vowels both in Baarin ($\varepsilon\varepsilon$, $\alpha \alpha$) and in Kalmuck ($\varepsilon\varepsilon$, $\phi \phi$). However, this had different consequences for vowel harmony in these two languages (see 10.11.3).

(6) Old Mongolian *i-diphthongs

Reflexes in initial/non-initial syllables are separated with a slash. See section 10.7.1 for the development of *ai in non-initial syllables when the first syllable has the vowel *o. The Dagur reflexes *wai* and *wvi* of *ui/oi and *yi are due to 'breaking' (see 10.7.5).

OM	*ai	*oi	*ui	*ei	*øi	*yi
Hlh	ai	oi	υi	ii	ui	(ui)
Cha	33	œœ	YI	(ii)	ii	(yi)
Baa	33	œœ	YY	(ii)	уу	(yy)
Bur	ai	oi	υi	ii	ui	(ui)
Kmn	ai	oi			(ui)	
Klm	εε/a	ØØ		(i)	ii	(yy)
Dag	ai/Ø	oi, (wai)	(wai)	(yi)	ui	(wyi)
ShY	ai, ei	oi		(ii, yy)	y(y)	(u)
Mgr	ai			ii	ui	(u)
San	ei	ui	(ui)	(ai, i)		(ui)
Bon	ii, wi	ii		(ii)	i(i)	(wi)
KJ	ai, ei		(i)	(e)	(ui)	(i)
Mog	(oi/ei)	oi	(ui)	(ei, a)		

Examples (9.13): */ai/: * k^haic^hi , *naiman, *sain, * kak^hai , * $k^hulakai$, * t^haulai ; */oi/: *hoi, * k^hoina , *noir, *oira; */ui/: * k^hui , *uila; */ei/: * $k^{h}ei$; * $nek^{h}ei$; */ ϕ i/: * $k^h\phi$ isyn, * $k^h\phi$ it^hen; */yi/: *yile.

10.5 VOWEL DELETION AND REDUCTION

Different vowel deletion and reduction processes have affected all modern Mongolic languages except Kamnigan and Moghol. In some languages, non-initial

⁵ Sanžeev (1940: 12f.); Pavlov (1963*a*; 1974); Verba (1980); Tömörtogoo (1985); Öljeyibürin (1996); Yu Rong (2002).

vowels were deleted or reduced, but in others this happened to initial vowels. This may have been due to different stress patterns in different languages, but virtually nothing is known about stress in the older stages of the languages, and we will not speculate on this.

The Old Mongolian sources generally preserve all vowels, except that Arabic Mongolian tends to delete some initial vowels (see Saitô 1996*a*).

10.5.1 Reduction of short non-initial vowels⁶

Some Old Mongolian short non-initial vowels correspond to schwa vowels in Halh (e.g. OM *kačar > Halh gacar rasap 'ground'), some have a zero reflex ($*emys > oms \ \Theta mc$ 'to wear'), and some correspond to a schwa at another place than in Old Mongolian ('metathesis', e.g. *arka > arag apra 'method'). It was shown in Chapter 6 that Halh schwas are epenthetic vowels which are not present in phonological representations, but are inserted by rules. This makes it possible to analyse all cases of vowel reduction in Mongolian by a vowel deletion rule:

(7) Deletion of short vowels

All short vowels in non-initial syllables are deleted from the phonological representation.

This rule applies to all dialects of Mongolian proper, as well as to Buriad (except for word-final vowels (see 9.2), Oirad, and Dagur. Other Mongolic languages normally preserve non-initial vowels. Numerous examples can be found in the comparative vocabulary (9.13).

If necessary, non-phonemic schwa vowels are inserted to make well-formed syllables after rule (7) has applied. The derivations of some words which illustrate the interaction of vowel deletion and schwa epenthesis in Halh are shown in (8).

(8)		'blue'	'to wear'	'ground'	'method'
	Old Mongolian	*k ^h øk ^h e	*emys	*kačar	*arka
	segmental changes	xoxe	omus	Gacar	arga
	vowel deletion	XOX	oms	Gacr	arg
	phonological represen- tation	/xox/	/oms/	/gacr/	/arg/
	schwa epenthesis			Gacər	arəg
	Halh surface form	хох хөх	oms өмс	Gacər газар	arәg арга

From examples as *arka > arac 'method', it can be seen that 'metathesis' does not result from a special metathesis rule, but from the interaction of vowel deletion and epenthesis. In this example, the second vowel is deleted, and then a schwa is inserted into the surface form to make well-formed syllables, using rule (6) in Chapter 6.

⁶ Arai (1960); Pavlov (1963*a*; 1982); Kuribayashi (1988; 1992); Engkebatu (1994*a*); Qasbagan-a (2000*b*).

Long vowels, which were formed from *VhV and *ijV groups (10.3), are not deleted in non-initial syllables, so their development must have preceded the vowel deletion rule (7). When the short vowels have been deleted, only long vowels and diphthongs remain in the phonological representation of non-initial syllables. Since only one vowel quantity remains, it is natural to reinterpret it as the unmarked quantity, that is, short (see section 1.1 for acoustic support for this). This process can be formulated as a rule, which applies in those languages where the vowel deletion rule applies (9).

(9) Long non-initial vowels become short.

Some derivations involving this rule are shown in (10). It is obvious that it must be applied after the deletion of medial *h, and after the deletion of short vowels. The order of the other segment-changing rules is irrelevant here, except that the rule which changes onset *k to σ must apply before short vowels are deleted.

(10)	Old Mongolian deletion of *h other segmental changes deletion of short vowels shortening of long vowels	ʻgate' *k ^h ahalka k ^h aalka xaatsga xaatsg —	ʻred' *hulahan ulaan otsaaŋ — otsaŋ	ʻwarm' *k ^h alahun k ^h alaun xakyon — xakyon	'beast' *kørehesy/n køreesy goroosu goroos goroos
	phonological representa- tion	/xaabg/	/oţsaŋ/	/xabuŋ/	/goros/
	vowel epenthesis	хаађәс	_		
	Halh surface form	хаађәс хаалга	ођађ улаан	хађођ халуун	goros гөрөөс

As mentioned in section 6.6, monosyllabic words consisting of an open syllable always have a long vowel in Mongolian proper. In Kalmuck as well, there is no vowel length contrast in this position. In the Cyrillic Kalmuck orthography, these words are written with single vowel letters, for example, cy < su > 'to sit'. Sanžeev (1978) says that these vowels are long, as in Mongolian (Halh *svv* cyy) and Buriad (*hvv*), but shows that the duration of these vowels are closer to short than to long vowels (see also Bitkeev 1970; 1975*a*; 1983: 8; Pavlov 1983: 43). The etymologically long vowels are thus shortened in this position, but turn up as long vowels in inflected forms, for example, in the future participle cyyx < suux > of the exemplified Kalmuck word.

There is a contrast between long and short vowels in monosyllabic words which end in a consonant both in Kalmuck and Halh. Monosyllabic words which have an etymologically long vowel (or a diphthong) have a short vowel in Kalmuck if they ended in a consonant in Old Mongolian (e.g. OM **sehyl* > KIm *syl* 'tail'; **noir* > *nør* 'sleep'), but a long vowel if they ended in a consonant+vowel combination (e.g. **ahula* > *uul* 'mountain', **oira* > *øør* 'near') (see also Benzing 1985: 177).

10.5.2 Deletion of initial vowels⁷

In the Mongolic languages spoken in the Gansu–Qinghai area (Shira Yugur, Monguor, Santa, Bonan, and Kangjia) there is an opposite tendency to delete initial vowels. This is only a tendency, and not a strict rule, however. Vowels in absolute word-initial position, not preceded by a consonant, are most liable to be deleted, but deletion is common also when the word-initial consonant was a fricative (*s or *h). After other initial consonants, vowels are deleted only sporadically. The vowel quality plays a role as well. The vowels *u, *y, *e are deleted rather often, *iand $*\phi$ more seldom, and *a and *o only rarely. Few words lost the initial vowel in Santa, and almost all of them had word-initial *e. It should be emphasized that the great majority of words retain their initial vowels in these languages. The deletion of initial vowels created initial consonant clusters, unknown in other Mongolic languages, but occurring in the Tibetan dialects in the same area, so this can be regarded as an areal feature (Kim Pang-han 1969; 1973; Janhunen 2001). Examples of words with initial vowel deletion (italicized) are given in (11).

(11) Deletion of initial vowels 8

OM	ShY	Mgr	San	Bon	KJ	'to wear'
*emys	mys	mosi	misui	muš	mysw	
*epesyn *unt ^h a	wesyn nta	wesi nt ^h aa	osuŋ	epsoŋ t ^h aa	weisun	ʻgrass' 'to sleep'
*urt ^h u	hʊ rt ʰʊ	<i>štur</i>	xutu	štur	štv	'long'
*yk ^h y	hkʰu	xuku	xuku	hku	ku	'to die'
*ite	ete	ite	itie	nte	ite	'to eat'
*øk ^h in	hk ^h on	ščun	ot ^h in	ok ^h uŋ	šči	'girl'
*hič ^h e	hč ^h e	ščee	šiče	<i>šče</i>		'ashamed'
*sityn	<i>štyn</i>	<i>šti</i>	šituŋ	štoŋ	šituŋ	'tooth'
*sumun	symyn	sumu	sumu	smo	sumu	'arrow'
*k ^h øk ^h e	hk ^h ø	k ^h uko	k ^h ukie	k ^h uko	k ^h uk ^h u	'blue'
*t ^h at ^h a	ht ^h a	t ^h ita	sta	<i>šta</i>	st ^h a	'to pull'

In a few words, including **umart*^h*a* 'to forget' (Halh *mart*^h MapT), the initial vowel disappeared in all modern languages (cf. Garudi 1996; 1999).

10.6 NON-INITIAL VOWELS AND VOWEL HARMONY

As seen in Chapter 8, there was a four-way vowel contrast in Old Mongolian noninitial syllables between *i, *A ($*a \sim *e$), *U ($*u \sim *y$), and *O ($*o \sim *\phi$). The third alternation pair, $*o \sim *\phi$, was rare, appearing only after *i in the initial syllable. The

⁷ Kotwicz (1939); Todaeva (1973); Bulucilagu (1989); Qasbagan-a (1998); Qasbagatur (1999).

⁸ These tables illustrate phenomena which apply only to some of the words. The words to which they apply are italicized. (See also (37), (38), (39), (40), (41).)

development of non-initial vowels is shown in (12). In Mongolian, Buriad, Oirad, and Dagur, short non-initial vowels were deleted from the phonological representation, appearing on the phonetic surface as schwas or not at all. In the remaining modern languages, non-initial vowels developed more or less like initial vowels. The contrast between non-initial *a/e and *u/y which existed in Old Mongolian was lost in Mongolian, Buriad, and Oirad, but is retained in Dagur as a contrast between plain and labialized consonants. The contrast with *i is retained in all languages, either in the vowel itself or as palatalization of the preceding consonant.

There is a special development in Kangjia, where a final **Vn* combination often became *z*, for example, OM **alt*^{*h*}*an* 'gold' > KJ *ant*^{*h*}*z*; OM **č*^{*h*}*ik*^{*h*}*in* 'ear' > KJ $č^{h}ixz$.

(12) Vowels in non-initial syllables

Mongolian, Buriad, Kalmuck, and Dagur schwas are written as \emptyset . See section 10.7.1 for the developments after initial **o* and * ϕ .

OM	*a	*u	*e	*у	*i
Hlh	Ø	Ø	Ø	Ø	Ø
Cha	Ø	Ø	Ø	Ø	Ø
Baa	Ø	Ø	Ø	Ø	Ø
Bur	Ø	Ø	Ø	Ø	Ø
Kmn	а	σ	e	u	i
Klm	Ø	Ø	Ø	Ø	Ø
Dag	Ø	Ø	Ø	Ø	Ø
ShY	а	γ, υ	e	v, u	¥
Mgr	а	u, i	e	i, u	i
San	а	u	ie, e	u	i
Bon	а	ш, о	ш	0	ш
KJ	a, (3)	σ	e	u	i
Mog	a, o	u	a	u	i

Examples (9.13): */a/: $*alt^han$, *kačar, *kurpan, $*k^hara$, $*mik^han$, *sira; */u/: $*\check{c}^hasun$, *nasun, *taru, $*t^hapun$, $*urt^hu$, *usun; */e/: *ene, *ite, $*k^helen$, *mete, *tehere, $*y\check{c}e$; */y/: $*k^hetyn$, $*k^hynty$, *nityn, $*p\phi hesyn$, *sityn, $*yk^hy$; */i/: $*\check{c}^hik^hin$, $*k^hari$, $*k^honin$, *morin, *pari, *peri.

10.6.1 Vowel harmony shifts⁹

The vowel shifts in the different Mongolic languages resulted in changes of the vowel harmony system or in some cases even in vowel harmony loss. The conse-

⁹ Hattori (1975; 1978); Gregerson (1976: 361ff.); Yu Shichang (1983: 28ff.); Rialland and Djamouri (1984); Svantesson (1985); Kim Chu-wŏn (1988*a*, *b*; 1992; 1993; 1997; 1999*a*). Vago (1973), S. Anderson (1980) and Binnick (1991) regard the vowel shifts as destructive for vowel harmony. See also Darbeeva (1996: 117ff.).

quences for vowel harmony roughly depend on the four types of vowel changes outlined in section 10.1. Since Old Mongolian vowel harmony was based on the three back ~ front vowel pairs $*a \sim *e$, $*o \sim *\phi$, and $*u \sim *y$, the development of these pairs is crucial for the fate of vowel harmony, as illustrated in (13).

(13) Vowel harmony shifts

0	М	Kalr	Kalmuck Monguor		I	Mongolian			Dagur		
*a	*e	a	e	a	e		a	e/v		a	Ŷ
*0	*ø	0	ø	0			э	0	Γ		
*u	*у	u	у	u			υ	u		э	u
*	*i i			i		i		i			

The four types of vowel changes resulted in four types of vowel harmony shifts:

- (a) Oirad (including Kalmuck) has retained the Old Mongolian vowel system, including palatal vowel harmony.
- (b) In the languages of the Monguor type (Monguor, Santa, Bonan, Moghol), velarization merged the front rounded vowels *y and *∅ with their vowel harmony alternants *u and *o. This destroyed the basis for vowel harmony, which was lost as a productive process in these languages.
- (c) In languages of the Mongolian type (Mongolian, Buriad, Kamnigan, Shira Yugur, Kangjia), the vowel shift resulted in a shift of the basis for vowel harmony from palatal to pharyngeal. Although velarization and pharyngealization took place to different extents in these languages, the vowels of alternation pairs $(a \sim e/y, z \sim o/\phi, v \sim u)$ are differentiated by the feature [pharyngeal].
- (d) In Dagur, the development went one step beyond this, by merger of the rounded back (>pharyngeal) vowels (*u > v > z; *o > z), and of the rounded front (>non-pharyngeal) vowels (*y > u, $*\phi > o > u$). Still, what remains of vowel harmony is based on the feature [pharyngeal].

10.6.2 Vowel harmony in Old Mongolian and Halh

Halh vowel harmony was described in section 5.2 and Old Mongolian vowel harmony in section 8.6. They differ from each other in several ways: (a) Old Mongolian vowel harmony was based on the feature [palatal] while Halh vowel harmony is based on the feature [pharyngeal]; (b) the vowel *i is completely neutral in Old Mongolian, but only partially neutral in Halh; (c) vowel harmony is cyclic in Halh but not in Old Mongolian; (d) Halh has rounding harmony, but Old Mongolian lacks it.

Using the vowel features introduced in section 5.1, the Old Mongolian vowels can be represented as in (14).

(14) Old Mongolian vowel features

-	-	fully specified	underspecified
back (non-palatal):	*a	[FO]	[O]
	*0	[VOR]	[OR]
	*u	[PVR]	[R]
front (palatal):	*e	[PO]	[PO]
	*ø	[POR]	[POR]
	*y	[PR]	[PR]
neutral:	*i	[P]	[]

In the full feature specification, front vowels are characterized by having the feature [palatal] as their only place feature, lacking the features [velar] and [pharyngeal]. We will, however, use the underspecified representations where the front vowels have the place feature [palatal], and the back vowels lack place features, so that front and back vowels differ only by the presence or absence of the feature [palatal]. This makes it possible to analyse back ~ front vowel harmony as spreading of the feature [palatal].

We represent Old Mongolian root morphemes so that the non-place features [round] and [open] are associated to the respective vowels, while the place feature [palatal] (if present) is attached to a root, but is not associated directly to the initial vowel (unlike the situation in Halh). This is because initial *i does not determine the palatality of the other vowels in a word. The feature [palatal] cannot be attached to the phonological representation of a suffix. When the place feature [palatal] is lifted off the non-initial vowels, they can be represented by each of the four combinations of the non-place features [open] and [round], forming the vowel *i and the archiphonemes *A, *O, *U:

(15)	archiphoneme				realization in front-vocalic		
			vocali	ic words	words		
	i	[]	[]	i	[P]	i	
	U	[R]	[R]	u	[PR]	у	
	А	[O]	[O]	a	[PO]	e	
	Ο	[OR]	[OR]	0	[POR]	ø	

The vowel shift changed only the place features, causing the harmonizing feature to shift from [palatal] in Old Mongolian to [pharyngeal] in Halh. The Old Mongolian marked vowels $*\phi$, *y became unmarked u, o in Halh, and the unmarked vowels *u, *o became marked v, o (16).

(16) Vowel harmony shift in Halh

Old Mongolia	n		Halh		
palatal:	*e	[PO]	non-pharyngeal:	e	[O]
	*ø	[POR]		0	[OR]
	*y	[PR]		u	[R]
non-palatal:	*a	[O]	pharyngeal:	a	[FO]
	*0	[OR]		э	[FOR]
	*u	[R]		υ	[FR]
neutral:	*i	[]	neutral:	i	[]

This is illustrated in (17), where the harmonizing feature [palatal] spreads in the Old Mongolian front-vocalic word **sehyl–eč^he* (17a), but not in the back-vocalic word **kačar–ač^ha* (17b); conversely, the Halh harmonizing feature [pharyngeal] spreads in *Gacr–as* in (b) but not in *suu* ξ –*es* in (a).

(17) Comparison of vowel harmony in Old Mongolian and Halh

Old Mongolian Halh (a) *sehvl-eč^he 'tail-ABL' suuk-es 'tail-ABL' $[P] \qquad [P] \\ s V h V 1 \rightarrow s V h V 1 - V \check{c}^{h} V$ [[P]] $s V V \mathfrak{B} - V s$ [O] [R] [R] [O] [0][R] [0] [0](b) *kačar–ač^ha 'ground–ABL' gacr-as 'ground-ABL' $[\mathbf{F}]$ [F] $\begin{array}{c|c} k \ V \ \check{c} \ V \ r - V \ \check{c}^h \ V \\ | \ | \ | \ | \end{array}$ $_{\rm GVcr} \rightarrow$ GVcr-Vs [O] [0]

Old Mongolian **i* is completely neutral: words with initial **i* can take either front or back vowels in the following syllables (e.g. **nityn* 'eye'; **č^hisun* 'blood'); the first vowel that is not an **i* decides the vowel harmony class. We assume that the harmonizing feature [palatal] is attached to the root morpheme, but not to any of its vowels. In contrast, all Halh words with initial *i* must have non-pharyngeal vowels in following syllables. Halh **i*-assimilation (10.7.3) eliminated all occurrences of **i* in the initial syllable of back-vocalic words (e.g. OM **č^hisun* 'blood' > Halh $c^h \upsilon s$ цус), and in many front-vocalic words as well (e.g. **nityn* 'eye' > *nut* нуд). This did not change the vowel harmony class of the words involved, but regularized vowel harmony in the sense that it made the presence or absence of the harmonizing feature [pharyngeal] in the initial vowel determine its presence or absence in all following vowels except non-initial *i*; it can be noted, however, that /*i*/ is non-

contrastively pharyngealized to [I] in pharyngeal words, except after palatalized and alveopalatal consonants (1.1.2). Thus we can assume that the harmonizing feature [pharyngeal] is attached to the initial vowel in Halh, as illustrated in (18).



Words with *i as the only vowel can be either front- or back-vocalic in Old Mongolian, but in Halh they must be non-pharyngeal. This is illustrated in (19), which shows that those (rather few) words which have *i as the only vowel and take backvocalic suffixes in Old Mongolian have changed the vowel harmony class in Halh. This can be seen as the consequence of the fact that [pharyngeal] became the harmonizing feature in Halh: since *i* does not have this feature, it does not spread, and suffix vowels will not contain it.

(19)	Words with i as the only vowel	
	Old Mongolian	Halh
	(a) $*k^{h}i-k^{h}y$ 'to do-futp'	xii–get 'to do–PFG'
	$\begin{cases} [P] & [P] \\ k^{h} V \rightarrow k^{h} V - k^{h} V \\ \end{vmatrix}$	
		x V V - g V t
	[R]	[O]
	(b) *nis-ka-pa 'fly-CAUS-PST'	nis-et 'fly-pfg'
	$ n V s - k V - p V $ $ \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad $	n V s - V t
	[0] [0]	[0]

In section 5.2.6 we showed that the Halh harmonizing feature [pharyngeal] spreads from the first vowel of a word by a local rule which is cyclic in relation to the

morphology. There are no indications that Old Mongolian vowel harmony is cyclic, and it is possible to analyse it with a global rule that spreads the harmonizing feature [palatal] from the root throughout the word after all suffixes have been attached. The development of cyclicity in Halh vowel harmony is due to the development of rounding harmony (see 10.7.1) and to a segmental rule which realizes the expected diphthong *[oi] as [e]; see section 5.2.6.

10.7 VOWEL ASSIMILATION

Several different vowel assimilation processes took place in the modern Mongolic languages. The most widespread one is assimilation of *i in the initial syllable to the vowel of the following syllable. In languages, such as Halh, which have developed palatalized consonants, the assimilated *i is often retained as palatalization of the preceding consonant. Ramstedt (1902: 45) used the term 'breaking' (*brechung*) for the combination of *i-assimilation and consonant palatalization (10.11.1). Dagur, which is the only Mongolic language that has developed labialized consonants, has a parallel process which assimilates *u to a following *a, leaving labialization as a trace on the preceding consonant (10.7.5).

10.7.1 *Progressive rounding assimilation* and the development of rounding harmony

Although we reconstruct Old Mongolian without rounding assimilation or rounding harmony (8.6.1), some rounding assimilation took place already in the Old Mongolian sources, and in several modern languages (Mongolian proper, Buriad, Kamnigan, Shira Yugur, Dagur, and Kangjia) this led to the development of more or less regular rounding harmony, affecting both root and suffix vowels. Oirad, which did not shift the basis for vowel harmony from palatal to pharyngeal, has not developed rounding harmony, except in the Alshaa and Hoshuud dialects (Sun et al. 1990; Gereltü 1991; 1992). The remaining languages do not have productive vowel harmony of any kind, but rounding assimilation took place in some words where the first vowel was *o or $*\phi$ and the second was *a or *e (20). The non-open rounded vowels *u and *y did not trigger rounding assimilation in any modern or ancient Mongolic language.

10.7.2 Regressive rounding assimilation¹⁰

An initial **e* is usually rounded by a following **y* in Mongolian proper where it appears as *o*, in Kalmuck where it became ϕ , and in Buriad and Kamnigan where the reflex is *u*. It usually remained unrounded in the other modern languages. Examples are OM **emys* 'to wear', **t*^h*emyr* 'iron', which became *oms* ΘM , *t*^h*ompr*

¹⁰ Qaserdeni (1980a); Kögjiltü (1987a); Möngkebuyan (2001).

(20)Progressive rounding assimilation

OM	*a	*aha	*ai	*e	*ehe
Hlh	Ø	э	oi	Ø	(0)
Cha	Ø	э	æ	Ø	(0)
Baa	Ø	э	э	Ø	(0)
Bur	э	33	oi	e	(00)
Kmn	э	(33)	(ji)	u	(ee)
Klm	Ø	а	а	Ø	(3)
Dag	Ø	(a, 5)	Ø	Ø	(y)
ShY	э	33	oi	ø	(ee)
Mgr	0	00	oi	0	
San	0	0	ei	ie	
Bon	0	0	wi	0	
KJ	σ	υ	vai	u	
Mog	(a, o)		(ai, oi)	(a)	

The table shows the development of non-initial vowels in words with initial *o or $*\phi$.

Examples (9.13): */a/: *k^hola, *ora, *olan: */aha/: *nokahan, *tolahan, *t^hokahan: */ai/: *mokai, *nok^hai: */e/: *k^høk^he, *nøk^her, *ønke: */ehe/: *kørehesvn.

төмөр in Halh, oms, t^homər in Chahar and Baarin, umdə, t^humər in Buriad, umut, $t^{h}umur$ in Kamnigan, and ϕms , $t^{h}\phi mr$ in Kalmuck.

10.7.3 Regressive *i-assimilation¹¹

The vowel *i was completely neutral in Old Mongolian vowel harmony, and could occur in the initial syllable of both front-vocalic and back-vocalic words. All occurrences of *i in the initial syllable of Old Mongolian back-vocalic words were eliminated in Halh by assimilation to the vowel of the following syllable (e.g. OM *k^hilpar 'easy' > Halh x^jagpər хялбар; *k^himusun 'nail' > хот хумс; *sink^hor 'falcon' > $\delta z \eta x \partial r$ inotation). The result is that *i* never occurs in the first syllable of a pharyngeal word in Halh. Initial *i was often assimilated to a following *y (> u)as well (OM **čiryken* 'heart' > *curx* 3ypx). When the assimilating vowel is *a, assimilation is often accompanied by palatalization of the consonant that precedes **i* (see 10.11.1).

Assimilation of *i is also common in Chahar, but in some pharyngeal words, initial i has not been eliminated by assimilation, but by pharyngealization, splitting the */i/ phoneme into /i/ in non-pharyngeal and /1/ in pharyngeal words (see 10.2). There was a similar development in Baarin, where ε corresponds to Chahar *i*.

Complete assimilation of initial *i to the vowel of the following syllable has taken place in some words in all the other modern Mongolic languages as well, but

¹¹ For sources, see App. J, 10.7.3, p. 229.
the number of words which have undergone this process varies a lot from language to language. The development of *i before different vowels is summarized in (21).

(21) *Reflexes of *i before different vowels in the second syllable*

	*Ø/i	*a/aha	*u/uhu	*ahu	*e/ehe/ihe	*y	*0
Hlh	i	а	υ	υ	i	u	э
Cha	i	а, 1	υ	υ	i	u	(3)
Baa	i	а, ε	υ	υ	i	u	(3)
Bur	e	a, i	U	υ	e	u	э
Kmn	i	i	(U)	i	i	i	(3)
Klm	i	a, i	u	0	i	у	0
Dag	i	a, i		э	i	u	
ShY	Ŷ	a			e		(y)
Mgr	i	i	i		i		
San	i	i	u		i	u	
Bon	i	i, a			i		(i)
KJ	i	i		(i)	i	i	
Mog	i				i	u	(i)

Vowel in the second syllable

Examples (see 9.13): */i/: see (3); */a/: * $\check{c}ikasun$, * $\check{c}ita$, * $k^{h}ilpar$, * $mik^{h}an$, *sira; */aha/: *imahan; */u/: * $\check{c}^{h}isun$, * $k^{h}imusun$; */uhu/: *niruhun; */ahu/: * $\check{c}^{h}ilahun$, *sipahun; */e/: * $hi\check{c}^{h}e$, *ire, *ite; */ehe/: *inehe; */ihe/: *sirihe; */y/: * $\check{c}iryken$, *nityn, *sityn; */o/: * $\check{c}^{h}ino$, * $sink^{h}or$.

Assimilation is common in Buriad, but some initial *i* remain in pharyngeal words, for example, $x^{j}izar$ 'border', $x^{j}ilb\partial r$ 'easy', $n^{j}ilx\partial$ 'baby' (Halh $x^{j}acgar$ хязгаар, $x^{j}a\underline{k}p\partial r$ хялбар, $n^{j}a\underline{k}x$ нялх). Assimilation occurs only in a few words in Kamnigan and Moghol, and initial **i* is often retained in Santa, Monguor, Bonan, and Kangjia as well.

There are also some cases of progressive assimilation of *i in non-initial syllables (see Poppe 1963).

10.7.4 Breaking of word-initial *i

When a word-initial **i* (i.e. one that was not preceded by a consonant) was assimilated to a back vowel, the result is the onset *j* plus the assimilated vowel in Halh and some other dialects of Mongolian proper, as well as in Buriad, Oirad, and sometimes Dagur. For example, OM **imahan* 'goat' became *jama* in Halh (ямаа), Chahar, and Baarin, *jaman* in Buriad and Kalmuck. Another example is OM **ilka* 'to separate' > Halh *jagəg* ялга. There was a similar development for the initial group **hi*: OM **hiruhar* 'bottom', **hiryher* 'blessing' became *jərəţ* ёроол, *joroţ* ерөөл in Halh and *joral*, *jørɛl* in Kalmuck.

10.7.5 Breaking of *u in Dagur

The normal Dagur reflex of Old Mongolian **u* is *z*, but when it was followed by an **a* in the next syllable it is often, but not always, assimilated to this vowel, and the preceding consonant is labialized (or a *w* appears if **u* is word-initial). Examples are *t*^want 'middle', *k*^warpə 'three', *want*^h 'to sleep' (OM **tumta*, **kurpan*, **unt*^ha). The same development took place with **o* in some words as well (e.g. in *walən* 'many', OM **olan*). In some Dagur words, **y* was broken to *wr*, for example, OM **yile* > *wril* 'deed' and OM **yje* > *wrj* 'joint' (see Poppe 1930*a*: 110; 1955: 31; Todaeva 1986: 11f.; Janhunen 1990*b*; Kuribayashi 1993). This process is analogous to **i*-assimilation and consonant palatalization ('**i*-breaking') in Halh, and is connected with the development of a series of labialized consonant phonemes in Dagur, unique for a Mongolic language.

10.8 ONSET CONSONANTS

In this section we describe the development of the Old Mongolian consonants in syllable onset position. Coda consonants are treated in section 10.9. The development of the consonants is also affected by deaspiration (10.10) and palatalization (10.11). A summary of the development of the onset consonants in environments without palatalization or deaspiration is given in (22).

(22) Development of onset consonants

The velar and uvular allophones of */k/ and $*/k^h/$ are shown separately. A slash separates reflexes in word-initial/word-medial position. The consonants *m, *n, *r, *j, which remain unchanged in all modern languages, are not shown in the table.

OM	*t ^h	*č ^h	*[k ^h]	$*[q^h]$	*p	*t	*č	*[k]	*[q]	*s	*h	*1
Hlh	t ^h	c^h	х	х	p/w	t	с	g	G	s	Ø	ե
Cha	t ^h	čh	Х	х	р	t	č	k	k	S	Ø	1
Baa	t ^h	čh	х	х	р	t	č	k	k	s	Ø	1
Bur	t ^h	s	х	х	b	d	z	g	g	h	Ø	1
Kmn	t ^h	c^h	k ^h	k ^h	р	t	с	k	k	h	Ø	1
Klm	t ^h	c^h	k ^h	х	b/w	d	z	g	G	s	Ø	1
Dag	th	čh	k ^h , x/k	x/k	p/w	t	č	k	k	s	x/Ø	1
ShY	th	čh	k ^h	x/k ^h	p/w	t	č	k	k	s	h/Ø	1
Mgr	th	čh	k ^h	х	p/w	t	č	k	q	s	x/Ø	1
San	th	čh	k ^h	q ^h , q/q	p/w	t	č	k/q	q ^h , q/q	s	x/Ø	1
Bon	th	čh	k ^h	x, q/q	w, p	t	č	k	x, q/q	s	h/Ø	1
KJ	th	čh	k ^h	x/k	p/w	t	č	k	x, k/k	s	h/Ø	1
Mog	t	č	k	q	b	d	ĭ	g	G	s	Ø	1

Examples (9.13): */t^h/: *t^hapun, *t^hani, *t^haulai, *alt^han, *ert^he, *umart^ha; */č^h/: *č^hak, *č^hakahan, *č^herik, *kakč^ha, *k^hakač^ha, *nič^hykyn; */k^h/ [k^h]: *k^helen, *k^hen, *k^høl, *k^hyr, *hyk^her, *nek^he, *nek^hei, *nøk^her, *yk^hy; */k^h/ [q^h]: *k^halahun, *k^hara, *k^honin, *k^huruhun, *ak^ha, *mik^han, *nok^hai; */p/: *pahu, *pari, *peri, *epesyn, *sipar, *t^hapun; */t/: *tahun, *tehere, *tørpen, *mete, *nityn, *sityn; */č/: *čahu, *čahun, *čaru, *kačar, *yče, *yčyhyr; */k/ [k]: *kem, *ker, *kerel, *kørehesyn, *hyneken, *ønteken, *øŋke; *k [q]: *kačar, *kal, *kar, *kurpan, *čikasun, *č^hakahan, *mokai; */s/: *sahu, *sain, *sara, *č^hasun, *jasun, *usun; */h/: *harpan, *hulahan, *hynesyn, *ahula, *čahu, *tehere; */m/: *mete, *morin, *motun, *emys, *aman, *naiman; */n/: *nahat, *narin, *nok^hai, *ene, *hyneken, *inehe; */l/: *hulahan, *k^halahun, *k^helen; */r/: *ire, *k^hara, *sira; */j/: *japu, *jasun, *josun, *huja, *pajan, *peje, *yje.

10.8.1 Stops

According to our reconstruction, Old Mongolian had two series of stops and affricates, voiceless aspirated and voiceless unaspirated. The Old Mongolian aspirated ~ unaspirated contrast is retained as such in most modern languages. It was replaced by a voiceless ~ voiced contrast in Moghol, and the unaspirated stops seem to have become voiced in Buriad and Oirad. The velar/uvular stops often became fricativized.

In Halh, *p was retained word-initially and after *l, *m, and *p (which became ξ , m, w in Halh). In other positions, it was fricativized to w. Similar developments took place in Oirad, Dagur, Shira Yugur, Monguor, Santa, Bonan, and Kangjia (Cenggeltei 1988; Qasbagan-a 2000a).

The dental stops are retained in all modern languages.

The Old Mongolian velar stops k^h , k^h had the uvular allophones $[q^h]$, [q] in back-vocalic words (see 8.6.2). The velar and uvular allophones are kept phonetically distinct in all modern languages, and their phonemic status depends on the development of the vowel system. Generally speaking, they became contrasting phonemes in those languages where back and front vowels merged ($y > u, \phi > o$), but remained allophones in the others, where the Old Mongolian seven-vowel system is kept structurally intact. The situation can be illustrated by comparing Moghol with Mongolian proper (23). Although the reflexes of k^h and q^h are retained as velar and uvular consonants in both languages, the vowel merger in Moghol made the uvulars and velars contrast, but they remained allophones conditioned by the pharyngeality of the following vowel in Mongolian.

(23)		'far'	'foot'	'finger'	'to reach'
	Old Mongolian	*k ^h ola	*k ^h øl	*k ^h uruhun	*k ^h yr
	allophones	[q ^h ola]	[k ^h øl]	[q ^h uruhun]	[k ^h yr]
	Moghol	qolo	kol	qurun	kur
	Halh	хэд [хэд]	хођ [хөђ]	χυτυ [χυτυ]	xur [xur]
		ХОЛ	хөл	хуруу	хүр

Although those languages which have not merged front and back rounded vowels usually retain the velars and uvulars as allophones, the situation is complicated by fricativization, which affected most modern languages (Buraev 1977), and created different reflexes for $*[k^h]$ and $*[q^h]$ in Oirad and several other modern languages. The pattern of mergers is complicated and differs from language to language; see (24) for an overview.

(24) Mergers of velar and uvular stops

A slash separates reflexes in word-initial/word-medial position.

OM	*[k ^h]	*[q ^h]	*[q]	*[k]
Cha, Baa	x		k	
Bur	x		g	
Kmn	k ^h		k	
Hlh	2	K	G	g
Dag	k ^h , x/k	x/k		κ
ShY	k ^h	x/k ^h		κ
Klm	k ^h	x	G	g
Mgr	k ^h	x	Q	k
Mog	k	q	G	g
KJ	k ^h	q ^h ,	x, k/k k	
San	k ^h		^h , q/q k/	
Bon	k ^h		x, q/q k	

The original state with an aspirated and an unaspirated stop, which have velar and uvular allophones, is retained in Kamnigan. Mongolian and Buriad retained the system, but changed the aspirated stop to a fricative x (with the uvular allophone $[\chi]$ in pharyngeal words). In Halh, the unaspirated stop *k became voiced. In originally front-vocalic words it became velar g and in originally back-vocalic words it became uvular G. The picture is, however, complicated by the fact that syllable-final *k developed to g both in front- and back-vocalic words, making velars and uvulars contrast in pharyngeal words in Halh after short word-final vowels were deleted (see (34) below).

In Oirad, k^h became a fricative x only in back-vocalic words, and it remained a stop in front-vocalic words. In words where a vowel has been palatalized (10.11.2), fricativization took place before palatalization, so that k^h and x may contrast (25).

(25)		'far'	'foot'	'sheep'	'back'
	Old Mongolian	*k ^h ola	*k ^h øl	*k ^h oni/n	*k ^h oina
	allophones	[q ^h ola]	[k ^h øl]	[q ^h oni]	[q ^h oina]
	fricativization	xola	—	xoni	xoina
	vowel palatalization	—	—	xøni	xøøna
	vowel deletion	xol	k ^h øl	xøn	xøøn
	Kalmuck	xol	k ^h øl	xøn	xøøn

DEVELOPMENT OF THE MONGOLIC LANGUAGES

In medial position, all uvular/velar stops merged to [y] in Dagur, analysed as an allophone of /k/ by us. Santa, Bonan, and Kangjia merged * $[q^h]$ and *[q], but there was a split, so that the reflex is either q^h , x, x in the three languages (26a), or q, q, k [q] (b). What caused this split is not known. The stops are realized as fricatives in word-medial position.

(26)	(a)	Old Mongolian *k ^h ara *k ^h alahun *kar	'black' 'warm' 'hand'	Santa q ^h ara q ^h aluŋ q ^h a	Bonan xara xoloŋ xar	<i>Kangjia</i> xara xulə xar
	(b)	*kal *k ^h ojar *k ^h uruhun *kačar *kurpan	'fire' 'two' 'finger' 'ground' 'three'	q ^h aŋ qua quru qača quraŋ	xal quar quruu qačar quraŋ	xar kuar kuru kačar kuro

10.8.2 Affricates¹²

The Old Mongolian affricates $*\check{c}$ and $*\check{c}^{h}$ are retained as alveopalatal affricates in most modern languages, but they were affected by two phonological processes, fricativization and depalatalization, in Oirad, Buriad, and Mongolian. Fricativization changed the affricates to the corresponding fricatives, and depalatalization made the alveopalatal affricates dental except before *i. Different dialects underwent these processes to different extents. Halh has no fricativization but depalatalization, with the result that $*\check{c}$ and $*\check{c}^{h}$ were retained only before *i. In other positions, they became dental *c* and *c*^h. This process together with *i-assimilation created a contrast between alveopalatal and dental affricates:

(27)		'hundred'	'spear'	'time'	'to boil'
	Old Mongolian	*čahu/n	*čita	*č ^h ak	*č ^h ina
	depalatalization	cahu		c ^h ak	—
	*i-assimilation		čata		č ^h ana
	segmental changes	CUU	—	chag	—
	short vowel deletion		čat		č ^h an
	Halh	соо зуу	čat жад	с ^h ag цаг	č ^h an чана

In Kamnigan, the reflexes of $*\check{c}$, $*\check{c}^h$ are \check{c} , \check{c}^h before *i (which is retained as i), and c, c^h elsewhere. This can be analysed as phonemes /c/ and /c^h/, with the palatalized allophones [\check{c}] and [\check{c}^h] before i. All affricates were fricativized in Buriad, and there is fricativization of $*\check{c}^h$, but not of $*\check{c}$, in the Horchin dialect of Mongolian (Bosson and Unensečen 1962).

Different Oirad dialects differ with respect to depalatalization and fricativization. In Kalmuck and in Xinjiang Torguud, only $*\check{c}$ was fricativized, becoming *z* except before *i. Alshaa (Cogtu 1985) and Hoshuud has depalatalization but no fricativi-

200

¹² Todaeva (1960*a*); Cenggeltei (1979*b*); Clark (1979); Rassadin (1981); Gantogtoh (1982); Badgaev (1987; 1988*a*, *b*; 1989*a*, *b*, *c*; 1990).

zation, the result being the same as in Halh. According to Kara (1958), z and c alternate in Oirad of Western Mongolia. The situation is summarized in (28).

(28) Development of Old Mongolian affricates

A slash separates reflexes before *i and in other positions

	depalatalization	frica of *		frica of *∂	tivization
Buriad	+	+	š/s	+	ž/z
Bargu	+	+	š/s	-	č/c
Kalmuck	+	-	č ^h /c ^h	-/+	j∕z
Torguud	+	-	č ^h /c ^h	-/+	č/z
Halh, Alshaa, Hoshuud	+	-	č ^h /c ^h	-	č/c
Kamnigan	+	-	c ^h [č ^h /c ^h]	-	c [č/c]
Horchin	-	+	š	-	č
Other languages	_	_	č ^h	-	č

These developments are exemplified in (29) with examples from our standard sources, or Sun et al. (1990).

(29)		'stone'	'time'	'spear'	'to bite'
	Old Mongolian	*č ^h ilahun	*č ^h ak	*čita	*čahu
	Buriad	šulun	sag	žadə	ΖŬŬ
	Bargu	รับไบ	sak	čat	CUU
	Kalmuck	č ^h olun	chag	jid	zu
	Torguud (Xinjiang)	č ^h olun	c ^h ak	čit	zuu
	Halh	č ^h ʊ�ʒʊ чулуу	с ^h ag цаг	čat жад	соо зуу
	Alshaa	č ^h ulu	c ^h ak	čit	cuu
	Hoshuud (Qinghai)	č ^h ulu	c ^h ak	čɛt	cuu
	Kamnigan	c ^h iləə [č ^h iləə]	c ^h ak [c ^h ak]	cita [čita]	cəə [cəə]
	Horchin	šʊlʊ	šak	čɛt	čυυ
	Chahar	č ^h ulu	č ^h ak	čıt	čυυ
	Baarin	č ^h ulu	č ^h ak	čɛt	čυυ

There is a fair number of words where especially $*\check{c}$ but also $*\check{c}^h$ were depalatalized in Halh, Buriad, and Kalmuck even when they occurred before Old Mongolian *i(30). Presumably, *i-assimilation took place before depalatalization in such words. (See also 10.11.1 for consonants that fail to palatalize under similar conditions.)

(30)	ОМ		Halh		Buriad	Kalmuck
	*čiha	'to point'	caa	заа	zaa	za
	*čikasun	'fish'	cagəs	загас	zagəhən	zagsn
	*čirkohan	'six'	curga	зургаа	zurgan	zurgan
	*k ^h ečije	'when'	xice	хэзээ	xeze	k ^h eze
	*č ^h isun	'blood'	chus	цус	šuhən	c ^h usn
	*eč ^h ike	'father'	ic ^h əg	эцэг	esəgə	ec ^h k ^h

10.8.3 Fricatives¹³

Old Mongolian */s/ had the allophones *[š] before **i* and *[s] in other positions. This pattern is usually preserved in the modern languages, in the sense that *[š] occurs before reflexes of Old Mongolian **i*, and *[s] occurs elsewhere. The allophone *[š] became phonemic by **i*-assimilation in all modern languages except Kamnigan; examples (see 9.13) are: **sihe*, **sini*, **sira*, **sityn*, **masi*, **kasihun*. The development is exemplified in (31) with the Old Mongolian pair **sara* 'moon' ~ **sira* 'yellow', which became *sar* ~ *šar* in Mongolian, Oirad, and Dagur (31).

(31)	Old Mongolian	*sara 'moon'	*sira 'yellow'
	allophonic variation	—	[šira]
	*i-assimilation	—	šara
	deletion of short vowels	sar	šar
	Halh	sar cap	šar шар

There is a special development in Buriad, where *s is weakened to h except before *i. For example, the words 'moon' and 'yellow' are har - sar in Buriad. The same development took place in (Urul'ga) Kamnigan. Since *i is normally not assimilated in Kamnigan, \check{s} occurs only before i and can be regarded as an allophone of h; the two above-mentioned words are hara - hira [šira] in Kamnigan. The sibilant s was reintroduced in Buriad by fricativization of \check{c}^h , creating a contrast between h, s and \check{s} (32).

(32)

	'moon'	'yellow'	'time'	'to boil'
Old Mongolian	*sara	*sira	*č ^h ak	*č ^h ina
allophonic variation	_	[šira]	_	_
depalatalization		_	c ^h ak	
* <i>i</i> -assimilation		šara		č ^h ana
sibilant weakening	hara			
fricativization	—	—	sak	šana
other segmental changes	harə	šarə	sag	šanə
Buriad	harə	šarə	sag	šanə

In some areas of the Jalaid dialect as well as in Gorlos, and in the language of the mongolized Dörbed of Heilongjiang province, *s became t^h (except before *i): *sahu > $t^{h}\upsilon\upsilon$ 'to sit', *sehyl > $t^{h}uul$ 'tail', *usun > υt^{h} 'water' (Rudnev 1911: 180f.; Ramstedt 1957: 70; Bao Xianglin 1985; Zhang and Shaobu 1990; Böke 1994; Chen 1995b: 664f.). This development took place in some Eastern Buriad dialects as well (Rassadin 1982: 80; Darbeeva 1996: 84).

Initial *h disappeared in Mongolian, Buriad, Kamnigan, Oirad, Moghol, and Hailar Dagur, but is retained in the other Mongolic languages. Its reflex is h in Shira Yugur and Bonan, x in Buthaa Dagur, Santa, Monguor, and Kangjia (in some of these languages with the allophone [f] before rounded vowels). It is palatalized to š before *i in Santa, Monguor, Bonan, and Kangjia (see * $hic^{h}e$ in 9.13 for an example).

¹³ Clark (1979); Boosiyang (1981); Badgaev and Omakaeva (1986); Buraev (1988*a*, *b*).

Non-initial h disappeared in all modern languages, normally causing the surrounding vowels to coalesce to one long vowel; see section 10.3.

10.8.4 Sonorants

The Old Mongolian sonorants, that is, the nasals *m and *n, the liquids *l and *r, and the glide *j (Ramstedt 1915), are very stable in syllable onset position and are normally retained unchanged in all modern Mongolic languages. Groups of the form *ijV became long vowels (10.3).

The change of phonetic realization from *l to b seems to be unique for Halh. Its exact distribution in the Halh dialects is not known. As far as we know, it did not take place in other Mongolian dialects, nor in other Mongolic languages.

10.9 CODA CONSONANTS

As seen in section 8.7.4, only the unaspirated voiceless stops, the fricative *s, and the sonorants occur as codas. The only Old Mongolian consonant that can be a coda but not an onset is the velar nasal $*\eta$. The nasal *n and the liquids *l and *r are much more frequent than other codas.

The development of the Old Mongolian syllable codas is summarized in (33). Although the individual coda consonants were fairly stable in most languages, the system of coda consonants changed considerably from Old Mongolian to most modern languages, in particular Mongolian proper, due to changes of the word and syllable structure (see 10.12).

(33) Development of coda consonants

N = nasal whose place of articulation is assimilated to a following obstruent.

OM	*p	*t	*[k]	*[q]	*s	*m	*n	*ŋ	*N	*1	*r
Hlh	w	t	g	g	S	m, N	ŋ	ŋ	Ν	ե	r
Cha	р	t	k	k	s	m, N	ŋ	ŋ	Ν	1	r
Baa	р	(t)	k	k	S	m, N	n	ŋ	Ν	1	r
Bur	b	d	g	g	d	m, N	n	n	n	1	r
Kmn	р	t	k	k	t	(m)	n		Ν	1	r
Klm	w	d	g	g	S	m	n	ŋ	Ν	1	r
Dag	u	(t)	(k)	r	s	(m)	n	(n)	n	1	r
ShY	р	t	k	k	S	m	n	ŋ	ŋ	1	r
Mgr	р	(t)	q	q	S	m	n	ŋ	Ν	1	r, š
San	Ø	(Ø, t)	ø	Ø	S	(n, ŋ)	ŋ	ŋ	Ν	ŋ, n	Ø
Bon	р	(Ø, t)	(k,x)	q	š	m	ŋ	(ŋ)	m	1	r
KJ		(Ø, t)	u		S	n	(3)		Ν	r	r
Mog	(b)	(t)	(g)		s		n			1	r

Examples are (see 9.13): */p/: *ap, *č ϕ p, *napč^hin; */t/: *k^hit^hat, *morit, *nahat, *setk^hil; */k/ [k]: *č^heč^hek, *č^herik, * ϕ k, *pič^hik; */k/ [q]: *alak, *č^hak, *pulak; */s/: *emys, *nis, *pos; */m/: *kem, *sam, *ertem, *k^hamt^hu, *nimken, *tumta; */n/: *č^hakahan, *hulahan, *k^halahun, *k^hen; */ŋ/: *tok^halaŋ, *tyhyreŋ, *maŋlai; */N/: *k^hynty, *ønteken, *yntyr, *k^høŋken, *øŋke; */I/: *čil, *kal, *k^høl, *alt^han, *čalki; */r/: *kačar, *kar, *ytyr, *ert^he, *kurpan, *tørpen, *umart^ha.

Unaspirated voiceless *p, *t, *k are the only stops that occur in Old Mongolian syllable codas. In most modern languages they developed in the same way as in onset position, but there are a few special developments. In Halh, coda *p became w, as did onset *p in word-medial position. In Santa, all coda stops were deleted, and in Dagur, *k > r in a number of words (cf. Todaeva 1986: 33ff.; Serbak 1992–3).

In Halh, coda *k became g in both non-pharyngeal and pharyngeal words, while onset *k became g in non-pharyngeal words and σ in pharyngeal words. Combined with deletion of short non-initial vowels, this led to a contrast between velar g and uvular σ in pharyngeal, but not in non-pharyngeal words (34) (see also 3.2.2). A similar development took place in Kalmuck, where the words exemplified in (34) are arg, alg, yg, ϕg .

(34)	Old Mongolian	*arka	*alak	*yke	*øk
		'method'	'motley'	'word'	'to give'
	allophones	[arqa]	[alaq]	[yke]	[øk]
	velarization of coda *[q]	—	alak		
	other segmental changes	arga	alag	uge	og
	vowel deletion	arG	alzg	ug	
	vowel epenthesis	arəg	abəg		
	Halh surface form	arәс арга	ађәд алаг	ug үг	og өг

Old Mongolian coda *s became d in Buriad, t in (Urul'ga) Kamnigan, and \check{s} in Bonan, but is retained in the other languages. Santa does not have s in syllable codas, but retains Old Mongolian coda *s by adding a vowel after it.

The liquids *l and *r are usually retained in coda position, except in Santa, where *l became a nasal and *r was deleted. In Kangjia and in Minhe Monguor (Jagunasutu and Li 1982), *l and *r merged to r.

In Huzhu Monguor, coda **r* developed to š before **t*^h: OM **umart*^h*a* > *muštaa* 'to forget'; **ert*^h*e* > š*te* 'early'; **urt*^h*u* > š*tur* 'long'. This suggests that OM **t*^h was preaspirated, devoicing the preceding **r* (see also 10.10 below).

In Mongolian, Oirad, Dagur, and Shira Yugur, coda **l* disappeared before **s* (Poppe 1954*b*); see the examples **mølsyn* 'ice' and * $\mathcal{E}^{h}ahalsun$ 'paper' (Halh *mos* Moc, $c^{h}aas$ цаас) in 9.13. If the **l* is not in coda position but separated from the **s* by a vowel, it does not disappear: * $k^{h}ølesyn$ 'sweat', Halh $xo\xi s$ xonc.

There were three coda nasals in Old Mongolian, *m, *n and *y. In word-medial position, *n and *y did not occur before non-homorganic stops, but *m did. In the modern languages, there is a tendency for *m to assimilate to the place of articulation of a following obstruent. All three nasals occur in word-final position. Word-final *m is retained except in Santa and Kangjia, but merger of *n and *y has taken

place in word-final position in many modern languages, including Halh and Buriad. Eastern Mongolian (including Baarin), Ordos, Oirad, and Shira Yugur have kept final *n and *y apart, however (35) (see also Bulucilagu 1960; 1998; Böke 1985; Lutajirgal 1990; Pèljei 1992).

(35) Word final *n and *n

OM	*k ^h alahun 'warm'	*tyhyreŋ 'full'
Hlh	хадиŋ халуун	tuurəŋ дүүрэн
Bur	xalon	duurən
Baa	xalon	tuurəŋ
Klm	xalun	dyynŋ
ShY	xaluun	tyreŋ

The contrast was reintroduced in Halh by the deletion of final short vowels:

Old Mongolian	*k ^h en 'who'	*ene 'this'
velarization of n	k ^h eŋ	—
final vowel deletion		en
other changes	xiŋ	in
Halh	хіŋ хэн	in энэ
	velarization of <i>n</i> final vowel deletion other changes	velarization of <i>n</i> k ^h eŋ final vowel deletion — other changes xiŋ

10.9.1 Unstable *n

Many Old Mongolian nouns had a vacillation between a subject form ending with *n (e.g. $*\check{c}^{h}isun$ 'blood') and a form without *n ($*\check{c}^{h}isu$) used as an indefinite object. In Halh, this *n was usually lost in the nominative ($c^{h}\upsilon s$ цус 'blood'), accusative ($c^{h}\upsilon sig$ цусыг), instrumental ($c^{h}\upsilon sar$ цусаар), and comitative ($c^{h}\upsilon snt$ цустай) while it is retained in the genitive ($c^{h}\upsilon sni$ цусны), dative ($c^{h}\upsilon snt$ цусанд), and ablative ($c^{h}\upsilon snas$ цуснаас).

Adjectives normally retain final *n (which became η in Halh, e.g. $*nokahan > n 2G2\eta$ HOFOOH 'green'), as do attributive nouns ($c^h \upsilon s \partial \eta$ цусан 'blood–'). Monosyllabic nouns (e.g $*hon > 2\eta$ OH 'year') and a few others, such as $*\phi k^h in > 2x^j \partial \eta$ OXUH 'girl', have also retained *n. In many modern languages, analogy has created apparent reflexes of *n in words where Old Mongolian did not have it. The different modern languages have dealt with unstable *n in different ways, but since this primarily is a morphosyntactic and semantic problem and not a phonological one, we will not go into it further. Many examples can be found in the comparative vocabulary (9.13).

10.10 DEASPIRATION AND RELATED PROCESSES¹⁴

Some Mongolic languages have eliminated the occurrence of two aspirated consonants in a word by deaspirating one of them in a way similar to Grassmann's

¹⁴ For sources, see App. J, 10.10, p. 229.

(1863) law in Indo–European. In addition to the aspirated stops and affricates (* t^h , * k^h , * ξ^h), the voiceless fricative *s also triggers deaspiration (although it is not changed by it).

In all dialects of Mongolian proper, except Northern Halh and Eastern Mongolian, an initial aspirated stop or affricate ($*t^h$, $*c^h$, $*k^h$) became unaspirated if the following syllable onset is an aspirated stop or affricate, or the voiceless fricative *s. Mongolian deaspiration is exemplified in (37a) with Chahar; the same pattern occurs, for example, in Ordos (Sun et al. 1990) and Naiman (Mönggöngerel 1998). Standard Halh forms are given for comparison. There is no deaspiration if the two aspirated consonants are separated by more than a short vowel (37b); note that x is the regular reflex of $*k^h$. This suggests that Old Mongolian had preaspiration, similar to that in Halh, making it difficult to realize both post- and preaspiration with devoicing of both the initial and the final part of one short vowel in words such as $*/t^hat^ha/[t^ha^hta]$, while a longer distance gives more space for the realization of two consecutive aspirations (see Svantesson 2003*a*). The deaspirating effect of *s can be related to the fact that [s] is aspirated and may devoice the final part of the preceding vowel in Halh (see 2.2).

Although Ulaanbaatar Halh does not have phonological deaspiration, postaspiration is significantly shorter for initial aspirated stops in those positions where Chahar and other dialects have deaspiration, as was pointed out already by Ramstedt (1902: 12f.). Our measurements (Karlsson and Svantesson 2002) showed that Halh postaspiration (VOT) is only about two-thirds of its normal value in these positions.

	OM	Chahar	Halh	
(a) 'to pull'	*t ^h at ^h a	tat^h	thath	тат
'fat'	*t ^h osun	təs	t ^h əs	тос
'ear'	*č ^h ik ^h in	čix	čhix	чих
'blood'	*č ^h isun	čus	c ^h us	цус
'China'	*k ^h it ^h at	kıt ^h at	x ^j at ^h ət	Хятад
'blue'	*k ^h øk ^h e	kox	XOX	хөх
(b) 'paper'	*č ^h ahalsun	č ^h aas	chaas	цаас
'old'	*k ^h auč ^h in	xuuč ^h əŋ	xuuč ^h əŋ	хуучин
'cold'	*k ^h øit ^h en	xiit ^h əŋ	xuit ^h əŋ	хүйтэн
'sweat'	*k ^h ølesyn	xols	xolzs	хөлс
'together'	*k ^h amt ^h u	xamt ^h	xamt ^h	хамт

(37) Deaspiration in Mongolian dialects

In Monguor, the second of two aspirated consonants is deaspirated (38a). Here as well, *s, and also *h, trigger deaspiration (b). This kind of dissimilation also takes place in Santa and Bonan, although not as regularly as in Monguor, and in a few words in Kangjia and Shira Yugur. (Words with deaspiration are italicized.)

206

		OM	Mgr	Santa	Bonan	Kangjia	ShY
(a)	'to pull'	*thatha	t ^h ita	sta	šta	stha	ht ^h a
	'together'	*k ^h amt ^h u	xamti	xant ^h u	hamtui		xamty
	'China'	*k ^h it ^h at	č ^h itar	q ^h utei	hti	xthe	k ^h ʊt ^h at
	'to bark'	*k ^h uč ^h a	xoča	q ^h uča	hča	k ^h ʊč ^h a	k ^h uč ^h a
	'old'	*k ^h auč ^h in	xaučin	q ^h uaič ^h eŋ	xiič ^h aŋ	xuaič ^h o	xuuč ^h yn
	'ear'	*č ^h ik ^h in	č ^h iki	čhiqeŋ	č ^h ixaŋ	čhixo	č ^h k ^h vn
	'blue'	*k ^h øk ^h e	k ^h uko	k ^h ukie	k ^h uko	k ^h uk ^h u	hk ^h ø
(b)	'axe'	*syk ^h e	suko	sukie	ške	suku	sukhe
	'to guard'	*sak ^h i	ski	saqi	saaqa	saxi	saaky
	'be ashamed'	*hič ^h e	ščee	šiče	šče	šči	hč ^h e
	'ox'	*hyk ^h er	xukor	xukie	okhor		hkor

(38) Monguor type deaspiration

A similar process is 'aspiration flip-flop', which occurs regularly in Monguor. This process converts an unaspirated–aspirated consonant sequence to an aspirated–unaspirated sequence (39). In this case as well, the spreading of preaspiration through a vowel is a possible explanation. For example, preaspiration of the second consonant in */tot^hara/*[to^htara], realized as devoicing of the final part of the vowel, may have spread through the vowel and been reinterpreted as postaspiration of the initial consonant, *[t^hotara]. At least in one word (39b), *[q] followed by *s does not have the expected reflex q in Monguor, but appears as x, as if it were the reflex of *[q^h].

(39) Aspiration flip-flop

	OM	Monguor	Santa	Bonan	Kangjia	Sh. Yugur
(a) 'inside'	*tot ^h ara	t ^h utor	sutoro			ht ^h ər
'forty'	*tøč ^h in	t ^h ičin				tøč ^h vn
ʻpig'	*kak ^h ai	xqai	q ^h uqei	qaquui	kykai	kak ^h ai
'to fit'	*čok ^h i	č ^h uqu		čoqui	čuku	čəkv
(b) 'bitter'	*kasihun	xašin	q ^h wšuŋ	qašoŋ	kaši	kašuun

This process created an aspirated stop phoneme p^h in some words in Monguor (40), and also in Santa, Bonan, Kangjia, and Shira Yugur, where aspiration flip-flop usually does not take place. This was also triggered by **s* in several words (40b). It seems that aspiration of the initial **p* is not triggered if the distance between it and the causing consonant is more than one short vowel (40c).

(40) Aspiration of *p

	OM	Monguor	Santa	Bonan	Kangjia	Sh. Yugur
(a) 'firm' 'to write'	-	p ^h ati p ^h uči	-	pat ^h uu p ^h uuč ^h i		pat ^h p ^h yč ^h y

DEVELOPMENT OF THE MONGOLIC LANGUAGES

	OM	Monguor	Santa	Bonan	Kangjia	Sh. Yugur
(b) 'also'	*pasa		p ^h ese	sa		p ^h ysa
'to rise'	*pos	posi	posu	oš	posu	$p^h \Im s$
'belt'	*pyse	p ^h usee	p ^h isie	se	p ^h use	p ^h ysyi
(c) 'tiger'	*pars	pas	pasu	pasu	pasu	parvs
'kidney'	*pøhesyn	poosi	posuŋ	poosoŋ	posun	pvisvn

In some words with initial vowel, aspiration apparently moved from a medial consonant to the beginning of a word as in flip-flop, but since there was no initial consonant, the moved aspiration appears as an initial consonant h or x (cf. Helimskij 1984). In this way, what looks like spurious reflexes of initial *h were created (41). There are, however, some words where there is no apparent source for this spurious *h (41b).

(41) Aspiration reinterpreted as initial *h

	OM	Monguor	Santa	Bonan	Kangjia	Sh. Yugur
(a) 'to die' 'gold' 'early'	*yk ^h y *alt ^h an *ert ^h e	xuku xaltan šte	<i>xuku</i> ant ^h aŋ et ^h ie	<i>hku</i> alt ^h aŋ ert ^h e	ku ant ^h ə et ^h e	hk ^h u alt ^h an hyrt ^h e
'long' 'girl' 'big' (b) 'to ride' 'to laugh'	*urt ^h u *øk ^h in *jek ^h e *unu *inehe	štur ščun ške xuni šinee	xutu ot ^h in xukie unu sinie	štur ok ^h uŋ ško honu šine	štu une šine	hort ^h o hk ^h on šik ^h e hony hnii

10.11 PALATALIZATION

The Old Mongolian vowel *i had a palatalizing effect on some of the preceding consonants in all Mongolic languages. In Halh and several other dialects of Mongolian proper, a series of palatalized consonant phonemes developed when the *i which conditioned palatalization disappeared by vowel reduction or by *i-assimilation, and similar developments took place in Buriad, Oirad, and Dagur. Similarly, the palatalizing effect of *i on a preceding vowel created new vowel phonemes in Oirad and in some Mongolian dialects.

10.11.1 Consonant palatalization¹⁵

Palatalized consonant phonemes occur only in pharyngeal words in Halh, where they developed due to the influence of a following *i. In initial syllables, all *i were eliminated from pharyngeal words by *i-assimilation (10.7.3), and in non-initial syllables, all short vowels, including *i, were deleted from the phonological rep-

¹⁵ Bitkeev (1975*b*); see also the literature referred to in 10.7.3.

208

(42)		'easy'	'black'	'to guard'	'beard'
	Old Mongolian	*k ^h ilpar	*k ^h ara	*sak ^h i	*sak ^h al
	palatalization	[k ^{jh} ilpar]		[sak ^{jh} i]	
	*i-assimilation	k ^{jh} alpar			
	vowel deletion	k ^{jh} alpr	k ^h ar	sak ^{jh}	sak ^h l
	segmental changes	x ^j al3pr	xar	sax ^j	saxb
	vowel epenthesis	x ^j abpər			saxə৳
	Halh	x ^j alzpər	xar	sax ^j	saxə৳
		хялбар	xap	сахь	сахал

resentation (10.5.1). The conditioning environments disappeared in both cases so that palatalized and plain consonants became contrastive:

The Old Mongolian consonants $*k^h$, *p, *k, *m, *n, *l, *r were palatalized to Halh x^j , $p^j(w^j)$, g^j , m^j , n^j , ξ^j , r^j in these environments. The consonants *s, *h, $*\check{c}$, $*\check{c}^h$ also developed in different ways depending on whether or not they preceded *i (see 10.8.2–3), but *t and $*t^h$ never preceded *i in Old Mongolian.

Similar developments took place in other dialects of Mongolian, and also in Buriad and Dagur. In Kalmuck, only the dentals *l, *n were palatalized. According to Bitkeev (1965), they became pure palatals, but we will write them as l^j , n^j . Examples of palatalization in these languages are given in (43).

(43) Consonant palatalization

		OM	Halh		Chahar	Buriad	Kalmuck	Dagur
p^*p	'cheese'	*pisilak	p ^j as k əg	бяслаг				
	'fifty'	*t ^h apin	t ^h aw ^j	тавь	t ^h ɛp ^j	t ^h ab ^j ən	t ^h ɛwn	t ^h ap ^j
*k	'to swallow'	*čalki	całz ^j g ^j	залги	čɛlʲk	zalg ^j ə	zal ^j g	čɛlk ^j
$*k^h$	'border'	*k ^h ičahar	x ^j acgar	хязгаар		x ^j izar	k ^h izer	k ^h ičar
	'nail'	*k ^h imusun	xoms	хумс	xums	x ^j umhən	xumsn	k ^h imč ^h
	'to offer'	*t ^h ak ^h i	t ^h ax ^j	тахь		t ^h ax ^j ə	t ^h ek ^h	t ^h ak ^{jh}
	'hen'	*t ^h ak ^h ija	t ^h ax ^j a	тахиа	tex ^j a	t ^h ax ^j a	t ^h ak ^h a	tvki
*m	'thousand'	*miŋk ^h an	m ^j aŋg	мянга	mıŋk	m ^j angən	miŋgn	m ^j ankə
	'meat'	*mik ^h an	max	мах	max	m ^j axən	maxn	m ^j ak
	'life'	*amin	am ^j	амь	ϵm^j	am ^j ən	εmn	am ^j
*n	'baby'	*nilk ^h a	n ^j al z x	нялх	nılx	n ^j ilxə	nilx	n ^j alk ^h
	'to hide'	*nihu	ηυυ	нуу	nuu	n ^j uu	nu	nəə
	'to recognize'	*t ^h ani	t ^h an ^j	тань	t ^h en ^j	t ^h an ^j ə	t ^h an ^j	t ^h an ^j
*l	'which'	*ali	aђ ^j	аль	εl ^j	al ^j ən	al ^j	
	'fruit'	*alima	alʒ ^j əm	алим	εl ^j əm	al ^j mə	al ^j mn	al ^j əm
*r	'to hold'	*pari	par ^j	барь	pɛrʲ	bar ^j ə	ber	par ^j
	'horse'	*morin	mər ^j	морь	mær ^j	mər ^j ən	mørn	mər ^j

There is no palatalization of initial consonants in Chahar, and palatalization took place in Halh only when the assimilating vowel was *a, and never when it was *u or *o, as illustrated in (43). Palatalization of initial consonants is more common in Buriad, where it occurs before v (< *u) as well, as in $n^{j}vv$ 'to hide' in (43), and even in non-pharyngeal words (Zolhoev 1973*b*); compare, for example, Buriad

 $n^{j}ud \partial n$ 'eye' and $un^{j}en$ 'cow' (OM *nityn, *ynijen) with Halh nut нүд, une үнээ. Some pharyngeal words in Buriad retain i in the initial syllable (e.g. $x^{j}izar$ 'border'); since all consonants became palatalized before i (Soktoeva 1988), there is no palatalized ~ plain contrast in this position.

There are some words where expected palatalization does not take place in Halh, for example, max max 'meat', OM *mik^han (cf. Buriad $m^{j}ax\partial n$). These irregularities have no obvious explanation. Apparently, **i*-assimilation took place in several waves, so that for instance the word *mik^han 'meat' assimilated its **i* before palatalization had taken place in Halh, while words as *mink^han 'thousand' did not assimilate **i* until after palatalization (44). Two or even three different 'breakings', possibly combined with a change of the vowel from non-palatalizing **u* to palatalizing **i* (cf. 8.6.4), were assumed by Vladimircov (1929: 176ff.), Sanžeev (1953: 104ff.), Janhunen (1990b), and others. This does not really solve the problem of why the initial consonant was palatalized in certain words but not in others, however.

(44)	Old Mongolian	*mik ^h an 'meat'	*miŋk ^h an 'thousand'
	*i-assimilation (1)	mak ^h an	_
	palatalization	_	[m ^j iŋk ^h an]
	* <i>i</i> -assimilation (2)	_	m ^j aŋk ^h an
	other changes	max	m ^j aŋg
	Halh	тах мах	т ^ј ађе мянга

Word-final consonants were often palatalized in Dagur, even when they did not occur before **i*. For example, OM $k^{h} \phi l$ 'foot', $k^{h} elen$ 'tongue', kar 'hand' became Dagur $k^{h} ul^{j}$, xxl^{j} , kar^{j} (cf. Poppe 1955: 59). There was a special development in Monguor, where k^{h} often became \check{c}^{h} before **i*: OM $k^{h} imusun$ 'nail' > Mgr $\check{c}^{h} imsi$, $k \phi k^{h} in$ 'girl' > $\check{s}\check{c}un$ (cf. Badgaev 1989c).

10.11.2 Vowel palatalization¹⁶

Eastern Mongolian dialects have no palatalized consonant phonemes. Instead, palatalized vowel phonemes developed when a word with an originally back vowel (*a, *o, *u) had an *i in the following syllable. It seems that the development took place in four steps: (i) *i palatalized the preceding consonant; (ii) palatalized consonant phonemes were created by loss of final vowels; (iii) the palatalized consonant made the preceding vowel palatal; and (iv) consonant palatalization was lost, making the palatalized vowel phonemic. Halh went through the first two steps (and also the third step on the phonetic level (see 1.3)), and Baarin went through all four (45).

¹⁶ Ramstedt (1932); Rudnev (1911: 188); Cenggeltei (1959); Bayancogtu (1962); Bosson and Unensečen (1962); Pavlov (1963*a*); Tömörcèrèn (1969*b*); Rassadin (1982: 116ff.); Togog-a (1983*a*); Kögjiltü (1984; 1986*a*); Kuribayashi (1985*a*); Sun et al. (1990); Kim Chu-wŏn (1998; 1999*b*); Gereltü (1999).

(45)	Old Mongolian	*ama/n 'mouth'	*ami/n 'life'
	(i) consonant palatalization		[am ^j i]
	(ii) deletion of short vowels	am	am ^j
	Halh	am ам	am ^j амь
	(iii) vowel palatalization	—	[ɛmʲ]
	(iv) loss of consonant palatalizatio	n —	εm
	Baarin	am	εm

A similar development took place in Oirad (including Kalmuck). Examples are given in (46), where Chahar and Halh forms are given for comparison.

(46)	Vowel palatalization
------	----------------------

		OM	Baarin	Kalmuck	Chahar	Halh	
*a	'life'	*amin	εm	εmn	εm ^j	am ^j	амь
	'to return'	*k ^h ari	xer	xer	xer ^j	xar ^j	харь
	'fifty'	*t ^h apin	t ^h ɛp	t ^h &wn	t ^h ɛp ^j	t ^h aw ^j	тавь
	'which'	*ali	εl	al ^j	εl ^j	al3 ^j	аль
	'to recognize'	*t ^h ani	t ^h ɛn	t ^h an ^j	t ^h ɛn ^j	t ^h an ^j	тань
*0	'sheep'	*k ^h onin	xœn	xøn	xœn ^j	xən ^j	хонь
	'horse'	*morin	mær	mørn	mær ^j	mər ^j	морь
*и	'to summon'	*uri	Υľ	yr		υr ^j	урь
	'destiny'	*k ^h upi	xœp	xyw	хүр ^ј	xuwj	хувь

Since Kalmuck did not undergo any vowel shift, the original back vowels **a*, **o*, **u* were not changed before they were palatalized by a following **i*, becoming ε , ϕ , *y*. The vowels ϕ and *y* merged with the unchanged reflexes of Old Mongolian * ϕ and **y* (e.g. OM **m* ϕ *r* 'path' > *m* ϕ *r*, **yče* 'to see' > *yz*) so the resulting vowel system has eight vowels. This is illustrated in Figure 10.3a. Kalmuck has only two palatalized consonant phonemes, *n^j* and *l^j* (and the marginal *d^j*, *t^{jh}*). Vowel palatalization usually took place only before non-dental consonants which have no corresponding palatalized consonant.

In Baarin, Old Mongolian **a* was palatalized to ε as in Kalmuck, but the Mongolian vowel shift changed **o*, **u* to *s*, *v* before they were palatalized, so that their palatalized reflexes are α , *y* (Figure 10.3b). Unlike in Kalmuck, the palatalized vowels did not merge with any existing vowels, so three new vowel phonemes were created. The vowel *y* was added to the system by monophthongization of the diphthongs **øi* and **yi* (10.4).

The Chahar dialect is analysed in different ways by different authors. Dobo (1983*a*) describes Shuluun Höh Chahar as having palatalized consonants as well as vowels. He writes the words 'life' and 'mouth' as εm^j and *am*. From the data given by Dobo, it seems that palatalized short vowels in initial syllables occur only before palatalized consonants, as in Halh. There is one difference from Halh, however: the diphthongs **ai*, **oi* developed to monophthongs $\varepsilon \varepsilon$, $\alpha \alpha$ (shortened in non-initial syllables) in Chahar, but are retained as *ai*, *oi* in Halh. This is an independent



FIGURE 10.3 Vowel palatalization

reason to assume vowel phonemes ϵ/ϵ , α/ϵ in Chahar (but not in Halh), and the [ϵ] in [ϵm^{j}] and [α] in [m αr^{j}] must belong to these phonemes.

Sun et al. (1990) do not write palatalized consonants word-finally for Shuluun Höh Chahar (so that 'life' ~ 'mouth' are $\varepsilon m \sim am$), but they write palatalized consonants in other positions (e.g. un^{j_y} 'cow', vl^{j_as} 'aspen'). Under both analyses, it is necessary to assume the existence of palatalized vowels as well as consonants. We will follow Dobo.

10.11.3 Vowel palatalization and vowel harmony¹⁷

Although palatalization restructured the vowel systems of both Eastern Mongolian and Oirad, it had quite different consequences for vowel harmony in these two languages, here exemplified with Baarin and Kalmuck.

```
<sup>17</sup> Poppe (1965: 184); Vago (1973).
```

212

Using the vowel features introduced in section 5.1, palatalization can be regarded as an addition of the feature [palatal] (47). The main acoustic affect is to increase F2.

(47)	u	[R]	>	у	[PR]
	a	[FO]	>	3	[PFO]
	0	[OR]	>	ø	[POR]
	υ	[FR]	>	Y	[PFO]
	э	[FOR]	>	œ	[PFOR]

In Baarin, vowel harmony is based on the feature [pharyngeal], and vowel palatalization did not change the harmony class of the vowels, because this feature is not involved (48).

(48) Baarin vowel palatalization and vowel harmony

a	[FO]	pharyngeal	\rightarrow	ε	[PFO]	pharyngeal
		pharyngeal				pharyngeal
υ	[FR]	pharyngeal	\rightarrow	Y	[PFR]	pharyngeal
u	[R]	non-pharyngeal	\rightarrow	у	[PR]	non-pharyngeal

In Kalmuck, vowel harmony is based on palatality, and palatalization changed the harmony class of a non-palatal vowel to palatal (49).

(49) Kalmuck vowel palatalization and vowel harmony

a	[FO]	non-palatal	\rightarrow	ε	[PFO]	palatal
0	[OR]	non-palatal	\rightarrow	ø	[POR]	palatal
u	[R]	non-palatal	\rightarrow	у	[PR]	palatal

Thus, palatalization changed the vowel harmony class of words with original back vowels in Kalmuck, but not in Baarin. This is illustrated in (50) with the instrumental suffix, $-\varepsilon r/ar$ in Kalmuck and -yr/or/ar/3r in Baarin; for verbs, the direct past (Kalmuck $-l\varepsilon/la$; Baarin $-ly/lo/la/l_3$) is used. Halh forms are given for comparison. The words in (50a), which had front vowels in Old Mongolian, take non-pharyngeal suffixes in Baarin and Halh, and front-vocalic suffixes in Kalmuck. Those in (b) and (c) are originally back-vocalic, and take pharyngeal suffixes in Baarin and Halh. They take back-vocalic suffixes in Kalmuck (b), but if the stem vowel has been fronted (c), the word changed its harmony class and takes front-vocalic suffixes.

(50)		OM		Kalmuck	Baarin	Halh		
	(a)	front ve	owel					
		*ker	'house'	ger-er	kyr–yr	gir–er	гэрээр	
		*mør	'path'	mør–er	mor-or	mor-or	мөрөөр	
		*yke	'word'	yg–er	uk–vr	ug-er	үгээр	
	(b)	back vowel						
		*aman	'mouth'	am–ar	am–ar	am–ar	амаар	
		*motun	'tree'	mod-ar	mət-ər	mət-ər	модоор	
		*sur	'to learn'	sur–la	sor–la	sur—ţa	сурлаа	

	OM		Kalmuck	Baarin	Halh	
(c)	fronted	back vowe	el –			
	*amin	'life'	em-er	Em–ar	am ^j –ar	амиар
	*morin	'horse'	mør–er	mær–ər	mər ^j –ər	мориор
	*uri	'invite'	yr–lɛ	yr–la	ur ^j —15a	урьлаа

As seen in section 10.4, **i*-diphthongs usually developed to front monophthongic vowels in Southern and Eastern Mongolian (including Baarin and Chahar), as well as in Oirad (including Kalmuck). In these languages, but not in Halh (1.3), long vowels (from OM **VhV* combinations) which were followed by **i* in the next syllable merged with the reflexes of **i*-diphthongs. Words whose vowels became fronted by monophthongization of **i*-diphthongs (51a) or by palatalization of long vowels (51b), changed their harmony class in Kalmuck (Bitkeev 1968) but not in Baarin or other dialects of Mongolian proper.

(51)		OM		Kalmuck	Baarin	Halh	
	(a)	*k ^h aič ^h i	'scissors'	xeeč ^h -er	xɛɛčʰ–ar	xaič ^h –ar	хайчаар
		*oira	'near'	øør–er	œœr–ər	oir-or	ойроор
		*k ^h ui	'sheath'	xyy–gɛr	xyy-kar	xui–gar	хуйгаар
	(b)	*tahari	'sore'	deer-er	teer-ar	taar ^j —ar	даариар
		*sahurin	'base'	syyr-er	syyr–ar	suur ^j –ar	сууриар

10.12 SYLLABLE AND WORD STRUCTURE¹⁸

As seen in Chapter 8 and in section 6.1, syllable structure was much simpler in Old Mongolian than in Halh. At the same time, the number of syllables and phonemes per word decreased considerably from Old Mongolian to Halh, as is shown in Figure 10.4, which is based on the 254 uninflected Old Mongolian words given in section 8.9 and their Halh cognates. This figure shows the number of words with a given number of syllables (1–4; top panel) or a given number of speech sounds, including Halh schwas (1–9; bottom panel). The average number of syllables per word decreased from 2.2 in Old Mongolian to 1.4 in Halh. Two-thirds of the words were disyllabic in Old Mongolian, but two-thirds are monosyllabic in Halh. Thus there is a clear tendency towards monosyllabic stems in Halh. There is a similar decrease in the number of speech sounds per word, from 4.8 in Old Mongolian to 3.5 in Halh.

The decrease of the number of syllables per word is accompanied by a corresponding increase of the number of possible syllables (52). The maximal syllable was CVC in Old Mongolian but is CVCCC in Halh. Furthermore, the number of different consonants has increased, and the number of consonants which are possible codas has increased as well. The number of possible consonants in a syllable

¹⁸ Tömörcèrèn (1970); Galsan (1976); Sun (1981); Tömörtogoo (1987*b*, *c*; 1992: 176ff.); Kuribayashi (1988; 1992); Qaserdeni (1995); Norjin (1998).

differs depending on the vowel harmony class of its vowel nucleus, and this was taken into account when the number of possible syllables was calculated.

(52) Possible number of phonemes and phoneme combinations in different positions

	Old Mongolian	Halh
Onset	16	23
Vowel nucleus	13	17
Coda	9	27
Complex codas		157
Number of possible syllables	1,872	50,328



FIGURE 10.4 Number of syllables (top) and speech sounds (bottom) per word in Old Mongolian and Halh

б

The lower number of syllables per word is mainly due to the larger set of different phonemes, and less to the existence of more complex syllables. The increase of the number of phonemes and the decrease of the number of syllables per word are mainly the consequences of two processes: the development of long vowels in connection with loss of **h*, and deletion of short non-initial vowels. Vowel deletion also increased the number of consonants which are possible in coda position, since several consonants which did not occur in syllable codas in Old Mongolian were transferred from onset to coda position in the resyllabification that was a consequence of vowel deletion. In Halh, these include \check{c} , c, j and the reflexes of aspirated consonants (t^h , c^h , \check{c}^h , x). Examples are OM * $k^hai\check{c}^hi > xai\check{c}^h$ xaŭr 'scissors'; *peje > pij δHe 'body'; * $huhut^ha > vot^h$ yyr 'bag'; * $k^h \phi k^h e > xox x \Theta x$ 'blue'. Furthermore, consonant palatalization created a number of new phonemes, which can occupy the coda (as well as the onset) position: * $k^honin > xon^j x OHE$; 'sheep'; *masi > maš MaIII 'very'; * $pari > par^j \delta \Theta P$ 'to hold', etc. Similar developments took place in other dialects of Mongolian proper, in Dagur, and in Oirad.

It is interesting to note that although Mongolian is typologically very different from the isolating tone languages spoken in East and South-east Asia (Sino-Tibetan, Kadai, Hmong-Mien, and some branches of Austroasiatic), it shows the same tendency to pack more and more information into each syllable which is typical for the South-east Asian linguistic area (see Svantesson 1988c).

10.13 CONCLUSION

The development of Mongolian proper, Oirad, and Buriad is more regular, and easier to describe, than the development of the other, 'peripheral' languages, which have been geographically isolated. This might indicate that Mongolian, Buriad, and Oirad are direct descendants of Old Mongolian, and that the peripheral languages are rather its sister languages. On the other hand, the peripheral languages have developed independently, in contact with Chinese, Tibetan, or Persian, and also without the stabilizing influence of the Mongolian written language. These facts may explain their more complicated linguistic history. It should also be noted that the peripheral languages have not been described in great detail, and that the available sources for them vary somewhat, sometimes due to dialect differences, but sometimes possibly due to different analyses by different investigators.

There is no generally accepted classification of the Mongolic languages into subgroups. Binnick (1987) and Rybatzki (2003) discuss this problem at some length. At least sixteen different classification schemes have been proposed: Rudnev (1908), Vladimircov (1929), Nomura (1940), Poppe (1955), Poppe (1960*a*; 1965), Doerfer (1964*a*), Bertagaev (1968*b*), Beffa and Hamayon (1983), Yu Shichang (1983), Cenggeltei (1989*a*), Binnick (1992), Jahontova (1997*a*), Nugteren (1997), Secencogtu (1999), Svantesson (2000), and Rybatzki (2003). One reason for this diversity is that the sets of languages for which different innovations have occurred often overlap, and furthermore it is difficult to sort out which innovations

216

are genetic and which are areal. Thus the choice of criteria affects the classification to a relatively high degree.

One might ask if it is at all meaningful to attempt a genetic classification of a group like the Mongolic languages, with a complicated history of language contacts, using standard criteria such as shared innovations. In an article devoted to Mongolic classification, Binnick (1987: 194) says that '. . . no simple *Stammbaum* can do justice to Mongolian linguistic history'. In our opinion, a definitive classification of the Mongolic languages must wait at least until all Mongolic languages have been studied thoroughly and a reliable reconstruction of Proto-Mongolic exists. Even then it can be doubted if a family tree model is appropriate; perhaps the 'punctuated equilibrium' model proposed by Dixon (1997) is more suitable for the Mongolic languages and dialects have been in constant contact resulting in linguistic convergence have been interrupted by short periods with rapid demographic and linguistic change.

APPENDICES

The appendices contain notes and details on the literature and sources used in the relevant sections.

Appendix A - Introduction

Additional literature on Mongolian and its phonology, and Mongolian speech processing

General descriptions of Mongolian and its phonology: Abel-Rémusat (1820); Schmidt (1831); A. I. Bobrovnikov (1835); Kowalewski (1835); A. A. Bobrovnikov (1849); Grunzel (1895); Kotwicz (1902); Ramstedt (1902); Rudnev (1905); Ôyabu (1920); Haenisch (1925); Whymant (1926); Vladimircov (1929); Poppe (1936; 1937; 1951*a*; 1954*a*; 1970); Takeuchi (1937); Kojima (1938); Hattori (1943); Hambis (1946); Todaeva (1951); Sanžeev (1959; 1964*b*); Luvsanvandan (1961*a*; 1964*b*; 1966*c*; 1967; 1975*b*; 1985); Vanduj (1961; 1970); Cenggeltei (1963; 1979*a*); Street (1963); Dobo (1964; 1983*a*); Posch (1964*a*); Rinčen (1966; 1979); Coloo (1967*b*; 1976); Qagancilagu (1970); Sühbaatar (1970); Beffa and Hamayon (1975); Hagiwara (1975); Möömöö (1977; 1979; 1997); Bayancogtu (1978); Nasunbayar et al. (1982); Jin (1983); Delige'erma (1987); Nadeljaev (1987); Sanžaa (1990; 1998); Nadmid and Sanžaa (1993); Qaserdeni et al. (1996); Sun (1996*b*); Jahontova (1997*b*); Orlovskaja (1997); Svantesson (2003*b*). See Chapter 9 for descriptions of other Mongolic languages.

Mongolian speech processing: Dawa, Ôkawa, and Shirai (1996; 1999*a*, *b*, *c*, *d*; 2001*a*, *b*); Dawa et al. (1997; 1998; 1999; 2000); Köke, Bao, and Coyijongjab (1997); Bao Huaiqiao and Chen (1998); Köke et al. (1998); Dawa, Shirai, and Ôkawa (2000); Köke, Chen, and Zheng (2001); Dawa and Shirai (2003); Köke (2003*a*); Radnaeva (2003).

Appendix B – Chapter 1

1.1 Acoustic properties of monophthongs. Duration: Sanžeev (1978); Bayancogtu (1982a); Rialland and Djamouri (1984); Köke (1997; 2001; 2003a). Formant frequencies: Obata and Teshima (1934; 1935); Jôo (1971; 1973a; 1975; 1976b; 1997); Rialland and Djamouri (1984); Svantesson (1985; 1995b); Coyijongjab (1989b); Coyijongjab and Jôo (1993); Bayarmendü (1997c; 1998a, b); Franzén and Svantesson (1999); Köke (1999b; 2003a); Franzén (2001). See the literature cited in section 3.1.1 for reduced vowels. Intrinsic pitch: Köke (1999c; 2003a).

The Mongolian vowels given here as $[u, \theta/o, v, \sigma]$ are often transcribed as \ddot{u}, \ddot{o}, u, o (e.g. by Poppe 1951*a*), which corresponds to their (reconstructed) pronunciation in Old Mongolian ([y, θ , u, o] in IPA symbols). In theoretical discussions,

especially of vowel harmony, Mongolists have used the terms 'front vowels' for the two first ones and 'back vowels' for the two last ones (e.g. Vladimircov 1929: 115; Poppe 1951a: 15). Their more detailed descriptions of vowel qualities show that they were aware of the phonetic incorrectness of this, and the use of this terminology may be due to the historical bias in Mongolian studies, or to the lack of an alternative terminology. This practice has been unfortunate since non-Mongolists have ignored the detailed descriptions and assumed that the vowel symbols denote phonetically front and back vowels.

According to Kim Chu-won (1997), sixteenth-century Korean sources transcribe Mongolian [u] and [ɔ] with Korean \mathfrak{L} [o]. A. I. Bobrovnikov (1835) says that the Mongolian letters \mathbf{v} (i.e. [0, v]) and \mathbf{v} [u, o] are similar to Russian [o] and [u], respectively, and that each Mongolian letter represents two sounds, one of which is 'milder' (*něžněe*) than the corresponding Russian sound. A. A. Bobrovnikov (1849) says that Mongolian \ddot{u} (our [u]) is almost like Russian [u], and u (our [v]) is between Russian [u] and [o]. Edkins (1871: 212) says that the vowels written by Schmidt (1831) as o, u, \ddot{o}, \ddot{u} (our $[0, \upsilon, \theta/0, u]$) are rather $\hat{o}, o, o, u \sim \ddot{u}$; he states that Schmidt's o and u are like o in fond and o in bone, and that ü varies between oo in tool and French u; he describes the difference between u and \ddot{o} as one of pitch, the former having lower pitch. Vitale and Sercey (1897) say that there is no \ddot{o} or \ddot{u} in Halh. Ramstedt (1902) writes the main allophones of the four rounded vowels (our $[u, \theta/0, \upsilon, \sigma]$) as u, \ddot{v}, u, o and uses the then current term 'mixed' (i.e. central) for the two first ones; he describes some of their allophones as front. He says that y is a sound intermediate between [u] and [o]. Poppe (1951a) follows Ramstedt's description. Rudney (1905) says that he was advised by the Iranist Carl Salemann to use the Cyrillic based letters \dot{y} , \dot{o} (for our $[u, \theta/o]$) in order to distinguish them from \ddot{u} , \ddot{o} , which are not found in Halh. He further says (p. 7) that y and o are similar to Russian [u] and that y and o are similar to Russian [o]. Cybikov (1915: 1-2) says that [our [u]] is pronounced as [u] with the tongue pulled backwards (s ottjagivaniem jazyka *nazad*). Kamiyama (1919) uses the symbols *i*, *e* (i.e. [y]), *u*, *o*, *a*, *o*, *u* and says that u and o are pronounced 'making an effort in the throat' (喉にカヲ入レテ nodo ni chikara o irete). Vladimircov (1929: 57) uses the Cyrillic based letters y, o, y, o (for our $[u, \theta/0, v, \vartheta]$ and the terms central series (*srednij rjad*) and back series (*zadnij* riad). He says (p. 159) that Mongolian u is intermediate between [u] and [o]. Obata and Teshima (1934) say that *ü* (our [u]) is similar to [u]. Hattori (1936; 1937*a*) says that \ddot{u}, \ddot{o}, u, o (our $[u, \theta/o, v, \vartheta]$) are pronounced as $u, \ddot{o}, o, \vartheta$ in the spoken language. Kojima (1938) says that u (our [v]) is between [u] and [o]. Sanžeev (1941) describes Buriad y, o as central (srednij), but uses the terms front ~ back (perednij ~ zadnij) when analysing vowel harmony. Aberle and Austin (1951) say that Chahar \ddot{u} (our [u]) is a back vowel and \ddot{o} (our $[\theta/o]$) is a central vowel. Cenggeltei and Sinedke (1959) use the IPA symbols u, o, ω [equivalent to v], o for the rounded vowels.

1.1.1 The phonetic basis for vowel harmony. A. I. Bobrovnikov (1835) and Kljukin (1926) use the terms loud (gromkij) and soft (tihij) for the vowel harmony classes, corresponding to the Mongolian traditional terms er-e 'male' and em-e

APPENDICES

'female'. Adam (1874) uses the terms strong ~ weak (forte ~ faible). Orloy (1878) uses the terms loud (gromkij) and dull (gluhoj) The terms hard (series) (tverdvj (riad)) and soft (series) (miagkii (riad)) are used by many Russian and Soviet authors in descriptions of Mongolian and Buriad: A. A. Bobrovnikov (1849: 6); Cybikov (1908); Rudney (1913–14); Nadeljaev (1957); Buraev (1959; 1960; 1985a; 1986). According to Buraev, the soft series vowels are characterized by raising of the central part of the tongue blade. The terms pharyngeal ~ palatal are used by Grunzel (1888) (guttural ~ palatal); Rudney (1905; 1913–14) and Cybikov (1908; 2) (gortannyj ~ nebnyj); Vladimircov (1929: 115) (guttural'nyj ~ palatal'nyj); Abematsu (1936) and Kojima (1938) (喉音 kôon ~ 口蓋音 kôgaion). Some authors use the terms back ~ front for modern Mongolian or Buriad: Vladimircov (1929) and Sanžeev (1941) (zadnij ~ perednij); Poppe (1951a: 20 (hintere ~ vordere); 1970); Todaeva (1951) (zadnij ~ perednij); Street (1963). The traditional terms cingg-a ~ köndei (čanga ~ höndij in Cyrillic Mongolian) are used by e.g. Tömörcèrèn (1967b; 1969c), who gives the Russian equivalents as sil'nyj 'strong' ~ slabyj 'weak'; Bayancogtu (1982b); Qaserdeni et al. (1996: 104 ff.); Jin (1987). Hattori (1975; 1978; 1980) uses the terms open ~ closed. Tömörcèrèn (1969c), Jôo (1976b), and Norjin (1992b) use the terms vertical harmony (uužimših zohicol), diagonal harmony, and wide-narrow harmony (aguu uvitan-u jokical), respectively.

The assumption that Mongolian vowel harmony is based on tongue root retraction was first formulated by Cenggeltei (1963). In Cenggeltei (1959) and Cenggeltei and Sinedke (1959), the traditional terms er-e 'male' and em-e 'female' are used. Cenggeltei and Sinedke (1959) interpret their X-ray pictures of vowel articulation as showing that the main difference between [0, 0, 1] and [u, 0, i] is that the whole tongue body is more retracted in the first set of vowels. In Cenggeltei (1963), the traditional terms cingg-a 'tense, strong' ~ köndei 'hollow' are used with the interpretation (p. 11) that tense vowels are pronounced with tensing of the upper pharynx cavity and with the tongue root pulled backwards. Cenggeltei (1982) says that tense vowels are produced with tensed pharynx muscles and a retracted tongue root. In the 1983 Chinese version of Cenggeltei (1982), the terms 紧 jin 'tense' ~ 松 song 'lax' are used, and these are subsequently used by other scholars from China, e.g. Dobo (1984); Sun (1996b: 73). Analyses similar to Cenggeltei's have been proposed by Gregerson (1976: 361ff.) (tongue root fronted ~ backed) and Svantesson (1985) (pharyngeal ~ non-pharyngeal). Rialland and Djamouri (1984) use the feature [arrière] ('back') which is realized phonetically as [-ATR].

Jôo (1991–2; 1992; 1997) found that for each vowel harmony pair $[u \sim v, o \sim v, v \sim a]$, the second member has higher oral airflow than the first one. Bao Huaiqiao and Lü (1992) measured the intensity level of the two first formants, and found that the 'tense' (pharyngeal) vowels have a relatively higher F2 level than the 'lax' vowels. (See also the literature cited in 5.2.)

Appendix C – Chapter 2

2.1 *Stops and affricates.* Vitale and Sercey (1897) say that Halh *d* (our [t]) is similar to *t*. Vladimircov (1929: 60ff., 384, 402), Poppe (1931; 1936: 11; 1951*a*; 1955:

APPENDICES

103), and Sanžeev (1953: 87) repeat Ramstedt's description of strong and weak stops (using the terms *sil'nyj* ~ *slabyj* when writing in Russian). Rudnev (1905: 7) and Todaeva (1951) characterize the two series as voiceless ~ voiced (*gluhoj* ~ *zvonkij*), as does Hattori (1943) (無声 *musei* ~ 有声 yûsei). Mongolists from China, including Qaserdeni (1958), Cenggeltei (1959; 1963; 1979a: 55ff.), Cenggeltei and Coyijongjab (1959), Qaserdeni et al. (1996: 101), and Sun (1996b), usually label the two series of stops and affricates as aspirated (Ch 送气 sòng qì, Mo (*keiyin*) *türilte-tei*) vs. unaspirated (Ch 不送气 bú sòng qì, Mo (*kei-yin*) *türilte ügei*).

Most Mongolian researchers use the Mongolian terms $\check{c}anga \sim sul$ 'strong/tense \sim weak/lax' or Russian $sil'nyj \sim slabyj$ 'strong \sim weak'. Luvsanvandan (1966c: 27) and Coloo (1976) interpret these terms as being more or less equivalent to voice-less \sim voiced (duugüi \sim duutai). Möömöö (1976; 1977; 1979) says that the strong \sim weak opposition involves the degree of tenseness (*naprjažennost'*); strong stops are voiceless and may be aspirated, while weak stops sometimes are voiced. He does not regard voicing as phonologically relevant, however. Sanžaa (1987a) says that strong stops are aspirated except when the next syllable has a strong stop and the intervening vowel is short (cf. 10.10). See also Luvsanvandan (1960).

2.1.1 Stops in other Mongolian dialects. Aberle and Austin (1951) say that the Chahar Mongolian stop series are voiceless fortis aspirated vs. voiceless lenis unaspirated. Austin, Hangin, and Onon (1956) say that Chahar stops are aspirated vs. unaspirated voiceless. Jôo (1973*b*) found that Chahar stops are aspirated vs. unaspirated voiceless, but that the Halh ones are unaspirated voiceless vs. voiced. VOT measurements for Chahar were published by Köke (1998*a*) and by Köke and Coyijongjab (1999: 121, 162).

2.1.2 Buriad and Kalmuck stops. Podgorbunskij (1910: 16) says that Halh t and d (our [t^h, t]) are very similar to each other, and that the corresponding consonants in Buriad (i.e. our [t^h, d]) are less similar. According to Ramstedt (1957: 42), the weak consonants in Western Halh, Kalmuck, and Buriad are often voiced, but in Southern and Eastern Halh, the strong consonants are more heavily aspirated, and the voicing of the weak consonants depends on sandhi rules (cf. deaspiration, 10.10). Poppe (1930b: 12) says that all Buriad dialects as well as Oirad have a voiced rather than voiceless d, and a t that is not very aspirated; he says that, in contrast, Mongolian d (our [t]) is voiceless. Poppe (1938: 39) says that the Buriad weak stops may be voiced depending on the dialect. According to Sanžeev (1941), the Buriad stop series are voiceless ~ voiced (gluhoj ~ zvonkij) or strong ~ weak. Buraev (1959: 41) analyses the two series in Buriad as strong vs. weak, the weak consonants often being voiced in syllable-initial position or before voiced sounds. Buraev (1987a: 42) says that the strong stops often, but not always, are aspirated, and that b often is voiceless initially. Bitkeev (1965; 1983) describes the two series of consonants in Kalmuck as strong ~ weak (*sil'nyi* ~ *slabyi*), and says that the weak consonants may be voiced, but that this is not their distinctive property. Iliškin (1973) and Pavlov (1983) use the terms voiceless ~ voiced (gluhoj ~ zvonkij).

Appendix D – Chapter 3

3.2.1 Palatalized consonants. Scholars who recognize the existence of palatalized consonant phonemes in Mongolian or Buriad are: Castrén (1857: 1); Rudnev (1913–14); Ramstedt (1932); Poppe (1938: 39); Nadeljaev (1957); Buraev (1959: 49); Sanžeev (1959: 23); Bitkeev (1975b); Coloo (1976; 1987: 28); Rassadin (1976); Möömöö (1977); I. D. Buhaeva (1980); Saitô (1986); Sanžaa (1987*a*); Zolhoev (1989; 1991). Todaeva (1951) says that there are palatalized consonants in Mongolian (p. 41), but does not include them in the table of 'all consonants' (p. 33). Similarly, Sanžeev (1941: 20) says that palatalized consonants are phonemes in Buriad, but does not include them in the consonant table (p. 21). Sanžeev (1982) says that Mongolian and Buriad palatalized consonants have not been completely phonemized, but does not explain what that means. Scholars who do not recognize the existence of palatalized consonant phonemes include Ramstedt (1902), Poppe (1951*a*; 1970), Street (1963), Luvsanvandan (1964*b*; 1967; 1982*b*; 1985), and Tömörcèrèn (1969*b*).

3.2.2 Velar and uvular consonants. Vladimircov (1929: 123ff.); Damdinsürèn (1958: 25); Tömörcèrèn (1966b); Luvsanvandan (1978; 1980a); Saitô (1985); Shimizu (1992). Badmaev (1980b) and Soktoeva (1980a) give X-ray pictures which show clearly the velar ~ uvular alternation in Tsongool Mongolian and Buriad, respectively.

Appendix E - Chapter 5

5.2 *Vowel harmony*. Literature on the phonetic basis of vowel harmony is cited in Appendix B, 1.1.1, p. 219. Mongolian vowel harmony, especially the transparent vowel *i* and the blocking of rounding harmony by the rounded non-open vowels *u* and v, has been debated quite extensively in the theoretical linguistic literature: Binnick (1969; 1980); Tretiakoff (1975); Odden (1977; 1980); Chinchor (1979); Jensen and Stong-Jensen (1979); Steriade (1979; 1987: 355ff.; 1995: 148); S. Anderson (1980); Hamp (1980); Cohen (1981); Halle and Vergnaud (1981: 12f.); Leben (1982: 178f.); Yamada (1983: 47ff.); Goldsmith (1985); van der Hulst and Smith (1986); Schein and Steriade (1986: 716); J. Anderson and Ewen (1987: 276ff.); Lieber (1987: 134ff.); Demirdache (1988: 74); Walker (1993). All these assume palatal harmony and are therefore of limited value as analyses of Halh Mongolian. Analyses which assume vowel harmony of the pharyngeal ([-ATR] or [RTR]) type are Cenggeltei (1963; 1982); Rialland and Djamouri (1984); Svantesson (1985); van der Hulst and Smith (1987; 1988); Kim Chu-wŏn (1988a, b; 1992; 1999a); Archangeli and Pulleyblank (1994: 266ff.); Ch'o Mi-hŭi (1994: 191ff.); Hong Sŏng-hun (1994a: 170-231; 1994b; 1996); Denwood (1995; 1996; 1997a, b); Kaun (1995: 39-65); Xiao (1995); Li Bing (1996; 1999); Esenova (1998); see also Kuribavashi (1981c) and Sun (1981).

Appendix F – Chapter 6

6.2 Syllabification of morphologically simple words. Mongolian syllabification has not been treated extensively in the literature. Cenggeltei (1979a: 148) gives a

APPENDICES

right-to-left syllabification rule, expressed in the form of a rule for converting the Classical Mongolian script into the modern pronunciation. Rialland and Djamouri (1984: 337ff.) also give a right-to-left syllabification rule for Mongolian; see also Kuribayashi (1988) and Harada (1999).

6.4 *Cyclic syllabification.* Various restrictions on resyllabification after morphological operations have been proposed in the literature. Booij (1992) proposes that resyllabification can only change the final coda of the input form. This is too strict for Mongolian, where the entire final syllable can be changed. On the other hand, resyllabification restrictions are crucial in Mongolian, since they explain why words with different morphology are syllabified (and given epenthetic vowels) in different ways.

Appendix G – Chapter 7

7. *Prosody*. Esenova (1985; 1986b; 1987c; 1992b, c); Pjurbeev (2000). *Mongolian intonation*: Möömöö (1985); Esenova (1992a); Ichinose (1992a); Bao Huaiqiao and Köke (2000); Karlsson (2001; 2003b); Köke (2003a, b). *Buriad intonation*: Zolhoev (1973c); Bjuraeva (1976a, b, c; 1977; 1978; 1983; 1986a, b, c; 1987; 1988; 1989; 1990; 1991; 1996; 1998; 1999); B. Budaev (1983; 1987b); Mohosoeva (1986a, b; 1988; 1989); Balagunova (1987b); Pavlova (1987a, b; 1988; 1989; 1991); Bjuraeva and Pavlova (1989); Žargalov (1989); Buraev (1991); Esenova (1992d); Dambueva (1998). *Kalmuck intonation*: Esenova (1979; 1980; 1984; 1986a; 1987a, b; 1989a, b); Pavlov and Esenova (1984); Mušaev (1987); Barangova and Esenova (1990).

7.5 Word stress. Shimizu (1979; 1985; 1991–2; 1996); Batujirgal (1981); Jôo and Mikami (1981); Kakudô (1982*a*; 1992); Bayancogtu (1987); Buraev (1987*c*); B. Budaev (1991); Coyijongjab (1993*a*); Bayarmendü (1997*b*); Erdenicugla (1998); Köke (1998*c*); Tugtambayar (2002). Kakudô (1982*a*) says that the first mora of a word is low, then there is a rise, and finally a fall before the last or next to last mora. *Buriad*: Poppe (1930*b*: 67ff.); Bajčura (1961: 232; 1971; 1978); B. Budaev (1979; 1981; 1984; 1986*a*, *b*; 1987*a*, *b*; 1989; 1991). Bajčura says that there is tonic stress (i.e. higher pitch) on the last syllable.

Appendix H – Chapter 8

8.2 Sino-Mongolian. The Mongolian name of the Secret history of the Mongols is Monggol-un niguca tobciyan. The Modern Standard Chinese pronunciation of the Chinese name form is Menggu mishi, but the obsolescent pronunciation Menggu bishi (Wade-Giles transcription: Mêng-ku pi-shih) is often used in the Mongolist literature. The book is also known as the Secret history of the Yuan dynasty (Yuan chao mishi in Chinese). Hung (1951) and de Rachewiltz (1965) deal with the history of the text. Additional literature on Sino-Mongolian phonology: Kobayashi (1954); Street (1957a; 1986); Manlažav (1973a, b); Saitô (1993b); Kökebars (1994); Žam"jan (1995).

8.3 Arabic Mongolian. In his publication of the Mongolian material of the Leiden manuscript, Poppe (1927–8) dates it to 1245 (following M. T. Houtsma),

APPENDICES

but in the introduction to a reprint of this article (Poppe 1972b), he says that the correct date is 1343. See Saitô (2003) for several aspects on Arabic Mongolian.

8.5 Old Mongolian vowels. A few authors, including Kögjiltü (1982), Cenggeltei (1985b: 24), Kuldinow (1985), and Möngkebuyan (1998), reconstruct Old Mongolian with only five vowels **i*, **e*, **a*, **u*, **o*. These authors do not make it clear, however, how the development to the Modern Mongolian seven-vowel system took place. Li Bing (1996; 1999) assumes that Old Mongolian had a vowel system similar to the one found in Modern Mongolian proper, and Kögjiltü (1986b; 1989*a*; 1991; 1993) makes a similar assumption for the Middle Mongolian vowels. Irincin (1976–8) and Nangrub (1981) assume that Old Mongolian **e* was pronounced [x] (as in most Mongolian dialects of South Mongolia), but Kögjiltü (1986*b*) reconstructs its sound value as [e]. *Additional literature*: Poppe (1933*c*); Rincindorji (1960); Nomura (1965–73; 1971; 1976; 1979); Luvsanvandan (1966*e*; 1977); Manlažav (1969); Tömörcèrèn (1976); Yu Shichang (1985); Jalsan (1989); Kürelbagatur (1989); Bayanbagatur (1994); Qaserdeni (1994); Tulgaguri (1994); Kökebars (1998); Bayar (2000); Dašcèdèn (2000).

8.5.1 *Primary long vowels*. Nomura (1953) and Poppe (1955: 74ff.) observed that Monguor has long vowels corresponding to Mongolian short vowels in some words, but did not derive them from Proto-Mongolic. The existence of long vowels in Proto-Mongolic, based on Monguor, Dagur, and Moghol data, and also on comparisons with some Turkic (Turkmen, Yakut) and Tungusic (Evenki) languages was proposed by Hattori (1959); Nomura (1959*a*, *b*; 1972; 1979); Poppe (1959; 1960*a*: 92, 96ff.; 1961; 1962; 1965: 179; 1967); Ligeti (1964); Pritsak (1964: 162); Kim Pang-han (1971); Kuz'menkov (1989); Kürelbagatur (1991*a*). Vladimircov (1929: 305) and Ramstedt (1957: 165) do not reconstruct long vowels, and Doerfer (1964*c*; 1970; 1974) finds that the theory of long vowels in Proto-Mongolic is far from being proved. See also Murayama (1970), Bulucilagu (1986), and Möng-kebuyan (2000*a*).

8.6 Vowels in non-initial syllables and vowel harmony. Almost everyone who wrote about vowel harmony in Proto-Mongolic or Classical Mongolian assumed palatal harmony: Poppe (1951b; 1954a; 1955: 84ff.; 1960a: 147ff.; 1965: 184); Ramstedt (1957: 150ff.); Hamp (1958); Sanžeev (1964b); Lightner (1965); Zimmer (1967); Kiparsky (1968: 19, 33); Ultan (1973: 42ff.); Kim Sŏk-tŭk (1976; 1982); Lieber (1987: 56ff.); Darbeeva (1996: 111ff.); Guw-a (1996). Li Bing (1996) assumes, however, that Proto-Mongolic had RTR harmony (pharyngeal harmony), like Modern Mongolian proper. *Additional literature*: Thomsen (1959); Poppe (1960a: 117ff.; 1966; 1968; 1969b; 1975b); Tömörtogoo (1969; 1971); Bayancogtu (1981a, c); Kuz'menkov (1986); Qasbagan-a (1988).

8.6.4 *Proto-Mongolic* *ul? The vowel **ul* is reconstructed in back-vocalic words by Vladimircov (1929: 118ff., 170ff.); Sanžeev (1953: 118; 1964*b*: 31ff.); Poppe (1955: 33, 42; 1960*a*: 92,112ff.); Tömörtogoo (1992: 54ff.). Ramstedt (1957: 137) says that Proto-Altaic might have had this vowel, but that it merged with **i* in Mongolic. See also Yu Shichang (1983: 12ff.); Darbeeva (1996: 31); Sinedke (1998).

8.7.1 Stops and affricates. According to Vladimircov (1929: 384), the two

series are voiceless ~ voiced (gluhie ~ zvonkie) in 'Written Mongolian'. Poppe (1955; 1960a; 9; 1976) uses the terms strong ~ weak (stark ~ schwach or fortis \sim lenis) and says that aspiration was important. Furthermore, Poppe (1955) says that the two series were voiceless \sim voiced in Common Mongolian (pp. 95–6), but also that 'Common Mongolian t [our t^{h}] was probably a strongly aspirated dental consonant' (p. 102) and 'Common Mongolian *d [our *t] was a voiceless weak consonant (media lenis)' (p. 105). Ramstedt (1957: 37) says that the two series (in Proto-Altaic) were perhaps voiceless ~ voiced (tenuis ~ media), but more probably fortis ~ lenis, and that aspiration was an important feature of the fortis series. Doerfer (1964b: 54) uses the terms voiceless ~ voiced (stimmlos ~ stimmhaft). Hattori (1972: 63; 1973: 36) says that Proto-Mongolic had voiceless aspirated fortes vs. (half-)voiced unaspirated lenes. Tömörtogoo (1992: 118) uses the terms voiceless ~ voiced (duugüi ~ duutaj). Darbeeva (1996) uses the symbols for voiceless and voiced consonants, but does not describe the two series, except that *t (our $*t^{h}$) is characterized as voiceless (gluhoj) and *d (our *t) as weak (slaby) (pp. 74– 5). Möngkebuyan (2002) reconstructs aspirated and unaspirated series. See also Šonggor (1987). Velars and uvulars: Poppe (1955: 129ff.); Odden (1980); Finch (1989–91); Möngkebuyan (2002). Poppe (1960a: 10) says that uvulars and velars were allophones in Proto-Mongolic (Urmongolisch), but on p. 16 he says that they had developed to two different phonemes already in Pre-Mongolic (Vormongolisch). Absence of *t and *th before *i: Ramstedt (1902: 11; 1914; 1932; 1957: 42, 81); Vladimircov (1929: 395, 405); Sanžeev (1953: 93); Poppe (1955: 103; 1956b; 1960a: 14, 22, 26; 1966); Clauson (1962: 206); Doerfer (1985b: 176); Badgaev (1989c); Tömörtogoo (1992: 123ff.); Kuz'menkov (1993b).

8.7.2 The fricative *h: Poppe (1955: 96; 1958; 1960a: 10; 1969a; 1976), Luvsanvandan (1975a), Krippes (1992), and Möngkebuyan (2002) reconstruct Proto-Mongolic *p. Tömörtogoo (1992: 121) reconstructs Proto-Mongolic *f (from Proto-Altaic *p), and says that it developed to x in Old Mongolian and h in Middle Mongolian. See also Širokogorov (1931: 113ff.); Kobayashi (1954); Aalto (1955); Clauson (1962); Luvsanbaldan (1962); Doerfer (1968; 1985b: 148ff.; 1996); Poppe (1969c); Vanduj (1978); Yu Shichang (1983: 17ff.); Bayanbagatur (1987); Kuz'menkov (1988b); Bulag (1990; 1993b); Wang (1992a, b); Huang (1993; 1997); Ardajab (1994); Rozycki (1994); Darbeeva (1996: 69ff.); Janhunen (1999).

Appendix I – Chapter 9

9. *The Mongolic languages*. Vladimircov (1929); Sanžeev (1952; 1953); Benzing (1953); Ramstedt (1957: 25ff.); Cenggeltei (1957–8; 1985*b*); Poppe (1965); Bertagaev (1968*b*); Comrie (1981); Möömöö and Mönh-Amgalan (1984); Schwarz (1984); Sun (1985); Ramsey (1987); Sun et al. (1990); Pjurbeev (2001*b*).

9.1 Mongolian. Dialects of Mongolia: Luvsanvandan (1959; 1961b); Nagy (1960); Coloo (1967c; 1985; 1987); Vanduj (1973); Rinčen (1979); Janhunen (2003d). Dialects and standard language of South Mongolia: Hattori (1937b); Nomura (1941b); Sturt (1941); Todaeva (1956; 1957; 1960a, b; 1985; 1997d); Coyijongjab (1957; 1978; 1993b); Cenggeltei (1957–8; 1978; 1979b); Qaserdeni

and Naranbatu (1978); Türgen (1978); Chen (1993; 1995b; 1999); Sun (1996b); Qaserdeni (2000); Pèljei (2001); Janhunen (2003d).

Northern Halh. Bayad: Öljeyibürin (1992). Darhad: Sanžeev (1931); Tömörtogoo (1967); Šarhüü (1979–80; 1984). Mongolian of Tuva: Bitkeev (1973; 1980); Ulanova (1987; 1988). South Selenge 'Buriad': Poppe (1934a; 1971); Čagdurov (1951); Radnaev (1960); C. Budaev (1962; 1965; 1977b; 1980); Buraev (1965); Badmaev (1980a, b); Sanžina (1989).

Southern Halh. Abaga: San (2002). Arhangai: Coloo (1962; 1969). Dariganga: Róna-Tas (1960a). Eastern Halh: Vanduj (1967). Gobi Halh: Coloo (1967a). Övörhangai: Tömörcèrèn (1967a). Sönid: Qadacilagu (1984); Cogbayar (1989); Öljeyibürin (2000; 2001). Western Halh: Bese (1961). Zavhan Sartuul: Vanduj (1964); Baatar (1987).

Southern Mongolian. Chahar: Aberle and Austin (1951); Hattori (1951); Bayancogtu (1962); Hangin and Street (1962); Tömörcèrèn (1966a); Jôo (1973b; 1975); Danzan (1978); Nadamid (1978; 1986; 1993; 1998); Qaserdeni (1980b); Secenbagatur (1982); Temür (1982); Uuda (1982); Sun (1983); Todaeva (1985); Türgen (1985, 1988); Coyijongjab (1989a, b); Qasbagan-a (1990); Ichinose (1992b); Coyijongjab and Jôo (1993); Kürelbagatur (1993); Idam (1994); Köke and Coyijongjab (1996); Köke and Sodobagatur (1996); Bayarmendü (1997c; 1998a); Naranbatu (1999). Ordos: Soulié (1903); Rudnev (1911); Whymant (1926); Mostaert (1926–7); Qaserdeni (1959; 1981a; 1985); Poppe (1964c); Street (1966); Batujirgal (1981); Kuribayashi (1982a); Todaeva (1985); Serengnorbu (1986; 1996); Erdenimöngke (1987); Möngkebuyan (1990); Banzargarudi (1998); Erdenibagatur (1998); Naranbatu (1999); Georg (2003a). Üjemchin: Kara (1962; 1963); Tömörcèrèn (1964; 1969a); Kökebars (1991); Nomintuyag-a (1992); Kökebars and Cimeg (1997).

Naiman. Bulu (1984); Sangdügüreng and Cegen (1989); Öljeyitogtaqu (1991); Secen (1994; 2000); Xi Caiyun (1995); Mönggöngerel (1998).

Eastern Mongolian. Arhorchin: Rudnev (1911); Secen (1988, 2000). *Baarin*: Cenggeltei (1959; 1961); Secenbagatur (1959; 1982); Bayancogtu (1962); Jalcib (1962); Dobo (1964; 1982); Todaeva (1985); Bayarmendü (1997*a*, *b*, *c*; 1998*a*); Secen (2000). *Gorlos*: Rudnev (1911). *Harchin*: Nomura (1940–1; 1941*a*; 1950; 1951; 1957; 1960); Todaeva (1985); Nolmajab (1986; 1991); Kögjiltü (1987*b*; 1990); Sodobagatur (1999; 2000; 2001); Qasbagatur (2002). *Horchin*: Bosson and Unensečen (1962); Caganqada (1979; 1981; 1985*b*; 1996); Coyijongjab (1982*a*); Togog-a (1983*a*, *b*); Öljei (1984); Todaeva (1985); Qayirqan (1990); Tulgaguri (1996; 1997); Bayarmendü (1997*b*, *c*; 1998*a*); Möngkebuyan (1997); Tong (1998); Cenggeltü (1999); Bayancogtu (2001); Soyol (2002). *Hüree*: Togog-a (1991*b*); Guw-a (1993); Sinquwa (1994); Xiuhua (2002). *Ih Minggan*: Todaeva (1985; 1988). *Jalaid*: Nasunbayar (1987*c*); Böke (1994); Öljeyibürin (1994). *Jaruud*: Togog-a (1991*a*); Xiao (1995). *Jirim*: Ilan and Saran (1998). *Monggoljin*: Qai (1987); Edquriyagci and Uran-a (1991); Edquriyagci (1996); Mansang (1998). *Tariach*: Boo (1997).

Urad. Bulu (1980); Batusayiqan (1985); Möngkebuyan (1992*a*; 1997); Naranbatu (1994; 1999); Ulagantuyag-a (1997).

APPENDICES

9.2 *Buriad.* The number of Buriad speakers in Russia was calculated from data given by D. D. Damdinova (www.belti.msk.ru/unesco/d10e.htm, accessed December 2001) (see also Dyrheeva, Budaev, and Bažeeva 1999).

Older sources. During the eighteenth century, Buriad word lists were recorded by Daniel Gottlieb Messerschmidt, Gerhard Friedrich Miller, Mihail Tatarinov, Peter Simon Pallas, Ivan Čemesov, and others. See C. Budaev (1992: 6ff.), who also uses these collections for tracing the development of Buriad dialects through time.

Phonology. Castrén (1857); Podgorbunskij (1910); Poppe (1933*a*; 1938; 1960*b*; 1964*b*); Sanžeev (1941; 1951); Bertagaev (1947; 1968*a*); Buraev (1962; 1983; 1987*a*, *b*; 1988*a*, *b*; 1989; 1993; 1995); Zolhoev (1963*a*, *b*; 1977; 1980); Buraev, Bažeeva, and Pavlova (1975); Rassadin (1976; 1982); Darbeeva (1978; 1988; 1992; 1997); Clark (1979); Soktoeva (1980*b*; 1986*a*, *b*); Matheev (1984); C. Budaev (1986; 1987); Damdinov (1986); Gantogtoh (1990); Walker (1994); Skribnik (2003). See Chapter 7 for literature on Buriad stress and intonation, and Chapter 2 for Buriad stops.

Dialects. Bálint (1877); Poppe (1933*b*; 1941*a*); Hattori (1940); Sanžeev (1953: 46ff.); Alekseev (1949; 1955); Rygdylon (1957); Cydendambaev (1960; 1968); Šagdarov and Rassadin (1968); C. Budaev (1978; 1986; 1992; 1993); Balagunova (1986; 1989); Cyrenov (1998).

Eastern Buriad. Aga: Poppe (1932*c*); Šagdarov (1968). *Buriad of Mongolia*: Gantogtoh (1990; 1993). *Hori*: Rudnev (1913–14); Bertagaev (1936); Sanžeev (1939); Bese (1964). *Ulan-Ude*: Cyrendoržieva (1977).

Northwestern Buriad. Baigal-Hudari: Abašeev (1951; 1956); Cydendambaev (1964); Matheev (1972b); Buraev (1976). Bargazhan: Radnaev (1958; 1965). Boohon: Homonov (1958; 1965); Bjuraeva (1996). Ehired-Bulagad: Matheev (1956; 1957; 1968). Ivalga: Abašeev (1951); Bese (1962); Cyrenov (1996). Kabansk: Čeremisov (1947). Kachug: Mitroškina (1968). North Selenge: C. Budaev (1977a). Ol'hoon: Babuev (1991). Osa: Balagunova (1987a).

Southwestern Buriad. Rassadin (1991; 1996*a*). *Aha*: Rassadin (1989*a*). *Alair*: Poppe (1930*b*); Buraev (1968). *Tünhen*: Čeremisov (1941); Abašeev (1958; 1965); Matheev (1972*a*).

Nizhneudinsk. Sanžeev (1930); Darbeeva (1960); Rassadin (1988; 1999).

Bargu. Poppe (1932*a*); Tömörcèrèn (1971); Boosiyang (1981); Pürbü (1987); Janhunen (1988); Rassadin (1989*b*); Boosiyang and Jirannige (1996).

9.4 *Oirad.* During the eighteenth century, Kalmuck vocabularies were collected by several persons including Nicolaas Witsen from Amsterdam, and the Swedes Filip Johan Stralenberg and Johan Christian Schnitscher (see Doerfer 1965 and Krueger 1975). The Swedish missionary Cornelius Rahmn, who was among the Kalmucks from 1819 to 1823, wrote unpublished Kalmuck–Swedish and Swedish–Kalmuck dictionaries, whose manuscripts are kept in Uppsala University Library. A Kalmuck–German dictionary was published in 1854 by the Herrnhutian missionary Heinrich August Zwick.

Phonology. Navrockij (1840); Popov (1847); A. A. Bobrovnikov (1849); Kotwicz (1929); Sanžeev (1940); Posch (1957–8; 1964*b*); Coyijongjab (1959); Bormanshi-

nov (1961); Pavlov (1963*a*); Bitkeev (1965; 1978; 1981; 1982; 1983; 1985*a*, *b*; 1988); Todaeva (1968); Benzing (1985); Secen (1989); Sambuudorž (1996; 2002); Sürüng (1997); Birtalan (2003); Bläsing (2003).

Dialects. Kara (1958); Kičikov (1967*a*); Ulanova (1992). *Alshaa*: Secenbilig (1982); Cogtu (1985); Gereltü (1986; 1991; 1992); Batugerel (1989). *Bayad*: Baldan and Coloo (1974). *Buzaawa*: Kičikov (1967*b*). *Dörbed*: Vanduj (1962; 1965); Kičikov (1963; 1967*a*). *Hoshuud*: Todaeva (1970); Jia (1982; 1986*a*, *b*); Caganqada (1985*a*); Sonam (1985); Oyunceceg (1986; 1987); Bayasqal (1987); Cüngkerja (1987); Dobo and Caganqada (1987); Dorjicang (1987); Dobo (1997). *Karakol Kalmuck*: Dondukov (1973; 1975); Tenišev (1976; 1997); Pavlov (1990); Esenova (2001). *Ööld*: Badmadorž (1984; 1987*a*, *b*). *Subei*: Qaserdeni (1990). *Torguud*: Coloo (1964); Ubušaev (1969*a*, *b*, *c*; 1979); Luvsanbaldan (1975); Bimba (1981); Taya (1992). *Tsaatan*: Ubušaev (1970); Badgaev (1989*b*). *Ural*: Pavlov (1964). *Urianhai*: Coloo (1973*a*). *Zahchin*: Coloo (1964; 1965).

9.5 *Dagur*. Aberle and Austin (1951); Austin (1952); Nomura (1960); Martin (1961); Poppe (1964*d*); Zhong (1965; 1980; 1982); Weiers (1978); Qagancilagu and Kögjimtei (1979); Engkebatu (1982); Namtsarai and Qaserdeni (1983); Sŏng Paeg-in (1983); Satô (1985); Sun (1985; 1996*a*); Todaeva (1986; 1997*b*); Kakudô (1987*a*); Üjügür (1987); Badarangg-a (1988); Kuribayashi (1993); Ardajab (1994); Chuluu (1994*c*; 1996); Tsumagari (2003). *Hailar dialect*: Poppe (1930*a*; 1934*b*); Tsumagari (1985). *Qiqihar dialect*: Wang (1989). *Xinjiang dialect*: Ding (1992; 1995).

Appendix J – Chapter 10

10. Development of the modern Mongolic languages. In addition to the standard works mentioned, other general works on the historical phonology include Šmits (1898); Osada (1952); Murayama (1960); Poppe (1964a; 1975a); Luvsanvandan (1966a); Weiers (1969); Nangrub (1981); Pürbü (1981); Liu (1984); Luvsanvandan and Bold (1985); Kuz'menkov (1988a); Sanžaa (1993; 1998); Ang (1994); Qasbagan-a (1996); Šongqor (1996); Janhunen (2003a, b); Rybatzki (2003a); Saitô (2003). Rassadin (1982) and Buraev (1987a) discuss the historical phonology of Buriad from a general Mongolic point of view.

10.1 *The Mongolic vowel shifts.* Mongolists have often ignored the vowel shifts in Mongolian proper and Buriad, implicitly regarding them as surface phenomena which only affect the pronunciation of the vowels, not the vowel system as such. An exception is Bosson and Unensečen (1962), who describe the vowel shift for Horchin (Eastern Mongolian) in terms of centralization and a shift backwards (corresponding to our velarization) and a shift downwards (pharyngealization). See also Hattori (1978; 1980); Janhunen (1981); Qaserdeni (1981*b*); Darbeeva (1996: 18ff.). Kögjiltü (1982; 1985; 1986*c*; 1989*a*, *b*; 1991) and Cenggeltei (1985*b*: 24), who refers to a 'young researcher', presumably Kögjiltü, give vowel correspondences corresponding to the Dagur and Monguor vowel shifts, but assume that the Old Mongolian vowel system was a five-vowel system similar to that in Monguor, and that Middle Mongolian had more or less the same vowel qualities as modern

Halh. Kögjiltü regards the front rounded vowels y, ϕ in Oirad as secondary, having developed due to Turkic influence.

10.2 Vowel splits and mergers. Split of *i in Mongolian dialects: Bayancogtu (1981b); Dobo (1983a); Batusayiqan (1985); Ünenci (1987); Ichinose (1992b); Möngkebuyan (1992b; 1997); Kürelbagatur (1993); Nadamid (1998); Qasbagan-a (2002). Bitkeev (1973) provides X-ray pictures which show that [u1] is the reflex of *i in back-vocalic words in the Mongolian dialect spoken in Tuva. Opening of *u: Kičikov (1967a); Bitkeev (1976); Sinquwa (1994); Böke (1999).

10.3 Long vowels. Ramstedt (1912); Kobayashi (1954); Posch (1957–8); Luvsanvandan (1966d); Tömörtogoo (1967; 1980; 1990; 1995); Bertagaev (1971); Ubušaev (1973); Bulag (1982; 1983–4); Rassadin (1982: 38ff.); Tian (1983); Kürelbagatur (1991; 2001); Nasunöljei and Nabcingerel (1997); Kökebars (1998); Jirannige (1999); Möngkebuyan (2000*b*); Li Meiling (2001).

10.7.3 *Regressive* *i-*assimilation.* A. A. Bobrovnikov (1849: 22ff.); Grunzel (1895: 22ff., 27); Poppe (1930b: 48ff.; 1956*a*); Nomura (1953); Kobayashi (1954); Sanžeev (1957; 1964*a*; 1970); Kałużyński (1965); Pavlov (1969); Tömörtogoo (1971; 1987*a*); Rassadin (1976; 1982: 23ff., 119ff.); Thomsen (1976; 1987); Kuribayashi (1981*a*, *b*; 1982*a*, *b*; 1985*b*); Hattori (1983*a*, *b*); Kögjiltü (1984); Bulucilagu (1985*b*); Erdenimöngke (1987); Janhunen (1990*b*); Qayirqan (1990); Kürelbagatur (1993); Garudi (1996); Kim Chu-wŏn (1999*b*); Möngkebuyan (2001); Qasbagan-a (2002).

10.10 Deaspiration and related processes. Regressive deaspiration: Luvsanvandan (1960; 1975c); Kara (1962; 1963); Coloo (1967a; 1969); Vanduj (1967); Tömörcèrèn (1969a); Gantogtoh (1978); Dobo (1981); Temür (1982); Cenggeltei (1985a; 1989b); Serengnorbu (1986); Sanžaa (1987b; 1988); Öljeyitogtaqu (1991); Kökebagatur (1992); Nadamid (1993); Idam (1994); Kökebars and Cimeg (1997). Progressive deaspiration: Poppe (1955: 98); Hattori (1972: 65ff.); Cenggeltei (1985a; 1989b); Satô (1991–2). Creation of p^h: Buraev (1987a: 41); Cenggeltei (1988); Darbeeva (1996: 142ff.).

REFERENCES

All references to the same author are collected under one name form; for ethnic Mongolian authors, the Mongolian name has been preferred. Cross-references are given in the Index.

For some periodicals, for example *Studia Mongolica*, the volumes are divided into fascicles. In some cases, a number of fascicles which are only part of a volume of the series are bound as a separate physical volume. This is indicated in the following way in the bibliography: *StMo* 3/6–11: fasc. 8 (pp. 14–35), that is, fascicle 8 of volume 3 of *Studia Mongolica*, a fascicle that consists of pages 14 to 35 in the physical volume consisting of fascicles 6 to 11 of volume 3 of the series. If the physical volume coincides with the series volume, the number of fascicles of the physical volume is not shown, for example, *StMo* 8: fasc. 14 (pp. 183–90).

Abbreviations

In general, we follow the abbreviations used in the Linguistic bibliography.

AH	Alt'ai hakpo (Journal of the Altaic society of Korea) (Sŏul).
AN	Akademija nauk [Academy of sciences].
AOH	Acta orientalia Academiae scientiarum Hungaricae (Budapest).
BF	Burjatskij filial Sibirskogo otdelenija Akademii nauk SSSR [Buriad branch of
	the Siberian division of the USSR Academy of sciences].
BIHP	Zhōngyāng yánjiūyuàn lìshǐ yǔyán yánjiūsuð jíkān (Bulletin of the Institute of
	history and philology, Academia Sinica) (Táiběi).
BION	Burjatskij institut obŝestvennyh nauk Sibirskogo otdelenija Akademii nauk
	SSSR [Buriad institute of social sciences at the Siberian division of the USSR
	Academy of sciences]; from 1992: Rossijskoj akademii nauk [of the
	Russian Academy of sciences] (Ulan-Udè).
BKI	Burjatskoe knižnoe izdatel'stvo.
BM	Burjat-Mongol'skij [Buriad Mongol].
BMNIIK	
	scientific research institute of culture] (Ulan-Udè).
BNC	Burjatskij naučnyj centr Sibirskogo otdelenija Akademii nauk SSSR [Buriad
	scientific centre at the Siberian division of the USSR Academy of sciences];
	from 1992: Rossijskoj akademii nauk [of the Russian Academy of sci-
	ences] (Ulan-Udè).
CAJ	Central Asiatic journal (The Hague; Wiesbaden).
CLS n	Papers from the nth regional meeting of the Chicago linguistic society.
GIJaLI	Gosudarstvennyj institut jazyka, literatury i istorii [State institute of language,
	literature, and history] (Ulan-Udè).
GK	Gengo kenkyû (Journal of the Linguistic society of Japan) (Tôkyô).

REFERENCES

HBK	Hoppô bunka kenkyû (Bulletin of the Institute for the study of North Eurasian cultures, Hokkaido University) (Sapporo).
HK	Hitotsubashi kenkyû (Hitotsubashi journal of social sciences).
HOM	Handbuch der Orientalistik I:V/2: Mongolistik (Leiden: Brill). 1964.
HZS	<i>Hèl zohiol(yn) (sudlal).</i> Ulaanbaatar. (The name forms <i>Hèl zohiol, Hèl zohiol sudlal</i> and <i>Hèl zohiolyn sudlal</i> occur.)
IAMS	International association for Mongol studies.
IIFF	Institut istorii, filologii i filosofii Sibirskogo otdelenija Akademii nauk SSSR
	[Institute of history, philology, and philosophy at the Siberian division of the USSR Academy of sciences].
IK	Ibaraki Kirisutokyô daigaku kiyô (Academic journal of Ibaraki Christian col-
	lege).
JSFOu	Suomalais-ugrilaisen seuran aikakauskirja (Journal de la société finno-ougri-
551 O u	enne) (Helsinki).
KKI	Kalmyckoe knižnoe izdateľstvo.
KNI	Kalmyckij naučno-issledovatel'skij institut istorii, filologii i ekonomiki [Kal-
KINIIITE	muck scientific research Institute of history, philology, and economics] (Èlis- ta).
KNIIJaLI	Kalmyckij naučno-issledovateľ skij institut jazyka, literatury i istorii [Kalmuck
IXI VIIJ al 21	scientific research institute of language, literature, and history] (Èlista).
KSBurNII	Kratkie soobŝenija Burjatskogo kompleksnogo naučno-issledovatel'skogo
	instituta (Ulan-Udè: SO AN SSSR).
KSINA	Kratkie soobŝenija Instituta narodov Azii (Moskva).
MK	Mongoru kenkyû (Bulletin of the Japan association of Mongolian studies). (Cf. NMGK.)
MKB	Monggol kele bicig (Mongolian language journal). Kökeqota. (1954-7, con-
	tinued as Monggol teüke kele bicig in 1958, and as Monggol kele jokiyal teüke
	in 1959–60; reestablished as <i>Monggol kele bicig</i> in 1979.)
MKJT	Monggol kele jokiyal teüke. Kökeqota. (Cf. MKB.)
MKSÖ	Monggol kele sinjilegen-ü ögülel-üd (Kökeqota: Öbör Monggol-un arad-un keblel-ün qoriy-a). 1987.
MKUJ	Monggol kele udq-a jokiyal (Mongolian language and literature) (Kökeqota).
MSYQ 2	Öbör Monggol-un yeke surgaguli-yin qoyadugar udagan-u Monggol sudulul-
	un olan ulus-un erdem sinjilegen-ü yarilcaqu qural: kele bicig; Nèiměnggŭ
	dàxué dì sān cì Měnggǔxué guójì xuéshù tǎolùnhuì: yǔwén; The second inter-
	national symposium on Mongolian studies under auspices of Inner Mongolia
	University: language (Kökeqota). 1991.
MSYQ 3	Öbör Monggol-un yeke surgaguli-yin gurbadugar udagan-u Monggol sudulul-
1115 I Q 3	un olan ulus-un erdem sinjilegen-ü yarilcaqu qural: ögülel-ün tobci; The third
	international symposium on Mongolology sponsored by Inner Mongolia Uni-
	versity: summaries of symposium papers (Kökeqota). 1998.
MTKASC	Monggol töröl-ün kele ayalgun-u sudulul-un cuburil; Měnggǔ yǔzú yǔyán
MIKASC	fängyán yánjiū cóngshū (Kökeqota: Öbör Monggol-un arad-un keblel-ün
MTVD	qoriy-a). Managal taika kala hisia (Kiikagata) (Cf. MKR.)
MTKB	Monggol teüke kele bicig (Kökeqota). (Cf. MKB.)
MUISÈ	Mongol ulsyn ih surguul': èrdèm šinžilgèènij bičig (Ulaanbaatar).
MY	Mínzú yŭwén (Minority languages of China) (Běijīng).
NagoKR	Nagoya daigaku bungakubu kenkyû ronshû (The journal of the faculty of lit- erature, Nagoya University) (Nagoya).
232	REFERENCES
--------------	--
NDX	Nèiměnggǔ dàxué xuébào: zhéxué shèhuì kēxué bǎn (Journal of Inner Mon- golia University: philosophy and social sciences). Subtitle changed in late 1990s to Rénwén, shèhuì kēxué bǎn (Humanities and social sciences) (Kökeqota). Chinese edition.
NELS	Proceedings of the North East linguistic society.
NLLT	Natural language and linguistic theory (Dordrecht).
NMGK	Nippon Mongoru gakkai kiyô (Bulletin of the Japan association for Mongol studies) (Tôkyô). (This name is used from issue No 19; the journal started as Nippon Mongoru gakkaihô (No. 1), continued as Nippon Mongoru gakkai kaihô (No. 2–9, also abbreviated NMGK), and then as Mongoru kenkyû (No. 10–18).
NOS	Nihon onkyô gakkai shûki kenkyû happyôkai kôen ronbunji (Proceedings of
$\hat{O}GDG$	the autumn meeting of the Acoustical society of Japan). Ôsaka gaikokugo daigaku gakuhô (Journal of Osaka University of foreign
ÔGDR	studies). Ôsaka gaikokugo daigaku ronshû (Journal of Osaka University of foreign studies, new series).
OGK	Onseigakkai kaihô (Bulletin of the Phonetic society of Japan) (Tôkyô).
OK	Onsei no kenkyû (The study of sounds) (Tôkyô).
ÖMBDS	Öbör Monggol-un bagsi-yin degedü surgaguli-yin erdem sinjilegen-ü sedgül (Kökeqota).
ÖMBYS	Öbör Monggol-un bagsi-yin yeke surgaguli-yin erdem sinjilegen-ü sedgül (Kökeqota). (Continuation of ÖMBDS after 1982/3.)
ÖMSK	Öbör Monggol-un surgan kömüjil-ün keblel-ün qoriy-a.
ÖMÜBDS	Öbör Monggol-un ündüsüten-ü bagsi-yin degedü surgaguli, erdem sinji- legen-ü sedgül: gün uqagan neyigem sinjilekü uqagan-u Monggol keblel (Journal of social sciences, Inner Mongolia teachers' college of the nation- alities, Mongolian edition) (Töngliáo).
ÖMÜYS	Öbör Monggol-un ündüsüten-ü yeke surgaguli, erdem sinjilegen-ü sedgül: neyigem-ün sinjilekü uqagan-u keblel (Journal of Inner Mongolia Univer- sity for nationalities: social sciences, Mongolian edition) (Töngliáo). (Con- tinuation of ÖMÜBDS, 2001–.)
ÖMYSS	Öbör Monggol-un yeke surgaguli-yin erdem sinjilegen-ü sedgül: gün uqa- gan neyigem sinjilekü uqagan (Kökeqota). (Mongolian edition. From 1991 the title is given in English as Journal of Nei Monggol University: philoso- phy and social sciences in Mongolian; 1992/4 changed to Journal of Inner Mongolia University: philosophy and social sciences in Mongolian.)
OShKR	Ozawa Shigeo kyôju taikan kinen ronbun to omoide no ki (Tôkyô: Jichôsha). 1993.
ÖΤ	<i>Ögülel-ün tegübüri</i> ([Xīníng]: Dumdadu Ulus-un Monggol kele bicig-ün sinjilegen-ü neyigemlig-ün Kökenagur qubiyari neyigemlig).
OUMÈ n IH	Olon ulsyn mongolč èrdèmtnij n ih hural [n-th International congress of Mongolists] (Ulaanbaatar).
PIAC 12	Hazai, György and Peter Zieme (eds.) (1974), Sprache, Geschichte und Kul- tur der Altaischen Völker (Schriften zur Geschichte und Kultur des alten Ori- ents, 5) (Berlin: Akademie-Verlag).
PIAC 14	Ligeti, Lajos (ed.) (1975), <i>Researches in Altaic languages</i> (Budapest: Akadémiai Kiadó).

- PIAC 25 Jarring, Gunnar and Staffan Rosén (eds.) (1984), Altaistic studies: papers presented at the 25th meeting of the Permanent international Altaistic conference at Uppsala June 7–11 1982 (Stockholm: Almqvist & Wiksell).
- PIAC 29 Istoriko-kul'turnye kontakty narodov altajskoj jazykovoj obŝnosti: tezisy dokladov XXIX sessii postojannoj meždunarodnoj altaističeskoj konferencii (PIAC), II: lingvistika (Moskva: AN SSSR). 1986.
- PIAC 32 Brendemoen, Bernt (ed.) (1990), Altaica Osloensia: proceedings from the 32nd meeting of the Permanent international Altaistic conference (Oslo: Universitetsforlaget).
- PIAC 38 Stary, Giovanni (ed.) (1996), Proceedings of the 38th Permanent international Altaistic conference (Wiesbaden: Harrassowitz).
- PIAC 39 Berta, Árpád (ed.) (1997), The historical and linguistic interaction between Inner-Asia and Europe: proceedings of the 39th Permanent international Altaistic conference (Studia Uralo-Altaica, 39) (Szeged: University of Szeged).
- *PIAC* 41 Janhunen, Juha and Volker Rybatzki (eds.) (1999), Writing in the Altaic world (*Studia orientalia*, 87) (Helsinki).
- RAN Rossijskaja Akademija nauk [Russian Academy of sciences].
- SO Sibirskoe otdelenie [Siberian division].
- StMoStudia Mongolica (Mongolyn sudlal) (Ulaanbaatar). (After eight volumes, the
issues were renumbered from 1, but both the old and new numbers are printed in
each volume. In this bibliography, this is shown as e.g. StMo 10(18), i.e. number
10 in the new series = number 18 in the old series. After StMo 13(21), the series
changed name to Mongolica, again with both the new and old number printed in
the book, indicated in this bibliography as e.g. Mongolica 2(23).)
- ŠUA Šinžlex uhaany akademi [Academy of sciences].
- *ŠUAM Šinžlex uhaany akademijn mèdèè (Bulletin of the Academy of sciences)* (Ulaanbaatar).
- SUS Suomalais-ugrilainen seura (Société finno-ougrienne) (Helsinki).
- SUST Suomalais-ugrilaisen seuran toimituksia (Mémoires de la société finno-ougrienne) (Helsinki).
- TBION Trudy Burjatskogo instituta obŝestvennyh nauk [Working Papers of the Buriad institute of social sciences] (Ulan-Udè). (Special volumes: 2 Issledovanie burjatskih govorov, 2 (1968); 13 Materialy po istorii i filologii Central'noj Azii, 5 (1970); 27 Issledovanija po istorii i filologii Central'noj Azii (1976); 29 Sibirskij fonetičeskij sbornik (1976).)
- TBurNII Trudy Burjatskogo kompleksnogo naučno-issledovateľskogo instituta (Ulan-Udè: SO AN SSSR). (Special volume: 17 Issledovanie burjatskih govorov, 1 (1965).)
- *TPV Teoretičeskie problemy vostočnogo jazykoznanija* [Theoretical problems in oriental linguistics], 5 (Moskva: Nauka). 1982.
- UAJb Ural-Altaische Jahrbücher (24–52 Wiesbaden; 53–4 Berlin).
- UAJb NF Ural-Altaische Jahrbücher, neue Folge (Wiesbaden).

VBF Voprosy burjatskoj filologii (Ulan-Udè: Burjatskij gosudarstvennyj pedagogičeskij institut imeni Dorži Banzarova).

- VGS Voprosy grammatičeskoj sistemy Mongol'skih jazykov [Problems about the grammatical system of Mongolian languages] (Èlista: KNIIIFÈ). 1980.
- VJa Voprosy jazykoznanija (Moskva).
- VLU Vestnik Leningradskogo universiteta (Leningrad).

234	REFERENCES
WPLund	Working papers (Deptartment of linguistics, Lund University).
XMY	Xīběi mínzú yánjiū (North west minorities research) (Lánzhōu).
ZASB	Zentralasiatische Studien (Wiesbaden).
ZBurNII	Zapiski Burjat-Mongol'skogo naučno-issledovatel'skogo instituta kul'tury (Ulan-
	Udè).
ZDMG	Zeitschrift der Deutschen Morgenländischen Gesellschaft.
ZKEl	Bičvr; (Učenye) Zapiski (Èlista: KNIIJaLI).
ZSMY	Zhōngguó shǎoshù mínzú yǔyán [The languages of the minority peoples in
	China] (Chéngdū: Sìchuān mínzú).
ZY	Zhōngguó yǔwén (Běijīng).

DEFEDENCES

Bibliographical references

<u>-----</u>

AALTO, Pentti (1955), 'On the Altaic initial *p*-', *CAJ* 1: 9–16.

- ABAEV, Nikolaj Vjačeslavovič, Regbi Ešievič Pubaev, and Širap Bodievič Čimitdoržiev (eds.) (1989), Cybikovskie čtenija: tezisy dokladov i soobŝenij [Cybikov lectures: abstracts of lectures and presentations] 5 (Ulan-Udè: BION).
- ABAŠEEV, Daniil Altaevič (1951), 'Predvaritel'nyj otčet po obsledovaniju govora burjatmongolov Ivolginskogo i Bajkalo–Kudarinskogo ajmakov' [Preliminary report on the investigation of the dialect of the Buriad Mongols of the Ivalga and Baigal–Hudari regions], ZBurNII 11: 140–53.
- ——(1956), 'Zametki po govoru bajkalo-kudarinskih burjat' [Remarks on the dialect of the Baigal-Hudari Buriads], ZBurNII 21: 132-45.

— (1958), 'Kratkie zamečanija po govoru tunkinskih burjat' [Short remarks on the dialect of the Tünhen Buriads], *Sbornik trudov po filologii*, 3: 269–76 (Ulan-Udè: BMNIIK).

-----(1965), 'Tunkinskij govor' [The Tünhen dialect], TBurNII 17: 3-34.

- ABEL-RÉMUSAT, Jean Pierre (1820), Recherches sur les langues Tartares, ou mémoires sur différens points de la grammaire et de la littérature des Mandchous, des Mongols, des Ouigours et des Tibetains (Paris: Imprimerie Royale).
- ABEMATSU Gen'ichi (1936), *Môkogo dokushûsho, dai-ichi ken: moji no yomikata, kakikata* [Textbook of Mongolian, vol. 1: reading and writing] (Tôkyô: Kôbundô).
- ABERLE, David F. and William M. Austin (1951), 'A lexical approach to the comparison of two Mongol social systems', *Studies in linguistics*, 9: 79–90.
- ADAM, Lucien (1874), De l'harmonie des voyelles dans les langues ouralo-altaïques (Paris: Maisonneuve).
- AHN, Mee-Jin (1999), 'Vowel length-driven syllable weight', NELS 29/2: 11-26.
- ALEKSEEV, Dmitrij Andrianovič (1949), 'Dialekty burjat-mongol'skoga jazyka' [Dialects of Buriad Mongolian], Učenye zapiski Leningradskogo gosudarstvennogo universiteta, 98 (Serija vostokovedčeskih nauk, 1): 161–202.
- ——(1955), 'O territorial'nyh dialektah burjat-mongol'skogo jazyka' [Territorial dialects of Buriad Mongol], in *Materialy naučnoj konferencii po voprosam burjat-mongol'skogo jazyka*, 74–114 (Ulan-Udè: BMNIIK).
- ANDERSON, John M. and Colin J. Ewen (1987), Principles of dependency phonology (Cambridge: Cambridge University Press).
- ANDERSON, Stephen R. (1980), 'Problems and perspectives in the description of vowel harmony', in Vago (1980), 1–48.
- ÁNG Qí (1994), 'Duì Yuán dài Měnggǔ kǒuyǔ yǔyīn de yánjiū' [A study of the phonetics of colloquial Yuan dynasty Mongolian], Nèiměnggǔ shèhuì kēxué, 1994/2: 80–7.

- ARAI Shin'ichi (1960), 'Shin Haruha Môkogo ni okeru gobi boin *a*, *e* no datsuraku genshô ni tsuite' [The phenomenon of deletion of final vowels *a* and *e* in New Halh Mongolian], ÔGDG 8: 51–8.
- ARCHANGELI, Diana B. (1988), Underspecification in Yawelmani phonology and morphology (New York: Garland).
- and Douglas G. Pulleyblank (1994), *Grounded phonology* (Cambridge, MA: MIT Press).
- ARDAJAB, Eldengtei-yin (1994), 'Dagur kelen-ü üge-yin ekin-ü šürgükü h abiy-a büküi üges' [Dagur words with the initial fricative h], ÖMBYS 1994/4: 58–63.
- AUSTIN, William M. (1952), 'A brief outline of Dagur grammar', *Studies in linguistics*, 10: 65–75.
- ——Gombojab Hangin, and Urgunge Onon (1956), A Mongol reader (Washington: American council of learned societies).
- BAATAR, Jadambatyn (1987), 'Zavhany sartuulyn salbar aman ajalguuny avia züjn zarim onclogoos' [Some phonetic characteristics of the sub-dialect of the Zavhan Sartuul], *HZS* 16: fasc. 14 (pp. 137–42).
- BABUEV, Sergej Damdinovič (1991), 'Fonetičeskie osobennosti ol'honskogo govora' [Phonetic characteristics of the Ol'hoon dialect], in Šagdarov (1991), 105–11.
- [BADARANGG-A] Bādárónggé (1988), 'Dáwò'ěryů yů Měnggůyů yìtóng bǐjiào: jiān tán Dáwò'ěryů de xìshů' [Comparison of Dagur and Mongolian: with an addendum on the classification of Dagur], MY 1988/6: 61–7.
- BADGAEV, Nikolaj Boktaevič (1987), 'K voprosu o spontannom peredviženii soglasnyh v burjatskom jazyke' [The problem of spontaneous consonant shift in Buriad], in A. Ju. Ajhenval'd and S. V. Volkov (eds.), *Tezisy konferencii aspirantov i molodyh naučnyh* sotrudnikov, 2: 12–14 (Moskva: Nauka).
- ——(1988a), 'Altajskie affrikaty c i 3 v svete areal'noj lingvistiki' [The Altaic affricates c and 3 in the light of areal linguistics], in Ubrjatova and Seljutina (1988), 154–8.
- —(1988b), 'Iz istorii razvitija mongol'skih affrikat' [From the developmental history of Mongolian affricates], in V. I. Podlesskaja (ed.), Sinhronija i diahronija v lingvističeskih issledovanijah, 1: 20–36 (Moskva: Nauka).
- ——(1989a), 'Affrikata č v mongol'skih jazykah' [The affricate č in Mongolian languages], in Šagdarov (1989), 74–83.
- —(1989b), 'O sisteme affrikat caatanskogo podgovora torgutskogo govora kalmyckogo jazyka' [The system of affricates in the Tsaatan sub-dialect of the Torguud dialect of Kalmuck], in Abaev, Pubaev, and Čimitdoržiev (1989), 153–5.
- (1989c), 'O prevraŝenii smyčnyh v affrikaty pered glasnymi perednego rjada v mongol'skih jazykah' [On the change of stops to affricates before front vowels in Mongolian languages], in *Vladimircovskie čtenija: tezisy dokladov i soobŝenij* 2: 72–4, (Moskva: Nauka).
 - —(1990), *Sistema affrikat v sovremennyh mongol'skih jazykah v areal'nom osveŝenii* [The system of affricates in modern Mongolian languages in an areal perspective], Avtoreferat kand. diss. (Ulan-Udè: BION).
- and Èllara Uljaevna Omakaeva (1986), 'K areal'noj harakteristike zvukoperehoda $s > h (s > h > \emptyset)$ v altajskih jazykah' [The areal characterization of the sound shift $s > h (s > h > \emptyset)$ in Altaic languages], *PIAC* 29: 15–17.
- BADMADORŽ, Dügèržavyn (1984), 'K voprosu o razvitii vokalizma ojratskogo dialekta na territorii MNR: redukcija glasnyh i monoftongizacija diftongov, Èletskij govor' [The development of the vocalism of Oirad dialects in the MPR: vowel reduction and monophthongization of diphthongs, Ööld dialect], MUISÈ 1984/3: 61–73.

- [BADMADORŽ, Dügèržavyn] Badamdorž (1987a), 'Ööld salbar aman ajalguuny zarim onclog' [Some characteristics of the Ööld subdialect], HZS 20: fasc. 7 (pp. 98–107).
- [——] Badamdorž (1987*b*), 'BNMAU dah' Ööldijn aman ajalguuny urt ègšgijn asuudald'; Badamudoruji, 'Mongoru Jinmin Kyôwakoku no Ûrudo hôgen ni okeru chôboin ni kansuru mondai' [The problem of long vowels in the Ööld dialect of the MPR], *MK* 18: 114–16; 117–20.
- BADMAEV, Andrej Radnažapovič (1980*a*), 'Malošumnye soglasnye 2-oj i 3-ej artikuljacij v congol'skom dialekte burjatskogo jazyka (po dannym statičeskogo rentgenografirovanija)' [Dental and palatal sonorants in the Tsongool dialect of Buriad (by static X-ray data)], in E. I. Ubrjatova (ed.), *Sibirskij dialektologičeskij sbornik*, 128–35 (Novosibirsk).
- ——(1980b), 'Soglasnye 4-oj artikuljacii v congol'skom dialekte burjatskogo jazyka (po dannym statičeskogo rentgenografirovanija)' [Velar consonants in the Tsongool dialect of Buriad (by static X-ray data)], in Nadeljaev (1980), 150–8.
- BAJARSAJHAN, Endonžamcyn, Gončigijn Gantogtoh, D. Dorž, G. Mižiddorž, Baldandoržijn Sum"jaabaatar, Osornamžilyn Süxbaatar, Anna Angadykovna Darbeeva, Grigorij Cerenovič Pjurbeev, and Zoja Vasil'evna Ševernina (2001–2), *Bol'šoj akademičeskij mongol'sko-russkij slovar'* [Large academic Mongolian–Russian dictionary], 4 vols (Moskva: Academia).
- BAJARSÜRÈN, Cèrèngijn (1978), 'Mongol hèlènd orson har' garaltai ügijn duudlagyn tuhaj' [The pronunciation of foreign words in Mongolian], *HZS* 13: fasc. 10 (pp. 82–6).
- BAJČURA, Uzbek Š. (1961), Zvukovoj stroj tatarskogo jazyka v svjazi s nekotorymi drugimi tjurkskimi i finno-ugorskimi jazykami, čast' 2 [The sound structure of Tatar in relation to some other Turkic and Finno-Ugric languages, part 2] (Kazan': Izd. Kazanskogo universiteta).
- ——(1971), 'Instrumental'nye dannye ob udarenii i intonacii v altajskih jazykah' [Instrumental data on stress and intonation in Altaic languages], in O. P. Sunik (ed.), *Problema obŝnosti altajskih jazykov*, 279–92 (Leningrad: Nauka).
- (1974), 'O slogovoj strukture slova v altajskih jazykah sravnitel'no s ural'skimi i indoevropejskimi po statističeskim i instrumental'nym dannym' [Syllable structure in Altaic languages compared to Uralic and Indoeuropean according to statistical and instrumental data], *PIAC* 12: 77–87.
- [-----] Baitchura (1978), 'Instrumental-phonetic data on the word stress and intonation in Khakas and Buryat', *CAJ* 22: 1–31.
- BALAGUNOVA, Sandulma Sanžievna (1986), 'O fonetičeskoj interferencii v burjatskoj reči treh"jazyčnyh tatar' [Phonetic interference in the Buriad speech of trilingual Tatars], in Rassadin (1986), 69–74.
- ------(1987*a*), 'Zvukovye osobennosti govora osinskih burjat' [Phonetic characteristics of the dialect of the Osa Buriads], in Rassadin (1987*a*), 94–101.
- ——(1987b), 'O jazykovyh sredstvah vyraženija modal'nosti v burjatskom jazyke' [Linguistic means of expressing modality in Buriad], in Rassadin (1987b), 46–62.
- ——(1989), 'O rečevoj interferencii u treh"jazyčnyh tatar Irkutskoj oblasti' [Phonetic interference among trilingual Tatars in the Irkutsk district], in Abaev, Pubaev, and Čimitdoržiev (1989), 155–6.
- BALDAN, Luvsangončigijn and Žavzangijn Coloo (1974), 'Bajad ajalguu, aman zohiol sudalsan zarim dün' [Some results of research on the Bayad dialect and folklore], ŠUAM 1974/3: 22–9.
- BALDANŽAPOV, Purbo Baldanovič (1962*a*), Jirüken-ü tolta-yin tayilburi: *mongol'skoe* grammatičeskoe sočinenie XVIII veka [Jirüken-ü tolta-yin tayilburi: a Mongolian grammatical treatise of the 18th century] (Ulan-Udè: BKI).

— (1962*b*), Kelen-ü čimeg: *mongol'skoe grammatičeskoe sočinenie XIX veka* [*Kelen-ü čimeg*: a Mongolian grammatical treatise of the 19th century] (Ulan-Udè: BKI).

- BÁLINT, Gábor (1877), Az éjszaki burját-mongol nyelvjárás rövid ismertetése [Short presentation of the northern Buriad Mongolian dialect] (Budapest).
- BANZARGARUDI (1998), 'Erten-ü Monggol bicig-ün kelen-ü oncalig sinji-yi qadagalaju bayig-a Ordos aman ayalgu' [The Ordos dialect which preserves characteristics of Old Written Mongolian], *MKUJ* 1998/1: 13–19.
- Bào Huáiqiào and Chén Jiāyóu (1998), 'Měnggǔyǔ yǔyīn shēngxué cānshù shùjùkù' [Parameter database for Mongolian acoustic phonetics], MSYQ 3: 151–3.
 - and [Köke] Hūhé (2000), 'Měnggǔyǔ jùzi yùnlǜ tèzhēng shēngxué fēnxī' [Acoustic analysis of Mongolian prosodic features], in *The fourth Seoul international Altaistic conference*, 193–7.
 - and Lů Shìnán (1992), 'Měnggùyù Cháhā' ěr huà yuányīn söngjǐn de shēngxué fēnxī' [Acoustic analysis of vowel tenseness in the Chahar dialect of Mongolian], MY 1992/1: 61–8.
- BĀO Xiánglín (1985), 'Měnggǔ shūmiànyǔ yuányīn i zài Dù'ěrbótè fāngyīn zhōng de yǎnbiàn' [The development of the Written Mongolian vowel i in the Dörbed dialect], Zhōngyāng mínzú xuéyuàn xuébào, 1985/4: 96.
- BARANGOVA, Nudla Nikolaevna and Tamara Sarangovna Esenova (1990), 'Nabljudenija nad pauzaciej razgovornoj reči kalmykov' [Observations on pausing in colloquial Kalmuck], in È. Č. Bardaev (ed.), *Puti soveršenstvovanija prepodavanija rodnogo jazyka*, 60-8 (Èlista: KNIIIFÈ).
- BATUGEREL (1989), 'Alaša aman ayalgun-daki abiyan-u jarim qubiral-un tuqai' [Some sound changes in the Alshaa dialect], *MKUJ* 1989/3: 40–51.
- BATUJIRGAL, N. (1981), 'Ordos aman ayalgun-u üge-yin ergülte-yin tuqai tursin ögülekü ni' [A note on word stress in the Ordos dialect], *MKUJ* 1981/3: 31–43.
- BATUSAYIQAN (1985), 'Urad aman ayalgun-u abiyalaburi-yin jarim oncalig' [Some characteristics of the phonemes of the Urad dialect], *MKB* 1985/1: 77–86.
- BAWDEN, Charles R. (1997), *Mongolian–English dictionary* (London: Kegan Paul International).
- BAYANBAGATUR (1987), 'Dumdadu jagun-u Monggol kelen-deki üge-yin ekin-ü *h*-un irelte ba tegün-ü baragsan siltagan-u tuqai' [The origin of word-initial *h* in Middle Mongolian and reasons for its disappearance], *ÖMBYS* 1987/2: 72–83.
- —(1994), 'Dumdadu jagun-u Monggol kelen-ü egesig abiyan-u sistèm-ün tuqai' [The vowel system of Middle Mongolian], in Soyol erdem-ün sudulul: Öbör Monggol-un bagsi-yin yeke surgaguli bayigulugdagsan döcin jil-ün oi-du jorigulba, 103–31 (Köke-qota: ÖMSK).
- BAYANCOGTU (1962), 'Bagarin, Caqar qoyar aman ayalgun-u urugsilagsan egesig' [Fronted vowels in the Baarin and Chahar dialects], *ÖMYSS* 1962/1: 61–90.
 - ——(1978), 'Odo üy-e-yin Monggol kelen-ü abiyan-u sistèm' [The sound system of modern Mongolian], ÖMYSS 1978/3: 97–137.
 - ——(1981a), 'Bicig-ün kelen-ü üge-yin ekin-ü busu üy-e-yin urugul-un egesig-ün tuqai' [Non-initial rounded vowels in the written language], ÖMYSS 1981/1: 38–53.
 - —(1981b), 'Monggol kelen-ü i egesig-ün tuqai' [The vowel i in Mongolian], ÖMYSS 1981/2: 129–40.
 - ——(1981c), 'Egesig jokicaqu yoson-u tuqai jarim asugudal' [Some problems about vowel harmony], ÖMYSS 1981/3: 70–87.
 - ——(1982a), 'Monggol kelen-ü urtu egesig-ün dagudalg-a-yin tuqai' [The pronunciation of long vowels in Mongolian], ÖMYSS 1982/1: 45–51.

- BAYANCOGTU (1982b), 'Monggol kelen-ü egesig-ün cingg-a köndei-yin tuqai' [Tenseness in Mongolian vowels], *ÖMYSS* 1982/1: 155–62.
 - —(1987), 'Monggol kelen-ü üge-yin ergülte-yin tuqai' [Word stress in Mongolian], ÖMYSS 1987/4: 1–19.

—(2001), 'Qorcin aman ayalgun-u abiyan jüi-yin tuqai' [Phonetics of the Horchin dialect], *ÖMYSS* 2001/1: 13–32.

- BAYAR (2000), 'The long vowels in the *Secret history of Mongols* and the ways of transliterating them', in Š. Bira (ed.), Mongolyn nuuc tovčoo-*ny 750 žilijn ojd zoriulsan olon ulsyn baga hural*, 9–13 (Ulaanbaatar: IAMS).
- BAYARMENDÜ, Borjigin (1989), 'Monggol kelen-ü jigeleke üge-yin tuqai' [Loan-words in Mongolian], *MKB* 1989/4: 15–24.
 - ——(1997*a*), *Bagarin aman ayalgun-u sudulul* [Studies of the Baarin dialect] (Kökeqota: Öbör Monggol-un arad-un keblel-ün qoriy-a).
- ——(1997b), 'Bagarin, Qorcin aman ayalgun-u üge-yin ergülte-yin tuqai' [Word stress in the Baarin and Horchin dialects], *MKB* 1997/6: 48–52.
 - —(1997c), 'Bagarin, Caqar, Qorcin aman ayalgun-daki tüsiglekü egesig bolon tegün-ü ejikü asagudal-un tuqai' [Dependent vowels in the Barin, Chahar, and Horchin dialects and their centralization], ÖMYSS 1997/3: 30–7.
- ——(1998a), 'Bagarin, Caqar, Qorcin aman ayalgun-u urugsilagsan egesig-ün tursilta-yin abiyan jüi-yin qaricagulul' [An experimental phonetic comparison of front vowels in the Baarin, Chahar, and Horchin dialects], ÖMYSS 1998/4: 1–5.
- [-----] Bayarmend (1998b), 'Acoustic analysis of Mongolian vowels', AH 8: 155-76.
- BAYASQAL (1987), 'Cayidam nutug-un ayalgu-daki geyigülügci *x*, *k*-yin tuqai' [The consonants *x*, *k* in the Chaidam dialect], *ÖT* 1: 337–49.
- BAZAROV, Boris Vandanovič and Mihail Innokent'evič Tulohonov (eds.) (1998), *Filologičeskij sbornik pamjati V. C. Najdakova* [Philological collection to the memory of V. C. Najdakov] (Ulan-Udè: BNC).
- BEFFA, Marie-Lise and Roberte Hamayon (1975), Éléments de grammaire mongole (Paris: Dunod).

——(1983), 'Les langues mongoles', Études mongoles . . . et sibériennes, 14: 121–69.

- BENZING, Johannes (1953), Einführung in das Studium der altaischen Philologie und der Turkologie (Wiesbaden: Harrassowitz).
- (1985), Kalmückische Grammatik zum Nachschlagen (Wiesbaden: Harrassowitz).
- BERNHARDT, Karl A. (1999), 'Teaching English to Mongolian L1 speakers: some notes on phonotactics', *Mongolian studies*, 22: 19–34.
- BERTAGAEV, Trofim Alekseevič (1936), 'Zametki lingvista o horinskom govore: otčet o poezdke v Horinskij ajmak s 30/III po 10/IV s. g.' [A linguist's remarks on the Hori dialect: account for a tour to the Hori district March 30 to April 10 this year], Zapiski BM GIJaLI, 1: 5–21.
 - —(1947), 'Vlijanie russkogo jazyka na razvitie fonetičeskoj sistemy burjat-mongol'skogo jazyka' [The influence of Russian on the development of the phonetic system of Buriad Mongolian], Učenye zapiski burjat-mongol'skogo gosudarstvennogo pedagogičeskogo instituta, 1: 80-4.
 - —(1968*a*), 'Burjatskij jazyk' [Buriad], in Skorik (1968), 13–33.
- (1968*b*), 'Mongolskie jazyki: vvedenie' [Mongolian languages: introduction], in Skorik (1968), 7–12.
- ——(1971), 'Količestvennye čeredovanija glasnyh i ablaut v mongol'skih jazykah' [Quantity changes of vowels and ablaut in the Mongolian languages], VJa 1971/6: 116–21.

BESE, Lajos (1961), 'Remarks on a western Khalkha dialect', AOH 13: 277-94.

(1962), 'Contributions to the Khori-Buriat sub-dialect of Ivolga', AOH 15: 15–21.

- [——] Bèšè (1964), 'Ob affrikatah horinskogo dialekta' [Affricates in the Hori dialect], KSINA 83: 46–8.
- BIMBA (1981), 'Qobog Sayiri-yin Torgud aman ayalgun-u egesig jokicaqu yoso' [Vowel harmony in the Torguud dialect of Hobogsair], ÖMYSS 1981/2: 73–82.
- BINNICK, Robert I. (1969), 'Non-high neutral vowels in modern Mongolian vowel harmony', CLS 5, 295-301.

—(1980), 'The underlying representation of harmonizing vowels: evidence from modern Mongolian', in Vago (1980), 113–34.

- (1987), 'On the classification of the Mongolian languages', CAJ 31: 178–95.
- (1991), 'Vowel harmony loss in Uralic and Altaic', in W. G. Boltz and M. C. Shapiro (eds.), *Studies in the historical phonology of Asian languages*, 35–52 (Amsterdam: Benjamins).
- ——(1992), 'Mongolian languages', in W. Bright (ed.), International encyclopedia of linguistics, 2: 434–8 (New York: Oxford University Press).
- BIRTALAN, Ágnes (2003), 'Oirat', in Janhunen (2003f), 210–28.
- BITKEEV, Petr Cedenovič (1964), 'Nejasnye glasnye kalmyckogo jazyka (v porjadke obsuždenija)' [Reduced vowels in Kalmuck (presented for discussion)], ZKEl 3: 69–91.
- ——(1965), Soglasnye fonemy kalmyckogo jazyka (na materiale èksperimental'nyh issledovanij) [Kalmuck consonant phonemes (based on experimental investigations)] (Ulan-Udè: BKI).
- (1968), 'Fonetičeskie pričiny singarmoničeskoj transformacii slov v mongol'skih jazykah' [Phonetic reasons for vowel harmony transformations of words in Mongolian languages], *Izvestija SO AN SSSR*, 1968/11 (*serija obŝestvennyh nauk*, 3): 123–9.
- ——(1970), 'Kvantitativnye izmenenija glasnyh v mongol'skih jazykah' [Quantitative vowel changes in Mongolian languages], *TBION* 13: 41–51.

—(1973), 'Tverdorjadnyj glasnyj ï v mongol'skom jazyke Tuvy' [The hard vowel ï in Tuva Mongolian], ZKEl 11: 110–17.

- —(1975*a*), *Problemy fonetiki kalmyckogo jazyka (kvantitativnye i kvalitativnye izmenenija glasnyh)* [Problems of Kalmuck phonetics (quantitative and qualitative vowel alternations)] (Èlista: KKI).
- —(1975b), 'Mjagkie soglasnye v altajskih jazykah' [Palatalized consonants in Altaic languages], in Bitkeev (1975c), 340–53.
- (ed.) (1975c), Problemy altaistiki i mongolovedenija, vyp. 2, serija lingvistiki: materialy vsesojuznoj konferencii, Èlista, 17–19 maja 1972 g. [Problems of Altaic and Mongolian studies, vol. 2, linguistics: materials from the All-union conference, Elista, May 17–19, 1972] (Moskva: Nauka).
- —(1976), 'K voprosu ob orfoèpičeskih normah sovremennogo kalmyckogo jazyka' [The problem of orthoepic norms in modern Kalmuck], in M. U. Monraev (ed.), Voprosy grammatiki i leksikologii sovremennogo kalmyckogo jazyka, 148–55 (Moskva: Nauka).
- —(1978), 'Sravnitel'no-tipologičeskaja harakteristika glasnyh fonem mongol'skogo i kalmyckogo jazykov' [Comparative and typological characterization of vowel phonemes in Mongolian and Kalmuck], in P. C. Bitkeev (ed.), *Issledovanija po mongol'skoj filologii*, 97–106 (Moskva: Nauka).
- (1980), 'Slovesnoe udarenie v kalmyckom jazyke i mongol'skom jazyke Tuvy' [Word stress in Kalmuck and Tuva Mongolian], VGS 65–92.
- ——(1981), 'Osnovnye zakonomernosti razvitija kalmyckih glasnyh' [Basic regularities in the development of Kalmyk vowels], in E. I. Ubrjatova (ed.), *Jazyk i fol'klor narodov Severa*, 178–83 (Novosibirsk: Nauka).

BITKEEV, Petr Cedenovič (1982), 'Tipologičeskaja harakteristika sistem glasnyh fonem mongol'skogo i kalmyckogo jazykov' [Typological characterization of the systems of vowel phonemes in Mongolian and Kalmuck], *TPV* 18–25.

—(1983), 'Fonetika' [Phonetics], in G. D. Sanžeev (ed.), *Grammatika kalmyckogo jazy*ka: fonetika i morfologija, 4–47 (Èlista: KKI).

——(1985*a*), 'Problema dolgotnosti v fonologičeskoj sisteme ojratskogo jazyka' [The problem of length in the Oirad phonological system], *VJa* 1985/5: 111–20.

- (1985*b*), 'Sravnitel'naja tipologija vokalizma v mongol'skih jazykah' [Comparative typology of the vocalism in Mongolian languages], *OUMÈ 4 IH* 2: 198–206.
- —(1988), 'Principy fonologičeskoj rekonstrukcii i sistema fonem ojratskogo jazyka XVII v.' [The principles of phonological reconstruction and the phoneme system of 17th century Oirad], in P. C. Bitkeev (ed.), *Problemy mongol'skoj filologii*, 3–9 (Èlista: KNIIIFÈ).
- BJURAEVA, Èleonora Ivanovna (1976*a*), 'Melodemy prostyh nerasprostranennyh povestvovatel'nyh dialogičeskih predloženij v horinskom dialekte burjatskogo jazyka' [Melodemes of simple narrative dialogue sentences in the Hori dialect of Buriad], *TBION* 27: 45–57.
- [----] Konjaeva (1976*b*), 'Melodemy prostyh nerasprostranennyh povestvovatel'nyh monologičeskih predloženij v burjatskom jazyke' [Melodemes of simple narrative monologue sentences in Buriad], *TBION* 29: 99–122.
- [—] Konjaeva (1976c), 'Melodemy dvusostavnyh nerasprostranennyh povestvovatel'nyh dialogičeskih predloženij burjatskogo jazyka' [Melodemes of simple narrative dialogue sentences with two elements in Buriad], in E. I. Ubrjatova (ed.), *Issledovanija po fone-tike sibirskih jazykov*, 137–52 (Novosibirsk: IIFF).
- [-----] Konjaeva(1977), 'Pobuditel'nye melodemy govora horinskih burjat (èksperimental'nofonetičeskoe issledovanie)' [Imperative melodemes in the dialect of the Hori Buriads (experimental phonetic investigation)], in Cydendambaev (1977), 95–113.
 - ——(1978), *Ritmomelodika prostyh nerasprostranennyh predloženij burjatskogo jazyka* [Rhythmomelodics of simple sentences in Buriad] (Ulan-Udè: BKI).
- ——(1983), 'Rol' intonacii v oformlenii pobuditel'nyh vyskazyvanij burjatskogo jazyka' [The role of intonation in the formation of imperative utterances in Buriad], in Nadeljaev (1983), 156–8. [Different from 1986b]
- ——(1986a), 'Intonacionnaja harakteristika povestvovanija v burjatskom jazyke' [Intonation characteristics of Buriad narration], in Rassadin (1986), 43–52.
- ——(1986b), 'Rol' intonacii v oformlenii pobuditel'nyh vyskazyvanij burjatskogo jazyka' [The role of intonation in the formation of imperative utterances in Buriad], in Nadeljaev (1986), 140–3.
- ——(1986c), 'Osnovnye tipy intonacionnyh edinic i ih upotreblenie v burjatskom jazyke' [Basic types of intonation units and their use in Buriad], in E. I. Ubrjatova (ed.), *Fonetičeskie struktury v sibirskih jazykah*, 116–24 (Novosibirsk: IIFF).
- (1987), 'Intonacionnye oformlenie pričastnyh i deepričastnyh oborotov' [Intonational form of participle and gerund constructions], in Rassadin (1987*a*), 15–36.
- (1988), 'Sootnošenie intonacionnyh i sintaksiko-segmentnyh edinic burjatskogo jazyka' [The correspondence between intonational and syntactic-segmental units in Buriad], in Ubrjatova and Seljutina (1988), 100–11.
- —(1989), 'Prosodičeskie sredstva vyraženija sub"ektivnoj modal'nosti v burjatskom jazyke' [Prosodic means of expressing subjective modality in Buriad], in Šagdarov (1989), 84–100.
- —(1990), 'O ritmomelodičeskih osobennostjah burjatskih uligerov' [Rhythmomelodic characteristics of Buriad tales], in Džangar i problemy èpičeskogo tvorčestva: tezisy dok-

ladov i soobŝenij meždunarodnoj naučnoj konferencii (22–24 avgusta 1990 goda), 213–15 (Èlista: Kalmyckij institut obŝestvennyh nauk).

—(1991), 'Rol' intonacii v formirovanii pragmatičeskih tipov vyskazyvanij v burjatskom jazyke' [The role of intonation in the formation of pragmatic types of utterances in Buriad], in Šagdarov (1991), 121–34.

— (1996), 'Intonacionnye osobennosti zapadnogo dialekta burjatskogo jazyka (na materiale bohanskogo govora)' [Intonational characteristics of the western dialect of Buriad (based on data from the Boohon dialect)], in Rassadin (1996*b*), 34–44.

- (1998), 'Intonacija kak sredstvo vyraženija sintaksičeskih svjazej v predloženii (na materiale burjatskogo jazyka)' [Intonation as a means of expressing syntactic relations in a sentence (based on Buriad data)], in Bazarov and Tulohonov (1998), 237–41.
- (1999), 'Intonacionnye risunki kommunikativnyh edinic burjatskogo jazyka' [Intonation patterns of communicative units in Buriad], in V. I. Rassadin (ed.), *Istorija razvitija* mongol'skih jazykov, 92–120 (Ulan-Udè: BNC).

— and Elena Semenovna Pavlova (1976), 'Melodemy prostyh nerasprostranennyh obŝevoprositel'nyh predloženij v burjatskom jazyke' [Melodemes of simple yes–no questions in Buriad], *TBION* 29: 123–46.

— (1989), 'Sootnošenie intonacii i leksiko-grammatičeskih sredstv pri vyraženii semantičeskoj suŝnosti vyskazyvanija' [The relation between intonation and lexico-grammatical means for expressing the semantic nature of an utterance], in Abaev, Pubaev, and Čimitdoržiev (1989), 156–9.

BLÄSING, Uwe (2003), 'Kalmuck', in Janhunen (2003f), 229-47.

- BOBROVNIKOV, Aleksandr Il'ič (1835), *Grammatika mongol'skago jazyka* [Grammar of Mongolian] (Sankt-Peterburg: Tipografija svjatago pravitel'stvujuŝego sinoda).
- BOBROVNIKOV, Aleksej Aleksandrovič (1849), Grammatika mongol'sko-kalmyckago jazyka [Grammar of the Mongolian–Kalmuck language] (Kazan': Universitetskaja tipografija).
- Вöке (1982), 'Düngsiyang kelen-ü egesig jokicaqu yoson-u tuqai' [Vowel harmony in Santa], *ÖMYSS* 1982/4: 120–33.
- (1983*a*), *Dōngxiāngyǔ cíhuì* [Santa vocabulary] (*MTKASC* 008).
- [——] Bùhé (1983*b*), 'Dōngxiāngyǔ de yuányīn héxié xiànzhuàng tànxī' [Preliminary analysis of Santa vowel harmony], *MY* 1983/4: 24–30.
- ——(1985), 'Düngsiyang kelen-ü qamar-un geyigülügci n kiged η-ün tuqai' [The nasal consonants n and η in Santa], ÖMYSS 1985/2: 1–6.
- ——(1986), Dōngxiāngyǔ hé Měnggǔyǔ [Santa and Mongolian] (MTKASC 007).
- ——(1987), 'Afganistan-u Mogol kelen-ü abiyan jüi: sudululgan-u bayidal-i qamtu-duni tanilcagulqu ni' [Phonetics of Moghol in Afghanistan: with an addendum on the study of Moghol], ÖMYSS 1987/3: 89–105.
- (1994), 'Jalayid aman yariyan-u jarim oncalig-un tuqai' [Some features of the Jalaid dialect], *MKB* 1994/1: 18–25, 60.
- (1995), 'Monggol töröl kelen-ü <u>K</u>angjiya aman yariy-a-yi sinjilekü ni' [A study of the Mongolian spoken language Kangjia], *ÖMYSS* 1995/4: 1–11.
- ——(1996), *Mogol kelen-ü sudulul* [Moghol studies] (Kökeqota: ÖMYS-yin keblel-ün qoriy-a).
- [——] Bùhé (1997), 'Mògēlèyǔ gàiyào' [A sketch of Moghol], in Jalsan and Badmaodser (1997), 320–31.
 - —(1998), 'Monggor, Düngsiyang, Boo An kelen-ü qaricagan-u tuqai sinjilege' [Investigation of the relations between Monguor, Santa, and Bonan], *MSYQ* 3: 1–4.
- (1999), 'Monggol kelen-ü nutug-un ayalgud-un $\mathfrak{I}, \mathfrak{I}$ egesig-ün soligdal' [Alternation of the vowels \mathfrak{I} and \mathfrak{I} in Mongolian dialects], *MKB* 1999/4: 11–17.

- BÖKE (2001), 'Monggor bülüg kelen-ü abiyan-u jarim üjegdel-ün qaricagulul' [Comparison of some phonetic phenomena in languages of the Monguor branch], *ÖMYSS* 2001/1: 1–12.
- [——] Bùhé and Chén Năixióng (1981), 'Tóngrén Bǎo'ānhuà gàiyào' [Sketch of Tongren Bonan], *MY* 1981/2: 61–75.
- [-----] Bùhé and Liú Zhàoxióng (1982), *Bǎo'ānyǔ jiǎnzhì* [Short description of Bonan] (Běijīng: Mínzú).
- Boo Liyan-ciyün (1997), 'Tariyaci-yin yariyan-u $y \sim y$: egesig-ün tuqai' [The vowels $y \sim y$: in the Tariach dialect], *MKB* 1997/2: 44–6.
- BOOLJ, Geert (1992), 'Lexical phonology and prosodic phonology', in W. U. Dressler et al. (eds.), *Phonologica 1988*, 49–62 (Cambridge: Cambridge University Press).
- BOOSIYANG (1981), 'Qagucin Bargu aman ayalgun-u geyigülügci $s \sim x$ -ün tuqai' [The consonants $s \sim x$ in the Old Bargu dialect], *ÖMYSS* 1981/2: 121–8.
- and B. Jirannige (1996), Bargu aman ayalgu [The Bargu dialect] (MTKASC 001).
- BORMANSHINOV, Arash (1961), Kalmyk manual (Washington: American council of learned societies).
- BOSSON, James E. (1964), *Modern Mongolian: a primer and reader (Uralic and Altaic series*, 38) (Bloomington: Indiana University).
- and B. Unensečen (1962), 'Some notes on the dialect of the Khorchin Mongols', in N. Poppe (ed.), *American studies in Altaic linguistics (Uralic and Altaic Series*, 13), 23–44 (Bloomington: Indiana University).
- BUDAEV, Bal'žinima Žambalovič (1979), 'Tonal'nost' v slovesnom udarenii burjatskogo jazyka' [Tonality in Buriad word stress], in Nadeljaev (1979), 121–45.
 - ——(1981), Akcentuacija burjatskogo jazyka: slovesnoe udarenie v horinskom dialekte [Accentuation in Buriad: word stress in the Hori dialect] (Moskva: Nauka).
 - ——(1983), 'Ritmika odnoslogov v burjatskom jazyke' [The rhythmics of monosyllables in Buriad], in Nadeljaev (1983), 153–5.
 - ——(1984), 'Slovesnaja akcentuacija burjatskogo jazyka' [Word accentuation in Buriad], in Nadeljaev (1984), 149–53.
- (1986*a*), 'Sintagmatičeskoe udarenie v burjatskom jazyke: ponjatie sintagmy, otličie ee ot slova, slovosočetanija i frazy' [Syntagmatic stress in Buriad: the concept of syntagm, its differentiation from word, word compound, and phrase], in Rassadin (1986), 52–8.
- (1986*b*), 'Slovesnoe udarenie v potoke reči (na materiale horinskogo dialekta burjatskogo jazyka)' [Word stress in the speech stream (based on the Hori dialect of Buriad)], in Nadeljaev (1986), 122–4.
- (1987*a*), 'Slogovoj akcent v burjatskom jazyke' [Syllable accent in Buriad], in Rassadin (1987*a*), 37–49.
 - (1987*b*), 'Slovosočetatel'noe udarenie v burjatskom jazyke (na materiale atributivnyh slovosočetanij)' [Stress of word combinations in Buriad (based on attributive word combinations)], in E. I. Ubrjatova (ed.), *Jazyki Sibiri i Mongolii*, 164–7 (Novosibirsk: IIFF).
- ——(1989), 'Pozicionnaja zakreplennost' udarenija v burjatskom jazyke' [Positional fixation of stress in Buriad], in Šagdarov (1989), 131–6.
- ——(1991), 'Slovesnoe udarenie v mongol'skom jazyke (fizičeskie harakteristiki)' [Word stress in Mongolian (physical characteristics)], in Šagdarov (1991), 150–60.
- BUDAEV, Cyrendaša Badmaevič (1962), 'Fonetičeskie osobennosti congol'skogo govora' [Phonetic characteristics of the Tsongool dialect], *KSBurNII* 4: 122–7.
 - ----(1965), 'Congol'skij govor' [The Tsongool dialect], TBurNII 17: 151-86.

—(1977*a*), 'Severnoselenginskij govor burjatskogo jazyka (opyt sravnitel'nostatističeskogo analiza materialov, sobrannyh po programme dialektologičeskogo atlasa)' [The North Selenge dialect of Buriad (comparative-statistical analysis of materials collected in the dialect atlas program)], in Cydendambaev (1977), 8–39.

- —(1977b), 'Formirovanie južnoburjatskih govorov kak rezul'tat vzaimodejstvija mongol'skogo i burjatskogo jazykov' [The formation of South Buriad dialects as the result of the interaction of Mongolian and Buriad], OUMÈ 3 IH 2: 32–7.
- ——(1978), Leksika burjatskih dialektov v sravnitel'no-istoričeskom osveŝenii [The lexicon of Buriad dialects from a comparative historical perspective] (Novosibirsk: Nauka).
- (1980), 'Nekotorye nabljudenija nad govorami central'nyh halhascev i južnoselenginskih burjat' [Some observations on the Central Halh and South Selenge Buriad dialects], in Š. B. Čimitdoržiev (ed.), *Issledovanija po kul'ture narodov Central'noj Azii*, 73–8 (Ulan-Udè: BION).
- ——(1986), 'Izmenenija v zvukovom sostave burjatskih dialektov v XVIII–XIX vv.' [Changes in the sound structure of Buriad dialects in the 18th–19th centuries], in Nadeljaev (1986), 159–62.
- ——(1987), 'Rannemongol'skie èlementy v sovremennom burjatskom jazyke' [Early Mongolian elements in contemporary Buriad], in Solncev (1987), 2: 14–20.
- ——(1992), Burjatskie dialekty (opyt diahroničeskogo issledovanija) [Buriad dialects (diachronic investigation)] (Novosibirsk: Nauka).
- —(1993), 'Iz istorii složenija dialektnoj sistemy sovremennogo burjatskogo jazyka' [From the history of the formation of the modern Buriad dialect system], in Rassadin (1993), 37–58.
- BUHAEVA, Irina Doržievna (1980), 'O kategorii tverdosti-mjagkosti burjatskogo jazyka v russkih zaimstvovanijah' [The Buriad category hard-soft in Russian loans], in *Voprosy burjatskoj filologii i literaturovedenija*, 44–53 (Irkutsk).
- BUHAEVA, Oktjabrina Doržievna (1989), 'O normah sočetaemosti soglasnyh fonem v russkom i burjatskom jazykah' [Norms for the phonotactics of consonant phonemes in Russian and Buriad], in Šagdarov (1989), 32–42.
 - —(1991a), *Tipologičeskoe sopostavlenie fonem i ih sočetaemosti v russkom i burjatskom jazykah* [Typological comparison of phonemes and their phonotactics in Russian and Buriad] (Novosibirsk: Izd. Novosibirskogo universiteta).
- ——(1991b), 'O segmentnoj organizacii slova v burjatskom i russkom jazykah' [The segmental organization of the word in Buriad and Russian], in Šagdarov (1991), 57–76.
- BULAG, Borjigin (1981), 'Düngsiyang kele kiged Monggol bicig-ün kelen-ü jarim abiyanu tokiralcag-a' [Some sound correspondences between Santa and Written Mongolian], ÖMYSS 1981/2: 157–64.
 - —(1982), 'Monggol töröl kele-nügüd-ün urtu qoos egesig-ün tuqai' [Long vowels and diphthongs in Mongolian languages], *ÖMYSS* 1982/3: 77–95.
- ——(1983), 'Sugurag-a ügei y ~ g geyigülügci-yin tuqai' [Non-disappearing y ~ g], MKUJ 1983/1: 4–19.
- (1983–4), 'Monggol kelen-ü urtu egesig-ün egüsül kögjil-ün tuqai' [The origin and development of long vowels in Mongolian], *MKB* 1983/6: 41–52; 1984/1: 17–33.
- [——] Bāo Lìgāo (1985), 'Měnggǔ shūmiànyǔ yuányīn jiān wèi tuōluò de fǔyīn y ~ g' [Nondisappearing intervocalic y ~ g in written Mongolian], MY 1985/1: 16–28.
 - (1988), 'Düngsiyang kelen-ü balarqai egesig u-yin tuqai' [The reduced vowel u in Santa], MKUJ 1988/3: 106–12.
 - (1990), 'Monggol töröl kele-nügüd-ün üge-yin ekin-ü *h* geyigülügci-yin tuqai' [The word-initial consonant *h* in Mongolian languages], *MKUJ* 1990/5: 28–44.

- BULAG, Borjigin (1993*a*), 'Dumdadu jagun-u Monggol kelen-ü üge-yin dumdaki geyigülügci *h*-un tuqai' [The word-internal consonant *h* in Middle Mongolian], *MKUJ* 1993/6: 4– 19, 34.
 - —(1993*b*), 'Monggol bicig-ün šürgükü *h* bolon teg geyigülügci üsüg' [The fricative *h* and the zero consonant in written Mongolian], *MKB* 1993/11: 8–15.
- [——] Bāo Lìgāo (1996), 'Měnggůwén zìmů G de gů dài dúyīn jí qí yănbiàn' [The old reading of the Mongolian letter G and its development], MY 1996/2: 57–63.
- [-----] Bāo Lìgāo (1997), 'Dōngxiāngyǔ tóng *Měnggǔ mìshǐ* de yǔyīn duìyìng' [Phonetic comparison between Santa and the language of the *Secret history of the Mongols*], in Jalsan and Badmaodser (1997), 283–98.
- BULU (1980), 'Urad aman ayalgun-u jarim oncalig-un tuqai tursin ögülekü ni' [A note on some characteristics of the Urad dialect], *ÖMBDS* 1980/2: 85–97.
- ——(1984), 'Naiman qosigun-u nutug-un ayalgu-yin abiyan-u qubiral-un tuqai angqan-u ajiglaltu' [Preliminary notes on sound changes in the dialect of Naiman banner], ÖMBYS 1984/1: 1–11, 23.
- BULUCILAGU (1960), 'Monggol kelen-ü egesig qamarsiqu jici geyigülügci n, ng-ün tuqai' [Vowel nasalization and the consonants n and ng in Mongolian], MKJT 1960/4: 41–5.
- ——(1982), 'Sir-a Yogor kelen-ü egesig jokical ba urugul-un egesig-ün tuqai' [Vowel harmony and rounded vowels in Shira Yugur], ÖMYSS 1982/1: 1–12.
- (1985*a*), *Dong bù Yùgùyŭ cíhuì* [Shira Yugur vocabulary] (*MTKASC* 017).
- ——(1985b), 'Jegün Yogor kelen-deki üge-yin nigedüger üy-e-yin *i egesig-ün ularil' [The development of the vowel *i in the first syllable of Shira Yugur words], ÖMYSS 1985/4: 1–13.
- [——] Bolbasural (1985*c*), 'Monggol kelen-ü türilte-tei *n* geyigülügci-yi ögülekü ni' [A note on the aspirated consonant *n* in Mongolian], *MKB* 1985/1: 62–8.
- ——(1986), 'Jegün Yogor kele ba Monggol kelen-ü urtu egesig-ün tokiralcag-a' [Correspondence between long vowels in Shira Yugur and Mongolian], ÖMYSS 1986/4: 1–14.
- ——(1989), 'Jegün Yogor kelen-ü üge-yin üy-e, üge-yin ergülte' [Syllables and word stress in Shira Yugur], ÖMYSS 1989/1: 86–94.
- ——(1991), 'Jegün Yogor kelen-ü geyigülügci abiyalaburi ba Monggol kele-lüge-yin tokiralcag-a (tobci)' [Correspondences between consonant phonemes in Shira Yugur and Mongolian (abstract)], *MKB* 1991/5: 39–49.
- (1992), 'Jegün Yogor kelen-ü bogoni egesig abiyalaburi ba Monggol kele-lüge-yin tokiralcag-a (tobci)' [Correspondences between short vowel phonemes in Shira Yugur and Mongolian (abstract)], *MKB* 1992/1: 43–50.
- [——] Bǎo Cháolǔ (1998), 'Měnggǔyǔ de címò shéjiān bí fǔyīn' [The final apical nasal consonant in Mongolian], *NDX* 1998/3: 33–40.
 - and Jalsan (1992), *Dong bù Yùgùyǔ hé Měnggǔyǔ* [Shira Yugur and Mongolian] (*MTKASC* 016).
- BURAEV, Ignatij Dmitrievič (1959), Zvukovoj sostav burjatskogo jazyka [The sound structure of Buriad] (Ulan-Udè: Burjatskij kompleksnyj naučno-issledovatel'skij institut).
- (1960), 'Singarmonizm v burjatskom jazyke' [Vowel harmony in Buriad], *TBurNII* 3: 89–94.
 - —(1962), 'Fonetika' [Phonetics], in G. D. Sanžeev (ed.), *Grammatika burjatskogo jazyka: fonetika i morfologija*, 7–48 (Moskva: Izdatel'stvo vostočnoj literatury).
 - (1965), 'Sartul'skij govor' [The Sartuul dialect], TBurNII 17: 108–50.
- -----(1968), 'Nekotorye fonetičeskie osobennosti govora alaro-unginskih burjat' [Some phonetic characteristics of the dialect of the Alair-Ungi Buriads], *TBION* 2: 117–35.
 - -----(1976), 'Dlitel'nost' glasnyh i soglasnyh zvukov bajkalo-kudarinskogo govora burjat-

skogo jazyka' [The duration of vowels and consonants in the Baigal–Hudari dialect of Buriad], *TBION* 29: 34–42.

-(1977), 'Spirantizacija soglasnyh v mongol'skih jazykah' [Fricativization of consonants in Mongolian languages], *OUMÈ 3 IH* 2: 38–43.

—(1983), 'Stanovlenie zvukovogo stroja sovremennogo burjatskogo jazyka' [The development of the sound structure of modern Buriad], in Nadeljaev (1983), 162–4.

—(1985a), 'Fonologičeskij aspekt mongol'skogo singarmonizma' [The phonological aspect of Mongolian vowel harmony], OUMÈ 4 IH 2: 126–34.

- (1985*b*), 'Pozicionnoe upotreblenie soglasnyh v burjatskom jazyke i osobennosti ih sočetaemosti meždu soboj' [Positional use of consonants in Buriad and characteristics of their phonotactics], in Sanžeev (1985), 9–20.
- —(1986), 'Singarmonizm: priznak artikuljacionnoj bazy jazyka' [Vowel harmony: a characteristic of the base of articulation of the language], in Rassadin (1986), 5–23.
- ——(1987*a*), *Stanovlenie zvukovogo stroja burjatskogo jazyka* [The development of the sound structure of Buriad] (Novosibirsk: Nauka).
 - —(1987*b*), 'Stanovlenie sistemy glasnyh fonem sovremennogo burjatskogo jazyka' [The development of the system of vowel phonemes in modern Buriad], in Rassadin (1987*a*), 3–14.
- (1987*c*), 'Udarenie i singarmonizm v mongol'skih jazykah' [Stress and vowel harmony in Mongolian languages], in Solncev (1987), 2: 20–8.

— (1988*a*), 'Nekotorye èlementy fonetičeskogo substrata v burjatskom jazyke' [Some elements of the phonetic substrate in Buriad], in Šagdarov (1988), 41–53.

- ——(1988*b*), 'Rezul'taty kontaktirovanija jazykov altajskoj obŝnosti v cirkumbajkal'skom regione' [Results of contacts between languages of the Altaic community in the Circum-Baikal region], *VJa* 1988/4: 123–8.
- (1989), 'Stanovlenie sistemy soglasnyh fonem sovremennogo burjatskogo jazyka' [The development of the consonant phoneme system in modern Buriad], in Šagdarov (1989), 3–12.
- ——(1991), 'Intonacija kak sredstvo aktualizacii reči' [Intonation as a means of actualization of speech], in Šagdarov (1991), 112–20.
- -----(1993), 'O stanovlenii burjatskogo jazyka' [The development of the Buriad language], in Rassadin (1993), 3–10.
- —(1995), 'Èvoljucija burjatskogo konsonantizma' [The development of the Buriad consonantism], in V. I. Rassadin (ed.), *Istoriko-sravnitel'noe izučenie mongol'skih jazykov*, 3–14 (Ulan-Udè: BNC).
- ——(1996), 'Problemy klassifikacii burjatskih dialektov' [Problems in the classification of Buriad dialects], in Rassadin (1996b), 3–16.
- Taťjana Petrovna Bažeeva, and Elena Semenovna Pavlova (1975), Atlas zvukov burjatskogo jazyka [Buriad sound atlas] (Ulan-Udè: BKI).
- [CAGANQADA] Chágān (1979), 'Kē'ěrqìn tǔyǔ yuányīn héxiélǜ de tèdiǎn' [Characteristics of vowel harmony in the Horchin dialect], *MY* 1979/4: 281.
- [-----] Chágānhādá (1981), 'Zài lùn Kē' ěrqìn tǔyǔ de yuányīn héxié' [A further discussion of vowel harmony in the Horchin dialect], *MY* 1981/4: 62–6.
- [-----] Chágānhādá (1985*a*), 'Měnggǔyǔ Qīnghǎi tǔyǔ de yǔyīn tèdiǎn' [Phonetic characteristics of the Qinghai dialect of Mongolian], *MY* 1985/4: 30–6.
- [——] Chágānhādá (1985*b*), 'Měnggǔyǔ Kē' ěrqìn tǔyǔ de yǔyīn xìtǒng' [The sound system of the Horchin dialect of Mongolian], *Yǔyán yánjiū*, 9: 180–95.
- [——] Chágānhādá (1996), *Měnggǔyǔ Kē 'ěrqìn tǔyǔ yánjiū* [Studies on the Horchin dialect of Mongolian] (Běijīng: Shèhuì kēxué wénxiàn).

- ČAGDUROV, Sergej Šagžievič (1951), 'Zametki o sartul'skom govore' [Remarks on the Sartuul dialect], *ZBurNII* 11: 154–65.
- CASTRÉN, M. Alexander (1857), Versuch einer burjätischen Sprachlehre nebst kurzem Wörterverzeichniss (Sankt-Peterburg: Kaiserliche Akademie der Wissenschaften).
- [CENGGELTEI] Cinggeltei (1957–8), 'Dumdadu Ulus-taki Monggol töröl-ün kele-nügüd ba Monggol kelen-ü ayalgu-nugud-un yerüngkei bayidal' [The general situation of Mongolian languages and dialects in China], *MKB* 1957/11: 44–57; 1957/12: 25–48; *MTKB* 1958/1: 44–8; 1958/2: 41–57; 1958/3: 32–9; 1958/4: 29–49; 1958/6: 35–9; 1958/7: 26–32; 1958/12: 19–57.
- [-----] Cinggeltei (1959), 'Monggol kelen-ü Bagarin ayalgun-u abiy-a-yin jüi ba üges-ün jüi' [Phonology and morphology of the Baarin dialect of Mongolian], *ÖMYSS* 1959/2: 1–96.
- [-----] Chingeltei (1961), 'Notes on the Barin phonology', AOH 13: 295–303.
- [----] Cinggeltei (1963), 'Monggol kelen-ü abiyan-u sistem' [The Mongolian sound system], *ÖMYSS* 1963/2: 1–84.
 - ——(1978), 'Dumdadu Ulus-un Monggol kelen-ü ayalgu-nugud-i qubiyarilaqu tuqai sanal' [Some thoughts on the dialect division of Mongolian in China], *ÖMYSS* 1978/3: 1–32.
- ——(1979*a*), *Odo üy-e-yin Monggol kelen-ü jüi* [Grammar of modern Mongolian] (Kökeqota: Öbör Monggol-un arad-un keblel-ün qoriy-a).
- [——] Qīnggé'ěrtài (1979b), 'Zhōngguó Měnggǔyǔ fāngyán de huàfēn' [Dialect division of Mongolian in China], MY 1979/1: 13–20; 1979/2: 108–12.
- ——(1982), 'Egesig jokicaqu yoson-u tuqai' [On vowel harmony], ÖMYSS 1982/1: 87–110.
 Chinese version 1983: Qīnggé' ertài, 'Guānyú yuányīn héxiélù', Zhōngguó yǔyánxué bào, 1: 200–20.
- [——] Qīnggé'ěrtài (1985*a*), 'Měnggǔyǔ sèyīn *q*, *k* de lìshǐ yǎnbiàn' [The historical development of the stops *q*, *k* in Mongolian], *MY* 1985/3: 1–10.
- [——] Qīnggé'ěrtài (1985*b*), 'Guānyú Měnggŭ yǔzú yǔyán jí qí yánjiū' [Languages of the Mongolian group and their investigation], *NDX* 1985/4: 18–28.
 - ——(1988), 'Tǔzúyǔ zhōng b fǔyīn de yǎnbiàn' [The development of the consonant b in Monguor], XMY 1988/2: 188–93.
- ——(1989a), 'Monggol töröl keled-ün qaricagan-u tuqai' [Interrelations between the Mongolian languages], ÖMYSS 1989/4: 6–19.
- [____] Qīnggé'ěrtài (1989b), 'Měnggǔ yǔzú yǔyán zhōng de yīnshì jiégòu' [The structure of phonological intensity in the Mongolian languages], MY 1989/1: 28–36.
 - (1991), Tůzúyů hé Měnggůyů [Monguor and Mongolian] (MTKASC 013).
- ——(1997), 'Monggor ündüsüten kiged Monggor kele' [The Monguor people and language], in *Kele bicig-ün tuqai ögülel-ün tegübüri*, 630–45 (Kökeqota: Öbör Monggolun keblel-ün qoriy-a). [Originally published in 1991]
- [-----] Cinggeltei and Coyijongjab (1959), 'Monggol kelen-ü geyigülügci-yin tuqai' [Mongolian consonants], *ÖMYSS* 1959/1: 42–60.
- [____] Cinggeltei and Sinedke (1959), 'Monggol kelen-ü ündüsün egesig-üd-ün tuqai' [The basic vowels of Mongolian], *ÖMYSS* 1959/2: 97–114.
- CENGGELTÜ, D. (1999), 'Qorcin Jegün Gar-un Qoyitu qosigun-u yariyan-u geyigülügci abiyan-u jarimdug selgücel' [Some consonant alternations in the dialect of Horchin Left Rear banner], MKB 1992/2: 48-50.
- ČEREMISOV, Konstantin Mihajlovič (1941), 'Zametki po burjat-mongol'skomu jazyku, III: neskol'ko zamečanij o tunkinskom dialekte' [Notes on Buriad Mongolian, III: some remarks on the Tünhen dialect], *Zapiski BM GIJaLI*, 3–4: 121–40.
 - -----(1947), 'Zametki po burjat-mongol'skomu jazyku, IV: o kabanskom podgovore'

[Notes on Buriad Mongolian, IV: the Kabansk subdialect], Zapiski Burjat-Mongol'skogo naučno-issledovatel'skogo instituta kul'tury i ekonomiki, 7: 125–36.

- —(1951), *Burjat-mongol'sko-russkij slovar'; Burjat-mongol-Orod slovar'* [Buriad Mongolian-Russian dictionary] (Moskva: Gosudarstvennoe izdatel'stvo inostrannyh i nacional'nyh slovarej).
- CÈRÈNSODNOM, Dalantajn and Manfred Taube (1993), Die Mongolica der Berliner Turfansammlung (Berliner Turfantexte, 16) (Berlin: Akademie Verlag).
- CHARETTE, Monik (1991–2), 'Mongolian and Polish meet government licensing', SOAS Working papers in linguistics and phonetics, 2: 275–91.
- CHÉN Năixióng (1986a), Bǎo'ānyǔ cíhuì [Bonan vocabulary] (MTKASC 011).
- ——(1986b), 'Guānyú Wǔtúnhuà' [On Wutun], Ajia, Afurika gengo bunka kenkyû (Journal of Asian and African studies), 31: 33–52.
 - ----(1987), Bǎo 'ānyǔ hé Měnggǔyǔ [Bonan and Mongolian] (MTKASC 010).
- ——(1989–90), Bǎo'ānyǔde yǔyīn hé cíhuì' [The phonetics and vocabulary of Bonan], *XMY* 1989/2: 16–40; 1990/1: 33–48.
 - —(1993), 'Xīng'ān méng Měnggůhuà tèdiǎn shílíng' [Notes on the characteristics of Mongolian in Hinggan league], in Dài Qìngxià, Luó Měizhēn, and Yáng Yìngxīn (eds.), *Mínzú yǔwén lùnwén jí: qìngzhù Mă Xuéliáng xiānshēng bāshí shòuchén wén jí*, 275–88 (Běijīng: Zhōngyāng mínzú xuéyuàn).
- ——(1994a), 'Yúnnán de Qìdān zú hòuyì hé Qìdān zì yícún' [Descendants of the Kitan people in Yunnan and remainders of the Kitan script], MY 1994/6: 20–6.
- —(1994b), 'Băo'ānyǔ jí qí fāngyán tǔyǔ' [The Bonan language and its dialects], in E. H. Kaplan and D. W. Whisenhunt (eds.), *Opuscula Altaica: essays presented in honor of Henry Schwarz*, 265–85 (Bellingham: Western Washington University).
- (1995a), 'Běnhuà zhōng de Ā'ěrtài yǔyán chéngfèn yícún' [Remainders of the Altaic component in Shidian U], Zhōngguó yǔyánxué bào, 6: 297–316.
- —(1995b), 'Xīng'ān méng Měnggǔ huà diàochá jì' [Survey of Mongolian dialects in Hinggan league], in *Chén Năixióng lùnwén jí*, 651–81 (Kökeqota: Nèiměnggǔ jiàoyù).
- ——(1999), 'Zhāo Wūdá Měnggŭhuà kăochá' [An investigation of Mongolian in Juu Ud], NDX 1999/6: 75-80.
- CHINCHOR, Nancy (1979), 'On the treatment of Mongolian vowel harmony', *CUNYForum:* papers in linguistics, 5–6: 171–86.
- Ch'o Mi-hŭi (1994), *Vowel harmony in Korean: a grounded phonology approach*. Ph.D. dissertation, Indiana University.
- CHULUU, Üjiyediin (Chaolu Wu) (1994a), Introduction, grammar, and sample sentences for Jegün Yogur (Sino-Platonic papers, 54) (Philadelphia: University of Pennsylvania).
- ——(1994b), Introduction, grammar, and sample sentences for Dongxiang (Sino-Platonic papers, 55) (Philadelphia: University of Pennsylvania).
- ——(1994c), Introduction, grammar, and sample sentences for Dagur (Sino-Platonic papers, 56) (Philadelphia: University of Pennsylvania).
- ——(1994d), Introduction, grammar, and sample sentences for Monguor (Sino-Platonic papers, 57) (Philadelphia: University of Pennsylvania).
- (1994*e*), Introduction, grammar, and sample sentences for Baoan (Sino-Platonic papers, 58) (Philadelphia: University of Pennsylvania).
 - (1996), Daur (Languages of the world/Materials, 93) (München: Lincom Europa).
- CLARK, Larry V. (1979), 'A problem in Buryat historical linguistics', in H. G. Schwarz (ed.), Studies on Mongolia: proceedings of the First North American conference on Mongolian studies, 14–25 (Bellingham: Western Washington University).
- CLAUSON, Gerard (1956), 'The case against the Altaic theory', CAJ 2: 181-7.

- CLAUSON, Gerard (1962), 'The phonetic structure of pre-thirteenth century Mongolian', ch. 10 (pp. 188–210) in G. Clauson, *Turkish and Mongolian studies* (London: Royal Asiatic Society of Great Britain and Ireland).
- CLEMENTS, George N. (1990), 'The role of the sonority cycle in core syllabification', in J. Kingston and M. Beckman (eds.), *Papers in laboratory phonology 1: between the grammar and physics of speech*, 283–333 (Cambridge: Cambridge University Press).
- and Samuel Jay Keyser (1983), CV phonology: a generative theory of the syllable (Cambridge: MIT Press).
- COGBAYAR (1989), 'Sönid aman ayalgun-u abiyan-u sistèm' [The sound system of the Sönid dialect], *MKUJ* 1989/4: 64–75.
- [COGTU] Süngrüb (1985), 'Alaša aman ayalgun-u abiyalaburi bolon barimjiy-a abiyan-u abiyalaburi-yin qaricagulul' [Comparison of the phonemes of the Alshaa dialect and the standard pronunciation], *MKB* 1985/2: 46–57.
- COHEN, David A. (1981), 'Mongolian vowel harmony', Texas linguistic forum, 17: 29-43.
- COLE, Jennifer (1995), 'The cycle in phonology', in Goldsmith (1995), 70-113.
- COLOO, Žavzangijn (1962), 'Arhangaj ajmgijn baruun hojt nutgijn halhyn aman ajalguuny tuhaj' [The Halh dialect of northwestern Arhangai], *ŠUAM* 1962/1: 44–8.
- ——(1964), 'Zahčin, torguud ajalguuny onclogoos' [Some characteristics of the Zahchin and Torguud dialects], HZS 2/10–19: fasc. 14 (pp. 93–117).
- -----(1965), Zahčiny aman ajalguu [The Zahchin dialect] (HZS 3: fasc. 1).
- (1967*a*), 'Govijn halhyn aman ajalguuny ügijn bürèldèhüünij zarim onclog' [Some characteristics of the word structure of the Gobi Halh dialect], *StMo* 6/15–21: fasc. 17 (pp. 23–37).
- ——(1967b), Mongol hèlnij aviany ögüülègdèh ës [Pronunciation rules for the sounds of Mongolian] (Ulaanbaatar: BNMAU gègèèrlijn jaam).
- ——(1967c), 'Halhyn zarim aman ajalguuny aviazüjn onclogoos' [Some phonetic characteristics of some Halh dialects], ŠUAM 1967/1: 61–70.
- (1969), 'Arhangaj ajmgijn hojt sumdyn halhyn aman ajalguu, aman zohiol sudalsan n''
 [Studies on the Halh dialect and folklore of the northern districts of Arhangai], *StMo* 7: fasc. 17 [18] (219–42).
- ——(1970), 'Baruun halhyn aman ajalguu' [The western Halh dialect], in Š. Barajšir (ed.), Halhyn ajalguu, 133–224 (Ulaanbaatar: ŠUA).
 - —(1973*a*), 'Urianhaj ajalguug sudalsan zarim dün' [Some results from a study on the Urianhai dialect], *HZS* 9: fasc. 7 (pp. 57–63).
- (1973*b*), 'Ob udarenii v halhaskom narečii' [Stress in the Halh dialect], *OUMÈ 2 IH* 2: 251–3.
- ——(1976), *Orčin cagijn mongol hèlnij avia züj* [Phonetics of modern Mongolian] (Ulaanbaatar: ŠUA).
- ——(1985), 'K voprosu klassifikacii dialektov v MNR' [The problem of the classification of dialects in MPR], in Sanžeev (1985), 124–8.
- (1987), 'BNMAU-yn nutgijn ajalguu' [Dialects of the MPR], *HZS* 18: fasc. 1 (pp. 3–142).
- COMRIE, Bernard (1981), *The languages of the Soviet Union* (Cambridge: Cambridge University Press).
- COYIJONGJAB (1957), 'Monggol-un udq-a jokiyal-un kelen-ü fonèm-üd ba tedger-i kerkin temdeglejü bayig-a tuqai' [The phonemes of the Mongolian literary language and how they are denoted], *MKB* 1957/8: 52–66.
 - ——(1959), 'Oyirad-un fonèm-un sistèm-ün tuqai' [The Oirad phoneme system], *MKJT* 1959/3: 15–26.

—(1978), 'Monggol kelen-ü saguri ayalgu, barimjiy-a abiyan-u tuqai kedün asagudal' [Some questions about the basic dialect of Mongolian and the standard pronunciation], *ÖMYSS* 1978/3: 55–84.

—(1982*a*), 'Qorcin aman ayalgun-u egesig abiyalaburi-yin jarim oncalig' [Some characteristics of the vowel phonemes of the Horchin dialect], *ÖMYSS* 1982/1: 71–86.

——(1982b), 'Köke Muncug kelen-ü abiyan-u togtaca kiged Monggol kelen-ü abiy-a' [The sound system of Kök Mungchaq and the sounds of Mongolian], ÖMYSS 1982/4: 89–119.

- [—] Quèjīngzhābù (1985), 'Cóng Kùkù Ménqiàkèyǔ yǔ Měnggǔyǔ yuányīn bǐjiào kàn Měnggǔyǔ yǔyīn shǐ de yìxiē wèntí' [Some problems of Mongolian historical phonology as seen from the comparison of vowels betweeen Kök Mungchaq and Mongolian], MY 1985/2: 1–10.
- —(1989a), 'Caqar aman ayalgun-u qoos egesig-tei qolbogdaqu kedün jüil-ün tuqai: tursilta-yin abiyan-u jüi-yin arg-a-bar sudulugsan angqan-u temdeglel' [Some problems concerning the diphthongs in the Chahar dialect: preliminary results of research by methods of experimental phonetics], ÖMYSS 1989/4: 20–30, 131–2.
- [——] Quèjīngzhābù (1989b), 'Měnggůyů Cháhā'ěr tůyů yuányīn de shíyàn yůyīnxué yánjiū' [An experimental phonetic study on vowels of the Chahar dialect of Mongolian], MY 1989/4: 1–5.
- ——(1993a), 'Monggol kelen-ü ergülte-yin tuqai: abiyan-u tursilta-yin jabsar-un medegülülte' [Stress in Mongolian: an interim report on phonetic experiments], ÖMYSS 1993/1: 1–14.
- (1993b), 'Monggol kelen-ü barimjiy-a abiy-a-tai qolbogdaqu kedün asagudal-un tuqai'
 [Some problems concerning the standard Mongolian pronunciation], ÖMYSS 1993/4: 35–48.
- [—] Quèjīngzhābù and Jôo Hakutarô (1993), 'Uchi-Môko Chaharu hôgen no boin ni kansuru onkyô onseigakuteki kenkyû' [An acoustic phonetic investigation of the vowels of the Chahar dialect of Inner Mongolia], *OShKR*, 29–49.
- CÜNGKERJA (1987), 'Degedü Monggol-un aman ayalgun-u egesig abiyalaburi kiged barimjiy-a abiyan-u egesig abiyalaburi-yin qaricagulul' [Comparison of the vowel phonemes in the Upper Mongol dialect and the standard pronunciation], *ÖT* 1: 292–303.
- Сувікоv, Gombožab Cèbekovič (1908), *Posobie dlja praktičeskago izučenija mongol'skago jazyka* [Textbook for the practical study of Mongolian] (Vladivostok).
- ——(1915), *Posobie k praktičeskomu izučeniju mongol'skago jazyka* [Textbook for the practical study of Mongolian] (*Izvěstija Vostočnago Instituta*, 56/3) (Vladivostok: Izdanie Vostočnago Instituta).
- CYDENDAMBAEV, Cybikžap Boboevič (1960), 'O dialektal'nyh različijah v razgovornom burjatskom jazyke' [Dialect differences in spoken Buriad], *TBurNII* 3: 111–17.
 - ——(1964), 'Kratkaja harakteristika govora kudarinskih burjat' [Short characterization of the dialect of the Hudari Buriads], KSINA 83: 57–68.
 - ——(1968), 'K itogam èkspedicionnogo izučenija govorov dobajkal'skih burjat' [Results of expedition research of the dialects of Cis-Baikal Buriads], *TBION* 2: 164–75.
- (ed.) (1977), *Issledovanie burjatskih i russkih govorov* [Investigation of Buriad and Russian dialects] (Ulan-Udè: BION).
- CYDENŽAPOV, Širab-Nimbu Rinčinovič (1973), 'Ob odnoj osobennosti konečnyh soglasnyh osnovy slov staromongol'skogo jazyka' [A characteristic of word stem final consonants in Old Mongolian], *OUMÈ 2 IH 2*: 280–2.
- CYRENDORŽIEVA, Cyrendulma Šogdupovna (1977), 'Iz nabljudenij nad reč'ju burjat goroda Ulan-Udè' [Observations on the speech of Ulan-Ude Buriads], in Cydendambaev (1977), 47–59.

- CYRENOV, Babasan Doržievič (1996), 'Ivalgyn burjaad hèlènèj fonetikyn onso ilgaa' [Phonetic characteristics of the Ivalga dialect of Buriad], in Rassadin (1996b), 165–77.
- ——(1998), 'Baruun burjaadaj ontohonuudaj hèlènèj onso ilgaanud tuhaj' [Characteristics of the language of western Buriad folktales], in Bazarov and Tulohonov (1998), 314–17.
- DAMBUEVA, Polina Petrovna (1998), 'Intonacija kak sredstvo vyraženija kategorii modal'nosti v burjatskom i russkom jazykah' [Intonation as a means of expressing the modality category in Buriad and Russian], in Bazarov and Tulohonov (1998), 241–6.
- DAMDINOV, Dašinima Galdanovič (1962), 'Predvaritel'nye dannye o jazyke hamniganov čitinskoj oblasti' [Preliminary facts on the language of the Kamnigan of Chita district], *KSBurNII* 4: 128–37.
 - —(1968), 'Ètno-lingvističeskij očerk hamniganskogo govora' [Ethno-linguistic sketch of the Kamnigan dialect], TBION 2: 74–116.
- ——(1975), 'Zapadno-mongol'skie čerty v govore ononskih hamnigan' [West Mongolian features in the dialect of the Onon Kamnigan], in Bitkeev (1975c), 151–61.
- (1977), 'O jazyke zakamenskih (armakskih) hamnigan' [On the language of the Zakamna (Armak) Kamnigan], in Cydendambaev (1977), 40–6.
- ——(1986), 'Ob odnoj osobennosti soglasnogo h v burjatskom jazyke' [One characteristic of the consonant h in Buriad], in Rassadin (1986), 107–16.
- ——(1988), 'Ob arhaičeskih osobennostjah govora ononskih hamnigan' [On archaic characteristics of the dialect of the Onon Kamnigan], in L. D. Šagdarov (ed.), *Razvitie i vzaimodejstvie dialektov Pribajkal'ja*, 69–79 (Ulan-Udè: BF).
- DAMDINSÜRÈN, Cèndijn (1946), *Mongol šinè üsgijn tovč dürèm* [Concise rules for the new Mongolian script] (Ulaanbaatar).
- ——(1958), 'Šinè üsgijn dürèm' [The new orthography], in A. Luvsandèndèv (ed.), Mongol hèl bičgijn zarim asuudal, 15–43 (Ulaanbaatar: Šinžlèh uhaan, dèèd bolovsrolyn hürèèlèn).

— and B. Osor (1983), *Mongol üsgijn dürmijn tol'* [Mongolian orthographic dictionary] (Ulaanbaatar: BNMAU Ardyn bolovsrolyn jaam).

- DANZAN (1978), 'Caqar aman ayalgun-u abiyan-u qubiral-un jarim oncalig' [Some characteristics of the sound changes in the Chahar dialect], ÖMBDS 1978/1: 165–9.
- DARBEEVA, Anna Angadykovna (1960), 'Predvaritel'nye dannye o jazyke nižneudinskih burjat' [Preliminary information about the language of Nizhneudinsk Buriads], TBur-NII 3: 118–26.
- (1978), Vlijanie dvujazyčija na razvitie izolirovannogo dialekta: na materiale mongol'skih jazykov [Influence of bilingualism on the development of an isolated dialect: based on data from Mongolian languages] (Moskva: Nauka).
- ——(1988), 'Vlijanie kontaktov na razvitie mongol'skih jazykov' [Contact influences on the development of Mongolian languages], in Šagdarov (1988), 53–61.
- (1992), 'Interferentnye javlenija i ih mesto v zvukovoj sisteme pis'mennyh i bespis'mennyh mongol'skih jazykov' [Interference phenomena and their place in the sound system of written and unwritten Mongolian languages], in È. R. Tenišev (ed.), *Mongol'skij lingvističeskij sbornik*, 3–13 (Moskva: Institut jazykoznanija RAN).
- —(1996), Istoriko-sopostavitel'nye issledovanija po grammatike mongol'skih jazykov: fonetika [Historical-comparative investigations on the grammar of Mongolian languages: phonetics] (Moskva: Nauka).
 - -(1997), 'Burjatskij jazyk' [Buriad], in Jarceva (1997), 37-51.
- DAŠCEDEN, T. (2000), 'Mongolyn nuuc tovčoony helnij egšig aviag sudlah asuudald' [The problem of research on the vowels in the language of the Secret history of the Mongols], in Š. Bira (ed.), Mongolyn nuuc tovčoo-ny 750 žilijn ojd zoriulsan olon ulsyn baga hural,

29-32 (Ulaanbaatar: IAMS).

DAWA, Idomjo-a, Ôkawa Shigeki, and Shirai Katsuhiko (1996), 'Môkogo hôgen 7 boin no forumanto shûhasû bunseki' [Formant frequency analysis of the seven vowels in Mongolian dialects], NOS 1996: 325–6 (1–P–22).

— — — (1999a), 'Měnggůyyů qīge yuányīn shēngpín tèxìng jìsuànjī fēnxī' [Computer analysis of the frequency characteristics of the seven vowels in Mongolian], Shēngxué xuébào (Acta acustica), 24/1: 94–7.

— (1999b), 'An introduction to Mongolian phonetic-linguistic model: on the acoustic properties of Mongolian language', *Proceedings of the XIVth International congress of phonetic sciences (ICPhS 99)*, 1657–60.

————(1999*c*), 'Design of Mongolian speech database considering dialectal characteristics', *Journal of the Acoustical society of Japan (E)*, 20/3: 181–8.

— (1999d), 'Design and assessment of Mongolian dialectal speech input system', in Oriental COCOSDA '99: second international workshop on East-Asian language resources and evaluation, 45–8 (Taibei).

— (2001*a*), 'Měnggůyů zhůyào fāngyán de shēngxué hé yīnlǜ tèzhēng fēnxī fēnlèi' [Classification of major Mongolian dialects by analysing acoustic and prosodic features], *MY* 2001/1: 26–32.

— and Shirai Katsuhiko (2003), 'A real time language and dialect identification based on the VQ error of the acoustic feature and prosodic cues', in Solé, Recasens, and Romero (2003), 1377–80.

— and Ôkawa Shigeki (2000), 'Jôhô riron ni motozuku Mongorugo kôgo kara bungo e no henkan shori shuhô' [An information theoretical method for converting spoken to written Mongolian], *NOS* 2000: 211–12 (2–1–19).

- Arai Daisuke, Ôkawa Shigeki, and Shirai Katsuhiko (1997), 'Mongorugo on'in no bunrui oyobi ninshiki seinô no hyôka' [Classification of Mongolian phonemes and evaluation of the recognition performance], *NOS* 1997: 129–30 (2–Q–7).
- Ôkawa Shigeki, Nakajima Takehiro, and Shirai Katsuhiko (1998), 'Hôgen kan no sai o kôryoshita Mongorugo onsei dêtabêsu no kôchiku' [Design of a Mongolian speech database considering dialect differences], NOS, 1998: 173–4 (1–R–28).
- —Nakajima Takehiro, Shirai Katsuhiko, and Ôkawa Shigeki (1999), 'Jôhôryô o kijun toshita Mongorugo shuyô ni dai hôgen no onkyô tokuchô no bunseki' [Acoustic feature analysis of the two major dialects of Mongolian based on the information measure], *NOS* 1999: 119–20 (3–Q–1).

—Ôkawa Shigeki, Shirai Katsuhiko, and Kobayashi Tetsunori (2000), 'Building and application of Mongolian multi-dialectal speech database', in *COCOSDA: notes from the Cocosda Workshop 2000*, 123–6 (Běijīng).

- DélìGé'ěrмă (1987), 'Měnggůyů' [Mongolian], ZSMY 576-91.
- DEMIRDACHE, Hamida (1988), 'Transparent vowels', in H. van der Hulst and N. Smith (eds.), *Features, segmental structure and harmony processes*, 2: 39–76 (Dordrecht: Foris).
- DENWOOD, Ann (1995), 'Could Khalkha Mongolian be a head-licensing language?: a comparison between Khalkha Mongolian and Vata, a typical ATR-harmony language', *SOAS working papers in linguistics and phonetics*, 5: 41–63.
 - (1996), 'Vowel harmony in Khalkha Mongolian', AH 6: 47-82.
 - (1997a), *The role of the element* I *in Khalkha Mongolian phonology*. Ph.D. dissertation, School of Oriental and African Studies, University of London.

- DENWOOD, Ann (1997b), 'Harmony: three for the price of one', SOAS working papers in linguistics and phonetics, 7: 193–207.
 - (1998), 'Onset to onset government in Khalkha Mongolian', AH 8: 143–54.
- DING Shíqing (1992), 'Xīnjiāng Dáwò'ěryǔ xiǎoshéyīn qiǎnxī' [A preliminary analysis of the Xinjiang Dagur uvulars], MY 1992/5: 53-7.
- ——(1995), 'Xīnjiāng Dáwò'ěryǔ jiǎnshù' [An outline of Xinjiang Dagur], Yúyán yánjiū, 28: 188–95.
- DIXON, Robert M. W. (1997), *The rise and fall of languages* (Cambridge: Cambridge University Press).
- [Dobo] Dàobù (1964), 'Měnggǔyǔ gàikuàng' [Sketch of Mongolian], ZY 1964/3: 240-53.
- [——] Dàobù (1981), 'Měnggǔyǔ kǒuyǔ zhōng de císhǒu fǔyīn ruòhuà xiànxiàng' [Weakening of initial consonants in spoken Mongolian], *MY* 1981/1: 43–5.
- [——] Dàobù (1982), 'Měnggǔyǔ Bālín tǔyǔ de fùfǔyīn, guòdùxìng yuányīn hé yīnjié jiégòu' [Consonant clusters, transient vowels and syllable structure in Baarin Mongolian], in Mínzú yǔwén yánjiū wén jí, 366–84 (Xīníng: Qīnghǎi mínzú).
- [——] Dàobù (1983a), Měnggůyů jiǎnzhì [Short description of Mongolian] (Běijīng: Mínzú).
 - ——(1983*b*), *Uyigurjin Monggol üsüg-ün durasqaltu bicig-üd* [Documents in the Uigur Mongolian script] (Běijīng: Mínzú).
- [——] Dàobù (1984), 'Měnggǔyǔ de yuányīn héxié yǔ yuányīn yīnwèi duìlì de zhōnghé' [Vowel harmony and the neutralization of vowel phoneme oppositions in Mongolian], *MY* 1984/2: 10–13.
- [——] Dàobù (1997), 'Qīnghǎi shěng Xiǎo Wàisī bùluò Měnggǔyǔ de jǐge yǔyīn tèdiǎn' [Some phonetic features of the Mongolian language of the Minor Beis tribe in Qinghai province], MY 1997/6: 63–7.
- [____] Dob and [Caganqada] Chaganhada (1987), 'Phonetic features of the Mongolian spoken by the Minor Beis tribe in Qinghai province', in D. C. Laycock and W. Winter (eds.), *A world of language: papers presented to professor S. A. Wurm on his 65th birthday* (*Pacific Linguistics C-100*), 141–7 (Canberra: Australian National University).
- DOERFER, Gerhard (1964*a*), 'Klassifikation und Verbreitung der mongolischen Sprachen', *HOM* 35–50.
- -----(1964b), 'Sprachbau', HOM 51-75.
- (1964c), 'Langvokale im Urmongolischen?', JSFOu 65: fasc. 4 (pp 1–21).
- ——(1965), Ältere westeuropäische Quellen zur kalmückischen Sprachgeschichte: Witsen 1692 bis Zwick 1827 (Asiatische Forschungen, 18) (Wiesbaden: Harrassowitz).
- ——(1966), 'Zur Verwandtschaft der altaischen Sprachen', Indogermanische Forschungen 71: 81–123.
- -----(1968), 'Zwei wichtige Probleme der Altaistik', JSFOu 69: fasc. 4 (pp. 1–21).
- -----(1970), 'Langvokale im Urmongolischen? II', JSFOu 70: fasc.1 (pp. 1–24).
- -----(1974), 'Bemerkungen zum Vokalismus des Monguor: Langvokale im Urmongolischen? III', *JSFOu* 73: 36–94.
- -----(1985a), 'Das Kamniganische', HBK 17: 69-75.
- ------(1985b), Mongolo-Tungusica (Wiesbaden: Harrassowitz).
- ——(1996), 'Primary *h- in Mongol?' CAJ 40: 173-7.
- DONDUKOV, Ulzy-Žargal Šojbonovič (1973), 'Nekotorye jazykovye osobennosti govora Issyk-Kul'skih sart-kalmykov (ojratov) v sravnitel'nom osveŝenii s mongol'skimi i kirgizskimi jazykami' [Some linguistic characteristics of the Ysyk Köl Sart Kalmucks (Oirads) in a comparative perspective with Mongolian and Kirgiz], *OUMÈ 2 IH* 1: 166–72.

——(1975), 'O nekotoryh jazykovyh osobennostjah issyk-kul'skih kalmykov' [On some linguistic characteristics of the Ysyk Köl Kalmucks], in Bitkeev (1975c), 16–33.

- DORJICANG (1987), 'Kökenagur Monggol-un aman ayalgun-u abiyan-u bütüce-yin toyimo' [Sketch of the sound structure of the dialect of the Qinghai Mongols], *ÖT* 1: 303–37.
- DYRHEEVA, Galina Aleksandrovna (1991), 'Statističeskij analiz sočetaemosti soglasnyh v burjatskom jazyke' [Statistical analysis of the phonotactics of consonants in Buriad], in Šagdarov (1991), 77–87.

— Bal'žinima Žambalovič Budaev, and Tat'jana Petrovna Bažeeva (1999), *Burjatskij jazyk: sovremennoe sostojanie (sociolingvističeskij aspekt)* [Buriad: the present state (sociolinguistic aspect)] (Ulan-Udè: BNC).

- and Oktjabrina Doržievna Buhaeva (1978), 'Statistiko-kombinatornaja harakteristika paradigm sočetajuŝihsja soglasnyh v burjatskom jazyke' [Statistic-combinatorial characteristics of paradigms of consonant combinations in Buriad], in I. D. Buraev (ed.), *Issledovanija po burjatskoj filologii*, 86–103 (Ulan-Udè: BF).
- EBLELTÜ, B. (1992), 'Monggol bicig-ün üy-e-yin bütüce ba üy-e-yin büridke' [The structure and inventory of written Mongolian syllables], *MKB* 1992/6: 43–57.
- EDKINS, Joseph (1871), China's place in philology: an attempt to show that the languages of Europe and Asia have a common origin (London: Trübner).
- EDQURIYAGCI (1996), *Monggoljin aman ayalgu* [The Mongoljin dialect] (Kökeqota: ÖMSK).
- and Uran-a (1991), 'Monggoljin aman ayalgun-u abiyan-u oncalig' [Phonetic characteristics of the Mongoljin dialect], *MKB* 1991/4: 51–6.
- EFIMOV, Valentin Aleksandrovič (1965), *Jazyk afganskih Hazara: jakaulangskij dialekt* [The language of the Afghan Hazara: the Yakaulang dialect] (Moskva: Nauka).
- ENGKEBATU, Merten (1982), 'Dagur kelen-ü abiyan-u sistèm' [The sound system of Dagur], ÖMYSS 1982/4: 134–74.
- (1984), Dáwò'ěryǔ cíhuì [Dagur vocabulary] (MTKASC 005).
- (1988), Dáwò'ěryǔ hé Měnggǔyǔ [Dagur and Mongolian] (MTKASC 004).
- [——] Ēnhébātú (1994a), 'Cóng Měng–Dá cídiăn kàn Dáwò'ěryǔ de yůyīn tuōluò xiànxiàng' [Sound deletion in Dagur as judged from A Mongolian–Dagur dictionary], NDX 1994/2: 40–3.
 - (1994*b*), 'Dagur kele-yi Manju üsüg-iyer biciglekü yoson-u tuqai' [On the method of writing Dagur with Manchu letters], *ÖMYSS* 1994/2: 1–25.
- [——] Ēnhébātú (1996), '19 shìjì Dáwò'er rén shǐyòng de wénzì' [The writing system used by the Dagur people during the 19th century], *NDX* 1996/6: 82–7.
- ERDENIBAGATUR (1998), 'Ordos yariyan-u abiyan-u qubiral kiged jöb ungsilg-a bicilge-yin jarim asagudal' [Some problem on the phonetic development of the Ordos dialect and the correct pronunciation and spelling], *MKUJ* 1998/2: 81–4.
- ERDENICUGLA (1988), 'Jegün Yogor kele bolon Monggol kelen-ü bogoni egesig abiyanu qaricagulul' [Comparison of short vowels of Shira Yugur and Mongolian], ÖMBYS 1988/4: 38–49.
- ——(1998), 'Orcin cag-un Monggol kelen-ü üge-yin ergülte-yin sinji cinar üiledül-ün tuqai' [Nature and function of modern Mongolian stress], MKUJ 1998/3: 81–6.
- ERDENIMÖNGKE (1987), 'Ordos aman ayalgun-u **i* egesig-ün tuqai' [The vowel **i* in the Ordos dialect], *MKB* 1987/2: 9–18.
- ESENOVA, Tamara Sarangovna (1979), 'Vosprijatie nekotoryh intonacionnyh tipov kalmyckogo jazyka' [Perception of some intonation types in Kalmuck], in B. D. Muniev (ed.), *Glagol i otglagol'nye formy sovremennogo kalmyckogo jazyka*, 187–97 (Èlista: KNIIIFÈ).

- ESENOVA, Tamara Sarangovna (1980), 'Tipy intonacii kalmyckih predloženij (èksperimental'no-fonetičeskoe issledovanie)' [Intonation types of Kalmuck clauses (experimental phonetic investigation)], VGS 148–54.
 - —(1984), 'Tipy intonacii kalmyckih predloženij: auditorskij analiz i ob"ektivnye harakteristiki' [Intonation types in Kalmuck clauses: auditory analysis and objective characteristics], in D. A. Pavlov (ed.), *Voprosy kalmyckoj filologii*, 29–35 (Èlista: Kalmyckij gosudarstvennyj universitet).
- —(1985), 'Ob istoričeskoj interpretacii otdel'nyh intonacionnyh dannyh mongol'skih jazykov' [The historical interpretation of certain intonation facts in Mongolian languages], in Voprosy sovetskoj tjurkologii: tezisy dokladov i soobŝenij, 98–100 (Ašhabad).
 - —(1986*a*), 'Izmenenie osnovnyh parametrov intonacii i raspredelenie ih maksimumov v kalmyckoj fraze' [Modification of basic intonation parameters and the distribution of their maxima in the Kalmuck phrase], in Rassadin (1986), 36–43.
- ——(1986*b*), 'O nekotoryh intonacionnyh paralleljah v altajskih jazykah' [Some intonation parallels in Altaic languages], *PIAC* 29: 56–7.
 - —(1987*a*), 'Intonacionnoe vyraženie stepeni svjazannosti složnogo bessojuznogo predloženija v kalmyckom jazyke' [Intonational expression of the degree of cohesion of complex sentences without conjunctions in Kalmuck], in Solncev (1987), 3: 183–8.
- ——(1987b), 'K prosodičeskoj harakteristike kalmyckih govorov' [The prosodic characterisation of Kalmuck dialects], in L. D. Šagdarov (ed.), *Dialektnaja leksika v mongol'skih jazykah*, 150–65 (Ulan-Udè: BF).
- [----] Yesenova (1987c), 'On tempo divergences in Mongolian languages', in Ü. Viks (ed.), *Proceedings of the Eleventh international congress of phonetic sciences*, 3: 141–4 (Tallinn: Academy of sciences of the Estonian SSR).
 - —(1989a), 'O sootnošenii dlitel'nosti i intensivnosti zvukov v voprositel'nyh frazah kalmyckogo jazyka' [The relation between duration and intensity of sounds in Kalmuck interrogative phrases], in Šagdarov (1989), 122–30.
 - —(1989b), 'Dinamičeskie harakteristiki tipov intonacii (èksperimental'no-fonetičeskoe issledovanie na materiale kalmyckogo jazyka)' [Dynamic characteristics of intonation types (experimental phonetic investigation based on Kalmuck data)], in Abaev, Pubaev, and Čimitdoržiev (1989), 165–6.
 - —(1992*a*), 'Realizacija i vosprijatie osnonvnyh tipov intonacii v mongol'skom jazyke' [The realization and perception of basic intonation types in Mongolian], in V. M. Solncev (ed.), *VI Meždunarodnyj kongress mongolovedov: doklady rossijskoj delegacii*, 2: 50–9 (Moskva: Institut vostokovedenija RAN).
- (1992b), 'Modifikacija dinamičeskoj programmy slova vo fraze' [Modification of the dynamic program of a word in a phrase], in È. R. Tenišev (ed.), *Mongol'skij lingvističeskij sbornik*, 13–25 (Moskva: Institut jazykoznanija RAN).
- —(1992c), Prosodičeskaja organizacija frazy i singarmonizm (èksperimental'no-fonetičeskoe issledovanie na materiale mongol'skih jazykov) [The prosodic organization of the phrase and vowel harmony (experimental phonetic investigation based on Mongolian languages)], Avtoreferat dokt. diss., Sankt-Peterburgskij gosudarstvennyj universitet.
- —(1992*d*), 'Vosprijatie tipov intonacii burjatskogo jazyka' [Perception of intonation types in Buriad], in *Tezisy i doklady naučnoj konferencii Banzarovskie čtenija*, 134–7 (Ulan-Udè: BION).
- —(1998), 'O tipah garmonii glasnyh v mongol'skih jazykah' [Types of vowel harmony in Mongolian languages], in Bazarov and Tulohonov (1998), 252–4.
- (2001), 'Kalmykov issykkul'skih jazyk' [The language of the Ysyk Köl Kalmucks], in Jarceva (2001), 25–9.

—and Kermen Uljumandžievna Mandžieva (1986), 'Fonetičeskoe oformlenie voprositel'nosti (po dannym melodiki)' [The phonetic formation of interrogativity (by melodicity data)], in Nadeljaev (1986), 146–9.

- FIELD, Kenneth L. (1997), A grammatical overview of Santa Mongolian. Ph.D. dissertation, University of California, Santa Barbara.
- FINCH, Roger (1986), 'Toward a reconstruction of the Chinese characters used for the Secret history of the Mongols', Kyôto sangyo daigaku kokusai gengo kagaku kenkyûsho shohô (The Bulletin of the International institute for linguistic sciences, Kyoto sangyo University), 7/2: 214–39.

-----(1989–91), 'The velar and uvular consonants in Middle Mongolian', ZASB 22: 164–89.

FRANZÉN, Vivan (2001), 'An acoustic investigation of Mongolian vowels', Mongolica 11(32), 342–51.

— and Jan-Olof Svantesson (1999), 'Remarks on the Mongolian vowel system', in M. Juntunen and B. Schlyter (eds.), *Return to the silk routes: current Scandinavian research on Central Asia*, 109–17 (London: Kegan Paul).

GALSAN, Sèrèènèngijn (1969), 'Mongol hèlnij avian züjn nèg üzègdèl' [A phenomenon in Mongolian phonetics], *HZS* 6: fasc. 8 (pp. 225–9).

——(1976), 'Sokraŝenie slogov v sistemah russkogo i mongol'skogo jazykov' [Reduction of syllables in the systems of Russian and Mongolian], *StMo* 3(11): fasc. 11 (pp. 183–92).

- GANTOGTOH, Gončigijn (1978), 'Halhyn ajalguuny hamžin šürgèh gijgüülègčijn sèlgèè' [Affricate alternations in the Halh dialect], *HZS* 13: fasc. 12 (pp. 98–101).
- ——(1982), 'Mongol hèlnüüdijn hamžin šürgèh gijgüülègčijn sèlgèh tohiroh bolomž' [The possibilities of affricate change and correspondence in Mongolian languages], *HZS* 15: fasc. 14 (pp. 115–32).
- [——] Gangtogtaqu (1990), 'Monggol bičig-ün kele kiged Buriyad ayalgu' [Written Mongolian and the Buriad dialect], MKUJ 1990/1: 72–9.
- ——(1993), 'Mongol bičgijn hèl hijgèèd Mongol dah' Buriad ajalguu' [The Mongolian written language and the Buriad dialect of Mongolia], in Rassadin (1993), 158–170.
- GARUDI (1996), 'Dumdadu erten-ü Monggol kelen-ü üge-yin ekin-ü üy-e-yin *u/ü* egesig-ün nigen nököceltü qubiral-un jim-e qauli' [A conditional law of change of the vowel *u/ü* in the first syllable in Middle Mongolian], *MKB* 1996/3: 8–16.
- ——(1999), 'Üge-yin ekin-ü egesig u/ü-yin nigen teüken qubiral-un ucir' [A historical change of the word initial vowel u/ü], MKB 1999/10: 11–15.
- GEORG, Stefan (2003*a*), 'Ordos', in Janhunen (2003*f*), 193–209.
- (2003*b*), 'Mongghul', in Janhunen (2003*f*), 286–306.
- GERASIMOVIČ, Ljudmila Konstantinovna (1970), 'K voprosu o haraktere udarenija v mongol'skom jazyke' [The problem of the character of stress in Mongolian], VLU 1970/14: 131–7.

——(1975), Mongol'skoe stihosloženie: opyt èksperimental'no-fonetičeskogo issledovanija [Mongolian versification: an experimental phonetic investigation] (Leningrad: Izd. Leningradskogo universiteta).

- GERELTÜ, N. (1986), 'Alaša aman ayalgun-u ündüsün geyigülügci abiyan-u tuqai' [The basic consonant sounds of the Alshaa dialect], *MKB* 1986/4: 20–5.
- —(1991), 'Alaša aman ayalgun-u egesig jokicaqu yoso-yi Qobog Sayiri-yin Torgud aman ayalgun-u egesig jokicaqu yoson-luga qaricagulju üjekü ni' [Vowel harmony in the Alshaa dialect compared with vowel harmony in the Hobogsair Torguud dialect], *MSYQ* 2: 1087–90.

- GERELTÜ, N. (1992), 'Alaša aman ayalgu kiged Qobog Sayiri-yin Torgud aman ayalgun-u egesig jokicaqu yoson-u qaricagulul' [Comparison of vowel harmony of the Alshaa and Hobogsair Torguud dialects], *MKB* 1992/7: 34–40.
- ——(1999), 'Oyirad ayalgun-u urugsilagsan egesig-ün tuqai' [Fronted vowels in the Oirad dialect], *MKUJ* 1999/6: 24–9, 56.
- GOLDSMITH, John A. (1985), 'Vowel harmony in Khalkha Mongolian, Yaka, Finnish and Hungarian', *Phonology yearbook*, 2: 253–75.

—(1990), Autosegmental and metrical phonology (Oxford: Blackwell).

(ed.) (1995), The handbook of phonological theory (Oxford: Blackwell).

- GRASSMANN, Hermann (1863), 'Ueber die aspiraten und ihr gleichzeitiges vorhandensein im an- und auslaute der wurzeln', Zeitschrift für vergleichende Sprachforschung auf dem Gebiete des deutschen, griechischen und lateinischen, 12: 81–138.
- GREGERSON, Kenneth J. (1976), 'Tongue-root and register in Mon-Khmer', in P. Jenner, L. Thompson, and S. Starosta (eds.), *Austroasiatic studies (Oceanic linguistics special publication*, 13), 1: 323–69 (Honolulu: University Press of Hawaii).
- GRIVELET, Stéphane (2001), 'Digraphia in Mongolia', International journal of the sociology of language, 150: 75-93.
- GRØNBECH, Kaare (1940), 'Der Akzent im Türkischen und Mongolischen', ZDMG 94: 375– 90.
- GRUNZEL, Josef (1888), 'Die Vocalharmonie der altaischen Sprachen', Sitzungsberichte der Kaiserlichen Akademie der Wissenschaften: philosophisch-historische Classe, 117/3 (pp. 1–42).
 - —(1895), Entwurf einer vergleichenden Grammatik der altaischen Sprachen nebst einem vergleichenden Wörterbuch (Leipzig: W. Friedrich).
- Guō Shǒuxiáng (1988), 'Měnggǔyǔ zhōng Éyǔ jiècí chūtàn' [Preliminary remarks on Russian loan-words in Mongolian], NDX 1988/1: 117–24.
- GUW-A, D. (1993), 'Küriy-e aman ayalgun-u egesig abiyalaburi-yi bicig-ün kele-tei qaricagulqu ni' [Comparison of the vowel phonemes in the Hüree dialect and the written language], *MKB* 1993/2: 24–9.
 - —(1995), 'Monggol kelen-ü geyigülügci abiyalaburi r jici tegün-ü qubiralta' [The Mongolian consonant phoneme r and its change], ÖMBYS 1995/3: 24–30.
- ——(1996), 'Monggol Manju bicig-ün kelen-ü egesig-ün jokical-un qaricagulul' [Comparison of vowel harmony in written Mongolian and Manchu], *MKB* 1996/2: 51–5.
- HAENISCH, Erich (1925), 'Beiträge zur mongolischen Schrift- und Volkssprache', Mitteilungen des Seminars für Orientalische Sprachen an der Friedrich-Wilhelms-Universität zu Berlin, 28/1: 170–85 (5 plates).
- HAGIWARA Shôzô (1975), Shin, kyû ryô bunki taishô shin Mongorugo bunpôsho [New Mongolian grammar contrasting the new and old scripts] (Tôkyô: Nippon Mongoru kyôkai).
- HALLE, Morris and William Idsardi (1995), 'General properties of stress and metrical structure', in Goldsmith (1995), 403–43.
 - and Jean-Roger Vergnaud (1981), 'Harmony processes', in W. Klein and W. Levelt (eds.), *Crossing the boundaries in linguistics: studies presented to Manfred Bierwisch*, 1–22 (Dordrecht: Reidel).

-(1987), An essay on stress (Cambridge, MA: MIT Press).

- HAMBIS, Louis (1946), *Grammaire de la langue mongole écrite (première partie)* (Paris: Maisonneuve).
- HAMMOND, Michael (1986), 'The obligatory-branching parameter in metrical theory', *NLLT* 4: 185–228.
- HAMP, Eric P. (1958), 'Vowel harmony in Classical Mongolian', Word, 14: 291-4.

----(1980), 'Mongolian vocalic features and the problem of harmony', in Vago (1980), 101-11.

- HANGIN, John G. and John C. Street (1962), *Chakhar survey* (New York: American council of learned societies).
- HARADA Ryûji (1999), 'Syllabification in Khalkha Mongolian and output-output correspondence', *Phonology at Santa Cruz*, 6: 13–24.
- HATTORI Shirô (1936), 'Man Mô chihô no gengoon ni tsuite' [Speech sounds in in the Manchurian–Mongolian area], OGK 43: 8–11.

-(1937a), 'Môkogo no enshin boin to sono tenshahô no yurai' [Rounded vowels in Mongolian and the origin of their transcription], *OGK* 47: 5–7.

—(1937b), 'Horonbairu no Môkogo' [Hölönbuir Mongolian], Môkogaku, 1: 97–111.

- ——(1939), 'Môkogo bungo no kigen ni tsuite' [The development of Written Mongolian] *GK* 3: 1–27.
- ——(1940), 'Buriyâto hôgen no bunrui' [Classification of Buriad dialects], Môko gakuhô, 1:40-86.
 - —(1943), 'Môkogo gaisetsu' [Outline of Mongolian], in *Môko to sono gengo*, 193–287 (Ôsaka: Yukawa kôbunsha).
- ——(1946), Genchô hishi no Môkogo o arawasu kanji no kenkyû [Studies on the Chinese characters which represent the Mongolian language in the Secret history of the Mongols] (Tôkyô: Ryûbun shokyoku).
- —(1951), 'Môkogo Chaharu hôgen no on'in taikei', *GK* 19–20: 68–102. English version 1993: 'The phonological system and structure of the Chakhar dialect of Mongolian', in *Hattori Shirô ronbun shû*, 4: 42–68 (Tôkyô: Sanseidô).
- ——(1959), 'Môko sogo no boin no nagasa', GK 36: 40-54. English version 1959: The length of vowels in Proto-Mongol (StMo 1: fasc. 12).
 - —(1972), 'Initial plosives of Proto-Mongolian and their later developments', *Gengo no kagaku*, 3: 63–92.
- ----(1973), 'The Chinese dialect on which the transcription of the Yüan-ch'ao mi-shih was based', Acta Asiatica (Bulletin of the Institute of Eastern culture [Tôhô gakkai]), 24: 35-44.
- —(1975), 'Boin chôwa to chûki Chôsengo no boin taikei' [Vowel harmony and the Middle Korean vowel system], *Gengo no kagaku*, 6: 1–22.
- —(1976), 'Some problems on the reconstruction of the Mongolian sounds of the *Secret history of the Mongols:* on the word *gu-un*', in Heissig et al. (1976), 261–72.
- (1978), 'Arutai sho gengo, Chôsengo, Nihongo no boin chôwa' [Vowel harmony in Altaic languages, Korean, and Japanese], *Gengo*, 7/4: 80–8. English version 1979: 'Vowel harmonies of the Altaic languages', *OUMÈ 3 IH* 3: 316–22.
- (1980), 'The vowel system of Old Japanese and its vowel harmony in relation to those of Korean and the Altaic languages', in *Mélanges offerts à M. Charles Haguenauer* en l'honneur de son quatre-vingtième anniversaire: études japonaises, 31-8 (Paris: L'Asiathèque).
- ----(1983a), 'Môko sho gengo no *i no ore' [Breaking of *i in Mongolian languages], *Gengo*, 12/6: 109–13.
- -(1983b), 'Môko sho gengo no **i* no ore saisetsu' [Further remarks on the breaking of **i* in Mongolian languages], *Gengo*, 12/11: 100–5; 12/12: 104–8.

—(1983c), 'Vowel harmony or consonant harmony?', in K. Sagaster and M. Weiers (eds.), *Documenta barbarorum: Festschrift für Walther Heissig zum 70. Geburtstag*, 196–202 (Wiesbaden: Harrassowitz).

HATTORI Shirô (1984), 'Pakupa ji (Pasupa ji) ni tsuite: toku ni *e* no ji to *ė* no ji ni kanshite' [On the 'Phags-pa script: especially concerning the letters *e* and *ė*], *Gengo*, 13/7: 100–4; 13/8: 116–21.

(1989), 'The hP'ags-pa letters e and e represent one and the same Mongolian vowel', in
 W. Heissig and K. Sagaster (eds.), Gedanke und Wirkung, Festschrift zum 90. Geburtstag
 von Nikolaus Poppe, 146–51 (Wiesbaden: Harrassowitz).

HAYES, Bruce (1985), A metrical theory of stress rules (New York: Garland).

HÉ Jírén (1989), 'Yúnnán Měnggǔzú yǔyán jí qí xìshǔ wèntí' [The language of the Mongols in Yunnan province and the problem of its affiliation], *MY* 1989/5: 25–36.

——(1998), 'Guānyú Yúnnán Měnggŭzú Kăzhuóyǔ de xíngchéng' [The formation of the Katso language of the Mongols in Yunnan], MY 1998/4: 51–4.

HEISSIG, Walther, John R. Krueger, Felix J. Oinas, and Edmond Schütz (eds.) (1976), *Tractata altaica Denis Sinor sexagenario optime de rebus altaicis merito dedicata* (Wiesbaden: Harrassowitz).

HELGASON, Pétur (2002), *Preaspiration in the Nordic languages: synchronic and diachronic aspects*. Ph.D. dissertation, Stockholm University.

[HELIMSKIJ] Helimski, Evgenij Arnol'dovič (1984), 'A distinctive feature which became a phoneme: the case of Monguor', in *Fifth international phonology meeting, June 25–28, 1984, Eisenstadt: abstracts,* 27 (Wien: Universität Wien).

HIGUCHI Kôichi (1999), 'Implications of mistakes in Mongolian Buddhist works', AH 9: 65–92.

Номолоv, Mihail Petrovič (1958), 'Nekotorye dannye o govore bohanskih burjat' [Some facts about the dialect of the Boohon Buriads], in *Sbornik trudov po filologii*, 3: 277–82 (Ulan-Udè: BMNIIK).

(1965), 'Bohanskij govor' [The Boohon dialect], TBurNII 17: 35-70.

HONG, [Sŏng-hun] Sung-Hoon (1994a), Issues in round harmony: grounding, identity and their interaction. Ph. D. dissertation, University of Arizona.

-----(1994*b*), 'Identity as an independent condition on phonological rules', *CLS* 30/1: 184–98.

- ——(1996), 'Halha Monggolö moüm chohwa-ŭi ch'oejöksöng iron bunsök' [Halh Mongolian vowel harmony in Optimality Theory], *Ŏnö (Korean journal of linguistics)*, 21: 959–79.
- HUÁNG Zōngjiàn (1993), '*Huá-Yi yìyǔ* de Měnggǔyǔ císhǒu h' [Initital h in the Mongolian language of *Hua-Yi yiyu*], MY 1993/4: 19–22.

[----] Huan Zün Žian (1997), 'Hua i i ju tolin dah' mongol hèlnij ügijn èhnij h-gijn tuhaj'
 [Initital h in the Mongolian language of Hua-Yi yiyu], Mongolica, 6(27) (OUMÈ 6 IH
 2): 78-88.

HULST, Harry van der (1987), 'The dual interpretation of lil, lul and lal', *NELS* 18/1: 208–22.

—and Norval Smith (1986), 'On neutral vowels', in K. Bogers, H. van der Hulst, and M. Mous (eds.), *The phonological representation of suprasegmentals*, 233–79 (Dordrecht: Foris).

— (1987), 'Vowel harmony in Khalkha and Buriat (East Mongolian)', in *Linguistics in the Netherlands 1987*, 81–9.

———(1988), 'Tungusic and Mongolian vowel harmony: a minimal pair', in *Linguistics in the Netherlands 1988*, 79–88 (Dordrecht: Foris).

and Jeroen van de Weijer (1995), 'Vowel harmony', in Goldsmith (1995), 495–534.

HUNG, William (1951), 'The transmission of the book known as the Secret history of the Mongols', Harvard journal of Asiatic studies, 14: 433–92.

- HYMAN, Larry (1977), 'On the nature of linguistic stress', in L. Hyman (ed.), Studies in stress and accent (Southern California occasional papers in linguistics, 4), 37-82 (Los Angeles, University of Southern California).
- [IBRAHIM, A.] Ā. Yībùlāhēimài (1985), 'Gānsù jìng nèi Tángwānghuà jìlüè' [Sketch of the Tangwang dialect in Gansu], MY 1985/6: 33-47.

-] Ā. Yībùlāhēimài (1988), 'Dōngxiāngyŭ de yīnwèi' [The phonemes of Santa], in Dongxiangyŭ lùnjí, 65–75 (Lánzhou: Gansù mínzú).

ICHINOSE Megumi (1992a), 'Mongorugo no fukugôgo no on'inteki tokuchô' [Phonemic characteristics of compound words in Mongolian], HBK 21: 105-19.

- -(1992b), 'Uchi-Môkogo Chaharu hôgen ni okeru *i* oto ni tsuite' [The sound *i* in the Chahar dialect in Inner Mongolia], Hokkaidô daigaku bungakubu kiyô, 73: left 169-88.
- IDAM, J. (1994), 'Caqar aman ayalgun-du ekin-ü abiy-a-yi soliju dagudadag üges-ün tuqai' [Words whose initial sound is changing in the Chahar dialect], MKUJ 1994/3: 84-6.
- ILAN, J. and Saran (1998), 'Jirim-ün yariyan-u abiyan-u tokiralcagan-daki olan talatu sinji' [The multilateral quality of the sound correspondences of the Jirim dialect], MKB 1998/5: 11-21.
- ILIŠKIN, Ivan Kuznecovič (1973), Očerki sopostavitel'noj grammatiki russkogo i kalmyckogo jazykov: fonetika i morfologija [Essays on the contrastive grammar of Russian and Kalmuck: phonetics and morphology] (Èlista: KKI).
- [IRINCIN, Yekeminggadai] Yìlínzhēn (1976–8), 'Wèiwútǐ Měngwén hé gǔ Měnggǔyǔ yǔyīn' [Uigur Mongolian script and Old Mongolian speech sounds], NDX 1976/2: 69-76; 1977/2: 70-2; 1978/1: 65-73.
- Itô Junko (1989), 'A prosodic theory of epenthesis', NLLT 7: 217-59.
- [JAGUNASUTU] Zhàonàsītú (1964), 'Tǔzúyǔ gàikuàng' [Sketch of Monguor], ZY 1964/6: 477-96.
- [----] Zhàonàsītú (1981a), Tǔzúyǔ jiǎnzhì [Short description of Monguor] (Běijīng: Mínzú).
- [-----] Zhàonàsītú (1981b), Dōng bù Yùgùyǔ jiǎnzhì [Short description of Shira Yugur] (Běijīng: Mínzú).
- [-----] Zhàonàsītú (1987*a*), 'Dōng bù Yùgùyǔ' [Shira Yugur], *ZSMY* 609–18. [-----] Zhàonàsītú (1987*b*), 'Tǔzúyǔ' [Monguor], *ZSMY* 619–29.
- [----] Zhàonàsītú (1987c), 'Bāsībāwén yuányīn zìmǔ zìxíng wèntí shàng de liǎng zhǒng tǐxì' [The two systems of 'Phags-pa vowel letters], MY 1987/4: 1-6.
- [-----] Zhàonàsītú (1988), 'Yǒuguān Bāsībā zìmǔ è de jǐgè wèntí' [Some problems of the letter è in the 'Phags-pa script], MY 1988/1: 1-7, the article continues on the unpaginated inside back cover.
- —] Zhàonàsītú (1989), 'Bāsībāzì zhōng de líng shēngmǔ fúhào' [Zero initial letters in Ĩ the 'Phags-pa script), MY 1989/2: 29-36.
- [-----] Zhàonàsītú (1992), 'Tǔwén' [The Monguor script], in [Dobo] Dàobù and Tán Kèràng (eds.), Zhōngguó shǎoshù mínzú wénzì, 194-9 (Běijīng: Zhōngguó zàngxué).
- [-----] Zhàonàsītú and Lǐ Kèyù (1982), 'Tǔzúyǔ Mínhé fāngyán gàishù' [Sketch of the Mínhé dialect of Monguor], in Mínzú yǔwén yánjiū wén jí, 458-87 (Xīníng: Qīnghǎi mínzú).
- JAHONTOVA, Natal'ja Sergeevna (1997a), 'Mongol'skie jazyki' [Mongolian languages], in Jarceva (1997), 10-18.
 - -(1997b), 'Mongol'skij jazyk' [Mongolian], in Jarceva (1997), 108-24.
- JALCIB (1962), 'Bagarin aman ayalgun-u qoos, urtu geyigülügci' [Consonant clusters and long consonants in the Baarin dialect], ÖMYSS 1962/1: 91-107.
- [JALSAN] Jiǎlāsēn (1989), 'Guānyú Bāsībāwén de yuányīn ぢに、 ぢしう hé ぢい' [The vowels Ø, y, and *i* in the 'Phags-pa script], NDX 1989/2: 58-63.

- JALSAN and BADMAODSER (eds.) (1997), Sinjilegen-üögülel durasqal-un jokiyal: Cenggeltei profèssor-un kiciyel kömüjil erdem sinjilegen-ü ajil-un tabin jil-ün oi-du; Lùnwén yŭ jìniànwén jí; A collection of papers and commemorative essays (Kökeqota: Öbör Monggol-un keblel-ün qoriy-a).
- JANHUNEN, Juha (1981), 'Korean vowel system in North Asian perspective', *Hangŭl* (*Hangeul*), 172: 129–46.

— (1988), 'Preliminary notes on the phonology of modern Bargut', *Studia orientalia*, 64: 353–66.

——(1990a), Material on Manchurian Khamnigan Mongol (Castrenianumin toimitteita, 37) (Helsinki: Helsinki University).

- (1990*b*), 'On breaking in Mongolic', *PIAC 32*: 181–91.
- (1992), 'On the position of Khamnigan Mongol', JSFOu 84: 115-43.
- (1998), 'Ethnicity and language in prehistoric Northeast Asia', in R. Blench and M. Spriggs (eds.), Archaeology and language, II: 195–208 (London: Routledge).
- ——(1999), 'Laryngeals and pseudolaryngeals in Mongolic: problems of phonological intepretation', *CAJ* 43/1: 114–31.
- (2001), 'On the phonological interpretation of the retroflex preinitials in the Mongolic languages of the Amdo region', in L. Károly and É. Kincses Nagy (eds.), Néptörténet – nyelvtörténet: a 70 éves Róna-Tas András köszöntése, 61–9 (Szeged).
- (2003*a*), 'Proto-Mongolic', in Janhunen (2003*f*), 1–29.
- (2003*b*), 'Written Mongol', in Janhunen (2003*f*), 30–56.
- —(2003c), 'Khamnigan Mongol', in Janhunen (2003f), 83–101.
- -----(2003*d*), 'Mongol dialects', in Janhunen (2003*f*), 177–92.
- -----(2003e), 'Para-Mongolic', in Janhunen (2003f), 391-402.
- (ed.) (2003f), The Mongolic languages (London: Routledge).
- JARCEVA, Viktorija Nikolaevna (ed.) (1997), Jazyki mira: mongol'skie jazyki, tungusoman'čžurskie jazyki, japonskij jazyk, korejskij jazyk [The languages of the world: Mongolian languages, Tungus-Manchu languages, Japanese, and Korean] (Moskva: Indrik).

——(ed.) (2001), Jazyki Rossijskoj federacii i sosednih gosudarstv: ènciklopedija [The languages of the Russian Federation and neighbouring states: encyclopedia], vol. 2 (Moskva: Nauka).

JENSEN, John T. and Margaret Stong-Jensen (1979), 'The relevancy condition and variables in phonology', *Linguistic analysis*, 5: 125–60.

JIĂXĪrú (1982), 'Hăixī Měnggǔyǔ de tèdiǎn' [Characteristics of Haixi Mongolian], in Mínzú yǔwén yánjiū wén jí, 405–17 (Xīníng: Qīnghǎi mínzú).

——(1986a), 'Wūtúměirén tǔyǔ k fǔyīn de yǎnbiàn' [Changes of the consonant k in the Urtmörön dialect], in Qīnghǎi shǎoshù mínzú yǔwén yánjiū wén jí, 330–8 (Xīníng: Qīnghǎi mínzú xuéyuàn). [Originally published in 1980].

— (1986b), 'Wūtúměirén tůyů zhông de i yuányin kǎo' [A study on the vowel i in the Urtmörön dialect], in Qinghǎi shǎoshù mínzú yùwén yánjiū wén jí, 339–50 (Xiníng: Qinghǎi mínzú) xuéyuàn. [Reprint].

- JIN Gang (1983), 'Monggol kelen-ü abiyan-u bütüce' [The sound structure of Mongolian], MKB 1983/4: 77–94; 1983/5: 82–95.
 - (1987), 'Monggol kelen-ü egesig-ün ilgaburi' [Distictive features for Mongolian vowels] *MKSÖ*, 520–53.
- JIRANNIGE, Jigucidai (1985), 'Monggol bicig-ün kelen-ü qoos egesig-ün asagudal-du' [The problem of diphthongs in Written Mongolian], *MKUJ* 1985/2: 18–26.
 - (1999), 'Orcin cag-un Monggol kelen-ü urtu egesig-ün garul egüsül-ün tuqai' [The origin of long vowels in modern Mongolian], *MKUJ* 1999/2: 4–19.

- Jôo Hakutarô (1971), 'Saundosupekutoru bunseki ni yoru gendai Mongorugo boin no ichikôsatsu' [Spectrographic analysis of modern Mongolian vowels], OGK 138: 14–16.
 - ----(1973*a*), 'Jikken onseigaku ni yoru Mongorugo no kansatsu' [The Mongolian language from the point of view of experimental phonetics], *OK* 16: 207–29.
 - —(1973*b*), 'Chaharu hôgen no /t/ to /d/ ni kansuru ichikôsatsu' [Investigation of /t/ and /d/ in the Chahar dialect], *Lexicon*, 2: 28–38.
- ——(1975), 'Onkyô onseigaku ni yoru Chaharu hôgen no boin bunseki' [Acoustic phonetic analysis of the vowels in the Chahar dialect], *Lexicon*, 4: 12–21.
- ——(1976a), 'Arutai go taishô kenkyû: onsetsuso no kôzô' [Contrastive study of the Altaic languages: the structure of syllabemes], OK 17: 241–57.
- ——(1976b), 'Mongorugo no boin chôwa' [Mongolian vowel harmony], Gengo, 5/6: 53–61.
- ——(1991–2), 'Mongorugo no boin chôwa to koki ryûryô: seiri jikken ni yoru kenkyû' [Mongolian vowel harmony and expiratory airflow: an experimental physiological study], NMGK 22–23: 104–17.
 - —(1992), 'Seiri onseigakuteki jikken ni yoru Mongorugo no boin chôwa' [A physiological interpretation of Mongolian vowel harmony], Bungei gengo kenkyû, gengo hen, Tsukuba daigaku (Studies in language and literature, Language, University of Tsukuba), 21: 1–52.
- —(1997), Jikken onseigaku kenkyû [Studies in experimental phonetics] (Tôkyô: Benseisha). [Section 1–1 (pp. 21–61): 'Haruha hôgen no boin bunseki' [Analysis of Halh vowels]; section 1–2 (pp. 62–86): 'Chaharu hôgen no boin bunseki' [Analysis of Chahar vowels]; section 3–2 (pp. 209–38): 'Gendai Mongorugo ni okeru boin chôwa' [Vowel harmony in modern Mongolian]]
- and Mikami [Satoshi] Tsukasa (1981), 'Mongorugo no akusento: ôtosegumento riron ni yoru bunseki' [Mongolian accent: an autosegmental analysis], Bungei gengo kenkyû, gengo hen, Tsukuba daigaku (Studies in language and literature, Language, The University of Tsukuba), 6: 143–65.
- KAKUDÔ Masayoshi (1974), 'Haruha hôgen no seishohô' [Halh orthography], NMGK 5: 29-36.
- ——(1982*a*), 'Haruha Mongorugo no pitchiakusento' [Pitch accent in Halh Mongolian], $\hat{O}GDG$ 56: 31–49.
- ——(1982b), 'Dunshan hôgen no on'in henka' [Phonological changes in Santa], ÔGDG 59: 17–35.
- —(1987a), 'Dagûrugo Nantun hôgen no tokuchô' [Characteristics of the Emüne Ayil dialect of Dagur], ÔGDG 74/1–2: 1–18.
- ——(1987b), 'Dozokugo no kai hôgen' [Subdialects of Monguor], ÔGDG 75/1–2: 49– 63.
 - —(1988), '*Geser Rëdzia-wu* (Dozokugo no ichi kai hôgen) no gengo' [The language of *Geser Rëdzia-wu* (a subdialect of Monguor)], ÔGDG 76/1–2: 25–50; 77: 23–44.
 - —(1990*a*), 'Dozokugo (Monguorugo) no ichi hôgen no jiyû kôtai: *Aus der Volksdichtung der Monguor* no gengo' [Free variation in a Monguor dialect: the language of *Aus der Volksdichtung der Monguor* [by Dominik Schröder]], *ÔGDR* 3: 65–91.
 - —(1990b), 'Dozokugo no ichi hôgen: Aus der Volksdichtung der Monguor no gengo' [A Monguor dialect: the language of Aus der Volksdichtung der Monguor], Nairiku Ajia kenkyû (Studies on the Inner Asian languages), 6 (Gaikokugaku kenkyû, Kôbe shi gaikokugo daigaku (Annals of foreign studies, Kobe city University of foreign studies), 23): 179–200.
 - -(1990c), 'Dozokugo no seishohô' [The orthography of Monguor], $\hat{O}GDR$ 4: 49–76.

KAKUDÔ Masayoshi (1992), 'Mongoru no uma to yûbokumin ni mirareru Mongorugo no chôboin no hyôki' [The representation of long vowels in the Mongolian language seen in Mongolian horses and nomads [by Nozawa Nobuyuki]], Shichôkaku gaikokugo kyôiku kenkyû, Ôsaka gaikokugo daigaku (Bulletin of the language laboratory, Osaka University of foreign studies), 15: 17–29.

—(1994), 'Xi Yuanlin *Tuzuyu yinwei xitong* ni okeru boin no bunrui ni tsuite' [The classification of vowels in Xi Yuanlin (1986)], *NMGK* 25: 15–28.

——(1995), 'Dunshango no on'in taikei' [The phoneme system of Santa], ÔGDR 13: 31– 56.

- ——(1997), 'Tenshû Dozokugo no tokuchô Gésà'ěr wénkù dai san kan no shiryô ni motozuite' [Remarks on the Tianzhu dialect of Monguor based on the third chapter of the Gesar wenku]). ÔGDR 17: 33–61.
- KAŁUŻYŃSKI, Stanisław (1965), 'Zur Frage der *i*-Brechung im Mongolischen', UAJb 36: 340-7.
- KAMIYAMA Kôhei (1919), *Môkogo nyûmon* [Introduction to Mongolian] (Tôkyô: Zenrin shoin).
- KARA, György (1958), 'Notes sur les dialectes oirat de la Mongolie occidentale', AOH 8: 111–68.
- ——(1962), 'Sur le dialecte üjümücin', *AOH* 14: 145–72.
- (1963), 'Zamečanija ob udžumčinskom dialekte' [Remarks on the Üjemchin dialect], in *Trudy Dvadcat' pjatogo meždunarodnogo kongressa vostokovedov*, 3: 429–31 (Moskva: Izd. vostočnoj literatury).
- ——(1995), AAKI and other Daurica in Cyrillic script: four rare Daur prints re-edited and introduced by G. Kara (Budapest: MTA Altajisztikai kutatócsoport).
- KARLSSON, Anastasia Mukhanova (2001), 'An introductory study of Mongolian intonation', WPLund 49: 82–5.
 - (2003*a*), 'Tonal gestures in Mongolian interrogatives', *Phonum*, 9: 189–92 (Umeå University).
 - (2003*b*), 'Rhythmical and accentual structure of Mongolian', in Solé, Recasens, and Romero (2003), 2465–7.
- and Jan-Olof Svantesson (2002), 'Preaspiration in Halh Mongolian'. Paper presented at the Workshop on Mongolic languages, Taibei, September 2002.
- KAUN, Abigail R. (1995), *The typology of rounding harmony: an optimality theoretic approach*. Ph. D. dissertation, University of California, Los Angeles.

KENSTOWICZ, Michael (1994), Phonology in generative grammar (Oxford: Blackwell).

- KIČIKOV, Anatolij Šalhakovič (1963), Derbetskij govor (fonetiko-morfologičeskoe issledovanie) [The Dörbed dialect (phonetic and morphologic investigation)] (Èlista: Kalmgosizdat).
 - —(1967*a*), 'Differencial'nye priznaki govorov kalmyckogo jazyka (o principah klassifikacii govorov)' [Differentiating features of Kalmuck dialects (on the principles of dialect classification)], *ZKEl* 5: 22–38.
- ——(1967*b*), 'O govore donskih (Buzava) kalmykov' [The dialect of the Don (Buzaawa) Kalmucks], *ZKEl* 5: 39–51.
- KIM Chu-wŏn (1988*a*), *Manju T'unggusŭ cheŏ-ŭi moŭm chohwa yŏn'gu* [A study on vowel harmony in the Manchu-Tungus languages], Ph.D. dissertation, Seoul University.

— (1988b), 'Moŭm chohwa-wa solch'un: Hunmin chongum haelye-ŭi solch'un-e taehayo [Vowel harmony and 'tongue retracted': on 'tongue retracted' in Hunmin chongum haelye], Onohak (Eoneohag, Journal of the Linguistic society of Korea), 9– 10: 29–43.

- (1992), '14 segi moŭm ch'ui kasŏl-e daehan kŏmt'o' [Investigation of the hypothesis of a vowel shift in the 14th century] *Ŏnŏhak (Eoneohag, Journal of the Linguistic society of Korea*), 14: 53–74.
- —(1993), *Moŭm chohwa-ŭi yŏn'gu* [A study of vowel harmony] (Kyŏngsan: Yŏngnam taehakkyo ch'ulp'anbu).
- [-----] Kim Ju-won (1997), 'Remarks on the phonology of Mongolian loanwords in Middle Korean', in Jalsan and Badmaodser (1997), 426–35.

——(1998), 'Monggolŏ-wa Hangugŏ-ŭi umraut'ŭ taejo yŏn'gu' [A contrastive study of umlaut in Mongolian and Korean], AH 8: 121–42.

- (1999a), 'Sŏlgŭn huch'uk (RTR)-gwa sŏlgŭn chŏnjin (ATR): Alt'ai cheŏ-wa 15 segi Hangugŏ-ŭi moŭm chohwa' [Retracted tongue root (RTR) and advanced tongue root (ATR): vowel harmony in Altaic languages and 15th Century Korean], in *Ŏnŏ-ŭi yŏksa: Sŏng Paeg-in kyosu chŏngnyŏn t'oeim kinyŏm nonmunjip*, 311–41 (Sŏul: T'aehaksa).
- ——(1999b), 'Altai cheŏ-wa Hangugŏ-ŭi chŏnsŏlgomoŭmhwa hyŏnsang' [Vowel fronting in Altaic and Korean], AH 9: 233–47.
- KIM Pang-han (1969), 'Monggoŏ pangŏn-ŭi ŏdu chaŭmgun-e kwanhan koch'al' [A study on initial consonant clusters in Mongolian dialects], *Paeksan hakpo*, 6: 79–112.
- ——(1971), 'Wŏnsi Monggoŏ-ŭi changmoŭm-e gwanha-yŏ' [Long vowels in Proto-Mongolian], in *Kim Hyŏng-gyu* [金亨奎] *paksa songsu kinyŏm nonch'ong*, 69–75.
- ——(1973), 'Monguor pangön-ŭi ŏdu chaŭmgun hyöngsöng-e kwanha-yö' [The development of initial consonant clusters in the Monguor dialects], *Öhak yön'gu (Language research)*, 9/2: 20–25.
- Кім Sök-tůk (1976), 'Hyöngt'aeůmso pyöndong yoin ŭrosŏ-ŭi moŭm chohwa: kojŏn Monggoŏ-ŭi kyŏngu' [Vowel harmony resulting from morphophonological alternation: the case of Classical Mongolian], *Hangŭl*, 158: 3–23.
- [----] Kim Suk Deuk (1982), 'A contrastive study of 15th century Korean and Classical Mongolian: vowel harmony', in G. de la Lama (ed.), *30th International congress of human sciences in Asia and North Africa: Japan and Korea*, 1, 34–44 (México: El Colegio de México).
- KIM, Stephen S. (2003), 'Santa', in Janhunen (2003f), 346-63.
- KIPARSKY, Paul (1968), How abstract is phonology? (Bloomington: Indiana University Linguistics Club).
- KLJUKIN, Innokentij Adrianovič (1926), Ključ k izučeniju živoj mongol'skoj reči i pis'mennosti: čast' 1 [Key to learning living Mongolian speech and writing, part 1] (Trudy Gosudarstvennogo dal'nevostočnogo universiteta, 6/4) (Vladivostok).
- KOBAYASHI Takashirô (1954), Genchô hishi no kenkyû [Studies on the Secret history of the Mongols] (Tôkyô: Nihon gakujutsu shinkô kai).
- KöGJILTÜ, Uriyangqai (1982), 'Monggol töröl kele-nügüd-ün ündüsün egesig-ün qaricagulul-un tobci agulg-a' [Summary of Comparison of the basic vowels in the Mongolian languages], ÖMYSS 1982/3: 297–304.
- -----(1983), 'Dörbeljin üsüg-ün l ba π (π)-yin tuqai' [On the letters *e* and *t* in the 'Phags-pa script], *ÖMYSS* 1983/3: 53–60.
- (1984), 'Monggol töröl kele-(ayalgun)-nügüd-ün abiyan-u qubiralta-du nölöge üjegülügsen egesig **i*-yin üiledül-ün tuqai angqan-u ajiglalta' [Preliminary observations on the function of the vowel **i* influencing sound changes in Mongolian languages and dialects], ÖMYSS 1984/4: 13–18.
- ——(1985), 'Monggol bicig-ün kelen-ü a ba Monggol töröl kele-(ayalgu)-nügüd-ün abiyan-u qaricagulul' [Comparison of a in written Mongolian and the sounds of Mongolian languages and dialects], MKB 1985/6: 19–31.

- [KöGJILTÜ, Uriyangqai] Hūgéjílètú (1986a), 'Yuányīn *i duì Měnggů yůzú yůyán huò fāngyán yůyīn yǎnbiàn de yǐngxiǎng chūtàn' [The function of the vowel *i in the sound changes of Mongolian languages and dialects], in Fù Màojì (ed.), Zhōngguó mínzú yǔyán lùnwén jí, 386–94 (Chéngdū: Sìchuān mínzú).
- [——] Hūgéjílètú (1986*b*), 'Shìlùn zhōng shìjì Měnggǔyǔ yuányīn *e* de yīnzhí' [Remarks on the quality of the vowel *e* in Middle Mongolian], *NDX* 1986/3: 41–7.
- ——(1986c), 'Monggol bicig-ün kelen-ü e egesig ba Monggol töröl kele-(ayalgu)-nügüdün abiyan-u qaricagulul' [Comparison of the vowel e in written Mongolian and the sounds of Mongolian languages (dialects)], MKB 1986/11: 6–14.
- [—] Hūgéjílètú (1987a), 'Guānyú shūmiàn Měnggǔyǔ yuányīn e duìyìng yú Měnggǔ yǔzú yǔyán (fāngyán) yuánchún yuányīn de xìngzhì wèntí' [The correspondence of the Written Mongolian vowel e to rounded vowels in Mongolian languages and dialects], NDX 1987/2: 93–100, 109.
- [——] Hūgéjílètú (1987b), 'Měnggůyů Kālāqìn tůyů yůyīn xìtŏng' [The sound system of the Harchin dialect of Mongolian], MY 1987/4: 17–22.
- [——] Hūgéjílètú (1987*c*), 'Zài lùn Bāsībā wén Ư yǔ Γ (Γξ)' [Further discussion of the 'Phagspa letters *e* and *I*), *NDX* 1987/1: 48–56.
- [——] Hūgéjílètú (1989*a*), 'Shìlùn zhōngshìjì Měnggǔyǔ de yuánchún yuányīn' [A discussion on rounded vowels in Middle Mongolian], *NDX* 1989/2: 40–57.
- ——(1989b), 'Monggol bicig-ün kelen-ü i egesig ba Monggol töröl kele-(ayalgu)-nügüdün abiyan-u qaricagulul' [Phonetic comparison of the vowel i in Written Mongolian and in Mongolian languages and dialects], MKUJ 1989/3: 22–39.
- ——(1990), 'Qaracin aman ayalgun-u abiyan-u sistèm' [Sound system of the Harchin dialect], Monggol-un sudulul, 1990/1: 62–72.
- ——(1991), 'Monggol bicig-ün kelen-ü urugul-un egesig ba Monggol töröl kele-(ayal-gu)-nügüd-ün abiyan-u qaricagulul' [Phonetic comparison of rounded vowels in written Mongolian and the Mongolian languages and dialects], *MKB* 1991/6: 50–6; 1991/7: 34–40; 1991/8: 25–30; 1991/9: 15–21.
- ——(1991–2), 'Mongorugo no onsei to onso ron: Mongorugo no jakka boin ni tsuite' [Phonetics and phonology of Mongolian: reduced vowels in Mongolian], *NMGK* 22–23: 144–50.
- [—] Fukejiruto (1993), 'Chûsei Môkogo ni okeru enshin boin no onka ni tsuite' [On the sound values of rounded vowels in Middle Mongolian], *OShKR*, 51–74.
- [-----] Hugjiltu (1999), 'On contraction in Mongolian hP'ags-pa documents', *PIAC* 41: 123-31.
- [-----] Hugjiltu (2003), 'Bonan', in Janhunen (2003f), 325-45.
- KŐHALMI, Katalin Uray- (1959), 'Der mongolisch-kamniganische Dialekt von Dadal Sum und die Frage der mongolisierung der Tungusen in der Nordmongolei und Transbajkalien', AOH 9: 163–204.
- Колма Takeo (1938), Môkogo bunten [Mongolian grammar] (Tôkyô: Bunkyûdô).
- [Köke, Qarnud] Hūhé (1996), 'Měnggǔyǔ r fǔyīn de shēngxué fēnxī' [Acoustic analysis of the Mongolian consonant r], NDX 1996/6: 88–94.
 - —(1997), 'Monggol kelen-ü egesig-ün urtu-yin qubilburi-yin tuqai: dagun-u uqagan-u arg-a bolon büridkel-ün uqagan-u arg-a-bar sinjilekü ni' [Long allophones of Mongolian vowels: an investigation with acoustical and statistical methods], ÖMYSS 1997/3: 38–51.
 - (1998*a*), 'Dagun-u uqagan-u abiy-a jüi-yin arg-a uqagan-iyar Monggol kelen-ü geyigülügci abiy-a-yi ilgal todorqayilaqu asagudal-du' [The problem of explaining the discrimination of Mongolian consonants with acoustic phonetic theory and methods], *ÖMYSS* 1998/4: 6–14.

- [——] Hūhé (1998b), 'Guānyú Měnggůyů de yīnjié wèntí' [The problem of the syllable in Mongolian], MY 1998/4: 73–8.
- ——(1998c), 'Monggol kelen-ü ergülte-yin asagudal-du: ergülte-yin sudulul-un jabsar-un medegülülte' [The problem of Mongolian stress: intermediate report on stress research], *MSYQ* 3: 146–8.

——(1999a), 'Monggol kelen-ü abiyan-u uyaldul-un asagudal' [The problem of coarticulation in Mongolian], MKB 1999/4: 18–22.

- [——] Hūhé (1999b), 'Měnggůyů yuányīn de shēngxué fēnxī' [Acoustic analysis of Mongolian vowels], MY 1999/4: 58–65.
- [—] Hūhé (1999c), 'Měnggůyů yuányīn gùyǒu yīndiào chūtán' [Preliminary research on intrinsic pitch in Mongolian], in Xiàndài yǔyīnxué lùnwénjí: dìsì jiè xiàndài yǔyīnxué xuéshù huìyì, 98–9 (Běijīng: Jīnchéng).
- ——(2001), 'Monggol kelen-ü üge-yin doboyilgalta-yin asagudal-du' [The problem of prominence in Mongolian words], ÖMYSS 2001/2: 1–18.
- [____] Harnud, Huhe (2003a), A basic study of Mongolian prosody (Publications of the Department of phonetics, University of Helsinki, 45) (Helsinki).
- [—] Harnud, Huhe (2003*b*), 'Stress on Mongolian disyllabic words', in Solé, Recasens, and Romero (2003), 2433–6.
- [——] Hūhé, Bào Huáiqiào, and [Coyijongjab] Quèjīngzhābù (1997), 'Guānyú Měnggůyů yůyīn shēngxué cānshù shùjùkù' [Parameter database for Mongolian acoustic phonetics], NDX 1997/5: 44–52.
- [----] Hūhé, Chén Jiāyóu, and Zhèng Yùlíng (2001), 'Měnggǔyǔ yùnlǜ tèzhēng shēngxué cānshù shùjùkù' [Acoustic parameter database of Mongolian prosodic features], NDX 2001/1: 39-43.
 - and Coyijongjab (1996), 'Caqar aman ayalgun-u qoyar üyetü üge-yin qoyadugar üy-eyin bogoni egesig-ün tuqai' [Short vowels in the second syllable of disyllabic words in the Chahar dialect], ÖMYSS 1996/1: 108–26.

———(1999), *Monggol kelen-ü abiyan-u dagun-u uqagan-u jadalulta* [Acoustic analysis of Mongolian speech sounds] (Kökeqota: ÖMYS-yin keblel-ün qoriy-a).

- —— and Sodobagatur (1996), 'Caqar aman ayalgun-u üge-yin ecüs-ün bogoni egesig-ün tuqai' [Short word-final vowels in the Chahar dialect], *ÖMYSS* 1996/3: 8–20.
- [——] Hūhé, Bào Huáiqiào, [Coyijongjab] Quèjīngzhābù, and Chén Jiāyóu (1998), 'Guānyú Měnggůyů yůyīn shēngxué cānshù shùjùkù' [Parameter database for Mongolian acoustic phonetics], AH 8: 201–10.
- [KÖKEBAGATUR, Borjigin T.] Fufubâtoru (1992), 'Mongoru gozoku sho gengo ni okeru gotô shiin no nan'onka' [Weakening of initial strong consonants in Mongolian languages], *HK* 17/3: 141–66.
- Кöкевая (1991), 'Üjümücin aman ayalgun-u abiy-a: kelen-ü jüi-yin ciqula oncalig' [Sounds of the Üjemchin dialect: important linguistic characteristics], *MKUJ* 1991/1: 52-60.
 - —(1994), '*Monggol-un niguca tobciyan-*u kelen-deki qamjiqu abiy-a *q, k, g-*un tuqai' [The stops *q, k, g* in the Secret history of the Mongols], in Monggol-un sudulul-un ögülelün tegübüri, 1–20 (Kökeqota: ÖMSK).
- ——(1996), 'Monggol kelen-ü abiyan-u uyaldul ba üy-e-yin bütüce-yin oncalig' [Sound combinations and syllable structure in Mongolian], ÖMBYS 1996/3: 1–14.
 - —(1998), 'Monggol-un niguca tobciyan-u kelen-ü egesig-ün bütüce-yin jarim asagudaldu' [Some problems on the vowel structure of the language of the Secret history of the Mongols], MKB 1998/11: 25–9.

- KÖKEBARS (1999), 'Monggol kelen-ü egesig abiyalaburi-yin sistèm-ün jarim asagudal-du' [Some problems about the Mongolian vowel phoneme system], *MKB* 1999/12: 13–17.
- —— and Cimeg (1997), *Üjümücin aman ayalgu* [The Üjemchin dialect] (Kökeqota: ÖMYSyin keblel-ün qoriy-a).
- [Kotwicz, Władysław] Kotvič (1902), *Lekcii po grammatikě mongol'skago jazyka* [Lectures on Mongolian grammmar] (Sankt-Peterburg: I. Trofimov).
- [——] Kotvič (1929), *Opyt grammatiki kalmyckogo razgovornogo jazyka* [Grammar of spoken Kalmuck], 2nd ed. (Rževnice u Pragi: Izdanie kalmyckoj komissii kul'turnyh rabotnikov v Čehoslovackoj respublike).
 - ——(1936), 'Contributions aux études altaïques, IV: sons intercalaires', Rocznik orientalistyczny, 12: 122–32.
- ——(1939), La langue mongole, parlée par les Ouïgours Jaunes près de Kan-tcheou: d'après les matériaux recueillis par S. E. Malov et autres voyageurs (Wilno).
- [KOWALEWSKI, Józef] Kovalevskij, Osip Mihajlovič (1835), *Kratkaja grammatika mongol'skago knižnago jazyka* [Short grammar of Written Mongolian] (Kazan': Universitetskaja tipografija).
- KRIPPES, Karl A. (1992), The reconstruction of proto-Mongolian *p-. Ph.D. dissertation, Indiana University, Bloomington.
- KRUEGER, John R. (1975), *The Kalmyk-Mongolian vocabulary in Stralenberg's* Geography *of 1730* (Stockholm: Almqvist & Wiksell).
- KUBO Tomoyuki (1997), 'Reduplication meduplication in Khalkha Mongolian', *GK* 112: 66–97.
- KULDINOW, Sodman (1985), 'The vocal set-up of Kalmyk (Mongolian) language', ZASB 18: 201–7.
- KÜRELBAGATUR, Uriyangqadai (1989), 'Dörbeljin bicig-ün egesig-ün temdeg-ün jarim öbermice sinji-ece' [Some characteristics of the representation of vowels in the 'Phagspa script], MKB 1989/3: 22–27.
 - (1991), 'Monggol töröl keled-ün urtu egesig-tei qamiy-a büküi jarim asagudal-i sinjilekü ni' [Investigation of some problems concerning long vowels in the Mongolian languages], ÖMYSS 1991/3: 14–25.
- ——(1993), 'Caqar aman ayalgun-u cingg-a ügen-ü ekin-ü üy-e-yin *i* egesig-ün tuqai' [The vowel *i* in initial syllables of tense words in the Chahar dialect], *MKB* 1993/11: 25–32.
- (2001), 'Monggol kelen-ü urtu egesig-ün bürildül-tü urugul-un egesig-ün üjegülügsen üiledül' [The function of round vowels in the formation of Mongolian long vowels], $\ddot{O}MYSS$ 2001/4: 12–20.
- KURIBAYASHI Hitoshi (1981*a*), '**i* no ore kô: Môkogo ni okeru **i* on no hatten no kisokusei to fukisokusei' [The breaking of **i*: the regularity and irregularity of the development of **i* in Mongolian], *MK* 12: 32–49.
- ----(1981b), 'Môkogo sho hôgen ni okeru gotô *i on no hatten' [The development of initial *i in Mongolian dialects], *HK* 6/3: 1–16.
- ——(1981c), 'Gendai Mongorugo ni okeru kuchibiru no boin chôwa ni tsuite' [Rounding harmony in modern Mongolian], HK 6/2: 98–112.
- ——(1982a), '*i no ore saisetsu: Haruha hôgen to Orudosu hôgen no hatten no heikôsei' [More on the breaking of *i: parallelism in the development of the Halh and Ordos dialects], MK 13: 37–55.
- —(1982b), 'Môkogo shi ni okeru *i no ore no mondaiten' [Some problems in the breaking of *i in the history of Mongolian], GK 82: 29–47.
- (1984), 'Môkogo ni okeru *uya to *üge no boin shukugô' [The development of Mongolian *uya and *üge], GK 85: 22–48.

- (1985*a*), 'Môkogo sho hôgen ni okeru umurauto genshô' [Umlaut changes in Mongolian dialects], *OK* 21: 89–99.
- (1985b), 'Some problems in the breaking of *i* in Mongolian', OUMÈ 4 IH 2: 118–25.
- (1988), 'Mongorugo ni okeru jakka boin no hattatsu to heionsetsuka genshô' [The development of reduced vowels and the creation of closed syllables in Mongolian], OK 22: 209–23.
- —(1992), 'How the reduced vowels have influenced the syllabic structure of Mongolian', OUMÈ 5 IH 2: 396–401.
- (1993), 'Onsei henka no kisokusei to sono reigai: Dagûrugo ni okeru enshin boin no ore' [The regularity of sound changes and its exceptions: the breaking of rounded vowels in Dagur], Kenkyû kiyô, Nihon daigaku jinbun kagaku kenkyûjo, 45: 37–63.
- (2003), Ka–I yakugo (kôshuhon) Mongorugo zen tango, gobi sakuin [Complete Mongolian word and suffix index to Hua–Yi yiyu] (Sendai: Tôhoku daigaku).
- and Coyijongjab (2001), Genchô hishi *Mongorugo zen tango, gobi sakuin* [Complete Mongolian word and suffix index to the *Secret history of the Mongols*] (Sendai: Tôhoku daigaku).
- KUZ'MENKOV, Evgenij Afanas'evič (1986), 'Labial'naja assimiljacija v srednemongol'skom' [Labial assimilation in Middle Mongolian], VJa 1986/3: 118–23.
- -----(1987), 'Oboznačenie mongol'skogo θ v transkripcii *Sokrovennogo skazanija*' [The representation of Mongolian θ in the transcription of the *Secret history*], in Solncev (1987), 2: 75–83.
- (1988a), 'Mongol'skie èlementy v man'čžurskom i dialektnaja baza staromongol'skoj pis'mennosti' [Mongolian elements in Manchu and the dialect basis of the Old Mongolian script], VJa 1988/4: 117–22.
- ——(1988b), 'O vozmožnoj predystorii mongorskih inicial'nyh f, x, š' [The possible prehistory of Monguor initial f, x, š], Vostokovedenie, 14: 55–65.
- ——(1989), 'Pervičnye dolgie glasnye v srednemongol'skom jazyke' [Primary long vowels in Middle Mongolian], in Šagdarov (1989), 43–9.
- (1993a), 'Fonetičeskaja rekonstrukcija mongol'skogo teksta Juan'-čao bi-ši' [Phonetic reconstruction of the Mongolian text of the Secret history of the Yuan dynasty], in V.
 M. Solncev (ed.), Mongolica: k 750-letiju Sokrovennogo skazanija, 294–327 (Moskva: Nauka).
 - —(1993*b*), 'K rekonstrukcii drevnemongol'skih affrikat' [The reconstruction of Old Mongolian affricates], *Vostokovedenie*, 18: 85–91.
 - —(1997), 'Mogol'skij jazyk' [Moghol], in Jarceva (1997), 91–7.
- LADD, D. Robert (1996), Intonational phonology (Cambridge: Cambridge University Press).
- LEBEN, William R. (1982), 'Metrical or autosegmental', in H. van der Hulst and N. Smith (eds.), *The structure of phonological representations*, 1: 177–90 (Dordrecht: Foris).
- Lǐ Bīng (1996), *Tungusic vowel harmony: description and analysis* (Amsterdam: Holland Institute of Generative Linguistics).
- ——(1999), 'Mănyǔ yuányīn xìtŏng de yănbiàn yǔ yuánshǐ Ā'ěrtàiyǔ yuányīn xìtŏng de chóngxīn gòusì' [The development of the Manchu vowel system and the reconstruction of the Proto-Altaic vowel system], MY 1999/3: 12–24.
- LI, Charles N. (1986), 'The rise and fall of tones through diffusion', in *Proceedings of the twelfth annual meeting of the Berkeley Linguistic Society*, 173–85.
- Lǐ Kèyù (1988), 'Menggǔ'ěr (Tǔzú) yǔ hé Měnggǔyù' [Monguor (Tu) and Mongolian], XMY 1988/1: 214–24, 23.
- Lǐ Měilíng (2001), 'Tǔzúyǔ cháng yuányīn de xíngchéng' [The formation of long vowels in Monguor], XMY 2001/1: 91–101.
- LIEBER, Rochelle (1987), An integrated theory of autosegmental processes (Albany: State University of New York Press).
- LIGETI, Lajos (1955), 'Le lexique Moghol de R. Leech', AOH 4: 119-58.

(1961), 'Trois notes sur l'écriture 'phags-pa', AOH 13: 201-37.

- -----(1964), 'Les voyelles longues en moghol', AOH 17: 1-48.
- LIGHTNER, Theodore (1965), 'On the description of vowel and consonant harmony', *Word*, 21: 244–50.
- LINDAU, Mona (1975), [Features] for vowels (UCLA working papers in phonetics, 30). (Los Angeles).
- (1979), 'The feature expanded', Journal of phonetics 7: 163-76.

LIÚ Zhàoxióng (1965), 'Dongxiangyù gàikuàng' [Sketch of Santa], ZY 1965/2: 153-67.

-----(1981), Döngxiāngyǔ jiǎnzhì [Short description of Santa] (Běijīng: Mínzú).

——(1984), 'Wǒguó Měnggǔ yǔzú yǔyán de yǔyīn duìyìng' [Sound correspondences between Mongolian languages in China], *MY* 1984/6: 1–19.

LUTAJIRGAL, Abqan (1990), 'Orcin cag-un Monggol kelen-ü $n \sim \eta$ abiyalaburi-yin tuqai' [The phonemes $n \sim \eta$ in modern Mongolian], *MKB* 1990/12: 26–34.

- LUVSANBALDAN, Haltaryn (1962), 'Èrtnij mongol hèlnij ügijn èhnij *h*-gijn tuhaj asuudald' [The problem of word-initial *h* in Old Mongolian], *StMo* 4/1–7: fasc. 4 (pp. 113–21).
 - —(1975), 'Torguud aman ajalguuny onclog' [Characteristics of the Torguud dialect], HZS 11: fasc. 5 (pp. 40–66).

——(1982), 'Regressivnaja assimiljacija nekotoryh glasnyh v halhaskom dialekte' [Regressive assimilation of some vowels in the Halh dialect], *StMo* 9(17): fasc. 9 (pp. 168–70).

- LUVSANDÈNDÈV, Amgaagijn (1958), 'Gadaad üg hèrhèn bičih tuhaj asuudald' [The problem of how to write foreign words], in A. Luvsandèndèv (ed.), *Mongol hèl bičgijn zarim asuudal*, 68–76 (Ulaanbaatar: Šinžlèh uhaan, dèèd bolovsrolyn hürèèlèn).
- LUVSANVANDAN, Šadavyn (1956), 'Mongol hèlnij hos ègšgijn tuhai' [Mongolian diphthongs], *MUISÈ* 3/1(8): 3–14.
- (1959), *Mongol hèl ajalguuny učir* [Mongolian languages and dialects] (*StMo* 1: fasc. 32).
- [----] Lubsangwangdan (1960), 'Monggol kelen-ü Qalq-a-yin aman ayalgun-u cingg-a, sula geyigülügci-yin jarim oncalig anu' [Some characteristics of strong and weak consonants in the Halh dialect of Mongolian], *MKJT* 1960/10: 1–28.
- [----] Lubsangwangdan (1961*a*), *Orcin cag-un Monggol kele* [Modern Mongolian] (Kökeqota: Öbör Monggol-un arad-un keblel-ün qoriy-a).
- (1961b), 'Mongol hèl ajalguuny učir' [Mongolian dialects], *StMo* 2/1–16 (*OUMÈ* 1 *IH* 1): fasc. 10 (pp. 143–64).
- ——(1964a), 'Mongol hèlnij halh ajalguuny balarhaj ègšgijn tuhaj asuudald' [The problem of reduced vowels in the Halh dialect of Mongolian], ŠUAM 1964/2: 88–103.
- (1964b), 'The Khalkha-Mongolian phonemic system', AOH 17: 175-85.
- ——(1966a), 'Èrtnij mongol hèlnij fonemyn sistem n'' [Phoneme system of Old Mongolian], StMo 5/2–7: fasc. 2 (pp. 3–47).
 - —(1966b), 'Mongol hèlnij ègšgijn bajrlal ba örgölt' [Vowel distribution and stress in Mongolian], *StMo* 6/9–14: fasc. 9 (pp. 3–10).
 - —(1966*c*), *Orčin cagijn mongol hèl züj* [Modern Mongolian grammar] (Ulaanbaatar: Ulsyn hèvlèlijn hèrèg èrhlèh horoo).
- (1966d), 'Orčin cagijn mongol helnij urt egšgijn garlyn tuhaj asuudald' [The problem of the origin of long vowels in modern Mongolian], HZS 4/1–4: fasc. 1/1 (pp. 5–26).

- (1966e), 'Dörvölžin üsgijn dursgalyn züjld ègšgijn urt boginyg jalgasan n'' [Differentiation between long and short vowels in 'Phags-pa script documents], HZS 4/1-4: fasc. 1/2 (pp. 27-35).
- —(1967), Orčin cagijn mongol hèlnij bütèc: avia, avialbar hoër n', nègdügèèr dèvtèr [The structure of modern Mongolian: phonetics and phonology, first book] (Ulaanbaatar: ŠUA).
- ——(1969), 'Fonemyn onolyn tuhaj asuudald' [A problem in phonological theory], HZS 6: fasc. 1 (pp. 13–52).
- ——(1975*a*), 'Èrtnij mongol hèlnij ügijn èhnij hamžin tèsrèh uruulyn čanga *p* gijgüülègč ba tüünij caašid hèrhèn huvirsan n''; 'An initial strong labial stop *p* in ancient Mongolian and its change', *HZS* 11: fasc. 1 (pp. 4–6; 7–9).
- ——(1975b), 'Orčin cagijn mongol hèlnij fonemyn bütèc' [The phoneme structure of modern Mongolian], HZS 11: fasc. 2 (pp. 10–18).
- ——(1975*c*), 'Mongol hèlnij avia zohicoh tuhaj asuudald' [The problem of assimilation in Mongolian], *StMo* 2(10): fasc. 2 (pp. 9–14).
- ——(1977), 'K probleme dolgih glasnyh srednemongol'skogo jazyka' [The problem of long vowels in Middle Mongolian], OUMÈ 3 IH 2: 94–8.
- ——(1978), 'Agšilga, ègšilgijn gijgüülègčijn tuhaj' [On the consonants y and g], *HZS* 13: fasc. 1 (pp. 3–4).
- (1980*a*), 'Mongol hèlnij agšilga *y*, ègšilgè *g* gijgüülègčijn tuhaj' [The consonants *y* and *g* in Mongolian], *StMo* 6(14): fasc. 5 (pp. 25–30). [Different from 1978]
- ——(1980b), 'Mongol hèlnij örgöltijn tuhaj asuudald' [The problem of stress in Mongolian], StMo 6(14): fasc. 4 (pp. 23–4).
- ——(1980c), 'Mongol hèlnij hos ègšig fonemyn huvilbar uruulših, ès uruulših n'' [Labialization and non-labialization of Mongolian diphthong allophones], *HZS* 14: fasc. 2 (pp. 6–8).
- ——(1982a), 'Orčin cagijn mongol hèlnij ègšgijn togtolcoo n'' [The vowel system of modern Mongolian], HZS 15: fasc. 4 (pp. 29–33).
- ——(1982b), 'Orčin cagijn mongol hèlnij gijgüülègčijn togtolcoo n'' [The consonant system of modern Mongolian], HZS 15: fasc. 5 (pp. 34–8).
- ——(1985), 'Fonemnaja struktura sovremennogo mongol'skogo jazyka' [Phoneme structure of modern Mongolian], in Sanžeev (1985), 62–6.

— and Luvsandoržijn Bold (eds.) (1985), *Mongol hèlnüüdijn har'cuulsan hèl züj* [Comparative grammar of the Mongolian languages] (Ulaanbaatar: ŠUA).

- MĂ Guóliáng and Liú Zhàoxióng (1986), 'Dōngxiāngyǔ yánjiū' [Santa studies], XMY 1986/1: 167-84, 232.
- [MANLAŽAV] Manalžav, Luvsanvandangijn (1969), 'Mongolyn nuuc tovčoony hèlnij urt ègšgijn tuhaj asuudald' [The problem of long vowels in the language of the Secret history of the Mongols], HZS 6: fasc. 4 (pp. 68–77).
- ——(1973*a*), *Fonologičeskaja sistema jazyka* Tajnoj istorii mongolov [The phonological system of the language of the *Secret history of the Mongols*], Avtoreferat kand. diss (Ulaanbaatar: ŠUA).
- [----] Manalžav (1973b), 'Nekotorye osobennosti distribucii soglasnyh fonem jazyka Sokrovennogo skazanija mongolov' [Some characteristics of the distribution of consonant phonemes in the Secret history of the Mongols], OUMÈ 2 IH 1: 282-4.
- MANSANG, Ö. (1998), 'Monggoljin aman ayalgun-daki egesig abiyan-u öbermice üjegdelün tuqai sinjilege' [A study of a characteristic phonetic feature of the vowels of the Mongoljin dialect], *ÖMÜBDS* 1998/4: 17–20.
- MARANTZ, Alec (1982), 'Re reduplication', Linguistic inquiry, 13: 435-82.

- MARTIN, Samuel E. (1961), Dagur Mongolian grammar, texts, and lexicon (Uralic and Altaic series, 4) (Bloomington: Indiana University).
- MATHEEV, Boris Vasil'evič (1956), 'Èhirit-bulagatskij govor (kratkoe soobŝenie ob osobennostjah fonetiki, leksiki)' [The Ehired-Bulagad dialect (short account for phonetic and lexical characteristics)], ZBurNII 22: 106–25.
 - (1957), 'Zvukovoj sostav èhirit-bulagatskogo govora' [The sound structure of the Ehired-Bulagad dialect], Učenye zapiski (Burjat-mongol'skij gosudarstvennyj pedagogičeskij institut imeni Dorži Banzarova), 11: 199–220 (Ulan-Udè).
 - —(1968), 'Očerki èhirit-bulagatskogo govora' [Essays on the Ehired–Bulagad dialect], *TBION* 2: 1–46.
 - (1972*a*), 'Nekotorye osobennosti modifikacii fonem v govore tunkinskih burjat' [Some characteristics of the modification of phonemes in the dialect of the Tünhen Buriads], *VBF* 1: 37–48.
 - -----(1972*b*), 'Obŝeburjatskaja fonema h v bajkalo--kudarinskom govore' [The common Buriad phoneme h in the Baigal--Hudari dialect], *VBF* 1: 49-54.
 - (1984), 'Formantnaja struktura dolgih monoftongov dobajkal'skogo burjatskogo dialekta' [The formant structure of long monophthongs of the Cis-Baikal Buriad dialect], in Nadeljaev (1984), 56–9.
- MAWKANULI, Talant (1999), *The phonology and morphology of Jungar Tuva*. Ph.D. dissertation, Indiana University, Bloomington.
- Мікамі Satoshi (1983), 'Reduplication in Mongolian', IK 17: 101-9.
- -----(1985), 'Mongorugo no jakka boin' [Reduced vowels in Mongolian], IK 19: 81-8.
- (1986), 'Mongorugo no boin' [Mongolian vowels], IK 20: 81–92.
- MIšIG, Lhagvažavyn (1961), 'Mongol Ard Ulsyn zarim nutgijn hamnigan ajalguug survalžilsan n'' [Investigation of Kamnigan dialects in some areas of the Mongolian People's Republic], *StMo* 2/1–16 (*OUMÈ 1 IH* 1): fasc. 12 (pp. 183–203).
- MITROŠKINA, Anastasija Grigor'evna (1968), 'Govor kačugskih (verholenskih) burjat' [The dialect of the Kačug (Upper Lena) Buriads], TBION 2: 47–73.
- MOHOSOEVA, Matrena (Viktorija) Matveevna (1978), 'Melodika voproso-otvetnyh predloženij v burjatskom jazyke' [Melodics of question–answer sentences in Buriad], in I. D. Buraev (ed.), *Issledovanija po burjatskoj filologii*, 76–85 (Ulan-Udè: BF).
- ——(1979), 'Melodemy nepolnyh voproso-otvetnyh predloženij v burjatskom jazyke' [Melodemes in incomplete question-answer sentences in Buriad], in Nadeljaev (1979), 146-54.
- ——(1981), Intonacija voproso-otvetnyh predloženij burjatskogo jazyka [Intonation in question–answer sentences in Buriad], Avtoreferat kand. diss (Ulan-Udè: BION).
- ——(1983), 'Rol' častotnyh harakteristik intonacii v differenciacii kommunikativnyh tipov (obŝego voprosa, otveta i sobstvenno povestvovanija) v burjatskom jazyke' [The role of frequency characteristics of intonation in the differentiation of communicative types (yes–no question, answer, and narration) in Buriad], in Nadeljaev (1983), 159–61.
- ——(1984), 'Vremennaja harakteristika v intonacionnyh strukturah voproso-otvetnyh predloženij v burjatskom jazyke' [Temporal characteristics in the intonation structures of question–answer clauses in Buriad], in Nadeljaev (1984), 153–9.
- —(1986a), 'Rol' častotnyh harakteristik intonacii v differenciacii kommunikativnyh tipov v burjatskom jazyke' [The role of frequency characteristics of intonation in the differentiation of communicative types in Buriad], in Nadeljaev (1986), 143–5.
- -----(1986b), 'Prosodija slova i frazovaja intonacija (na materiale burjatskogo jazyka)'

[Word prosody and phrase intonation (based on Buriad data)], in Rassadin (1986), 58-66.

- (1988), 'K voprosu o ritmiko-intonacionnyh osobennostjah burjatskogo jazyka' [The problem of rhythmic and intonational characteristics of Buriad], in Ubrjatova and Seljutina (eds.), 112–17.
- (V. M.) (1989), 'Ritmomelodičeskie harakteristiki burjatskogo jazyka' [Rhythmomelodic characteristics of Buriad], in Abaev, Pubaev, and Čimitdoržiev (1989), 167–9.
- Mönggöngerel (1998), *Naiman aman ayalgu* [The Naiman dialect] (Kökeqota: ÖMYSyin keblel-ün qoriy-a).
- MÖNGKEBUYAN (1989), 'Jegün Yogor kele ba Monggol kelen-ü geyigülügci-yin tokiralcag-a' [Correspondences of Shira Yugur and Mongolian consonants], *MKUJ* 1989/1: 50-64, 74.
- ——(1990), 'Ordos aman ayalgun-u abiyalaburi-yin sistèm' [The phoneme system of the Ordos dialect], MKB 90/3: 25–36.
- ——(1992a), 'Urad aman ayalgun-u abiyalaburi-yin sistèm' [The phoneme system of the Urad dialect], MKB 1992/7: 55–62, 40.
- —(1992b), 'Monggol kelen-ü aman ayalgu-nugud-un *i* abiyalaburi' [The phoneme *i* in Mongolian dialects], ÖMBYS 1992/1: 1–7.
- ——(1997), 'Qorcin aman ayalgu ba Urad aman ayalgun-u egesig abiyalaburi-yin qaricagulul: ayimag-un negüdellel ba aman ayalgu' [Comparison of Horchin and Urad vowel phonemes: clan migration and dialect], *MKB* 1997/6: 38–47.
- [-----] Mènghébǎoyīn (1998), 'Měnggǔ yǔzú yǔyán yuánchún yuányīn de yǎnbiàn guǐjī' [The line of development of round vowels in Mongolian languages], *MSYQ* 3: 56–8.
- ——(2000a), 'Monggol kelen-ü angqadagci urtu egesig-ün sudulul' [A study on primary long vowels in Mongolian], *MKB* 2000/9: 9–16.
- (2000b), 'Monggol kelen-ü egüsümel urtu egesig-ün tuqai' [Secondary long vowels in Mongolian], MKB 2000/12: 3–12.
- (2001), 'Monggol kelen-ü egesig abiyan-u ijilisil' [Vowel assimilation in Mongolian], MKB 2001/7: 14–18.
- [——] Mènghébǎoyīn (2002), 'Yuánshǐ Měnggůyǔ fǔyīn gòunǐ de jīchǔ' [The basis for the reconstruction of Proto-Mongolian consonants], *Zhōngyāng mínzú dàxué xuébào* 2002/3: 127–31.
- MONTGOMERY, Robert (1994), Buriat language policy, 19th C. 1928: a case study in Tsarist and Soviet nationality practices. Ph.D. dissertation, Indiana University.
- Мöömöö, Sürèngijn (1974), 'Ob obŝih fonetičeskih osobennostjah mongol'skogo jazyka' [General phonetic characteristics of Mongolian], *HZS* 10: fasc. 4 (pp. 32–42).
- ——(1975), 'Sistema vokalizma sovremennogo mongol'skogo jazyka' [The system of vocalism in modern Mongolian], *StMo* 2(10): fasc. 12 (pp. 184–210).
- (1976), 'K voprosu sistemy konsonantizma sovremennogo mongol'skogo jazyka' [The problem of the system of consonantism in modern Mongolian], *StMo* 3(11): fasc. 17 (pp. 273–304).
- ——(1977), Sistema fonem sovremennogo mongol'skogo jazyka [The phoneme system of modern Mongolian] (Warszawa: Wydawnictwa Uniwersytetu Warszawskiego).
- ——(1979), *Orčin cagijn mongol hèlnij avian züj* [Modern Mongolian phonetics] (Ulaanbaatar: Ulsyn hèvlelijn gazar).
- -----(1985), 'Grammatika intonacii v sovremennom mongol'skom jazyke' [The grammar of intonation in modern Mongolian], *OUMÈ 4 IH* 2: 87–90.
- -----(ed.) (1997), Orčin cagijn mongol hèl [Modern Mongolian] (Ulaanbaatar: ŠUA).
- and Jumžirijn Mönh-Amgalan (1984), *Orčin üeijn mongol hèl ajalguu* [Modern Mongolian languages and dialects] (Ulaanbaatar: ŠUA).

- MOSTAERT, Antoine (1926–7), 'Le dialecte des Mongols Urdus (Sud): étude phonétique', *Anthropos*, 21: 851–69; 22: 160–86; 25: 725 (corrections).
 - ----(1931), 'The Mongols of Kansu and their language', *Bulletin of the Catholic University of Peking*, 8: 75–89.
 - (1941–4), *Dictionnaire ordos (Monumenta Serica monograph*, 5). 3 vols (1941, 1942, 1944) (Peking: Catholic University).
- and Albert de Smedt (1929–31), 'Le dialecte Monguor parlé par les Mongols du Kansu occidental, lère partie: phonétique', *Anthropos*, 24: 145–65, 801–15 (1929); 25: 657–69, 961–73 (1930); 26: 253–4 (1931),
- MUDRAK, Oleg Alekseevič (1986), 'O fonologičeskoj sisteme mongorskogo jazyka i ee transkribirovanie' [The phonological system of Monguor and its transcription], *PIAC* 29: 77–9.
- MUNIEV, Bembja Džalykovič (1977), Kalmycko-russkij slovar'; Hal'mg-ors tol' [Kalmuck-Russian dictionary] (Moskva: Russkij jazyk).
- MURAYAMA Shichirô (1960), 'Chûki Môkogo no sho mondai' [Some problems of Middle Mongolian], *Tôyô gakuhô*, 43/1: 65–100; 43/2: 43–62.
- ——(1970), 'Die Entwicklung der Theorie von den primären langen Vokalen im Mongolischen', in L. Ligeti (ed.), *Mongolian studies*, 359–70 (Budapest: Akadémiai kiadó).
- MUŠAEV, Vladimir Naranovič (1987), 'Osobennosti ritmomelodiki vvodnyh i vključennyh obrazovanij v kalmyckom jazyke' [Characteristics of the rhythmomelodics of inserted and included structures in Kalmuck], in *Issledovanija po kalmyckomu jazykoznaniju*, 48–57 (Èlista: Kalmyckij gosudarstvennyj universitet).
- NADAMID (1978), 'Silagun Köke qosigun-u aman ayalgun-u abiyan-u sistèm' [The sound system of the dialect of Shuluun Höh banner], *ÖMYSS* 1978/3: 138–46.
 - -----(1984), 'Düngsiyang kelen-ü egesig-ün tuqai' [Santa vowels], MKB 1984/2: 33-46.
- [-----] Nàdémùdé (1986), 'Měnggǔyǔ Cháhā'ěr tǔyǔ de yuányīn hé fǔyīn' [Vowels and consonants in the Chahar dialect of Mongolian], *MY* 1986/5: 47–55.
- ——(1993), 'Caqar aman ayalgun-u jarim geyigülügci qarilcan gadagurcilaqu asagudaldu' [The problem of mutual exclusion of some consonants in the Chahar dialect], *MKB* 1993/7: 9–15.
- ——(1998), 'Caqar aman ayalgun-u cingg-a i [1] egesig' [The tense vowel i [1] in the Chahar dialect], MSYQ 3: 149–50.
- NADELJAEV, Vladimir Mihajlovič (1957), 'Sostav fonem v zvukovoj sisteme sovremennogo mongol'skogo jazyka' [The phoneme inventory in the sound system of modern Mongolian], VLU 8: 123–31.
- ——(ed.) (1979), Fonetika sibirskih jazykov [The phonetics of Siberian languages] (Novosibirsk: IIFF).
 - ——(ed.) (1980), Fonetičeskie issledovanija po sibirskim jazykam [Phonetic investigations of Siberian languages] (Novosibirsk: IIFF).
- ——(ed.) (1983), *Sibirskij fonetičeskij sbornik* [Siberian phonetic collection] (Novosibirsk: IIFF).
- ——(ed.) (1984), Issledovanija zvukovyh sistem jazykov Sibiri [Investigations of the sound systems of Siberian languages] (Novosibirsk: Nauka).
 - (ed.) (1986), *Fonetika jazykov Sibiri i sopredel'nyh regionov* [Phonetics of the languages of Siberia and adjacent areas] (Novosibirsk: Nauka).
- ——(1987), 'Sovremennyj mongol'skij jazyk' [Modern Mongolian], in E. I. Ubrjatova (ed.), Jazyki Sibiri i Mongolii, 3–74 (Novosibirsk: IIFF).
- NADMID, Žamsranžavyn and Žamsranžavyn Sanžaa (1993), Orčin cagijn mongol hèlnij avia züj [Modern Mongolian phonetics] (Ulaanbaatar).

- NAGY, Lajos J. (1960), 'A contribution to the phonology of an unknown East-Mongolian dialect', *AOH* 10: 269–94.
- NAMTSARAI and B. Qaserdeni (1983), *Dagur kele Monggol kelen-ü qaricagulul* [Dagur compared with Mongolian] (Kökeqota: Öbör Monggol-un arad-un keblel-ün qoriy-a).
- NANGRUB (1981), 'Monggol kelen-ü abiyan-u kögjil-ün teüke-yin tobci' [Outline history of the sound development of Mongolian], *MKUJ* 1981/3: 1–30.
- NARANBATU, Ö. (1994), 'Urad aman ayalgun-u tuqai' [The Urad dialect], ÖMBYS 1994/2: 28-35.

—(1999), 'Urad Ordos Caqar aman ayalgun-u jarim egesig-ün tuqai' [Some vowels in the Urad, Ordos, and Chahar dialects], *MKB* 1999/12: 5–8.

- NASUNBAYAR, Borjigin (1961), 'Düngsiyang kelen-ü temdeglel' [Description of Santa], *StMo* 3/1–5: fasc. 3 (pp. 53–106).
 - ——(1981), 'Düngsiyang kelen-ü üge-yin ergülte' [Word stress in Santa], MKUJ 1981/1: 10–15.
- [-----] Nàsēnbǎi (1987a), 'Dōngxiāngyŭ' [Santa], ZSMY 630-43.
- [-----] Nàsēnbǎi (1987b), 'Bǎo'ānyǔ' [Bonan], ZSMY 644-54.
- ——(1987c), 'Jalayid aman yariyan-u kedün uridugurki egesig' [Some fronted vowels in the Jalaid dialect], MKUJ 1987/1: 4–13, 26.
- B. Qaserdeni, W. Secen, Cogtu, Dawadagba, B. Türgen, and Ö. Naranbatu (1982), *Orcin cag-un Monggol kele* [Modern Mongolian] (Kökeqota: ÖMSK). [Second edition: Qaserdeni et al. 1996]
- NASUNÖLJEI and Nabcingerel (1997), 'Urtu egesig-ün bütüce-deki abiyan-u ijilsil-i ajiglaqu ni' [Observations on assimilation in the structure of long vowels], *MKB* 1997/10: 40–1.
- NAVROCKU, N. (1840), 'Kratkija zaměčanija o različii meždu kalmyckim i mongol'skim jazykami' [Short remarks on the differences between Kalmuck and Mongolian], *Učenyja* zapiski Imperatorskago Kazanskago universiteta, 1840/3: 160–76.
- NOLMAJAB, S. (1986), 'Qaracin Tümed aman ayalgun-u x geyigülügci-yin tuqai' [The consonant x in the Harchin Tümed dialect], *MKB* 1986/10: 14–24.
- ——(1991), 'Qaracin Tümed aman ayalgun-u x geyigülügci-yin qubiral kögjil-ün jim-e qauli-yin tuqai' [Development patterns for the consonant x in the Harchin Tümd dialect], ÖMYSS 1991/4: 20–9.
- NOMINTUYAG-A (1992), 'Üjümücin aman ayalgun-u egesig abiyalaburi' [Vowel phonemes of the Üjemchin dialect], *ÖMBYS* 1992/2: 43–55.
- NOMURA Masayoshi (1940), 'Môkogo no hôgen' [Mongolian dialects], *Manshûkokugo*, 6: 11–20; 8: 10–15.
- ——(1940–1), 'Môkogo Harachin Uki Ôfu hôgen no tan boin' [Short vowels in the dialect of Wángfǔ (王府), Harchin Right banner], *Minzokugaku nenpô* (Annual report of the Institute of ethnology), 3: 313–41.
- ——(1941a), 'Môkogo Harachin Chûki hôgen ni kansuru jakkan no oboegaki: Uchi-Môko hôgen kenkyû shiryô' [Remarks on the Harchin Central banner dialect of Mongolian: data for research on Inner Mongolian dialects], *GK* 9: 80–91.
- ——(1941b), 'Môkogo no kôgo toku ni Uchi-Môko no kôgo ni tsuite: Môkogo hikaku on'inron e no jakkan no kiyo' [On colloquial Mongolian, especially colloquial Inner Mongolian: some contributions to Mongolian comparative phonology], *Minzokugaku kenkyû*, 7/2: 34–75.
- —(1950), 'Remarks on the diphthong [wa] in the Kharachin dialect of the Mongol language', GK 16: 126–41.
- --(1951), 'Supplementary notes and additions to "Remarks on the diphthong [wa] in the Kharachin dialect of the Mongol language", *GK* 17–18: 149–55.

- NOMURA Masayoshi (1953), 'Arutai shogo ni okeru isshu no boin kôtai ni tsuite' [Vowel alternation in the Altaic languages], *NagoKR* 4: 25-44.
 - —(1957), 'On some phonological developments in the Kharachin dialect', in *Studia Altaica: Festschrift für Nikolaus Poppe zum 60. Geburtstag am 8. August 1957*, 132–6 (Wiesbaden: Harrassowitz).
 - (1959a), 'Monguoru hôgen no chôboin to Gen Môkogo ni okeru chôboin sonzai no kanôsei ni tsuite' [Long vowels in Monguor and the possibility that long vowels exist in Proto-Mongolian], in Nagoya Daigaku bungakubu jisshûnen kinen ronshû, 621–32. [Cf. Nomura 1979]
 - (1959b), 'Materials for the historical phonology of the Mongol language', Memoirs of the research department of the Toyo Bunko (The oriental library), 18: 59–89.
- ——(1960), 'Môkogo on'inshi kara mita Harachin hôgen to Dagûru hôgen to no heikôsei ni tsuite' [The parallelism of Harchin and Dagur dialects seen from the phonological history of Mongolian], *NagoKR* 22: 1–16.
- (1965–73), 'Gen Môkogo no boin taikei ni tsuite no kenkyû' [Studies of the Proto-Mongolian vowel system], NagoKR 37: 1–40 (1965) (I); 40: 1–10 (1966) (II); 52: 1–30 (1971) (IV); 55: 5–18 (1972) (V); 58: 5–33 (1973) (VI); Nagoya Daigaku bungakubu nijisshûnen kinen ronshû, 483–89 (1968) (III). [Cf. Nomura 1979]
- ——(1971), 'Môko shogo ni okeru isshu no boin kôtai *ä* ~ *i* ni tsuite' [The vowel alternation *ä* ~ *i* in Mongolian languages], in *Kindaichi* [金田一] *hakase beiju kinen ronshû*, 892–87 (Tôkyô: Sanseidô). [Cf. Nomura 1979]
- (1972), 'Kyôtsû kigo saikô no tejun ni okeru on'inronteki kasetsu no settei ni tsuite: Gen Môkogo ni sonzaishita chôboin no suimetsu to sono sho jôken' [The establishment of phonological hypotheses in the process of reconstruction of a proto-language: the loss of long vowels in Proto-Mongolian and its conditions], in *Gendai gengogaku: Hattori Shirô sensei teinen taikan kinen ronbun shû*, 589–98 (Tôkyô: Sanseidô).
- —(1976), 'Shuju na jôken no suiteisareru Monguoru hôgen no chôboin: Gen Môkogo no boin taikei ni tsuite no kenkyû dai-shi chô dai-shi' [Long vowels in Monguor assumed under different conditions: chapter IV §4 of Studies of the Proto-Mongolian vowel system], NagoKR 67: 5–10. [Cf. Nomura 1979]
- ——(1979), Gen Môkogo no boin taikei ni tsuite no kenkyû: Môkogo hikaku on'inron kenkyû [Studies of the Proto-Mongolian vowel system: research on Mongolian comparative phonology] (Nagoya: Saika shorin). [Cf. Nomura 1959a; 1965–73; 1971; 1976]
- Norлл, Tserin-ü (1991), 'Orcin üy-e-yin Monggol kelen-ü egesig-ün üiledül kiged čidabki' [The function and potential of modern Mongolian vowels], *MSYQ* 2: 1073–80.
- ——(1992a), 'Orcin üy-e-yin Monggol kelen-ü egesig-ün ilgagdaqu sinji-yin ucir' [Distinctive features of modern Mongolian vowels], *MKUJ* 1992/1: 37–49, 127.
- ——(1992b), 'Orcin üy-e-yin Monggol kelen-ü egesig-ün aguu uyitan-u jokical: Caqar aman ayalgun-u matèriyal-du ündüsülebe' [Wide–narrow vowel harmony in modern Mongolian: based on material from the Chahar dialect], *MKB* 1992/1: 11–21.
- ——(1998), 'Monggol kelen-ü üy-e bütükü yoson-u ularil: Monggol kelen-ü geyigülügci ugsaraqu jim-e' [Change of syllable structure in Mongolian: consonant combination rules], *MKB* 1998/9: 3–8.
- NUGTEREN, Hans (1997), 'On the classification of the peripheral Mongolic languages', *PIAC* 39: 207–16.
 - —(2003), 'Shira Yughur', in Janhunen (2003*f*), 265–85.
- OBATA Jûichi and [Teshima] Tesima Takehiko (1934), 'Physico-phonetical studies of the Mongolian language', *Proceedings of the Imperial Academy (of Japan)*, 10: 636–9.
 - -----(1935), 'Môkogo no butsuri onseigakuteki kenkyû' [An acoustic study of Mon-

274

golian], Nippon sugaku butsuri gaku kaishi (Proceedings of the Physico-Mathematical Society of Japan), 9/1: 1–17.

ODDEN, David (1977), 'Overspecification and variables in phonology', *Linguistic analysis*, 3: 177–96.

—(1979), 'Principles of stress assignment: a crosslinguistic view', *Studies in the linguistic sciences*, 9/1: 157–75.

----(1980), 'The irrelevancy of the relevancy condition: evidence for the feature specification constraint', *Linguistic analysis*, 6: 261–304.

- ÖLJEI (1984), 'Qorcin aman ayalgun-u jarim oncalig-un tuqai tursin ögülekü ni' [A note on some characteristics of the Horchin dialect], *MKB* 1984/3: 79–83.
- ÖLJEYIBÜRIN (1992), 'Bayad aman ayalgun-u abiyan-u sinjilel' [A study of the phonetics of the Bayad dialect], *ÖMBYS* 1992/2: 1–18.
- ——(1994), 'Jalayid aman ayalgun-u kele matariqu geyigülügci' [Retroflex consonants in the Jalaid dialect], ÖMBYS 1994/4: 64–71.
- —(1996), 'Monggol aman ayalgun-u qoos egesig-ün qandusi' [The trend of Written Mongolian diphthongs], *MKUJ* 1996/4: 85–90.
- (2000), 'Sönid aman ayalgun-u egesig abiyalaburi-yin sistèm' [The vowel phoneme system of the Sönid dialect], MKB 2000/10: 16–21.
- ——(2001), 'Sönid aman ayalgun-u geyigülügci abiyalaburi-yin sistèm' [The consonant phoneme system of the Sönid dialect], *MKB* 2001/1: 16–23.
- ÖLJEYITOGTAQU (1991), 'Naiman aman ayalgun-u kedün geyigülügci abiyan-u ondoosiqu asagudal ni' [The question of dissimilation of some consonants in the Naiman dialect], *MKUJ* 1991/3: 91–5.
- Омакаеva, Èllara Uljaevna (ed.) (2000), *Halimyg kèlnä čikär bičilhenä zokal bolyn toli;* Orfografičeskie pravila i orfografičeskij slovar' kalmyckogo jazyka [Orthographic rules and orthographic dictionary of Kalmuck] (Èleste: Halimyg dègter harhači).
- ORLOV, Aleksandr Matveevič (1878), *Grammatika mongolo-burjatskago razgovornago jazyka* [Grammar of the Buriad Mongolian colloquial language] (Kazan': Izdanie pravoslavnago missionerskago obŝestva).
- ORLOVSKAJA, Marija Nikolaevna (1997), 'Staropis'mennyj mongol'skij jazyk' [Old Written Mongolian], in Jarceva (1997), 18–29.
- OSADA Natsuki (1952), 'Jûni seiki ni okeru Môko sho buzoku no gengo' [Languages of the Mongolian tribes in the 12th century], *Tôhôgaku*, 5: 42–55.
- ŌUYÁNG Juéyà and Zhōu Yàowén (eds.) (1994), *Zhōngguó shǎoshù mínzú yǔyán shǐyòng qíngkuàng* [The situation of language use among the minority peoples of China] (Běijīng: Zhōngguó Zàngxué).
- ÔYABU Shôtarô (1920), Nihongo to Môkogo [Japanese and Mongolian] (Dairen: Manshû hibi shinbunsha).
- OYUNCECEG (1986), 'Degedü Monggol-un Kürlüg aman ayalgun-u egesig abiyan-u oncalig' [Characteristics of the vowels of the Hürleg dialect of the Upper Mongols], *MKB* 1986/9: 54–61.
- ——(1987), 'Degedü Monggol-un aman ayalgun-daki öbermice geyigülügcid' [Characteristic consonants of the dialect of the Upper Mongols], *MKUJ* 1987/2: 34–41.
- Ozawa Shigeo (1997), 'Dundad zuuny mongol hèlnij nutgijn ajalguuny tuhaj' [Dialects of Middle Mongolian], *Mongolica*, 6(27) (*OUMÈ 6 IH* 2): 44–5.
- PAVLOV, Dordži Antonovič (1963a), 'O nekotoryh fonetičeskih osobennostjah v kalmyckom jazyke' [Some phonetic characteristics of Kalmuck], *Trudy Dvadcat' pjatogo meždunarodnogo kongressa vostokovedov*, 3: 417–22 (Moskva: Izd. vostočnoj literatury).

- PAVLOV, Dordži Antonovič (1963*b*), *Sostav i klassifikacija fonem kalmyckogo jazyka* [The structure and classification of Kalmuck phonemes] (Èlista: Kalmgosizdat).
 - (1964), 'Ural'skij govor i nekotorye voprosy razvitija fonetičeskoj sistemy kalmyckogo jazyka' [The Ural dialect and some problems in the development of the phonetic system of Kalmuck], *ZKEl* 3: 40–68.
 - —(1968), *Sovremennyj kalmyckij jazyk: fonetika i grafika* [Modern Kalmuck: phonetics and script] (Èlista: KKI).
 - —(1969), 'Nekotorye voprosy razvitija fonetiki kalmyckogo jazyka po Todo bičig' [Some problems in the development of Kalmuck phonetics according to the Clear script], *ZKEl* 7: 239–59.
 - (1974), 'Razvitie diftongov v monoftongi v kalmyckom jazyke' [Development of diphthongs to monophthongs in Kalmuck], in N. A. Baskakov (ed.), *Issledovanija po vostočnoj filologii k semidesjatiletiju professora G. D. Sanžeeva*, 218–25 (Moskva: Nauka).
 - —(1982), 'O funkcii reducirovannyh glasnyh v kalmyckom jazyke' [The function of reduced vowels in Kalmuck], *TPV* 125–31.
 - ——(1983), *Fonetika sovremennogo kalmyckogo jazyka* [Modern Kalmuck phonetics] (Èlista: KKI).
- [-----] Pavla Dorž (1990), *Har Holyn hal'mgud boln tednä keln* [The Karakol Kalmucks and their language] (Èlst: Hal'mg gosudarstvenn universitet).
 - and Tamara Sarangovna Esenova (1984), 'Variativnost' dlitel'nosti zvukov v raznyh intonacionnyh tipah predloženij (èksperimental'no-fonetičeskoe issledovanie)' [Variability of the duration of sounds in different sentence intonation types (experimental phonetic investigation)], in D. A. Pavlov (ed.), *Voprosy kalmyckoj filologii*, 21–8 (Èlista: Kalmyckij gosudarstvennyj universitet).
 - (1986), 'Fonetičeskaja harakteristika i fonologičeskij status glasnyh kalmyckogo i mongol'skogo jazykov' [Phonetic characterization and phonological status of Kalmuck and Mongolian vowels], in Nadeljaev (1986), 49–52.
- PAVLOVA, Elena Semenovna (1987*a*), 'Melodičeskaja harakteristika pobuditel'nyh vyskazyvanij v burjatskom jazyke' [Melodic characteristics of imperative utterances in Buriad], in Rassadin (1987*a*), 72–82.
 - —(1987b), 'Vzaimodejstvie grammatičeskih i intonacionnyh sredstv pri vyraženii pobuždenija v burjatskom jazyke' [The interaction of grammatical and intonational means for expressing imperatives in Buriad], in Rassadin (1987b), 72–6.
 - ——(1988), 'Intonacija perečislenija v burjatskom jazyke' [The intonation of enumeration in Buriad], in Ubrjatova and Seljutina (1988), 126–9.
- ——(1989), 'Intonacija nezaveršennosti v burjatskom jazyke' [Continuation intonation in Buriad], in Šagdarov (1989), 113–21.
- (1991), 'Sopostavlenie nejtral'noj i modal'noj realizacii intonacionnyh edinic v burjatskom jazyke' [Comparison of neutral and modal realizations of intonational units in Buriad], in Šagdarov (1991), 135–43.
- PÈLJEI, J. (1992), 'Monggol barimjiy-a abiyan-u n, η geyigülügči-yin asagudal-du' [The problem of the consonants n, η in standard Mongolian pronunciation], *MKUJ* 1992/2: 30-43, 55.
 - (2001), 'Barimjiy-a abiyan-u abiyalaburi-yin sistèm-ün asagudal-du' [The problem of the phoneme system of the standard spoken language], *MKB* 2001/9: 4–16; 2001/10: 9-15.
- PELLIOT, Paul (1925), 'Les mots à *h* initiale, aujourd'hui amuie, dans le mongol des XIIIe et XIVe siècles', *Journal Asiatique*, 206: 193–263.

276

----(1944), 'Les formes avec et sans q- (k-) initial en turc et en mongol', *T'oung Pao*, 37: 73–101.

- PJURBEEV, Grigorij Cerenovič (2000), 'Dejstvie intonacionnogo faktora v predloženii mongol'skih jazykov' [The function of the intonation factor in the sentence of Mongolian languages], in *Vladimircovskie čtenija: doklady i tezisy*, 4: 87–90 (Moskva: RAN).
 ——(2001a), 'Kalmyckij jazyk' [Kalmuck], in Jarceva (2001), 29–41.
- (2001b), 'Mongol'skie jazyki' [Mongolian languages], in Jarceva (2001), 291–7.
- PODGORBUNSKIJ, Innokentij Aleksandrovič (1910), 'Něskol'ko zamětok po fonetikě burjatskago jazyka v svjazi s voprosom o transkripcii burjatskih tekstov' [Some remarks on Buriad phonetics in relation to the problem of transcription of Buriad texts], *Izvěstija Vostočno-Sibirskago otděla Imperatorskago Russkago geografičeskago obŝestva*, 41: 10-39.
- POPOV, Aleksandr Vasil'evič (1847), *Grammatika kalmyckago jazyka* [Kalmuck grammar] (Kazan': Universitetskaja tipografija).
- POPPE, Nikolaj Nikolaevič (1927–8), 'Das mongolische Sprachmaterial einer Leidener Handschrift', *Izvestija Akademii nauk SSSR*, 1927: 1009–40, 1251–74; 1928: 55–80.
 - ——(1930a), Dagurskoe narečie [The Dagur dialect] (Materialy Komissii po issledovaniju Mongol'skoj i Tannu-Tuvinskoj narodnyh respublik i Burjat-Mongol'skoj ASSR, 6) (Leningrad: AN SSSR).
 - —(1930b), Alarskij govor, časť pervaja [The Alair dialect, first part] (Materialy Komissii po issledovaniju Mongol'skoj i Tuvinskoj narodnyh respublik i Burjat-Mongol'skoj ASSR, 11) (Leningrad: AN SSSR).
 - —(1931), Praktičeskij učebnik mongol'skogo razgovornogo jazyka (halhaskoe narečie) [Practical textbook of spoken Mongolian (Halh dialect)] (Leningrad: Izd. Leningradskogo vostočnogo instituta im. A. S. Enukidze).
- (1932a), 'Skizze der Phonetik des Bargu-burjätischen', Asia Major, 7: 307–78.
- ——(1932b), 'K latinizacii mongol'skogo alfavita i perehodu na novyj literaturnyj jazyk' [The Latinization of the Mongolian alphabet and the transition to a new literary language], Zapiski Instituta vostokovedenija AN SSSR, 1: 19–34.
- ——(1932c), Zametki o govore aginskih burjat [Remarks on the dialect of the Aga Buriads] (Trudy Mongol'skoj komissii AN SSSR, 8) (Leningrad: AN SSSR).
- -----(1933*a*), *Burjat-mongol'skoe jazykoznanie* [Buriad Mongolian linguistics] (Leningrad: AN SSSR).
- (1933b), 'Die burjätischen Mundarten', JSFOu 67: 331–5.
- (1933*c*), 'Über einen Vokalwechsel in Mongolischen', *Ungarische Jahrbücher*, 13/1–2: 112–22.
- —(1934a), Jazyk i kolhoznaja poèzija burjat-mongolov Selenginskogo ajmaka [The language and kolkhoz poetry of the Buriad Mongols in the Selenga district] (Obrazcy narodnoj slovesnosti mongolov, 4) (Leningrad: AN SSSR).
- (1934*b*), 'Über die Sprache der Daguren', *Asia Major*, 10: 1–32, 183–220.
- ——(1936), *Stroj halha-mongol'skogo jazyka* [The structure of Halh Mongolian] (Leningrad: Izd. Leningradskogo naučno-issledovatel'skogo instituta jazykoznanija).
- ——(1937), Grammatika pis'menno-mongol'skogo jazyka [Grammar of written Mongolian] (Moskva: AN SSSR).
 - —(1938), *Grammatika burjat-mongol'skogo jazyka* [Buriad Mongolian grammar] (Moskva: AN SSSR).
- (1938–9), Mongol'skij slovar' Mukaddimat al-Adab [The Mongolian dictionary Muqaddimat al-adab] (Trudy Instituta vostokovedenija, 14), čast' 1–2 (1938); čast' 3 (index, 1939) (Moskva: AN SSSR).

- POPPE, Nikolaj Nikolaevič (1941*a*), 'Izučenie govorov burjat-mongol'skogo jazyka' [Investigations of Buriad Mongolian dialects], *Zapiski GIJaLI*, 1941/3-4: 3-12.
 - ——(1941*b*), *Kvadratnaja pis'mennost'* [The square script] (Moskva: AN SSSR). [English version 1957]
 - (1950), 'The groups **uya* and **üge* in Mongol languages', *Studia orientalia*, 14: fasc. 8 (pp. 1–15).
- -----(1951a), Khalkha-mongolische Grammatik (Wiesbaden: Steiner).
- (1951b), 'Remarks on the vocalism of the second syllable in Mongolian', *Harvard journal of Asiatic studies*, 14: 189–207.
- -----(1954a), Grammar of written Mongolian (Wiesbaden: Harrassowitz).
- -----(1954b), 'Über einen Konsonantenschwund im Mongolischen', UAJb 26: 240-2.
- -----(1955), Introduction to Mongolian comparative studies (SUST 110) (Helsinki: SUS).
- -----(1956a), 'On the so-called breaking of *i in Mongolian', UAJb 28: 43-8.
- (1956b), 'The Mongolian affricates $*\check{c}$ und $*\check{z}$ ', CAJ 2: 204–15.
 - —(1957), The Mongolian monuments in hP'ags-pa script (Göttinger Asiatische Forschungen, 8), 2nd ed. [of Poppe 1941b], trans. and ed. by John R. Krueger (Wiesbaden: Harrassowitz).
 - —(1958), 'Einige Lautgesetze und ihre Bedeutung zur Frage der mongolisch-türkischen Sprachbeziehungen', *UAJb* 30: 93–7.
- (1959), 'On the velar stops in intervocalic position in Mongolian', UAJb 31: 270–3.
- ——(1960a), Vergleichende Grammatik der altaischen Sprachen, Teil 1: Vergleichende Lautlehre (Wiesbaden: Harrassowitz).
- ——(1960b), Buriat grammar (Uralic and Altaic series, 2) (Bloomington: Indiana University).
- ——(1961), 'Preliminary remarks on the Zirni Manuscript', in Sh. Iwamura, *The Zirni Manuscript: a Persian–Mongolian glossary and grammar*, v–ix (Kyôto: Kyôto University).
- (1962), 'The primary long vowels in Mongolian', JSFOu 63: fasc. 2 (pp. 3–19).
- (1963), 'On some cases of assimilation of **i* in Mongolian', *CAJ* 8: 215–18.
- (1964a), 'Das Mittelmongolische', HOM 96-103.
- (1964b), 'Die burjätische Sprache', HOM 108-14.
- -----(1964c), 'Das Ordossische', HOM 134-6.
- ——(1964*d*), 'Die dagurische Sprache', *HOM* 137–42.
- (1965), Introduction to Altaic linguistics (Wiesbaden: Harrassowitz).
- (1966), 'On some ancient Mongolian loan-words in Tungus', CAJ 11: 187-98.
- -----(1967), 'On the long vowels in common Mongolian', JSFOu 68: fasc. 4 (pp 1-31).
- ——(1968), 'Über einige Vokalentsprechungen in mongolischen Lehnwörtern im Tuvinischen', ZDMG 118: 113–23.
- (1969*a*), 'On the initial *h* in the Yüan-ch'ao pi-shih', BIHP 39/2: 267–75.
- (1969*b*), 'On some vowel correspondences in Mongolian loan-words in Turkic', *CAJ* 13: 207–14.
- ——(1969c), 'Bemerkungen zum vorklassischen Schriftmongolisch (anläßlich des Buches von M. Weiers [1969])', ZASB 3: 105–28.
- -----(1970), Mongolian language handbook (Washington: Center for applied linguistics).
- -----(1971), 'Über einige Besonderheiten des Tsongol-Dialektes', ZASB 5: 145-55.
- (1972*a*), 'On some Mongolian loan words in Evenki', *CAJ* 16: 95–103.
- (1972b), 'Introduction', in N. Poppe, Mongolica (Farnborough: Gregg International).
- (1975*a*), 'Altaic linguistics: an overview', *Gengo no kagaku* (*Sciences of language*), 6: 130–86.

— (1975*b*), 'On some cases of fusion and vowel alternation in the Altaic languages', *CAJ* 19: 307–22.

POSCH, Udo (1957–8), 'The syllabic groups and their further development in the written Oirat language, Part 1', CAJ 3: 206–19.

——(1964*a*), 'Khalkha und Verwandtes', *HOM* 115–33.

(1964b), 'Das Kalmükische und verwandte Dialekte', HOM 200-26.

POZDNEEV, Aleksej Matveevič (1879), Review of Orlov (1878), Žurnal ministerstva narodnago prosvěšenija, dekabr' 1879, časť 206, otdel 1–2: 170–208.

PRINCE, Alan (1983), 'Relating to the grid', Linguistic inquiry, 14: 19–100.

PRITSAK, Omeljan (1964), 'Das Mogholische', HOM 159-84.

PULLEYBLANK, Edwin G. (1984), *Middle Chinese: a study in historical phonology* (Vancouver: UBC Press).

——(1991), Lexicon of reconstructed pronunciation in Early Middle Chinese, Late Middle Chinese, and Early Mandarin (Vancouver: UBC Press).

PÜRBÜ (1981), 'Erten-ü Monggol kelen-ü abiyan-u sistem-ün qubisul ularigsan ündüsün siltagan-u tuqai' [On the basic reasons for the change of the Classical Mongolian sound system], *MKUJ* 1981/1: 28–38.

- ——(1987), 'Bicig-ün kele ba Sin-e Bargu aman ayalgun-u qaricagulul' [Comparison of the written language and the New Bargu dialect], MKSÖ 282–333.
- QADACILAGU (1984), 'Sönid aman ayalgun-u jarim abiy-a-yi bicig-ün kele-tei qaricagulqu ni' [Comparison of some sounds in the Sönid dialect with the written language], *MKB* 1984/5: 83–92.
- [QAGANCILAGU, Qarnud] Hākānchŭlún (1970), Měnggů yůwén [Mongolian language and script], ([Táiwān]: Zhôngguó biānjiāng lìshǐ yǔwén xuéhuì).

[——] Hākānchǔlún and [Kögjimtei] Húgéjīntái (1979), *Dáhū'ěr fāngyán yǔ Mǎn, Měng yǔ zhī yìtóng bǐjiào* [Comparison of Dagur with Mongolian and Manchu] ([Táiběi]).

QAI Jüng-yan (1987), 'Barimjiy-a abiy-a bolon Monggoljin aman ayalgun-u egesig abiyalaburi-yin qaricagulul' [Comparison of the vowel phonemes in the standard language and the Monggoljin dialect], *MKUJ* 1987/4: 24–7.

QASBAGAN-A (1988), 'Dumdadu jagun-u Monggol kelen-ü egesig urugul-iyar ijilsikü tuqai' [Vowel rounding assimilation in Middle Mongolian], *MKUJ* 1988/1: 59–63.

——(1990), 'Caqar aman ayalgun-u abiyalaburi-yin sistèm-ün jarim oncalig' [Some characteristics of the phoneme system of the Chahar dialect], ÖMBYS 1990/4: 91–103.

——(1996), *Dumdadu jagun-u Monggol kelen-ü sudulul* [Middle Mongolian studies] (Kökeqota: ÖMSK).

——(1998), 'Monggol töröl-ün kelen-ü üge-yin ekin-ü neyilemel geyigülügci-yin tuqai' [Word initial consonant clusters in Mongolian languages], MKUJ 1998/6: 74–9.

——(2000*a*), 'Monggol töröl-ün keled-ün el-e kele ayalgun-daki *b* geyigülügci-yin tuqai' [The consonant *b* in the Mongolian languages and dialects], *MKUJ* 2000/3: 4–11, 74.

[——] Hāsībāgēn (2000*b*), 'Guānyú Měnggǔyǔ fēi dìyī yīnjié yuányīn ruòhuà hé tuōluò guòchéng zhōng de *ð*-huà xiànxiàng' [Reduction and change to *ð* in the deletion process of Mongolian vowels in non-initial syllables], *NDX* 2000/3: 53–6.

— (2002), 'Monggol kelen-ü cingg-a i egesig-i sinjilekü ni' [A study of the tense vowel i in Mongolian], *MKB* 2002/3: 21–5.

QASBAGATUR (1986), Tůzúyů cíhuì [Monguor vocabulary] (MTKASC 014).

[—] Hasbaatar (1999), 'Mongol hèlnij ügijn èhnij ègšig gèègdèh tuhaj' [The loss of initial vowels in Mongolian], in V. I. Rassadin (ed.), *Istorija razvitija mongol'skih jazykov*, 84–91 (Ulan-Udè: BNC).

^{—(1976), &#}x27;Ancient Mongolian', in Heissig et al. (1976), 463–78.

- QASBAGATUR (2002), 'Qaracin aman ayalgun-u abiyan-u jarim oncalig' [Some phonetic characteristics of the Harchin dialect], *ÖMYSS* 2002/4: 1–10.
- QASERDENI, B. (1958), 'Kelen-ü abiyan-u gurban sinji ba Monggol kelen-ü cingg-a sula geyigülügci-yin tuqai' [Three features of speech sounds and the strong and weak consonants in Mongolian], *MTKB* 1958/12: 14–19.

——(1959), 'Ordos-un aman ayalgun-u oncalig' [Characteristics of the Ordos dialect], MKJT 1959/1: 30-42; 1959/4: 3-14.

——(1980*a*), 'Monggol kelen-ü uqurin ijilsikü egesig jokicaqu yoson-u tuqai' [Regressive vowel harmony in Mongolian], *MKB* 1980/5: 49–54, 48.

- ——(1980b), 'Caqar aman ayalgun-u abiyan-u togtaca' [The sound system of the Chahar dialect], ÖMBDS 1980/3: 1–15.
- [——] Hāsī'è'ěrdūn (1981*a*), 'Měnggǔyǔ È'ěrduōsī fāngyán yǔyīn tèdiǎn' [Phonetic characteristics of the Ordos dialect of Mongolian], in *Mínzú yǔwén lùn jí*, 443–58 (Běijīng: Zhōngguó shèhuì kēxué).
 - (1981*b*), 'Monggol bicig-ün kele ba nutug-un ayalgu aman ayalgun-u bogoni egesigün qaricagulul' [Comparison of short vowels in written Mongolian and local dialects], *MKUJ* 1981/1: 16–27.
 - -----(1985), 'Barimjiy-a abiy-a bolon Ordos ayalgun-u abiyan-u qaricagulul' [Comparison of the standard pronunciation and the Ordos dialect], *MKUJ* 1985/3: 4–17.
 - ——(1990), 'Su Bèi Monggolcud-un yariyan-u abiy-a bolon barimjiy-a-luga-yin qaricagulul' [Comparison between the Subei Mongolian dialect and standard pronunciation], *MKB* 1990/9: 36–46.
 - (1994), 'Kitad üsüg-iyer galiglagsan Monggol-un niguca tobciyan-aca e egesig-ün abiyalaca-yi sinjikü ni' [An inquiry into the the pronunciation of the vowel e from its transcription with Chinese characters in the Secret history of the Mongols], MKUJ 1994/5: 30-3, 88.
 - —(1995), 'Monggol kelen-ü teüken yabuca-daki abiy-a qasugdagsan üjegdel-ün tuqai: kelen-ü dökomcilegdekü qauli-yi dagaldugulun ögülekü ni' [Sound deletion in the history of Mongolian: with a note on linguistic simplification laws], *MKB* 1995/1: 60-4.
 - (2000), 'Monggol bicig-ün kele bolon barimjiy-a yariyan-u kelen-ü üges-ün dagudalga-yin qaricagulul' [Comparison of Written Mongolian and the pronunciation of words in the standard spoken language], ÖMÜBDS 2000/2: 76–82.

— and Ö. Naranbatu (1978), 'Man-u ulus-un Monggol kelen-ü nutug-un ayalgu-yi anggilaqu tuqai sanal' [Thoughts on the division of Mongolian dialects in our country], *ÖMBDS* 1978/1: 117–32.

- —Goncogsürüng, [Cogtu] Süngrüb, W. Secen, Dawadagba, Togog-a, and Ö. Naranbatu (1996), *Orcin cag-un Monggol kele* [Modern Mongolian] (Kökeqota: ÖMSK). [Second edition of Nasunbayar et al. 1982]
- QAYIRQAN (1990), 'Monggol bicig-ün kelen-ü *i* egesig-ün Qorcin aman ayalgun-du nuguraqu oncalig' [Characteristics of breaking of the Written Mongolian vowel *i* in the Horchin dialect], *MKB* 90/6: 56–60.
- RACHEWILTZ, Igor de (1965), 'Some remarks on the dating of the Secret history of the Mongols', Monumenta Serica, 24: 185–206.
 - (1999), 'Some reflections on so-called Written Mongolian', in H. Eimer et al. (eds.), *Studia Tibetica et Mongolica (Festschrift Manfred Taube)*, 235–46 (Swisttal–Odendorf: Indica et Tibetica).
- RADNAEV, Èrhito Radnaevič (1958), 'O fonetičeskih svoeobrazijah barguzinskogo govora' [Phonetic characteristics of the Bargazhan dialect], Sbornik trudov po filologii, 3: 296– 301 (Ulan-Udè).

280

—(1960), 'Zametka o congol'skom govore' [A note on the Tsongool dialect], *TBurNII* 3: 127–33.

RADNAEVA, Ljubov' Dašinimaevna (1988), 'Èksperimental'no-fonetičeskij analiz diftongov s ispol'zovaniem ÈVM (na materiale burjatskogo jazyka)' [Experimental phonetic analysis of diphthongs using the computer (based on Buriad data)], VLU 2/3: 120–2.

——(2003), 'The two types of transcription (The Buryat language database)', in Solé, Recasens, and Romero (2003), 2737–40.

RAMSEY, S. Robert (1987), The languages of China (Princeton: PUP).

RAMSTEDT, G. John (1902), 'Das schriftmongolische und die Urgamundart phonetisch verglichen', JSFOu 21: fasc. 2 (pp. 1–55).

—(1906), 'Mogholica: Beiträge zur Kenntnis der Moghol-Sprache in Afghanistan, *JSFOu* 23: fasc. 4 (pp. I–IV, 1–60).

— (1912), 'Zur Geschichte des labialen Spiranten im Mongolischen', in Festschrift Vilhelm Thomsen zur Vollendung des siebzigsten Lebensjahres am 25. januar 1912, 182–7 (Leipzig: Harrassowitz).

—(1913), 'Az ŋ hang a mongolban és a törökben' [The sound ŋ in Mongolian and Turkish], Nyelvtudományi közlemények, 42: 229–38.

— (1914), 'Zur mongolisch-türkischen Lautgeschichte, I: Mongolisch-türkisches č; II: Mongolisches ž und seine Vertretung im Türkischen', *Keleti szemle (Revue orientale)*, 15: 134–50.

— (1915), 'Zur mongolisch-türkischen Lautgeschichte, III: der *j*-laut und damit zusammenhängende fragen', *Keleti szemle (Revue orientale)*, 16: 66–84.

—(1916–20), 'Ein anlautender stimmloser Labial in der mongolisch-türkischen Ursprache', *JSFOu* 32: fasc. 2 (pp. 1–10).

— (1932), 'Die Palatalisation in den altaischen Sprachen', Suomalaisen Tiedeakatemian toimituksia (Annales Academiæ scientiarum Fennicæ), B/27 (Mélanges de philologie offerts à M. J. J. Mikkola), 239–51.

----(1935), Kalmückisches Wörterbuch (Helsinki: SUS).

—(1957), Einführung in die altaische Sprachwissenschaft, I: Lautlehre (SUST 104/1), ed. by Pentti Aalto (Helsinki: SUS).

RAO, C. Radhakrishna (1965), *Linear statistical inference and its applications* (New York: Wiley).

RASSADIN, Valentin Ivanovič (1976), 'Istorija razvitija otdel'nyh fonetičeskih javlenij burjatskogo jazyka' [The developmental history of certain phonetic features in Buriad], in *Lingvističeskie issledovanija 1976: voprosy fonetiki, dialektologii i istorii jazyka*, 160– 76 (Moskva: AN SSSR).

——(1981), 'Razvitie mongol'skih affrikat v svete dannyh drugih altajskih jazykov' [The development of Mongolian affricates in the light of facts from other Altaic languages], Učenye zapiski Tartuskogo gosudarstvennogo universiteta, 558: 3–16.

——(1982), Očerki po istoričeskoj fonetike burjatskogo jazyka [Essays on the historical phonology of Buriad] (Moskva: Nauka).

(ed.) (1986), *Issledovanija po fonetike jazykov i dialektov Sibiri* [Investigations on the phonetics of Siberian languages and dialects] (Novosibirsk: Nauka).

— (ed.) (1987*a*), *Fonetičeskie issledovanija jazykov i dialektov Burjatii* [Phonetic investigations on languages and dialects of Buriatia] (Ulan-Udè: BF).

— (ed.) (1987b), *Grammatičeskoe svoeobrazie burjatskogo jazyka* [Grammatical characteristics of Buriad] (Ulan-Udè).

^{(1965), &#}x27;Barguzinskij govor' [The Bargazhan dialect], TBurNII 17: 71–107.

- RASSADIN, Valentin Ivanovič (1988), 'O tjurkskom vlijanii na razvitie govora nižneudinskih burjat' [Turkic influences on the development of the dialect of the Nizhneudinsk Buriad], in Šagdarov (1988), 62–75.
 - (1989*a*), 'Osobennosti zvukovogo stroja jazyka okinskih burjat i soëtov' [Characteristics of the sound structure of the language of the Aha Buriads and Soyots], in Šagdarov (1989), 50–73.
 - ——(1989*b*), 'Istoričeskaja svjaz' burjatskogo jazyka s jazykom bargutov' [The historical relation between Buriad and the language of the Bargu], in Abaev, Pubaev, and Čimitdoržiev (1989), 173–6.
 - —(1991), 'Zvukovoj sostav prisajanskoj gruppy burjatskih govorov' [The sound structure of the Sayan group of Buriad dialects], in Šagdarov (1991), 3–34.
 - (ed.) (1993), *Issledovanija po istorii mongol'skih jazykov* [Investigations on the history of the Mongolian languages] (Ulan-Udè: BNC).
- ——(1996a), *Prisajanskaja gruppa burjatskih govorov* [The Sajan group of Buriad dialects] (Ulan-Udè: BNC).
 - —(ed.) (1996b), *Problemy burjatskoj dialektologii* [Problems in Buriad dialectology] (Ulan-Udè: BNC).
 - ----(1999), *Stanovlenie govora nižneudinskih burjat* [The development of the language of the Nizhneudinsk Buriads] (Ulan-Udè: BNC).
- RIALLAND, Annie and Redouane Djamouri (1984), 'Harmonie vocalique, consonantique et structures de dépendance dans le mot en mongol khalkha', *Bulletin de la Société de linguistique de Paris*, 79: 333–83.
- RINČEN, Bjambyn (1958–9), 'L'inscription sinomongole de la stèle en l'honneur de Möngke qagan', *CAJ* 4: 130–43.
- ——(1966), Mongol bičgijn hèlnij züj: dèd dèvtèr, avian züj [Grammar of written Mongolian: 2nd vol., phonetics] (Ulaanbaatar: ŠUA).
- —(1969), *Mongol Ard Ulsyn hamnigan ajalguu* [The dialect of the Kamnigan in the Mongolian People's Republic] (Ulaanbaatar: ŠUA).
- ——(1979), Mongol Ard Ulsyn ugsaatny sudlal hèlnij šinžlèlijn atlas [Ethnological and linguistic atlas of the Mongolian People's Republic] (Ulaanbaatar: ŠUA).
- RINCINDORJI (1960), 'Monggol-un niguca tobciyan-u kelen-dü urtu egesig bürildükü nököcel togtagsan ni' [The shaping of environments for the development of long vowels in the language of the Secret history of the Mongols], MKJT 1960/4: 30–40.
- RÓNA-TAS, András (1960a), 'A study on the Dariganga phonology', AOH 10: 1-29.
- (1960b), 'Remarks on the phonology of the Mongour language', AOH 10: 263-7.
- -----(1962), 'On some finals of the Mongour language', AOH 14: 283-90.
- -----(1966), Tibeto-Mongolica: the Tibetan loanwords of Monguor and the development of the archaic Tibetan dialects (The Hague: Mouton).
- ROZYCKI, William V. (1994), 'Primary *h- in Mongol: the evidence considered', *CAJ* 38/1: 71–9.
- RUDNEV, Andrej Dmitrievič (1905), *Lekcii po grammatikě mongol'skago pis'mennago jazyka, čitannyja v 1903–1904 akademičeskom godu* [Lectures on the grammar of Written Mongolian given in the 1903–4 academic year] (Sankt-Peterburg: B. Avidon).
 - —(1908), 'Opyt klassifikacii mongolov po narěčijam' [Classification of the Mongols by dialects], in G. I. Ramstedt, *Sravnitel'naja fonetika mongol'skago pis'mennago jazy-ka i halha'sko-urginskago govora* [Russian translation of Ramstedt (1902)], pp. IV–VI (Sankt-Peterburg: Tipografija Imperatorskoj Akademii nauk).
 - —(1911), *Materialy po govoram vostočnoj Mongolii* [Materials on the dialects of eastern Mongolia] (Sankt-Peterburg: Kiršbaum).

282

—(1913–14), Hori-burjatskij govor: opyt izslědovanija, teksty, perevod i priměcanija [The Hori Buriad dialect: investigation, texts, translation, and remarks] (Petrograd: Kiršbaum).

RYBATZKI, Volker (2003a), 'Middle Mongol', in Janhunen (2003f), 57-82.

(2003*b*), 'Intra-Mongolic taxonomy', in Janhunen (2003*f*), 364–90.

- [RYGALOFF] Rigaloff, Alexis (1973), 'Le vocalism du mongol moderne (khalkha)', *OUMÈ* 2 *IH* 2: 69–73.
- RYGDYLON, Èrdemto Rinčinovič (1957), 'O nekotoryh osobennostjah govorov burjat Irkutskoj oblasti (po materialam èkspedicii 1955 g.)' [Some characteristics of the dialects of the Buriads in the Irkutsk area (based on data from the 1955 expedition)], ZBurNII 23: 184–97.
- ŠAGDAROV, Lubsan Doržievič (1968), 'O nekotoryh jazykovyh osobennostjah tugnujskih i aginskih burjat i stepeni ih otraženija v literaturnom jazyke' [Some linguistic characteristics of the Tugnui and Aga Buriads and the extent to which they are expressed in the literary language], *TBION* 2: 154–63.
- (1969), 'Burjatskij jazyk' [Buriad], in N. A. Baskakov (ed.), Zakonomernosti razvitija literaturnyh jazykov narodov SSSR v sovetskuju epohu: osnovnye processy vnutristrukturnogo razvitija tjurkskih, finno-ugorskih i mongol'skih jazykov, 383–464.
- ——(ed.) (1988), *Problemy mongol'skogo jazykoznanija* [Problems in Mongolian linguistics] (Novosibirsk: Nauka).
- ——(ed.) (1989), *Zvukovoj stroj mongol'skih jazykov* [The sound structure of Mongolian languages] (Ulan-Udè: BNC).
- (ed.) (1991), Segmentnye i prosodičeskie edinicy jazykov bajkal'skogo regiona [Segmental and prosodic units in the languages of the Baikal region] (Ulan-Udè: BNC).
- and Valentin Ivanovič Rassadin (1968), 'Ob upotreblenii tofalarami burjatskogo jazyka' [The use of Buriad by Tofalars], *TBION* 2: 176–86.
- SAITÔ Yoshio (1983), 'Monguorugo no on'in taikei' [The phoneme system of Monguor], Gengo bunka kenkyû, Tôkyô gaikokugo daigaku, 1: 9-17.
- ——(1984), 'Gendai Mongorugo no jakka boin to boin chôwa' [Reduced vowels and vowel harmony in modern Mongolian], *Lexicon*, 13: 57–71.
- ——(1985), 'Gendai Mongorugo no on'in kaishaku: kôgaisui shiin to kôkôgai on' [Phonological interpretation in modern Mongolian: uvular consonants and palatal sounds], *Kyôto sangyô daigaku kokusai gengo kagaku kenkyûsho shohô*, 7/1: 5–10.
- (1986), 'The consonant system of modern Khalkha Mongolian', in *In honor of Shigeru Takebayashi*, 115–28 (Tokyo: Iwasaki Linguistic Circle).
- (1993*a*), 'Monguorugo no on renzoku' [Sound sequences in Monguor], *NMGK* 24: 33-48.
- ——(1993b), 'Chûki Mongorugo kanji on'yaku bunken ni okeru shiin chôfuku' [Consonant gemination in Middle Mongolian documents transcribed with Chinese characters], OShKR 115–30. [Originally published in 1989]
- (1996*a*), 'À propos de l'accent en Mongol médiéval de l'Ouest', *PIAC* 38: 299–309.
- ——(1996b), 'Consonant gemination in Mongolian', Bulletin: the IAMS news information on Mongol studies, 17–18: 72–9.
 - (2000), 'Mukadimatto-aru-adabu ni oite 'alif to hâ' de onshasareta Mongorugo no gomatsu boin ni tsuite' [Mongolian word-final vowels transcribed with 'alif and hâ' in Muqaddimat al-adab], Ajia, Afurika gengo bunka kenkyû (Journal of Asian and African studies), 60: 159–68.
 - —(2003), *Chûki Mongorugo no moji to onsei* [Scripts and sounds in Middle Mongolian] (Kyôto: Shôkadô).

- SAMBUUDORŽ, Očirbatyn (1996), 'Ojrdyn aman ajalguunuudyn urt egšig aviany togtolcoo' [The system of long vowels in Oirat dialects], NMGK 27: 15–30.
- ——(2002), *Mongol hèlnij ojrd ajalguuny fonemijn togtolcoo* [The phoneme system of the Oirad dialect of Mongolian] (Ulaanbaatar: ŠUA).
- SAN Yüwè (2002), 'Abag-a-yin yariyan-u abiyan-u oncalig' [Sound characteristics of the Abaga dialect], *MKB* 2002/7: 16–21.
- SANGDÜGÜRENG and Cegen (1989), 'Naiman aman ayalgun-u abiyan-u sistèm' [The sound system of the Naiman dialect], *ÖMYSS* 1989/1: 95–103.
- [SANŽAA, Žamsranžavyn] Sandža (1987a), 'K voprosu o pozicijah i pozicionnyh čeredovanijah soglasnyh v sovremennom mongol'skom literaturnom jazyke' [The problem of the positions and positional alternation of consonants in modern literary Mongolian], Archív orientální, 55: 29–40.
 - —(1987b), 'Mongol hèl ajalguund čanga, sul gijgüülègč sèlgèh ës' [Laws for alternation of tense and lax consonant in Mongolian dialects], HZS 21: fasc. 10 (pp. 63–9).
 - ——(1988), 'Mongol hèl ajalguund gijgüülègč avia sèlgèh tügèèmèl züj togtol' [General consonant alternation laws in Mongolian], HZS 22: fasc. 9 (pp. 79–85).
- (1990), Osnovnye zakonomernosti čeredovanija soglasnyh zvukov v mongol'skom jazyke (pp. 1–36); Mongol hèlènd gijgüülègč avia sèlgèh ündsèn züj togtol (pp. 37–64)
 [Basic consonant alternation laws in Mongolian], Avtoreferat kand. diss, Ulaanbaatar.
- ——(1993), 'Vlijanie vnutrennih i vnešnih faktorov na razvitie zvukovoj sistemy halhamongol'skogo jazyka (na materiale soglasnyh zvukov)' [The influence of internal and external factors on the development of the Halh sound system (based on consonant data)], in Rassadin (1993), 122–33.

——(1998), Osnovnye zakonomernosti fonetičeskih izmenenij zvukov v mongol'skih jazykah i dialektah [Basic laws of phonetic sound changes in Mongolian languages and dialects], Avtoreferat dokt. diss., Ulaanbaatar.

- SANŽEEV, Garma Dancaranovič (1930), Fonetičeskie osobennosti govora nižneudinskih burjat [Phonetic characteristics of the dialect of the Nizhneudinsk Buriad] (Materialy Komissii po issledovaniju Mongol'skoj i Tannu-Tuvinskoj narodnyh respublik i Burjat-Mongol'skoj ASSR, 8) (Leningrad: AN SSSR).
- ——(1931), Darhatskij govor i fol'klor [Darhad dialect and folklore] (Materialy Komissii po issledovaniju Mongol'skoj i Tuvinskoj narodnyh respublik i Burjat-Mongol'skoj ASSR, 15) (Leningrad: AN SSSR).
- (1939), 'Lingvističeskie nabljudenija v Eravne i Hori' [Linguistic observations in Eravna and Hori], *Zapiski GIJaLI*, 1: 7–23.
- (1940), Grammatika kalmyckogo jazyka [Kalmuck grammar] (Moskva: AN SSSR).
- (1941), *Grammatika burjat-mongol'skogo jazyka* [Buriad Mongolian grammar] (Moskva: AN SSSR).
- ——(1951), 'Nekotorye voprosy burjat-mongol'skogo jazykoznanija v svete trudov I. V. Stalina' [Some problems in Buriad Mongolian linguistics in the light of the works of I. V. Stalin], ZBurNII 11: 88–121.
- ——(1952), 'Mongol'skie jazyki i dialekty' [Mongolian languages and dialects], Učenye zapiski Instituta vostokovedenija, 4: 30–125.
- ——(1953), *Sravnitel'naja grammatika mongol'skih jazykov, I* [Comparative grammar of the Mongolian languages, I] (Moskva: AN SSSR).
- ——(1957), 'Nekotorye voprosy sravniltel'no-istoričeskoj mongolistiki' [Some problems in comparative historical mongolistics], *Rocznik orientalistyczny*, 21: 373–84.
- —(1959), *Sovremennyj mongol'skij jazyk* [Modern Mongolian] (Moskva: Izd. vostočnoj literatury). English translation by D. M. Segal 1973: G. D. Sanzheyev, *The modern*

Mongolian language (Moskva: Nauka).

- (1964*a*), 'K assimiljacii glasnyh v mongol'skih dialektah' [Vowel assimilation in Mongolian dialects], *Zeitschrift für Phonetik Sprachwissenschaft und Kommunikationsfor*schung, 17: 627–30.
- (1964b), Staropis'mennyj mongol'skj jazyk [Old-script Mongolian] (Moskva: Nauka). English translation by John R. Krueger 1988: G. D. Sanzheyev, The old-script Mongolian language and its development in Khalkha (Bloomington, IN: Mongolia Society).
- —(1967), 'Stečenie soglasnyh v mongol'skom jazyke' [Consonant clusters in Mongolian languages], in Istoriko-filologičeskie issledovanija: sbornik statej k semidesjatiletiju akademika N. A. Konrada, 141–5 (Moskva: Nauka).
- —(1970), 'Zametki po tjurksko-mongol'skomy vokalizmu' [Remarks on Turkic-Mongolian vocalism], in R. Jakobson and S. Kawamoto (eds.), *Studies in general and oriental linguistics presented to Shirô Hattori on the occasion of his sixtieth birthday*, 508–15 (Tôkyô: TEC Company).
- ——(1978), 'Ob allofonah dolgih glasnyh fonem v mongol'skih jazykah' [Allophones of long vowel phonemes in Mongolian languages], VJa 1978/1: 88–99.
- ——(1982), 'Fonematičny li mjagkie soglasnye v mongol'skih jazykah?' [Are the palatalized consonants phonematic in Mongolian languages?], TPV 131–8.
- ——(ed.) (1985), *Mongol'skij lingvističeskij sbornik* [Mongolian linguistic collection] (Moskva: Nauka).
- SANŽINA, Darima Dabaevna (1989), 'Ètnolingvističeskoe opisanie irojskih burjat' [Ethnolinguistic description of the Iroi Buriad], in Abaev, Pubaev, and Čimitdoržiev (1989), 176–9.
- [ŠARHÜÜ, Če.] Sharufû (1979–80), 'Daruhato hôgen' [The Darhad dialect], MK 10: 17–37; 11: 55–77. (Trans. Shimizu Mikio).
- ——(1984), 'Darhadyn aman ajalguuny urt ègšgijn üüslijn tuhaj' [The origin of long vowels in the Darhad dialect], *MUISÈ* 1984/3: 133–9.
- SATÔ Nobuharu (1985), 'Dagûru hôgen no boin henka' [Vowel changes in the Dagur dialect], MK 8: 111–18.
- ——(1987), 'Mongorugo koritsu shô hôgen no chôboin ni tsuite: Shera Yögur' [Long vowels of an isolated Mongolian dialect: Shira Yugur], MK 18: 78–93.

——(1991–2), 'Chûgoku Kanshuku, Seikai shô no Mongoru kei shogengo ni okeru gotô heisaon no nan'onka to kôonka ni tsuite' [Lenition and fortition of initial stops in in the Mongolian languages of the Gansu–Qinghai area], NMGK 22–23: 14–28.

- SCHANE, Sanford A. (1984), 'The fundamentals of particle phonology', Phonology yearbook, 1: 129–55.
- SCHEIN, Barry and Donca Steriade (1986), 'On geminates', Linguistic inquiry, 17: 691-744.
- SCHMIDT, Isaak Jakob (1831), *Grammatik der mongolischen Sprache* (Sankt-Peterburg: Kaiserliche Akademie der Wissenschaften).
- SCHWARZ, Henry G. (1982), 'A script for the Dongxiang', ZASB 16: 153–64.
 (1984), The minorities of Northern China: a survey (Bellingham: Western Washing
 - ton University).
- SECEN, W. (1988), 'Aru Qorcin yariyan-u egesig ba geyigülügci abiyalaburi' [The vowel and consonant phonemes of the Arhorchin dialect], *MKUJ* 1988/2: 16–28.
 - ——(1989), 'Barimjiy-a abiy-a kiged Oyirad ayalgun-u abiyalaburi-yin qaricagulul' [Comparison of the phonemes of the standard pronunciation and the Oirad dialect], ÖMBYS 1989/2: 94–105.
 - —(1994), 'Naiman qosigun-u yariyan-u kele-yi dürimjigülkü-dü qolbogdaqu abiyan-u asagudal' [Phonetic problems related to the standardization of the spoken language of Naiman banner], ÖMÜBDS 1994/3: 92–8.

- SECEN, W. (2000), 'Bagarin, Aru Qorcin, Naiman-u abiyan-u qaricagulul' [Phonetic comparison of Baarin, Arhorchin, and Naiman], MKUJ 2000/2: 9–14.
- SECENBAGATUR, B. (1959), 'Bagarin-u aman ayalgun-u tuqai' [The Baarin dialect], *MKJT* 1959/6: 14–23; 1959/11: 46–51.

—(1982), 'Caqar barimjiy-a abiy-a kiged Bagarin aman ayalgun-u abiyan-u qaricagulul' [Comparison of Chahar standard pronunciation and the Baarin dialect], *MKUJ* 1982/1: 4–22.

- SECENBILIG (1982), 'Alaša aman ayalgun-daki jarim abiyan-u tuqai' [Some sounds in the Alshaa dialect], *MKB* 1982/4: 29–36.
- [SECENCOGTU] Sīqīncháokètú (1999), *Kāngjiāyǔ yánjiū* [Studies on Kangjia] (Shànghǎi: Shànghǎi yuǎndōng).
- [——] Sīqīncháokètú (2002), 'Kāngjiāyǔ gàikuàng' [Sketch of Kangjia], MY 2002/6: 66– 77.
- ŜERBAK, Aleksandr Mihajlovič (1992–3), 'Sur le soi-disant rhotacisme mongolien', AOH 46: 323–6.
- SERENGNORBU, B. (1986), 'Ordos aman ayalgun-daki abiyan-u jarim üjegdel-ün tuqai' [Some phonetic phenomena in the Ordos dialect], ÖMYSS 1986/2: 90–5.
- ——(1996), 'Ordos aman ayalgun-u e, *i*-yin soligdal-un tuqai' [Merger of e and *i* in the Ordos dialect], *MKB* 1996/12: 38–40.
- SHIMIZU Mikio (1979), 'Mongorugo akusento no tokuchô' [A characteristic of the tone accent in Mongolian], OGK 162: 9–11.

—(1980), 'Mongorugo Haruha hôgen no onsetsu kôzô' [The syllable structure of Halh Mongolian], *Onsei gengo no kenkyû*, 1: 67–77.

- (1985), 'The accent of modern Mongolian', OUMÈ 4 IH 2: 394-7.
- ——(1991–2), 'Mongorugo no on'in henka to akusento' [Phonological change and tone accent in Mongolian], NMGK 22–3: 118–35.
- [——] Shinizu (1992), 'On the consonant G in modern Khalkha dialect', OUMÈ 5 IH 2: 395.
- ——(1996), 'The accent of modern Mongolian', Bulletin: the IAMS news information on Mongol studies, 17–18: 67–71.
- SINEDKE (1998), 'Sagarmag egesig qudum Monggol bicig-tü cuqum orosiju bayiqu-uu esekü-üü?!' [Is there really a neutral vowel in the Old Mongolian writing system?], MKUJ 1998/1: 4–12.
- SINQUWA (1994), 'Küriy-e aman ayalgun-u ɔ, ʊ egesig-ün tuqai' [The vowels ɔ, ʊ in the Hüree dialect], *Monggol-un sudulul*, 1994/1: 119–24.
- [ŠIROKOGOROV] Shirokogoroff, Sergej Mihajlovič (1931), Ethnological and linguistical aspects of the Ural-Altaic hypothesis (Běipíng: Commercial Press). (Reprinted from Qīnghuá xuébào (Tsing Hua journal), 6).
- SKORIK, Petr Jakovlevič (ed.) (1968), Jazyki narodov SSSR [The languages of the peoples of the USSR], 5 (Leningrad: Nauka).
- SKRIBNIK, Elena Konstantinovna (2003), 'Buryat', in Janhunen (2003f), 102–28.
- SLATER, Keith W. (1998), *Minhe Mangghuer: a mixed language of the Inner Asian frontier*. Ph.D. dissertation, University of California, Santa Barbara.
 - —(2003a), A grammar of Mangghuer: a Mongolic language of China's Qinghai-Gansu sprachbund (London: Routledge Curzon).
 - (2003*b*), 'Mangghuer', in Janhunen (2003*f*), 307–24.
- [ŠMITS] Schmidt, Pēteris (1898), 'Der Lautwandel im Mandschu und Mongolischen', Journal of the Peking oriental society, 4: 29–78.
- SODOBAGATUR (1999), 'Qaracin aman ayalgun-u ekin-ü busu üy-e-yin bogoni egesig-ün

tuqai' [Short vowels in non-initial syllables in the Harchin dialect], MKB 1999/10: 23-5.

- (2000), 'Qaracin Tümed aman ayalgun-u urtu egesig-ün tuqai' [Long vowels in the Harchin Tümd dialect], MKB 2000/11: 15–18.
- ——(2001), 'Qaracin aman ayalgun-u bogoni egesig abiyalaburi' [Short vowel phonemes in the Harchin dialect], *MKB* 2001/11: 11–12.
- SOKTOEVA, Svetlana Purbuevna (1980*a*), 'Soglasnye 4-oj artikuljacii v horinskom dialekte burjatskogo jazyka (po dannym statičeskogo rentgenografirovanija)' [Velars in the Hori dialect of Buriad (according to static X-ray data)], in Nadeljaev (1980), 139–49.
 - —(1980b), 'Malošumnye soglasnye 2-oj i 3-ej artikuljacii v horinskom dialekte burjatskogo jazyka (po dannym rentgeno-dentopalatografirovanija)' [Dental and palatal sonorants in the Hori dialect of Buriad (according to X-ray dentopalatography)], in E. I. Ubrjatova (ed.), *Sibirskij dialektologičeskij sbornik*, 136–58 (Novosibirsk).
 - —(1981a), 'Šumnye soglasnye 2-j i 3-j artikuljacii v horinskom dialekte burjatskogo jazyka (po dannym rentgenografirovanija)' [Dental and palatal obstruent consonants in the Hori dialect of Buriad (according to radiographic data)], in *Teoretičeskie voprosy fonemiki i grammatiki jazykov narodov SSSR*, 139–45 (Novosibirsk: Novosibirskij gosudarstvennyj universitet).
- (1981b), 'Gubnye soglasnye v horinskom dialekte burjatskogo jazyka' [Labial consonants in the Hori dialect of Buriad], in V. M. Nadeljaev (ed.), *Eksperimental'no*fonetičeskie issledovanija sibirskih jazykov, 100–18 (Novosibirsk: IIFF).
- (1983), 'Dlitel'nost' soglasnyh fonem v horinskom dialekte burjatskogo jazyka' [The duration of consonant phonemes in the Hori dialect of Buriad], in Nadeljaev (1983), 59–62.
- —(1984), 'Kačestvenno-količestvennye harakteristiki allofonov gubnyh soglasnyh fonem horinskogo dialekta burjatskogo jazyka (po dannym pnevmooscillografirovanija)' [Qualitative and quantitative characteristics of allophones of labial consonant phonemes in the Hori dialect of Buriad (by pneumooscillographic data)], in V. M. Nadeljaev (ed.), *Fonetika jazykov Sibiri*, 106–14 (Novosibirsk: IIFF).
- (1985), 'Dlitel'nost' malošumnyh soglasnyh 2-oj artikuljacii v burjatskom jazyke' [Duration of dental sonorants in Buriad], in V. M. Nadeljaev (ed.), *Fonetika sibirskih jazykov*, 80–93 (Novosibirsk: IIFF).
- ——(1986a), 'Osobennosti artikuljacionno-akustičeskoj bazy burjatskogo jazyka v oblasti konsonantizma' [Characteristics of the articulatory-acoustic base of Buriad in the domain of consonantism], in Rassadin (1986), 66–8.
- ——(1986b), 'Spornye i nerešennye problemy konsonantizma burjatskogo jazyka' [Controversial and unsolved problems in Buriad consonantism], in Nadeljaev (1986), 82–4.
- ——(1986c), 'Perednejazyčnye mgnovennopregradnye fonemy v burjatskom jazyke' [Apical vibrant phonemes in Buriad], in E. I. Ubrjatova (ed.), *Fonetičeskie struktury v sibirskih jazykah*, 72–9 (Novosibirsk: IIFF).
 - —(1988), Konsonantizm horinskogo dialekta burjatskogo jazyka: èksperimental'nofonetičeskoe issledovanie [The consonantism of the Hori dialect of Buriad: experimental phonetic investigation] (Novosibirsk: Nauka).
 - (1989a), 'Kačestvenno-količestvennaja harakteristika perednejazyčnyh i perednejazyčno-srednejazyčnyh soglasnyh horinskogo dialekta burjatskogo jazyka' [Qualitative and quantitative characteristics of apical and apico-dorsal consonants in the Hori dialect of Buriad], in Šagdarov (1989), 154–8.
 - —(1989b), 'Morfologičeskij i distributivnyj analizy soglasnyh v burjatskom jazyke' [Morphological and distributive analysis of Buriad consonants], in M. N. Vall (ed.), *Lingvističeskie edinicy, konstrukcii i tekst*, 18–23 (Novosibirsk: Nauka).

- SOLÉ, Maria-Josep, Daniel Recasens, and Joaquin Romero (eds.) (2003), *Proceedings of the 15th International congress of phonetic sciences* (Barcelona: Universitat Autònoma de Barcelona).
- SOLNCEV, Vadim Mihajlovič (ed.) (1987), *V Meždunarodnyj kongress mongolovedov: doklady sovetskoj delegacii* [V International congress of Mongolists: papers by the Soviet delegation] (Moskva: Nauka).
- SONAM, B. (1985), 'Kökenagur Monggol-un yariyan-u kelen-deki y geyigülügci-yin tuqai dökom ögüleki ni' [Discussion on the consonant y in spoken Qinghai Mongolian], *MKUJ* 1985/4: 17–23.
- SONG Chae-mok (1999*a*), 'Halha Monggolŏ-ŭi ŭmjŏl kujo: changmoŭm-gwa ijungmoŭmŭl chungsim-ŭro' [Syllable structure of Halh Mongolian: focused on long vowels and diphthongs], *AH* 9: 147–61.
- ——(1999b), 'Halha Monggolŏ-ŭi chungch'ŏp hyŏnsang-e taeha-yŏ' [The reduplication phenomenon in Halh Mongolian], Ŏnŏ-ŭi yŏksa: Sŏng Paeg-in kyosu chŏngnyŏn t'oeim kinyŏm nonmunjip kanhaengwŏnhoe p'yŏn (History of languages in honor of Professor Baeg-in Song), 343–80 (Sŏul: T'aehaksa).
- SŏNG Paeg-in (1983), 'Monggoŏ Dagurŭ pangŏn-ŭi ŭmun ch'egye yŏngu' [A study on the phonological system of Dagur Mongolian], *Dongbang hakchi (The journal of Korean studies)*, 40: 119–52.
- ŠONGQOR (1987), 'Dumdadu erten-ü oor Monggol ayalgu-yin geyigülügci-yin togtalcag-a' [The consonant system of Middle and Old Mongolian], *MKUJ* 1987/4: 4–23.
 - ——(1996), *Erten-ü Monggol kelen-ü sudulul* [Studies in Classical Mongolian] (Kökeqota: ÖMSK).
- SOULIÉ, Georges (1903), Éléments de grammaire mongole (dialecte ordoss) (Paris: Imprimerie nationale).
- SOYOL (2002), 'Qorcin aman ayalgun-u abiyan-u qubiralta-yin jarim asagudal' [Some problems of sound changes in the Horchin dialect], *ÖMÜYS* 2002/3: 32–42.
- STERIADE, Donca (1979), 'Vowel harmony in Khalkha Mongolian', MIT Working papers in linguistics, 1: 25–50.
- ——(1987), 'Redundant values', CLS 23, Parasession on autosegmental and metrical phonology, 339–62.
- (1995), 'Underspecification and markedness', in Goldsmith (1995), 114–74.
- STREET, John C. (1957*a*), *The language of the* Secret history of the Mongols (New Haven, CT: American oriental society).
- (1957b), Review of Poppe (1955), Language, 33: 81-7.
- ——(1962), 'Kalmyk shwa', in N. Poppe (ed.), American studies in Altaic linguistics (Uralic and Altaic series, 13), 263–91 (Bloomington: Indiana University).
- ——(1963), *Khalkha stucture* (*Uralic and Altaic series*, 24) (Bloomington, IN: Indiana University).
- (1966), 'Urdus phonology: a restatement', UAJb 38: 92-111.
- (1986), On the 14th century punctuation of Mongolian in the Yüan-ch'ao pi-shih (The Mongolia Society occasional papers, 12) (Bloomington, IN: Mongolia Society).
- STUART, Don Graham and Matthew M. Haltod (1957), 'The phonology of the word in modern standard Mongolian', *Word*, 13: 65–99.
- STURT, Reginald W. (1941), A Mongolian grammar. [Typescript] (Chìfēng).
- SÜHBAATAR, Cègmèdijn (1970), 'Phonétique mongole', Études mongoles, 1: 55-62.
- ——(1973), 'Orčin cagijn mongol hèlnij fonemyn bajrlalyn asuudald' [The problem of phoneme distribution in modern Mongolian], HZS 9: fasc. 18 (pp. 193–8).
- SŪN Zhú (1981), 'Xiàndài Měnggǔyǔ de ruòhuà yuányīn' [Vowel reduction in modern Mongolian], MY 1981/1: 46–54.

- —(1983), 'Měnggůyů Cháhā' ěr făngyīn yǔ shūmiànyǔ yǔyīn de bǐjiào' [Comparison of the speech sounds of the Chahar dialect and Written Mongolian], *MY* 1983/2: 7–16.
- (1985), *Měnggǔyǔ wén jí* [Collected works on Mongolian languages] (Xīníng: Qīnghǎi rénmín).

—(1996*a*), 'Lùn Dáwò'ěrzú yǔyán: jiān tán Dáwò'ěryǔ yǔ Měnggǔyǔ de mǒuxiē yìtóng' [On the language of the Dagurs: with an addendum on some differences between Dagur and Mongolian], in Sūn Zhú, *Měnggǔ yǔzú yǔyán yánjiū*, 178–230 (Kökeqota: Nèiměnggǔ dàxué). [Originally published in 1984]

- (1996b), 'Xiàndài Měnggǔyǔ biāozhǔnyīn yīnxì fēnxī' [Analysis of the sound system of modern standard Mongolian], in Sūn Zhú, *Měnggǔ yǔzú yǔyán yánjiū*, 68–125 (Kökeqota: Nèiměnggǔ dàxué). [Originally published in 1986]
- [Jagunasutu] Zhàonàsītú, Chén Năixióng, Wú Jùnfēng, and Lǐ Kèyù (1990), Měnggŭ yůzú yúyán cídiăn [Dictionary of the Mongolian languages] (Xīníng: Qīnghǎi rénmín).
- SÜRÜNG, Ni. (1997), 'Oyirad ayalgu bolon tegün-ü abiyan-u gool oncalig' [The Oirad dialect and its main phonetic characteristics], *Kele ba orcigulg-a (Language and translation)*, 1997/4: 13–21.
- SVANTESSON, Jan-Olof (1985), 'Vowel harmony shift in Mongolian', *Lingua*, 67: 283–327.
 (1988a), 'Sonority and syllabification in Mongolian', in W. Dressler et al. (eds.), *Discussion papers for the Sixth international phonology meeting and Third international morphology meeting*, 70–2 (Wien).
- (1988b), 'U', Linguistics of the Tibeto-Burman area, 11: 64–133.
- (1988*c*), 'The East Asian code shift', *WPLund* 33: 193–214.
- ——(1990), 'Phonetic correlates of stress in Mongolian', in *Proceedings ICSLP 90: 1990 International conference on spoken language processing*, 1: 617–20 (Tôkyô: Acoustical society of Japan).
- ——(1991a), 'Vowel palatalization in Mongolian', in Actes du XIIème Congrès international des sciences phonétiques, 5: 102–5 (Aix-en Provence: Université de Provence).
- ——(1991b), 'Tradition and reform in China's minority languages', International journal of applied linguistics, 1: 70–88.
- ——(1992), 'Stress in Khalkha Mongolian', OUMÈ 5 IH 2: 410-12.
- (1994), 'Mongolian syllable structure', WPLund 42: 225-39.
- (1995*a*), 'Cyclic syllabification in Mongolian', *NLLT* 13: 755–66.
- (1995b), 'Phonetic evidence for the great Mongolian vowel shift', in K. Elenius and P. Branderud (eds.), *Proceedings of the XIIIth International congress of phonetic sciences ICPhS 95*, 1: 416–19 (Stockholm University).
- (1996), 'Glides in Mongolian phonology', in L. Heltoft and H. Haberland (eds.), Proceedings of the Thirteenth Scandinavian conference of linguistics, 209–16 (Roskilde University).
- ——(1997), 'Reduplication in Mongolian phonology', *Phonum*, 4: 133–6 (Umeå University).
- —(2000), 'Mongolic vowel shifts and the classification of the Mongolic languages', *AH* 10: 193–207.
- (2003*a*), 'Preaspiration in Old Mongolian?', *Phonum*, 9: 5–8 (Umeå University).
- (2003b), 'Khalkha', in Janhunen (2003f), 154–76.
- and Vivan Franzén (1996), 'Mongolian schwa', in SICOPS '96: the First Seoul international conference on phonetic sciences, 70–4 (Seoul: Phonetic society of Korea).
- TAKEUCHI Ikunosuke (1937), 'Môkogo bungo kaitei' [Manual of written Mongolian], Môkogaku, 1: 174–200.
- TAYA, D (1992), 'Qobogsayiri-yin Torgud aman ayalgu-yin jarim abiyalaburi-yin qubiral-

un tuqai' [Some phoneme changes in the Torguud dialect in Hobogsair], *MKUJ* 1992/6: 27–30.

TEMÜR (1982), 'Caqar aman ayalgun-u jarim geyigülügci abiyan-u ondoosil' [Dissimilation of some consonant sounds in the Chahar dialect], *ÖMBDS* 1982/1: 109–14.

TENIŠEV, Èdgem Rahimovič (1976), 'O jazyke kalmykov Issyk-kulja' [The language of the Kalmucks of Ysyk Köl], *VJa* 1976/1: 82–7.

——(1997), 'Issyk-kul'skih kalmykov jazyk' [Ysyk Köl Kalmuck], in Jarceva (1997), 67– 73.

- THOMSEN, Kaare (1958), 'Die Entwickelung der Gruppe *uya (*üge) im Mongolischen', Acta orientalia, 23: 263–7.
 - —(1959), 'Bemerkungen über das mongolische Vokalsystem der zweiten Silbe', Acta orientalia, 24: 1–4.
- (1976), 'Über einige Vokalphoneme im Mongolischen', in Heissig et al. (1976), 641-6.
- (1987), 'Zur Aussprache des mittelalterlichen Mongolischen', UAJb NF 7: 171-80.
- [TIĀN Fēng] Tiyanfèng (1983), 'Monggol kelen-ü urtu egesig-ün bürildül ba kögjil-ün tuqai' [The origin and development of long vowels in Mongolian], *MKUJ* 1983/3: 14–23.
- TODAEVA, Buljaš Hojčievna (1951), Grammatika sovremennogo mongol'skogo jazyka: fonetika i morfologija [Grammar of modern Mongolian: phonetics and morphology] (Moskva: AN SSSR).
- [—] Todayèwa (1956), 'Dumdadu Ulus-un Monggol kele ayalgu-nugud-un ündüsün oncalig ba tegün-i Qalq-a Monggol-un udq-a jokiyal-un kele-tei qarsigulugsan ni' [The basic characteristics of the Mongolian languages and dialects in China and their contrasts with literary Halh], *MKB* 1956/4: 2–50.
- [—] Bù. Hā. Tuōdáyèwá (1957), 'Yánjiū Zhōngguó gè Měnggǔ yǔ hé fāngyán de chūbù zŏngjié' [Preliminary report on research on Mongolian languages and dialects in China], ZY 1957/9: 32–40.
- -----(1959), 'Über die Sprache der Tung-hsiang', AOH 9: 273-310.
- -----(1960*a*), 'Mongolische dialekte in China', *AOH* 10: 141–69.
- ——(1960*b*), *Mongol'skie jazyki i dialekty Kitaja* [Mongolian languages and dialects in China] (Moskva: Izd. vostočnoj literatury).
- -----(1961), Dunsjanskij jazyk [Santa] (Moskva: Izd. vostočnoj literatury).
- -----(1963), 'Einige besonderheiten der Paoan-Sprache', AOH 16: 175-97.
- -----(1964), Baoan'skij jazyk [Bonan] (Moskva: Nauka).
- (1966), 'Jazyk šira jugurov' [The language of the Shira Yugur], in È. R. Tenišev and B.
- H. Todaeva, Jazyk želtyh ujgurov, 45–83 (Moskva: Nauka).
- ——(1968), 'Kalmyckij jazyk' [Kalmuck], in Skorik (1968), 34–52.
- (1970), 'Zametki po jazyku hajjan'skih (kuku-norskih) mongolov' [Notes on the language of the Haiyan (Qinghai) Mongols], in R. Jakobson and S. Kawamoto (eds.), *Studies in general and oriental linguistics presented to Shirô Hattori on the occasion of his sixtieth birthday*, 627–31 (Tôkyô: TEC Company).
- ——(1973), Mongorskij jazyk [Monguor] (Moskva: Nauka).
- ——(1975), 'Ob izučenii jazyka šira jugurov' [Investigation of the language of the Shira Yugur], in Bitkeev (1975*c*), 234–42.
- [-----] Todayèwa (1979), 'Boo An kele' [Bonan], ÖMBDS 1979/1: 155–74.
- ——(1985), *Jazyk mongolov Vnutrennej Mongolii: očerk dialektov* [The language of the Mongols in Inner Mongolia: dialect descriptions] (Moskva: Nauka).
- —(1986), Dagurskij jazyk [Dagur] (Moskva: Nauka).

——(1988), 'Nekotorye dannye o jazyke ih-mjangan' [Some data on the Ih Minggan language], in Šagdarov (1988), 134–9.

-----(1997a), 'Baoan'skij jazyk' [Bonan], in Jarceva (1997), 29-36.

——(1997b), 'Dagurskij jazyk' [Dagur], in Jarceva (1997), 51–60.

——(1997*c*), 'Dunsjanskij jazyk' [Santa], in Jarceva (1997), 60–6.

——(1997d), 'Mongolov Vnutrennej Mongolii jazyk' [The language of the Mongols of Inner Mongolia], in Jarceva (1997), 98–107.

——(1997*e*), 'Mongorskij jazyk' [Mongour], in Jarceva (1997), 124–32.

——(1997f), 'Šira jugurov jazyk' [Shira Yugur], in Jarceva (1997), 148–52.

TOGOG-A (1983*a*), 'Qorcin aman ayalgun-u urugsilagsan egesig-ün tuqai' [Fronted vowels in the Horchin dialect], *ÖMYSS* 1983/2: 24–35.

— (1983b), 'Qorcin aman ayalgu-yi barimjiy-a abiy-a-bar dürimjigülkü-dü qolbogdaqu abiyan-u tal-a-yin kedün asagudal' [Some phonetic problems connected with the regularization of the Horchin dialect with the standard pronunciation], *MKUJ* 1983/3: 4–13.

——(1991*b*), 'Küriy-e qosigun-u Monggol kele ayalgun-u tuqai' [The Mongolian dialect in Hüree banner], *MSYQ* 2: 1019–26.

Тöмörcèrèn, Žogdovžavyn (1964), 'Üzèmčin ajalguuny zarim onclogoos' [Some characteristics of the Üjemchin dialect], *StMo* 4/10[13]–21: fasc. 10[13] (pp. 1–14).

——(1966*a*), 'Cahar ajalguuny zarim onclogoos' [Some characteristics of the Chahar dialect], *StMo* 5/2–7: fasc. 3 (pp. 48–60).

——(1966b), 'Halhyn ajalguun dah' gijgüülegč zohicohyn ul mör' [Traces of consonant harmony in Halh], ŠUAM 1966/1: 99–102.

——(1967*a*), 'Övörhangajn halh aman ajalguuny zarim onclogoos' [Some characteristics of the Halh dialect in Övörhangai], *StMo* 6/15–21: fasc. 16 (pp. 18–22).

—(1967b), 'Saarmag ègšgijn tuhaj' [On neutral vowels], ŠUAM 1967/1: 71–9.

——(1968), 'Mongol hèlnij üe ès bütèèh ègšgijn učir' [Non-syllabic vowels in Mongolian], ŠUAM 1968/1: 61–8.

— (1969*a*), 'Üzèmčin ajalguug sudalsan tèmdèglèlèès' [Notes on the study of the Uzemchin dialect], *StMo* 6/22–33: fasc. 33 (pp. 184–215).

— (1969*b*), 'Halhyn ajalguuny ègšig aviany tagnajšral' [Vowel palatalization in the Halh dialect], *HZS* 7: fasc. 4 (pp. 143–50).

——(1969*c*), 'Halhyn aman ajalguuny uužimših zohicol' [Vertical harmony in the Halh dialect], *StMo* 7: fasc. 19 (pp. 243–56).

(1970), 'Mongol hèlnij üe huraagdah ës' [Syllable deletion rules in Mongolian], *HZS* 8: fasc. 6 (pp. 179–200).

——(1971), 'Bargyn tuhaj tovč tèmdèglèl' [A short description of Bargu], StMo 8: fasc. 16 (pp. 203–20).

——(1976), 'Iz istorii mongol'skogo vokalizma' [From the history of the Mongolian vocalism], *StMo* 3(11): fasc. 14 (pp. 229–37).

Тöмörтogoo, Domijn (1967), 'Darhadyn aman ajalguuny urt ègšig, tèdgèèrijn üüsèlijn tuhaj tovč tèmdèglèl' [Long vowels in the Darhad dialect, and a note on their origin], *HZS* 4: fasc. 3 (pp. 196–203).

—(1969), 'Mongol bičgijn dursgaluudad ègšig aviany ižilsèl, uruulšil hijgèèd hoër ègšgijn dundah zarim gijgüülègčijn sugarlyg tusgasan n'' [Reflections on vowel assimilation, labialization and deletion of some consonants between two vowels in Mongolian written documents], *HZS* 6: fasc. 5 (pp. 79–82).

- Тöмörтоgoo, Domijn (1971), 'Dundad üeijn mongol hèlnij ègšig aviany ižilšil' [Vowel assimilation in Middle Mongolian], *StMo* 8: fasc. 15 (pp. 191–202).
- ——(1980), Orčin cagijn mongol hèlnij urt ègšigijn garal huv'sal [The origin and development of long vowels in modern Mongolian] (Ulaanbaatar).
- ——(1985), *Mongol hèlnij hos ègšig, tèdgèèrijn garal, huv'sal* [Mongolian diphthongs, their origin and development] (Ulaanbaatar).
- ——(1987*a*), 'Mongol hèlnij *i*-gijn nugaral gègčijn tuhajd' [The so-called breaking of *i* in Mongolian], *StMo* 12(20): fasc. 9 (pp. 149–54).
- (1987*b*), 'Mongol hèlnij üe bütèèh balarhaj ègšig' [Syllable-forming reduced vowels in Mongolian], *HZS* 21: fasc. 13 (pp. 87–90).
- ——(1987c), 'Mongol hèlnij üeijn bütèc huv'sal' [Changes of syllable structure in Mongolian], StMo 13(21): fasc. 12 (pp. 83–91).
- (1990), 'The consonant weakening in the Secret history of the Mongols', Mongolica, 1(22), 146-50.
 - —(1992), Mongol hèlnij tüühèn hèlzüjn ündès: Mongol hèlnij tüühèn avia züj, nègdügèèr dèvtèr [Principles of Mongolian historical linguistics: Mongolian historical phonology, 1st vol.] (Ulaanbaatar).
- ——(1995), 'Dundad üeijn mongol hèlnij hoër ègšigijn hoorondoh šurgèh gijgüülègčijn sularal' [A study of intervocalic fricatives in Middle Mongolian], *Monggolhak (Mongolian studies)*, 3: 219–24.
- [TÓNG Jīnróng] Tüng Jin-žüng (1998), 'Qorcin aman ayalgun-daki jarim abiyan-u öbermice qubiral-i dökom ögülekü ni' [A note on some characteristic sound changes in the Horchin dialect], MKUJ 1998/3: 76-80.
- TRETIAKOFF, A. (1975), 'Comparaison des lois de succession des voyelles en turc et en mongol', *PIAC* 14: 285–97.
- TSUMAGARI Toshirô (1985), 'Dagûrugo Hairaru hôgen no on'in taikei' [The phoneme system of the Hailar dialect of Dagur], *HBK* 17: 227–40.
 - (2003), 'Dagur', in Janhunen (2003*f*), 129–53.
- TUGTAMBAYAR, L. (2002), 'Monggol töröl keled-ün üge-yin ergülte-yin bayidal-i sinjilekü ni' [Analysis of word stress in the Mongolian languages], *ÖMYSS* 2002/2: 20–7.
- TULGAGURI (1994), 'Dumdadu erten-ü Monggol kelen-deki bogoni egesig abiyan-u sistèmün tölöbsirel' [The formation of the short vowel system in Middle Mongolian], MKUJ 1994/6: 4–16.
- ——(1996), 'Qorcin aman ayalgun-u egesig abiyalaburi-yin oncalig' [Characteristics of the vowel phonemes in the Horchin dialect], ÖMÜBDS 1996/4: 49–62.
- ——(1997), 'Qorcin aman ayalgun-u abiyalaburi-yin sistèm-ün oncalig' [Characteristics of the phoneme system of the Horchin dialect], *MKUJ* 1997/2: 4–25.
- TÜRGEN, B. (1978), 'Ulus-un dotogadu-yin Monggol kelen-ü nutug-un ayalgu-yi qubiyarilaqu ba saguri ayalgu bolun barimjiy-a abiyan-u tuqai asagudal' [The problem of the dialect division of Mongolian within the country, the basic dialect and the standard pronunciation], ÖMYSS 1978/3: 33–54.
- ——(1985), 'Orcin cag-un Monggol kelen-ü Caqar barimjiy-a abiyan-u abiyalaburi-yin bütüce' [The phoneme system of the modern Mongolian Chahar standard pronunciation], *MKB* 1985/2: 8–22; 1985/3: 9–25.
- (1988), 'Orcin cag-un Caqar aman ayalgun-u abiyan-u qubiral' [Sound changes in the modern Chahar dialect], in *Erdem sinjilgen-ü ögülel-ün tegübüri*, 1: 77–103 (Běijīng: Mínzú).
- UBRJATOVA, Elizaveta Ivanovna and Iraida Jakovlevna Seljutina (eds.) (1988), *Fonetika i grammatika jazykov Sibiri* [Phonetics and grammar of Siberian languages] (Novosibirsk: IIFF).

UBUŠAEV, Nikolaj Nadbitovič (1969*a*), 'Torgutskij dialekt i ego otnošenie k kalmyckomu literaturnomu jazyku' [The Torguud dialect and its relation to the Kalmuck literary language], *ZKEl* 7: 186–98.

—(1969*b*), 'O nekotoryh osobennostjah torgutskogo dialekta (po materialam èkspedicii v Justinskij rajon)' [Some characteristics of the Torguud dialect (based on data from an expedition to the Justa region)], *ZKEl* 7: 230–3.

- ——(1969c), 'Ikicohurovskij govor torgutskogo dialekta' [The Ikitsohur variety of the Torguud dialect], ZKEl 7: 234–8.
- —(1970), 'Nekotorye osobennosti caatanskogo podgovora' [Some characteristics of the Tsaatan subdialect], in 320 let starokalmyckoj pis'mennosti, 167–77 (Èlista: KNIIJaLI).
- ——(1973), 'O vozniknovenii dolgih glasnyh v mongol'skih jazykah' [The development of long vowels in Mongolian languages], in *Lingvogeografija, dialektologija i istorija jazyka*, 232–8 (Kišinev: Štiinca).
- ——(1979), Fonetika torgutskogo govora kalmyckogo jazyka [The phonetics of the Torguud dialect of Kalmuck] (Èlista: KKI).
- [ÜJÜGÜR] Wūzhū'ěr (1987), 'Dáwò'ěryŭ' [Dagur], ZSMY 592-608.
- ULAGANTUYAG-A (1997), 'Urad aman ayalgun-u egesig jokicaqu yoson-u tuqai' [Vowel harmony in the Urad dialect], *MKUJ* 1997/4: 71–81.
- ULANOVA, Alevtina Èrdni-Gorjaevna (1987), 'O nekotoryh osobennostjah mongol'skogo jazyka jugo-vostočnoj Tuvy' [Some characteristics of Mongolian in southeast Tuva], in Solncev (1987), 2: 156–63.
 - —(1988), Osobennosti mongol'skogo jazyka jugo-vostočnoj Tuvy [Characteristics of the Mongolian language of southeastern Tuva], Avtoreferat kand. diss (Ulan-Udè: BION).
- —(1992), 'K sravnitel'no-istoričeskomu issledovaniju ojratskih jazykov i dialektov' [Comparative-historical investigation of Oirad languages and dialects], in V. M. Solncev (ed.), VI Meždunarodnyj kongress mongolovedov: doklady rossijskoj delegacii, 2: 179– 86 (Moskva: Institut vostokovedenija RAN).
- ULTAN, Russell (1973), 'Some reflections on vowel harmony', *Working papers on language universals*, 12: 37–67 (Stanford University).
- ÜNENCI, D. (1987), 'Ersü üsüg i-yin tuqai' [The neutral vowel i], MKB 1987/1: 14-17.
- UUDA (1982), 'Caqar aman ayalgu-yin abiyalaburi-yin sistèm-ün jarim oncalig' [Some characteristics of the phoneme system of the Chahar dialect], *ÖMYSS* 1982/4: 184–209.
- VACEK, Jaroslav (1996), 'Dravidian and Mongolian: summary of results', Archiv orientální, 64: 31–46.
- VAGO, Robert (1973), 'Abstract vowel harmony systems in Uralic and Altaic languages', Language, 49: 579–605.
 - (ed.) (1980), Issues in vowel harmony (Amsterdam: John Benjamins).
- VANDUJ, Èrdènijn (1961), 'Mongol utga zohiolyn hèlnij avialbar (fonema)' [The phonemes of literary Mongolian], *HZS* 1/6–12: fasc. 6 (pp. 3–17).
- ——(1962), *Dörvödijn aman ajalguuny zarim onclogoos* [Some characteristics of the Dörbed dialect] (*HZS* 2: fasc. 3 (pp. 1–37)).
- (1964), 'Zavhany Sartuul ba tüünij aman ajalguu' [The Zavhan Sartuul and their dialect], HZS 2/10–19: fasc. 17 (pp. 189–211).
- —(1965), Dörvöd aman ajalguu [The Dörbed dialect] (Ulaanbaatar: ŠUA).
- (1967), 'Halhyn dornod dèd aman ajalguu' [The Eastern sub-dialect of Halh], *StMo* 6/15-21: fasc. 15 (pp. 3-17).
 - (1970), 'Halhyn ajalguu, tüünij ügsijn san' [The Halh dialect and its vocabulary], in Š. Barajšir (ed.), *Halhyn ajalguu*, 4–132 (Ulaanbaatar: ŠUA).
 - —(1973), 'K voprosu klassifikacii govorov i podgovorov v halhaskom narečii' [The problem of classification of dialects and sub-dialects of Halh], *OUMÈ 2 IH* 1: 82–7.

- VANDUJ, Èrdènijn (1978), 'Ügijn èhnij zarim gijgüülègč gèègdèž bajgaa n'' [The loss of some word initial consonants], *HZS* 13: fasc. 6 (pp. 54–6).
- VERBA, Nelli Karlovna (1980), 'Mongol'skie pis'mennye sočetanija V + j + i i ih reprezentacija v sovremennyh mongol'skih jazykah' [V + j + i clusters in written Mongolian and their representation in modern Mongolian languages], VGS 40–52.
- VIETZE, Hans-Peter and Ludwig Zenker (1976), Rückläufiges Wörterbuch der mongolischen Sprache (Leipzig: Verlag Enzyklopädie).
- VITALE, Guido Amedeo and Félix Édouard de Sercey (1897), *Grammaire et vocabulaire de la langue mongole (dialecte des Khalkhas)* (Péking: Imprimerie des Lazaristes).
- VLADIMIRCOV, Boris Jakovlevič (1923), 'O dvuh smešannyh jazykah zapadnoj Mongolii' [Two mixed languages in western Mongolia], *Jafetičeskij sbornik*, 2: 32–52.
- ——(1929), Sravnitel'naja grammatika mongol'skogo pis'mennogo jazyka i halhaskogo narečija: vvedenie i fonetika [Comparative grammar of written Mongolian and the Halh dialect: introduction and phonetics] (Leningrad: Izd. Leningradskogo vostočnogo instituta imeni A.S. Enukidze).
- WALKER, Rachel (1993), 'A vowel feature hierarchy for contrastive specification', *Toronto* working papers in linguistics, 12/2: 179–97.
- (1994), 'Buriat syllable weight and head prominence', *Phonology at Santa Cruz*, 3: 99–110.
- ——(1995), 'Mongolian stress: typological implications for nonfinality in unbounded systems', *Phonology at Santa Cruz*, 4: 85–102.
- (1996), 'A third parameter for unbounded stress', *NELS* 26: 441–55.
- WÁNG Pénglín (1989), 'On the emergence of geminate consonants in Dagur', Working Papers in Linguistics (University of Hawaii at Mānoa), 21/2: 91–104.
- (1992a), 'On limitations of Ramstedt's hypothesis concerning the Middle Mongolian initial h-', in B. H.-K. Luk (ed.), *Contacts between cultures: Eastern Asia, literature and humanities*, 3: 380–84 (Lewiston, NY: Edwin Mellen Press).

— (1992*b*), 'On the origin of the Middle Mongolian initial *h*- and the motivation for its loss', *Archív orientální*, 60: 389–408.

WEIERS, Michael (1969), Untersuchungen zu einer historischen Grammatik des präklassischen Schriftmongolisch (Wiesbaden: Harrassowitz).

- (1970*a*), 'Zu den langen Vokalen in der Moghol Sprache', ZASB 4: 467–73.
- ——(1970b), 'Zur Frage des Verhältnisses des Altmongolischen zum Mittelmongolischen', in L. Ligeti (ed.), *Mongolian studies*, 581–90 (Budapest: Akadémiai kiadó).
- ——(1972), Die Sprache der Moghol der Provinz Herat in Afghanistan (Sprachmaterial, Grammatik, Wortliste) (Opladen: Westdeutscher Verlag).
- ——(1978), 'Bemerkungen zur Phonetik der dagurischen Mundart von Butha', ZDMG 128: 125–60.
- (1992–3), 'Das Schriftmogolī der Mongolen von Afghanistan: ein Abriß', ZASB 23: 60–6.
- (2003), 'Moghol', in Janhunen (2003*f*), 248–64.
- WHITNEY, William Dwight (1862), 'On Lepsius's standard alphabet', *Journal of the Ameri*can oriental society, 7: 299–332.
 - —(1866), 'On the relation of vowels and consonants', *Journal of the American oriental society*, 8: 357–73.
- WHYMANT, Neville (1926), A Mongolian grammar outlining the Khalkha Mongolian with notes on the Buriat, Kalmuck, and Ordoss Mongolian (London: Kegan Paul, Trench, Trubner).

- WOOD, Sidney (1975), 'The weaknesses of the tongue arching model of vowel articulation', WPLund 11: 55–107.
- -----(1979), 'A radiographic analysis of constriction locations for vowels', *Journal of phon-etics*, 7: 25–43.
- WURM, Stephen A. and Lǐ Róng (main eds.) (1987), Language atlas of China (Pacific linguistics C/102) (Hong Kong: Longman).
- [Xī Cǎiyún] Si Tsai-yün (1995), 'Naiman aman ayalgu-yin abiyalaburi-yin oncalig-un tuqai' [Characteristics of the phonemes of the Naiman dialect], MKUJ 1995/3: 21–33.
- Xí Yuánlín (1986), 'Tŭzúyữ yīnwèi xìtŏng' [The phoneme system of Monguor], in Fù Màojì (ed.), *Zhōngguó mínzú yŭyán lùnwén jí*, 395–405 (Chéngdū: Sìchuān mínzú).
- XIÃO Sùyīng (1995), 'Zhālŭtè qí Měnggůhuà yīnyùn' [The phonology of Mongolian in Jaruud banner], BIHP 66/1: 107–244.
- [XIÙHUĀ] Siu-quwa (2002), 'Küriy-e aman ayalgu bolon bicig-ün kelen-ü abiy-a-lug-a-yin qaricagulul' [Comparison of the sounds of the Hüree dialect and the written language] ÖMÜYS 2002/2: 55–64.
- YAMADA Norio (1983), 'A constraint on phonological variables', *Linguistic analysis*, 12: 29-84.
- YAN Qíxiāng and Zhōu Zhízhì (1995), Zhōngguó Mèng-Gāomián yǔzú yǔyán yǔ Nányà yǔxì [Mon-Khmer languages in China and the Austroasiatic languages] (Běijīng: Zhōngyāng Mínzú Dàxué).
- [YI Ki-mun] Lee Ki-Moon (1964), 'Mongolian loan-words in Middle Korean', UAJb 35: 188–97.
- [-----] Lee Ki-Moon (1991), 'Mongolian loan-words in Korean', AH 3: 35-44.
- [Yù Róng] Iü Rüng (2002), 'Monggol kelen-ü ündüsün busu urtu egesig-ün tuqai' [Nonbasic long vowels in Mongolian], *MKUJ* 2002/6: 51–3.
- Yù Shìcháng (1983), Lùn Měnggǔ yǔzú de xíngchéng hé fāzhǎn [The formation and development of the Mongolian languages] (Běijīng: Mínzú).

—(1985), 'Měnggǔ mìshǐ zhōng yuánchún yuányīn de hànzì biǎoshìfǎ' [The method of denoting rounded vowels with Chinese characters in the Secret history of the Mongols], Zhōngguó yǔyánxué bào, 2: 174–84.

- ŽAM'JAN, Galsangijn (1995), 'Mongolyn nuuc tovčoony hèl ba dörvöd aman ajalguu' [The language of the Secret history of the Mongols and the Dörbed dialect], in V. I. Rassadin (ed.), Istoriko-sravnitel'noe izučenie mongol'skih jazykov, 165–72 (Ulan-Udè: BNC).
- ŽANČIV, Ëndonžamcyn (2002), *Dörvölžin üsgijn mongol dursgal* [Mongolian documents in the 'Phags-pa script] (Ulaanbaatar: National University of Mongolia).
- ŽARGALOV, Aleksandr Sandykovič (1986), Intonacionnoe oformlenie kommunikativnomodal'nyh raznovidnostej obŝego voprosa v burjatskom jazyke v sopostavlenii s russkim (èksperimental'no-fonetičeskoe issledovanie) [The intonational formation of communicative and modal varieties of yes-no questions in Buriad in comparison with Russian (experimental phonetic investigation)], Avtoreferat kand. diss (Ulan-Udè: SO AN SSSR).
 - (1987*a*), 'Sopostavitel'naja harakteristika intonacii obŝih voprosov v burjatskom i russkom jazykah' [Comparative characteristics of the intonation of yes-no questions in Buriad and Russian], in Rassadin (1987*a*), 50-71.
 - —(1987*b*), 'Intonacionnye kontury kommunikativno-modal'nyh raznovidnostej obŝego voprosa v burjatskom jazyke (èksperimental'no-fonetičeskoe issledovanie)' [Intonation contours for communicative and modal varieties of yes–no questions in Buriad and Russian], in Rassadin (1987*b*), 63–71.

ŽARGALOV, Aleksandr Sandykovič (1988), 'Intonacija modal'nyh raznovidnostej obŝego voprosa v burjatskom jazyke' [The intonation of modal variants of yes–no questions in Buriad], in Ubrjatova and Seljutina (1988), 118–25.

— (1989), 'Intonacionnoe vyraženie problematičeskoj i kategoričeskoj modal'nosti v voprosiltel'nyh vyskazyvanijah burjatskogo jazyka' [Intonational expression of problematic and categorial modality in interrogative utterances in Buriad], in Šagdarov (1989), 101–12.

ZHĀNG Yàguāng and B. Shàobù (1990), 'Měnggǔyǔ Dù'ěrbótè huà de tèdiǎn' [Characteristics of the Dörbed dialect of Mongolian], MY 1990/2: 53–55, 61.

ZHÒNG Sùchún (1965), 'Dáwò'ěryǔ gàikuàng' [Sketch of Dagur], ZY 1965/4: 317-35.

——(1980), 'Dáwò'ěryŭde yuányīn héxié' [Vowel harmony in Dagur], MY 1980/4: 18–27.
 ——(1982), Dáwò'ěryŭ jiănzhì [Short description of Dagur] (Běijīng: Mínzú).

ZIMMER, Karl E. (1967), 'A note on vowel harmony', International journal of American linguistics, 33: 166–71.

ZOLHOEV, Valentin Ignat'evič (1961), 'O treh tipah udarenija v mongol'skih jazykah' [Three types of stress in Mongolian languages], *TBurNII* 6: 105–15.

—(1962), 'Udarenie v burjatskom i russkom jazykah' [Stress in Buriad and Russian], *Trudy kafedr russkogo jazyka vuzov vostočnoj Sibiri i Dal'nego Vostoka*, 2: 149–55 (Irkutsk).

— (1963a), 'Slogoobrazujuŝie soglasnye v burjatskom jazyke' [Syllabic consonants in Buriad], in *Kratkie soobŝenija o naučno-issledovatel'skih rabotah za 1961 god*, 282–5 (Irkutsk: Irkutskij gosudarstvennyj universitet).

- ——(1963b), 'Fonologičeskie funkcii burjatskih glasnyh' [Phonological functions of Buriad vowels], in *Kratkie soobŝenija o naučno-issledovatel'skih rabotah za 1961 god*, 286-8 (Irkutsk: Irkutskij gosudarstvennyj universitet).
- (1963c), 'Pozicionnye ograničenija pri upotreblenii burjatskih soglasnyh' [Position constraints on the use of Buriad consonants], in *Otčetnaja konferencija Vostočno-Sibirskogo soveta po koordinacii i planirovaniju gumanitarnyh nauk: tezisy dokladov i soobŝenij*, 174–8. (Irkutsk).

(1970), 'Udarenie v mongol'skom jazyke' [Stress in Mongolian], TBION 13: 59-63.

-----(1972), 'Sočetaemost' soglasnyh i redukcija glasnyh v burjatskom jazyke' [Consonant phonotactics and vowel reduction in Buriad], *VBF* 1: 18–24.

(1973a), 'Sil'nye i slabye pozicii glasnyh fonem v burjatskom jazyke' [Strong and weak positions of vowel phonemes in Buriad], *VBF* 2: 44–51.

——(1973b), 'Sil'nye i slabye pozicii soglasnyh fonem' [Strong and weak positions of consonant phonemes], VBF 2: 52–62.

——(1973c), 'Fonetičeskie sredstva členenija potoka reči na slova v burjatskom jazyke' [Phonetic means of dividing the speech stream into words in Buriad], VBF 2: 29–43.

— (1977), 'Različija v slogodelenii v burjatskih govorah i literaturnom jazyke' [Differences in syllable division in Buriad dialects and the literary languge], in C. B. Cydendambaev (1977), 89–94.

——(1980), *Fonologija i morfonologija aggljutinativnyh jazykov* [The phonology and morphophonology of agglutinative languages] (Novosibirsk: Nauka).

— (1989), 'Ob oppozicijah fonem po ih differencial'nym priznakam (na materiale mongol'skih glasnyh)' [Phoneme oppositions according to their distinctive features (based on data from Mongolian vowels)], in Šagdarov (1989), 13–31.

—(1991), 'Oppozicii mongol'skih soglasnyh po ih differencial'nym priznakam' [Mongolian consonant oppositions according to their distinctive features], in Šagdarov (1991), 35-56.

296

- International standard ISO 9: 1995(E), information and documentation: Transliteration of Cyrillic characters into Latin characters, Slavic and non-Slavic languages. 2nd edn. (Genève). 1995.
- Monggol Kitad toli (nemen jasagsan debter); Měng–Hàn cídiăn (zēngdìngběn) [Mongolian–Chinese dictionary (enlarged and revised edition)] (Kökeqota: ÖMYS-yin keblelün qoriy-a). 1999.

— Xīn Měng–Hàn cídiăn; Sin-e Monggol Kitad toli; Šinè Mongol–Hjatad tol' [New Mongolian–Chinese dictionary] (Běijīng: Shāngwù yìnshūguǎn). 1999.

This page intentionally left blank

INDEX

/A/ 191

Aalto, Pentti (1917-98) 225 Abaga, Mo dial. (WM Abag-a, Mo Avga) 226 Abašeev, Daniil Altaevič (1890–1966) 227 Abel-Rémusat, Jean Pierre (1788–1832) 218 Abematsu Gen'ichi 精松源一(1903-) 220 Aberle, David Friend (1918-) 97, 219, 221, 226, 228 Adam, Lucien (1833-1918) 220 [advanced tongue root] 8, 43, 220 affricates 12-13, 119-21, 200-1, 224 Afghanistan 140, 154 Aga, Bur dial. (WM Ag-a, Bur, Ru Aga) 141, 144-5, 227 Agvangdandar (Ngag-dbang-bstan-dar) (1758-1830) 7 Aha, Bur dial. (Bur Aha, Ru Oka) 141, 145, 227 Ahn, Mee-Jin (An Mi-jin) 97 Ainu xvii Akan 8 Alair, Bur dial. (WM Alayir, Bur Alajr, Ru Alar') 141.145.227 Alekseev, Dmitrij Andrianovič (1908-73) 227 Alshaa (WM Alaša(n), Mo Alšaa) 148 Oir dial. 141-2, 148-9, 194, 200-1, 228 Altaic languages xvii Anderson, John Mathieson (1941-) 44, 222 Anderson, Stephen R. 189, 222 Áng Qí 昂奇 228 Angkuic languages 155 [apical] 45 Arabic 107, 154 Arabic Mongolian 98-9, 107-8, 111, 115-17, 120-1, 123-4, 127, 186, 223-4 Arai Daisuke 荒井大輔 251 Arai Shin'ichi 荒井伸-186 Archangeli, Diana B. 45, 222 archiphoneme 47-8, 53 Ardajab, Eldengtei-yin (1936-) 225, 228 Arhangai, Mo dial. 226 Arhorchin, Mo dial. (WM Aru Qorcin, Mo Arhorčin) 143, 226 Armenian 98 aspiration 12-20, 25, 120, 124, 198-9, 204, 206-8, 216, 221 aspiration flip-flop 207-8 assimilation i 28, 32, 192, 194, 195-7, 200-2, 208-10, 229

rounding 114-15, 194-5 u 197 velar nasal 57-8 [ATR], see [advanced tongue root] Austin, William Mandeville (1914–) 97, 219, 221, 226, 228 Austroasiatic languages 155, 216 Baarin, Mo dial. (WM Bagarin, Mo Baarin) 11, 97, 123, 141-4, 157, 180, 182-5, 195-6, 201, 205, 211-14, 226 Baarin right banner 143 Baatar, Jadambatyn (1957-) 226 Babuev, Sergej Damdinovič (1949–) 227 [back] 43, 220 Badamdorž, see Badmadorž Badarangg-a 228 Bādárónggē 巴达容哦, see Badarangg-a Badgaev, Nikolaj Boktaevič 200, 202, 210, 225, 228 Badmadorž, Dügèržavyn 228 Badmaev, Andrej Radnažapovič 222, 226 Baigal-Hudari, Bur dial. (Ru Bajkalo-Kudara) 145, 227 Baikal, lake 144-5 Baitchura, see Bajčura Bajarsajhan, Ëndonžamcyn xvi Bajarsürèn, Cèrèngijn 30 Bajčura, Uzbek Š. (1923–) 17, 62, 223 Balagunova, Sandulma Sanžievna (1947-) 223, 227 Baldan, Luvsangončigijn 228 Baldanžapov, Purbo Baldanovič (1921–91) 7 Bálint, Gábor (1844-1913) 227 Banzargarudi 226 Bǎo Cháolǔ 保朝鲁, see Bulucilagu Bào Huáiqiào 鲍怀翘 218, 220, 223 Bāo Lìgāo 包力高, see Bulag Bāo Xiánglín 包祥林 148, 202 Bao'an, see Bonan Baonan, see Bonan Barangova, Nudla Nikolaevna 223 Bargazhan, Bur dial. (WM Bargujin, Bur Bargažan, Ru Barguzin) 141, 145, 227 Bargu, Bur dial. (WM Bargu) 141, 144-5, 201, 227 Batugerel 228

- Batujirgal, N. (1942–) 223, 226
- Batusayiqan 182, 226, 229

Bawden, Charles Roskelly (1924-) xvi Bayad, Oir and Mo dial. (WM Bayad, Mo Bajad) 148, 226, 228 Bayanbagatur 224-5 Bayancogtu (1934-) 9, 210, 218, 220, 223-4, 226, 229 Bayar 224 Bayarmendü, Borjigin (1959-) xvii, 30, 218, 223.226 Bayasqal 228 Bažeeva, Tat'jana Petrovna (1950–) 8, 146, 227 Beffa, Marie-Lise (1943-) 97, 216, 218 Běnrén 本人, see U Benzing, Johannes (1913-2001) 187, 225, 228 Bernhardt, Karl A. 62 Bertagaev, Trofim Alekseevič (1905-76) 30, 216, 225, 227, 229 Bese, Lajos (1926-88) 226-7 Beskow, Jonas (1970-) 1 Bimba 228 Binnick, Robert Ira (1945-) 189, 216-17, 222 Birtalan, Ágnes 228 Bitkeev, Petr Cedenovič (1937-) xvii, 22, 97, 149, 187, 208-9, 214, 221-2, 226, 228-9 Bjuraeva, Èleonora Ivanovna (1939–) 93, 223, 227 Bläsing, Uwe 228 Bobrovnikov, Aleksandr II'ič (1793–1832) 97, 218 - 19Bobrovnikov, Aleksej Aleksandrovič (1821– 65) 97, 124, 218-20, 227, 229 Boersma, Paul 1 Bohan, see Boohon Böke (1929-) xviii, 151-4, 202, 205, 226, 229 Bolbasural, see Bulucilagu Bold, Luvsandoržijn 228 Bonan (Baonan; Ch Bǎo'ān 保安) 113, 140-1, 151, 153, 157, 180, 183, 188-90, 196-8, 200, 202-4, 206-8 Boo Liyan-ciyün 226 Boohon, Bur dial. (WM Bogan, Bur Boohon, Ru Bohan) 141, 145, 227 Booij, Geert Evert 223 Boosiyang (1934-) 202, 227 Bormanshinov, Arash (Araš Bormanžinov) (1922-) 227 Bosson, James Evert (1933-) 23, 97, 200, 210, 226, 228 boundary tone 88-9, 91-3 breaking 194, 196-7; see also assimilation: i; palatalization: consonant Budaev, Bal'žinima Žambalovič (1940–) 223, 227 Budaev, Cyrendaša Badmaevič (1923–) 142, 144, 226-7 Buhaeva, Irina Doržievna 30, 222 Buhaeva, Oktjabrina Doržievna 78

Bùhé 布和, see Böke Bulag, Borjigin (1938-) 123, 152, 225, 229 Bulagad, Bur dial. (WM Bulagadu, Bur Bul(a)gad, Ru Bulagat) 141, 145, 227 Bulu 226 Bulucilagu (1929-) 18, 150-1, 188, 205, 224, 229 Buraev, Ignatij Dmitrievič (1929-) xvii, 8, 78, 120, 142, 144, 146, 199, 202, 220-3, 226 - 9Buriad (WM Buriyad, Bur Burjaad, Ru Burjat) 17-18, 30, 34-5, 40, 96-7, 123-4, 140-2, 144-6, 147, 157, 180, 182-7, 189-90, 194-205, 208-10, 216, 221-3, 226-7 Cyrillic script 34-5, 40, 145 dialect division 144-5, 227 Eastern 141, 145, 202, 227 North-western 145, 227 phonetics 17-18, 40, 96-7, 221-3 South-western 145, 227 Western 141, 145 Buriatia xvii, 142, 144 Buthaa, Dag dial. (WM Budg-a) 149-50, 202 Buzaawa, Oir dial. 148, 228 Cagangada (1934-) 226, 228 Čagdurov, Sergej Šagžievič (1928–) 226 Caspian Sea 147 Castrén, Matthias Alexander (1813-52) 222, 227 -če, indirect past suffix 52-3 Cegen 226 Čemesov, Ivan 227 Cenggeltei (1924-) xvii, 8, 63, 96, 142, 148, 151, 198, 200, 210, 216, 218-22, 224-6, 228 - 9Cenggeltü, D. 226 Čeremisov, Konstantin Mihajlovič (1899– 1982) 146, 227 Cèrènsodnom, Dalantain (1937-) 127 Chágān 查干, see Caganqada Chágānhādá 查干哈达, see Cagangada Chahar, Mo dial. (WM Cagar, Mo Cahar) 11, 17, 123, 140-4, 157, 180, 182, 185, 195-6, 201, 206, 209, 211-12, 214, 221, 226 Chaolu Wu, see Chuluu Charette, Monik 65 Chén Jiāyóu 陈嘉猷 (1941-) 218 Chén Nǎixióng 陈乃雄 (1933-2002) xviii, 153, 155, 202, 226 China xvii-xix, 34, 40, 140-1, 143-6, 154 Chinchor, Nancy Ann 222 Chinese 102-8, 133, 140, 150-3, 155, 216 loan-words 23, 28-30, 48 Yuan dynasty Chinese (Early Mandarin) 102-7 Chinggis Khan (?1162–1227) 99, 102, 108

Chinggis Khan's stone 100, 126

300

Ch'o Mi-hŭi 222 Chuluu, Üjiyediin 150-3, 228 Cimeg 226, 229 cingg-a (čanga) 'tense, strong' 7, 220 Cinggeltei, see Cenggeltei Cinggis qagan-u ijagur 102 Clark, Larry V. (1943-) 142, 200, 202, 227 Classical Written Mongolian xvii, 23, 52, 100, 117, 121-3, 126, 140, 145, 216; see also Mongolian script Clauson, Gerard Leslie Makins (1891–1974) xvii, 98, 225 Clear script 145, 148 Clements, George N. 45, 63, 67 coda 62-72, 124-5, 203-5, 216 Cogbayar 226 Cogtu (1933-) 200, 228 Cohen, David A. 222 Cole, Jennifer (1961-) 56 Coloo, Žavzangijn (1937–) xviii, 73, 75, 96–7, 142, 148, 218, 221-2, 225-6, 228-9 Comrie, Bernard (1947-) 225 consonants 12-21, 25-30, 118-25, 197-210 Buriad 17-18, 146, 221-2 Chahar 17, 221 features 45 Kalmuck 17-18, 149, 221 Covijongjab (1931–) xvii, 9, 17, 23, 59, 102, 122, 127, 218, 221, 223, 225-7 Cüngkeria 228 Cybikov, Gombožab Cèbekovič (1873–1930) 97, 219-20 cyclicity syllabification 73-6, 223 vowel harmony 56-8, 190, 193-4 Cydendambaev, Cybikžap Boboevič (1915–83) 227Cydenžapov, Širap-Nimbu Rinčinovič (1943–) 124Cyrendoržieva, Cyrendulma Šogdupovna 227 Cyrenov, Babasan Doržievič 145, 227 Cyrillic script Buriad 34-5, 40, 145 Dagur 150 Kalmuck 35, 40, 75, 148-9, 187 Mongolian 4, 6-7, 9, 13, 21, 24, 30, 34-40, 50, 73, 75-6, 78 Dagur (Daur; WM Dagur; Ch Dáwò'ěr 达斡尔) 113, 140-1, 149-51, 154-7, 180-6, 189-90, 194-200, 202-4, 208-10, 216, 224, 228 - 9Dàhéjiā 大河家, Bon dial. 153 Dambueva, Polina Petrovna 223 Damdinov, Dašinima Galdanovič (1925-) 146, 227

Damdinova, Dašima Darižapovna 227

Damdinsürèn, Cèndijn (1908-86) xvi, 51, 73, 75.222 Danzan 226 Danzandagba (bsTan-'jin-grags-pa) 7 Dàobù 道布, see Dobo Darbeeva, Anna Angadykovna (1917-) 30, 98, 124, 178, 189, 202, 224-5, 227-9 Darhad, Mo dial. (WM Dargad, Mo Darhad) 141-3, 226 Dari 154 Dariganga, Mo dial. (WM Darigangg-a, Mo Dariganga) 141-3,226 Dašcèdèn, Tümenbajaryn 224 Daur, see Dagur Dawa, Idomjo-a (Idomuso) 伊·达瓦 (1964-) xvii, 218 Dawadagba (1934-) 280 deaspiration 18, 142-3, 197, 205-7, 229 Deed Mongol, Oir dial. (WM Degedü Monggol, Mo Dèèd Mongol), see Hoshuud deletion of short vowels 185-9, 200, 202, 204-5, 208-11, 216 Délìgé'ěrmǎ 德力格尔玛 (1955-) 218 Demirdache, Hamida Khadiga 222 Denwood, Margaret Ann 65, 222 depalatalization 141-3, 200-2 derivation 50, 75 devoicing 14-16, 19 Dīng Shíqìng 丁石庆 228 diphthongs 9-11, 20-4, 28, 30, 37, 45, 47, 50, 59, 76-8, 111-12, 185, 187, 214 directionality 68-9 Dixon, Robert Malcolm Ward (1939-) 217 Djamouri, Redouane 23-4, 55, 189, 218, 220, 222 - 3Dobo (1934–) 62, 65, 100, 126, 143, 211–12, 218, 220, 226, 228-9 Doerfer, Gerhard (1920-) xvii, 98, 123, 146, 151, 216, 224-5, 227 domain of vowel harmony 52-3 Don Kalmucks 148 Dondukov, Ulzy-Žargal Šojbonovič (1923–) 228 Dongxiang 东乡, nationality 155; see also Santa Dongxiang autonomous county 155 Dörbed, Oir dial. (WM Dörbed, Klm Dörvd, Ru Derbet) 17, 141, 148-9, 179, 183, 228 Heilongjiang, Mo dial. 143, 148, 202 Dorjicang 228 Dorž, D. 236 Doržiev, Agvan (1853-1938) 145 Dravidian languages xvii duration consonant xvii, 13-15, 17 vowel 1-4, 24, 94, 96, 187, 218 Dyrheeva, Galina Aleksandrovna (1951-) 78, 227

302

INDEX

/E/47-9.53 East Yugur, see Shira Yugur Ebleltü, B. 62 Edkins, Joseph (1823-1905) 219 Edgurivagci 226 Efimov, Valentin Aleksandrovič (1933-) 154 Ehired, Bur dial. (WM Ekiridü, Bur Èhirèd, Ru *Èhirit*) 141, 145, 227 /Ei/ 47-9. 51. 56-7 Ejnee, Oir dial. (WM Ejen-e) 148 em-e 'female' 7, 220 emphatic consonants 108 Engkebatu, Merten (1936-) xviii, 150, 186, 228English 30,96 Ēnhébātú, Mò'ěrdīng 莫尔丁·恩和巴图, see Engkebatu epenthetic consonant 55-6, 76-7 epenthetic vowel (reduced vowel, schwa) 1, 3, 5-7, 23-5, 32, 36, 44, 57, 62-78, 146, 148-50, 186-7, 189, 204, 209 Erdenibagatur 226 Erdenicugla 150, 223 Erdenimöngke 226, 229 er-e 'male' 7, 219-20 ersü 'hermaphrodite' 7 Esenova, Tamara Sarangovna (1953-) xvii, 23, 93, 222-3, 228 Evenki 146, 224 Ewen, Colin J. 44, 222 expressive word 30, 50 feature 43-6, 54, 191 feature spreading 53-6, 190-4 Field, Kenneth L. (1961-) 152 final prominence tone 90–1 Finch, Roger 105, 225 Finland xvii focal accent 53, 85-9, 91, 93 focus marker 85, 89, 91 formant frequencies xvii, 1, 3-11, 179 Franzén, Vivan (1950-2004) 23, 218 fricatives 18-19, 121-4, 202-3 fricativization 198-202 Fufubâtoru, see Kökebagatur Fukejiruto, see Kögjiltü fundamental frequency 85, 94, 96 Gaelic 16 Galsan, Sèrèènèngijn (1931-) 15, 214 Gangtogtaqu, see Gantogtoh Gānsù 甘肃 150, 152-3, 155 Gansu-Qinghai linguistic area 140 Gantogtoh, Gončigijn 200, 227, 229 Garudi 188, 229 genioglossi muscles 43-4 Georg, Ralf Stefan 151, 226

Georgian 98 Gerasimovič, Ljudmila Konstantinovna (1923-) 97 Gereltü, N. (1954–) 149, 194, 210, 228 glides 20, 76-8, 118, 203 Goldsmith, John A. (1951-) 76, 97, 222 Gorlos, Mo dial. (WM, Mo Gorlos) 143, 202, 226 Grassmann, Hermann Günther (1809-77) 206 Grassmann's law 206 Gregerson, Kenneth J. 189, 220 Grivelet, Stéphane 34 Grønbech, Kaare (1901-57) 96 Grunzel, Josef (1866-1934) 218, 220, 229 -gui, negation suffix 52-3, 92 Guō Shǒuxiáng 郭守祥 30 Guw-a, D. 19, 224, 226 h 98, 113, 120-4, 183-5, 187, 196, 202-3, 216, 225 Haenisch, Erich (1880-1966) 218 Hagiwara Shôzô 萩原正三 (1914-) 218 Hailar (Ch Hǎilā'ěr 海拉尔) 146, 149 Dag dial. 149-50, 202, 228 Hākānchǔlún 哈勘楚倫, see Qagancilagu Halh, Mo dial. (Khalkha; WM Oalg-a, Mo Halh) xvi-xviii, 1-98, 113-14, 116, 120-1, 123, 133, 135, 140-8, 178-216, 221, 226 Central 6, 142-3, 182 Eastern 142-3, 221, 226 Gobi 142-3, 226 Northern 141, 143, 148, 206, 226 Southern 141, 143, 221, 226 Western 143, 221, 226 Halle, Morris (1923-) 97, 222 Haltod, Matthew Magadbürin (1917-75) 24,97 Hamayon, Roberte (1939-) 97, 216, 218 Hambis, Louis (1906-78) 218 Hammond, Michael Theodore 97 Hamp, Eric Pratt (1920-) 222, 224 Hangin, John Gombojab (1921-89) 221, 226 Harada Ryûji 原田龍二 (1956-) 223 Harchin, Mo dial. (WM Qaracin, Mo Harčin) 141–3, 182–3, 226 Harnud, Huhe, see Köke Hasbaatar, see Qasbagatur Hāsībāgēn 哈斯巴根, see Qasbagan-a Hāsī'è'ěrdūn 哈斯额尔敦, see Oaserdeni Hattori Shirô 服部四郎 (1908-95) xviii, 26, 78, 97, 105, 110, 120, 122-3, 133, 189, 218-21, 224 - 9Hayes, Bruce (1955-) 97 Hazara 154 Hé Jírén 和即仁 155 Hēilóngjiāng 黑龙江 149

Helgason, Pétur (1965-) 16

Helimskij, Evgenij Arnol'dovič 208 Herat 154 Heshigten, Mo dial. (WM Kesigten, Mo Hèšigtèn) 143 hiatus avoidance 55-6 Higuchi Kôichi 樋口康一 (1951-) 100 Hilok, Bur dial. 145 Hilvat al-insān wa halbat al-lisān 107 History of the Yuan dynasty (Yuán shǐ 元史) 99 Hmong-Mien languages 216 Hölönbuir (WM Kölön Buyir, Mo Hölönbujr) 144-5 Homonov, Mihail Petrovič (1913-95) 227 Hong Sŏng-hun (Sung-Hoon) 222 Horchin, Mo dial. (WM Qorcin, Mo Horčin) 59, 141-3, 200-1, 226, 228 Hori, Bur dial. (WM Qori, Bur Hori) 17, 141, 145 - 6.227Hoshuud, Oir dial. (WM Qošud, Mo Hošuud) 141, 148-9, 228 Hotgoid, Mo dial. (WM Qotogoyid, Mo Hotgojd) 141-3, 148 Houtsma, Martijn Theodoor (1851–1943) 223 Hsiao, Stella Su-ying, see Xiao Suying Huá-Yí yì yǔ 華夷譯語 102, 106, 127 Huáng Zōngjiàn 黄宗鉴 225 Hūgéjílètú 呼格吉勒图, see Kögjiltü Húgéjīntái 胡格金台, see Kögjimtei Hugjiltu, see Kögjiltü Hūhé 呼和. see Köke Huí 🗉 (Chinese Muslims) 155 Hulst, Harry van der (1953-) 43, 53-4, 222 Hung, William Shih-hao (Hóng Shìháo 洪士豪) (1893-1980) 223 Hüree, Mo dial. (WM Küriy-e, Mo Hürèè) 143, 226 Hùzhù 互助 151 Mgr dial. 151-2, 204 Hyman, Larry (1947-) 97 hyoglossi muscles 8, 43-4 *i* 6, 8–9, 21, 23–4, 26, 28–9, 32, 37–9, 45–6, 49-50, 54-5, 76-8, 116-17, 120-1, 191-3, 195-7, 200-2, 208-10, 222, 229 -i, optative suffix 76–8 ibn-Muhannā, Jamāl al-Dīn 107, 127 Ibrahim, A. 152, 155 Icelandic 16 Ichinose Megumi - ノ瀬恵 (1957-) 52, 223, 226.229 Idam, J. 226, 229 Idsardi, William James 97 Ih Minggan, Mo dial. (WM Yeke Minggan, Mo *Ih Mjangan*) 143, 226 Ilan, J. 226 Ili (Ch Yīlí 伊犁) 149 Iliškin, Ivan Kuznecovič 221

Indo-European languages 206 inflection 50.75 information structure 90 'Inner Mongolia'. see South Mongolia Inner Mongolian dialect 148 intonation 85-96, 223 intrinsic duration 3 intrinsic pitch 218 IPA transcription xv, 100, 155, 157 Iran 154 Iranian languages 99, 154 Irincin, Yekeminggadai (1931-99) 224 ISO transliteration 34-5 Itô Junko 伊藤順子 68 lü Rüng, see Yu Rong Ivalga, Bur dial. (Bur Ivalga, Ru Ivolga) 145, 227 *j* 11, 37–8, 61, 76–8, 196, 203 -j, optative suffix 76-8 Jagunasutu (1934–) xviii, 109–10, 150–1, 204 Jahontova, Natal'ja Sergeevna (1955-) 216, 218 Jainca (Ch Jiānzhā 尖扎) 153 Jalaid, Mo dial. (WM Jalayid, Mo Zalajd) 143, 202.226 Jalcib 65, 226 Jalsan (1947-) 151, 224 Janhunen, Juha (1952-) 120-1, 123, 142, 144, 147, 155, 197, 188, 210, 225-9 Japan xviii Japanese xvii Jaruud, Mo dial. (WM Jarud, Mo Zaruud) 141-3.226 Jensen, John Thayer (1942-) 222 Jiǎ Xīrú 贾晞儒 (1936-) 228 Jiǎlāsēn 贾拉森, see Jalsan Jīn Gāng 金刚 218, 220 Jirannige, B. 227 Jirannige, Jigucidai (1945-) 111, 229 Jirim, Mo dial. (WM Jirim, Mo Žirèm) 226 Jirüken-ü tolta-yin tayilburi 7 Jīshíshān 积石山, Bon dial. 153, 183 Jôo Hakutarô 城生伯太郎 (1946-) xviii, 62, 218, 220-1, 223, 226 Kabansk, Bur dial. 145, 227 Kachug, Bur dial. (Ru Kačug) 145, 227 Kadai languages 216 Kakudô Masayoshi 角道正佳 (1948-) 24, 65, 151-2, 223, 228 Kalmuck, Oir dial. (WM Qalimag, Klm Hal'mg, Ru Kalmyk) xvii, 17-18, 23, 34-5, 40, 75, 97, 123, 140, 147-9, 179-80, 183-5, 187, 189-90, 195-6, 199-201, 204, 209,

211-14, 221, 223

Cyrillic script 34-5, 40, 75, 148-9, 187

304

Kalmuck, Oir dial. (cont.) dialect division 148 phonetics 17-18, 97, 221, 223 schwas 18, 23, 40 Kalmuckia xvii, 147-8 Kalmucks 147-8 Kałuzyński, Stanisław (1925-) 229 Kamiyama Kôhei 神谷衡平 219 Kamnigan (WM Qamnigan, Mo Hamnigan) 123, 140-1, 144, 146-7, 180, 183-5, 190, 194-6, 199-204 Kāngjiā 康家 140-1, 153-4, 157, 180, 183, 188-90, 194-200, 202-5, 207-8 Kānglè 康乐, ShY dial. 151 Kāngyáng 康杨 153 Kara, György (1935–) 142, 150, 201, 226, 228 - 9Karakol, Oir dial. 148, 228 Karlsson, Anastasia Mukhanova (Anastasija Ivanovna Muhanova) (1972-) 13, 85, 93, 206, 223 Kaun, Abigail Rhoades 222 Kazakstan 147 Kelen-ü cimeg 7 Kenstowicz, Michael J. 97 Keyser, Samuel Jay (1935-) 45 Khalkha, see Halh Khatso 155 Kičikov, Anatolij Šalhakovič (1921–) 148, 228 - 9Kim Chu-wŏn (Kim Ju-won) 金周源 (1956-) 189, 210, 219, 222, 229 Kim Pang-han 金芳漢 150-1, 188, 224 Kim Sŏk-tŭk (Kim Suk Deuk) 金錫得 224 Kim, Stephen S. 152 Kiparsky, Paul (1941-) 224 Kitāb majmū' tarjumān Turkī wa 'Ajamī wa Mughulī wa Fārsī 107, 127, 223 Kitan 154-5 Kljukin, Innokentij Adrianovič (1889–) 96, 219Kobayashi Takashirô 小林高四郎 (1905-97) 223, 225, 229 Kobavashi Tetsunori 小林哲則 251 Kögjiltü, Uriyangqai (1949-) xviii, 23, 110, 153, 195, 210, 224, 226, 228-9 Kögjimtei, Alhal 228 Kőhalmi, Katalin (Käthe) Uray-(1926-) 146 Kojima Takeo 小島武男 218-20 Kök Mungchak, see Tuvan Kök Nuur, Oir dial. (WM Kökenagur, Mo Höhnuur), see Hoshuud Köke, Qarnud (1962–) xvii, 12, 17–19, 23, 62, 97, 218, 221, 223, 226 Kökebagatur, Borjigin T. 229 Kökebars (1948-) 22, 62, 223-4, 226, 229 köndei (höndij) 'hollow' 7, 220

Konjaeva, È. I., *see* Bjuraeva Korean xvii, 97-8, 219 Kotvič, Vladislav Ljudvigovič, see Kotwicz Kotwicz, Władysław (1872–1944) 55, 150, 188, 218, 227 Kovalevskij, Osip Mihajlovič see Kowalewski Kowalewski, Józef Szczepan (1801-78) 97, 218 Krippes, Karl Anthony 225 Krueger, John Richard (1927–) 227 Kubo Tomoyuki 久保智之 58 Kuldinow, Sodman 224 Kürelbagatur, Uriyangqadai (1961-) 224, 226, 229 Kuribayashi Hitoshi 栗林均 (1951–) xviii, 10, 23, 102, 127, 184, 186, 197, 210, 214, 222-3, 226, 228-9 Kuz'menkov, Evgenij Afanas'evič (1945-) 106-7, 151, 154, 224-5, 228 Kyrgyzstan 148 # 19-20, 30, 59 k 19-20, 203 [labial] 45 labialization 23, 59, 150, 189, 194, 197 Ladd, D. Robert (1947-) 85 Lánzhōu 兰州 152 Latin alphabet 34, 145, 148, 150-2 Leben, William Ronald (1943-) 222 Lee Ki-Moon, see Yi Ki-mun Leiden Manuscript, see Kitāb maimū' tarjumān Turkī wa 'Ajamī wa Mughulī wa Fārsī lexical phonology 56 Lǐ Bīng 李兵 222, 224 Li, Charles N. (Lǐ Nè 李讷) (1940-) 153, 155 Lǐ Kèyù 李克郁 (1936-) 151, 204 Lǐ Měilíng 李美玲 (1964-) 151, 229 Lǐ Róng 李荣 (1920-) 140 Liáo 遼 dynasty (916-1125) 154 Lieber, Rochelle (1954-) 222, 224 Ligeti, Lajos (1902-87) 110, 122, 154, 224 Lightner, Theodore (1934-) 224 Lindau, Mona (1943-) 8 Línxià 临夏 153 liquids 19-20, 203-4 Liú Zhàoxióng 刘照雄 (1932-) 152-3, 228 loan-word 30-3, 35, 46 Chinese 23, 28-30, 48 English 30 Manchu 30 Russian 30-3 Sanskrit 30, 124 Tibetan 19-20, 30, 48 Loloish languages 155 long vowels (development) 183-4, 203, 216, 229; see also primary long vowels; vowel: length

Lǚ Shìnán 吕士楠 220

Lubsangwangdan, see Luvsanvandan Luóbùsāngwàngdān 罗布桑旺丹, see Luvsanvandan Lutaiirgal, Abgan 25, 205 Luvsanbaldan, Haltaryn (1925-92) 182, 225, 228Luvsandèndèv, Amgaagijn (1927-97) 30 Luvsanvandan, Šadavyn (1910-83) xviii, 9, 22-3, 25, 53, 96, 218, 221-2, 224-5, 228 - 9Mǎ Guóliáng 马国良 152 Manalžav, see Manlažav Manchu 30, 150 Mandžieva, Kermen Uljumandžievna 93 Mangghuer, Mgr dial., see Minhe Man'kovo, Kmn dial. 147 Manlažav, Luvsanvandangijn 107, 223-4 Mansang, Ö. (1935-) 226 Marantz, Alec 58-9 Martin, Samuel Elmo (1924-) 228 Matheev, Boris Vasil'evič (1928-82) 227 Mawkanuli, Talant 122 maximality 68-9 Měnggǔ mìshǐ (Měnggǔ bìshǐ) 蒙古秘史, see Secret history of the Mongols Mènghébǎoyīn 孟和宝音, see Möngkebuyan Mergel, river 146 merger back and front vowels 180-1, 198 *i and *e 6, 182 palatalized and original vowels 211 *y and *ø 145, 183 Messerschmidt, Daniel Gottlieb (1685-1735) 227 metathesis 186 Middle Mongolian 98 Mikami Satoshi 三上司 (1951-) 23, 52, 58, 223 Miller, Gerhard Friedrich (1705-83) 227 Mínhé 民和 151 Mgr dial. 151-2, 183, 204 Mišig, Lhagvažavyn 146 Mitroškina, Anastasija Grigor'evna 227 Mižiddorž, G. 236 Modern Written Mongolian 40-2, 100, 117, 126 Moghol 123, 140, 154, 180, 183-5, 190, 196-8, 202, 224 Mohosoeva, Matrena (Viktorija) Matveevna 93, 223 Mongghul, Mgr dial., see Huzhu Monggol Kitad toli 42, 143 Monggoljin, Mo dial. (WM Monggoljin, Mo Mongolžin) 143, 226 Monggol-un niguca tobciyan, see Secret history of the Mongols

Mönggöngerel (1965-) 59, 206, 226

Möngkebuyan (1953-) 122, 150, 182, 195, 224 - 6, 229Mongolia, Republic of ('Outer Mongolia', Mo Mongol) xvii-xix, 34, 140-2, 144-8 Bur dial. 145, 227 Mongolian, modern (WM Monggol, Mo Mongol) xvi-xviii, 1-98, 113-16, 120-1, 123, 133, 135, 140-8, 150, 178-216, 221, 225-6dialect division 141-3, 225-6 Eastern 141-3, 178, 182-3, 185, 205-6, 210, 212, 214, 226; see also Baarin Southern 141, 143, 182, 185, 214, 226; see also Chahar Mongolian script 34, 40-2, 98-100, 117 Mongolic languages xvi-xviii, 140-77, 225-9 classification 216-7 Monguor 107, 113, 140-1, 151-2, 157, 180-4, 188, 190, 196-8, 202-4, 206-8, 210, 224, 229 Mönh-Amgalan, Jumžirijn 6, 142, 182, 225 monophthongization 185, 211-2, 214 monosyllabic words 63-4 Montgomery, Robert Walker 145 Möömöö, Sürèngijn xvii, 6, 8, 22, 25, 96, 142, 182, 218, 221-3, 225 mora 78-9, 86-8, 92-3 Morin Dawaa (Ch Mòlìdáwă 莫力达瓦) 149 Mostaert, Antoine (1881–1971) 41–2, 151, 226 Mostaert transcription 41-2 Mudrak, Oleg Alekseevič 151 Muhar-Sheber, Bur dial. (WM Muqursiber, Bur Muhar-Šèbèr, Ru Muhoršibir') 145 Muniev, Bembja Džalykovič 149 Muqaddimat al-adab 107-8, 127 Murayama Shichirô 村山七郎 (1908-95) 107, 122. 224. 228 Mušaev, Vladimir Naranovič 223 Nabcingerel 229 Nadamid (1935-) 152, 226, 229 Nadeljaev, Vladimir Mihajlovič (1911-85) 218, 220, 222 Nàdémùdé 那德木德, see Nadamid Nadmid, Žamsranžavyn (1923–91) 218 Nagy, Lajos J. 225 Naiman, Mo dial. (WM Naiman, Mo Najman) 59, 141–3, 206, 226 Nakajima Takehiro 中?雄大 251 Namtsarai (1926-) 228 Nangrub 224, 228 Naranbatu, Ö. (1940–) 226 nasals 18-19, 57-8, 67, 124, 203-5 nasalization 18-19, 58 Nàsēnbǎi 那森柏, see Nasunbayar Nasunbayar, Borjigin (1925-87) 152-3, 218, 226

306

Nasunöljei 229 natural language processing xvii, 218 Navrockij, N. 227 Naxi 155 neutral (transparent) i 26, 49-50, 54, 192-3, 195. 222 Niándūhū 年都乎, Bon dial. 153 Nizhneudinsk, Bur dial. 141, 145, 227 Nolmajab, S. (1937-) 226 Nomintuyag-a 226 Nomura Masayoshi 野村正良(1914-91) xviii, 151, 216, 224-6, 228-9 Nonni, river (Mo Noon, Ch Nèn Jiāng 嫩江) 149 Norjin, Tserin-ü (1936–) 22, 214, 220 nucleus 62-3 Nugteren, Hans 150, 216 /0/ 4.22 /0/ 191 Obata Jûichi 小幡重一 (1888-1947) xvii, 218-19 occlusion phase 13-15 Odden, David Arnold (1955-) 97, 222, 225 Oirad (WM Oyirad, Klm Öörd) xvii, 17-18, 23, 34-5, 40, 75, 97-8, 115, 123, 140-2, 145, 147-9, 178-80, 183-7, 189-90, 194-204, 208-9, 211-14, 216, 221, 223, 226-9: see also Kalmuck Oka. see Aha Ôkawa Shigeki 大川茂樹 218 Old Mongolian xvi, 18, 28-9, 42, 50, 76, 98-139, 141-2, 145, 147-8, 150, 178-216, 223-5, 228-9 Ol'hoon, Bur dial. (WM Olgon, Bur Ol'hoon, Ru Ol'hon) 145, 227 Öliei 226 Öljeyibürin (1959-) 148, 185, 226 Öljeyitogtaqu 226, 229 Omakaeva, Èllara Uljaevna 40, 149, 202 Ongniud, Mo dial. (WM Ongnigud, Mo Ogniud) 143 onomatopoeics 30 Onon, river 146 Onon, Peter M. Urgunge (1919-) 221 onset 62-3, 67, 69-70, 118, 216 Ööld, Oir dial. (WM Ögeled) 148, 228 opaque /U/ 50-1 [open] 44-6, 53, 178, 182, 191 opening of *u and *y 148, 183, 229 Ordos, Mo dial. (WM, Mo Ordos) 141-3, 182, 205 - 6, 226Orlov, Aleksandr Matveevič 97, 220 Orlovskaja, Marija Nikolaevna (1926-) 218 Osa, Bur dial. 145, 227 Osada Natsuki 長田夏樹 (1920-) 228 Osor, B. (1935-) xvi, 51 'Outer Mongolia', see Mongolia, Republic of Ōuyáng Juéyà 欧阳觉亚 (1930-) 140, 155

Övörhangai, Mo dial. 226 Ôvabu Shôtarô 大薮鉦太郎 218 Oyunceceg 228 Ozawa Shigeo 小沢重男 (1926-) 123 p^h 30, 123-4, 207, 225, 229 [palatal] 43-5, 50, 178-9, 190-2, 194, 213 palatalization consonant 9-10, 12, 18, 20-1, 23-4, 26, 28-9, 32, 36, 38, 50, 60-1, 142-3, 150, 189, 194-5, 197, 201, 208-12, 216, 222 loss 142-3, 210-11 vowel 10-11, 142-3, 178, 199, 210-14 Palaungic languages 155 Pallas, Peter Simon (1741-1811) 227 Pao Kuo-yi, see Unensečen partial geminate 67 Pavla Dorž. see Pavlov Pavlov, Dordži Antonovič (1912-97) xvii, 23, 149, 185-7, 221, 223, 228-9 Pavlova, Elena Semenovna (1936-) 8, 93, 146, 223, 227 Pèljei 25, 205, 226 Pelliot, Paul (1878-1945) 123 Persian 216 'Phags-pa Lama (bLo-gros-rgyalmtshan) (1235-80) 109 'Phags-pa Mongolian 98-9, 108-11, 113, 116-17, 120-1, 123-4, 127, 133 [pharyngeal] 8, 26, 43-6, 53-7, 178-9, 182, 190 - 3.213pharyngeal constrictors 8, 43-4 pharyngealization 178-81, 190, 193, 195 phonemes 22-33 phonetic realization rule 44-5 Pjurbeev, Grigorij Cerenovič (1940-) 147, $223, \bar{225}$ place feature 44, 54, 191 Podgorbunskij, Innokentij Aleksandrovič 221, 227 Poirot, Jean (1873-1924) xvii polarization 181-2 Popov, Aleksandr Vasil'evič (1808-80) 227 Poppe, Nikolaj Nikolaevič (Nikolaus, Nicholas) (1897-1991) xvii, xviii, 17, 22, 24, 34, 42, 88, 96-8, 100, 107, 109-11, 115, 120, 122-4, 126-7, 142, 178, 184, 196-7, 204, 212, 216, 218-28 Posch, Udo (1922-65) 218, 227, 229 Pozdneev, Aleksej Matveevič (1851–1920) 97 Praat 1, 13, 85 preaspiration 12-18, 120, 204, 206 Pre-Classical Mongolian 100 Pre-Mongolic 28, 32, 123 primary long vowels 107, 111, 113, 122-3, 224 Prince, Alan 97 Pritsak, Omeljan (1919-) 154, 224

privative feature 44 prosodic phrase 85, 91 Proto-Altaic 123 Proto-Mongolic 98-9, 113, 117, 122-4, 183, 217, 224-5 Prževal'sk, see Karakol Pulleyblank, Douglas George 222 Pulleyblank, Edwin George (1922-) 102, 105, 107 punctuated equilibrium 217 Pürbü 227-8 Qadacilagu 226 Qagancilagu, Qarnud (1926-97) 218, 228 Qai Jüng-yan 226 Qasbagan-a (1957-) 186, 188, 198, 224, 226, 228 - 9Qasbagatur (1946–) 151, 188, 226 Qaserdeni, B. (1934-) 96, 195, 214, 218, 220-1, 224-6, 228 Qayirgan 226, 229 Qīnggé'ěrtài 清格尔泰, see Cenggeltei Qīnghǎi 青海 148, 150-1, 155, 201 Oir dial., see Hoshuud Qīnglóng 青龙, ShY dial. 151 Qiqihar (Ch Qíqíhā'ěr 齐齐哈尔) 149 Dag dial. 149, 228 Qubilai (1216-94) 108 Quèjīngzhābù 确精札布, see Coyijongjab question 78, 85, 93-5 Rachewiltz, Igor de (1929-) 98, 102, 115, 223 Radnaev, Èrhito Radnaevič (1930–) 226–7 Radnaeva, Ljubov' Dašinimaevna xvii, 9, 218 Rahmn, Cornelius (1785-1853) 227 Ramsey, Samuel Robert (1941-) 225 Ramstedt, Gustaf John (1873–1950) xvii, xviii, 9, 12, 17, 19, 22, 55, 96-8, 122-3, 125, 154, 178, 194, 202-3, 206, 210, 218-19, 221-2, 224-5, 229 Ramstedt-Pelliot's law 123 Rao, Calyampudi Radhakrishna (1920-) 6 Rassadin, Valentin Ivanovič (1939-) xviii, 200, 202, 210, 222, 227-9 reduced vowel, see epenthetic vowel reduplication 58-61 resyllabification 74, 216, 223 [retracted tongue root] 8 rheme 90 rhyme 62, 69-70 Rialland, Annie 23-4, 55, 189, 218, 220, 222-3 Rigaloff, see Rygaloff Rinčen, Bjambyn (1905-77) 36, 146, 218, 225 Rincindorji 224 Róna-Tas, András (1931–) 151, 226 [round] 44-6, 53-4, 178, 191 rounding assimilation 114-15, 194-5

rounding harmony 46-50, 53-4, 56-7, 115, 146-7. 149-51. 190. 194. 222 Rozycki, William Vincent (1949-) 225 [RTR], see [retracted tongue root] Rudnev, Andrej Dmitrievič (1878-1958) 96, 202, 210, 216, 218-22, 226-7 Russia xviii, xix, 40, 144, 146-8 Russian 13, 17, 30-4, 96, 140, 147 Rybatzki, Volker 216-17, 228 Rygaloff, Alexis (Aleksej Nikolaevič Rygalov) (1922–) 22 Rygdylon, Èrdemto Rinčinovič (1906–57) 227 sagarmag 'neutral' 7 Šagdarov, Lubsan Doržievič (1930–) 145, 227 Saitô Yoshio 斎藤純男 (1958-) xviii, 19, 22-3, 25, 57, 73, 76, 108, 151, 186, 222-4, 228 Sajan, Bur dial. 145 Salemann, Carl Gustav Hermann (Karl Germanovič Zaleman) (1849-1916) 219 Sambuudorž, Očirbatyn 228 San Yüwè 226 Sandža, see Sanžaa Sangdügüreng 226 Sanskrit 30, 124 Santa (Ch Dongxiang 东乡) 140-1, 152, 155-7, 180, 183-4, 188-90, 196-200, 202-8 Sanžaa, Žamsranžavyn (1947–) 25–6, 218, 221-2, 228-9 Sanžeev, Garma Dancaranovič (1902-82) 30, 63, 65, 78, 96, 142, 148, 178, 185, 187, 210, 218-22, 224-7, 229 Sanžina, Darima Dabaevna (1949-) 226 Saran 226 Šarhüü, Če. 226 Sart Kalmuck, Oir dial., see Karakol Sartuul, Mo dial. (WM Sartagul, Bur, Mo Sartuul) 141-5, 226 Satô Nobuharu 佐藤暢治 150, 228-9 Schane, Sanford A. 44 Schein, Barry 67, 222 Schmidt, Isaak Jakob (1779-1847) 96, 219 Schmidt, Peter, see Šmits Schnitscher, Johan Christian 227 schwa, see epenthetic vowel schwa ~ zero alternation 71-2 Schwarz, Henry G. (1928-) 152, 225 Secen, W. 226, 228 Secenbagatur, B. (1936-) 226 Secenbilig 228 Secencogtu (1951-) 153, 216 Secret history of the Mongols 102, 107, 127, 223 Selenge, river (WM Selengge, Bur, Mo Sèlèngè) 142 North Selenge, Bur dial. 145, 227 South Selenge 'Buriad', Mo dial. 142-5, 226

308

Ŝerbak, Aleksandr Mihajlovič (1926–) 204 Sercey, Félix Édouard de 219-20 Serengnorbu, B. 182, 226, 229 Ševernina, Zoja Vasil'evna (1928–) 236 Shàobù, B. 波·少布 148, 202 Sharufû. see Šarhüü Shīdiàn 施甸 155 Shiliin Gol, Mo dial. (WM Sili-yin Gool, Mo *Šilijn Gol*) 141–3, 179–80, 182 Shimizu Mikio 清水幹夫 62, 222-3 Shira Yugur (East Yugur; Yellow Uigur; Ch Dōngbù Yùgùyǔ 东部裕固语) 113, 140-1, 150-1, 153, 157, 180, 184, 188-90, 194, 198, 202-5, 207-8 Shirai Katsuhiko 白井克彦 218 Shirokogoroff, see Širokogorov shortening of long vowels 183, 187 Shuluun Höh, Mo dial. (WM Silugun Köke, Mo Šuluun Höh, Ch Zhènglán 正蓝) 141, 143, 185, 211-12; see also Chahar Sì bù cóngkān 四部叢刊 102 Si Tsai-yün, see Xi Caiyun Siberia 147 Sìchuān 四川 155 Sìjiǎjí 四甲集, San dial. 152 Sinedke (1927-) 8, 219-20, 224 Sino-Mongolian 98-9, 102-7, 111-13, 116-17, 120-1, 123-4, 127, 133, 223 Sino-Tibetan languages 216 Singuwa 226, 229 Sīgīncháokètú 斯钦朝克图, see Secencogtu Širokogorov, Sergej Mihajlovič (1887– 1939) 225 Siu-quwa, see Xiuhua Sjölander, Kåre 1 Skribnik, Elena Konstantinovna 227 Slater, Keith William (1965-) 151 Smedt, Albert de (1884-1941) 151 Smith, Norval 222 Šmits, Pēteris (1869–1938) 228 Sodobagatur 23, 226 Sogdian 199 Soktoeva, Svetlana Purbuevna (1952-) 17-19, 146, 210, 222, 227 Sonam, B. 228 Song Chae-mok 宋在穆 58,62 Sòng Lián 宋濂 (1310-81) 99 Sǒng Paeg-in 成百仁 228 Šongqor (1955–) 225, 228 Sönid, Mo dial. (WM Sönid, Mo Sönöd) 141-3, 226 sonority 67 Soulié de Morant, Charles Georges (1878-1955) 226 South Mongolia ('Inner Mongolia', WM Öbör Monggol, Ch Nèi Měnggǔ 内蒙古) 34, 182, 140-4, 148-9

standard language 140, 143, 225-6 Southeast Asian linguistic area 216 Southern 'Buriad', Mo dial. 143-4 Soviet Union xvii, xviii, 30, 140 Sovol 226 split of *i 179, 182, 229 Steriade, Donca 67, 222 Stong-Jensen, Margaret 222 stops 12-18, 119-21, 198-200, 203-4, 220-1, 224 - 5Stralenberg, Philip Johan (orig. Tabbert) (1676-1747) 227 Street, John Charles (1930-) 22-4, 97, 123, 149, 218, 220, 222-3, 226 stress 94-7, 122, 223; see also focal accent Stuart, Don Graham 24, 97 Sturt, Reginald W. 97, 225 styloglossi muscles 43-4 Sùběi 肃北, Oir dial. 228 Sühbaatar, Cègmèdijn (1942-) 62, 218 Sühbaatar, Osornamžilyn (1945–) 236 Sum" jaabaatar, Baldandoržijn (1937-) 236 Sūn Zhú 孙竹 (1931-95) 23, 143, 149, 194, 201, 206, 210, 212, 214, 218, 220-2, 225-6, 228 Süngrüb, see Cogtu Suǒnánbà 销南坝, San dial. 152 Sürüng, Ni. 228 Svantesson, Jan-Olof (1944-) 8-10, 13, 23, 34, 43, 58, 62, 65, 73, 76, 97, 120, 148, 155, 179, 189, 206, 216-18, 220, 222 syllabification 62-76, 222-3 cyclic 73-6, 223 syllable structure 62-8, 118, 203, 214-16 Syrian 99 -t, plural suffix 74–5 Takeuchi Ikunosuke 竹内幾之助 218 Tángjiācūn 唐家村 155 Tángwāng 唐汪 155 target segment 54 Tariach, Mo dial. (WM Tariyaci, Mo Tariač) 226 Tatarinov, Mihail 227 Tǎtǎtǒng'ā 塔塔統阿 99 Taube, Manfred 127 Taya, D. 228 Temür 226, 229 Tenišev, Èdgem Rahimovič (1921-) 228 Teshima Takehiko 豊島武彦 xvii, 218-19 Thomsen, Kaare (1924-97) 123, 184, 224, 229 Tiān Fēng 天峰 229 Tibetan 108, 110, 153, 155, 216 loan-word 19-20, 30, 48 Tibetan nationality 155 Tibeto-Burman languages 155 timing tier 45

Tivanfèng, see Tian Feng Todaeva, Buliaš Hoičievna (1915–) xviii, 63, 96, 149-53, 188, 197, 200, 204, 218, 220-2, 225-6, 228 Togog-a (1937-) 210, 226 Tömörcèrèn, Žogdovžavyn 23, 78, 210, 214, 220, 222, 224, 226-7, 229 Tömörtogoo, Domijn (1938-) xviii, 98, 125, 178, 185, 214, 224-6, 229 Tóng Jīnróng 佟金荣 226 Tōnghǎi 通海 155 Tóngrén 同仁 153, 155 Bon dial. 153 Torguud, Oir dial. (WM Torgud) 141, 148-9, 183, 200-1, 228 transcription and transliteration xv, 34-5, 99-101, 140, 155-7 transparent i, see neutral i Tretiakoff, A. 222 Tsaatan, Oir dial. (WM Catan, Mo Caatan) 228 Tsendina, Anna (Anna Damdinovna Cendina; Damdinsürèngijn Dulmaa) (1954-) Tsongool, Mo dial. (WM Conggool, Bur Congool) 141-5, 222 Tsumagari Toshirô 津曲敏郎 (1951-) 228 Tǔ ± nationality 153, 155; see also Monguor Tugnui, Bur dial. 145 Tugtambayar, L. 223 Tulgaguri (1955-) 224, 226 Tümd, Mo dial. (WM Tümed, Mo Tümd) 143 Tüng Jin-žüng, see Tong Jinrong Tungusic languages xvii, 123-4, 224 Tünhen, Bur dial. (WM Tüngkin, Bur Tünhèn, Ru Tunka) 141, 145, 227 Tuōdáyèwá 托达叶娃, see Todaeva Türgen, B. (1930-) 143, 226 Turkic languages xvii, 99, 107, 117, 122, 124, 224, 229 Turkmen 224 Tuva, Mo dial. in 226, 229 Tuvan, Junggar 122 U (Ch Wū 乌) 155 /U/ 47-8, 50, 53, 191 *w* 111, 117–18, 210, 224 va 23, 30, 59 vai 23, 30, 62 Ubušaev, Nikolaj Nadbitovič 228-9 /Ui/47-8 Uigur Mongolian 98, 99-101, 102, 111, 113, 117, 120-1, 123-6 Uigur script 34, 40-1 Uigurs 99 Üjemchin, Mo dial. (WM Üjümücin, Mo Üzèmčin) 141, 143, 226 Üjügür 228 Ulaanbaatar xvi, 6, 11, 50-1, 182, 206

Ulaanchab, Mo dial. (WM Ulagancab, Mo Ulaancav) 141-3 Ulagantuyag-a 226 Ulan-Ude (Ru Ulan-Udè, Bur Ulaan Üdè) 145 Bur dial. 226 Ulanova, Alevtina Èrdni-Gorjaevna 226, 228 Ultan, Russell 224 underspecification 44, 46, 191 Ünenci, D. 229 Unensečen, B. (Pao Kuo-vi) (1916-) 23, 200, 210, 226, 228 Ungi, Bur dial. (WM Unggi) 145 United States xviii universal default rule 45 unstable n 125-6, 205 Urad, Mo dial. (WM, Mo Urad) 141-3, 182, 226 Ural, Oir dial. 228 Uran-a 226 Urianhai, Oir dial. (WM Uriyangqai, Mo Urianhaj) 148, 228 Urul'ga, Kmn dial. 147, 183, 202, 204 -Ut, plural suffix 125 Uuda (1932-) 226 Vacek, Jaroslav (1943-) xvii Vago, Robert Michael 189, 212 Vanduj, Èrdènijn (1921-77) 218, 225-6, 228-9 [velar] 43-5, 55, 178-9, 191 velar ~ uvular consonants 12, 18, 28-9, 39, 45, 55-6, 115-18, 120, 198-200, 204, 222 velarization 178-82, 190 Verba, Nelli Karlovna 185 Vergnaud, Jean-Roger 97, 222 Vietze, Hans-Peter (1939-) xvi Vitale, Guido Amedeo 219-20 Vladimircov, Boris Jakovlevič (1884-1931) xvii, xviii, 22-3, 30, 96-8, 111, 115, 122, 124, 148, 178, 182-3, 210, 216, 218-20, 222, 224 - 5voice onset time (VOT) 14-15, 17, 221 vowel 1-11, 22-5, 111-18, 178-97, 210-14 alternation patterns 47-8 features 43-6, 54, 191 length 1-7, 22-24, 36-7, 40, 45, 59, 78, 87-8, 144, 146, 149-54, 183, 187, 214; see also long vowels; primary long vowels marked 45, 191 non-initial 1-3, 5-6, 23-5, 36, 40, 48-90, 53; see also epenthetic vowel; vowel harmony place of articulation 43-4 vowel harmony 7-8, 26, 32-3, 42-3, 45, 46-58, 96, 113-17, 144, 146-7, 149-51, 154, 188-94, 195, 212-14, 219-20, 222, 224 cyclic 56-8, 190, 193-4 diagonal 220 loss 189-90 palatal (back ~ front) 189-94, 213-14, 220

vowel harmony (cont.) pharvngeal 8, 46-50, 57, 190-4, 213-14, 220. 222 phonetic basis 7-8, 219-20 rounding 46-50, 53-4, 56-7, 115, 146-7, 149-51, 190, 194, 222 shift 43, 189-92, 194, 213-14 tense ~ lax 220 vertical 220 vowel shift 178-82, 211-13, 228-9 Walker, Rachel (1967–) 97, 222, 227 Wáng Pénglín 王鹏林 225, 228 Wāngjiācūn 汪家村 155 Wāngjiājí 汪家集, San dial. 152 Waves 1,4 WaveSurfer 1, 15 Weeninck, David 1 Weiers, Michael (1937-) 76, 98, 100, 126, 154, 228 Weijer, Jeroen Maarten van de (1965-) 53-4 White Mongol, see Monguor Whitney, William Dwight (1827-94) 67, 76 Whymant, A. Neville John 218, 226 Witsen, Nicolaas (1641–1717) 227 Wood, Sidney A. J. (1934-) 43-4 word structure 78-9, 203, 214-16 Written Mongolian, see Classical Written Mongolian, Modern Written Mongolian, Mongolian script Wú Chéngyì 吴丞义 153 Wurm, Stephen Adolphe (1922-2001) 140 Wǔtún 五屯 155 Wūzhū'ěr, Ōunán 欧南·乌珠尔, see Üjügür -x, future participle suffix 67, 75 Xī Cǎiyún 希彩云 226 Xí Yuánlín 席元麟 151 Xiāo Sùyīng 蕭素英 62, 222, 226 -xii, verbalizing suffix 53 Xīnjiāng 新疆 122, 148-9, 200-1 Dag dial. 149, 228

Xiùhuā 秀花 (1966-) 226

Yakut 224 Yamada, Norio 山田宣夫 (1947-) 222 Yán Qíxiāng 颜其香 (1933-) 155 Yányuán 盐源 155 Yellow Uigur, see Shira Yugur Yellow Yugur, see Shira Yugur Yesenova, see Esenova Yi Ki-mun 李基文 97 Yībùlāhēimài 伊布拉黑麦. see Ibrahim Yìlínzhēn 亦邻真, see Irincen Ysyk Köl 148 Oir dial., see Karakol Yù Róng 玉荣 185 Yù Shìcháng 喻世长 (1916-) 178, 189, 216, 224 - 5Yuán cháo mìshǐ 元朝秘史, see Secret history of the Mongols Yuán 元 dynasty (1271-1368) 108-9 Yúnnán 云南 155 Zahchin, Oir dial. (WM Jagacin, Mo Zahčin) 141, 148, 228 Zakamna, Bur dial. (Ru Zakamna) 145 al-Zamaxšarī, Abū'l Oāsim Mahmūd ibn-'Umar (1075-1144) 107 Žam'jan, Galsangijn (1936–) 223 Žančiv, Endonžamcvn 127 Žargalov, Aleksandr Sandykovič (1954–97) 93, 223Zava Pandita (1599-1662) 148 Zenker, Ludwig xvi Zhāng Yàguāng 张亚光 148, 202 Zhàonàsītú 照那斯图, see Jagunasutu Zhèng Yùlíng 郑玉玲 (1949-) 218 Zhòng Sùchún 仲素纯 228 Zhōngyuán yīnyùn 中原音韻 102 Zhōu Déqīng 周德清 (1277-1365) 102 Zhōu Yàowén 周耀文 140, 155 Zhōu Zhízhì 周植志 (1932-) 155 Zimmer, Karl E. 224 Zolhoev, Valentin Ignat'evič (1928-) xvii, 97,

146, 210, 222-3, 227

Zwick, Heinrich August (1796-1855) 227

INDEX OF OLD MONGOLIAN WORDS

- * $a\check{c}^{h}$ ihan 'burden' ($a\check{c}^{h}a$ ayaa) 128, 158, 184
- *ahula 'mountain' (*оођ* уул) 119, 122, 128, 144, 158, 184, 187, 198
- *aimak 'district' (*aiməg* аймаг) 125, 128
- *ak^ha 'elder brother' (*ax* ax) 128, 158, 198
- *alak 'motley' (*аξэg* алаг) 128, 158, 204
- *ali 'which' (*а*ξ^{*j*}аль) 128, 158, 209, 211
- *alima 'fruit' (*ақ^jәт* алим) 128, 158, 209
- *alt^han 'gold' (*аtt*^{*h*} алт) 112, 119, 125, 128, 147, 158, 180, 189, 198, 204, 208
- *aman 'mouth' (*am* ам) 28, 128, 158, 198, 211, 213
- *amin 'life' (*ат*^{*j*} амь) 28, 112, 119, 128, 158, 209, 211, 214
- *aŋkita 'separate' (*аŋgⁱt* ангид) 118, 128
- *ap 'to take' (aw ав) 125, 128, 158, 180, 204
- *arihun 'clean' (*ar^jvŋ* ариун) 128, 158, 184
- *arka 'method' (агәс арга) 128, 158, 186, 204
- *čahu 'to bite' (cuu 3yy) 128, 159, 184, 198, 201
- *čahun 'hundred' (cuu syy) 129, 159, 198, 200
- *čalahu 'young' (*са*<u></u><u></u>^{*t*} залуу) 122, 129, 142, 159, 184
- *čalki 'to swallow' (*сађ'д'* залги) 118, 129, 159, 204, 209
- *čaru 'to use' (*car* 3ap) 129, 159, 198
- *čiha 'to point' (caa 3aa) 129, 159, 184, 201
- *čikasun 'fish' (*cacəs* загас) 129, 159, 196, 198, 201
- *čil 'year' (čіђ жил) 129, 145, 159, 180, 204
- *čirkohan 'six' (*сигса* зургаа) 119, 129, 159, 184, 201
- *čiryken 'heart' (*curx* зүрх) 129, 159, 195–6, 201
- *čita 'spear' (čat жад) 129, 159, 182, 196, 200-1
- *čok^hi 'to fit' (*сэх*^{*j*}зохь) 112, 118, 119, 129, 159, 207
- *čøhelen 'soft' (соођәŋ зөөлөн) 129, 160, 184
- *čøp 'right' (соw зөв) 129, 160, 204
- *č^hahalsun 'paper' (*c*^haas цаас) 129, 160, 204, 206
- *č^hak 'time' (*c^hag* цаг) 125, 129, 160, 198, 200– 2, 204
- *č^hakahan 'white' (*c^hagaŋ* цагаан) 129, 160, 184, 198, 204

- *č^hasun 'snow' (*c^has* цас) 129, 160, 180, 189, 198
- *č^heč^hek 'flower' (*c^hic^həg* цэцэг) 129, 160, 204
- *č^herik 'soldier' (*c^hirэg* цэрэг) 119, 125, 129, 160, 198, 204
- *č^hik^hin 'ear' (*č^hix* чих) 116, 129, 160, 180, 189, 206–7
- *č^hilahun 'stone' (*č^hυξυ* чулуу) 129, 160, 184, 196, 201
- *č^hina 'to boil' (*č*^hап чана) 130, 160, 200, 202
- *č^hino 'wolf' (*č^hэп* чоно) 130, 160, 196
- *č^hiŋkis 'Chinggis' (č^hiŋgəs Чингэс) 117, 130
- *č^hisun 'blood' (*c*^hυs цус) 116, 130, 161, 192–3, 196, 201, 205–6
- *č^hitk^hør 'devil' (*č^hot^hgər* чөтгөр) 112, 116, 130, 161
- *eč^hike 'father' (*ic^həg* эцэг) 116, 130, 161, 201
- *ehyten 'door' (*uut* үүд) 112, 130, 161, 184
- *emehel 'saddle' (imet эмээл) 130, 161, 184
- *emys 'to wear' (*oms* өмс) 130, 161, 186, 188, 195, 198, 204
- *ene 'this' (*in* энэ) 130, 161, 180, 189, 198, 205
- *epesyn 'grass' (*ows* өвс) 101, 130, 161, 188, 198
- *ertem 'learning' (*irtəm* эрдэм) 119, 125, 130, 161, 204
- *erthe 'early' (*irth* эрт) 130, 161, 198, 204, 208
- *harpan 'ten' (агәw арав) 130, 161, 198
- *hič^he 'to be ashamed' (*ič^h* ич) 130, 161, 188, 196, 203, 207
- *hiruhar 'bottom' (*jэгэђ* ёроол) 130, 162, 184, 196
- *hiryher 'blessing' (*jогођ* ерөөл) 112, 130, 162, 184, 196
- *hoi 'forest' (лі ой) 77, 130, 162, 184
- *hon 'year' (*эŋ* он) 113, 119, 125, 130, 162, 180, 205
- *huhut^ha 'bag' (*uut^h* yyr) 130, 162, 184, 216
- *huja 'to tie' (*uj* уя) 77, 130, 162, 198
- *hulahan 'red' (*иђаŋ* улаан) 130, 162, 180, 184, 187, 198, 204
- *hyk^her 'ox' (*ихәr* үхэр) 130, 162, 183, 198, 207
- *hyneken 'fox' (ипәд үнэг) 130, 162, 198
- *hynesyn 'ash' (uns үнс) 130, 162, 198

- *ilka 'to separate' (*јађад* ялга) 131, 162, 196
- *imahan 'goat' (jama ямаа) 131, 162, 184, 196
- *inehe 'to laugh' (*ine* инээ) 131, 163, 184, 196, 198, 208
- *ire 'to come' (*ir* ир) 131, 163, 196, 198
- *irken 'people' (*irgəŋ* иргэн) 112, 131
- *ite 'to eat' (*it* ид) 131, 163, 182, 188-9, 196
- *japu 'to go' (*jaw* яв) 112, 115, 119, 131, 163, 198
- *jasun 'bone' (jas яс) 131, 163, 198
- *jek^he 'big' (*ix* их) 131, 163, 208
- *josun 'rule' (jos ëc) 131, 163, 198
- *kačar 'ground' (*Gacər* raзap) 119, 131, 163, 186, 189, 192, 198, 200, 204
- *kakč^ha 'single' (*Ganc^h* ганц) 131, 163, 198
- *kak^hai ʻpigʻ (*Gaxai* гахай) 112, 119, 131, 163, 185, 207
- *kal 'fire' (*са*5 гал) 131, 163, 198, 200, 204
- *kar 'hand' (*Gar* rap) 131, 163, 180, 198, 200, 204, 210
- *kasihun 'bitter' (*саšи*у гашуун) 131, 164, 184, 202, 207
- *kem 'fault' (gim гэм) 131, 164, 198, 204
- *ker 'house' (*gir* гэр) 119, 131, 164, 180, 198, 213
- *kerel 'light' (girəţ гэрэл) 131, 164, 198
- *kørehesyn 'beast' (goros гөрөөс) 131, 149, 164, 187, 195, 198
- *kurpan 'three' (*догэw* гурав) 108, 131, 164, 180, 189, 197–8, 200, 204
- *khaha 'to shut' (xaa xaa) 131, 164, 184
- *k^hahalka 'gate' (*хаађэс* хаалга) 122, 131, 164, 187
- *k^haič^hi 'scissors' (*хаič^h* хайч) 131, 164, 185, 214, 216
- *k^hakač^ha 'to leave' (*хадэс^h* хагац) 131, 164, 198
- *k^halahun 'warm' (*хађи*ј халуун) 131, 164, 184, 187, 198, 200, 204–5
- *k^hamt^hu 'together' (*хатt^h* хамт) 125, 132, 165, 204, 206–7
- *k^hara 'black' (*xar* xap) 132, 145, 165, 180, 189, 198, 200, 209
- *k^hari 'to return' (*хаг^і* харь) 132, 165, 189, 198, 211
- *k^hauč^hin 'old' (*хооё^hәŋ* хуучин) 132, 165, 184, 206–7
- *k^hauli 'law' (хоођ' хууль) 112, 132, 165
- *khečije 'when' (xice xэзээ) 132, 165, 184, 201
- *k^hei [']wind' (*xii* хий) 112, 119, 132, 165, 185
- *k^helen 'tongue' (*хіђ* хэл) 101, 132, 165, 180, 182, 189, 198, 210
- *k^hen 'who' (*xiŋ* хэн) 132, 165, 198, 204, 205
- *k^hetyn 'how many' (xit хэд) 132, 165, 189
- *k^hi 'to do' (*хіі* хий) 116-7, 132, 193

- *k^hičahar 'border' (*x^jacgar* хязгаар) 112, 118– 19, 123, 132, 165, 209
- *k^hič^hihe 'to strive' (*хič^hе* хичээ) 132, 165, 184
- *k^hilpar 'easy' (*х^jақрәг* хялбар) 28, 132, 166, 182, 195–6, 209
- *k^himusun 'nail' (*хотя* хумс) 132, 166, 195–6, 209–10
- *k^hit^hat 'Chinese' (*x^jat^hət* Хятад) 132, 166, 204, 206–7
- *k^hohasun 'empty' (*хээгэŋ* хоосон) 132, 166, 184
- *k^hoina 'back' (*хэіп* хойно) 112, 119, 132, 166, 185, 199
- *k^hojar 'two' (*xɔjər* xoëp) 112, 115, 119, 125, 132, 166, 200
- *khola 'far' (хэд хол) 132, 166, 195, 198-9
- *k^honin 'sheep' (*хэп^j* хонь) 132, 166, 189, 198– 9, 211, 216
- *k^høisyn 'navel' (xuis xyŭc) 112, 132, 166, 185
- *k^høit^hen 'cold' (*xuit^hәŋ* хүйтэн) 112. 132, 166, 185, 206
- *k^høk^he 'blue' (*xox* xøx) 133, 166, 180, 186, 188, 195, 206–7, 216
- *k^høl 'foot' (*хођ* хөл) 115, 133, 166, 180, 198–9, 210
- *k^hølesyn 'sweat' (*хођs* хөлс) 133, 167, 180, 204, 206
- *k^høŋken 'light' (*хоŋдәŋ* хөнгөн) 133, 167, 204
- * $k^{h}uč^{h}a$ 'to bark' (*xuch* xyII) 133, 167, 207
- *k^hui 'sheath' (*хоі* хуй) 133, 167, 185, 214
- *k^hulakai 'thief' (хођ*саі* хулгай) 133, 167, 185
- *k^hupi 'destiny' (хоw¹ хувь) 133, 167, 183, 211
- *k^hura 'rain' (*xur* xyp) 119, 133
- *k^huruhun 'finger' (*xuru* xypyy) 133, 167, 180, 184, 198, 200
- *k^hynty 'heavy' (*xunt* хүнд) 133, 167, 180, 189, 204
- *k^hyr 'to reach' (*xur* xyp) 133, 167, 180, 198
- *k^hyreken 'son-in-law' (*хигдәŋ* хүргэн) 133, 167, 180
- *maŋlai 'forehead' (*magnai* магнай) 133, 167, 204
- *masi 'very' (*таš* маш) 119, 133, 167, 202, 216
- *mete 'to know' (*mit* мэд) 115, 133, 168, 180, 189, 198
- *mik^han 'meat' (*max* мах) 133, 168, 189, 196, 198, 209–10
- *miŋk^han 'thousand' (*m^jаŋ*д мянга) 125, 133, 168, 182, 209–10
- *mokai 'snake' (*тэдэі* могой) 122, 134, 168, 195, 198
- *moŋkal 'Mongol' (*тэŋдэђ* Монгол) 115, 125, 134
- *morin 'horse' (*тэг*^{*i*}морь) 113, 119, 134, 142, 168, 189, 198, 209, 211, 214
- *morit 'horses' (*тог*^{*i*} морьд) 125, 134, 168, 204

- *motun 'tree' (mot мод) 113, 134, 168, 198, 214
- *mølsyn 'ice' (mos мөс) 101, 112, 134, 168, 204
- *mør 'path' (*mor* мөр) 134, 168, 180, 211, 213
- *nahat 'to play' (*naat* наад) 134, 168, 184, 198, 204
- *naiman 'eight' (*naim* найм) 112, 134, 168, 185, 198
- *napč^hin 'leaf' (*nawč^h* навч) 119, 125, 134, 168, 204
- *narin 'fine' (*nar^jiŋ* нарийн) 113, 134, 169, 198
- *nasun 'age' (*nas* нас) 134, 147, 169, 189
- *nehe 'to open' (nee нээ) 134, 169, 184
- *nek^he 'to weave' (*nix* нэх) 134, 169, 198
- *nek^hei 'sheepskin' (*nixi* нэхий) 112, 134, 169, 185, 198
- *nere 'name' (nir нэр) 112, 119, 134, 169, 180
- *nič^hykyn 'naked' (*nuc^hgəŋ* нүцгэн) 134, 169, 198
- *nihu 'to hide' (*поо* нуу) 61, 112, 134, 169, 184, 209
- *nilkha 'baby' (*пⁱаţx* нялх) 134, 169, 209
- *nimken 'thin' (*nimgəŋ* нимгэн) 125, 134, 169, 204
- *niruhun 'spine' (пого нуруу) 134, 169, 184, 196
- *nis 'to fly' (*nis* нис) 116, 134, 169, 180, 193, 204
- *nisi 'to beat' 116, 134
- *nityn 'eye' (*nut* нүд) 112, 116, 135, 170, 189, 192–3, 198, 210, 196
- *noir 'sleep' (*nэіг* нойр) 135, 170, 185, 187
- *nokahan 'green' (*пэдэŋ* ногоон) 135, 170, 195, 205
- *nok^hai 'dog' (*пэхэі* нохой) 115, 135, 170, 180, 195, 198
- *nøk^her 'friend' (*noxər* нөхөр) 115, 119, 135, 170, 180, 195, 198
- *oira 'near' (*sir* ойр) 112, 135, 170, 185, 187, 214
- *ol 'to find' (эђ ол) 135, 170, 180
- *olan 'many' (эдэŋ олон) 135, 170, 195, 197
- *ora 'to enter' (*>r* op) 115, 135, 170, 180, 195
- *øher 'self' (oor өөр) 123, 135
- *øk 'to give' (og өг) 135, 170, 180, 183, 204
- *øk^hin 'girl' (*эх^јәŋ* охин) 112, 135, 170, 188, 205, 208, 210
- *ønteken 'egg' (*ontəg* өндөг) 135, 170, 198, 204
- *øŋke 'colour' (*о*уд өнгө) 125, 135, 171, 195, 198, 204
- *pahu 'to go down' (puu буу) 135, 171, 198
- *pajan 'rich' (*pajəŋ* баян) 135, 171, 198
- *parahun 'right' (paroŋ баруун) 122, 135
- *pari 'to hold' (*parⁱ*барь) 135, 171, 189, 198, 209, 216
- *pars 'tiger' (par 6ap) 125, 135, 171, 208

- *pasa 'also' (pas 6ac) 135, 171, 208
- *pat^hu 'firm' (pat^h бат) 135, 171, 208
- *peje 'body' (pij бие) 135, 171, 198, 216
- *peri 'daughter-in-law, bride' (*pir* бэр) 113, 136, 171, 180, 189, 198
- *pič^hi 'to write' (*pič^h* бич) 116, 119, 136, 171, 180, 208
- *ріč^ьік 'letter' (*ріč^hәд* бичиг) 112, 125, 136, 171, 180, 204
- *pisilak 'cheese' (pⁱasţəg бяслаг) 136, 209
- *pos 'to rise' (*pэs* бос) 115, 125, 136, 171, 180, 204, 208
- *pøhere 'kidney' (poor бөөр) 136, 172, 184
- *pøhesyn 'louse' (*poos* бөөс) 136, 172, 184, 189, 208
- *pulak 'spring' (*риђад* булаг) 125, 136, 172, 183, 204
- *pyrin 'each' (pur бүр) 112, 136
- *pyse 'belt' (pus бүс) 136, 172, 208
- *saha 'to milk' (saa caa) 136, 172, 184
- *sahu 'to sit' (suu cyy) 136, 172, 184, 198, 202
- *sahurin 'base' (suur^j суурь) 136, 172, 214
- *sain 'good' (saiŋ сайн) 112, 136, 172, 185, 198
- *sakhal 'beard' (saxəb сахал) 136, 172, 209
- *sak^hi 'to guard' (*sax^j* сахь) 113, 136, 172, 207, 209
- *sam 'comb' (sam сам) 136, 172, 204
- *sara 'moon' (sar cap) 136, 145, 172, 198, 202
- *sehyl 'tail' (*suuţ* сүүл) 136, 173, 184, 187, 192, 202
- *setk^hil 'mind' (*sit^hgəţ* сэтгэл) 112, 117, 125, 136, 173, 204
- *sihe 'to piss' (šee шээ) 136, 173, 184, 202
- *sini 'new' (*šin* шинэ) 119, 136, 173, 180, 202
- *siŋk^hor 'falcon' (*šэŋхәт* шонхор) 112, 116, 136, 173, 195–6
- *sipahun 'bird' (*šuwu* шувуу) 136, 173, 196
- *sipar 'mud' (*šawər* шавар) 137, 173, 198
- *sira 'yellow' (*šar* шар) 137, 173, 189, 196, 198, 202
- *sirihe 'table' (šire ширээ) 137, 173, 184, 196
- *sityn 'tooth' (*šut* шүд) 137, 173, 188–9, 196, 198, 202
- *suhu 'armpit' (sug суга) 123, 137
- *sumun 'arrow' (*sum* сум) 137, 173, 183, 188
- *sur 'to learn' (sur cyp) 119, 137, 173, 180, 214
- *sykhe 'axe' (sux cyx) 137, 174, 180, 207
- *syme 'temple' (*sum* сүм) 137, 174, 183
- *tahari 'sore' (taar^j даарь) 137, 174, 214
- *tahun 'sound' (tuu дуу) 137, 174, 198
- *taru 'to press' (tar дар) 137, 174, 189
- *tehel 'gown' (teeţ дээл) 123, 137, 174
- *tehere 'above' (*teer* дээр) 119, 137, 174, 184, 189, 198
- *tehesyn 'rope' (tees дээс) 137, 174, 184

- *tehy 'younger brother' (tuu дүү) 137, 174, 184
- *tokhalaŋ 'lame' (*tэсţэŋ* доголон) 137, 174, 204
- *tolahan 'seven' (tэţэ долоо) 137, 174, 195
- *tot^hara 'inside' (*tэt^hәr* дотор) 137, 174, 207
- *tøč^hin 'forty' (*toč^h* дөч) 137, 175, 207
- *tørpen 'four' (*tогэw* дөрөв) 107, 112–13, 115, 119, 137, 175, 198, 204
- *tumta 'middle' (*tont* дунд) 112, 125, 137, 175, 197, 204
- *tyhyren 'full' (*tuurəŋ* дүүрэн) 125, 137, 175, 184, 204–5
- *t^hak^hi 'to offer' (t^hax^j тахь) 137, 175, 209
- *t^hak^hija 'hen' (*t^haxia* тахиа) 116, 137, 175, 184, 209
- *t^hani 'to recognize' (*t^han^j*тань) 138, 175, 198, 209, 211
- *thapin 'fifty' (thaw тавь) 65, 138, 175, 209, 211
- *t^hapun 'five' (*t*^h*aw* тав) 65, 113, 138, 175, 180, 189, 198
- *thatha 'to pull' (thath TAT) 138, 175, 188, 206-7
- *t^haulai 'hare' (*t^hооţаi* туулай) 77, 112, 119, 138, 175, 184–5, 198
- *themehen 'camel' (thime тэмээ) 138, 175, 184
- *t^hemyr 'iron' (t^homər төмөр) 138, 176, 195
- *thohan 'number' (thээ тоо) 138, 176, 184
- *t^hokahan 'kettle' (*t*^hэдэ тогоо) 138, 176, 195 *t^hosun 'fat' (*t*^hэз тос) 42, 112–13, 119, 138,
 - 142, 176, 180, 206

- *uhu 'to drink' (00 yy) 138, 176, 184
- *uila 'to cry' (viţ уйл) 112, 138, 176, 185
- *ukija 'to wash' (*uga* yraa) 138, 176, 180, 184
- *ulus 'state' (*иђs* улс) 126, 138
- *umart^ha 'to forget' (*mart^h* март) 138, 176, 188, 198, 204
- *untha 'to sleep' (Unth унт) 138, 176, 188, 197
- *unu 'to ride' (*оп* уна) 138, 176, 208
- *uri 'to invite' (*ог*^{*j*}урь) 138, 176, 211, 214
- *urita 'before' (*urit* урьд) 112, 116, 138
- *urt^hu 'long' (*urt^h* ypt) 125, 138, 176, 188–9, 204, 208
- *urus 'to flow' (*urs* ypc) 125, 138
- *usun 'water' (*us* yc) 112, 139, 177, 180, 189, 198, 202
- *yče 'to see' (uc y3) 139, 177, 180, 189, 198, 211
- *yčyhyr 'point' (*ucur* үзүүр) 139, 177, 184, 198
- *yile 'deed' (*ui*5 үйл) 112, 119, 126, 139, 177, 185, 197
- *yje 'joint' (uj ye) 77, 139, 177, 197-8
- *yke 'word' (ug yr) 112, 119, 139, 177, 204, 213
- *yk^hy 'to die' (*ux* yx) 112, 139, 177, 180, 188–9, 198, 208
- *ynijen 'cow' (*une* үнээ) 116, 139, 177, 184, 210
- *yntyr 'high' (*ontər* өндөр) 125, 139, 177, 183, 204
- *ytyr 'day' (*оtәr* өдөр) 139, 177, 183, 204